### SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

## AMENDMENT NO. 3 To Form 10

GENERAL FORM FOR REGISTRATION OF SECURITIES Pursuant to Section 12(b) or (g) of the Securities Exchange Act of 1934

## Mallinckrodt public limited company

(Exact name of Registrant as specified in its charter)

Ireland

(State or other jurisdiction of incorporation or organization)

98-1088325 (I.R.S. Employer Identification Number)

1st Floor, 20 On Hatch Lower Hatch Street, Dublin 2, Ireland (Address of principal executive offices)

+353 (1) 438-1700 (Registrant's telephone number, including area code)

Securities to be registered pursuant to Section 12(b) of the Act:

Title of Each Class to be so Registered Ordinary Shares, par value \$0.20 Name of Each Exchange on which Each Class is to be Registered

New York Stock Exchange

Securities to be registered pursuant to Section 12(g) of the Act: None

#### MALLINCKRODT PLC

#### INFORMATION REQUIRED IN REGISTRATION STATEMENT CROSS-REFERENCE SHEET BETWEEN INFORMATION STATEMENT AND ITEMS OF FORM 10

Certain information required to be included herein is incorporated by reference to specifically identified portions of the body of the information statement filed herewith as Exhibit 99.1. None of the information contained in the information statement shall be incorporated by reference herein or deemed to be a part hereof unless such information is specifically incorporated by reference.

#### Item 1. Business.

The information required by this item is contained under the sections of the information statement entitled "Information Statement Summary," "Risk Factors," "Cautionary Statement Concerning Forward-Looking Statements," "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Business," "Certain Relationships and Related Person Transactions," and "Where You Can Find More Information" and is incorporated herein by reference.

#### Item 1A. Risk Factors.

The information required by this item is contained under the section of the information statement entitled "Risk Factors" and is incorporated herein by reference.

#### Item 2. Financial Information.

The information required by this item is contained under the sections of the information statement entitled "Capitalization," "Unaudited Pro Forma Condensed Combined Financial Statements," "Selected Historical Combined Financial Data," and "Management's Discussion and Analysis of Financial Condition and Results of Operations" and is incorporated herein by reference.

#### Item 3. Properties.

The information required by this item is contained under the sections of the information statement entitled "Business—Manufacturing," "Business—Sales, Marketing and Distribution," and "Business—Properties" and is incorporated herein by reference.

#### Item 4. Security Ownership of Certain Beneficial Owners and Management.

The information required by this item is contained under the section of the information statement entitled "Security Ownership of Certain Beneficial Owners and Management" and is incorporated herein by reference.

#### Item 5. Directors and Executive Officers.

The information required by this item is contained under the section of the information statement entitled "Management" and is incorporated herein by reference.

#### Item 6. Executive Compensation.

The information required by this item is contained under the sections of the information statement entitled "Compensation Discussion and Analysis," "Executive Compensation," and "Certain Relationships and Related Person Transactions" and is incorporated herein by reference.

#### Item 7. Certain Relationships and Related Transactions.

The information required by this item is contained under the sections of the information statement entitled "Management" and "Certain Relationships and Related Person Transactions" and is incorporated herein by reference.

#### Item 8. Legal Proceedings.

The information required by this item is contained under the section of the information statement entitled "Business-Legal Proceedings" and is incorporated herein by reference.

#### Item 9. Market Price of, and Dividends on, the Registrant's Common Equity and Related Stockholder Matters.

The information required by this item is contained under the sections of the information statement entitled "Dividends," "Capitalization," "Executive Compensation," "The Separation," and "Description of Mallinckrodt's Share Capital" and is incorporated herein by reference.

#### Item 10. Recent Sales of Unregistered Securities.

The information required by this item is contained under the section of the information statement entitled "Description of Mallinckrodt's Share Capital— Sale of Unregistered Securities" and is incorporated herein by reference.

#### Item 11. Description of Registrant's Securities to be Registered.

The information required by this item is contained under the sections of the information statement entitled "Dividends," "The Separation," and "Description of Mallinckrodt's Share Capital" and is incorporated herein by reference.

#### Item 12. Indemnification of Directors and Officers.

The information required by this item is contained under the section of the information statement entitled "Description of Mallinckrodt's Share Capital— Limitations on Liability, Indemnification of Directors and Officers and Insurance" and is incorporated herein by reference.

#### Item 13. Financial Statements and Supplementary Data.

The information required by this item is contained under the section of the information statement entitled "Index to Combined Financial Statements" and the financial statements referenced therein and is incorporated herein by reference.

#### Item 14. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure.

None.

#### Item 15. Financial Statements and Exhibits.

#### (a) Financial Statements

The information required by this item is contained under the section of the information statement entitled "Index to Combined Financial Statements" and the financial statements referenced therein and is incorporated herein by reference.

#### (b) Exhibits

See below.

The following documents are filed as exhibits hereto:

Exhibit Number	Exhibit Description
2.1	Form of Separation and Distribution Agreement by and between Covidien plc and Mallinckrodt plc*
3.1	Form of Memorandum and Articles of Association of Mallinckrodt plc*
3.2	Certificate of Incorporation of Mallinckrodt plc**
4.1	Indenture, dated as of April 11, 2013, by and among Mallinckrodt International Finance S.A., Covidien International Finance S.A. and Deutsche Bank Trust Company Americas, as trustee**
4.2	Registration Rights Agreement, dated as of April 11, 2013, by and among Mallinckrodt International Finance S.A., Goldman, Sachs & Co., J.P. Morgan Securities LLC and the other purchasers named therein**
10.1	Form of Transition Services Agreement by and between Covidien plc and Mallinckrodt plc*
10.2	Form of Tax Matters Agreement by and between Covidien plc and Mallinckrodt plc*
10.3	Form of Employee Matters Agreement by and between Covidien plc and Mallinckrodt plc*
10.4	Credit Agreement, dated as of March 25, 2013, by and among Mallinckrodt International Finance S.A., JPMorgan Chase Bank, National Association, as administrative agent, and the other lenders and agents party thereto**
10.5	Letter Agreement, dated as of February 9, 2012, by and between Covidien plc and Mark Trudeau**†
10.6	Letter Agreement, dated as of August 1, 2011, by and between Covidien plc and Matthew K. Harbaugh**†
10.7	Letter Agreement, dated as of August 1, 2011, by and between Covidien plc and David E. Silver**†
10.8	Letter Agreement, dated as of August 1, 2011, by and between Covidien plc and Thomas E. Berry**
10.9	Letter Agreement, dated as of August 1, 2011, by and between Covidien plc and Peter G. Edwards**
10.10	Form of Mallinckrodt Pharmaceuticals Stock and Incentive Plan*
10.11	Form of Mallinckrodt Employee Stock Purchase Plan*
10.12	Form of Mallinckrodt Savings Related Share Plan*
10.13	Form of Deed of Indemnification by and between Mallinckrodt plc and Directors and Secretary**
10.14	Form of Indemnification Agreement by and between Mallinckrodt Brand Pharmaceuticals, Inc. and Directors and Secretary**
21.1	Subsidiaries of Mallinckrodt plc*
99.1	Information Statement of Mallinckrodt plc, preliminary and subject to completion, dated May 31, 2013*

\* Filed herewith.

\*\* Previously filed.

<sup>†</sup> Confidential treatment requested as to certain terms in this agreement; these terms have been omitted from this filing and filed separately with the Securities and Exchange Commission.

SIGNATURES

Pursuant to the requirements of Section 12 of the Securities Exchange Act of 1934, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized.

#### MALLINCKRODT PUBLIC LIMITED COMPANY

By: /s/ Mark Trudeau

Name: Mark Trudeau Title: Director

Date: May 31, 2013

SEPARATION AND DISTRIBUTION AGREEMENT BY AND BETWEEN COVIDIEN PLC AND MALLINCKRODT PLC DATED AS OF [—], 2013

ARTICLE I DEFINITIONS
-----------------------

#### AI

ARTICLE I DEFINITIONS ARTICLE II THE SEPARATION		Page 2
		15
2.1	Transfer of Assets and Assumption of Liabilities	15
2.2	Mallinckrodt Assets	17
2.3	Mallinckrodt Liabilities	20
2.4	Transfer of Excluded Assets; Assumption of Excluded Liabilities	22
2.5	Approvals and Notifications	22
2.6	Novation of Mallinekrodt Liabilities	25
2.7	Novation of Excluded Liabilities	25
2.8	Intercompany Agreements and Arrangements	26
2.9	Treatment of Shared Contracts	27
2.10	Bank Accounts; Cash Balances	29
2.11		29
2.12	6	29
2.13	1	29
2.14		30
2.15		30
2.16	Adjustment Amount	31
ARTICLE III THE DISTRIBUTION		32
3.1	The Distribution	32
3.2	Actions Prior to the Distribution	33
3.3	Conditions to Distribution	33
3.4	Certain Stockholder Matters	35
ARTICLE	IV MUTUAL RELEASES; INDEMNIFICATION	36
4.1	Release of Pre-Distribution Claims	36
4.2	Indemnification by Mallinckrodt	39
4.3	Indemnification by Covidien	39
4.4	Indemnification Obligations Net of Insurance Proceeds and Other Amounts	40
4.5	Procedures for Indemnification of Third-Party Claims	40
4.6	Additional Matters	41
4.7	Remedies Cumulative	42
4.8	Survival of Indemnities	43
4.9	Guarantees, Letters of Credit or Other Obligations	43
4.10		43
4.11		43
ARTICLE	V INSURANCE	44
5.1	Cooperation	44
5.2	Policies and Rights Included Within Assets	44
5.3	Claims Made Tail Policies	44

i

Claims Made Tail Policies 5.3

5.4	Occurrence Based Policies	46
5.5	Administration; Other Matters	46
5.6	Agreement for Waiver of Conflict and Shared Defense	48
ARTICLE	VI CERTAIN OTHER MATTERS	49
6.1	Late Payments	49
6.2	Grant of License for Mallinckrodt Name	49
ARTICLE	VII EXCHANGE OF INFORMATION; CONFIDENTIALITY	49
7.1	Agreement for Exchange of Information; Archives	49
7.2	Ownership of Information	50
7.3	Compensation for Providing Information	50
7.4	Record Retention	50
7.5	Limitations of Liability	50
7.6	Production of Witnesses; Records; Cooperation	50
7.7	Confidentiality	51
7.8	Protective Arrangements	52
ARTICLE	VIII DISPUTE RESOLUTION	53
8.1	Good Faith Negotiation	53
8.2	Mediation	53
8.3	Litigation	54
ARTICLE	IX FURTHER ASSURANCES AND ADDITIONAL COVENANTS	54
9.1	Further Assurances	54
ARTICLE	X TERMINATION	55
10.1	Termination	55
10.2	Effect of Termination	55
ARTICLE	XI MISCELLANEOUS	55
11.1	Counterparts; Entire Agreement; Corporate Power	55
11.2		56
11.3		57
11.4	Third-Party Beneficiaries	57
11.5	Notices	57
11.6		58
11.7		59
11.8	5	59
11.9	1	59
	0 Headings	59
11.11		59
11.12	2 Waivers of Default	59
11.13		59
	4 Amendments	60
	5 Interpretation	60
11.16	6 Attorney-Client Privilege	60

ii

# 11.17 Limitations of Liability11.18 Performance

#### SCHEDULES

Schedule 1.1	Intercompany Agreements
Schedule 1.2	Legacy Indebtedness
Schedule 1.3	Certain Hedging Arrangements
Schedule 1.4	Certain Mallinckrodt Policies
Schedule 1.5	Certain Products Liability Policies
Schedule 2.1(a)	Plan of Reorganization
Schedule 2.2(a)(i)	Certain Mallinckrodt Assets
Schedule 2.2(a)(ii)	Certain Mallinckrodt Subsidiaries
Schedule 2.2(b)(ii)	Certain Excluded Assets
Schedule 2.3(a)(iv)	Certain Divested Businesses
Schedule 2.3(a)(v)	Certain Actions
Schedule 2.3(b)(iii)	Certain Excluded Liabilities
Schedule 2.9(a)	Certain Shared Contracts
Schedule 2.12	Certain Covidien Controlled Actions
Schedule 2.16	Sample Closing Statement and Transaction Accounting Principles
Schedule 4.9(b)	Certain Guarantees
Schedule 5.5(e)	Certain Insurer Cost-Sharing Agreements
Schedule 5.5(f)	Old Colony Policies
Schedule 6.2	Grantees of License for Mallinckrodt Name

iii

#### SEPARATION AND DISTRIBUTION AGREEMENT

This SEPARATION AND DISTRIBUTION AGREEMENT, dated as of [--], 2013 (this "<u>Agreement</u>"), is by and between Covidien plc, an Irish public limited company ("<u>Covidien</u>"), and Mallinckrodt plc, an Irish public limited company ("<u>Mallinckrodt</u>"). Mallinckrodt and Covidien are referred to together as the "<u>Parties</u>" and individually as a "<u>Party</u>." Capitalized terms used herein shall have the respective meanings assigned to them in <u>Article I</u> or elsewhere in this Agreement.

#### RECITALS

WHEREAS, Covidien currently owns and operates both the Covidien Business and the Mallinckrodt Business;

WHEREAS, the board of directors of Covidien (the "Covidien Board") has determined that it is in the best interests of Covidien and its shareholders that the Mallinckrodt Business be operated by a newly incorporated publicly traded company;

WHEREAS, Mallinckrodt has been incorporated for these purposes and has not engaged in activities except those incidental to its formation and in preparation for the transactions described herein;

WHEREAS, in furtherance of the foregoing, the Covidien Board and the board of directors of Mallinckrodt (the "<u>Mallinckrodt Board</u>") have determined that it is appropriate and desirable for Covidien and its applicable Subsidiaries to transfer the Mallinckrodt Assets to Mallinckrodt and certain entities designated by Mallinckrodt that will be Subsidiaries of Mallinckrodt as of the Distribution Date (any such entities, the "<u>Mallinckrodt Designees</u>"), and for Mallinckrodt and the Mallinckrodt Designees to assume the Mallinckrodt Liabilities, in each case as more fully described in this Agreement and the Ancillary Agreements and including the steps set forth in the Plan of Reorganization (the "<u>Separation</u>");

WHEREAS, Covidien currently intends that, on the Distribution Date, it will make a distribution in specie of the Mallinckrodt Business to the holders of Covidien Ordinary Shares on the Record Date ("Qualifying Covidien Shareholders"), effected by (i) the transfer of Covidien's entire legal and beneficial interest in the issued share capital of the Mallinckrodt Holding Companies to Mallinckrodt; and (ii) Mallinckrodt issuing Mallinckrodt Ordinary Shares directly to Qualifying Covidien Shareholders on a pro-rata basis in return, as more fully described in this Agreement and the Ancillary Agreements (the "Distribution");

WHEREAS, the Distribution and certain related transactions, taken together, are intended to qualify as a reorganization under Section 368 of the Code for U.S. federal income tax purposes and under various reorganization provisions contained in Irish tax law;

WHEREAS, this Agreement is intended to be a "plan of reorganization" within the meaning of Treasury Regulation Section 1.368-2(g); and

WHEREAS, each of Covidien and Mallinckrodt has determined that it is appropriate and desirable to set forth the principal corporate transactions required to effect the Separation and the Distribution and to set forth certain other agreements that shall govern certain matters relating to the Separation and the Distribution and the relationship of Covidien, Mallinckrodt and their respective Subsidiaries following the Distribution.

NOW, THEREFORE, in consideration of the mutual agreements, provisions and covenants contained in this Agreement, and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

#### ARTICLE I DEFINITIONS

For purposes of this Agreement, the following terms shall have the following meanings:

"<u>Action</u>" shall mean any demand, action, claim, dispute, suit, countersuit, arbitration, settlement, inquiry, subpoena, proceeding or investigation of any nature (whether criminal, civil, legislative, administrative, regulatory, prosecutorial or otherwise) by or before any federal, state, local, foreign or international Governmental Authority or any arbitration or mediation tribunal.

"<u>Adjustment Amount</u>" shall mean an amount, which may be positive or negative, equal to (a) Specified Working Capital as of immediately after the Distribution, <u>less</u> (b) Specified Indebtedness as of immediately after the Distribution, <u>plus</u> (c) the aggregate amount paid by any member of the Covidien Group or the Mallinekrodt Group in respect of Capital Expenditures in the period beginning on October 1, 2012 and ending immediately after the Distribution.

"<u>Affiliate</u>" (including with a correlative meaning, "<u>affiliated</u>") shall mean, when used with respect to a specified Person, a Person that, directly or indirectly, through one (1) or more intermediaries, controls, is controlled by or is under common control with such specified Person. For the purpose of this definition, "<u>control</u>" (including with correlative meanings, "<u>controlled by</u>" and "<u>under common control with</u>"), when used with respect to any specified Person shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or other interests, by contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment, undertaking or otherwise. It is expressly agreed that, prior to, on and after the Distribution Date, for purposes of this Agreement and the Ancillary Agreements, (1) no member of the Mallinckrodt Group shall be deemed to be an Affiliate of any member of the Covidien Group and (2) no member of the Covidien Group shall be deemed to be an Affiliate of any member of doubt, after the Effective Time, the members of the Covidien Group and the members of the Mallinckrodt Group shall not be deemed to be under common control for purposes hereof due solely to the fact that Covidien and Mallinckrodt may have common shareholders.

"Agent" shall mean Computershare Trust Company, N.A., or such other trust company or bank duly appointed by Covidien to act as distribution agent, transfer agent and/or registrar for the Mallinckrodt Ordinary Shares in connection with the Distribution.

"Agreement" shall have the meaning set forth in the Preamble.

"Ancillary Agreement" shall mean the Transition Services Agreement, the Tax Matters Agreement, the Employee Matters Agreement, the Intercompany Agreements and the Transfer Documents.

"<u>Approvals or Notifications</u>" shall mean any consents, waivers, approvals, permits or authorizations to be obtained from, notices, registrations or reports to be submitted to, or other filings to be made with, any third Person, including any Governmental Authority.

"<u>Assets</u>" shall mean, with respect to any Person, the assets, properties, claims and rights (including goodwill) of such Person, wherever located (including in the possession of vendors or other third Persons or elsewhere), of every kind, character and description, whether real, personal or mixed, tangible, intangible or contingent, in each case whether or not recorded or reflected or required to be recorded or reflected on the books and records or financial statements of such Person, including the following:

(a) all accounting and other books, records and files whether in paper, microfilm, microfiche, computer tape or disc, magnetic tape, electronic or any other form;

(b) all apparatus, computers and other electronic data processing and communications equipment, fixtures, machinery, equipment, furniture, office equipment, automobiles, trucks, vessels, motor vehicles and other transportation equipment and other tangible personal property;

(c) all inventories of materials, parts, raw materials, components, supplies, works-in-process and finished goods and products;

(d) all interests in real property of whatever nature, including easements, whether as owner, mortgagee or holder of a Security Interest in real property, lessor, sublessor, lessee, sublessee or otherwise;

(e) (i) all interests in any capital stock or other equity interests of any Subsidiary, Affiliate or any other Person, (ii) all bonds, notes, debentures or other securities issued by any Subsidiary, Affiliate or any other Person, (iii) all loans, advances or other extensions of credit or capital contributions to any Subsidiary, Affiliate or any other investments in securities of any Person;

(f) all license agreements, leases of personal property, open purchase orders for raw materials, supplies, parts or services and other contracts, agreements or commitments;

(g) all deposits, letters of credit and performance and surety bonds;

(h) all written (including in electronic form) or oral technical information, data, specifications, research and development information, engineering drawings and specifications, operating and maintenance manuals, and materials and analyses prepared by consultants and other third Persons;

(i) all Intellectual Property and Technology;

(j) all Software;

(k) all cost information, sales and pricing data, customer prospect lists, supplier records, customer and supplier lists, customer and vendor data, correspondence and lists, product data and literature, artwork, design, formulations and specifications, quality records and reports and other books, records, studies, surveys, reports, plans and documents;

(l) all prepaid expenses, trade accounts and other accounts and notes receivable;

(m) all rights under insurance policies and all rights in the nature of insurance, indemnification or contribution;

(n) all rights under contracts, consent decrees, orders or agreements, all claims or rights against any Person arising from the ownership of any Asset, all rights in connection with any bids or offers and all claims, choses in action or similar rights, whether accrued or contingent;

(o) all licenses, permits, approvals and authorizations that have been issued by any Governmental Authority;

(p) all cash or cash equivalents, bank accounts, lock boxes and other deposit arrangements; and

(q) all interest rate, currency, commodity or other swap, collar, cap or other hedging or similar agreements or arrangements.

"Balance Sheet Date" shall mean March 29, 2013.

"Business Day" shall mean any day that is not a Saturday, a Sunday or other day that is a statutory holiday under the federal Laws of the United States. In the event that any action is required or permitted to be taken under this Agreement on or by a date that is not a Business Day, such action may be taken on or by the Business Day immediately following such date.

"Capital Expenditures" shall have the meaning set forth in Schedule 2.16.

"<u>Claims Administration</u>" shall mean the processing of claims made under the Shared Policies and Mallinckrodt Policies, including the reporting of losses or claims to the insurance carriers and management and defense of claims, including the right to exhaust, settle, release, commute, buy-back or otherwise resolve disputes with respect to any such claims.

4

"Closing Statement" shall have the meaning set forth in Section 2.16(b).

"Code" shall mean the Internal Revenue Code of 1986, as amended, and the regulations promulgated thereunder.

"Confidential Information" shall have the meaning set forth in Section 7.7(a).

"Covidien" shall have the meaning set forth in the Preamble.

"Covidien Accounts" shall have the meaning set forth in Section 2.10(a).

"Covidien Board" shall have the meaning set forth in the Recitals.

"Covidien Business" shall mean the businesses and operations of the Covidien Group other than the Mallinckrodt Business.

"Covidien Group" shall mean Covidien, each Subsidiary of Covidien and each other Person that is controlled directly or indirectly by Covidien (in each case other than any member of the Mallinckrodt Group).

"Covidien Indemnitees" shall have the meaning set forth in Section 4.2.

"<u>Covidien Intellectual Property</u>" shall mean (i) the Covidien Name and Covidien Marks and (ii) all other Intellectual Property that is owned or licensed by any member of the Covidien Group or the Mallinekrodt Group, other than the Mallinekrodt Intellectual Property.

"<u>Covidien Name and Covidien Marks</u>" shall mean the names, marks, trade dress, logos, monograms, domain names and other source or business identifiers of Covidien or any of its Affiliates using or containing "Covidien" (in block letters or otherwise), "Covidien" either alone or in combination with other words or elements and all names, marks, trade dress, logos, monograms, domain names and other source or business identifiers confusingly similar to or embodying any of the foregoing either alone or in combination with other words or elements, together with the goodwill associated with any of the foregoing.

"Covidien Ordinary Shares" shall mean the ordinary shares, par value \$0.20 per share, of Covidien.

"<u>Covidien Software</u>" shall mean all Software that is owned or licensed by any member of the Covidien Group or the Mallinckrodt Group, other than the Mallinckrodt Software.

"Covidien Technology" shall mean all Technology that is owned or licensed by any member of the Covidien Group or the Mallinckrodt Group, other than the Mallinckrodt Technology.

"Covidien Transfer Documents" shall have the meaning set forth in Section 2.1(b).

"CPR" shall have the meaning set forth in Section 8.2.

"Credit Facility" shall mean the Credit Agreement, dated as of March 25, 2013, by and among MIFSA, as borrower, from the Distribution Date, Mallinckrodt, as guarantor, the lenders party thereto from time to time and JPMorgan Chase Bank, National Association, as administrative agent.

"Disclosure Document" shall mean any registration statement (including the Form 10) filed with the SEC by or on behalf of any Party or any of its controlled Affiliates, and also includes any information statement (including the Information Statement), prospectus, offering memorandum (including the offering memorandum in connection with the offering of Senior Notes), offering circular, periodic report or similar disclosure document, whether or not filed with the SEC or any other Governmental Authority, in each case which describes the Separation or the Distribution or the Mallinckrodt Group or primarily relates to the transactions contemplated hereby.

"Dispute" shall have the meaning set forth in Section 8.1.

"Dispute Notice" shall have the meaning set forth in Section 2.16(c).

"Dispute Resolution Period" shall have the meaning set forth in Section 2.16(c).

"Distribution" shall have the meaning set forth in the Recitals.

"Distribution Date" shall mean the date of the consummation of the Distribution, which shall be determined by Covidien in its sole discretion.

"Distribution Ratio" shall mean a fraction the numerator of which shall be one (1) and the denominator of which shall be eight (8).

"D&O Tail Policies" shall have the meaning set forth in Section 5.3(b).

"Effective Time" shall mean the time at which the Distribution occurs on the Distribution Date, which shall be deemed to be 6:59 p.m., New York City time, on the Distribution Date, or such other time as Covidien may determine.

"Employee Matters Agreement" shall mean the Employee Matters Agreement, dated as of the date hereof, by and between Covidien and Mallinckrodt, as such Employee Matters Agreement may be amended from time to time.

"Environmental Law" shall mean any Law relating to pollution, protection or restoration of or prevention of harm to the environment or natural resources, including the use, handling, transportation, treatment, storage, disposal, Release or discharge of Hazardous Materials or the protection of or prevention of harm to human health and safety.

"Environmental Liabilities" shall mean all Liabilities relating to, arising out of or resulting from any Hazardous Materials, Environmental Law or contract or agreement relating to environmental, health or safety matters (including all removal, remediation or cleanup costs, investigatory costs, response costs, natural resources damages, equipment upgrades or replacements, asbestos survey and removal costs, property damages, personal injury damages, costs of compliance, including with any product take back requirements, or with any settlement, judgment or other determination of Liability and indemnity, contribution or similar obligations) and all costs and expenses, interest, fines, penalties or other monetary sanctions in connection therewith.

"Exchange Act" shall mean the U.S. Securities Exchange Act of 1934, as amended, together with the rules and regulations promulgated thereunder.

"Excluded Assets" shall have the meaning set forth in Section 2.2(b).

"Excluded Liabilities" shall have the meaning set forth in Section 2.3(b).

"Fiduciary Tail Policies" shall have the meaning set forth in Section 5.3(c).

"Force Majeure" shall have the meaning set forth in Section 11.7.

"Form 10" shall mean the registration statement on Form 10 filed by Mallinckrodt with the SEC to effect the registration of Mallinckrodt Ordinary Shares pursuant to the Exchange Act in connection with the Distribution, as such registration statement may be amended or supplemented from time to time prior to the Effective Time.

"GAAP" means United States generally accepted accounting principles, consistently applied.

"Governmental Approvals" shall mean any notices, reports or other filings to be made, or any consents, registrations, approvals, permits or authorizations to be obtained from, any Governmental Authority.

"Governmental Authority" shall mean any nation or government, any state, municipality or other political subdivision thereof, and any entity, body, agency, commission, department, board, bureau, court, tribunal or other instrumentality, whether federal, state, local, domestic, foreign or multinational, exercising executive, legislative, judicial, regulatory, administrative or other similar functions of, or pertaining to, government and any executive official thereof.

"Group" shall mean either the Mallinckrodt Group or the Covidien Group, as the context requires.

"Guarantee Release" shall have the meaning set forth in Section 4.9(b).

"Hazardous Materials" shall mean any chemical, radiological isotope, material, substance, waste, pollutant, emission, discharge, release or contaminant that could result in liability under, or that is prohibited, limited or regulated by or pursuant to, any Environmental Law, and any natural or artificial substance (whether solid, liquid or gas, noise, ion, vapor or electromagnetic) that could cause harm to human health or the environment, including petroleum, petroleum products and byproducts, asbestos and asbestos-containing materials, urea formaldehyde foam insulation, electronic, medical or infectious wastes, polychlorinated biphenyls, radon gas, radioactive substances, chlorofluorocarbons and all other ozone-depleting substances.

"Indemnifying Party" shall have the meaning set forth in Section 4.4(a).

"Indemnitee" shall have the meaning set forth in Section 4.4(a).

"Indemnity Payment" shall have the meaning set forth in Section 4.4(a).

"Independent Accounting Firm" shall have the meaning set forth in Section 2.16(c).

"Information" shall mean information, whether or not patentable or copyrightable, in written, oral, electronic or other tangible or intangible forms, stored in any medium, including studies, reports, records, books, contracts, instruments, surveys, discoveries, ideas, concepts, know-how, techniques, designs, specifications, drawings, blueprints, diagrams, models, prototypes, samples, flow charts, data, computer data, disks, diskettes, tapes, computer programs or other software, marketing plans, customer names, communications by or to attorneys (including attorney-client privileged communications), memos and other materials prepared by attorneys or under their direction (including attorney work product), and other technical, financial, employee or business information or data.

"Information Statement" shall mean the information statement to be sent to each holder of Covidien Ordinary Shares in connection with the Distribution, as filed with the SEC, as such information statement may be amended or supplemented from time to time prior to the Effective Time.

"Initial Share Capital" shall mean all of the shares in the capital of Mallinckrodt issued and outstanding as of immediately prior to the consummation of the Distribution, which consists of seven Mallinckrodt Ordinary Shares and 40,000 ordinary A shares, par value  $\in 1.00$  per share, of Mallinckrodt.

"Insurance Administration" shall mean, with respect to each Shared Policy and Mallinckrodt Policy, the accounting for premiums, retrospectively-rated premiums, defense costs, indemnity payments, deductibles and retentions, as appropriate, under the terms and conditions of each of the Shared Policies and Mallinckrodt Policies; discussions or negotiations with insurers and the control of any Actions relating to such Shared Policy or Mallinckrodt Policy; the reporting to excess insurance carriers of any losses or claims which may cause the per-occurrence, per claim or aggregate limits of any Shared Policy to be exceeded; and the distribution of Insurance Proceeds as contemplated by this Agreement.

"Insurance Proceeds" shall mean those monies (i) received by an insured from an insurance carrier, including due to premium adjustments, whether or not retrospectively rated, or

(ii) paid by an insurance carrier on behalf of an insured, in either case net of any applicable premium deductible or self-insured retention. For the avoidance of doubt, "Insurance Proceeds" shall not include any costs or expenses incurred by a Party in pursuing insurance coverage.

"Insured Claims" shall mean those Liabilities that, individually or in the aggregate, are covered within the terms and conditions of any of the Shared Policies, whether or not subject to deductibles, co-insurance, self-insured retentions, or uncollectibility due to insure insolvency.

"Intellectual Property" shall mean all of the following whether arising under the Laws of the United States or of any other foreign or multinational jurisdiction: (i) patents, patent applications (including patents issued thereon) and statutory invention registrations, including reissues, divisions, continuations, continuations in part, substitutions, renewals, extensions and reexaminations of any of the foregoing, and all rights in any of the foregoing provided by international treaties or conventions, (ii) trademarks, service marks, trade names, service names, trade dress, logos and other source or business identifiers, including all goodwill associated with any of the foregoing, and any and all common law rights in and to any of the foregoing, registrations and applications for registration of any of the foregoing, all rights in and to any of the foregoing provided by international treaties or conventions, (iii) Internet domain names, (iv) copyrightable works, copyrights, moral rights, mask work rights, database rights and design rights, in each case, other than Software, whether or not registered, and all registrations and applications for registration of any of the foregoing provided by international treaties or convention, including trade secrets, invention disclosures, processes and know-how, in each case, other than Software, and (vi) intellectual property rights arising from or in respect of any Technology.

"Intercompany Agreements" shall mean the agreements listed on Schedule 1.1.

"Intercompany Balances" shall mean the intercompany accounts receivable and accounts payable between any member of the Covidien Group, on the one hand, and any member of the Mallinckrodt Group, on the other hand.

"IRS" shall mean the United States Internal Revenue Service.

"IRS Ruling" shall have the meaning set forth in Section 3.3(a)(i).

"Law" shall mean any national, supranational, federal, state, provincial, local or similar law (including common law), statute, code, order, ordinance, rule, regulation, treaty (including any income tax treaty), license, permit, authorization, approval, consent, decree, injunction, binding judicial or administrative interpretation or other requirement, in each case, enacted, promulgated, issued or entered by a Governmental Authority.

"Legacy Indebtedness" shall mean the indebtedness listed on Schedule 1.2.

"Liabilities" shall mean any and all debts, guarantees, assurances, commitments, liabilities, responsibilities, Losses, Taxes, remediation, deficiencies, reimbursement obligations in respect of letters of credit, damages, fines, penalties, settlements, sanctions, costs, expenses, interest and obligations of any nature or description, whether accrued or fixed, absolute or contingent, matured or unmatured, accrued or not accrued, asserted or unasserted, liquidated or

unliquidated, foreseen or unforeseen, known or unknown, reserved or unreserved, or determined or determinable, including those arising under any Law, claim (including any Third-Party Claim), demand, Action, or order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority or arbitration tribunal, and those arising under any contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment or undertaking, or any fines, damages or equitable relief that is imposed, in each case, including all costs and expenses relating thereto.

"linked" shall have the meaning set forth in Section 2.10(a).

"Losses" shall mean actual losses (including any diminution in value), costs, damages, penalties, Taxes and expenses (including legal and accounting fees and expenses and costs of investigation and litigation), whether or not involving a Third-Party Claim.

"Mallinckrodt" shall have the meaning set forth in the Preamble.

"Mallinckrodt Accounts" shall have the meaning set forth in Section 2.10(a).

"Mallinckrodt Assets" shall have the meaning set forth in Section 2.2(a).

"<u>Mallinckrodt Balance Sheet</u>" shall mean the unaudited pro forma balance sheet of the Mallinckrodt Business, as of the Balance Sheet Date, including the notes thereto, as reflected in the Form 10.

"Mallinckrodt Board" shall have the meaning set forth in the Recitals.

"<u>Mallinckrodt Business</u>" shall mean: (a) (i) the business and operations of the Pharmaceuticals Business and (ii) such other businesses and operations relating thereto carried on by the Pharmaceuticals Business, (b) except as otherwise expressly provided herein, any terminated, divested or discontinued businesses or operations that at the time of termination, divestiture or discontinuation primarily related to the Mallinckrodt Business (as described in the foregoing clause (a)) as then conducted and (c) the business and operations of Mallinckrodt Inc., a New York corporation, any Person that was a Subsidiary thereof as of the acquisition thereof by Tyco International Ltd. on October 17, 2000 and any predecessor-in-interest or successor-in-interest to any of the foregoing, as such business and operations were conducted at any time prior to or after such acquisition (whether or not any such business and operations or Subsidiary was terminated, divested or discontinued (as applicable) by Mallinckrodt Inc., Tyco International Ltd. or Covidien prior to the date hereof), excluding, in the case of each of clauses (a) through (c), the businesses and operations primarily related to the Excluded Assets.

"Mallinckrodt Cash" shall have the meaning set forth in Section 2.2(a)(vii).

"<u>Mallinckrodt Contracts</u>" shall mean the following contracts and agreements to which Covidien or any of its Subsidiaries is a party or by which it or any of its Subsidiaries or any of their respective Assets is bound, whether or not in writing, in each case immediately prior to the Distribution (including, for the avoidance of doubt, any Person that will be a member of the Mallinckrodt Group at the time of the Distribution), except for any such contract or agreement that is contemplated to be retained by Covidien or any member of the Covidien Group pursuant to any provision of this Agreement or any Ancillary Agreement:

(a) any customer, distribution, supply or vendor contracts or agreements entered into prior to the Effective Time that relate exclusively to the Mallinckrodt Business;

(b) any contract or agreement entered into in the name of, or expressly on behalf of, any division, business unit or member of the Mallinckrodt Group;

(c) any joint venture agreement or, subject to Section 2.14, any license agreement that relates primarily to the Mallinckrodt Business;

(d) any guarantee, indemnity, representation, warranty or other Liability of any member of the Mallinckrodt Group or the Covidien Group in respect of any other Mallinckrodt Contract, any Mallinckrodt Liability or the Mallinckrodt Business;

(e) any employment, change of control, retention, consulting, indemnification, termination, severance or other similar agreements with any Mallinckrodt Group Employee or consultants of the Mallinckrodt Group that are in effect as of the Distribution Date;

(f) any consent order, decree or agreement with any third party including but not limited to Governmental Authorities entered into in the name of, or expressly on behalf of, any division, business unit or member of the Mallinckrodt Group;

(g) any contract or agreement that is otherwise expressly contemplated pursuant to this Agreement or any of the Ancillary Agreements to be assigned to Mallinckrodt or any member of the Mallinckrodt Group; and

(h) any interest rate, currency, commodity or other swap, collar, cap or other hedging or similar agreements or arrangements entered into by or on behalf of any member of the Mallinckrodt Group, including the hedging arrangements listed on <u>Schedule 1.3</u>.

"Mallinckrodt Designees" shall have the meaning set forth in the Recitals.

"Mallinckrodt Employee" shall have the meaning set forth in the Employee Matters Agreement.

"Mallinckrodt Financing Arrangements" shall mean the Senior Notes and the Credit Facility.

<sup>11</sup> 

"Mallinckrodt Group" shall mean Mallinckrodt, each Subsidiary of Mallinckrodt and each other Person that is controlled directly or indirectly by Mallinckrodt.

"Mallinckrodt Holding Companies" means MIFSA and Mallinckrodt Belgium BVBA;

"Mallinckrodt Indemnitees" shall have the meaning set forth in Section 4.3.

"<u>Mallinckrodt Intellectual Property</u>" shall mean (a) all patents, patent applications, statutory invention registrations, registered trademarks, registered service marks, registered Internet domain names and copyright registrations (collectively, "<u>Registrable IP</u>") that are owned exclusively by any member of the Mallinckrodt Group at or prior to the Distribution Date, excluding any such Registrable IP that has been assigned by any member of the Mallinckrodt Group to any member of the Covidien Group prior to the Distribution Date, and (b) all Intellectual Property, other than Registrable IP, that is owned by any member of the Covidien Group or Mallinckrodt Group and that is used or held for use primarily in the Mallinckrodt Business as of the Distribution Date.

"Mallinckrodt Liabilities" shall have the meaning set forth in Section 2.3(a).

"Mallinckrodt Lines of Credit" shall mean any third-party line of credit in favor of any member of the Mallinckrodt Group.

"Mallinckrodt Ordinary Shares" shall mean the ordinary shares, par value \$0.20 per share, of Mallinckrodt.

"<u>Mallinckrodt Policies</u>" shall mean all Policies in the name of Mallinckrodt Inc., a New York corporation, any Subsidiary thereof and any predecessor-ininterest to any of the foregoing, in each case as of October 17, 2000, including the Policies set forth on <u>Schedule 1.4</u>.

"Mallinckrodt Software" shall mean all Software owned or licensed by any member of the Covidien Group or Mallinckrodt Group and that is primarily used or held for use in the Mallinckrodt Business as of the Distribution Date.

"<u>Mallinckrodt Spin Shares</u>" shall mean those Mallinckrodt Ordinary Shares to be issued, with effect from the Effective Time, to Qualifying Covidien Shareholders pursuant to the Distribution and in accordance with <u>Section 3.4(b)</u>;

"<u>Mallinckrodt Technology</u>" shall mean all Technology owned or licensed by any member of the Covidien Group or Mallinckrodt Group and that is primarily used or held for use in the Mallinckrodt Business as of the Distribution Date.

"Mallinckrodt Transfer Documents" shall have the meaning set forth in Section 2.4(b).

"Mediation Request" shall have the meaning set forth in Section 8.2.

"MIFSA" shall mean Mallinckrodt International Finance S.A., a Luxembourg company.

"<u>NYSE</u>" shall mean the New York Stock Exchange.

"Parties" or "Party" shall have the meaning set forth in the Preamble.

"Person" shall mean an individual, a general or limited partnership, a corporation, a trust, a joint venture, an unincorporated organization, a limited liability entity, any other entity or any Governmental Authority.

"<u>Pharmaceuticals Business</u>" shall mean the pharmaceuticals business segment of Covidien described in Covidien's Annual Report on Form 10-K for the period ended September 28, 2012, which business develops, manufactures and distributes specialty pharmaceuticals, active pharmaceutical ingredients, contrast products and radiopharmaceuticals.

"Plan of Reorganization" shall have the meaning set forth in Section 2.1(a).

"Policies" shall mean insurance policies and insurance contracts of any kind (other than life and benefits policies or contracts), including primary, excess and umbrella policies, comprehensive general liability policies, director and officer liability, fiduciary liability, automobile, aircraft, marine, property and casualty, workers' compensation and employee dishonesty insurance policies, bonds and self-insurance and captive insurance company arrangements, together with the rights, benefits and privileges thereunder.

"Prime Rate" shall mean the rate that Citibank, N.A. (or any successor thereto or other major money center commercial bank agreed to by the Parties) announces from time to time as its prime lending rate, as in effect from time to time.

"Procedure" shall have the meaning set forth in Section 8.2.

"Qualifying Covidien Shareholder" shall have the meaning set forth in the Recitals.

"Record Date" shall mean the close of business on June 19, 2013 or the close of business on another date if determined by the Covidien Board as the record date for determining holders of Covidien Ordinary Shares entitled to receive Mallinckrodt Ordinary Shares pursuant to the Distribution.

"Registrable IP" shall have the meaning set forth in the definition of Mallinckrodt Intellectual Property.

"<u>Release</u>" shall mean any release, spill, emission, discharge, leaking, pumping, pouring, dumping, injection, deposit, disposal, dispersal, leaching or migration of Hazardous Materials into the environment (including ambient air, surface water, groundwater and surface or subsurface strata).

"Reorganization Agreement" means any contract, agreement, arrangement, commitment, understanding, instrument, loan note, security, transfer document, or other document executed or presented for the purposes of, in relation to or arising from, the implementation of the Plan of Reorganization.

"Representatives" shall mean, with respect to any Person, any of such Person's directors, officers, employees, agents, consultants, advisors, accountants, attorneys or other representatives.

"<u>Respiratory Business</u>" shall mean (1) the Oximetry and Monitoring Products business line of Covidien, which business line develops, manufactures and distributes sensors, monitors and temperature management products, and (2) the Airway and Ventilation Products business line of Covidien, which business line develops, manufactures and distributes airway, ventilator, breathing systems and inhalation therapy products, in each of cases (1) and (2) as described in Covidien's Annual Report on Form 10-K for the period ended September 28, 2012 and any terminated, divested or discontinued businesses or operations that at the time of termination, divestiture or discontinuation primarily related to any of the foregoing.

"Sample Closing Statement" shall have the meaning set forth in Section 2.15(a).

"SEC" shall mean the U.S. Securities and Exchange Commission.

"Security Interest" shall mean any mortgage, security interest, pledge, lien, charge, claim, option, right to acquire, voting or other restriction, right-of-way, covenant, condition, easement, encroachment, restriction on transfer or other encumbrance of any nature whatsoever.

"Senior Notes" shall mean the 3.500% Senior Notes due 2018 and the 4.750% Senior Notes due 2023 issued by MIFSA under an indenture dated April 11, 2013.

"Separation" shall have the meaning set forth in the Recitals.

"Shared Contract" shall have the meaning set forth in Section 2.9(a).

"<u>Shared Policies</u>" shall mean all Policies, current or past, which are owned or maintained by or on behalf of Covidien or any of its Subsidiaries which relate to the Covidien Business or the Mallinckrodt Business, other than the Mallinckrodt Policies but including Mallinckrodt Policies to the extent they relate to workers compensation; provided that any products liability Policy shall not be a Shared Policy hereunder (and will be deemed to be an Excluded Asset hereunder), except for the products liability Policies listed on Schedule 1.5, which shall be treated as Shared Policies hereunder.

"Specified Indebtedness" means: (a) the Senior Notes, (b) the Legacy Indebtedness and (c) the Mallinckrodt Lines of Credit (to the extent drawn upon).

"Specified Working Capital" shall have the meaning set forth in Schedule 2.16.

"Software" shall mean any and all (i) computer programs, including any and all software implementation of algorithms, models and methodologies, whether in source code, object code, human readable form or other form, (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (iii) descriptions, flow charts and other work products used to design, plan, organize and develop any of the foregoing, (iv) screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons, and (v) documentation, including user manuals and other training documentation, relating to any of the foregoing.

"Subsidiary" shall mean, with respect to any Person, any corporation, limited liability company, joint venture or partnership of which such Person (i) beneficially owns, either directly or indirectly, more than fifty percent (50%) of (A) the total combined voting power of all classes of voting securities of such Person, (B) the total combined equity interests or (C) the capital or profit interests, in the case of a partnership, or (ii) otherwise has the power to vote, either directly or indirectly, sufficient securities to elect a majority of the board of directors or similar governing body.

"<u>Target Accounts Payable</u>" shall mean the product of (a) the cost of goods sold of the Mallinekrodt Business for the fiscal quarter ending on June 28, 2013 (determined in a manner consistent with GAAP), multiplied by (b) 37, divided by (c) 90.

"Target Accounts Receivable" shall mean the product of (a) the sales of the Mallinckrodt Business for the fiscal quarter ending on June 28, 2013 (determined in a manner consistent with GAAP), multiplied by (b) 52.1, divided by (c) 90.

"Target Adjustment Amount" shall mean an amount, which may be positive or negative, equal to (a) Target Accounts Receivable, plus (b) Target OUS Inventory, minus (c) Target Accounts Payable, minus (d) \$253 million.

"Target OUS Inventory" shall mean the product of (a) the cost of goods sold with respect to sales of the Mallinckrodt Business outside the United States for the fiscal quarter ending on June 28, 2013 (determined in a manner consistent with GAAP), multiplied by (b) 50.3, divided by (c) 90.

"<u>Tax Matters Agreement</u>" shall mean the Tax Matters Agreement, dated as of the date hereof, by and between Covidien and Mallinckrodt, as such Tax Matters Agreement may be amended from time to time.

"Tax Return" shall have the meaning set forth in the Tax Matters Agreement.

"Taxes" shall have the meaning set forth in the Tax Matters Agreement.

"<u>Technology</u>" shall mean all technology, designs, formulae, algorithms, procedures, methods, discoveries, processes, techniques, ideas, know-how, research and development, technical data, tools, materials, specifications, processes, inventions (whether patentable or unpatentable and whether or not reduced to practice), apparatus, creations, improvements, works of authorship in any media, confidential, proprietary or nonpublic information, and other similar materials, and all recordings, graphs, drawings, reports, analyses and other writings, and other tangible embodiments of the foregoing in any form whether or not listed herein, in each case, other than Software.

"Third-Party Claim" shall have the meaning set forth in Section 4.5(a).

"Transaction Accounting Principles" means GAAP applied on a basis consistent with the accounting principles, practices, methodologies and policies used in preparing the Mallinckrodt Balance Sheet, except as otherwise described on <u>Schedule 2.16</u>.

"Transfer Documents" shall have the meaning set forth in Section 2.4(b).

"Transferred Entities" shall have the meaning set forth in Section 2.2(a)(ii).

"Transition Services Agreement" shall mean the Transition Services Agreement, dated as of the date hereof, by and between Covidien and Mallinckrodt, as such Transition Services Agreement may be amended from time to time.

"Unreleased Excluded Liability" shall have the meaning set forth in Section 2.7(b).

"Unreleased Mallinckrodt Liability" shall have the meaning set forth in Section 2.6(b).

#### ARTICLE II THE SEPARATION

2.1 Transfer of Assets and Assumption of Liabilities.

(a) On or prior to the Distribution Date, but in any case prior to the Effective Time, in accordance with the plan and structure set forth on <u>Schedule 2.1(a)</u> (such plan and structure being referred to as the "<u>Plan of Reorganization</u>") and to the extent not previously effected pursuant to the steps of the Plan of Reorganization that have been completed prior to the date hereof:

(i) Covidien shall, and shall cause its applicable Subsidiaries to, assign, transfer, convey and deliver to Mallinckrodt or the applicable Mallinckrodt Designees, and Mallinckrodt or such Mallinckrodt Designees shall

accept from Covidien and its applicable Subsidiaries, all of Covidien's and such Subsidiaries' respective direct or indirect right, title and interest in and to all of the Mallinckrodt Assets (it being understood that if any Mallinckrodt Asset shall be held by a Transferred Entity or a wholly owned Subsidiary of a Transferred Entity, such Mallinckrodt Asset may be assigned, transferred, conveyed and delivered to Mallinckrodt as a result of the transfer of all or substantially all of the equity interests in such Transferred Entity from Covidien or its applicable Subsidiaries to Mallinckrodt or its applicable Subsidiaries);

(ii) subject to <u>Section 2.5(c)</u>. Mallinckrodt and the applicable Mallinckrodt Designees shall accept, assume and agree faithfully to perform, discharge and fulfill all the Mallinckrodt Liabilities in accordance with their respective terms. Mallinckrodt and such Mallinckrodt Designees shall be responsible for all Mallinckrodt Liabilities, regardless of when or where such Mallinckrodt Liabilities arose or arise, or whether the facts on which they are based occurred prior to or subsequent to the Effective Time, regardless of where or against whom such Mallinckrodt Liabilities are asserted or determined (including any Mallinckrodt Liabilities arising out of claims made by Covidien's or Mallinckrodt's respective directors, officers, employees, agents, Subsidiaries or Affiliates against any member of the Covidien Group or the Mallinckrodt Group) or whether asserted or determined prior to the date hereof, and regardless of whether arising from or alleged to arise from negligence, recklessness, violation of Law, fraud, misrepresentation or any other cause by any member of the Covidien Group, or any of their respective directors, officers, employees, agents, Subsidiaries;

(iii) Covidien shall cause the Mallinckrodt Designees to assign, transfer, convey and deliver to certain of its other Subsidiaries designated by Covidien, and such other Subsidiaries shall accept from the Mallinckrodt Designees, the Mallinckrodt Designees' respective right, title and interest in and to any Excluded Assets specified by Covidien to be so assigned, transferred, conveyed and delivered; and

(iv) Covidien and certain of its Subsidiaries designated by Covidien shall accept and assume from the Mallinckrodt Designees and agree faithfully to perform, discharge and fulfill certain Excluded Liabilities of the Mallinckrodt Designees, and Covidien and its applicable Subsidiaries shall be responsible for all Excluded Liabilities, regardless of when or where such Excluded Liabilities arose or arise, or whether the facts on which they are based occurred prior to or subsequent to the Effective Time, regardless of where or against whom such Excluded Liabilities are asserted or determined (including any such Excluded Liabilities arising out of claims made by Covidien's or Mallinckrodt's respective directors, officers, employees, agents, Subsidiaries or Affiliates against any member of the Covidien Group or the Mallinckrodt Group) or whether asserted or determined prior to the date hereof, and regardless of whether arising from or alleged to arise from negligence, recklessness, violation of Law, fraud, misrepresentation or any other cause by any member of the Covidien Group or the Mallinckrodt Group, or any of their respective directors, officers, employees, agents, Subsidiaries.

(b) In furtherance of the assignment, transfer, conveyance and delivery of the Mallinckrodt Assets and the assumption of the Mallinckrodt Liabilities in accordance with <u>Sections 2.1(a)(i)</u> and <u>2.1(a)(ii)</u>, on or before the date that such Mallinckrodt Assets are assigned, transferred, conveyed or delivered or such Mallinckrodt Liabilities are assumed (i) Covidien shall execute and deliver, and shall cause its applicable Subsidiaries to execute and deliver, such bills of sale, quitclaim deeds, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent necessary to evidence the transfer, conveyance and assignment of all of Covidien's and its applicable Subsidiaries' (other than Mallinckrodt's Subsidiaries) right, title and interest in and to the Mallinckrodt Assets to Mallinckrodt and/or the Mallinckrodt Designees, and (ii) Mallinckrodt shall execute and deliver, and shall cause the applicable Mallinckrodt Designees to execute and deliver, such assumptions of contracts and other instruments of assumption as and to the extent necessary to evidence the valid and effective assumption of the Mallinckrodt Liabilities by Mallinckrodt and the Mallinckrodt Designees. All of the foregoing documents contemplated by this Section 2.1(b) shall be referred to collectively herein as the "Covidien Transfer Documents."

(c) In the event that, in connection with the Separation, any Party (or any member of such Party's respective Group) shall receive or otherwise possess any Asset or Liability that is allocated to any other Person pursuant to this Agreement or any Ancillary Agreement, such Party shall promptly transfer, or cause to be transferred, such Asset or Liability, as the case may be, to the Person entitled to such Asset or responsible for such Liability, as the case may be. Prior to any such transfer, the Person receiving, possessing or responsible for such Asset or Liability shall be deemed to be holding such Asset or Liability, as the case may be, in trust for any such other Person.

(d) Mallinckrodt hereby waives compliance by each and every member of the Covidien Group with the requirements and provisions of any "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer or sale of any or all of the Mallinckrodt Assets to any member of the Mallinckrodt Group.

(e) Covidien hereby waives compliance by each and every member of the Mallinckrodt Group with the requirements and provisions of any "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer or sale of any or all of the Excluded Assets to any member of the Covidien Group.

#### 2.2 Mallinckrodt Assets.

(a) For the purposes of this Agreement, "Mallinckrodt Assets" shall mean (without duplication):

(i) all Assets that are expressly provided by this Agreement or any Ancillary Agreement (including for the avoidance of doubt the Schedules hereto or thereto) as Assets to be transferred to Mallinckrodt or any other member of the Mallinckrodt Group, including the Assets listed on <u>Schedule 2.2(a)(i)</u>;

(ii) (A) all the Mallinckrodt Contracts and all rights, interests or claims of either Covidien or Mallinckrodt or any of their respective Subsidiaries thereunder and (B) all issued and outstanding capital stock or other equity interests held by Covidien or its Subsidiaries in the wholly owned Subsidiaries of Covidien that shall have been contributed to, or otherwise transferred, conveyed, or assigned to, the Mallinckrodt Group pursuant to the Plan of Reorganization on or prior to the Distribution Date, including the wholly owned Subsidiaries listed on <u>Schedule 2.2(a)(ii)</u> (such Subsidiaries, the "<u>Transferred Entities</u>");

(iii) all Assets reflected as assets of Mallinckrodt and its Subsidiaries on the Mallinckrodt Balance Sheet, subject to any dispositions of such Assets subsequent to the date of the Mallinckrodt Balance Sheet; provided that the amounts set forth on the Mallinckrodt Balance Sheet with respect to any Assets shall not be treated as minimum amounts or limitations on the amount of such Assets that are included in the definition of Mallinckrodt Assets pursuant to this subclause (iii);

(iv) subject to <u>Section 6.2</u>, all rights, interests and claims of either Covidien or Mallinckrodt or any of their respective Subsidiaries to any Mallinckrodt Intellectual Property, Mallinckrodt Software and Mallinckrodt Technology;

(v) all other rights, interests and claims of either Party or any of its Subsidiaries with respect to Information that is exclusively related to the Mallinckrodt Assets, the Mallinckrodt Liabilities, the Mallinckrodt Business or the Transferred Entities and, subject to the provisions of the applicable Ancillary Agreements, a nonexclusive right to all Information that is related to the Mallinckrodt Assets, the Mallinckrodt Liabilities, the Mallinckrodt Business or the Transferred Entities and, subject to the Mallinckrodt Liabilities, the Mallinckrodt Business or the Transferred Entities (but is not exclusively related to such matters);

(vi) subject to, and to the extent provided in, <u>Article V</u>, any and all rights of any member of the Mallinckrodt Group under any Shared Policies and Mallinckrodt Policies, including any rights thereunder arising after the Effective Time in respect of any Policies that are occurrence policies;

(vii) all cash or cash equivalents of Mallinckrodt or any Transferred Entity (the "Mallinckrodt Cash");

(viii) any cash or cash equivalents withdrawn from Covidien Accounts in accordance with Section 2.10(e); and

(ix) except as contemplated by Section 2.5(b), any and all Assets, other than Intellectual Property, Software and Technology, owned and used or held for

use immediately prior to the Effective Time by Covidien or any of its Subsidiaries that are used primarily in the Mallinckrodt Business. The intention of this clause (ix) is only to rectify any inadvertent omission of transfer or conveyance of any Assets that, had the Parties given specific consideration to such Asset as of the date hereof, would have otherwise been classified as a Mallinckrodt Asset. No Asset shall be deemed to be a Mallinckrodt Asset solely as a result of this clause (ix) if such Asset is within the category or type of Asset expressly covered by the terms of this Agreement or an Ancillary Agreement unless the Party claiming entitlement to such Asset can establish that the omission of the transfer or conveyance of such Asset was inadvertent.

Notwithstanding the foregoing, the Mallinckrodt Assets shall not in any event include the Excluded Assets referred to in Section 2.2(b).

(b) For the purposes of this Agreement, "Excluded Assets" shall mean (without duplication):

(i) any and all Assets that are expressly contemplated by this Agreement or any Ancillary Agreement (or the Schedules hereto or thereto) as Assets to be retained by Covidien or any other member of the Covidien Group,

(ii) the Assets described on Schedule 2.2(b)(ii);

(iii) any cash or cash equivalents withdrawn from Mallinckrodt Accounts in accordance with Section 2.10(e);

(iv) all rights, interests and claims of either Party or any of its Subsidiaries to any Covidien Intellectual Property, Covidien Software or Covidien Technology;

(v) any and all Assets that are primarily related to the Respiratory Business;

(vi) any and all Shared Contracts (other than Mallinckrodt Assets arising under any Shared Contracts);

(vii) except to the extent provided in <u>Article V</u>, any and all rights of any member of the Covidien Group and/or the Mallinckrodt Group under any Shared Policies or Mallinckrodt Policies, including any rights thereunder arising before or after the Effective Time in respect of such Policies; and

(viii) subject to Section 2.2(a)(ix), any and all Assets of any members of the Covidien Group that are not Mallinckrodt Assets.

#### 2.3 Mallinckrodt Liabilities

(a) For the purposes of this Agreement, "Mallinckrodt Liabilities" shall mean (without duplication):

(i) all Liabilities, including any Environmental Liabilities and any Liability relating to the protection of human and occupational health and safety, the protection or restoration of, or prevention of harm to, the environment or natural resources, relating to, arising out of or resulting from:

(A) the operation or ownership of the Mallinckrodt Business, as conducted at any time prior to, on or after the Distribution Date (including any Liability relating to, arising out of or resulting from any act or failure to act by any Representative (whether or not such act or failure to act is or was within such Person's authority));

(B) the operation or ownership of any business conducted by any member of the Mallinckrodt Group at any time after the Effective Time (including any Liability relating to, arising out of or resulting from any act or failure to act by any Representative (whether or not such act or failure to act is or was within such Person's authority)); or

(C) any Mallinckrodt Assets (including any Mallinckrodt Contracts and any Mallinckrodt Assets arising under any Shared Contracts, to the extent related to the Mallinckrodt Business, and any real property and leasehold interests) in any such case whether arising before, on or after the Distribution Date;

(ii) any and all Liabilities that are expressly provided by this Agreement or any Ancillary Agreement (or the Schedules hereto or thereto) as Liabilities to be assumed by Mallinckrodt or any other member of the Mallinckrodt Group, and all agreements, obligations and Liabilities of any member of the Mallinckrodt Group under this Agreement or any of the Ancillary Agreements;

(iii) all Liabilities relating to, arising out of or resulting from the Mallinckrodt Financing Arrangements;

(iv) all Liabilities relating to, arising out of or resulting from any of the terminated, divested or discontinued businesses and operations of the Mallinckrodt Business, including the businesses listed on <u>Schedule 2.3(a)(iv)</u>;

(v) all Liabilities reflected as liabilities or obligations of Mallinckrodt and its Subsidiaries on the Mallinckrodt Balance Sheet, subject to any discharge of such Liabilities subsequent to the date of the Mallinckrodt Balance Sheet; provided that the amounts set forth on the Mallinckrodt Balance Sheet with respect to any Liabilities shall not be treated as minimum amounts or limitations on the amount of such Liabilities that are included in the definition of Mallinckrodt Liabilities pursuant to this subclause (v);

(vi) all Liabilities relating to, arising out of or resulting from the Actions listed on Schedule 2.3(a)(v);

(vii) all Liabilities relating to, arising out of or resulting from the Specified Indebtedness; and

(viii) all Liabilities arising out of claims made by Covidien's or Mallinckrodt's respective directors, officers, shareholders, employees, agents, Subsidiaries or Affiliates against any member of the Covidien Group or the Mallinckrodt Group to the extent relating to, arising out of or resulting from the Mallinckrodt Business or the other businesses, operations, activities or Liabilities referred to in clauses (i) through (vii) above, inclusive.

Notwithstanding the foregoing, the Mallinckrodt Liabilities shall not include the Excluded Liabilities referred to in Section 2.3(b).

(b) For the purposes of this Agreement, "Excluded Liabilities" shall mean (without duplication):

(i) any and all Liabilities that are expressly contemplated by this Agreement or any Ancillary Agreement (or the Schedules hereto or thereto) as Liabilities to be retained or assumed by Covidien or any other member of the Covidien Group, and all agreements and obligations of any member of the Covidien Group under this Agreement or any of the Ancillary Agreements;

(ii) any and all Liabilities of a member of the Covidien Group to the extent relating to, arising out of or resulting from any Excluded Assets (other than Liabilities arising under any Shared Contracts to the extent such Liabilities relate to the Mallinckrodt Business);

(iii) the Liabilities described on Schedule 2.3(b)(iii);

(iv) any and all Liabilities that are primarily related to the Respiratory Business; and

(v) any and all Liabilities of any members of the Covidien Group that are not Mallinckrodt Liabilities.

#### 2.4 Transfer of Excluded Assets; Assumption of Excluded Liabilities.

(a) To the extent any Excluded Asset is transferred or assigned to, or any Excluded Liability is assumed by, a member of the Mallinckrodt Group upon consummation of the Distribution or is owned or held by a member of the Mallinckrodt Group after the Effective Time, from and after the Distribution Date:

(i) Mallinckrodt shall, and shall cause its applicable Subsidiaries to, promptly assign, transfer, convey and deliver to Covidien or certain of its Subsidiaries designated by Covidien, and Covidien or such Subsidiaries shall accept from Mallinckrodt and its applicable Subsidiaries, all of Mallinckrodt's and such Subsidiaries' respective right, title and interest in and to such Excluded Assets; and

(ii) Covidien and certain of its Subsidiaries designated by Covidien shall promptly accept, assume and agree faithfully to perform, discharge and fulfill all such Excluded Liabilities in accordance with their respective terms.

(b) In furtherance of the assignment, transfer, conveyance and delivery of Excluded Assets and the assumption of Excluded Liabilities set forth in Sections 2.1(a)(ii), 2.1(a)(iv), 2.4(a)(i) and 2.4(a)(i) and without any additional consideration therefor: (i) Mallinckrodt shall execute and deliver, and shall cause its applicable Subsidiaries to execute and deliver, such bills of sale, quitclaim deeds, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent necessary to evidence the transfer, conveyance and assignment of all of Mallinckrodt's and its applicable Subsidiaries' right, title and interest in and to the Excluded Assets to Covidien and its applicable Subsidiaries, and (ii) Covidien shall execute and deliver, such assumptions of contracts and other instruments of assumption as and to the extent necessary to evidence the transfer and other instruments of assumption as and to the extent necessary to evidence the valid and effective assumption of the Excluded Liabilities by Covidien and such Subsidiaries. All of the foregoing documents contemplated by this Section 2.4(b) shall be referred to collectively herein as the "Mallinckrodt Transfer Documents" and, together with the Covidien Transfer Documents."

#### 2.5 Approvals and Notifications.

(a) To the extent that the transfer or assignment of any Excluded Assets or the assumption of any Excluded Liabilities requires any Approvals or Notifications, the Parties shall use their commercially reasonable efforts to obtain or make such Approvals or Notifications as soon as reasonably practicable; provided, however, that, except to the extent expressly provided in this Agreement or any of the Ancillary Agreements or as otherwise agreed between Covidien and Mallinckrodt, neither Covidien nor Mallinckrodt shall be obligated to contribute capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any Person in order to obtain or make such Approvals or Notifications.

(b) If and to the extent that the valid, complete and perfected transfer or assignment to the Covidien Group of any Excluded Assets or the assumption by the Covidien Group of any Excluded Liabilities would be a violation of applicable Law, or require any Approval or Notification that has not been obtained or made on or before the Distribution Date, then, unless the Parties shall otherwise mutually determine, the transfer or assignment to the Covidien Group of such Excluded Assets or the assumption by the Covidien Group of such Excluded Liabilities, as the case may be, shall be automatically deemed deferred and any such purported transfer, assignment or assumption shall be null and void until such time as all legal

impediments are removed or such Approvals or Notifications have been obtained or made. Notwithstanding the foregoing, any such Excluded Assets or Excluded Liabilities shall continue to constitute Excluded Assets or Excluded Liabilities for all other purposes of this Agreement.

(c) If any transfer or assignment of any Excluded Asset or any assumption of any Excluded Liability not intended to be transferred, assigned or assumed hereunder, as the case may be, is consummated on or prior to the Distribution Date, then, insofar as reasonably possible, the member of the Mallinckrodt Group holding or owning such Excluded Asset or such Excluded Liability, as the case may be, shall thereafter hold such Excluded Asset or Excluded Liability, as the case may be, for the use and benefit of the member of the Covidien Group entitled thereto (at the expense of the member of the Covidien Group entitled thereto). In addition, the member of the Mallinckrodt Group retaining such Excluded Asset or such Excluded Liability shall, insofar as reasonably possible and to the extent permitted by applicable Law, treat such Excluded Asset or Excluded Liability in the ordinary course of business in accordance with past practice and take such other actions as may be reasonably requested by the member of the Covidien Group to whom such Excluded Asset is to be transferred or assigned, or which will assume such Excluded Liability, as the case may be, in order to place such member of the Covidien Group in a substantially similar position as if such Excluded Liability, as the case may be, including use, risk of loss, potential for gain, and dominion, control and command over such Excluded Asset or Excluded Liability, as the case may be, and all costs and expenses related thereto, shall inure from and after the Distribution Date to the Covidien Group.

(d) If and when the Approvals or Notifications, the absence of which caused the deferral of transfer or assignment of any Excluded Asset or the deferral of assumption of any Excluded Liability, are obtained or made, and, if and when any other legal impediments for the transfer or assignment of any Excluded Asset or the assumption of any Excluded Liability have been removed, the transfer or assignment of the applicable Excluded Asset or the assumption of the applicable Excluded Liability, as the case may be, shall be effected in accordance with the terms of this Agreement and/or the applicable Ancillary Agreement.

(e) Any member of the Mallinckrodt Group retaining an Excluded Asset or Excluded Liability due to the deferral of the transfer or assignment of such Excluded Asset or the deferral of the assumption of such Excluded Liability, as the case may be, shall not be obligated, in connection with the foregoing, to expend any money unless the necessary funds are advanced (or otherwise made available) by Covidien or the member of the Covidien Group entitled to the Excluded Asset or Excluded Liability, other than reasonable out-of-pocket expenses, attorneys' fees and recording or similar fees, all of which shall be promptly reimbursed by Covidien or the member of the Covidien Group entitled to such Excluded Asset or Excluded Liability.

(f) To the extent that the transfer or assignment of any Mallinckrodt Asset, the assumption of any Mallinckrodt Liability, the Separation, or the Distribution requires any Approvals or Notifications, the Parties shall use their commercially reasonable efforts to obtain or make such Approvals or Notifications as soon as reasonably practicable; <u>provided</u>, <u>however</u>, that, except to the extent expressly provided in this Agreement or any of the Ancillary Agreements or as otherwise agreed between Covidien and Mallinckrodt, neither Covidien nor

Mallinckrodt shall be obligated to contribute capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any Person in order to obtain or make such Approvals or Notifications.

(g) If and to the extent that the valid, complete and perfected transfer or assignment to the Mallinckrodt Group of any Mallinckrodt Asset or assumption by the Mallinckrodt Group of any Mallinckrodt Liability would be a violation of applicable Law, or require any Approvals or Notifications in connection with the Separation or the Distribution that have not been obtained or made on or before the Distribution Date, then, unless the Parties shall otherwise mutually determine, the transfer or assignment to the Mallinckrodt Group of such Mallinckrodt Assets or the assumption by the Mallinckrodt Group of such Mallinckrodt Liabilities, as the case may be, shall be automatically deemed deferred and any such purported transfer, assignment or assumption shall be null and void until such time as all legal impediments are removed or such Approvals or Notifications have been obtained or made. Notwithstanding the foregoing, any such Mallinckrodt Assets or Mallinckrodt Liabilities for all other purposes of this Agreement.

(h) If any transfer or assignment of any Mallinckrodt Asset or any assumption of any Mallinckrodt Liability intended to be transferred, assigned or assumed hereunder, as the case may be, is not consummated on or prior to the Distribution Date, whether as a result of the provisions of <u>Section 2.5(g)</u> or for any other reason, then, insofar as reasonably possible, the member of the Covidien Group retaining such Mallinckrodt Asset or such Mallinckrodt Liability, as the case may be, shall thereafter hold such Mallinckrodt Asset or Mallinckrodt Liability, as the case may be, for the use and benefit of the member of the Mallinckrodt Group entitled thereto). In addition, the member of the Covidien Group retaining such Mallinckrodt Liability shall, insofar as reasonably possible and to the extent permitted by applicable Law, treat such Mallinckrodt Asset or Mallinckrodt Liability in the ordinary course of business in accordance with past practice and take such other actions as may be reasonably requested by the member of the Mallinckrodt Group in a substantially similar position as if such Mallinckrodt Asset or Mallinckrodt Liability, as the case may be, in order to place such member of the Mallinckrodt Group in a substantially similar position as if such Mallinckrodt Asset or Mallinckrodt Liability, as the case may be, including use, risk of loss, potential for gain, and dominion, control and command over such Mallinckrodt Asset or Mallinckrodt Liability, as the case may be, and all costs and expenses related thereto, shall increase form and after the Distribution Date to the Mallinckrodt Group.

(i) If and when the Approvals or Notifications, the absence of which caused the deferral of transfer or assignment of any Mallinckrodt Asset or the deferral of assumption of any Mallinckrodt Liability pursuant to <u>Section 2.5(g)</u>, are obtained or made, and, if and when any other legal impediments for the transfer or assignment of any Mallinckrodt Asset or the assumption of any Mallinckrodt Liability have been removed, the transfer or assignment of the applicable Mallinckrodt Liability, as the case may be, shall be effected in accordance with the terms of this Agreement and/or the applicable Ancillary Agreement.

(j) Any member of the Covidien Group retaining a Mallinckrodt Asset or Mallinckrodt Liability due to the deferral of the transfer or assignment of such Mallinckrodt Asset or the deferral of the assumption of such Mallinckrodt Liability, as the case may be, shall not be obligated, in connection with the foregoing, to expend any money unless the necessary funds are advanced (or otherwise made available) by Mallinckrodt or the member of the Mallinckrodt Group entitled to the Mallinckrodt Asset or Mallinckrodt Liability, other than reasonable out-of-pocket expenses, attorneys' fees and recording or similar fees, all of which shall be promptly reimbursed by Mallinckrodt or the member of the Mallinckrodt Group entitled to such Mallinckrodt Asset or Mallinckrodt Liability.

#### 2.6 Novation of Mallinckrodt Liabilities.

(a) Each of Covidien and Mallinckrodt, at the request of the other, shall use its commercially reasonable efforts to obtain, or to cause to be obtained, as soon as reasonably practicable, any consent, substitution, approval or amendment required to novate or assign all obligations under agreements, leases, licenses and other obligations or Liabilities of any nature whatsoever that constitute Mallinckrodt Liabilities, or to obtain in writing the unconditional release of all parties to such arrangements other than any member of the Mallinckrodt Group, so that, in any such case, the members of the Mallinckrodt Group will be solely responsible for such Liabilities; provided, however, that, except as otherwise expressly provided in this Agreement or any of the Ancillary Agreements, neither Covidien nor Mallinckrodt shall be obligated to contribute any capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any third Person from whom any such consent, substitution, approval, amendment or release is requested.

(b) If Covidien or Mallinckrodt is unable to obtain, or to cause to be obtained, any such required consent, substitution, approval, amendment or release and the applicable member of the Covidien Group continues to be bound by such agreement, lease, license or other obligation or Liability (each, an "<u>Unreleased Mallinckrodt Liability</u>"), Mallinckrodt shall, to the extent not prohibited by Law, as indemnitor, guarantor, agent or subcontractor for such member of the Covidien Group, as the case may be, (i) pay, perform and discharge fully all the obligations or other Liabilities of such member of the Covidien Group that constitute Unreleased Mallinckrodt Liabilities from and after the Distribution Date and (ii) use its commercially reasonable efforts to effect such payment, performance, or discharge prior to any demand for such payment, performance, or discharge is permitted to be made by the obligee thereunder on any member of the Covidien Group. If and when any such consent, substitution, approval, amendment or release shall be obtained or the Unreleased Mallinckrodt Liabilities shall otherwise become assignable or able to be novated, Covidien shall promptly assign, or cause to be assigned, and Mallinckrodt or the applicable Mallinckrodt Group member shall assume, such Unreleased Mallinckrodt Liabilities without exchange of further consideration.

2.7 Novation of Excluded Liabilities.

(a) Each of Covidien and Mallinckrodt, at the request of the other, shall use its commercially reasonable efforts to obtain, or to cause to be obtained, as soon as reasonably practicable, any consent, substitution, approval or amendment required to novate or assign all obligations under agreements, leases, licenses and other obligations or Liabilities for which a member of the Covidien Group and a member of the Mallinckrodt Group are jointly or severally liable and that constitute Excluded Liabilities, or to obtain in writing the unconditional release of all parties to such arrangements other than any member of the Covidien Group, so that, in any such case, the members of the Covidien Group will be solely responsible for such Liabilities; provided, however, that, except as otherwise expressly provided in this Agreement or any of the Ancillary Agreements, neither Covidien nor Mallinckrodt shall be obligated to contribute any capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any third Person from whom any such consent, substitution, approval, amendment or release is requested.

(b) If Covidien or Mallinckrodt is unable to obtain, or to cause to be obtained, any such required consent, substitution, approval, amendment or release and the applicable member of the Mallinckrodt Group continues to be bound by such agreement, lease, license or other obligation or Liability (each, an "<u>Unreleased Excluded Liability</u>"), Covidien shall, to the extent not prohibited by Law, as indemnitor, guarantor, agent or subcontractor for such member of the Mallinckrodt Group that constitute Unreleased Excluded Liabilities from and after the Distribution Date and (ii) use its commercially reasonable efforts to effect such payment, performance, or discharge prior to any demand for such payment, performance, or discharge is permitted to be made by the obligee thereunder on any member of the Mallinckrodt Group. If and when any such consent, substitution, approval, amendment or release shall be obtained or the Unreleased Excluded Liabilities shall otherwise become assignable or able to be novated, Mallinckrodt shall promptly assign, or cause to be assigned, and Covidien or the applicable Covidien Group member shall assume, such Unreleased Excluded Liabilities without exchange of further consideration.

#### 2.8 Intercompany Agreements and Arrangements.

(a) Except as set forth in <u>Section 2.8(b)</u>, in furtherance of the releases and other provisions of <u>Section 4.1</u> hereof, Mallinckrodt and each member of the Mallinckrodt Group, on the one hand, and Covidien and each member of the Covidien Group, on the other hand, hereby terminate any and all agreements, arrangements, commitments or understandings, whether or not in writing, between or among Mallinckrodt and/or any member of the Mallinckrodt Group, on the other hand, effective as of the Effective Time. No such terminated agreement, arrangement, commitment or understanding (including any provision thereof which purports to survive termination) shall be of any further force or effect after the Effective Time. Each Party shall, at the reasonable request of any other Party, take, or cause to be taken, such other actions as may be necessary to effect the foregoing.

(b) The provisions of <u>Section 2.8(a)</u> shall not apply to any of the following agreements, arrangements, commitments or understandings (or to any of the provisions thereof): (i) this Agreement and the Ancillary Agreements (and each other agreement or instrument

expressly contemplated by this Agreement or any Ancillary Agreement to be entered into by any of the Parties or any of the members of their respective Groups or to be continued following the Effective Time); (ii) any agreements, arrangements, commitments or understandings to which any Person other than the Parties and their respective Affiliates is a party; (iii) any intercompany accounts payable or accounts receivable accrued as of the Effective Time that are reflected in the books and records of the Parties or otherwise documented in writing in accordance with past practices, which shall be settled in the manner contemplated by Section 2.8(d); (iv) any agreements, arrangements, commitments or understandings to which any non-wholly owned Subsidiary of Covidien or Mallinckrodt, as the case may be, is a party (it being understood that directors' qualifying shares or similar interests will be disregarded for purposes of determining whether a Subsidiary is wholly owned); (v) any Shared Contracts; (vi) any agreements, arrangements, commitments or understandings relating to the purchase and sale of products in the ordinary course of business between any member of the Mallinckrodt Group and any member of the Covidien Group; (vii) the Reorganization Agreements; and (viii) any other agreements, arrangements, commitments or understandings that this Agreement or any Ancillary Agreement expressly contemplates will survive past the Effective Time.

(c) The Parties acknowledge and agree that all of the Intercompany Balances as of five (5) Business Days prior to the date hereof have been repaid, settled or otherwise eliminated by means of cash payments, a dividend, capital contribution, a combination of the foregoing or otherwise, as determined by Covidien.

(d) All Intercompany Balances outstanding as of the date hereof shall, as promptly as practicable after the Effective Time, be repaid, settled or otherwise eliminated by means of cash payments, a dividend, capital contribution, a combination of the foregoing or otherwise, as determined by Covidien.

#### 2.9 Treatment of Shared Contracts.

(a) Without limiting the generality of the obligations set forth in <u>Section 2.1</u>, unless the Parties otherwise agree or the benefits of any contract, agreement, arrangement, commitment or understanding described in this <u>Section 2.9</u> are expressly conveyed to the applicable Party pursuant to this Agreement or an Ancillary Agreement, (i) any contract, agreement, arrangement, commitment or understanding that is listed on <u>Schedule 2.9(a)</u> shall be assigned in part to the applicable member(s) of the applicable Group, if so assignable, or appropriately amended prior to, on or after the Distribution Date, so that each Party or the members of its respective Group shall, as of the Distribution Date, be entitled to the rights and benefits, and shall assume the related portion of any Liabilities, inuring to its respective businesses, in each case, in accordance with the allocation of benefits and burdens set forth on <u>Schedule 2.9(a)</u>, and (ii) (A) any contract, agreement, arrangement, commitment or understanding that is an Excluded Asset or Excluded Liability but, prior to the Effective Time, inured in part to the benefit or burden of any member of the Mallinckrodt Group (other than any such contract, agreement, arrangement, commitment or understanding covering substantially the same services or arrangements that are covered by a contract, agreement, arrangement, commitment or understanding entered into by a member of the Mallinckrodt Group in

connection with the Separation), and (B) any contract, agreement, arrangement, commitment or understanding that is a Mallinckrodt Asset or a Mallinckrodt Liability but, prior to the Effective Time, inured in part to the benefit or burden of any member of the Covidien Group (other than any such contract, agreement, arrangement, commitment or understanding covering substantially the same services or arrangements that are covered by a contract, agreement, arrangement, commitment or understanding entered into by a member of the Covidien Group in connection with the Separation), shall be assigned in part to the applicable member(s) of the applicable Group, if so assignable, or appropriately amended prior to, on or after the Distribution Date, so that each Party or the members of its respective Group shall, as of the Distribution Date, be entitled to the rights and benefits, and shall assume the related portion of any Liabilities, inuring to its respective businesses (any contract, agreement, arrangement, commitment or understanding referred to in clause (i) or (ii) above, a "Shared Contract"); provided, however, that, in the case of each of clause (i) and (ii), (1) in no event shall any member of any Group be required to assign (or amend) any Shared Contract in its entirety or to assign a portion of any Shared Contract which is not assignable (or cannot be amended) by its terms (including any terms imposing consents or conditions on an assignment where such consents or conditions have not been obtained or fulfilled) and (2) if any Shared Contract cannot be so partially assigned by its terms or otherwise, or cannot be amended or if such assignment or amendment would impair the benefit the Parties thereto derive from such Shared Contract, then the Parties shall, and shall cause each of their respective Subsidiaries to, take such other reasonable and permissible actions (including by providing prompt notice to the other Party with respect to any relevant claim of Liability or other relevant matters arising in connection with a Shared Contract so as to allow such other Party the ability to exercise any applicable rights under such Shared Contract) to cause a member of the Mallinckrodt Group or the Covidien Group, as the case may be, to receive the rights and benefits of that portion of each Shared Contract that relates to the Mallinckrodt Business or the businesses retained by Covidien, as the case may be (in each case, to the extent so related), as if such Shared Contract had been assigned to (or amended to allow) a member of the applicable Group pursuant to this Section 2.9, and to bear the burden of the corresponding Liabilities (including any Liabilities that may arise by reason of such arrangement), as if such Liabilities had been assumed by a member of the applicable Group pursuant to this Section 2.9.

(b) Each of Covidien and Mallinckrodt shall, and shall cause the members of its Group to, (i) treat for all Tax purposes the portion of each Shared Contract inuring to its respective businesses as Assets owned by, and/or Liabilities of, as applicable, such Party, or its subsidiaries, as applicable, not later than the Distribution Date, and (ii) neither report nor take any Tax position (on a Tax Return or otherwise) inconsistent with such treatment (unless required by applicable Law).

(c) Nothing in this <u>Section 2.9</u> shall require any member of any Group to make any payment (except to the extent advanced, assumed or agreed in advance to be reimbursed by any member of the other Group), incur any obligation or grant any concession for the benefit of any member of any other Group in order to effect any transaction contemplated by this <u>Section 2.9</u>.

## 2.10 Bank Accounts; Cash Balances.

(a) Covidien and Mallinckrodt each agrees to take, or cause the respective members of their respective Groups to take, on the Distribution Date (or such earlier time as Covidien and Mallinckrodt may agree), all actions necessary to amend all contracts or agreements governing each bank and brokerage account owned by Mallinckrodt or any other member of the Mallinckrodt Group (collectively, the "<u>Mallinckrodt Accounts</u>") and all contracts or agreements governing each bank or brokerage account owned by Covidien or any other member of the Covidien Group (collectively, the "<u>Covidien Accounts</u>") so that each such Mallinckrodt Account and Covidien Account, if currently linked (whether by automatic withdrawal, automatic deposit or any other authorization to transfer funds from or to, hereinafter "<u>linked</u>") to any Covidien Account or Mallinckrodt Account, respectively, is delinked from such Covidien Account or Mallinckrodt Account, respectively.

(b) It is intended that, following consummation of the actions contemplated by <u>Section 2.10(a)</u>, there will be in place a centralized cash management process pursuant to which the Mallinckrodt Accounts will be managed centrally and funds collected will be transferred into one (1) or more centralized accounts maintained by Mallinckrodt.

(c) It is intended that, following consummation of the actions contemplated by <u>Section 2.10(a)</u>, there will continue to be in place a centralized cash management process pursuant to which the Covidien Accounts will be managed centrally and funds collected will be transferred into one (1) or more centralized accounts maintained by Covidien.

(d) With respect to any outstanding payments initiated by Covidien, Mallinckrodt or any of their respective Subsidiaries prior to the Effective Time, such outstanding payments shall be honored following the Effective Time by the Person or Group owning the account from which the payment was initiated.

(e) As between Covidien and Mallinckrodt (and the members of their respective Groups) all payments made and reimbursements received after the Effective Time by either Party (or member of its Group) that relate to a business, Asset or Liability of the other Party (or member of its Group) shall be held by such Party in trust for the use and benefit of the Party entitled thereto and, promptly following receipt by such Party of any such payment or reimbursement, such Party shall pay over, or shall cause the applicable member of its Group to pay over, to the other Party the amount of such payment or reimbursement without right of set-off.

2.11 <u>Ancillary Agreements</u>. Effective on or prior to the Distribution Date, each of Covidien and Mallinckrodt will execute and deliver all Ancillary Agreements to which it is a party.

2.12 Certain Litigation Matters. Notwithstanding anything herein to the contrary, from and after the Distribution Date, (a) Covidien shall have the exclusive right to control in its sole discretion all proceedings and negotiations relating to the Actions specified on <u>Schedule 2.12</u>, including the exclusive right to settle such Actions in its sole discretion and (b) all monies received by any of the Parties or their respective Subsidiaries in respect of such Actions shall be shared, after reduction from such recovery of all legal fees and other out-of-pocket expenses incurred by the Parties or their respective Subsidiaries in connection therewith, 70% by Covidien and 30% by Mallinckrodt.

2.13 <u>Disclaimer of Representations and Warranties</u>. EACH OF COVIDIEN (ON BEHALF OF ITSELF AND EACH MEMBER OF THE COVIDIEN GROUP) AND MALLINCKRODT (ON BEHALF OF ITSELF AND EACH MEMBER OF THE MALLINCKRODT GROUP) UNDERSTANDS AND AGREES THAT, EXCEPT AS EXPRESSLY SET FORTH HEREIN, IN ANY REORGANIZATION AGREEMENT OR IN ANY ANCILLARY AGREEMENT, NO PARTY

TO THIS AGREEMENT, ANY ANCILLARY AGREEMENT, ANY REORGANIZATION AGREEMENT OR ANY OTHER AGREEMENT OR DOCUMENT CONTEMPLATED BY THIS AGREEMENT, ANY ANCILLARY AGREEMENT, ANY REORGANIZATION AGREEMENT OR OTHERWISE, IS REPRESENTING OR WARRANTING IN ANY WAY AS TO THE ASSETS, BUSINESSES OR LIABILITIES TRANSFERRED OR ASSUMED AS CONTEMPLATED HEREBY OR THEREBY, AS TO ANY CONSENTS, NOTIFICATIONS OR APPROVALS REQUIRED IN CONNECTION HEREWITH OR THEREWITH, AS TO THE VALUE OR FREEDOM FROM ANY SECURITY INTERESTS OF, OR ANY OTHER MATTER CONCERNING, ANY ASSETS OF SUCH PARTY, OR AS TO THE ABSENCE OF ANY DEFENSES OR RIGHT OF SETOFF OR FREEDOM FROM COUNTERCLAIM WITH RESPECT TO ANY CLAIM OR OTHER ASSET, INCLUDING ANY ACCOUNTS RECEIVABLE, OF ANY PARTY, OR AS TO THE LEGAL SUFFICIENCY OF ANY ASSIGNMENT, DOCUMENT OR INSTRUMENT DELIVERED HEREUNDER TO CONVEY TITLE TO ANY ASSET OR THING OF VALUE UPON THE EXECUTION, DELIVERY AND FILING HEREOF OR THEREOF. EXCEPT AS MAY EXPRESSLY BE SET FORTH HEREIN, IN ANY REORGANIZATION AGREEMENT OR IN ANY ANCILLARY AGREEMENT, ALL SUCH ASSETS ARE BEING TRANSFERRED ON AN "AS IS," "WHERE IS" BASIS (AND, IN THE CASE OF ANY REAL PROPERTY, EXCEPT AS OTHERWISE AGREED BY COVIDIEN, BY MEANS OF A QUITCLAIM OR SIMILAR FORM OF DEED OR CONVEYANCE) AND THE RESPECTIVE TRANSFEREES SHALL BEAR THE ECONOMIC AND LEGAL RISKS THAT (I) ANY CONVEYANCE WILL PROVE TO BE INSUFFICIENT TO VEST IN THE TRANSFEREE GOOD AND MARKETABLE TITLE, FREE AND CLEAR OF ANY SECURITY INTEREST, AND (II) ANY NECESSARY APPROVALS OR NOTIFICATIONS ARE NOT OBTAINED OR MADE OR THAT ANY REQUIREMENTS OF LAWS OR JUDGMENTS ARE NOT COMPLIED WITH.

2.14 <u>Intellectual Property</u>. Notwithstanding anything to the contrary in this Agreement or any Ancillary Agreement, except for Intellectual Property related agreements which relate specifically to the Mallinckrodt Business and were executed or entered into by the Mallinckrodt Business, Covidien will retain all licenses, rights and royalty payments in and to any and all existing Intellectual Property license agreements with third parties, including the sole right to amend or modify such agreements.

### 2.15 Mallinckrodt Financing Arrangements.

(a) Prior to or as of the date hereof, Mallinckrodt and MIFSA entered into the Mallinckrodt Financing Arrangements. Mallinckrodt and its Subsidiaries agree to take all such reasonable action as Covidien shall request after the date hereof to ensure that Mallinckrodt and its Subsidiaries, as the case may be, shall be solely and exclusively liable for all obligations under the Mallinckrodt Financing Arrangements and each of Covidien and any other member of the Covidien Group are fully released and discharged of any and all of their obligations thereunder as of the Distribution Date.

(b) On or prior to the Distribution Date, MIFSA shall redeem a portion of its equity interest for an amount in cash that equals (i) \$889.3 million (which represents the net proceeds of the Senior Notes Offering), plus (ii) Covidien's estimate as of the time of such redemption of (x) the amount of Mallinckrodt Cash and (y) the amount drawn under the Mallinckrodt Lines of Credit, in each case as of a then-recent date, and without duplication, less (iii) \$168 million.

### 2.16 Adjustment Amount.

(a) <u>Schedule 2.16</u> sets forth a sample calculation of the Adjustment Amount and the Target Adjustment Amount as of the Balance Sheet Date (the "<u>Sample Closing Statement</u>"), including the asset, liability and other line items and accounting principles used in such calculation, and assuming that all of such asset and liability line items that constitute Mallinckrodt Assets or Mallinckrodt Liabilities under this Agreement will be transferred to Mallinckrodt as of the Distribution.

(b) Within sixty (60) days after the Distribution Date, Mallinckrodt shall cause to be prepared and delivered to Covidien a statement (the "<u>Closing</u> <u>Statement</u>") setting forth (i) the Adjustment Amount and the calculation of the Adjustment Amount and (ii) the Target Adjustment Amount and the calculation of the Target Adjustment Amount. The Closing Statement shall be prepared in accordance with the Transaction Accounting Principles, including the use of the same line items and line item entries, set forth on and used in the preparation of the Sample Closing Statement; <u>provided</u>, <u>however</u>, that assets newly acquired and liabilities newly incurred following the date of the Sample Closing Statement which cannot be appropriately placed in line items previously used by Mallinckrodt, but that constitute Mallinckrodt Assets or Mallinckrodt Liabilities, will also be included to the extent consistent with the Transaction Accounting Principles.

(c) Within thirty (30) days following receipt by Covidien of the Closing Statement, Covidien shall deliver written notice to Mallinckrodt of any dispute Covidien has with respect to the preparation or content of the Closing Statement (the "Dispute Notice"); provided, however, that if Covidien does not deliver any Dispute Notice to Mallinckrodt within such thirty (30)-day period, the Closing Statement will be final, conclusive and binding on the Parties. Any Dispute Notice shall (i) set forth in reasonable detail the basis for any dispute included therein, the amounts involved and Covidien's determination of the Adjustment Amount and/or the Target Adjustment Amount (as applicable) and (ii) include only disagreements based on the Adjustment Amount and/or the Target Adjustment Amount (as applicable) not being calculated properly in accordance with this Agreement or containing mathematical errors. Upon receipt by Mallinckrodt of a Dispute Notice, Mallinckrodt and Covidien shall negotiate in good faith to resolve any dispute set forth therein. If Mallinckrodt and Covidien, such good faith effort notwithstanding, fail to resolve any such dispute within fifteen (15) Business Days following receipt by Mallinckrodt of the Dispute Notice (the "Dispute Resolution Period"), then Mallinckrodt and Covidien jointly shall engage, within ten (10) Business Days following the expiration of the Dispute Resolution Period, Ernst & Young LLP or, if Ernst & Young LLP is unavailable or conflicted, another nationally recognized major accounting firm selected jointly by Covidien and Mallinckrodt (the "Independent Accounting Firm") to resolve any such dispute. If Ernst & Young LLP is unavailable or conflicted and Covidien and Mallinckrodt are unable to agree on the Independent Accounting Firm, then each of Covidien and Mallinckrodt shall select a nationally recognized major accounting firm, and the two (2) firms will mutually select a third nationally recognized major accounting firm to serve as the Independent Accounting Firm. As promptly as practicable, and in any event not more than fifteen (15) days following the engagement of the Independent Accounting Firm, Mallinckrodt and Covidien shall each prepare and submit a presentation detailing each Party's complete statement of proposed resolution of each issue still in dispute to the Independent Accounting Firm. Mallinckrodt and Covidien shall cause the Independent Accounting Firm to, as soon as practicable after the submission of the presentations described in the immediately preceding sentence and in any event not more than thirty (30) days following such presentations, make a final determination, binding on the Parties, of the appropriate amount of

each of the line items that remain in dispute as indicated in the Dispute Notice. With respect to each disputed line item, such determination, if not in accordance with the position of either Covidien or Mallinckrodt, shall not be in excess of the higher, nor less than the lower, of the amounts set forth by Mallinckrodt in the Closing Statement or by Covidien in the Dispute Notice, as applicable. Notwithstanding the foregoing, the scope of the disputes to be resolved by the Independent Accounting Firm shall be limited to whether any determination of the Adjustment Amount and/or the Target Adjustment Amount (as applicable) was properly calculated in accordance with the Transaction Accounting Principles, and the Independent Accounting Firm is not to make any other determination, including any determination as to whether GAAP was followed, to the extent GAAP is inconsistent with the Transaction Accounting Principles. All fees and expenses relating to the work, if any, to be performed by the Independent Accounting Firm shall be borne equally by Covidien and Mallinckrodt. All determinations made by the Independent Accounting Firm, and the Closing Statement, as modified by the Independent Accounting Firm, will be final, conclusive and binding on the Parties, absent fraud or manifest error.

(d) For purposes of complying with the terms set forth in this <u>Section 2.16</u>, Mallinckrodt and Covidien shall cooperate with and make available to each other and their respective Representatives all information, records, data and working papers, in each case, to the extent related to the Mallinckrodt Assets, Mallinckrodt Liabilities or Mallinckrodt Business, and shall permit access to its facilities and personnel, as may be reasonably required in connection with the preparation and analysis of the Closing Statement and the resolution of any disputes thereunder.

(e) If the Adjustment Amount, as finally determined pursuant to Section 2.16(c), is greater than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), by at least \$20 million, then Mallinckrodt shall pay or cause to be paid an amount in cash equal to the difference from the first dollar (i.e., without regard to the \$20 million threshold) to Covidien by wire transfer of immediately available funds to an account or accounts designated in writing by Covidien to Mallinckrodt. If the Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to Section 2.16(c), is less than the Target Adjustment Amount, as finally determined pursuant to the section 2.16(c) is to be made within five (5) Business Days of the date on which the Adjustment Amount and the Target Adjustment Amount have been finally determined pursuant to this Section 2.16(c).

## ARTICLE III THE DISTRIBUTION

### 3.1 The Distribution.

(a) Subject to the terms and conditions of this Agreement (including the conditions set out in <u>Section 3.3</u>), Covidien agrees that, on the Distribution Date and with effect from the Effective Time, it will effect the Distribution.

(b) Mallinckrodt agrees that the Mallinckrodt Spin Shares shall be allotted credited as fully paid up and free from any liens, charges and encumbrances whatsoever and shall have the rights described in Mallinckrodt's Memorandum and Articles of Association adopted pursuant to <u>Section 3.2(d)</u>.

(c) Notwithstanding any other provision of this Agreement, Covidien shall, in its sole and absolute discretion, determine the Distribution Date and all terms of the Distribution, including, without limitation, the form, structure and terms of any transaction(s) and/or offering(s) to effect the Distribution and the timing and conditions to the consummation of the Distribution. In addition, Covidien may, at any time and from time to time until the consummation of the Distribution, modify or change the terms of the Distribution, including, without limitation, by accelerating or delaying the timing of the consummation of all or part of the Distribution. For the avoidance of doubt, nothing in the foregoing shall in any way limit Covidien's right to terminate this Agreement or the Distribution as set forth in <u>Article X</u> or alter the consequences of any such termination from those specified in such Article.

(d) Mallinckrodt shall cooperate with Covidien to accomplish the Distribution and shall, at Covidien's direction, promptly take any and all actions necessary or desirable to effect the Distribution, including, without limitation, the registration under the Exchange Act of Mallinckrodt Ordinary Shares on an appropriate registration form or forms to be designated by Covidien. Covidien shall select any investment bank or manager in connection with the Distribution, as well as any financial printer, solicitation and/or exchange agent and financial, legal, accounting and other advisors for Covidien. Mallinckrodt and Covidien, as the case may be, will provide to the Agent any information required in order to complete the Distribution.

### 3.2 Actions Prior to the Distribution.

(a) Covidien and Mallinckrodt shall prepare and mail, prior to the Distribution Date, to the holders of Covidien Ordinary Shares, such information concerning Mallinckrodt, its business, operations and management, the Distribution and such other matters as Covidien shall reasonably determine and as may be required by Law. Covidien and Mallinckrodt will prepare, and Mallinckrodt will, to the extent required under applicable Law, file with the SEC any such documentation and any requisite no-action letters which Covidien determines are necessary or desirable to effectuate the Distribution and Covidien and Mallinckrodt shall each use its reasonable best efforts to obtain all necessary approvals from the SEC with respect thereto as soon as practicable.

(b) Covidien and Mallinckrodt shall take all such action as may be necessary or appropriate under the securities or blue sky laws of the United States (and any comparable Laws under any foreign jurisdiction) in connection with the Distribution.

(c) Mallinckrodt shall prepare and file, and shall use its reasonable best efforts to have approved, an application for the listing of the Mallinckrodt Spin Shares on the NYSE, subject to official notice of issuance.

(d) Covidien and Mallinckrodt shall take all such action as may be necessary or appropriate to provide for the adoption by Mallinckrodt of the Memorandum and Articles of Association in such form as may be reasonably determined by Covidien and Mallinckrodt.

(e) Covidien shall take all such action as may be necessary or appropriate so that, prior to the Distribution, the board of directors of each of the Mallinekrodt Holding Companies shall meet to consider, and if thought fit, approve: (i) the transfer of its entire issued share capital from Covidien to Mallinekrodt, conditional only upon the Distribution being effected; and (ii) the updating of all statutory registers to reflect such transfer.

### 3.3 Conditions to Distribution.

(a) The consummation of the Distribution will be subject to the satisfaction, or waiver by Covidien in its sole and absolute discretion, of the following conditions:

(i) The continued validity of a private letter ruling received by Covidien from the IRS (the "<u>IRS Ruling</u>") prior to the date hereof in connection with the transactions contemplated hereby, which shall continue in full force and effect and which shall not be modified or amended in any respect adversely affecting the intended tax-free treatment of the Distribution and certain related transactions.

(ii) The receipt of a tax opinion from Skadden, Arps, Slate, Meagher & Flom LLP, tax counsel to Covidien, dated as of the Distribution Date to be in form and substance satisfactory to Covidien in its sole and absolute discretion, which tax opinion shall rely on the effectiveness of the IRS Ruling, substantially to the effect that, for U.S. federal income tax purposes, the Distribution and certain related transactions, taken together, will qualify as transactions under Sections 355(a) and/or 368(a) of the Code.

(iii) The receipt of one or more opinions from Houlihan Lokey or another independent firm acceptable to Covidien in its sole and absolute discretion, confirming the solvency and financial viability of each of Covidien and Mallinckrodt and the satisfaction of any legal capital requirements in connection with the Separation, which opinions shall be in form and substance acceptable to Covidien in its sole and absolute discretion and which opinions shall not have been withdrawn or rescinded.

(iv) The Reorganization shall have been completed in accordance with the Plan of Reorganization.

(v) The financing contemplated to be obtained in connection with the Separation as described in Section 2.15 herein shall have been obtained.

(vi) Each of the Ancillary Agreements shall have been duly executed and delivered by the applicable parties thereto.

(vii) No order, injunction or decree issued by any Governmental Authority of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the Separation, the Distribution or any of the transactions related thereto shall be pending, threatened, issued or in effect.

(viii) The actions and filings necessary or appropriate under applicable U.S. federal, U.S. state or other securities Laws or blue sky Laws and the rules and regulations thereunder shall have been taken or made, and, where applicable, have become effective or been accepted.

(ix) All Governmental Approvals necessary to consummate the Separation, the Distribution and the transactions related thereto and to permit the operation of the Mallinckrodt Business after the Distribution Date shall have been obtained and be in full force and effect.

(x) The Separation and the Distribution shall not violate or result in a breach of applicable law or any material contract of Covidien or Mallinckrodt or any of their respective Subsidiaries.

(xi) The approval for listing on the NYSE for the Mallinckrodt Ordinary Shares to be delivered to the Covidien shareholders in the Distribution shall have been obtained, subject to official notice of issuance.

(xii) The SEC declaring effective the Form 10, with no order suspending the effectiveness of the Form 10 in effect and no proceedings for such purposes pending before or threatened by the SEC.

(xiii) The Information Statement and such other information concerning Mallinckrodt, its business, operations and management, the Distribution and such other matters as Covidien shall determine in its sole and absolute discretion and as may otherwise be required by law shall have been mailed to the Qualifying Covidien Shareholders.

(xiv) No other events or developments shall exist or shall have occurred that, in the judgment of the Covidien Board, in its sole and absolute discretion, makes it inadvisable to effect the Separation, the Distribution or the transactions related thereto.

(b) The foregoing conditions are for the sole benefit of Covidien and shall not give rise to or create any duty on the part of Covidien or the Covidien Board to waive or not waive such conditions or in any way limit Covidien's right to terminate this Agreement as set forth in <u>Article X</u> or alter the consequences of any such termination from those specified in such Article. Any determination made by the Covidien Board prior to the Distribution concerning the satisfaction or waiver of any or all of the conditions set forth in this <u>Section 3.3</u> shall be conclusive and binding on the Parties.

## 3.4 Certain Stockholder Matters.

(a) Subject to Section 3.3, on or prior to the Distribution Date, Mallinckrodt will deliver to the Agent for the benefit of Qualifying Covidien Shareholders all of the Mallinckrodt Ordinary Shares to be delivered in the Distribution, and shall cause the transfer agent for the Covidien Ordinary Shares to instruct the Agent to distribute on the Distribution Date the appropriate number of Mallinckrodt Ordinary Shares to each such holder or designated transferee or transferees of such holder by way of direct registration in book-entry form. Mallinckrodt will not issue paper stock certificates. The Distribution shall be effective at the Effective Time.

(b) Subject to Section 3.3, each Qualifying Covidien Shareholder will be entitled to receive in the Distribution a number of whole Mallinckrodt Ordinary Shares equal to the number of Covidien Ordinary Shares held by such holder on the Record Date multiplied by the Distribution Ratio and rounded down to the nearest whole number, with any residual fractional interest dealt with in accordance with paragraph (c) below.

(c) No fractional interests in Mallinckrodt Ordinary Shares will be distributed or credited to book-entry accounts in connection with the Distribution. As soon as practicable after the Distribution Date, Covidien shall direct the Agent to determine the fractional interests in Mallinckrodt Ordinary Shares which would have been allocable to each holder of record or beneficial owner of Covidien Ordinary Shares as of the Record Date had no rounding down occurred as part of the calculation in paragraph (b) above, to aggregate all such fractional interests into whole Mallinckrodt Ordinary Shares and to sell those whole shares in open market transactions (with the Agent, in its sole and absolute discretion, determining when, how and through which broker-dealer and at what price to make such sales), and to cause to be distributed to each such holder or for the benefit of each such beneficial

owner, in lieu of any fractional interest, such holder's or owner's ratable share of the proceeds of such sale, after deducting any Taxes required to be withheld and after deducting an amount equal to all brokerage charges, commissions and transfer Taxes attributed to such sale. Neither Covidien nor Mallinckrodt will be required to guarantee any minimum sale price for the relevant Mallinckrodt Ordinary Shares. Neither Covidien nor Mallinckrodt will be required to pay any interest on the proceeds from the sale of such Mallinckrodt Ordinary Shares.

(d) Until the Mallinckrodt Ordinary Shares are delivered in accordance with this <u>Section 3.4</u> and applicable Law, from and after the Effective Time, Mallinckrodt will regard the Persons entitled to receive such Mallinckrodt Ordinary Shares as record holders of Mallinckrodt Ordinary Shares in accordance with the terms of the Distribution without requiring any action on the part of such Persons. Mallinckrodt agrees that, subject to any transfers of such shares, from and after the Effective Time (i) each such holder will be entitled to receive all dividends payable on, and exercise voting rights and all other rights and privileges with respect to, the Mallinckrodt Ordinary Shares then held by such holder, and (ii) each such holder will be entitled, without any action on the part of such holder, to receive evidence of ownership of the Mallinckrodt Ordinary Shares then held by such holder.

(e) At the Effective Time, Mallinckrodt shall acquire and cancel, for no consideration, the Initial Share Capital.

## ARTICLE IV MUTUAL RELEASES; INDEMNIFICATION

### 4.1 Release of Pre-Distribution Claims.

(a) Except as provided in (i) Sections 4.1(c) and 4.1(d) and (ii) any Ancillary Agreement, effective as of the Effective Time, Mallinckrodt does hereby, for itself and each other member of the Mallinckrodt Group, their respective Affiliates (other than any member of the Covidien Group), successors and assigns, and all Persons who at any time prior to the Effective Time have been shareholders, directors, officers, agents or employees of any member of the Mallinckrodt Group, their respective Affiliates (other than any members of the Covidien Group, their respective Affiliates (other than any members of the Mallinckrodt Group, their respective Affiliates (other than any members of the Covidien Group, their respective Affiliates (other than any member of the Mallinckrodt Group), successors and assigns, and all Persons who at any time prior to the Effective Time have been shareholders, directors, officers, agents or employees of any member of the Covidien Group (in each case, in their respective capacities as such), remise, release and forever discharge Covidien Group (in each case, in their respective Time have been shareholders, directors, officers, agents or employees of any member of the Covidien Group (in each case, in their respective capacities as such), and their respective heirs, executors, administrators, successors and assigns, from any and all Liabilities whatsoever, whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, existing or arising from any acts or events occurring or failing to occur or alleged to have occurred or to have failed to occur or any conditions existing or alleged to have existed on or before the Effective Time, including in connection with the transactions and all other activities to implement the Separation and the Distribution.

(b) Except as provided in (i) Sections 4.1(c) and 4.1(d) and (ii) any Ancillary Agreement, effective as of the Effective Time, Covidien does hereby, for itself and each other member of the Covidien Group, their respective Affiliates (other than any member of the

Mallinckrodt Group), successors and assigns, and all Persons who at any time prior to the Effective Time have been shareholders, directors, officers, agents or employees of any member of the Covidien Group (in each case, in their respective capacities as such), remise, release and forever discharge Mallinckrodt, the respective members of the Mallinckrodt Group, their respective Affiliates (other than any member of the Covidien Group), successors and assigns, and all Persons who at any time prior to the Effective Time have been shareholders, directors, officers, agents or employees of any member of the Mallinckrodt Group (in each case, in their respective heirs, executors, administrators, successors and assigns, from any and all Liabilities whatsoever, whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, existing or arising from any acts or events occurring or failing to occur or alleged to have occurred or to have failed to occur or any conditions existing or alleged to have existed on or before the Effective Time, including in connection with the transactions and all other activities to implement the Separation and the Distribution.

(c) Nothing contained in Section 4.1(a) or (b) shall impair any right of any Person to enforce this Agreement, any Ancillary Agreement or any agreements, arrangements, commitments or understandings that are specified in Section 2.8(b) or the applicable Schedules thereto not to terminate as of the Effective Time, in each case in accordance with its terms. Nothing contained in Section 4.1(a) or (b) shall release any Person from:

(i) any Liability provided in or resulting from any agreement among any members of the Covidien Group or the Mallinckrodt Group that is specified in <u>Section 2.8(b)</u> or the applicable Schedules thereto as not to terminate as of the Effective Time, or any other Liability specified in such <u>Section 2.8(b)</u> as not to terminate as of the Effective Time;

(ii) any Liability, contingent or otherwise, assumed, transferred, assigned or allocated to the Group of which such Person is a member in accordance with, or any other Liability of any member of any Group under, this Agreement or any Ancillary Agreement;

(iii) any Liability for the sale, lease or receipt of goods, property or services purchased, obtained or used in the ordinary course of business by a member of one Group from a member of the other Group prior to the Effective Time;

(iv) any Liability for unpaid amounts for products or services or refunds owing on products or services due on a value-received basis for work done by a member of one Group at the request or on behalf of a member of the other Group;

(v) any Liability that the Parties may have with respect to indemnification or contribution pursuant to this Agreement, any Ancillary Agreement or otherwise for claims brought against the Parties by third Persons, which Liability shall be governed by the provisions of this <u>Article IV</u> and <u>Article V</u> and, if applicable, the appropriate provisions of the Ancillary Agreements; or

(vi) any Liability the release of which would result in the release of any Person other than a Person released pursuant to this Section 4.1.

In addition, nothing contained in <u>Section 4.1(a)</u> shall release any member of the Covidien Group from honoring its existing obligations to indemnify any director, officer or employee of Mallinckrodt who was a director, officer or employee of any member of the Covidien Group on or prior to the Distribution Date, to the extent such director, officer or employee is or becomes a named defendant in any Action with respect to which such director, officer or employee was entitled to such indemnification pursuant to then-existing obligations; it being understood that, if the underlying obligation giving rise to such Action is a Mallinckrodt Liability, Mallinckrodt shall indemnify, or procure from a Subsidiary the effective indemnification of, Covidien for such Liability (including Covidien's costs to indemnify the director, officer or employee) in accordance with the provisions set forth in this <u>Article IV</u>.

(d) Mallinckrodt shall not make, and shall not permit any member of the Mallinckrodt Group to make, any claim or demand, or commence any Action asserting any claim or demand, including any claim of contribution or any indemnification, against Covidien or any other member of the Covidien Group, or any other Person released pursuant to <u>Section 4.1(a)</u>, with respect to any Liabilities released pursuant to <u>Section 4.1(a)</u>. Covidien shall not make, and shall not permit any member of the Covidien Group to make, any claim or demand, or commence any Action asserting any claim or demand, including any claim of contribution or any indemnification asserting any claim or demand, including any claim of contribution or any indemnification against Mallinckrodt or any other member of the Mallinckrodt Group, or any other Person released pursuant to <u>Section 4.1(b)</u>, with respect to any Liabilities released pursuant to <u>Section 4.1(b)</u>.

(e) It is the intent of each of Covidien and Mallinckrodt, by virtue of the provisions of this <u>Section 4.1</u>, to provide for a full and complete release and discharge of all Liabilities existing or arising from all acts and events occurring or failing to occur or alleged to have occurred or to have failed to occur and all conditions existing or alleged to have existed on or before the Distribution Date, between or among Mallinckrodt or any other member of the Mallinckrodt Group, on the one hand, and Covidien or any other member of the Covidien Group, on the other hand (including any contractual agreements or arrangements existing or alleged to exist between or among any such members on or before the Distribution Date), except as expressly set forth in <u>Section 4.1(c)</u>. At any time, at the request of any other Party, each Party shall cause each member of its respective Group to execute and deliver releases reflecting the provisions hereof.

(f) Any breach of the provisions of this <u>Section 4.1</u> by either Covidien or Mallinckrodt shall entitle the other Party to recover reasonable fees and expenses of counsel in connection with such breach or any Action resulting from such breach.

4.2 Indemnification by Mallinckrodt. Except as provided in Section 4.4, Mallinckrodt shall, and shall cause the other members of the Mallinckrodt Group to, indemnify, defend and hold harmless Covidien, each member of the Covidien Group and each of their respective directors, officers, employees and agents, in each case in their respective capacities as such, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "Covidien Indemnitees"), from and against any and all Liabilities of the Covidien Indemnitees relating to, arising out of or resulting from, directly or indirectly, any of the following items (without duplication):

(a) the failure of Mallinckrodt or any other member of the Mallinckrodt Group or any other Person to pay, perform or otherwise promptly discharge any Mallinckrodt Liabilities or Mallinckrodt Contract in accordance with its respective terms, whether prior to, on or after the Distribution Date;

(b) the Mallinckrodt Business (except to the extent it relates to an Excluded Liability), any Mallinckrodt Liability or any Mallinckrodt Contract;

(c) any breach by Mallinckrodt or any other member of the Mallinckrodt Group of this Agreement or any of the Ancillary Agreements;

(d) except to the extent it relates to an Excluded Liability, any guarantee, indemnification obligation, letter of credit reimbursement obligation, surety, bond or other credit support agreement, arrangement, commitment or understanding for the benefit of any member of the Mallinckrodt Group by any member of the Covidien Group that survives following the Distribution; and

(e) any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, with respect to all information contained in the Form 10, the Information Statement, the preliminary or final offering memorandum with respect to the Senior Notes or any other Disclosure Document, in each case, as amended or supplemented.

4.3 <u>Indemnification by Covidien</u>. Covidien shall, and shall cause the other members of the Covidien Group to, indemnify, defend and hold harmless Mallinckrodt, each member of the Mallinckrodt Group and each of their respective directors, officers, employees or agents, in each case in their respective capacities as such, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "<u>Mallinckrodt Indemnitees</u>"), from and against any and all Liabilities of the Mallinckrodt Indemnitees relating to, arising out of or resulting from, directly or indirectly, any of the following items (without duplication):

(a) the failure of Covidien or any other member of the Covidien Group or any other Person to pay, perform or otherwise promptly discharge any Excluded Liabilities in accordance with their terms, whether prior to, on or after the Distribution Date;

(b) the Excluded Liabilities;

(c) the Covidien Business (except to the extent it relates to a Mallinckrodt Liability and other than the conduct of business, operations or activities for the benefit of the Mallinckrodt Group pursuant to any Ancillary Agreement); and

(d) any breach by Covidien or any other member of the Covidien Group of this Agreement or any of the Ancillary Agreements.

### 4.4 Indemnification Obligations Net of Insurance Proceeds and Other Amounts.

(a) The Parties intend that any Liability subject to indemnification or reimbursement pursuant to this <u>Article IV</u> or <u>Article V</u> will be net of Insurance Proceeds that actually reduce the amount of the Liability. Accordingly, the amount which any Party (an "<u>Indemnifying Party</u>") is required to pay to any Person entitled to indemnification hereunder (an "<u>Indemnitee</u>") will be reduced by any Insurance Proceeds theretofore actually recovered by or on behalf of the Indemnitee in respect of the related Liability. If an Indemnitee receives a payment (an "<u>Indemnity Payment</u>") required by this Agreement from an Indemnifying Party in respect of any Liability and subsequently receives Insurance Proceeds, then the Indemnitee will pay to the Indemnifying Party an amount equal to the excess of the Indemnity Payment received over the amount of the Indemnity Payment that would have been due if the Insurance Proceeds had been received, realized or recovered before the Indemnity Payment was made.

(b) An insurer who would otherwise be obligated to pay any claim shall not be relieved of the responsibility with respect thereto or, solely by virtue of the indemnification provisions hereof, have any subrogation rights with respect thereto, it being expressly understood and agreed that no insurer or any other third party shall be entitled to a "windfall" (*i.e.*, a benefit they would not be entitled to receive in the absence of the indemnification provisions) by virtue of the indemnification provisions hereof. Nothing contained in this Agreement or any Ancillary Agreement shall obligate any member of any Group to seek to collect or recover any Insurance Proceeds.

(c) The Parties intend that any indemnification or reimbursement payment in respect of a Liability pursuant to this <u>Article IV</u> or <u>Article V</u> shall be (i) reduced by the Tax Benefit Amount (as defined in the Tax Matters Agreement), if any, realized by such indemnified or reimbursed Person as a result of such payment and (ii) increased so that the amount of such payment, reduced by the amount of all Income Taxes (as defined in the Tax Matters Agreement) payable with respect to the receipt thereof (but taking into account, for the avoidance of doubt, all correlative Tax Benefit Amounts resulting from the payment of such Income Taxes), shall equal the amount of the payment which the Person receiving such payment would otherwise be entitled to receive pursuant to this Agreement.

#### 4.5 Procedures for Indemnification of Third-Party Claims.

(a) If an Indemnitee shall receive notice or otherwise learn of the assertion by a Person (including any Governmental Authority) who is not a member of the Covidien Group or the Mallinckrodt Group of any claim or of the commencement by any such Person of any Action (collectively, a "<u>Third-Party Claim</u>") with respect to which an Indemnifying Party may be obligated to provide indemnification to such Indemnitee pursuant to <u>Section 4.2</u> or <u>4.3</u>, or any other Section of this Agreement or any Ancillary Agreement, such Indemnitee shall give such Indemnifying Party written notice thereof as promptly as practicable (and no later than thirty (30) days or sooner, if the nature of the Third-Party Claim so requires) after becoming aware of such Third-Party Claim. Any such notice shall describe the Third-Party Claim in reasonable detail and include copies of all notices and documents (including court papers) received by the Indemnitee relating to the Third-Party Claim. Notwithstanding the foregoing, the failure of an

Indemnitee to provide notice in accordance with this <u>Section 4.5(a)</u> shall not relieve an Indemnifying Party of its indemnification obligations under this Agreement, except to the extent to which the Indemnifying Party is actually prejudiced by the Indemnitee's failure to provide notice in accordance with this <u>Section 4.5(a)</u>.

(b) An Indemnifying Party may elect to defend (and, unless the Indemnifying Party has specified any reservations or exceptions, to seek to settle or compromise), at such Indemnifying Party's own expense and by such Indemnifying Party's own counsel, any Third-Party Claim. Within thirty (30) days after the receipt of notice from an Indemnitee in accordance with <u>Section 4.5(a)</u> (or sooner, if the nature of such Third-Party Claim so requires), the Indemnifying Party shall notify the Indemnitee of its election whether the Indemnifying Party will assume responsibility for defending such Third-Party Claim, which election shall specify any reservations or exceptions. After notice from an Indemnifying Party to an Indemnitee of its election to assume the defense of a Third-Party Claim, such Indemnitee shall have the right to employ separate counsel and to participate in (but not control) the defense, compromise or settlement thereof, but the fees and expenses of such counsel shall be the expense of such Indemnitee except as set forth in the next sentence.

(c) In the event that the Indemnifying Party has elected to assume the defense of the Third-Party Claim but has specified, and continues to assert, any reservations or exceptions in such notice, then, in any such case, the reasonable fees and expenses of one (1) separate counsel for all Indemnitees shall be borne by the Indemnifying Party.

(d) If an Indemnifying Party elects not to assume responsibility for defending a Third-Party Claim, or fails to notify an Indemnitee of its election as provided in <u>Section 4.5(b)</u>, such Indemnitee may defend such Third-Party Claim at the cost and expense of the Indemnifying Party.

(e) Unless the Indemnifying Party has failed to assume the defense of the Third-Party Claim in accordance with the terms of this Agreement, no Indemnitee may settle or compromise any Third-Party Claim without the consent of the Indemnifying Party.

(f) In the case of a Third-Party Claim, no Indemnifying Party shall consent to entry of any judgment or enter into any settlement of the Third-Party Claim without the consent of the Indemnitee if the effect thereof is to permit any injunction, declaratory judgment, other order or other non-monetary relief to be entered, directly or indirectly against any Indemnitee.

(g) For the avoidance of doubt, the provisions of this <u>Article IV</u> shall apply to Third-Party Claims that have already been asserted as well as Third-Party Claims asserted after the date hereof, and there shall be no requirement under this <u>Section 4.5</u> to give notice with respect to any Third-Party Claims that have already been asserted as of the Effective Time.

### 4.6 Additional Matters.

(a) Indemnification payments in respect of any Liabilities for which an Indemnitee is entitled to indemnification under this <u>Article IV</u> shall be paid by the Indemnifying Party to the Indemnitee as such Liabilities are incurred upon demand by the Indemnitee,

including reasonably satisfactory documentation setting forth the basis for the amount of such indemnification payment, including documentation with respect to calculations made and consideration of any Insurance Proceeds that actually reduce the amount of such Liabilities. The indemnity agreements contained in this <u>Article IV</u> shall remain operative and in full force and effect, regardless of (i) any investigation made by or on behalf of any Indemnitee, (ii) the knowledge by the Indemnitee of Liabilities for which it might be entitled to indemnification hereunder and (iii) any termination of this Agreement.

(b) Any claim on account of a Liability which does not result from a Third-Party Claim shall be asserted by written notice given by the Indemnitee to the related Indemnifying Party. Such Indemnifying Party shall have a period of thirty (30) days after the receipt of such notice within which to respond thereto. If such Indemnifying Party does not respond within such thirty (30)-day period, such Indemnifying Party shall be deemed to have refused to accept responsibility to make payment. If such Indemnifying Party does not respond within such thirty (30)-day period or rejects such claim in whole or in part, such Indemnitee shall be free to pursue such remedies as may be available to such party as contemplated by this Agreement and the Ancillary Agreements.

(c) In the event of payment by or on behalf of any Indemnifying Party to any Indemnitee in connection with any Third-Party Claim, such Indemnifying Party shall be subrogated to and shall stand in the place of such Indemnitee as to any events or circumstances in respect of which such Indemnitee may have any right, defense or claim relating to such Third-Party Claim against any claimant or plaintiff asserting such Third-Party Claim or against any other Person. Such Indemnitee shall cooperate with such Indemnifying Party in a reasonable manner, and at the cost and expense of such Indemnifying Party, in prosecuting any subrogated right, defense or claim.

(d) In the event of an Action in which the Indemnifying Party is not a named defendant, if either the Indemnife or Indemnifying Party shall so request, the Parties shall endeavor to substitute the Indemnifying Party for the named defendant. If such substitution or addition cannot be achieved for any reason or is not requested, the named defendant shall allow the Indemnifying Party to manage the Action as set forth in this <u>Section 4.6</u>, and the Indemnifying Party shall fully indemnify the named defendant against all costs of defending the Action (including court costs, sanctions imposed by a court, attorneys' fees, experts fees and all other external expenses), the costs of any judgment or settlement, and the cost of any interest or penalties relating to any judgment or settlement.

(e) For all claims as to which indemnification or contribution is provided under this <u>Article IV</u>, other than Third-Party Claims (as to which <u>Section 4.5</u> shall apply), the reasonable fees and expenses of counsel to the Indemnite for the enforcement of the indemnity obligations shall be borne by the Indemnifying Party.

4.7 <u>Remedies Cumulative</u>. The remedies provided in this <u>Article IV</u> shall be cumulative and, subject to the provisions of <u>Article VIII</u>, shall not preclude assertion by any Indemnitee of any other rights or the seeking of any and all other remedies against any Indemnifying Party.

4.8 <u>Survival of Indemnities</u>. The rights and obligations of each of Covidien and Mallinckrodt and their respective Indemnitees under this <u>Article IV</u> shall survive the sale or other transfer by any Party of any Assets or businesses or the assignment by it of any Liabilities.

4.9 Guarantees, Letters of Credit or Other Obligations. In furtherance of, and not in limitation of, the obligations set forth in Section 2.6 and this Article IV:

(a) On or prior to the Distribution Date or as soon as practicable thereafter, Mallinckrodt shall (with the reasonable cooperation of the applicable member(s) of the Covidien Group) use its reasonable best efforts to have any member(s) of the Covidien Group removed as guarantor of or obligor for any Mallinckrodt Liability to the extent that they relate to Mallinckrodt Liabilities.

(b) On or prior to the Distribution Date, to the extent required to obtain a release from a guarantee or letter of credit, including the guarantees listed on <u>Schedule 4.9(b)</u> (a "<u>Guarantee Release</u>"), of any member of the Covidien Group, Mallinckrodt shall execute a guarantee agreement in the form of the existing guarantee or letter of credit, as applicable, or such other form as is agreed to by the relevant parties to such guarantee agreement or letter of credit, except to the extent that such existing guarantee or letter of credit contains representations, covenants or other terms or provisions either (i) with which Mallinckrodt would be reasonably unable to comply or (ii) which would be reasonably expected to be breached.

(c) If the Parties are unable to obtain, or to cause to be obtained, any such required removal as set forth in clauses (a) and (b) of this Section 4.9, (i) Mallinckrodt shall, and shall cause the other members of the Mallinckrodt Group to, indemnify, defend and hold harmless each of the Covidien Indemnitees for any Liability arising from or relating to such guarantee and shall, as agent or subcontractor for the applicable Covidien Group guarantor or obligor, pay, perform and discharge fully all the obligations or other Liabilities of such guarantor or obligor thereunder, and (ii) Mallinckrodt shall not, and shall cause the other members of the Mallinckrodt Group not to, agree to renew or extend the term of, increase any obligations under, or transfer to a third Person, any loan, guarantee, letter of credit, lease, contract or other obligation for which a member of the Covidien Group is or may be liable unless all obligations of the members of the Covidien Group with respect thereto are thereupon terminated by documentation satisfactory in form and substance to Covidien in its sole and absolute discretion.

4.10 <u>Contribution</u>. If the indemnification provided for in <u>Section 4.2</u> is unavailable to, or insufficient to hold harmless, the Covidien Indemnitees under this <u>Article IV</u>, then Mallinckrodt shall, or shall cause the other members of the Mallinckrodt Group to, contribute to the amount paid or payable to such Covidien Indemnitee as a result of such Liabilities (or actions in respect thereof).

4.11 <u>Taxes</u>. The provisions of this Agreement, including this <u>Article IV</u>, shall not apply to any matters relating to Taxes to the extent such matters are addressed in the Tax Matters Agreement or the Employee Matters Agreement. In the case of any conflict between this Agreement and either the Tax Matters Agreement or the Employee Matters Agreement in relation to any matters related to Taxes, the Tax Matters Agreement or the Employee Matters Agreement, as applicable, shall prevail.

### ARTICLE V INSURANCE

5.1 <u>Cooperation</u>. Covidien and Mallinckrodt agree to use their respective reasonable best efforts to cooperate in good faith to arrange insurance coverage for Mallinckrodt to be effective no later than the Effective Time. In no event shall Covidien, any other member of the Covidien Group or any Covidien Indemnitee have any liability or obligation whatsoever to any member of the Mallinckrodt Group in the event that any insurance policy or other contract or policy of insurance shall be terminated or otherwise cease to be in effect for any reason, shall be unavailable or inadequate to cover any Liability of any member of the Mallinckrodt Group for any reason whatsoever or shall not be renewed or extended beyond the current expiration date. Covidien and Mallinckrodt further agree to use their respective reasonable best efforts to cooperate with each other and the other members of their respective Groups with respect to the various insurance matters contemplated by this Agreement and to provide assistance in accessing coverage under any Shared Policy or Mallinckrodt Policy, as applicable, in a manner contemplated by this Agreement.

5.2 Policies and Rights Included Within Assets. The Mallinckrodt Assets shall include any and all rights of an insured party under each of the Shared Policies, subject to the terms of such Shared Policies and any limitations or obligations of Mallinckrodt contemplated by this <u>Article V</u>, specifically including rights of indemnity and the right to be defended by or at the expense of the insurer, with respect to all actual, contingent or alleged wrongful acts, occurrences, events, Actions, proceedings, injuries, Losses, Liabilities, damages and expenses which occurred or are alleged to have occurred, in whole or in part, or were incurred or claimed to have been incurred prior to the Effective Time by any Party in or in connection with the conduct of the Mallinckrodt Business, and which actual or alleged wrongful acts, occurrences, events, Actions, proceedings, injuries, Losses, Liabilities, damages and expenses, Liabilities, damages and expenses may arise out of an insured or insurable occurrence or wrongful act under one or more of such Shared Policies; <u>provided</u>, <u>however</u>, that nothing in this clause shall be deemed to constitute (or to reflect) an assignment of such Shared Policies, or any of them, to Mallinckrodt. Notwithstanding the foregoing, with regard to the Mallinckrodt Assets in respect of any claims made Policy that is not put into run-off as further described below in <u>Section 5.3</u>, nothing in this Agreement is intended to provide coverage for alleged wrongful acts, occurrences, events, Actions, proceedings, injuries, Losses, Liabilities, damages and expenses which occurred or are alleged to have occurred or are alleged to have occurred, in whole or in part, prior to the Effective Time and are covered under a claims made policy form, that were not reported to Covidien's Director of Risk Management prior to the Effective Time.

## 5.3 Claims Made Tail Policies.

(a) The claims made tail policies provided for in this Section 5.3 will solely provide coverage for any claim arising from any wrongful act occurring, in whole or in part, prior to the Effective Time.

(b) Subject to prevailing market conditions and underwriting, Covidien shall purchase Directors and Officers Liability Insurance Policies having total limits of \$250 million, consisting of \$200 million of traditional Side A/B/C coverage and \$50 million of Side A DIC coverage and having a policy period incepting at the Effective Time, or the expiration date of the current Covidien Directors and Officers Liability Insurance Policies, whichever date is earlier, and ending on a date that is six years after the Effective Time ("<u>D&O Tail Policies</u>"). The premium for the D&O Tail Policies shall be pre-paid for the full six-year term of the D&O Tail Policies. Such D&O Tail Policies shall cover Covidien and Mallinckrodt and the insured persons thereof and shall have material terms and conditions no less favorable than those contained in the Policies comprising the Covidien Directors and Officers Liability Insurance program incepting on June 29, 2012, except for the policy period, premium and provisions excluding coverage for wrongful acts, errors or omissions, post-dating the Effective Time. Covidien (i) shall provide Mallinckrodt with copies of the D&O Tail Policies upon Mallinckrodt's written request but no sooner than a reasonable time after such Policies are issued and (ii) shall not amend the terms of, nor cancel or permit cancellation of, any such Policies without ninety (90) days prior written notice to Mallinckrodt.

(c) Subject to prevailing market conditions and underwriting, Covidien shall purchase Fiduciary Liability Insurance Policies having total limits of \$50 million and having a policy period incepting at the Effective Time, or the expiration date of the current Covidien Fiduciary Liability Insurance Policies, whichever date is earlier, and ending on a date that is six years after the Effective Time ("Fiduciary Tail Policies"). The premium for the Fiduciary Tail Policies shall be pre-paid for the full six-year term of the Fiduciary Tail Policies. Such Fiduciary Tail Policies shall cover Covidien and Mallinckrodt and the insured persons thereof and shall have material terms and conditions no less favorable than those contained in the Policies comprising the Covidien Fiduciary Liability Insurance program incepting on October 1, 2012, except for the policy period, premium and provisions excluding coverage for wrongful acts, errors and omissions, post-dating the Effective Time. Covidien (i) shall provide Mallinckrodt with copies of the Fiduciary Tail Policies upon Mallinckrodt's written request but no sooner than a reasonable time after such Policies are issued and (ii) shall not amend the terms of, nor cancel or permit cancellation of, any such Policies without ninety (90) days prior written notice to Mallinckrodt.

(d) Subject to prevailing market conditions and underwriting, to the extent that Covidien is unable prior to the Effective Time to obtain any of the Policies as provided for in paragraphs (a), (b) and (c) of this Section 5.3, then, with respect to suits or claims based on wrongful acts, errors or omissions on or before the Effective Time, Covidien shall use commercially reasonable efforts to secure alternative insurance arrangements on the applicable standalone insurance policies for Mallinckrodt to provide benefits on terms and conditions (including policy limits) in favor of Mallinckrodt and the insured persons thereof no less favorable than the benefits (including policy limits) that were to be afforded by the policies described in paragraphs (a), (b) and (c) of this Section 5.3. With respect to such alternative insurance arrangements, Covidien and Mallinckrodt shall be responsible for their own costs under their applicable standalone insurance policies. Covidien shall not under any circumstances purchase any such alternative coverage containing an exclusion for suits or claims based on wrongful acts, errors or omissions up to and including the Effective Time to the extent such exclusion would preclude coverage for Mallinckrodt and/or the insured persons thereof, but would not preclude coverage for Covidien and/or the insured persons thereof.

### 5.4 Occurrence Based Policies.

(a) With respect to Shared Policies of workers' compensation, automobile liability and general liability insurance, for suits or claims that are filed or made either before, on or after the Effective Time, with respect to occurrences which took place, in whole or in part, prior to the Effective Time and for which Old Colony State Insurance Company funds claim payments and claim adjustment expenses, Mallinckrodt shall pay to Old Colony State Insurance Company a one-time separation payment equal to [-]. Payment by Mallinckrodt will be due upon demand by Old Colony State Insurance Company.

(b) With respect to all other occurrence based Shared Policies, for suits or claims relating to the Mallinckrodt Business that are filed or made based upon occurrences that occurred or are alleged to have occurred in whole or in part prior to the Effective Time, Mallinckrodt shall be responsible for bearing the full amount of the deductible, self-insured retention and/or any claims, costs and expenses that are not covered under such insurance policies including that portion of any premium adjustments, tax, assessment or similar regulatory surcharges that relates to claims based on occurrences that predate the Effective Time.

### 5.5 Administration; Other Matters.

(a) <u>Administration</u>. Except as otherwise provided in <u>Section 5.4</u> hereof, from and after the Effective Time, Covidien shall have responsibility for and shall have the exclusive right to control (i) Insurance Administration of the Shared Policies and the Mallinckrodt Policies and (ii) subject to <u>Section 5.5</u>. Claims Administration under the Shared Policies and Mallinckrodt Policies; <u>provided</u>, that the retention of such responsibilities by Covidien is in no way intended to limit, inhibit or preclude any right to insurance coverage for any Insured Claim of a named insured under such Policies as contemplated by the terms of this Agreement; <u>provided</u>, <u>further</u>, that Covidien's retention of the administrative responsibilities for the Shared Policies and Mallinckrodt Policies shall not relieve the Party submitting any Insured Claim of the primary responsibility for reporting such Insured Claim accurately, completely and in a timely manner or of such Party's authority to settle any such Insured Claim within any period or amount permitted or required by the relevant Policy; <u>provided</u>, <u>further</u>, that notwithstanding the foregoing, with respect to Mallinckrodt Liabilities, Mallinckrodt shall have responsibility for reporting to excess insurance carriers of any Losses or claims which may cause the per-occurrence, per-claim or aggregate limits of any Shared Policy to be exceeded. Covidien may discharge its administrative responsibilities under this <u>Section 5.5</u> by contracting for the provision of services by independent parties. Each of the applicable Parties shall pay any costs relating to defending its respective Insured Claims Administration and Insurance Administration are not covered under such Policies. Each of the Parties shall be responsible for obtaining or reviewing the appropriateness of releases upon settlement of its respective Insured Claims under Shared Policies. Covidien shall retain the exclusive right to amend, modify or waive any rights under the Shared Policies and Mallinckrodt Policies, notwit

Mallinckrodt Policies apply to any Mallinckrodt Liabilities and/or claims Mallinckrodt has made or could make in the future, and no member of the Mallinckrodt Group shall, without the prior written consent of Covidien, erode, exhaust, settle, release, commute, buy-back or otherwise resolve disputes with any insurer with respect to any of the Shared Policies or Mallinckrodt Policies, or amend, modify or waive any rights under any such Shared Policies or Mallinckrodt Policies; provided that to the extent any such amendment, modification or waiver adversely affects the rights of any member of the Mallinckrodt Group with respect to coverage for any Mallinckrodt Liabilities and/or claims Mallinckrodt has made, then Covidien shall use its commercially reasonable efforts to provide advance written notice of any such amendment, modification or waiver to Mallinckrodt. Mallinckrodt shall cooperate with Covidien and share such information at Mallinckrodt's cost as is reasonably necessary in order to permit Covidien to manage and conduct its insurance matters as it deems appropriate. Neither Covidien nor any of its Affiliates' shall have any obligation to secure extended reporting for any claims under any of Covidien's or its Affiliates' liability Policies for any acts or omissions by any member of the Mallinckrodt Group incurred prior to the Effective Time. To the extent reasonably practicable, Covidien will notify Mallinckrodt ta least ten (10) days prior to terminating or finalizing any buy-back of any rights under any Shared Policy or Mallinckrodt Policy with respect to which Mallinckrodt has asserted a claim or given written notice to Covidien that it proposes to submit a claim.

(b) Exceeding Policy Limits. Where Mallinckrodt Liabilities are specifically covered under the same Shared Policy for occurrences, acts or events prior to the Effective Time, then Mallinckrodt may claim coverage for Insured Claims under such Shared Policy as and to the extent that such insurance is available up to the full extent of the applicable limits of liability of such Shared Policy (and may receive any Insurance Proceeds with respect thereto as contemplated by Section 5.3, Section 5.4 or Section 5.5, Covidien and Mallinckrodt shall not be liable to one another for claims not reimbursed by insurers for any reason not within the control of Covidien or Mallinckrodt, as the case may be, including coinsurance provisions, deductibles, quota share deductibles, self-insured retentions, bankruptcy or insolvency of an insurance carrier, Shared Policy limitations or restrictions, any coverage disputes, any failure to timely claim by Covidien or Mallinckrodt or any defect in such claim or its processing. For the avoidance of doubt, with respect to the Mallinckrodt Liabilities, Mallinckrodt shall exclusively bear (and neither Covidien nor any member of the Covidien Group shall have any obligation to repay or reimburse Mallinckrodt or members of the Mallinckrodt Group for) and shall be liable for all uninsured, uncovered, unavailable or uncollectible amounts of all such claims made by Mallinckrodt or any member of the Mallinckrodt Group under the Shared Policies as provided for in this Article V. Mallinckrodt and members of the Mallinckrodt Group shall indemnify, hold harmless and reimburse Covidien and members of the Covidien Group for any coinsurance provisions, deductibles, quota share deductibles, self-insured retentions, fees and expenses incurred by Covidien or members of the Covidien Group to the extent resulting from any such access to, or any claims made by Mallinckrodt or members of the Mallinckrodt Group under, any Shared Policy insurance provided pursuant to this Article V, including any indemnity payments, settlements, judgments, legal fees and allocated claims expenses and claim-handling fees, whether such claims are made by Mallinckrodt, its employees or third Persons. It is expressly understood that the foregoing shall not limit any Party's liability to the other Party for indemnification pursuant to Article IV.

(c) <u>Allocation of Insurance Proceeds</u>. Except as otherwise provided in <u>Section 5.4</u>, Insurance Proceeds received with respect to suits, occurrences, claims, costs and expenses covered under the Shared Policies shall be paid to Covidien with respect to Excluded Liabilities and to Mallinckrodt with respect to Mallinckrodt Liabilities. In the event that the

aggregate limits on any Shared Policies are exhausted by the payment of Insured Claims by the relevant parties, such parties agree to allocate the Insurance Proceeds received thereunder based upon their respective percentage of the total insured claim or claims which were covered under such Shared Policy (their "<u>allocable portion of Insurance Proceeds</u>"), and any Party who has received Insurance Proceeds in excess of such Party's allocable portion of Insurance Proceeds shall pay to the other Party the appropriate amount so that each Party will have received its allocable portion of Insurance Proceeds. Each of the Parties agrees to use their respective reasonable best efforts to maximize available coverage under those Shared Policies applicable to it for the benefit of both Parties, and to take all commercially reasonable steps to recover from all other responsible parties (except the other Party hereto) in respect of an Insured Claim to the extent coverage limits under a Shared Policy have been exceeded or would be exceeded as a result of such Insured Claim.

(d) <u>Allocation of Aggregate Deductibles</u>. In the event that both Parties have insured claims under any Shared Policy for which an aggregate deductible is payable, the Parties agree that the aggregate amount of the total deductible paid shall be borne by the Parties in the same proportion to which the Insurance Proceeds received by each such Party bears to the total Insurance Proceeds received under the applicable Shared Policy (their "<u>allocable share of the deductible</u>"), and any Party who has paid more than its allocable share of the deductible shall be entitled to receive from the other Party an appropriate amount such that each Party will only have to bear its allocable share of the deductible.

(e) <u>Mallinckrodt Policies</u>. Notwithstanding anything to the contrary herein, (i) Covidien shall have the exclusive right to assert claims under and control all proceedings relating to the Mallinckrodt Policies, including discussions or negotiations with insurers and any pending or future Actions relating to such claims (including with respect to settlement thereof), (ii) Mallinckrodt shall have no right to any Insurance Proceeds under any Mallinckrodt Policy except (A) any amounts payable to Mallinckrodt pursuant to the cost-sharing agreements listed on <u>Schedule 5.5(e)</u> and (B) amounts payable with respect to claims for any Mallinckrodt Liabilities under any workers' compensation policies that are included in the Mallinckrodt Policies, and (iii) all costs and other Liabilities relating to such Policies, claims or Actions, shall be shared, after reduction from such recovery of all legal fees and other out-of-pocket expenses incurred by Covidien to date of such recovery, 70% by Covidien and 30% by Mallinckrodt, provided that such allocation of costs and Insurance Proceeds shall not apply to (x) any amounts payable to Mallinckrodt pursuant to the preceding clause (ii) or (y) any amounts payable with respect to claims for any Excluded Liabilities under any workers' compensation policies that are included in the Mallinckrodt Policies.

(f) <u>Old Colony Policies</u>. Notwithstanding anything to the contrary herein, any and all claims in respect of the Policies provided by Old Colony State Insurance Company shall be administered, paid, accounted for and otherwise managed as provided on <u>Schedule 5.5(f)</u>.

5.6 <u>Agreement for Waiver of Conflict and Shared Defense</u>. In the event that Insured Claims of more than one of the Parties exist relating to the same occurrence, the relevant Party (on behalf of itself and the other members of its respective Group) shall jointly defend and waive any conflict of interest necessary to the conduct of the joint defense. Nothing in this <u>Article V</u> shall be construed to limit or otherwise alter in any way the obligations of the Parties, including those created by this Agreement, by operation of Law or otherwise.

### ARTICLE VI CERTAIN OTHER MATTERS

6.1 <u>Late Payments</u>. Except as expressly provided to the contrary in this Agreement or in any Ancillary Agreement, any amount not paid when due pursuant to this Agreement or any Ancillary Agreement (and any amounts billed or otherwise invoiced or demanded and properly payable that are not paid within thirty (30) days of such bill, invoice or other demand) shall accrue interest at a rate per annum equal to the Prime Rate plus five percent (5%).

6.2 <u>Grant of License for Mallinckrodt Name</u>. Subject to the terms, conditions and limitations contained herein, Mallinckrodt, on its own behalf and on behalf of the other members of the Mallinckrodt Group, hereby grants to the members of the Covidien Group listed on <u>Schedule 6.2</u> a non-exclusive, worldwide, irrevocable, royalty-free license to use and display the name "Mallinckrodt" in their legal names and for related incidental uses following the Effective Time (e.g., in payroll checks, regulatory filings and bank accounts). The members of the Covidien Group's use of the "Mallinckrodt" name is limited to incidental, non-substantive use, such as use for payroll, banking, regulatory and other similar purposes. In no event shall the members of the Covidien Group create, reproduce or arrange for the creation or reproduction of the "Mallinckrodt" name or use the "Mallinckrodt" name in any advertising or marketing materials.

## ARTICLE VII

## EXCHANGE OF INFORMATION; CONFIDENTIALITY

7.1 <u>Agreement for Exchange of Information; Archives</u>. Subject to <u>Section 7.7</u> and any other applicable confidentiality obligations, each of Covidien and Mallinckrodt, on behalf of its respective Group, agrees to provide, or cause to be provided, to the other Group, at any time before, on or after the Distribution Date, as soon as reasonably practicable after written request therefor, any Information in the possession or under the control of such respective Group which the requesting Party reasonably needs (i) to comply with reporting, disclosure, filing or other requirements imposed on the requesting Party (including under applicable securities or Tax Laws) by a Governmental Authority having jurisdiction over the requesting Party, (ii) for use in any other judicial, regulatory, administrative, Tax or other proceeding or in order to satisfy audit, accounting, claims, regulatory, litigation, Tax or other similar requirements, in each case other than claims or allegations that one Party to this Agreement has against the other, or (iii) subject to the foregoing clause (ii), to comply with its obligations under this Agreement or any Ancillary Agreement; provided, however, that, in the event that any Party determines that any such provision of Information could be commercially detrimental, violate any Law or agreement, or waive any privilege otherwise available under applicable Law, including the attorney-client privilege, the Parties shall take all reasonable measures to permit the compliance with such obligations in a manner that avoids any such harm or consequence. For the avoidance of doubt, the rights and obligations of any Party described in this <u>Section 7.1</u> with respect to the sharing of Information related to Taxes are subject to the rights and obligations described in the Tax Matters Agreement.

7.2 <u>Ownership of Information</u>. Any Information owned by one Group that is provided to a requesting Party pursuant to <u>Section 7.1</u> or <u>Section 7.6</u> shall be deemed to remain the property of the providing Party. Unless specifically set forth herein, nothing contained in this Agreement shall be construed as granting or conferring rights of license or otherwise in any such Information.

7.3 <u>Compensation for Providing Information</u>. The Party requesting Information agrees to reimburse the other Party for the reasonable out-of-pocket costs, if any, of creating, gathering and copying such Information, to the extent that such costs are incurred for the benefit of the requesting Party.

7.4 <u>Record Retention</u>. To facilitate the possible exchange of Information pursuant to this <u>Article VII</u> and other provisions of this Agreement after the Effective Time, the Parties agree to use their reasonable best efforts to retain all Information in their respective possession or control on the Distribution Date in accordance with the policies of Covidien as in effect on the Distribution Date or such other policies as may be adopted by Covidien after the Effective Time (<u>provided</u>, in the case of Mallinckrodt, that Covidien notifies Mallinckrodt of any such material change). No Party will destroy, or permit any of its Subsidiaries to destroy, any Information which the other Party may have the right to obtain pursuant to this Agreement prior to the end of the retention period set forth in such policies without first notifying the other Party of the proposed destruction and giving the other Party the opportunity to take possession of such information prior to such destruction; <u>provided</u>, <u>however</u>, that in the case of any Information relating to Taxes, employee benefits or Environmental Liabilities, such retention period shall be extended to the expiration of the applicable statute of limitations (giving effect to any extensions thereof). Notwithstanding the foregoing, <u>Section 8.01</u> of the Tax Matters Agreement shall govern the retention of Tax Records (as defined in the Tax Matters Agreement).

7.5 <u>Limitations of Liability</u>. No Party shall have any liability to any other Party in the event that any Information exchanged or provided pursuant to this Agreement which is an estimate or forecast, or which is based on an estimate or forecast, is found to be inaccurate in the absence of willful misconduct by the Party providing such Information. No Party shall have any liability to any other Party if any Information is destroyed after reasonable best efforts by such Party to comply with the provisions of <u>Section 7.4</u>.

### 7.6 Production of Witnesses; Records; Cooperation.

(a) After the Effective Time, except in the case of an adversarial Action by one Party against another Party, each Party shall use its commercially reasonable efforts to make available to the other Party, upon written request, the former, current and future directors, officers, employees, other personnel and agents of the members of its respective Group as witnesses and any books, records or other documents within its control or which it otherwise has the ability to make available, to the extent that any such person (giving consideration to business demands of such directors, officers, employees, other personnel and agents) or books, records or other documents may reasonably be required in connection with any Action in which the requesting Party may from time to time be involved, regardless of whether such Action is a matter with respect to which indemnification may be sought hereunder. Without limiting any indemnification obligations of the non-requesting Party pursuant to <u>Article IV</u>, the requesting

Party shall bear all costs and expenses in connection therewith. For the avoidance of doubt, the rights and obligations of any Party described in this <u>Section 7.6</u> are subject to the rights and obligations described in the Tax Matters Agreement.

(b) If an Indemnifying Party chooses to defend or to seek to compromise or settle any Third-Party Claim, the other party shall make available to such Indemnifying Party, upon written request, the former, current and future directors, officers, employees, other personnel and agents of the members of its respective Group as witnesses and any books, records or other documents within its control or which it otherwise has the ability to make available, to the extent that any such person (giving consideration to business demands of such directors, officers, employees, other personnel and agents) or books, records or other documents may reasonably be required in connection with such defense, settlement or compromise, or such prosecution, evaluation or pursuit, as the case may be, and shall otherwise cooperate in such defense, settlement or compromise, or such prosecution or pursuit, as the case may be.

(c) Without limiting the foregoing, the Parties shall cooperate and consult to the extent reasonably necessary with respect to any Actions.

(d) Without limiting any provision of this <u>Section 7.6</u>, each of the Parties agrees to cooperate, and to cause each member of its respective Group to cooperate, with each other in the defense of any infringement or similar claim with respect to any Intellectual Property and shall not claim to acknowledge, or permit any member of its respective Group to claim to acknowledge, the validity or infringing use of any Intellectual Property of a third Person in a manner that would hamper or undermine the defense of such infringement or similar claim.

(e) The obligation of the Parties to provide witnesses pursuant to this <u>Section 7.6</u> is intended to be interpreted in a manner so as to facilitate cooperation and shall include the obligation to provide as witnesses inventors and other officers (subject to the exception set forth in the first sentence of <u>Section 7.6(a)</u>).

(f) In connection with any matter contemplated by this <u>Section 7.6</u>, the Parties will enter into a mutually acceptable joint defense agreement so as to maintain to the extent practicable any applicable attorney-client privilege or work product immunity of any member of any Group.

### 7.7 Confidentiality.

(a) Subject to Section 7.8, until the five (5)-year anniversary of the Distribution Date, each of Covidien and Mallinckrodt, on behalf of itself and each member of its respective Group, agrees to hold, and to cause its respective Representatives to hold, in strict confidence, with at least the same degree of care that applies to Covidien's confidential and proprietary information pursuant to policies in effect as of the Distribution Date (and in no event less than a reasonable degree of care), all confidential or proprietary Information ("Confidential Information") concerning each such other Group that is either in its possession (including Confidential Information in its possession prior to the date hereof) or furnished by any such other Group or its respective Representatives at any time pursuant to this Agreement, any Ancillary



Agreement or otherwise, and shall not use any such Confidential Information other than for such purposes as shall be expressly permitted hereunder or thereunder, except, in each case, to the extent that such Confidential Information has been (i) in the public domain through no fault of such Party or any member of such Group or any of their respective Representatives, (ii) later lawfully acquired from other sources by such Party (or any member of such Party's Group) which sources are not themselves bound by a confidentiality obligation, or (iii) independently generated without reference to any Confidential Information of the other Party. Each Party shall maintain, and shall cause its respective Group members and Representatives to maintain, policies and procedures, and develop such further policies and procedures as will from time to time become necessary or appropriate, to ensure compliance with this <u>Section 7.7</u>.

(b) Mallinckrodt acknowledges that it and other members of the Mallinckrodt Group may have in its or their possession Confidential Information of third Persons that was received under a confidentiality or nondisclosure agreement with such third Person while part of Covidien. Mallinckrodt will, and will cause its respective Group members and its Representatives to, hold in strict confidence the Confidential Information of third Persons to which any member of the Mallinckrodt Group has access, in accordance with the terms of any agreements entered into prior to the Effective Time between members of the Covidien Group and such third Persons.

(c) Each Party agrees not to release, communicate or disclose, or permit to be released, communicated or disclosed, directly or indirectly, any Confidential Information to any other Person, except its Representatives who need to know such Confidential Information (who shall be advised of their obligations hereunder with respect to such Confidential Information), except in compliance with <u>Section 7.8</u>. Without limiting the foregoing, when any Confidential Information is no longer needed for the purposes contemplated by this Agreement or any Ancillary Agreement, each Party will promptly after request of the other Party either return to the other Party all Confidential Information in a tangible form (including all copies thereof and all notes, extracts or summaries based thereon) or certify to the other Party that it has destroyed such Confidential Information (and such copies thereof and such notes, extracts or summaries based thereon).

(d) Each Party shall be liable for any failure by its respective Representatives to comply with the restrictions on use and disclosure of Confidential Information contained in this Agreement.

7.8 Protective Arrangements. In the event that any Party or any member of its Group either determines on the advice of its counsel that it is required to disclose any Confidential Information pursuant to applicable Law or receives any demand under lawful process or from any Governmental Authority to disclose or provide Information of any other Party (or any member of any other Party's Group) that is subject to the confidential Information and shall cooperate at the expense of the requesting Party in seeking any reasonable protective arrangements requested by such other Party. Subject to the foregoing, the Person that received such request may thereafter disclose or provide Confidential Information to the extent required by such Law (as so advised by counsel) or by lawful process or such Governmental Authority. The disclosing Party shall

promptly provide the Party owning such Confidential Information with a copy of the Information so disclosed, in the same form and format so disclosed, together with a list of all Persons to whom such Information was disclosed, in each case to the extent permitted by law.

### ARTICLE VIII DISPUTE RESOLUTION

8.1 <u>Good Faith Negotiation</u>. Subject to <u>Section 8.3</u>, either Party hereto seeking resolution of any dispute, controversy or claim arising out of or relating to this Agreement, the Transition Services Agreement, the Employee Matters Agreement or the validity, interpretation, breach or termination of this Agreement, the Transition Services Agreement or the Employee Matters Agreement (a "<u>Dispute</u>"), shall provide written notice thereof to the other Party hereto, and following delivery of such notice, the Parties shall attempt in good faith to negotiate a resolution of the Dispute. The negotiations shall be conducted by executives who have authority to settle the Dispute and who are at a higher level of management than the persons with direct responsibility for the subject matter of the Dispute. All such negotiations shall be confidential and shall be treated as compromise and settlement negotiations for purposes of applicable rules of evidence. If the Parties are unable for any reason to resolve a Dispute within thirty (30) days after the delivery of such notice or if a Party reasonably concludes that the other Party is not willing to negotiate as contemplated by this <u>Section 8.1</u>, the Dispute shall be submitted to mediation in accordance with <u>Section 8.2</u>.

8.2 Mediation. Any Dispute not resolved pursuant to Section 8.1 shall, at the written request of any Party hereto (a "Mediation Request"), be submitted to nonbinding mediation in accordance with the then-current International Institute for Conflict Prevention and Resolution ("CPR") Mediation Procedure (the "Procedure"), except as modified herein. The mediation shall be held in New York, New York or such other place as the Parties may mutually agree. The Parties shall have twenty (20) days from receipt by a Party (or Parties) of a Mediation Request to agree on a mediator. If no mediator has been agreed upon by the Parties within twenty (20) days of receipt by a Party (or Parties) of a Mediation Request, then any Party may request (on written notice to the other Party) that CPR appoint a mediator in accordance with the Procedure. All mediation pursuant to this clause shall be confidential and shall be treated as compromise and settlement negotiations for purposes of applicable rules of evidence, and no oral or documentary representations made by the Parties during such mediation shall be admissible for any purposes in any subsequent proceedings. No Party hereto shall disclose or permit the disclosure of any information about the evidence adduced or the documents produced by any other Party in the mediation proceedings or about the existence, contents or results of the mediation without the prior written consent of such other Party except in the course of a judicial or regulatory proceeding or as may be required by law or requested by a Governmental Authority or securities exchange. Before making any disclosure permitted by the preceding sentence, the Party intending to make such disclosure shall, to the extent reasonably practicable, give the other Party reasonable written notice of the intended disclosure and afford the other Party areasonable opportunity to protect its interests. If the Dispute has not been resolved within sixty (60) days of the appointment of a mediator, or within ninety (90) days after receip

### 8.3 Litigation.

(a) Notwithstanding the foregoing provisions of this <u>Article VIII</u>, (i) any Party may seek preliminary provisional or injunctive judicial relief without first complying with the procedures set forth in <u>Sections 8.1</u> and <u>8.2</u> if such action is reasonably necessary to avoid irreparable damage and (ii) either Party may initiate litigation before the expiration of the periods specified in <u>Section 8.2</u> if such Party has submitted a Mediation Request and the other Party has failed, within fourteen (14) days after the appointment of a mediator, to agree upon a date for the first mediation session to take place within thirty (30) days after the appointment of such mediator or such longer period as the Parties may agree to in writing.

(b) All applicable statutes of limitations and defenses based upon the passage of time shall be tolled while the procedures specified in <u>Sections 8.1</u> and <u>8.2</u> are pending. The Parties shall take any necessary or appropriate action required to effectuate such tolling.

## ARTICLE IX FURTHER ASSURANCES AND ADDITIONAL COVENANTS

### 9.1 Further Assurances.

(a) In addition to the actions specifically provided for elsewhere in this Agreement, each of the Parties shall use its reasonable best efforts, prior to, on and after the Distribution Date, to take, or cause to be taken, all actions, and to do, or cause to be done, all things, reasonably necessary, proper or advisable under applicable Laws, regulations and agreements to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements.

(b) Without limiting the foregoing, prior to, on and after the Distribution Date, each Party hereto shall cooperate with the other Party, and without any further consideration, but at the expense of the requesting Party, to execute and deliver, or use its reasonable best efforts to cause to be executed and delivered, all instruments, including instruments of conveyance, assignment and transfer, and to make all filings with, and to obtain all Approvals or Notifications of, any Governmental Authority or any other Person under any permit, license, agreement, indenture or other instrument (including any consents or Governmental Approvals), and to take all such other actions as such Party may reasonably be requested to take by any other Party from time to time, consistent with the terms of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements and the transfers of the Mallinckrodt Assets and the assignment and assumption of the Mallinckrodt Liabilities and the other ransactions contemplated hereby and thereby. Without limiting the foregoing, each Party will, at the reasonable request, cost and expense of any other Party, take such other actions as may be reasonably necessary to vest in such other Party good and marketable title to the Assets allocated to such Party under this Agreement or any of the Ancillary Agreements.

(c) On or prior to the Distribution Date, Covidien and Mallinckrodt in their respective capacities as direct and indirect shareholders of their respective Subsidiaries, shall each ratify any actions which are reasonably necessary or desirable to be taken by Covidien, Mallinckrodt or any of their respective Subsidiaries, as the case may be, to effectuate the transactions contemplated by this Agreement and the Ancillary Agreements.

(d) Covidien and Mallinckrodt, and each of the members of their respective Groups, waive (and agree not to assert against any of the others) any claim or demand that any of them may have against any of the others for any Liabilities or other claims relating to or arising out of: (i) the failure of Mallinckrodt or any other member of the Mallinckrodt Group, on the one hand, or of Covidien or any other member of the Covidien Group, on the other hand, to provide any notification or disclosure required under any state Environmental Law in connection with the Separation or the other transactions contemplated by this Agreement, including the transfer by any member of any Group to any member of the other Group of ownership or operational control of any Assets not previously owned or operated by such transferee; or (ii) any inadequate, incorrect or incomplete notification or disclosure under any such state Environmental Law by the applicable transferor. To the extent any Liability to any Governmental Authority or any third Person arises out of any action or inaction described in clause (i) or (ii) above, the transferee of the applicable Asset hereby assumes and agrees to pay any such Liability.

### ARTICLE X TERMINATION

10.1 <u>Termination</u>. This Agreement may be terminated by Covidien at any time, in its sole and absolute discretion, prior to the Effective Time. After the Effective Time, this Agreement may not be terminated except by an agreement in writing signed by each of the Parties.

10.2 Effect of Termination. In the event of any termination of this Agreement prior to the Effective Time, no Party (or any of its directors or officers) shall have any Liability or further obligation to any other Party.

### ARTICLE XI MISCELLANEOUS

### 11.1 Counterparts; Entire Agreement; Corporate Power.

(a) This Agreement and each Ancillary Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Party.

(b) This Agreement, the Ancillary Agreements, the Exhibits, the Schedules and appendices hereto and thereto contain the entire agreement between the Parties with respect to the subject matter hereof, supersede all previous agreements, negotiations, discussions, writings, understandings, commitments and conversations with respect to such subject matter and there are no agreements or understandings between the Parties other than those set forth or referred to herein or therein.

(c) Covidien represents on behalf of itself and each other member of the Covidien Group, and Mallinckrodt represents on behalf of itself and each other member of the Mallinckrodt Group, as follows:

(i) each such Person has the requisite corporate or other power and authority and has taken all corporate or other action necessary in order to execute, deliver and perform each of this Agreement and each Ancillary Agreement to which it is a party and to consummate the transactions contemplated hereby and thereby; and

(ii) this Agreement and each Ancillary Agreement to which it is a party has been duly executed and delivered by it and constitutes a valid and binding agreement of it enforceable in accordance with the terms thereof.

(d) Each Party acknowledges that it and each other Party is executing certain of the Ancillary Agreements by facsimile, stamp or mechanical signature. Each Party expressly adopts and confirms each such facsimile, stamp or mechanical signature made in its respective name as if it were a manual signature, agrees that it will not assert that any such signature is not adequate to bind such Party to the same extent as if it were signed manually and agrees that at the reasonable request of any other Party at any time it will as promptly as reasonably practicable cause each such Ancillary Agreement to be manually executed (any such execution to be as of the date of the initial date thereof).

(e) Notwithstanding any provision of this Agreement or any Ancillary Agreement, neither Covidien nor Mallinckrodt shall be required to take or omit to take any act that would violate its fiduciary duties to any minority shareholders of any non-wholly owned Subsidiary of Covidien or Mallinckrodt, as the case may be (it being understood that directors' qualifying shares or similar interests will be disregarded for purposes of determining whether a Subsidiary is wholly owned).

## 11.2 Governing Law; Submission to Jurisdiction; Waiver of Jury Trial.

(a) The construction, interpretation and performance of this Agreement shall be governed and construed according to the laws of the State of New York, without regard to conflicts of laws principles (other than Section 5-1401 and Section 5-1402 of the General Obligations Law of the State of New York).

(b) Each of Covidien and Mallinckrodt, on behalf of itself and the members of its Group, hereby irrevocably (i) agrees that any Dispute shall be subject to the exclusive jurisdiction of the state and federal courts located in New York, New York, (ii) waives any claims of forum non conveniens, and agrees to submit to the jurisdiction of such courts, as provided in New York General Obligations Law § 5-1402, (iii) agrees that service of any process, summons, notice or document by U.S. registered mail to its respective address set forth in Section 11.5 shall be effective service of process for any litigation brought against it in any such court or for the taking of any other acts as may be necessary or appropriate in order to effectuate any judgment of said courts and (iv) UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO TRIAL BY JURY IN CONNECTION WITH ANY DISPUTE.

11.3 <u>Assignability</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Except as otherwise provided for in this Agreement, this Agreement shall not be assignable, in whole or in part, directly or indirectly, by either Party without the express written consent of the other Party, and any attempt to assign any rights or obligations arising under this Agreement without such consent shall be void. A Party may assign its respective rights or delegate its respective obligations under this Agreement to any Affiliate of such Party; provided, however, that in connection with each such assignment or delegation, the assigning Party provides a guarantee to the non-assigning Party for any liability or obligation assigned or delegated pursuant to this <u>Section 11.3</u>; provided, further, that Mallinckrodt shall only be entitled to assign its rights or delegate its obligations under this Agreement with the prior written consent of Covidien.

11.4 <u>Third-Party Beneficiaries</u>. Except for the indemnification rights under this Agreement of any Covidien Indemnitee or Mallinckrodt Indemnitee in their respective capacities as such, (i) the provisions of this Agreement and each Ancillary Agreement are solely for the benefit of the Parties and are not intended to confer upon any Person except the Parties any rights or remedies hereunder, and (ii) there are no third-party beneficiaries of this Agreement or any Ancillary Agreement and neither this Agreement nor any Ancillary Agreement shall provide any third person with any remedy, claim, liability, reimbursement, claim of action or other right in excess of those existing without reference to this Agreement or any Ancillary Agreement.

11.5 Notices. All notices, requests, claims, demands or other communications under this Agreement and, to the extent applicable and unless otherwise provided therein, under each of the Ancillary Agreements shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by facsimile or electronic transmission with receipt confirmed (followed by delivery of an original via overnight courier service) or by registered or certified mail (postage prepaid, return receipt requested) to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this <u>Section 11.5</u>):

If to Covidien, to:

Covidien plc 1st Floor, 20 on Hatch Lower Hatch Street Dublin 2 Ireland Attn: General Counsel Facsimile: +353-1-438-1798

and

Covidien 15 Hampshire Street Mansfield, MA 02048 Attn: General Counsel Facsimile: (508) 261-8544

with a copy to: Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Attention: Adam O. Emmerich Benjamin M. Roth Facsimile: (212) 403-2000 If to Mallinckrodt to: Mallinckrodt plc Damastown, Mulhuddart Dublin 15 Ireland Attn: General Counsel Facsimile: (314) 654-5366 and Mallinckrodt 675 James S. McDonnell Blvd. Hazelwood, MO 63042 Attn: General Counsel Facsimile: 314-654-5366 with a copy to: Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Attention: Adam O. Emmerich Benjamin M. Roth Facsimile: (212) 403-2000

Any Party may, by notice to the other Party, change the address and contact person to which any such notices are to be given.

11.6 Severability. If any provision of this Agreement or any Ancillary Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof or thereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby. Upon such determination, the Parties shall negotiate in good faith in an effort to agree upon such a suitable and equitable provision to effect the original intent of the Parties.

11.7 Force Majeure. No Party shall be deemed in default of this Agreement or any Ancillary Agreement to the extent that any delay or failure in the performance of its obligations under this Agreement or any Ancillary Agreement, other than a delay or failure to make a payment, results from any cause beyond its reasonable control and without its fault or negligence, such as acts of God, acts of civil or military authority, embargoes, epidemics, war, riots, insurrections, fires, explosions, earthquakes, floods, unusually severe weather conditions, labor problems or unavailability of parts, or, in the case of computer systems, any failure in electrical or air conditioning equipment (each such cause, a "Force Majeure"). In the event of any such excused delay, the time for performance shall be extended for a period equal to the time lost by reason of the delay.

11.8 <u>Publicity</u>. Prior to the Effective Time, each of Mallinckrodt and Covidien shall consult with each other prior to issuing any press releases or otherwise making public statements with respect to the Separation, the Distribution or any of the other transactions contemplated hereby or under any Ancillary Agreement and prior to making any filings with any Governmental Authority with respect thereto.

11.9 Expenses. Except as expressly set forth in this Agreement (including Sections 2.15, 6.1, 7.6(a), 7.8 and 9.1(b) and Articles IV and  $\underline{V}$ ) or in any Ancillary Agreement, all fees, costs and expenses incurred in connection with the preparation, execution, delivery and implementation of this Agreement and any Ancillary Agreement, and with the consummation of the transactions contemplated hereby and thereby, will be borne by the Party incurring such fees, costs or expenses.

11.10 <u>Headings</u>. The article, section and paragraph headings contained in this Agreement and in the Ancillary Agreements are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement or any Ancillary Agreement.

11.11 <u>Survival of Covenants</u>. Except as expressly set forth in this Agreement or any Ancillary Agreement, the covenants, representations and warranties contained in this Agreement and each Ancillary Agreement, and liability for the breach of any obligations contained herein, shall survive the Separation and the Distribution and shall remain in full force and effect.

11.12 <u>Waivers of Default</u>. Waiver by any Party of any default by the other Party of any provision of this Agreement or any Ancillary Agreement shall not be deemed a waiver by the waiving Party of any subsequent or other default, nor shall it prejudice the rights of the other Party. No failure or delay by any Party in exercising any right, power or privilege under this Agreement or any Ancillary Agreement shall operate as a waiver thereof nor shall a single or partial exercise thereof prejudice any other or further exercise thereof or the exercise of any other right, power or privilege.

11.13 <u>Specific Performance</u>. Subject to the provisions of <u>Article VIII</u>, in the event of any actual or threatened default in, or breach of, any of the terms, conditions and provisions of this Agreement or any Ancillary Agreement, the Party or Parties who are, or are to be, thereby aggrieved shall have the right to specific performance and injunctive or other equitable relief in respect of its or their rights under this Agreement or such Ancillary

Agreement, in addition to any and all other rights and remedies at law or in equity, and all such rights and remedies shall be cumulative. The Parties agree that the remedies at law for any breach or threatened breach, including monetary damages, are inadequate compensation for any loss and that any defense in any action for specific performance that a remedy at law would be adequate is waived. Any requirements for the securing or posting of any bond with such remedy are waived by each of the Parties.

11.14 <u>Amendments</u>. No provisions of this Agreement or any Ancillary Agreement shall be deemed waived, amended, supplemented or modified by any Party, unless such waiver, amendment, supplement or modification is in writing and signed by the authorized representative of the Party against whom it is sought to enforce such waiver, amendment, supplement or modification.

11.15 Interpretation. In this Agreement and any Ancillary Agreement, (a) words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other genders as the context requires; (b) the terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement (or the applicable Ancillary Agreement) as a whole (including all of the Schedules, Exhibits and Appendices hereto and thereto) and not to any particular provision of this Agreement (or such Ancillary Agreement); (c) Article, Section, Exhibit, Schedule and Appendices hereto and thereto) and not to any particular provision of this Agreement (or such Ancillary Agreement); (c) Article, Section, Exhibit, Schedule and Appendix references are to the Articles, Sections, Exhibits, Schedules and Appendices to this Agreement (or the applicable Ancillary Agreement) unless otherwise specified; (d) the word "including" and words of similar import when used in this Agreement (or the applicable Ancillary Agreement) shall mean "including, without limitation"; (e) the word "or" shall not be exclusive; (f) unless expressly stated to the contrary in this Agreement or in any Ancillary Agreement, all references to "the date hereof," "the date of this Agreement," "hereby" and "hereupon" and words of similar import shall all be references to [—], 2013, regardless of any amendment or restatement hereof; and (g) except where the context otherwise requires, references to Subsidiaries of Mallinckrodt upon consummation of the Distribution. Covidien and Mallinckrodt have each participated in the negotiation and drafting of this Agreement and if an ambiguity or question of interpretation should arise, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or burdening either Party by virtue of the authorship of any of the provisions in this Agreement or any interim drafts of this Agreeme

11.16 <u>Attorney-Client Privilege</u>. Mallinckrodt agrees that, in the event of any Dispute or other litigation, dispute, controversy or claim between Covidien or a member of the Covidien Group, on the one hand, and Mallinckrodt or a member of the Mallinckrodt Group, on the other hand, Mallinckrodt will not, and will cause the members of its Group not to, seek any waiver of attorney-client privilege with respect to any communications relating to advice given prior to the Effective Time by counsel to Covidien or any Person that was a subsidiary of Covidien prior to the Distribution Date, regardless of any argument that such advice may have affected the interests of both Parties. Moreover, Mallinckrodt will, and will cause the members of its Group to, honor any such attorney-client privilege between Covidien and the members of its Group and its or their counsel, and will not assert that Covidien or a member of its Group has waived, relinquished or otherwise lost such privilege. For the avoidance of doubt, in the event of

any litigation, dispute, controversy or claim between Covidien or a member of its Group, on the one hand, and a third party other than a member of the Mallinckrodt Group, on the other hand, Covidien shall retain the right to assert attorney-client privilege with respect to any communications relating to advice given prior to the Distribution Date by counsel to Covidien or any Person that was a subsidiary of Covidien prior to the Distribution Date.

11.17 <u>Limitations of Liability</u>. Notwithstanding anything in this Agreement to the contrary, neither Mallinckrodt or its Affiliates, on the one hand, nor Covidien or its Affiliates, on the other hand, shall be liable under this Agreement to the other for any special, indirect, punitive, exemplary, remote, speculative or similar damages in excess of compensatory damages of the other arising in connection with the transactions contemplated hereby (other than any such liability with respect to a Third-Party Claim), whether or not advised of the possibility of such damages and whether or not such damages are reasonably foreseeable.

11.18 <u>Performance</u>. Covidien will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the Covidien Group. Mallinckrodt will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the Covidien Group. Mallinckrodt will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the Mallinckrodt Group. Each Party (including its permitted successors and assigns) further agrees that it will (a) give timely notice of the terms, conditions and continuing obligations contained in this Section 11.18 to all of the other members of its Group, and (b) cause all of the other members of its Group not to take any action or fail to take any such action inconsistent with such Party's obligations under this Agreement, any Ancillary Agreement or the transactions contemplated hereby or thereby.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed on the date first written above by their duly authorized representatives.

COVIDIEN PLC

By: Name:

Title:

MALLINCKRODT PLC

By:

Name: Title:

Companies Acts 1963 to 2012

## A PUBLIC COMPANY LIMITED BY SHARES

## MEMORANDUM and ARTICLES OF ASSOCIATION

of

# MALLINCKRODT PUBLIC LIMITED COMPANY

# ARTHUR COX

DUBLIN

Cert. No.: 522227

#### Companies Acts 1963 to 2012

## A PUBLIC COMPANY LIMITED BY SHARES

# MEMORANDUM OF ASSOCIATION

### of

## MALLINCKRODT PUBLIC LIMITED COMPANY

- 1. The name of the Company is Mallinckrodt public limited company.
- 2. The Company is to be a public limited company.
- 3. The objects for which the Company is established are:
  - 3.1 (a) To carry on the business of a healthcare services development company operating in the healthcare field, and to design, manufacture, produce, supply and provide generic and branded pharmaceuticals, contrast media, radiopharmaceuticals, active pharmaceutical ingredients and dosage pharmaceuticals and other devices or products of a surgical, pharmaceutical, diagnostic, medical imaging or medical character necessary or suitable for the proper treatment of sick or injured persons or patients and to carry on business as merchants of and dealers in all supplies required for use in the treatment and care of the sick and injured and to do all things usually dealt in by persons carrying on the above mentioned businesses or any of them or likely to be required in connection with any of the said businesses.
    - (b) To carry on the business of a holding company and to co-ordinate the administration, finances and activities of any subsidiary companies or associated companies, to do all lawful acts and things whatever that are necessary or convenient in carrying on the business of such a holding company and in particular to carry on in all its branches the business of a management services company, to act as managers and to direct or coordinate the management of other companies or of the business, property and estates of any company or person and to undertake and carry out all such services in connection therewith as may be deemed expedient by the Company's board of directors and to exercise its powers as a shareholder of other companies.
    - (c) To acquire the entire issued share capital of Mallinckrodt International Finance S.A., a Luxembourg registered company and Mallinckrodt Belgium BVBA, a Belgian registered company.
  - 3.2 To acquire shares, stocks, debentures, debenture stock, bonds, obligations and securities by original subscription, tender, purchase, exchange or otherwise and to subscribe for the same either conditionally or otherwise, and to guarantee the subscription thereof and to exercise and enforce all rights and powers conferred by or incidental to the ownership thereof.
  - 3.3 To facilitate and encourage the creation, issue or conversion of and to offer for public subscription debentures, debenture stocks, bonds, obligations, shares, stocks, and securities and to act as trustees in connection with any such securities and to take part in the conversion of business concerns and undertakings into companies.
  - 3.4 To purchase or by any other means acquire any freehold, leasehold or other property and in particular lands, tenements and hereditaments of any tenure, whether subject or not to any charges or incumbrances, for any estate or interest whatever, and any rights, privileges or easements over or in respect of any property, and any buildings, factories, mills, works, wharves, roads, machinery, engines, plant, live and dead stock, barges, vessels or things, and any real or personal property or rights whatsoever which may be necessary for, or may conveniently be used with, or may enhance the value or property of the Company, and to hold or to sell, let, alienate, mortgage, charge or otherwise deal with all or any such freehold, leasehold, or other property, lands, tenements or hereditaments, rights, privileges or easements.

- 3.5 To sell or otherwise dispose of any of the property or investments of the Company.
- 3.6 To establish and contribute to any scheme for the purchase of shares in the Company to be held for the benefit of the Company's employees and to lend or otherwise provide money to such schemes or the Company's employees or the employees of any of its subsidiary or associated companies to enable them to purchase shares of the Company.
- 3.7 To grant, convey, transfer or otherwise dispose of any property or asset of the Company of whatever nature or tenure for such price, consideration, sum or other return whether equal to or less than the market value thereof and whether by way of gift or otherwise as the Directors shall deem fit and to grant any fee, farm grant or lease or to enter into any agreement for letting or hire of any such property or asset for a rent or return equal to or less than the market or rack rent therefor or at no rent and subject to or free from covenants and restrictions as the Directors shall deem appropriate.
- 3.8 To acquire and undertake the whole or any part of the business, good-will and assets of any person, firm or company carrying on or proposing to carry on any of the businesses which this Company is authorised to carry on, and as part of the consideration for such acquisition to undertake all or any of the liabilities of such person, firm or company, or to acquire an interest in, amalgamate with, or enter into any arrangement for sharing profits, or for co-operation, or for limiting competition or for mutual assistance with any such person, firm or company and to give or accept by way of consideration for any of the acts or things aforesaid or property acquired, any shares, debentures, debenture stock or securities that may be agreed upon, and to hold and retain or sell, mortgage or deal with any shares, debentures, debenture stock or securities so received.
- 3.9 To apply for, purchase or otherwise acquire any patents, brevets d'invention, licences, concessions and the like conferring any exclusive or nonexclusive or limited rights to use or any secret or other information as to any invention which may seem capable of being used for any of the purposes of the Company or the acquisition of which may seem calculated directly or indirectly to benefit the Company, and to use, exercise, develop or grant licences in respect of or otherwise turn to account the property, rights or information so acquired.
- 3.10 To enter into partnership or into any arrangement for sharing profits, union of interests, co-operation, joint venture, reciprocal concession or otherwise with any person or company carrying on or engaged in or about to carry on or engage in any business or transaction which the Company is authorised to carry on or engage in or any business or transaction capable of being conducted so as directly to benefit this Company.
- 3.11 To invest and deal with the moneys of the Company not immediately required upon such securities and in such manner as may from time to time be determined.
- 3.12 To lend money to and guarantee the performance of the contracts or obligations of any company, firm or person, and the repayment of the capital and principal of, and dividends, interest or premiums payable on, any stock, shares and securities of any company, whether having objects similar to those of this Company or not, and to give all kinds of indemnities.
- 3.13 To engage in currency exchange and interest rate transactions including, but not limited to, dealings in foreign currency, spot and forward rate exchange contracts, futures, options, forward rate agreements, swaps, caps, floors, collars and any other foreign exchange or interest rate hedging arrangements and such other instruments as are similar to, or derived from, any of the foregoing whether for the purpose of making a profit or avoiding a loss or managing a currency or interest rate exposure or any other exposure or for any other purpose.
- 3.14 To guarantee, support or secure, whether by personal covenant or by mortgaging or charging all or any part of the undertaking, property and assets (both present and future) and uncalled capital of the Company, or by both such methods, the performance of the obligations of, and

the repayment or payment of the principal amounts of and premiums, interest and dividends on any securities of, any person, firm or company including (without prejudice to the generality of the foregoing) any company which is for the time being the Company's holding company as defined by section 155 of the Companies Act, 1963 (or any successor legislation) or a subsidiary as therein defined of any such holding company or otherwise associated with the Company in business.

- 3.15 To borrow or secure the payment of money in such manner as the Company shall think fit, and in particular by the issue of debentures, debenture stocks, bonds, obligations and securities of all kinds, either perpetual or terminable and either redeemable or otherwise and to secure the repayment of any money borrowed, raised or owing by trust deed, mortgage, charge, or lien upon the whole or any part of the Company's property or assets (whether present or future) including its uncalled capital, and also by a similar trust deed, mortgage, charge or lien to secure and guarantee the performance by the Company of any obligation or liability it may undertake.
- 3.16 To draw, make, accept, endorse, discount, execute, negotiate and issue promissory notes, bills of exchange, bills of lading, warrants, debentures and other negotiable or transferable instruments.
- 3.17 To subscribe for, take, purchase or otherwise acquire and hold shares or other interests in, or securities of any other company having objects altogether or in part similar to those of this Company, or carrying on any business capable of being conducted so as directly or indirectly to benefit this Company.
- 3.18 To hold in trust as trustees or as nominees and to deal with, manage and turn to account, any real or personal property of any kind, and in particular shares, stocks, debentures, securities, policies, book debts, claims and chases in actions, lands, buildings, hereditaments, business concerns and undertakings, mortgages, charges, annuities, patents, licences, and any interest in real or personal property, and any claims against such property or against any person or company.
- 3.19 To constitute any trusts with a view to the issue of preferred and deferred or other special stocks or securities based on or representing any shares, stocks and other assets specifically appropriated for the purpose of any such trust and to settle and regulate and if thought fit to undertake and execute any such trusts and to issue, dispose of or hold any such preferred, deferred or other special stocks or securities.
- 3.20 To give any guarantee in relation to the payment of any debentures, debenture stock, bonds, obligations or securities and to guarantee the payment of interest thereon or of dividends on any stocks or shares of any company.
- 3.21 To construct, erect and maintain buildings, houses, flats, shops and all other works, erections, and things of any description whatsoever either upon the lands acquired by the Company or upon other lands and to hold, retain as investments or to sell, let, alienate, mortgage, charge or deal with all or any of the same and generally to alter, develop and improve the lands and other property of the Company.
- 3.22 To provide for the welfare of persons in the employment of or holding office under or formerly in the employment of or holding office under the Company including Directors and ex-Directors of the Company and the wives, widows and families, dependants or connections of such persons by grants of money, pensions or other payments and by forming and contributing to pension, provident or benefit funds or profit sharing or co-partnership schemes for the benefit of such persons and to form, subscribe to or otherwise aid charitable, benevolent, religious, scientific, national or other institutions, exhibitions or objects which shall have any moral or other claims to support or aid by the Company by reason of the locality of its operation or otherwise.
- 3.23 To remunerate by cash payments or allotment of shares or securities of the Company credited as fully paid up or otherwise any person or company for services rendered or to be rendered to the Company whether in the conduct or management of its business, or in placing or assisting

to place or guaranteeing the placing of any of the shares of the Company's capital, or any debentures or other securities of the Company or in or about the formation or promotion of the Company.

- 3.24 To enter into and carry into effect any arrangement for joint working in business or for sharing of profits or for amalgamation with any other company or association or any partnership or person carrying on any business within the objects of the Company.
- 3.25 To distribute in specie or otherwise as may be resolved, any assets of the Company among its members and in particular the shares, debentures or other securities of any other company belonging to this Company or of which this Company may have the power of disposing.
- 3.26 To vest any real or personal property, rights or interest acquired or belonging to the Company in any person or company on behalf of or for the benefit of the Company, and with or without any declared trust in favour of the Company.
- 3.27 To transact or carry on any business which may seem to be capable of being conveniently carried on in connection with any of these objects or calculated directly or indirectly to enhance the value of or facilitate the realisation of or render profitable any of the Company's property or rights.
- 3.28 To accept stock or shares in or debentures, mortgages or securities of any other company in payment or part payment for any services rendered or for any sale made to or debt owing from any such company, whether such shares shall be wholly or partly paid up.
- 3.29 To pay all costs, charges and expenses incurred or sustained in or about the promotion and establishment of the Company or which the Company shall consider to be preliminary thereto and to issue shares as fully or in part paid up, and to pay out of the funds of the Company all brokerage and charges incidental thereto.
- 3.30 To procure the Company to be registered or recognised in any part of the world.
- 3.31 To do all or any of the matters hereby authorised in any part of the world or in conjunction with or as trustee or agent for any other company or person or by or through any factors, trustees or agents.
- 3.32 To make gifts or grant bonuses to the Directors or any other persons who are or have been in the employment of the Company including substitute and alternate directors.
- 3.33 To do all such other things that the Company may consider incidental or conducive to the attainment of the above objects or as are usually carried on in connection therewith.
- 3.34 To carry on any business which the Company may lawfully engage in and to do all such things incidental or conducive to the business of the Company.
- 3.35 To make or receive gifts by way of capital contribution or otherwise.

The objects set forth in any sub-clause of this clause shall be regarded as independent objects and shall not, except where the context expressly so requires, be in any way limited or restricted by reference to or inference from the terms of any other sub-clause, or by the name of the Company. None of such sub-clauses or the objects therein specified or the powers thereby conferred shall be deemed subsidiary or auxiliary merely to the objects mentioned in the first sub-clause of this clause, but the Company shall have full power to exercise all or any of the powers conferred by any part of this cause in any part of the world notwithstanding that the business, property or acts proposed to be transacted, acquired or performed do not fall within the objects of the first sub-clause of this clause.

NOTE: It is hereby declared that the word "company" in this clause, except where used in reference to this Company shall be deemed to include any partnership or other body of persons whether incorporated or not incorporated and whether domiciled in Ireland or elsewhere and the intention is that the objects specified in each paragraph of this clause shall except where otherwise expressed in such paragraph be in no way limited or restricted by reference to or inference from the terms of any other paragraph.

- 4. The share capital of the Company is US\$200,000,000 and €40,000 divided into 500,000,000 Ordinary Shares of US\$0.20 each, 500,000,000 Preferred Shares of US\$0.20 each and 40,000 Ordinary A Shares of €1.00 each.
- 5. The liability of the members is limited.
- 6. The shares forming the capital, increased or reduced, may be increased or reduced and be divided into such classes and issued with any special rights, privileges and conditions or with such qualifications as regards preference, dividend, capital, voting or other special incidents, and be held upon such terms as may be attached thereto or as may from time to time be provided by the original or any substituted or amended articles of association and regulations of the Company for the time being, but so that where shares are issued with any preferential or special rights attached thereto such rights shall not be alterable otherwise than pursuant to the provisions of the Company's articles of association for the time being.

We, the several persons whose names and addresses are subscribed, wish to be formed into a company in pursuance of this memorandum of association and we agree to take the number of shares in the capital of the company set opposite our respective names.

Names, addresses and descriptions of subscribers	Number of shares taken by each subscribe
J. MCGOWAN-SMYTH For and on behalf of	
For and on benall of Fand Limited	One Ordinary Share
Arthur Cox Building	One Ordinary Share
Earlsfort Terrace	
Dublin 2	
J. MCGOWAN-SMYTH	
For and on behalf of	
DIJR Nominees Limited	One Ordinary Share
Arthur Cox Building	
Earlsfort Terrace	
Dublin 2	
J. MCGOWAN-SMYTH	
For and on behalf of	One Ordinary Share
AC Administration Services Limited	
Arthur Cox Building	
Earlsfort Terrace	
Dublin 2	
J. MCGOWAN-SMYTH	
For and on behalf of	
Arthur Cox Nominees Limited	One Ordinary Share
Arthur Cox Building	
Earlsfort Terrace	
Dublin 2	
J. MCGOWAN-SMYTH	
For and on behalf of	
Arthur Cox Registrars Limited	One Ordinary Share
Arthur Cox Building	,
Earlsfort Terrace	
Dublin 2	
J. MCGOWAN-SMYTH	
For and on behalf of	
Arthur Cox Trust Services Limited	One Ordinary Share
Arthur Cox Building	
Earlsfort Terrace	
Dublin 2	
J. MCGOWAN-SMYTH	
For and on behalf of	
Arthur Cox Trustees Limited	One Ordinary Share
Arthur Cox Building	-
Earlsfort Terrace	
Dublin 2	
Solicitor	
	7

# Dated 21 December 2012

Witness to the above signatures:

Name:	MAIREAD FOLEY
Address:	ARTHUR COX BUILDING EARLSFORT TERRACE DUBLIN 2

Occupation: COMPANY SECRETARY

## **COMPANIES ACTS 1963 TO 2012**

# A PUBLIC COMPANY LIMITED BY SHARES

## ARTICLES OF ASSOCIATION

#### -of-

### MALLINCKRODT PUBLIC LIMITED COMPANY

# PRELIMINARY

1. The regulations contained in Table A in the First Schedule to the Companies Act, 1963 shall not apply to the Company.

2. (a) In these articles:

"1983 Act" means the Companies (Amendment) Act 1983.

"1990 Act" means the Companies Act 1990 (No. 33 of 1990).

"Act" means the Companies Act, 1963 (No. 33 of 1963) as amended by the Companies Acts 1977 to 2012 and Parts 2 and 3 of the Investment Funds, Companies and Miscellaneous Provisions Act 2006, the Companies (Amendment) Act 2009, the Companies (Miscellaneous Provisions) Act 2009 and the Companies (Amendment) Act 2012, all enactments which are to be read as one with, or construed or read together as one with, the Acts and every statutory modification and re-enactment thereof for the time being in force.

"Acts" means the Companies Acts 1963 to 2005 and Parts 2 and 3 of the Investment Funds, Companies and Miscellaneous Provisions Act 2006, the Companies (Amendment) Act 2009, the Companies (Miscellaneous Provisions) Act 2009 and the Companies (Amendment) Act 2012, all enactments which are to be read as one with, or construed or read together as one with, the Companies Acts and every statutory modification and re-enactment thereof for the time being in force.

"address" includes any number or address used for the purposes of communication by way of electronic mail or other electronic communication.

"Assistant Secretary" means any person appointed by the Secretary from time to time to assist the Secretary.

"Clear Days" in relation to the period of notice, means that period excluding the day when the notice is given or deemed to be given and the day for which it is given or on which it is to take effect.

"Chairman" means the Director who is elected by the Directors from time to time to preside as chairman at all meetings of the Board and at general meetings of the Company.

"electronic communication" has the meaning given to those words in the Electronic Commerce Act 2000.

"electronic signature" has the meaning given to those words in the Electronic Commerce Act 2000.

"Ordinary Resolution" means an ordinary resolution of the Company's members within the meaning of section 141 of the Act.

"public announcement" means disclosure in a press release reported by a national news service or in a document publicly filed by the Company with the U.S. Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act and the rules and regulations promulgated thereunder.

"Redeemable Shares" means redeemable shares in accordance with section 206 of the 1990 Act.

"Register" means the register of members to be kept as required in accordance with section 116 of the Act.

"Special Resolution" means a special resolution of the Company's members within the meaning of section 141 of the Act.

"the Company" means the company whose name appears in the heading to these articles.

"the Directors" or "the Board" means the directors from time to time and for the time being of the Company or the directors present at a meeting of the board of directors and includes any person occupying the position of director by whatever name called.

"the Group" means the Company and its subsidiaries from time to time and for the time being.

"the Holder" in relation to any share, means the member whose name is entered in the Register as the holder of the share or, where the context permits, the members whose names are entered in the Register as the joint holders of shares.

"the Office" means the registered office from time to time and for the time being of the Company.

"the seal" means the common seal of the Company.

"the Secretary" means any person appointed to perform the duties of the secretary of the Company.

"these articles" means the articles of association of which this article 2 forms part, as the same may be amended and may be from time to time and for the time being in force.

- (b) Expressions in these articles referring to writing shall be construed, unless the contrary intention appears, as including references to printing, lithography, photography and any other modes of representing or reproducing words in a visible form except as provided in these articles and/or where it constitutes writing in electronic form sent to the Company, and the Company has agreed to its receipt in such form. Expressions in these articles referring to execution of any document shall include any mode of execution whether under seal or under hand or any mode of electronic signature as shall be approved by the Directors. Expressions in these articles referring to receipt of any electronic communications shall, unless the contrary intention appears, be limited to receipt in such manner as the Company has approved.
- (c) Unless the contrary intention appears, words or expressions contained in these articles shall bear the same meaning as in the Acts or in any statutory modification thereof in force at the date at which these articles become binding on the Company.
- (d) A reference to a statute or statutory provision shall be construed as a reference to the laws of Ireland unless otherwise specified and includes:
  - (i) any subordinate legislation made under it including all regulations, by-laws, orders and codes made thereunder;
  - (ii) any repealed statute or statutory provision which it re-enacts (with or without modification); and

(iii) any statute or statutory provision which modifies, consolidates, re-enacts or supersedes it.

- (e) The masculine gender shall include the feminine and neuter, and vice versa, and the singular number shall include the plural, and vice versa, and words importing persons shall include firms or companies.
- (f) Reference to US\$, USD, or dollars shall mean the currency of the United States of America and to €, euro, EUR or cent shall mean the currency of Ireland.

#### SHARE CAPITAL AND VARIATION OF RIGHTS

- 3. (a) The share capital of the Company is US\$200,000,000 and €40,000 divided into 500,000,000 ordinary shares of US\$0.20 each, 500,000,000 preferred shares of US\$0.20 each and 40,000 ordinary A shares of €1.00 each.
  - (b) The rights and restrictions attaching to the ordinary shares shall be as follows:
    - subject to the right of the Company to set record dates for the purposes of determining the identity of members entitled to notice of and/or to vote at a general meeting, the right to attend and speak at any general meeting of the Company and to exercise one vote per ordinary share held at any general meeting of the Company;
    - (ii) the right to participate pro rata in all dividends declared by the Company; and
    - (iii) the right, in the event of the Company's winding up, to participate pro rata in the total assets of the Company.

The rights attaching to the ordinary shares may be subject to the terms of issue of any series or class of preferred shares allotted by the Directors from time to time in accordance with article 3(d).

- (c) The Directors may issue and allot ordinary A shares subject to the rights, privileges, limitations and restrictions set out in this article 3(c):
  - (i) Income

The holder of an ordinary A share shall not be entitled to receive any dividend or distribution declared, made or paid or any return of capital (save as provided for in this article) and shall not entitle its holder to any further or other right of participation in the assets of the Company.

(ii) Capital

On a winding up of, or other return of capital (other than on a redemption of any class of shares in the capital of the Company) by the Company, the holders of ordinary A shares shall be entitled to participate in such return of capital or winding up of the Company, such entitlement to be limited to the repayment of the amount paid up or credited as paid up on such ordinary A shares and shall be paid only after the holders of ordinary shares shall have received payment in respect of such amount as is paid up or credited as paid up on those ordinary shares held by them at that time, plus the payment in cash of \$100,000,000 on each such ordinary share.

(iii) Acquisition of Ordinary A Shares

The Company as agent for the holders of ordinary A shares shall have the irrevocable authority to authorise and instruct the Secretary (or any other person appointed for the purpose by the Directors) to acquire, or to accept the surrender of, the ordinary A shares for no consideration and to execute on behalf of such holders such documents as are necessary in connection with such acquisition or surrender, and pending such

acquisition or surrender to retain the certificates, to the extent issued, for such ordinary A shares. Any request by the Company to acquire, or for the surrender of, any ordinary A shares may be made by the Directors depositing at the Office a notice addressed to such person as the Directors shall have nominated on behalf of the holders of ordinary A shares. A person whose shares have been acquired or surrendered in accordance with this article shall cease to be a member in respect of such ordinary A shares but shall notwithstanding remain liable to pay the Company all monies which, at the date of acquisition or surrender, were payable by him or her to the Company in respect of such shares, but his or her liability shall cease if and when the Company has received payment in full of all such monies in respect of such shares. A notice issued pursuant to this paragraph shall be deemed to be validly issued notwithstanding the provisions of articles 134 to 139 inclusive.

(iv) Voting

The holders of ordinary A shares shall not be entitled to receive notice of, nor attend, speak or vote at, any general meeting.

The rights attaching to the ordinary A shares may be subject to the terms of issue of any series or class of preferred shares allotted by the Directors from time to time in accordance with article 3(d).

- (d) The Directors are authorised to issue all or any of the authorised but unissued preferred shares from time to time in one or more classes or series, and to fix for each such class or series such voting power, full or limited, or no voting power, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board providing for the issuance of such class or series, including, without limitation, the authority to provide that any such class or series may be:
  - (i) redeemable at the option of the Company, or the Holders, or both, with the manner of the redemption to be set by the Board, and redeemable at such time or times, including upon a fixed date, and at such price or prices;
  - entitled to receive dividends (which may be cumulative or non-cumulative) at such rates, on such conditions and at such times, and payable in preference to, or in such relation to, the dividends payable on any other class or classes of shares or any other series;
  - (iii) entitled to such rights upon the dissolution of, or upon any distribution of the assets of, the Company; or
  - (iv) convertible into, or exchangeable for, shares of any other class or classes of shares, or of any other series of the same or any other class or classes of shares, of the Company at such price or prices or at such rates of exchange and with such adjustments as the Directors determine,

which rights and restrictions may be as stated in such resolution or resolutions of the Directors as determined by them in accordance with this article 3(d). The Board may at any time before the allotment of any preferred share by further resolution in any way amend the designations, preferences, rights, qualifications, limitations or restrictions, or vary or revoke the designations of such preferred shares.

The rights conferred upon the Holder of any pre-existing shares in the share capital of the Company shall be deemed not to be varied by the creation, issue and allotment of preferred shares in accordance with this article 3(d).

(e) An ordinary share shall be deemed to be a Redeemable Share on, and from the time of, the existence or creation of an agreement, transaction or trade between the Company and any third party pursuant to which the Company acquires or will acquire ordinary shares, or an interest in

ordinary shares, from such third party. In these circumstances, the acquisition of such shares or interest in shares by the Company shall constitute the redemption of a Redeemable Share in accordance with Part XI of the 1990 Act.

- 4. Subject to the provisions of Part XI of the 1990 Act and the other provisions of this article, the Company may:
  - (a) pursuant to section 207 of the 1990 Act, issue any shares of the Company which are to be redeemed or are liable to be redeemed at the option of the Company or the member on such terms and in such manner as may be determined by the Company in general meeting (by Special Resolution) on the recommendation of the Directors; or
  - (b) subject to and in accordance with the provisions of the Acts and without prejudice to any relevant special rights attached to any class of shares pursuant to section 211 of the 1990 Act, purchase any of its own shares (including any Redeemable Shares and without any obligation to purchase on any pro rata basis as between members or members of the same class) and may cancel any shares so purchased or hold them as treasury shares (as defined in section 209 of the 1990 Act) and may reissue any such shares as shares of any class or classes.
- 5. Without prejudice to any special rights previously conferred on the Holders of any existing shares or class of shares, any share in the Company may be issued with such preferred or deferred or other special rights or such restrictions, whether in regard to dividend, voting, return of capital or otherwise, as the Company may from time to time by Ordinary Resolution determine.
- 6. (a) Without prejudice to the authority conferred on the Directors pursuant to article 3 to issue preferred shares in the capital of the Company, if at any time the share capital is divided into different classes of shares, the rights attached to any class may, whether or not the Company is being wound up, be varied or abrogated with the consent in writing of the Holders of three-fourths of the issued shares in that class, or with the sanction of a Special Resolution passed at a separate general meeting of the Holders of the shares of that class, provided that, if the relevant class of Holders has only one Holder, that person present in person or by proxy, shall constitute the necessary quorum. To every such meeting the provisions of article 35 shall apply.
  - (b) The redemption or purchase of preferred shares or any class of preferred shares shall not constitute a variation of rights of the preferred Holders where the redemption or purchase of the preferred shares has been authorised solely by a resolution of the ordinary Holders.
  - (c) The issue, redemption or purchase of any of the US\$500,000,000 preferred shares of US\$0.20 shall not constitute a variation of the rights of the Holders of ordinary shares.
  - (d) The issue of preferred shares or any class of preferred shares which rank pari passu with, or junior to, any existing preferred shares or class of preferred shares shall not constitute a variation of the existing preferred shares or class of preferred shares.
- 7. The rights conferred upon the Holders of the shares of any class issued with preferred or other rights shall not, unless otherwise expressly provided by the terms of issue of the shares of that class, be deemed to be varied by the creation or issue of further shares ranking pari passu therewith.
- 8. (a) Subject to the provisions of these articles relating to new shares, the shares shall be at the disposal of the Directors, and they may (subject to the provisions of the Acts) allot, grant options over or otherwise dispose of them to such persons, on such terms and conditions and at such times as they may consider to be in the best interests of the Company and its members, but so that no share shall be issued at a discount save in accordance with sections 26(5) and 28 of the 1983 Act, and so that, in the case of shares offered to the public for subscription, the amount payable on application on each share shall not be less than one-quarter of the nominal amount of the share and the whole of any premium thereon.
  - (b) Subject to any requirement to obtain the approval of members under any laws, regulations or the rules of any stock exchange to which the Company is subject, the Board is authorised, from time to time, in its discretion, to grant such persons, for such periods and upon such

terms as the Board deems advisable, options to purchase or subscribe for such number of shares of any class or classes or of any series of any class as the Board may deem advisable, and to cause warrants or other appropriate instruments evidencing such options to be issued.

- (c) The Directors are, for the purposes of section 20 of the 1983 Act, generally and unconditionally authorised to exercise all powers of the Company to allot and issue relevant securities (as defined by the said section 20) up to the amount of Company's authorised share capital and to allot and issue any shares purchased by the Company pursuant to the provisions of Part XI of the 1990 Act and held as treasury shares and this authority shall expire five years from the date of adoption of these articles. The Company may before the expiry of such authority make an offer or agreement which would or might require equity securities to be allotted after such expiry and the Directors may allot equity securities in pursuance of such an offer or agreement notwithstanding that the authority hereby conferred has expired.
- (d) The Directors are hereby empowered pursuant to sections 23 and 24(1) of the 1983 Act to allot equity securities within the meaning of the said section 23 for cash pursuant to the authority conferred by paragraph (c) of this article as if section 23(1) of the said 1983 Act did not apply to any such allotment. The Company may before the expiry of such authority make an offer or agreement which would or might require equity securities to be allotted after such expiry and the Directors may allot equity securities in pursuance of such an offer or agreement as if the power conferred by this paragraph (d) had not expired.
- (e) Nothing in these articles shall preclude the Directors from recognising a renunciation of the allotment of any shares by any allottee in favour of some other person.
- 9. If by the conditions of allotment of any share the whole or part of the amount or issue price thereof shall be payable by instalments, every such instalment when due shall be paid to the Company by the person who for the time being shall be the Holder of the share.
- 10. The Company may pay commission to any person in consideration of a person subscribing or agreeing to subscribe, whether absolutely or conditionally, for any shares in the Company or procuring or agreeing to procure subscriptions, whether absolute or conditional, for any shares in the Company on such terms and subject to such conditions as the Directors may determine, including, without limitation, by paying cash or allotting and issuing fully or partly paid shares or any combination of the two. The Company may also, on any issue of shares, pay such brokerage as may be lawful.
- 11. Except as required by law, no person shall be recognised by the Company as holding any share upon any trust, and the Company shall not be bound by or be compelled in any way to recognise (even when having notice thereof) any equitable, contingent, future or partial interest in any share or any interest in any fractional part of a share or (except only as by these articles or by law otherwise provided) any other rights in respect of any share except an absolute right to the entirety thereof in the Holder.
- 12. No person shall be entitled to a share certificate in respect of any ordinary share held by them in the share capital of the Company, whether such ordinary share was allotted or transferred to them, and the Company shall not be bound to issue a share certificate to any such person entered in the Register.
- 13. The Company shall not give, whether directly or indirectly and whether by means of a loan, guarantee, the provision of security or otherwise, any financial assistance for the purpose of or in connection with a purchase or subscription made or to be made by any person of or for any shares in the Company or in its holding company, except as permitted by section 60 of the Act.
- 14. (a) The Company shall have a first and paramount lien on every share (not being a fully paid share) for all moneys (whether presently payable or not) payable at a fixed time or called in respect of that share. The Directors, at any time, may declare any share to be wholly or in part exempt from the provisions of this article. The Company's lien on a share shall extend to all moneys payable in respect of it.
  - (b) The Company may sell in such manner as the Directors determine any share on which the Company has a lien if a sum in respect of which the lien exists is presently payable and is not paid within fourteen Clear Days after notice demanding payment, and stating that if the notice is not complied with the share may be sold, has been given to the Holder of the share or to the person entitled to it by reason of the death or bankruptcy of the Holder.

- (c) To give effect to a sale, the Directors may authorise some person to execute an instrument of transfer of the share sold to, or in accordance with the directions of, the purchaser. The transferee shall be entered in the Register as the Holder of the share comprised in any such transfer and he shall not be bound to see to the application of the purchase moneys nor shall his title to the share be affected by any irregularity in or invalidity of the proceedings in reference to the sale, and after the name of the transferee has been entered in the Register, the remedy of any person aggrieved by the sale shall be in damages only and against the Company exclusively.
- (d) The net proceeds of the sale, after payment of the costs, shall be applied in payment of so much of the sum for which the lien exists as is presently payable and any residue (upon surrender to the Company for cancellation of the certificate for the shares sold and subject to a like lien for any moneys not presently payable as existed upon the shares before the sale) shall be paid to the person entitled to the shares at the date of the sale.
- 15. (a) Subject to the terms of allotment, the Directors may make calls upon the members in respect of any moneys unpaid on their shares and each member (subject to receiving at least fourteen Clear Days' notice specifying when and where payment is to be made) shall pay to the Company as required by the notice the amount called on his shares. A call may be required to be paid by instalments. A call may be revoked before receipt by the Company of a sum due thereunder, in whole or in part and payment of a call may be postponed in whole or in part. A person upon whom a call is made shall remain liable for calls made upon him notwithstanding the subsequent transfer of the shares in respect of which the call was made.
  - (b) A call shall be deemed to have been made at the time when the resolution of the Directors authorising the call was passed.
  - (c) The joint Holders of a share shall be jointly and severally liable to pay all calls in respect thereof.
  - (d) If a call remains unpaid after it has become due and payable the person from whom it is due and payable shall pay interest on the amount unpaid from the day it became due until it is paid at the rate fixed by the terms of allotment of the share or in the notice of the call or, if no rate is fixed, at the appropriate rate (as defined by the Acts) but the Directors may waive payment of the interest wholly or in part.
  - (e) An amount payable in respect of a share on allotment or at any fixed date, whether in respect of nominal value or as an instalment of a call, shall be deemed to be a call and if it is not paid the provisions of these articles shall apply as if that amount had become due and payable by virtue of a call.
  - (f) Subject to the terms of allotment, the Directors may make arrangements on the issue of shares for a difference between the Holders in the amounts and times of payment of calls on their shares.
  - (g) The Directors, if they think fit, may receive from any member willing to advance the same all or any part of the moneys uncalled and unpaid upon any shares held by him, and upon all or any of the moneys so advanced may pay (until the same would, but for such advance, become payable) interest at such rate, not exceeding (unless the Company in general meeting otherwise directs) fifteen percent per annum, as may be agreed upon between the Directors and the member paying such sum in advance.
  - (h) (i) If a member fails to pay any call or instalment of a call on the day appointed for payment thereof, the Directors, at any time thereafter and during such times as any part of the call or instalment remains unpaid, may serve a notice on him requiring payment of so much of the call or instalment as is unpaid together with any interest which may have accrued.

- (ii) The notice shall name a further day (not earlier than the expiration of fourteen Clear Days from the date of service of the notice) on or before which the payment required by the notice is to be made, and shall state that in the event of non-payment at or before the time appointed the shares in respect of which the call was made will be liable to be forfeited.
- (iii) If the requirements of any such notice as aforesaid are not complied with then, at any time thereafter before the payment required by the notice has been made, any shares in respect of which the notice has been given may be forfeited by a resolution of the Directors to that effect. The forfeiture shall include all dividends or other moneys payable in respect of the forfeited shares and not paid before forfeiture. The Directors may accept a surrender of any share liable to be forfeited hereunder.
- (iv) On the trial or hearing of any action for the recovery of any money due for any call it shall be sufficient to prove that the name of the member sued is entered in the Register as the Holder, or one of the Holders, of the shares in respect of which such debt accrued, that the resolution making the call is duly recorded in the minute book and that notice of such call was duly given to the member sued, in pursuance of these articles, and it shall not be necessary to prove the appointment of the Directors who made such call nor any other matters whatsoever, but the proof of the matters aforesaid shall be conclusive evidence of the debt.
- (i) A forfeited share may be sold or otherwise disposed of on such terms and in such manner as the Directors think fit and at any time before a sale or disposition the forfeiture may be cancelled on such terms as the Directors think fit. Where for the purposes of its disposal such a share is to be transferred to any person, the Directors may authorise some person to execute an instrument of transfer of the share to that person. The Company may receive the consideration, if any, given for the share on any sale or disposition thereof and may execute a transfer of the share in favour of the person to whom the share is sold or disposed of and thereupon he shall be registered as the Holder of the share and shall not be bound to see to the application of the purchase money, if any, nor shall his title to the share be affected by any irregularity or invalidity in the proceedings in reference to the forfeiture, sale or disposal of the share.
- (j) A person whose shares have been forfeited shall cease to be a member in respect of the forfeited shares, but nevertheless shall remain liable to pay to the Company all moneys which, at the date of forfeiture, were payable by him to the Company in respect of the shares, without any deduction or allowance for the value of the shares at the time of forfeiture but his liability shall cease if and when the Company shall have received payment in full of all such moneys in respect of the shares.
- (k) A statutory declaration that the declarant is a Director or the Secretary of the Company, and that a share in the Company has been duly forfeited on the date stated in the declaration, shall be conclusive evidence of the facts therein stated as against all persons claiming to be entitled to the share.
- (1) The provisions of these articles as to forfeiture shall apply in the case of non-payment of any sum which, by the terms of issue of a share, becomes payable at a fixed time, whether on account of the nominal value of the share or by way of premium, as if the same had been payable by virtue of a call duly made and notified.
- (m) The Directors may accept the surrender of any share which the Directors have resolved to have been forfeited upon such terms and conditions as may be agreed and, subject to any such terms and conditions, a surrendered share shall be treated as if it has been forfeited.

# TRANSFER OF SHARES

16. (a) The instrument of transfer of any share may be executed for and on behalf of the transferor by the Secretary, an Assistant Secretary or any such person that the Secretary or an Assistant Secretary nominates for that purpose (whether in respect of specific transfers or pursuant to a

general standing authorisation), and the Secretary, Assistant Secretary or the relevant nominee shall be deemed to have been irrevocably appointed agent for the transferor of such share or shares with full power to execute, complete and deliver in the name of and on behalf of the transferor of such share or shares all such transfers of shares held by the members in the share capital of the Company. Any document which records the name of the transferor, the name of the transferee, the class and number of shares agreed to be transferred, the date of the agreement to transfer shares and the price per share, shall, once executed by the transferor or the Secretary, Assistant Secretary or the relevant nominee as agent for the transferor, be deemed to be a proper instrument of transfer for the purposes of section 81 of the Act. The transferor shall be deemed to remain the Holder of the share until the name of the transferee is entered on the Register in respect thereof, and neither the title of the transferee nor the title of the transferor shall be affected by any irregularity or invalidity in the proceedings in reference to the sale should the Directors so determine.

- (b) The Company, at its absolute discretion, may, or may procure that a subsidiary of the Company shall, pay Irish stamp duty arising on a transfer of shares on behalf of the transfere of such shares of the Company. If stamp duty resulting from the transfer of shares in the Company which would otherwise be payable by the transferee is paid by the Company or any subsidiary of the Company on behalf of the transferee, then in those circumstances, the Company shall, on its behalf or on behalf of its subsidiary (as the case may be), be entitled to (i) seek reimbursement of the stamp duty from the transferee, (ii) set-off the stamp duty against any dividends payable to the transferee of those shares and (iii) claim a first and permanent lien on the shares on which stamp duty has been paid by the Company or its subsidiary for the amount of stamp duty paid. The Company's lien shall extend to all dividends paid on those shares.
- (c) Notwithstanding the provisions of these articles and subject to any regulations made under section 239 of the 1990 Act, title to any shares in the Company may also be evidenced and transferred without a written instrument in accordance with section 239 of the 1990 Act or any regulations made thereunder. The Directors shall have power to permit any class of shares to be held in uncertificated form and to implement any arrangements they think fit for such evidencing and transfer which accord with such regulations and in particular shall, where appropriate, be entitled to disapply or modify all or part of the provisions in these articles with respect to the requirement for written instruments of transfer and share certificates (if any), in order to give effect to such regulations.
- 17. Subject to such of the restrictions of these articles and to such of the conditions of issue of any share warrants as may be applicable, the shares of any member and any share warrant may be transferred by instrument in writing in any usual or common form or any other form which the Directors may approve.
- 18. (a) The Directors in their absolute discretion and without assigning any reason therefor may decline to register:
  - (i) any transfer of a share which is not fully paid; or
  - (ii) any transfer to or by a minor or person of unsound mind;

but this shall not apply to a transfer of such a share resulting from a sale of the share through a stock exchange on which the share is listed.

- (b) The Directors may decline to recognise any instrument of transfer unless:
  - (i) the instrument of transfer is accompanied by any evidence the Directors may reasonably require to show the right of the transferor to make the transfer;
  - (ii) the instrument of transfer is in respect of one class of share only;
  - (iii) the instrument of transfer is in favour of not more than four transferees; and

(iv) it is lodged at the Office or at such other place as the Directors may appoint.

- 19. If the Directors refuse to register a transfer, they shall, within two months after the date on which the transfer was lodged with the Company, send to the transfere notice of the refusal.
- 20. (a) The Directors may from time to time fix a record date for the purposes of determining the rights of members to notice of and/or to vote at any general meeting of the Company. The record date shall not precede the date upon which the resolution fixing the record date is adopted by the Directors, and the record date shall be not more than eighty nor less than ten days before the date of such meeting. If no record date is fixed by the Directors, the record date for determining members entitled to notice of or to vote at a meeting of the members shall be the close of business on the day next preceding the day on which notice is given. Unless the Directors determine otherwise, a determination of members of record entitled to notice of or to vote at a meeting.
  - (b) In order that the Directors may determine the members entitled to receive payment of any dividend or other distribution or allotment of any rights or the members entitled to exercise any rights in respect of any change, conversion or exchange of shares, or for the purpose of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than thirty nor less than two days prior to such action. If no record date is fixed, the record date for determining members for such purpose shall be at the close of business on the day on which the Directors adopt the resolution relating thereto.
- 21. Registration of transfers may be suspended at such times and for such period, not exceeding in the whole 30 days in each year, as the Directors may from time to time determine subject to the requirements of section 121 of the Act.
- 22. All instruments of transfer shall upon their being lodged with the Company remain the property of the Company and the Company shall be entitled to retain them.
- 23. Subject to the provisions of these articles, whenever as a result of a consolidation of shares or otherwise any members would become entitled to fractions of a share, the Directors may sell or cause to be sold, on behalf of those members, the shares representing the fractions for the best price reasonably obtainable to any person and distribute the proceeds of sale (subject to any applicable tax and abandoned property laws) in due proportion among those members, and the Directors may authorise some person to execute an instrument of transfer of the shares to, or in accordance with the directions of, the purchaser. The transferee shall not be bound to see to the application of the purchase money nor shall his title to the shares be affected by any irregularity in or invalidity of the proceedings in reference to the sale.

## TRANSMISSION OF SHARES

- 24. In the case of the death of a member, the survivor or survivors where the deceased was a joint Holder, and the personal representatives of the deceased where he was a sole Holder, shall be the only persons recognised by the Company as having any title to his interest in the shares; but nothing herein contained shall release the estate of a deceased joint Holder from any liability in respect of any share which had been jointly held by him with other persons.
- 25. Any person becoming entitled to a share in consequence of the death or bankruptcy of a member may, upon such evidence being produced as may from time to time properly be required by the Directors and subject as herein provided, elect either to be registered himself as Holder of the share or to have some person nominated by him registered as the transferee thereof, but the Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the shares by that member before his death or bankruptcy, as the case may be.
- 26. If the person so becoming entitled elects to be registered himself, he shall deliver or send to the Company a notice in writing signed by him stating that he so elects. If he elects to have another person registered, he shall testify his election by executing to that person a transfer of the share. All the

limitations, restrictions and provisions of these regulations relating to the right to transfer and the registration of transfers of shares shall be applicable to any such notice or transfer as aforesaid as if the death or bankruptcy of the member had not occurred and the notice of transfer were a transfer signed by that member.

27. A person becoming entitled to a share by reason of the death or bankruptcy of the Holder shall be entitled to the same dividends and other advantages to which he would be entitled if he were the registered Holder of the share, except that he shall not, before being registered as a member in respect of the share, be entitled in respect of it to exercise any right conferred by membership in relation to the meetings of the Company, so, however, that the Directors may at any time give notice requiring such person to elect either to be registered himself or to transfer the share, and if the notice is not complied with within 90 days, the Directors may thereupon withhold payment of all dividends, bonuses or other moneys payable in respect of the share until the requirements of the notice have been complied with.

### ALTERATION OF CAPITAL

- 28. The Company may from time to time by Ordinary Resolution increase the authorised share capital by such sum, to be divided into shares of such amount, as the resolution shall prescribe.
- 29. The Company may by Ordinary Resolution:
  - (a) consolidate and divide all or any of its share capital into shares of larger amount than its existing shares;
  - (b) subdivide its existing shares, or any of them, into shares of smaller amount than is fixed by the memorandum of association subject, nevertheless, to section 68(1)(d) of the Act; or
  - (c) cancel any shares which, at the date of the passing of the resolution, have not been taken or agreed to be taken by any person and reduce the amount of its authorised share capital by the amount of the shares so cancelled.
- 30. The Company may by Special Resolution reduce its share capital, any capital redemption reserve fund or any share premium account in any manner and with and subject to any incident authorised, and consent required, by law.

### GENERAL MEETINGS

- 31. The Company shall in each year hold a general meeting as its annual general meeting in addition to any other meeting in that year, and shall specify the meeting as such in the notices calling it. Not more than fifteen months shall elapse between the date of one annual general meeting of the Company and that of the next. This article shall not apply in the case of the first general meeting, in respect of which the Company shall convene the meeting within the time periods required by the Act.
- 32. Subject to section 140 of the Act, all general meetings of the Company may be held outside of Ireland.
- 33. All general meetings other than annual general meetings shall be called extraordinary general meetings.
- 34. The Directors may, whenever they think fit, convene an extraordinary general meeting, and extraordinary general meetings shall also be convened on such requisition, or in default may be convened by such requisitionists, as provided in section 132 of the Act.
- 35. All provisions of these articles relating to general meetings of the Company shall, mutatis mutandis, apply to every separate general meeting of the Holders of any class of shares in the capital of the Company, except that:
  - (a) the necessary quorum shall be two or more persons holding or representing by proxy (whether or not such Holder actually exercises his voting rights in whole, in part or at all at the relevant general meeting) at least one-half in nominal value of the issued shares of the class or, at any adjourned meeting of such Holders, one Holder present in person or by proxy, whatever the amount of his holding, shall be deemed to constitute a meeting;

- (b) any Holder of shares of the class present in person or by proxy may demand a poll; and
- (c) on a poll, each Holder of shares of the class shall have one vote in respect of every share of the class held by him.
- 36. A Director shall be entitled, notwithstanding that he is not a member, to attend and speak at any general meeting and at any separate meeting of the Holders of any class of shares in the Company.

#### NOTICE OF GENERAL MEETINGS

- 37. (a) Subject to the provisions of the Acts allowing a general meeting to be called by shorter notice, an annual general meeting, and an extraordinary general meeting called for the passing of a special resolution, shall be called by not less than twenty-one Clear Days' notice and all other extraordinary general meetings shall be called by not less than fourteen Clear Days' notice.
  - (b) Any notice convening a general meeting shall specify the time and place of the meeting and, in the case of special business, the general nature of that business and, in reasonable prominence, that a member entitled to attend and vote is entitled to appoint a proxy to attend, speak and vote in his place and that a proxy need not be a member of the Company. It shall also give particulars of any Directors who are to retire at the meeting and of any persons who are recommended by the Directors for appointment or re-appointment as Directors at the meeting or in respect of whom notice has been duly given to the Company of the intention to propose them for appointment or re-appointment as Directors at the meeting. Provided that the latter requirement shall only apply where the intention to propose the person has been received by the Company in accordance with the provisions of these articles. Subject to any restrictions imposed on any shares, the notice of the meeting shall be given to all the members of the Company as of the record date set by the Directors and to the Directors and the Auditors.
  - (c) The accidental omission to give notice of a meeting to, or the non-receipt of notice of a meeting by, any person entitled to receive notice shall not invalidate the proceedings at the meeting.
- 38. Where, by any provision contained in the Acts, extended notice is required of a resolution, the resolution shall not be effective (except where the Directors of the Company have resolved to submit it) unless notice of the intention to move it has been given to the Company not less than twenty-eight days (or such shorter period as the Acts permit) before the meeting at which it is moved, and the Company shall give to the members notice of any such resolution as required by and in accordance with the provisions of the Acts.

### PROCEEDINGS AT GENERAL MEETINGS

- 39. All business shall be deemed special that is transacted at an extraordinary general meeting, and also all that is transacted at an annual general meeting, with the exception of declaring a dividend, the consideration of the accounts, balance sheets and the reports of the Directors and auditors, the election of Directors, the re-appointment of the retiring auditors and the fixing of the remuneration of the auditors.
- 40. At any annual general meeting of the members, only such nominations of persons for election to the Board shall be made, and only such other business shall be conducted or considered, as shall have been properly brought before the meeting. For nominations to be properly made at an annual general meeting, and proposals of other business to be properly brought before an annual meeting, nominations and proposals of other business must be: (a) specified in the Company's notice of meeting (or any supplement thereto) given by or at the direction of the Board, (b) otherwise properly made at the annual general meeting, by or at the direction of the Board or (c) otherwise properly requested to be brought before the annual general meeting by a member of the Company in accordance with these articles. For nominations of persons for election to the Board or proposals of other business to be properly requested by a member to be made at an annual general meeting, a member must (i) be a member at the time of giving of notice of such annual general meeting by or at the direction of the Board and at the time of the annual general meeting, (ii) be entitled to vote at such annual general meeting and (iii) comply with

the procedures set forth in these articles as to such business or nomination. The immediately preceding sentence shall be the exclusive means for a member to make nominations or other business proposals (other than matters properly brought under Rule 14a-8 under the U.S. Securities Exchange Act of 1934, as amended (the "Exchange Act") and included in the Company's notice of meeting) before an annual general meeting of members.

- 41. At any extraordinary general meeting of the members, only such business shall be conducted or considered, as shall have been properly brought before the meeting pursuant to the Company's notice of meeting. To be properly brought before an extraordinary general meeting, proposals of business must be (a) specified in the Company's notice of meeting (or any supplement thereto) given by or at the direction of the Board, (b) otherwise properly brought before the extraordinary general meeting, by or at the direction of the Board, or (c) otherwise properly brought before the meeting by any members of the Company pursuant to the valid exercise of power granted to them under the Acts.
- 42. Nominations of persons for election to the Board may be made at an extraordinary general meeting of members at which directors are to be elected pursuant to the Company's notice of meeting (a) by or at the direction of the Board, (b) by any members of the Company pursuant to the valid exercise of power granted to them under the Acts, or (c) provided that the Board has determined that directors shall be elected at such meeting, by any member of the Company who (i) is a member at the time of giving of notice of such extraordinary general meeting and at the time of the extraordinary general meeting, (ii) is entitled to vote at the meeting and (iii) complies with the procedures set forth in these articles as to such nomination. The immediately preceding sentence shall be the exclusive means for a member to make nominations (other than matters properly brought under Rule 14a-8 under the Exchange Act and included in the Company's notice of meeting) before an extraordinary general meeting of members.
- 43. Except as otherwise provided by law, the memorandum of association or these articles, the Chairman of any general meeting shall have the power to determine whether a nomination or any other business proposed to be brought before the general meeting was made or proposed, as the case may be, in accordance with these articles and, if any proposed nomination or other business is not in compliance with these articles, to declare that no action shall be taken on such nomination or other proposal and such nomination or other proposal shall be disregarded.
- 44. No business shall be transacted at any general meeting unless a quorum is present at the time when the meeting proceeds to business. The Holders of shares, present in person or by proxy (whether or not such Holder actually exercises his voting rights in whole, in part or at all at the relevant general meeting), entitling them to exercise a majority of the voting power of the Company on the relevant record date shall constitute a quorum.
- 45. Any general meeting duly called at which a quorum not present shall be adjourned and the Company shall provide notice pursuant to article 37 in the event that such meeting is to be reconvened.
- 46. The Chairman, if any, of the Board shall preside as Chairman at every general meeting of the Company, or if there is no such Chairman, or if he is not present within fifteen minutes after the time appointed for the holding of the meeting or is unwilling to act, the Directors present shall elect one of their number to be Chairman of the meeting.
- 47. If at any meeting no Director is willing to act as Chairman or if no Director is present within fifteen minutes after the time appointed for holding the meeting, the members present shall choose one of their number to be Chairman of the meeting.
- 48. The Chairman may, with the consent of any meeting at which a quorum is present, and shall if so directed by the meeting, adjourn the meeting from time to time and from place to place without notice other than by announcement of the time and place of the adjourned meeting by the Chairman of the meeting. The Chairman of the meeting may at any time without the consent of the meeting adjourn the meeting to another time and/or place if, in his opinion, it would facilitate the conduct of the business of the meeting to do so or if he is so directed by the Board. Save as aforesaid, it shall not be necessary to give any notice of an adjournment or of the business to be transacted at an adjourned meeting.
- 49. At any general meeting a resolution put to the vote of the meeting shall be decided on a show of hands unless a poll is (before or on the declaration of the result of the show of hands) demanded by:
  - (a) the Chairman; or

- (b) by at least three members present in person or by proxy; or
- (c) by any member or members present in person or by proxy and representing not less than one-tenth of the total voting rights of all the members having the right to vote at the meeting; or
- (d) by a member or members holding shares in the Company conferring the right to vote at the meeting being shares on which an aggregate sum has been paid up equal to not less than one-tenth of the total sum paid up on all the shares conferring that right.

Unless a poll is so demanded, a declaration by the Chairman that a resolution has, on a show of hands, been carried or carried unanimously, or by a particular majority, or lost, and an entry to that effect in the book containing the minutes of the proceedings of the Company, shall be conclusive evidence of the fact without proof of the number or proportion of the votes recorded in favour of or against such resolution.

The demand for a poll may be withdrawn.

- 50. Except as provided in article 51, if a poll is duly demanded it shall be taken in such manner as the Chairman directs, and the result of the poll shall be deemed to be the resolution of the meeting at which the poll was demanded.
- 51. A poll demanded on the election of the Chairman or on a question of adjournment shall be taken forthwith. A poll demanded on any other question shall be taken at such time as the Chairman of the meeting directs, and any business other than that on which a poll has been demanded may be proceeded with pending the taking of the poll.
- 52. Where there is an equality of votes, whether on a show of hands or on a poll, the Chairman of the meeting at which the show of hands takes place or at which the poll is demanded shall be entitled to a casting vote in addition to any other vote he may have.
- 53. Unless the Directors otherwise determine, no member shall be entitled to vote at any general meeting or any separate meeting of the Holders of any class of shares in the Company, either in person or by proxy, or to exercise any privilege as a member in respect of any share held by him unless all monies then payable by him in respect of that share have been paid.

# ADVANCE NOTICE OF MEMBER BUSINESS AND NOMINATIONS

- 54. Without qualification or limitation, subject to article 67, for any nominations or any other business to be properly brought before an annual general meeting by a member pursuant to article 40, the member must have given timely notice thereof (including, in the case of nominations, the completed and signed questionnaire, representation and agreement required by article 68), and timely updates and supplements thereof, in writing to the Secretary, and such other business must otherwise be a proper matter for member action.
- 55. To be timely, a member's notice shall be delivered to the Secretary at the Office not earlier than the close of business on the 120th day and not later than the close of business on the 90th day prior to the first anniversary of the preceding year's annual general meeting; provided, however, that in the event that the date of the annual general meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the member must be so delivered not earlier than the close of business on the 120th day prior to the date of such annual general meeting and not later than the close of business on the later of the 90th day prior to the date of such annual general meeting or, if the first public announcement of the date of such annual general meeting is less than 100 days prior to the date of such annual general meeting, the 10th day following the day on which public announcement of the date of such meeting is first made by the Company; provided, further, that with respect to the 2014 annual general meeting, notice by the member must be so delivered not later than the 10th day following the day on which public announcement of the date of such meeting is first made by the Company; provided, further, that with respect to the 2014 annual general meeting, notice by the member must be so delivered not later than the 10th day following the day on which public announcement of the date of such meeting is first made by the Company. In no event shall any adjournment or postponement of an annual general meeting, or the public announcement thereof, commence a new time period for the giving of a member's notice as described above.

- 56. Notwithstanding anything in article 55 to the contrary, in the event that the number of directors to be elected to the Board is increased by the Board, and there is no public announcement by the Company naming all of the nominees for director or specifying the size of the increased Board at least 100 days prior to the first anniversary of the preceding year's annual general meeting, a member's notice required by articles 54-57 shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delivered to the Secretary at the Office not later than the close of business on the 10th day following the day on which such public announcement is first made by the Company.
- 57. In addition, to be considered timely, a member's notice shall further be updated and supplemented, if necessary, so that the information provided or required to be provided in such notice shall be true and correct as of the record date for the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to the Secretary at the Office not later than five (5) business days after the record date for the meeting or any adjournment required to be made as of the record date, and not later than eight (8) business days prior to the date for the meeting or any adjournment or postponement thereof.
- 58. Subject to article 67, in the event the Company calls an extraordinary general meeting of members for the purpose of electing one or more directors to the Board, any member may nominate a person or persons (as the case may be) for election to such position(s) as specified in the Company's notice of meeting, provided that the member gives timely notice thereof (including the completed and signed questionnaire, representation and agreement required by article 68), and timely updates and supplements thereof, in writing, to the Secretary.
- 59. To be timely, a member's notice shall be delivered to the Secretary at the Office not earlier than the close of business on the 120th day prior to the date of such extraordinary general meeting and not later than the close of business on the later of the 90th day prior to the date of such extraordinary general meeting or, if the first public announcement of the date of such extraordinary general meeting is less than 100 days prior to the date of such extraordinary general meeting, the 10th day following the day on which public announcement is first made of the date of the extraordinary general meeting, and of the nominees proposed by the Board to be elected at such meeting. In no event shall any adjournment or postponement of an extraordinary general meeting, or the public announcement thereof, commence a new time period for the giving of a member's notice as described above.
- 60. In addition, to be considered timely, a member's notice shall further be updated and supplemented, if necessary, so that the information provided or required to be provided in such notice shall be true and correct as of the record date for the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to the Secretary at the Office not later than five (5) business days after the record date for the meeting, any adjournment or postponement thereof in the case of the update and supplement required to be made as of the record date, and not later than eight (8) business days prior to the date for the meeting, any adjournment or postponement thereof in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof.
- 61. To be in proper form, a member's notice (whether given pursuant to articles 54-57 or articles 58-60) to the Secretary must include the following, as applicable:
- 62. As to the member giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made, a member's notice must set forth: (i) the name and address of such member, as they appear on the Company's books, of such beneficial owner, if any, and of their respective affiliates or associates or others acting in concert therewith, (ii) (A) the class or series and number of shares of the Company which are, directly or indirectly, owned beneficially and of record by such member, such beneficial owner and their respective affiliates or associates or others acting in concert therewith, (ii) (A) the class or series and number of shares of the Company which are, directly or indirectly, owned beneficially and of record by such member, such beneficial owner and their respective affiliates or associates or others acting in concert therewith, (B) any option, warrant, convertible security, share appreciation right, or similar right with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class or series of

shares of the Company or with a value derived in whole or in part from the value of any class or series of shares of the Company, or any derivative or synthetic arrangement having the characteristics of a long position in any class or series of shares of the Company, or any contract, derivative, swap or other transaction or series of transactions designed to produce economic benefits and risks that correspond substantially to the ownership of any class or series of shares of the Company, including due to the fact that the value of such contract, derivative, swap or other transaction or series of transactions is determined by reference to the price, value or volatility of any class or series of shares of the Company, whether or not such instrument, contract or right shall be subject to settlement in the underlying class or series of shares of the Company, through the delivery of cash or other property, or otherwise, and without regard to whether the member, the beneficial owner, if any, or any affiliates or associates or others acting in concert therewith, may have entered into transactions that hedge or mitigate the economic effect of such instrument, contract or right, or any other direct or indirect opportunity to profit or share in any profit derived from any increase or decrease in the value of shares of the Company (any of the foregoing, a "Derivative Instrument") directly or indirectly owned beneficially by such member, the beneficial owner, if any, or any affiliates or associates or others acting in concert therewith, (C) any proxy, contract, arrangement, understanding, or relationship pursuant to which such member has a right to vote any class or series of shares of the Company, (D) any agreement, arrangement, understanding, relationship or otherwise, including any repurchase or similar so-called "stock borrowing" agreement or arrangement, involving such member, directly or indirectly, the purpose or effect of which is to mitigate loss to, reduce the economic risk (of ownership or otherwise) of any class or series of the shares of the Company by, manage the risk of share price changes for, or increase or decrease the voting power of, such member with respect to any class or series of the shares of the Company, or which provides, directly or indirectly, the opportunity to profit or share in any profit derived from any decrease in the price or value of any class or series of the shares of the Company (any of the foregoing, a "Short Interest"), (E) any rights to dividends on the shares of the Company owned beneficially by such member that are separated or separable from the underlying shares of the Company, (F) any proportionate interest in shares of the Company or Derivative Instruments held, directly or indirectly, by a general or limited partnership in which such member is a general partner or, directly or indirectly, beneficially owns an interest in a general partner of such general or limited partnership, (G) any performance-related fees (other than an asset-based fee) that such member is entitled to based on any increase or decrease in the value of shares of the Company or Derivative Instruments, if any, including without limitation any such interests held by members of such member's immediate family sharing the same household, (H) any significant equity interests or any Derivative Instruments or Short Interests in any principal competitor of the Company held by such member, and (I) any direct or indirect interest of such member in any contract with the Company, any affiliate of the Company or any principal competitor of the Company (including, in any such case, any employment agreement, collective bargaining agreement or consulting agreement), and (iii) any other information relating to such member and beneficial owner, if any, that would be required to be disclosed in a proxy statement and form or proxy or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal and/or for the election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder.

- 63. If the notice relates to any business other than a nomination of a director or directors that the member proposes to bring before the meeting, a member's notice must, in addition to the matters set forth in article 62 above, also set forth: (i) a brief description of the business desired to be brought before the meeting, the reasons for conducting such business at the meeting and any material interest of such member and beneficial owner, if any, in such business, (ii) the text of the proposal or business (including the text of any resolutions proposed for consideration and, in the event that such proposal or business includes a proposal to amend these articles, the text of the proposed amendment), and (iii) a description of all agreements, arrangements and understandings between such member and beneficial owner, if any, and any other person or persons (including their names) in connection with the proposal of such business by such member.
- 64. As to each person, if any, whom the member proposes to nominate for election or re-election to the Board, a member's notice must, in addition to the matters set forth in article 62 above, also set forth: (i) all information relating to such person that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder (including such person's written consent to being named in the proxy

statement as a nominee and to serving as a director if elected) and (ii) a description of all direct and indirect compensation and other material monetary agreements, arrangements and understandings during the past three years, and any other material relationships, between or among such member and beneficial owner, if any, and their respective affiliates and associates, or others acting in concert therewith, on the one hand, and each proposed nominee, and his or her respective affiliates and associates, or others acting in concert therewith, on the other hand, including, without limitation all information that would be required to be disclosed pursuant to Rule 404 promulgated under Regulation S-K under the Exchange Act if the member making the nomination and any beneficial owner on whose behalf the nomination is made, if any, or any affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of such rule and the nominee were a director or executive officer of such registrant.

- 65. With respect to each person, if any, whom the member proposes to nominate for election or re-election to the Board, a member's notice must, in addition to the matters set forth in articles 62 and 64 above, also include a completed and signed questionnaire, representation and agreement required by article 68 of these articles. The Company may require any proposed nominee to furnish such other information as may reasonably be required by the Company to determine the eligibility of such proposed nominee to serve as an independent director of the Company or that could be material to a reasonable member's understanding of the independence, or lack thereof, of such nominee.
- 66. Notwithstanding the provisions of these articles, a member shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in articles 54-68; provided, however, that any references in these articles to the Exchange Act or the rules promulgated thereunder are not intended to and shall not limit the separate and additional requirements set forth in these articles with respect to nominations or proposals as to any other business to be considered pursuant to articles 39-43.
- 67. Nothing in these articles shall be deemed to affect any rights (i) of members to request inclusion of proposals in the Company's proxy statement pursuant to Rule 14a-8 under the Exchange Act, (ii) of the holders of any series of preferred shares if and to the extent provided for under law, the memorandum of association or these articles or (iii) of members of the Company to bring business before an extraordinary general meeting pursuant to the valid exercise of power granted to them under the Acts. Subject to Rule 14a-8 under the Exchange Act, nothing in these articles shall be construed to permit any member, or give any member the right, to include or have disseminated or described in the Company's proxy statement any nomination of director or directors or any other business proposal.
- 68. Subject to the rights of members of the Company to propose nominations at an extraordinary general meeting pursuant to the valid exercise of power granted to them under the Acts, to be eligible to be a nominee for election or re-election as a director of the Company, a person must deliver (in accordance with the time periods prescribed for delivery of notice under articles 54-67) to the Secretary at the Office a written questionnaire with respect to the background and qualification of such person and the background of any other person or entity on whose behalf the nomination is being made (which questionnaire shall be provided by the Secretary upon written request), and a written representation and agreement (in the form provided by the Secretary upon written request) that such person (A) is not and will not become a party to (1) any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such person, if elected as a director of the Company, will act or vote on any issue or question (a "Voting Commitment") that has not been disclosed to the Company or (2) any Voting Commitment that could limit or interfere with such person's ability to comply, if elected as a director of the Company with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a director that has not been disclosed therein, and (C) in such person's individual capacity and on behalf of any person or entity on whose behalf the nomination is being made, would be in compliance, if elected as a director of the Company, and will comply with all applicable corporate governance, conflict of interest, confidentiality and share ownership and trading policies and guidelines of the Company publicly disclosed from time to time.

#### VOTES OF MEMBERS

- 69. Subject to any special rights or restrictions as to voting for the time being attached by or in accordance with these articles to any class of shares, on a show of hands every member present in person and every proxy shall have one vote, but so that no one member shall on a show of hands have more than one vote in respect of the aggregate number of shares of which he is the Holder, and on a poll every member who is present in person or by proxy shall have one vote for each share of which he is the Holder.
- 70. When there are joint Holders, the vote of the senior who tenders a vote, whether in person or by proxy, shall be accepted to the exclusion of the votes of the other joint Holders; and for this purpose, seniority shall be determined by the order in which the names stand in the Register.
- 71. A member of unsound mind, or in respect of whom an order has been made by any court having jurisdiction (whether in Ireland or elsewhere) in matters concerning mental disorder, may vote, whether on a show of hands or on a poll, by his committee, receiver, guardian or other person appointed by that court and any such committee, receiver, guardian or other person may vote by proxy on a show of hands or on a poll. Evidence to the satisfaction of the Directors of the authority of the person claiming to exercise the right to vote shall be received at the Office or at such other address as is specified in accordance with these articles for the receipt of appointments of proxy, not less than forty-eight hours before the time appointed for holding the meeting or adjourned meeting at which the right to vote is to be exercised and in default the right to vote shall not be exercisable.
- 72. No objection shall be raised to the qualification of any voter except at the meeting or adjourned meeting at which the vote objected to is given or tendered, and every vote not disallowed at such meeting shall be valid for all purposes. Any such objection made in due time shall be referred to the Chairman of the meeting, whose decision shall be final and conclusive.
- 73. Votes may be given either personally or by proxy.
- 74. (a) Every member entitled to attend and vote at a general meeting may appoint a proxy to attend, speak and vote on his behalf and may appoint more than one proxy to attend, speak and vote at the same meeting. The appointment of a proxy shall be in any form which the Directors may approve and, if required by the Company, shall be signed by or on behalf of the appointor. In relation to written proxies, a body corporate may sign a form of proxy under its common seal or under the hand of a duly authorised officer thereof or in such other manner as the Directors may approve. A proxy need not be a member of the Company. The appointment of a proxy in electronic or other form shall only be effective in such manner as the Directors may approve.
  - (b) Without limiting the foregoing, the Directors may from time to time permit appointments of a proxy to be made by means of an electronic or internet communication or facility and may in a similar manner permit supplements to, or amendments or revocations of, any such electronic or internet communication or facility to be made. The Directors may in addition prescribe the method of determining the time at which any such electronic or internet communication or facility is to be treated as received by the Company. The Directors may treat any such electronic or internet communication or facility which purports to be or is expressed to be sent on behalf of a Holder of a share as sufficient evidence of the authority of the person sending that instruction to send it on behalf of that Holder.
- 75. Any body corporate which is a member of the Company may authorise such person as it thinks fit to act as its representative at any meeting of the Company or of any class of members of the Company and the person so authorised shall be entitled to exercise the same powers on behalf of the body corporate which he represents as that body corporate could exercise if it were an individual member of the Company. The Company may require evidence from the body corporate of the due authorisation of such person to act as the representative of the relevant body corporate.
- 76. An appointment of proxy relating to more than one meeting (including any adjournment thereof) having once been received by the Company for the purposes of any meeting shall not require to be delivered, deposited or received again by the Company for the purposes of any subsequent meeting to which it relates.

- 77. Receipt by the Company of an appointment of proxy in respect of a meeting shall not preclude a member from attending and voting at the meeting or at any adjournment thereof. An appointment proxy shall be valid, unless the contrary is stated therein, as well for any adjournment of the meeting as for the meeting to which it relates.
- 78. (a) A vote given or poll demanded in accordance with the terms of an appointment of proxy or a resolution authorising a representative to act on behalf of a body corporate shall be valid notwithstanding the death or insanity of the principal, or the revocation of the appointment of proxy or of the authority under which the proxy was appointed or of the resolution authorising the representative to act or transfer of the share in respect of which the proxy was appointed or the authorisation of the representative to act was given, provided that no intimation in writing (whether in electronic form or otherwise) of such death, insanity, revocation or transfer shall have been received by the Company at the Office, at least one hour before the commencement of the meeting or adjourned meeting at which the appointment of proxy is used or at which the representative acts; provided, however, that where such intimation is given in electronic form it shall have been received by the Company at least 24 hours (or such lesser time as the Directors may specify) before the commencement of the meeting.
  - (b) The Directors may send, at the expense of the Company, by post, electronic mail or otherwise, to the members forms for the appointment of a proxy (with or without stamped envelopes for their return) for use at any general meeting or at any class meeting, either in blank or nominating any one or more of the Directors or any other persons in the alternative.
- 79. The instrument appointing a proxy shall, be deemed to confer authority to demand or join in demanding a poll.
- 80. Subject to Section 141 of the 1963 Act, a resolution in writing signed by all of the members for the time being entitled to attend and vote on such resolution at a general meeting (or being bodies corporate by their duly authorised representatives) shall be as valid and effective for all purposes as if the resolution had been passed at a general meeting of the Company duly convened and held, and may consist of several documents in like form each signed by one or more persons, and if described as a special resolution shall be deemed to be a special resolution within the meaning of the 1963 Act. Any such resolution shall be served on the Company.

# DIRECTORS

- 81. The number of Directors shall not be less than two nor more than 15. The continuing Directors may act notwithstanding any vacancy in their body, provided that if the number of the Directors is reduced below the prescribed minimum the remaining Director or Directors shall appoint forthwith an additional Director or additional Directors to make up such minimum or shall convene a general meeting of the Company for the purpose of making such appointment. If, at any annual general meeting of the Company, the number of Directors is reduced below the prescribed minimum due to the failure of any Directors to be re-elected, then in those circumstances, the two Directors which receive the highest number of votes in favour of re-election shall be re-elected and shall remain Directors until such time as additional Directors have been appointed to replace them as Directors. If, at any annual general meeting of the Company, the number of Director which (excluding the re-elected Director) receives the highest number of votes in favour of re-elected, then that Director shall hold office until the next annual general meeting and the Director which (excluding the re-elected Director) receives the highest number of votes in favour of re-election shall be re-elected and shall remain a Director or Directors able or which (excluding the re-elected Director) receives the highest number of votes in favour of re-election shall be re-elected and shall remain a Director until such time as one or more additional Directors have been appointed to replace him or her. If there are no Director or Directors able or willing to act then any two members may summon a general meeting for the purpose of appointing Directors. Any additional Director so appointed shall hold office (subject to the provisions of the Acts and these articles) only until the conclusion of the annual general meeting of the Company next following such appointment unless he is re-elected during such meeting.
- 82. Each Director shall be paid a fee for their services at such rate as may from time to time be determined by the Board. The Directors may also be paid all travelling, hotel and other expenses properly incurred by them in attending and returning from meetings of the Directors or any committee of the Directors or general meetings of the Company or in connection with the business of the Company.

- 83. If any Director shall be called upon to perform extra services which in the opinion of the Directors are outside the scope of the ordinary duties of a Director, the Company may remunerate such Director either by a fixed sum or by a percentage of profits or otherwise as may be determined by a resolution passed at a meeting of the Directors and such remuneration may be either in addition to or in substitution for any other remuneration to which he may be entitled as a Director.
- 84. A Director (whether or not a member of the Company) shall be entitled to attend and speak at general meetings.
- 85. Unless the Company otherwise directs, a Director of the Company may be or become a Director or other officer of, or otherwise interested in, any company promoted by the Company or in which the Company may be interested as Holder or otherwise, and no such Director shall be accountable to the Company for any remuneration or other benefits received by him as a Director or officer of, or from his interest in, such other company.

### **BORROWING POWERS**

86. Subject to Part III of the 1983 Act, the Directors may exercise all the powers of the Company to borrow or raise money, and to mortgage or charge its undertaking, property, assets and uncalled capital or any part thereof and to issue debentures, debenture stock and other securities whether outright or as collateral security for any debt, liability or obligation of the Company or of any third party, without any limitation as to amount.

### POWERS AND DUTIES OF THE DIRECTORS

- 87. The business of the Company shall be managed by the Directors, who may pay all expenses incurred in promoting and registering the Company and may exercise all such powers of the Company as are not, by the Acts or by these articles, required to be exercised by the Company in general meeting, subject, nevertheless, to any of these articles and to the provisions of the Acts.
- 88. The Directors may from time to time and at any time by power of attorney appoint any company, firm or person or body of persons, whether nominated directly or indirectly by the Directors, to be the attorney or attorneys of the Company for such purposes and with such powers, authorities and discretions (not exceeding those vested in or exercisable by the Directors under these articles) and for such period and subject to such conditions as they may think fit, and any such power of attorney may contain such provisions for the protection of persons dealing with any such attorney as the Directors may think fit, and may also authorise any such attorney to delegate all or any of the powers, authorities and discretions vested in him.
- 89. The Company may exercise the powers conferred by section 41 of the Act with regard to having an official seal for use abroad and such powers shall be vested in the Directors.
- 90. A Director who is in any way, whether directly or indirectly, interested in a contract or proposed contract with the Company shall declare the nature of his interest at a meeting of the Directors in accordance with section 194 of the Act.
- 91. Save as otherwise provided by these articles, a Director shall not vote at a meeting of the Directors or a committee of Directors on any resolution concerning a matter in which he has, directly or indirectly, an interest which is material or a duty which conflicts or may conflict with the interests of the Company. A Director shall not be counted in the quorum present at a meeting in relation to any such resolution on which he is not entitled to vote.
  - (a) A Director shall be entitled (in the absence of some other material interest than is indicated below) to vote (and be counted in the quorum) in respect of any resolutions concerning any of the following matters, namely:
    - the giving of any security, guarantee or indemnity to him in respect of money lent by him to the Company or any of its subsidiary or associated companies or obligations incurred by him or by any other person at the request of or for the benefit of the Company or any of its subsidiary or associated companies;

- the giving of any security, guarantee or indemnity to a third party in respect of a debt or obligation of the Company or any of its subsidiary or associated companies for which he himself has assumed responsibility in whole or in part and whether alone or jointly with others under a guarantee or indemnity or by the giving of security;
- (iii) any proposal concerning any offer of shares or debentures or other securities of or by the Company or any of its subsidiary or associated companies for subscription, purchase or exchange in which offer he is or is to be interested as a participant in the underwriting or subunderwriting thereof;
- (iv) any proposal concerning any other company in which he is interested, directly or indirectly and whether as an officer or member or otherwise howsoever, provided that he is not the Holder of or beneficially interested in 1% or more of the issued shares of any class of such company or of the voting rights available to members of such company (or of a third company through which his interest is derived) (any such interest being deemed for the purposes of this article to be a material interest in all circumstances);
- (v) any proposal concerning the adoption, modification or operation of a superannuation fund or retirement benefits scheme under which he may benefit and which has been approved by or is subject to and conditional upon approval for taxation purposes by the appropriate Revenue authorities;
- (vi) any proposal concerning the adoption, modification or operation of any scheme for enabling employees (including full time executive Directors) of the Company and/or any subsidiary thereof to acquire shares in the Company or any arrangement for the benefit of employees of the Company or any of its subsidiaries under which the Director benefits or may benefit; or
- (vii) any proposal concerning the giving of any indemnity pursuant to article 143(a) or the discharge of the cost of any insurance coverage purchased or maintained pursuant to article 97 and article 143(b).
- (b) Where proposals are under consideration concerning the appointment (including fixing or varying the terms of appointment) of two or more Directors to offices or employments with the Company or any company in which the Company is interested, such proposals may be divided and considered in relation to each Director separately and in such case each of the Directors concerned (if not debarred from voting under subparagraph (a)(iv) of this article) shall be entitled to vote (and be counted in the quorum) in respect of each resolution except that concerning his own appointment,
- (c) If a question arises at a meeting of Directors or of a committee of Directors as to the materiality of a Director's interest or as to the right of any Director to vote and such question is not resolved by his voluntarily agreeing to abstain from voting, such question may be referred, before the conclusion of the meeting, to the Chairman of the meeting and his ruling in relation to any Director other than himself shall be final and conclusive. In relation to the Chairman, such question may be resolved by a resolution of a majority of the Directors (other than the Chairman) present at the meeting at which the question first arises.
- (d) For the purposes of this article, an interest of a person who is the spouse or a minor child of a Director shall be treated as an interest of the Director.
- (e) The Company by Ordinary Resolution may suspend or relax the provisions of this article to any extent or ratify any transaction not duly authorised by reason of a contravention of this article.
- 92. A Director may hold and be remunerated in respect of any other office or place of profit under the Company or any other company in which the Company may be interested (other than the office of auditor of the Company or any subsidiary thereof) in conjunction with his office of Director for such period and on such terms as to remuneration and otherwise as the Directors may determine, and no Director or intending Director shall be disqualified by his office from contracting or being interested,

directly or indirectly, in any contract or arrangement with the Company or any such other company either with regard to his tenure of any such other office or place of profit or as vendor, purchaser or otherwise nor shall any Director so contracting or being so interested be liable to account to the Company for any profits and advantages accruing to him from any such contract or arrangement by reason of such Director holding that office or of the fiduciary relationship thereby established.

- 93. The Directors may exercise the voting powers conferred by shares of any other company held or owned by the Company in such manner in all respects as they think fit and in particular they may exercise their voting powers in favour of any resolution appointing the Directors or any of them as Directors or officers of such other company or providing for the payment of remuneration or pensions to the Directors or officers of such other company.
- 94. Any Director may act by himself or his firm in a professional capacity for the Company, and he or his firm shall be entitled to remuneration for professional services as if he were not a Director, but nothing herein contained shall authorise a Director or his firm to act as auditor to the Company.
- 95. All cheques, promissory notes, drafts, bills of exchange and other negotiable instruments and all receipts for money paid to the Company shall be signed, drawn, accepted, endorsed or otherwise executed, as the case may be, by such person or persons and in such manner as the Directors shall from time to time by resolution determine.
- 96. The Directors shall cause minutes to be made in books provided for the purpose:
  - (a) of all appointments of officers made by the Directors;
  - (b) of the names of the Directors present at each meeting of the Directors and of any committee of the Directors; and
  - (c) of all resolutions and proceedings at all meetings of the Company and of the Directors and of committees of Directors.
- 97. The Directors may procure the establishment and maintenance of or participate in, or contribute to any non-contributory or contributory pension or superannuation fund, scheme or arrangement or life assurance scheme or arrangement for the benefit of, and pay, provide for or procure the grant of donations, gratuities, pensions, allowances, benefits or emoluments to any persons (including Directors or other officers) who are or shall have been at any time in the employment or service of the Company or of any company which is or was a subsidiary of the Company or of the predecessor in business of the Company or any such subsidiary or holding Company and the wives, widows, families, relatives or dependants of any such persons. The Directors may also procure the establishment and subsidy of or subscription to and support of any institutions, associations, clubs, funds or trusts calculated to be for the benefit of any such persons as aforesaid or otherwise to advance the interests and well being of the Company or of any such other Company as aforesaid, or its members, and payments for or towards the insurance of any such persons as aforesaid and subscriptions or guarantees of money for charitable or benevolent objects or for any exhibition or for any public, general or useful object. Provided that any Director shall be entitled to retain any benefit received by him under this article, subject only, where the Acts require, to disclosure to the members and the approval of the Company in general meeting.

## **DISQUALIFICATION OF DIRECTORS**

- 98. The office of a Director shall be vacated ipso facto if the Director:
  - (a) is restricted or disqualified to act as a Director under the provisions of Part VII of the 1990 Act; or
  - (b) resigns his office by notice in writing to the Company or in writing offers to resign and the Directors resolve to accept such offer; or
  - (c) is removed from office under article 103.

### APPOINTMENT, ROTATION AND REMOVAL OF DIRECTORS

- 99. At every annual general meeting of the Company, all of the Directors shall retire from office unless re-elected by Ordinary Resolution at the annual general meeting. A Director retiring at a meeting shall retain office until the close or adjournment of the meeting.
- 100. Every Director shall be eligible to stand for re-election at an annual general meeting.
- 101. If a Director offers himself for re-election, he shall be deemed to have been re-elected, unless at such meeting the Ordinary Resolution for the re-election of such Director has been defeated.
- 102. The Company may from time to time by Special Resolution increase or reduce the maximum number of Directors.
- 103. The Company may, by Ordinary Resolution, of which extended notice has been given in accordance with section 142 of the Act, remove any Director before the expiration of his period of office notwithstanding anything in these articles or in any agreement between the Company and such Director. Such removal shall be without prejudice to any claim such Director may have for damages for breach of any contract of service between him and the Company.
- 104. The Company may, by Ordinary Resolution, appoint another person in place of a Director removed from office under article 103 and without prejudice to the powers of the Directors under article 81 the Company in general meeting by Ordinary Resolution may appoint any person to be a Director either to fill a casual vacancy or as an additional Director, subject to the maximum number of Directors set out in article 81.
- 105. The Directors may appoint a person who is willing to act to be a Director, either to fill a vacancy or as an additional Director, provided that the appointment does not cause the number of Directors to exceed any number fixed by or in accordance with these articles as the maximum number of Directors. A Director so appointed shall hold office only until the next following annual general meeting. If not re-appointed at such annual general meeting, such Director shall vacate office at the conclusion thereof.
- 106. The Directors may appoint any person to fill the following positions:
  - (a) Secretary:

It shall be the duty of the Secretary to make and keep records of the votes, doings and proceedings of all meetings of the members and Board of the Company, and of its committees, and to authenticate records of the Company. The Secretary shall be appointed by the Directors for such term, at such remuneration and upon such conditions as they may think fit; and any Secretary so appointed may be removed by them.

A provision of the Acts or these articles requiring or authorising a thing to be done by or to a Director and the Secretary shall not be satisfied by its being done by or to the same person acting both as Director and as, or in place of, the Secretary.

The Secretary may delegate any of his functions to such one or more persons (including individuals, bodies corporate or firms) as may be nominated by the Secretary from time to time.

(b) Assistant Secretaries:

The Assistant Secretaries shall have such duties as the Secretary shall determine.

In addition to the Board's power to delegate to committees pursuant to article 111, the Board may delegate any of its powers to any individual Director or member of the management of the Company or any of its subsidiaries as it sees fit; any such individual shall, in the exercise of the powers so delegated, conform to any regulations that may be imposed on them by the Board. The Board shall also have the power to appoint and remove officers of the Company including, but not limited to, chief executive officer, president, vice president, treasurer, controller and assistant treasurer.

#### **PROCEEDINGS OF DIRECTORS**

- 107. (a) The Directors may meet together for the dispatch of business, adjourn and otherwise regulate their meetings as they may think fit. The quorum necessary for the transaction of the business of the Directors shall be a majority of the Directors in office at the time when the meeting is convened. Questions arising at any meeting shall be decided by a majority of votes. Each director present and voting shall have one vote.
  - (b) Any Director may participate in a meeting of the Directors by means of telephonic or other such communication whereby all persons participating in the meeting can hear each other speak, and participation in a meeting in this manner shall be deemed to constitute presence in person at such meeting and any director may be situated in any part of the world for any such meeting.
- 108. The Chairman or any four Directors may, and the Secretary on the requisition of the Chairman or any four Directors shall, at any time summon a meeting of the Directors.
- 109. The continuing Directors may act notwithstanding any vacancy in their number but, if and so long as their number is reduced below the number fixed by or pursuant to these articles as the necessary quorum of Directors, the continuing Directors or Director may act for the purpose of increasing the number of Directors to that number or of summoning a general meeting of the Company but for no other purpose.
- 110. The Directors may elect a Chairman of their meetings and determine the period for which he is to hold office. Any Director may be elected no matter by whom he was appointed but if no such Chairman is elected, or if at any meeting the Chairman is not present within five minutes after the time appointed for holding the same, the Directors present may choose one of their number to be Chairman of the meeting.
- 111. The Board may from time to time designate committees of the Board, with such powers and duties as the Board may decide to confer on such committees, and shall, for those committees and any others provided for herein, elect a director or directors to serve as the member or members, designating, if it desires, other directors as alternate members who may replace any absent or disqualified member at any meeting of the committee. Adequate provision shall be made for notice to members of all meetings; a majority of the members shall constitute a quorum unless the committee shall consist of one or two members, in which event one member shall constitute a quorum; and all matters shall be determined by a majority vote of the members present. Action may be taken by any committee without a meeting if all members thereof consent thereto in writing, and the writing or writings are filed with the minutes of the proceedings of such committees.
- 112. A committee may elect a chairman of its meeting. If no such chairman is elected, or if at any meeting the chairman is not present within five minutes after the time appointed for holding the same, the members present may choose one of their number to be chairman of the meeting.
- 113. All acts done by any meeting of the Directors or of a committee of Directors or by any person acting as a Director shall, notwithstanding that it be afterwards discovered that there was some defect in the appointment of any such Director or person acting as aforesaid, or that they or any of them were disqualified, be as valid as if every such person had been duly appointed and was qualified to be a Director.
- 114. Notwithstanding anything in these articles or in the Acts which might be construed as providing to the contrary, notice of every meeting of the Directors shall be given to all Directors either by mail not less than forty-eight (48) hours before the date of the meeting, by telephone, email, or any other electronic means on not less than twenty-four (24) hours' notice, or on such shorter notice as person or persons calling such meeting may deem necessary or appropriate and which is reasonable in the circumstances. Any director may waive any notice required to be given under these articles, and the attendance of a director at a meeting shall be deemed to be a waiver by such Director.
- 115. A resolution or other document in writing (in electronic form or otherwise) signed (whether by electronic signature, advanced electronic signature or otherwise as approved by the Directors) by all the Directors entitled to receive notice of a meeting of Directors or of a committee of Directors shall be as valid as if it had been passed at a meeting of Directors or (as the case may be) a committee of Directors

duly convened and held and may consist of several documents in the like form each signed by one or more Directors, and such resolution or other document or documents when duly signed may be delivered or transmitted (unless the Directors shall otherwise determine either generally or in any specific case) by facsimile transmission, electronic mail or some other similar means of transmitting the contents of documents.

### THE SEAL

- 116. (a) The Directors shall ensure that the Seal (including any official securities seal kept pursuant to the Acts) shall be used only by the authority of the Directors or of a committee authorised by the Directors and that every instrument to which the seal shall be affixed shall be signed by a Director or some other person appointed by the Directors for that purpose.
  - (b) The Company may exercise the powers conferred by the Acts with regard to having an official seal for use abroad and such powers shall be vested in the Directors.

#### DIVIDENDS AND RESERVES

- 117. The Company in general meeting may declare dividends, but no dividends shall exceed the amount recommended by the Directors.
- 118. The Directors may from time to time pay to the members such interim dividends as appear to the Directors to be justified by the profits of the Company.
- 119. No dividend or interim dividend shall be paid otherwise than in accordance with the provisions of Part IV of the 1983 Act.
- 120. The Directors may, before recommending any dividend, set aside out of the profits of the Company such sums as they think proper as a reserve or reserves which shall, at the discretion of the Directors, be applicable for any purpose to which the profits of the Company may be properly applied and pending such application may at the like discretion either be employed in the business of the Company or be invested in such investments as the Directors may lawfully determine. The Directors may also, without placing the same to reserve, carry forward any profits which they may think it prudent not to divide.
- 121. Subject to the rights of persons, if any, entitled to shares with special rights as to dividend, all dividends shall be declared and paid according to the amounts paid or credited as paid on the shares in respect whereof the dividend is paid. All dividends shall be apportioned and paid proportionately to the amounts paid or credited as paid on the shares during any portion or portions of the period in respect of which the dividend is paid; but if any share is issued on terms providing that it shall rank for dividend as from a particular date, such share shall rank for dividend accordingly.
- 122. The Directors may deduct from any dividend payable to any member all sums of money (if any) immediately payable by him to the Company in relation to the shares of the Company.
- 123. Any general meeting declaring a dividend or bonus and any resolution of the Directors declaring an interim dividend may direct payment of such dividend or bonus or interim dividend wholly or partly by the distribution of specific assets and in particular of paid up shares, debentures or debenture stocks of any other company or in any one or more of such ways, and the Directors shall give effect to such resolution, and where any difficulty arises in regard to such distribution, the Directors may settle the same as they think expedient, and in particular may fix the value for distribution of such specific assets or any part thereof and may determine that cash payments shall be made to any members upon the footing of the value so fixed, in order to adjust the rights of all the parties, and may vest any such specific assets in trustees as may seem expedient to the Directors.
- 124. Any dividend or other moneys payable in respect of any share may be paid by cheque or warrant sent by post, at the risk of the person or persons entitled thereto, to the registered address of the Holder or, where there are joint Holders, to the registered address of that one of the joint Holders who is first named on the members Register or to such person and to such address as the Holder or joint Holders may in writing direct. Every such cheque or warrant shall be made payable to the order of the person to whom it is sent and payment of the cheque or warrant shall be a good discharge to the Company. Any

joint Holder or other person jointly entitled to a share as aforesaid may give receipts for any dividend or other moneys payable in respect of the share. Any such dividend or other distribution may also be paid by any other method (including payment in a currency other than US\$, electronic funds transfer, direct debit, bank transfer or by means of a relevant system) which the Directors consider appropriate and any member who elects for such method of payment shall be deemed to have accepted all of the risks inherent therein. The debiting of the Company's account in respect of the relevant amount shall be evidence of good discharge of the Company's obligations in respect of any payment made by any such methods.

- 125. No dividend shall bear interest against the Company.
- 126. If the Directors so resolve, any dividend which has remained unclaimed for twelve years from the date of its declaration shall be forfeited and cease to remain owing by the Company. The payment by the Directors of any unclaimed dividend or other moneys payable in respect of a share into a separate account shall not constitute the Company a trustee in respect thereof.

### ACCOUNTS

- 127. (a) The Directors shall cause to be kept proper books of account, whether in the form of documents, electronic form or otherwise, that:
  - (i) correctly record and explain the transactions of the Company;
  - (ii) will at any time enable the financial position of the Company to be determined with reasonable accuracy;
  - (iii) will enable the Directors to ensure that any balance sheet, profit and loss account or income and expenditure account of the Company complies with the requirements of the Acts; and
  - (iv) will enable the accounts of the Company to be readily and properly audited.

Books of account shall be kept on a continuous and consistent basis and entries therein shall be made in a timely manner and be consistent from year to year. Proper books of account shall not be deemed to be kept if there are not kept such books of account as are necessary to give a true and fair view of the state of the Company's affairs and to explain its transactions.

The Company may send by post, electronic mail or any other means of electronic communication a summary financial statement to its members or persons nominated by any member. The Company may meet, but shall be under no obligation to meet, any request from any of its members to be sent additional copies of its full report and accounts or summary financial statement or other communications with its members.

- (b) The books of account shall be kept at the Office or, subject to the provisions of the Acts, at such other place as the Directors think fit and shall be open at all reasonable times to the inspection of the Directors.
- (c) In accordance with the provisions of the Acts, the Directors shall cause to be prepared and to be laid before the annual general meeting of the Company from time to time such profit and loss accounts, balance sheets, group accounts and reports as are required by the Acts to be prepared and laid before such meeting.
- (d) A copy of every balance sheet (including every document required by law to be annexed thereto) which is to be laid before the annual general meeting of the Company together with a copy of the Directors' report and Auditors' report shall be sent by post, electronic mail or any other means of communication (electronic or otherwise), not less than twenty-one Clear Days before the date of the annual general meeting, to every person entitled under the provisions of the Acts to receive them; provided that in the case of those documents sent by electronic mail or any other means of electronic communication, such documents shall be sent with the consent of the recipient, to the address of the recipient notified to the Company by the recipient for such purposes.

128. The Directors shall determine from time to time whether and to what extent and at what times and places and under what conditions or regulations the accounts and books of the Company or any of them shall be open to the inspection of members, not being Directors, and no member (not being a Director) shall have any right of inspecting any account or book or document of the Company except as conferred by the Acts or authorised by the Directors or by the Company in general meeting. No member shall be entitled to require discovery of or any information respecting any detail of the Company's trading, or any matter which is or may be in the nature of a trade secret, mystery of trade, or secret process which may relate to the conduct of the business of the Company and which in the opinion of the Directors it would be inexpedient in the interests of the members of the Company to communicate to the public.

# **CAPITALISATION OF PROFITS**

- 129. Without prejudice to any powers conferred on the Directors as aforesaid and subject to the Directors' authority to issue and allot shares under articles 8(c) and 8(d), the Directors may resolve to capitalise any part of the amount for the time being standing to the credit of any of the Company's reserve accounts (including any capital redemption reserve fund, share premium account or other reserve account not available for distribution) or to the credit of the profit and loss account which is not available for distribution by applying such sum in paying up in full unissued shares to be allotted as fully paid bonus shares to those members of the Company who would have been entitled to that sum if it were distributable and had been distributed by way of dividend (and in the same proportions). Whenever such a resolution is passed in pursuance of this article, the Directors shall make all appropriations and applications of the amounts resolved to be capitalised thereby and all allotments and issues of fully paid shares or debentures, if any.
- 130. Without prejudice to any powers conferred on the Directors by these articles, and subject to the Directors' authority to issue and allot shares under articles 8(c) and 8(d), the Directors may resolve that any sum for the time being standing to the credit of any of the Company's reserve accounts (including any reserve account available for distribution) or to the credit of the profit and loss account be capitalised and applied on behalf of the members who would have been entitled to receive that sum if it had been distributed by way of dividend (and in the same proportions) either in or towards paying up amounts for the time being unpaid on any shares held by them respectively, or in paying up in full unissued shares or debentures of the Company of a nominal amount equal to the sum capitalised (such shares or debentures to be allotted and distributed and credited as fully paid up to and amongst such Holders in the proportions aforesaid) or partly in one way and partly in another, so, however, that the only purposes for which sums standing to the credit of the capital redemption reserve fund or the share premium account shall be applied shall be those permitted by the Acts.
- 131. The Directors may from time to time at their discretion, subject to the provisions of the Acts and, in particular, to their being duly authorised pursuant to Section 20 of the 1983 Act, to allot the relevant shares, offer to the Holders of Ordinary Shares the right to elect to receive in lieu of any dividend or proposed dividend or part thereof an allotment of additional Ordinary Shares credited as fully paid. In any such case the following provisions shall apply.
  - (i) The basis of allotment shall be determined by the Directors so that, as nearly as may be considered convenient in the Directors' absolute discretion, the value (calculated by reference to the average quotation) of the additional Ordinary Shares (excluding any fractional entitlement) to be allotted in lieu of any amount of dividend shall equal such amount. For such purpose the "average quotation" of an Ordinary Share shall be the average of the five amounts resulting from determining whichever of the following ((A), (B) or (C) specified below) in respect of Ordinary Shares shall be appropriate for each of the first five business days on which Ordinary Shares are quoted "ex" the relevant dividend and as determined from the information published by the New York Stock Exchange reporting the business done on each of these five business days:

- (A) if there shall be more than one dealing reported for the day, the average of the prices at which such dealings took place; or
- (B) if there shall be only one dealing reported for the day, the price at which such dealing took place; or
- (C) if there shall not be any dealing reported for the day, the average of the closing bid and offer prices for the day;

and if there shall be only a bid (but not an offer) or an offer (but not a bid) price reported, or if there shall not be any bid or offer price reported, for any particular day then that day shall not count as one of the said five business days for the purposes of determining the average quotation. If the means of providing the foregoing information as to dealings and prices by reference to which the average quotation is to be determined is altered or is replaced by some other means, then the average quotation shall be determined on the basis of the equivalent information published by the relevant authority in relation to dealings on the New York Stock Exchange or its equivalent.

- (ii) The Directors shall give notice in writing (whether in electronic form or otherwise) to the Holders of Ordinary Shares of the right of election offered to them and shall send with or following such notice forms of election and specify the procedure to be followed and the place at which, and the latest date and time by which, duly completed forms of election must be lodged in order to be effective. The Directors may also issue forms under which Holders may elect in advance to receive new Ordinary Shares instead of dividends in respect of future dividends not yet declared (and, therefore, in respect of which the basis of allotment shall not yet have been determined).
- (iii) The dividend (or that part of the dividend in respect of which a right of election has been offered) shall not be payable on Ordinary Shares in respect of which the right of election as aforesaid has been duly exercised (the "Subject Ordinary Shares") and in lieu thereof additional Ordinary Shares (but not any fraction of a share) shall be allotted to the Holders of the Subject Ordinary Shares on the basis of allotment determined aforesaid and for such purpose the Directors shall capitalise, out of such of the sums standing to the credit of any of the Company's reserves (including any capital redemption reserve fund or share premium account) or to the credit of the profit and loss account as the Directors may determine, a sum equal to the aggregate nominal amount of additional Ordinary Shares to be allotted on such basis and apply the same in paying up in full the appropriate number of unissued Ordinary Shares for allotment and distribution to and amongst the holders of the Subject Ordinary Shares on such basis.
- 132. (a) The additional Ordinary Shares allotted pursuant to articles 129, 130 or 131 shall rank pari passu in all respects with the fully paid Ordinary Shares then in issue save only as regards participation in the relevant dividend or share election in lieu.
  - (b) The Directors may do all acts and things considered necessary or expedient to give effect to any capitalisation pursuant to articles 129, 130 or 131 with full power to the Directors to make such provisions as they think fit where shares would otherwise have been distributable in fractions (including provisions whereby, in whole or in part, fractional entitlements are disregarded and the benefit of fractional entitlements accrues to the Company rather than to the holders concerned). The Directors may authorise any person to enter on behalf of all the Holders interested into an agreement with the Company providing for such capitalisation and matters incidental thereto and any agreement made under such authority shall be effective and binding on all concerned.
  - (c) The Directors may on any occasion determine that rights of election shall not be offered to any Holders of Ordinary Shares who are citizens or residents of any territory where the making or publication of an offer of rights of election or any exercise of rights of election or any purported acceptance of the same would or might be unlawful, and in such event the provisions aforesaid shall be read and construed subject to such determination.

## AUDIT

133. Auditors shall be appointed and their duties regulated in accordance with sections 160 to 163 of the Act or any statutory amendment thereof.

#### NOTICES

- 134. Any notice to be given, served, sent or delivered pursuant to these articles shall be in writing (whether in electronic form or otherwise).
- 135. (a) A notice or document to be given, served, sent or delivered in pursuance of these articles may be given to, served on or delivered to any member by the Company;
  - (i) by handing same to him or his authorised agent;
  - (ii) by leaving the same at his registered address;
  - (iii) by sending the same by the post in a pre-paid cover addressed to him at his registered address; or
  - (iv) by sending, with the consent of the member, the same by means of electronic mail or other means of electronic communication approved by the Directors, with the consent of the member, to the address of the member notified to the Company by the member for such purpose (or if not so notified, then to the address of the member last known to the Company).
  - (b) For the purposes of these articles and the Act, a document shall be deemed to have been sent to a member if a notice is given, served, sent or delivered to the member and the notice specifies the website or hotlink or other electronic link at or through which the member may obtain a copy of the relevant document.
  - (c) Where a notice or document is given, served or delivered pursuant to sub-paragraph (a)(i) or (ii) of this article, the giving, service or delivery thereof shall be deemed to have been effected at the time the same was handed to the member or his authorised agent, or left at his registered address (as the case may be).
  - (d) Where a notice or document is given, served or delivered pursuant to sub-paragraph (a)(iii) of this article, the giving, service or delivery thereof shall be deemed to have been effected at the expiration of twenty-four hours after the cover containing it was posted. In proving service or delivery it shall be sufficient to prove that such cover was properly addressed, stamped and posted.
  - (e) Where a notice or document is given, served or delivered pursuant to sub-paragraph (a)(iv) of this article, the giving, service or delivery thereof shall be deemed to have been effected at the expiration of 48 hours after despatch.
  - (f) Every legal personal representative, committee, receiver, curator bonis or other legal curator, assignee in bankruptcy, examiner or liquidator of a member shall be bound by a notice given as aforesaid if sent to the last registered address of such member, or, in the event of notice given or delivered pursuant to sub-paragraph (a)(iv), if sent to the address notified by the Company by the member for such purpose notwithstanding that the Company may have notice of the death, lunacy, bankruptcy, liquidation or disability of such member.
  - (g) Notwithstanding anything contained in this article the Company shall not be obliged to take account of or make any investigations as to the existence of any suspension or curtailment of postal services within or in relation to all or any part of any jurisdiction or other area other than Ireland.

- (h) Any requirement in these articles for the consent of a member in regard to the receipt by such member of electronic mail or other means of electronic communications approved by the Directors, including the receipt of the Company's audited accounts and the directors' and auditor's reports thereon, shall be deemed to have been satisfied where the Company has written to the member informing him/her of its intention to use electronic communications for such purposes and the member has not, within four weeks of the issue of such notice, served an objection in writing on the Company to such proposal. Where a member has given, or is deemed to have given, his/her consent to the receipt by such member of electronic mail or other means of electronic communications approved by the Directors, he/she may revoke such consent at any time by requesting the Company to communicate with him/her in documented form; provided, however, that such revocation shall not take effect until five days after written notice of the revocation is received by the Company.
- (i) Without prejudice to the provisions of sub-paragraphs (a)(i) and (ii) of this article, if at any time by reason of the suspension or curtailment of postal services in any territory, the Company is unable effectively to convene a general meeting by notices sent through the post, a general meeting may be convened by a public announcement and such notice shall be deemed to have been duly served on all members entitled thereto at noon on the day on which the said public announcement is made. In any such case the Company shall put a full copy of the notice of the general meeting on its website.
- 136. A notice may be given by the Company to the joint Holders of a share by giving the notice to the joint Holder whose name stands first in the Register in respect of the share and notice so given shall be sufficient notice to all the joint Holders.
- 137. (a) Every person who becomes entitled to a share shall before his name is entered in the Register in respect of the share, be bound by any notice in respect of that share which has been duly given to a person from whom he derives his title.
  - (b) A notice may be given by the Company to the persons entitled to a share in consequence of the death or bankruptcy of a member by sending or delivering it, in any manner authorised by these articles for the giving of notice to a member, addressed to them at the address, if any, supplied by them for that purpose. Until such an address has been supplied, a notice may be given in any manner in which it might have been given if the death or bankruptcy had not occurred.
- 138. The signature (whether electronic signature, an advanced electronic signature or otherwise) to any notice to be given by the Company may be written (in electronic form or otherwise) or printed.
- 139. A member present, either in person or by proxy, at any meeting of the Company or the Holders of any class of shares in the Company shall be deemed to have received notice of the meeting and, where requisite, of the purposes for which it was called.

#### WINDING UP

- 140. If the Company shall be wound up and the assets available for distribution among the members as such shall be insufficient to repay the whole of the paid up or credited as paid up share capital, such assets shall be distributed so that, as nearly as may be, the losses shall be borne by the members in proportion to the capital paid up or credited as paid up at the commencement of the winding up on the shares held by them respectively. And if in a winding up the assets available for distribution among the members shall be more than sufficient to repay the whole of the share capital paid up or credited as paid up at the commencement of the winding up the assets in proportion to the capital paid up or credited as paid up at the commencement of the winding up, the excess shall be distributed among the members in proportion to the capital at the commencement of the winding up paid up or credited as paid up on the said shares held by them respectively. Provided that this article shall not affect the rights of the Holders of shares issued upon special terms and conditions.
- 141. (a) In case of a sale by the liquidator under section 260 of the Act, the liquidator may by the contract of sale agree so as to bind all the members for the allotment to the members directly of the proceeds of sale in proportion to their respective interests in the Company and may further by the contract limit a time at the expiration of which obligations or shares not accepted or required to be sold shall be deemed to have been irrevocably refused and be at the disposal of the Company, but so that nothing herein contained shall be taken to diminish, prejudice or affect the rights of dissenting members conferred by the said section.

- (b) The power of sale of the liquidator shall include a power to sell wholly or partially for debentures, debenture stock, or other obligations of another company, either then already constituted or about to be constituted for the purpose of carrying out the sale.
- 142. If the Company is wound up, the liquidator, with the sanction of a Special Resolution and any other sanction required by the Acts, may divide among the members in specie or kind the whole or any part of the assets of the Company (whether they shall consist of property of the same kind or not), and, for such purpose, may value any assets and determine how the division shall be carried out as between the members or different classes of members. The liquidator, with the like sanction, may vest the whole or any part of such assets in trustees upon such trusts for the benefit of the contributories as, with the like sanction, he determines, but so that no member shall be compelled to accept any assets upon which there is a liability.

#### INDEMNITY

- (a) Subject to the provisions of and so far as may be admitted by the Acts, every Director and the Secretary of the Company shall be entitled to be indemnified by the Company against all costs, charges, losses, expenses and liabilities incurred by him in the execution and discharge of his duties or in relation thereto including any liability incurred by him in defending any proceedings, civil or criminal, which relate to anything done or omitted or alleged to have been done or omitted by him as an officer or employee of the Company and in which judgement is given in his favour (or the proceedings are otherwise disposed of without any finding or admission of any material breach of duty on his part) or in which he is acquitted or in connection with any application under any statute for relief from liability in respect of any such act or omission in which relief is granted to him by the Court.
  - (b) The Directors shall have power to purchase and maintain for any Director, the Secretary or other employees of the Company insurance against any such liability as referred to in section 200 of the Act.
  - (c) As far as is permissible under the Acts, the Company shall indemnify any current or former executive officer of the Company (excluding any present or former Directors of the Company or Secretary of the Company), or any person who is serving or has served at the request of the Company as a director or executive officer of another company, joint venture, trust or other enterprise, including any Company subsidiary (each individually, a "Covered Person"), against any expenses, including attorney's fees, judgements, fines, and amounts paid in settlement actually and reasonably incurred by him or her in connection with any threatened, pending, or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, to which he or she was or is threatened to be made a party, or is otherwise involved (a "proceeding"), by reason of the fact that he or she is or was a Covered Person; provided, however, that this provision shall not indemnify any Covered Person against any liability arising out of (a) any fraud or dishonesty in the performance of such Covered Person's duty to the Company, or (b) such Covered Party's conscious, intentional or wilful breach of the obligation to act honestly and in good faith with a view to the best interests of the Company. Notwithstanding the preceding sentence, this section shall not extend to any matter which would render it void pursuant to the Acts or to any person holding the office of auditor in relation to the Company.
  - (d) In the case of any threatened, pending or completed action, suit or proceeding by or in the name of the Company, the Company shall indemnify each Covered Person against expenses, including attorneys' fees, actually and reasonably incurred in connection with the defence or the settlement thereof, except no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable for fraud or dishonesty in the performance of his or her duty to the Company, or for conscious, intentional or wilful breach of his or her obligation to act honestly and in good faith with a view to the best interests of the Company, unless and only to the extent that the High Court of Ireland or the court in which such action or suit was brought shall determine upon application that despite

the adjudication of liability, but in view of all the circumstances of the case, such Covered Person is fairly and reasonably entitled to indemnity for such expenses as the court shall deem proper. Notwithstanding the preceding sentence, this section shall not extend to any matter which would render it void pursuant to the Acts or to any person holding the office of auditor in relation to the Company.

- (e) Any indemnification under this article (unless ordered by a court) shall be made by the Company only as authorised in the specific case upon a determination that indemnification of the Covered Person is proper in the circumstances because such person has met the applicable standard of conduct set forth in this article. Such determination shall be made by any person or persons having the authority to act on the matter on behalf of the Company. To the extent, however, that any Covered Person has been successful on the merits or otherwise in defence of any proceeding, or in defence of any claim, issue or matter therein, such Covered Person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith, without necessity of authorisation in the specific case.
- (f) As far as permissible under the Acts, expenses, including attorneys' fees, incurred in defending any proceeding for which indemnification is permitted pursuant to this article shall be paid by the Company in advance of the final disposition of such proceeding upon receipt by the Board of an undertaking by the particular indemnitee to repay such amount if it shall ultimately be determined that he or she is not entitled to be indemnified by the Company pursuant to these articles.
- (g) It being the policy of the Company that indemnification of the persons specified in this article shall be made to the fullest extent permitted by law, the indemnification provided by this article shall not be deemed exclusive (a) of any other rights to which those seeking indemnification or advancement of expenses may be entitled under these articles, any agreement, any insurance purchased by the Company, vote of members or disinterested directors, or pursuant to the direction (however embodied) of any court of competent jurisdiction, or otherwise, both as to action in his or her official capacity and as to action in another capacity while holding such office, or (b) of the power of the Company to indemnify any person who is or was an employee or agent of the Company or of another company, joint venture, trust or other enterprise which he or she is serving or has served at the request of the Company, to the same extent and in the same situations and subject to the same determinations as are hereinabove set forth. As used in this article, references to the "Company" include all constituent companies in a scheme of arrangement, consolidation or merger in which the Company or a predecessor to the Company by scheme of arrangement, consolidation or merger was involved. The indemnification provided by this article shall continue as to a person who has ceased to be a Covered Person and shall inure to the benefit of their heirs, executors, and administrators.

#### UNTRACED HOLDERS

- 144. (a) The Company shall be entitled to sell at the best price reasonably obtainable any share or stock of a member or any share or stock to which a person is entitled by transmission if and provided that:
  - (i) for a period of twelve years (not less than three dividends having been declared and paid) no cheque or warrant sent by the Company through the post in a prepaid letter addressed to the member or to the person entitled by transmission to the share or stock at his address on the Register or other last known address given by the member or the person entitled by transmission to which cheques and warrants are to be sent has been cashed and no communication has been received by the Company from the member or the person entitled by transmission; and
  - (ii) at the expiration of the said period of twelve years the Company has given notice by advertisement in a leading Dublin newspaper and a newspaper circulating in the area in which the address referred to in paragraph (a) of this article is located of its intention to sell such share or stock; and

- (iii) the Company has not during the further period of three months after the date of the advertisement and prior to the exercise of the power of sale received any communication from the member or person entitled by transmission.
- (b) To give effect to any such sale the Company may appoint any person to execute as transferor an instrument of transfer of such share or stock and such instrument of transfer shall be as effective as if it had been executed by the registered Holder of or person entitled by transmission to such share or stock. The Company shall account to the member or other person entitled to such share or stock for the net proceeds of such sale by carrying all monies in respect thereof to a separate account which shall be a permanent debt of the Company and the Company shall be deemed to be a debtor and not a trustee in respect thereof for such member or other person. Monies carried to such separate account may either be employed in the business of the Company or invested in such investments (other than shares of the Company or its holding company if any) as the Directors may from time to time think fit.
- (c) To the extent necessary in order to comply with any laws or regulations to which the Company is subject in relation to escheatment, abandonment of property or other similar or analogous laws or regulations ("Applicable Escheatment Laws"), the Company may deal with any share of any member and any unclaimed cash payments relating to such share in any manner which it sees fit, including (but not limited to) transferring or selling such share and transferring to third parties any unclaimed cash payments relating to such share.
- (d) The Company may only exercise the powers granted to it in sub-paragraph (a) above in circumstances where it has complied with, or procured compliance with, the required procedures (as set out in the Applicable Escheatment Laws) with respect to attempting to identify and locate the relevant member of the Company.
- (e) Any stock transfer form to be executed by the Company in order to sell or transfer a share pursuant to sub-paragraph (a) may be executed in accordance with Article 16(a).

### **DESTRUCTION OF DOCUMENTS**

- 145. The Company may implement such document destruction policies as it so chooses in relation to any type of documents (whether in paper, electronic or other formats), and in particular (without limitation to the foregoing) may destroy:
  - (a) any dividend mandate or any variation or cancellation thereof or any notification of change of name or address, at any time after the expiry of two years from the date such mandate variation, cancellation or notification was recorded by the Company;
  - (b) any instrument of transfer of shares which has been registered, at any time after the expiry of six years from the date of registration; and
  - (c) any other document on the basis of which any entry in the Register was made, at any time after the expiry of six years from the date an entry in the Register was first made in respect of it,

and it shall be presumed conclusively in favour of the Company that every share certificate (if any) so destroyed was a valid certificate duly and properly sealed and that every instrument of transfer so destroyed was a valid and effective instrument duly and properly registered and that every other document destroyed hereunder was a valid and effective document in accordance with the recorded particulars thereof in the books or records of the Company provided always that:

- (i) the foregoing provisions of this article shall apply only to the destruction of a document in good faith and without express notice to the Company that the preservation of such document was relevant to a claim;
- (ii) nothing contained in this article shall be construed as imposing upon the Company any liability in respect of the destruction of any such document earlier than as aforesaid or in any case where the conditions of proviso (a) above are not fulfilled; and

(iii) references in this article to the destruction of any document include references to its disposal in any manner.

## SALE, LEASE OR EXCHANGE OF ASSETS

- 146. The Directors are hereby expressly authorised to sell, lease or exchange all or substantially all of the Company's property and assets, including the Company's goodwill and its corporate franchises, upon such terms and conditions and for such consideration, which may consist in whole or in part of money or other property, including shares of stock in, and/or other securities of, any other company or companies, as the Directors deem expedient and for the best interests of the Company subject to authorisation by an Ordinary Resolution of members and any additional vote required by article 151. Notwithstanding authorisation or consent to a proposed sale, lease or exchange of the Company's property and assets by the members, the Board may abandon such sale, lease or exchange without further action of the members, subject to the rights, if any, of third parties under any contract relating thereto. Notwithstanding the foregoing, no resolution adopted by the members shall be required for a sale, lease or exchange of property and assets of the Company to a subsidiary. For the purposes of this article 146:
  - (a) the property and assets of the Company include the property and assets of any subsidiary of the Company; and
  - (b) "subsidiary" means any entity wholly owned and controlled, directly or indirectly, by the Company and includes, without limitation, companies, partnerships, limited partnerships, limited liability partnerships, limited liability companies, and/or statutory trusts.

### SHAREHOLDER RIGHTS PLAN

- 147. Subject to applicable law, the Directors are hereby expressly authorised to adopt any shareholder rights plan (a "**Rights Plan**"), upon such terms and conditions as the Directors deem expedient and in the best interests of the Company, including, without limitation, where the Directors are of the opinion that a Rights Plan could grant them additional time to gather relevant information or pursue strategies in response to or anticipation of, or could prevent, a potential change of control of the Company or accumulation of shares in the Company or interests therein.
- 148. The Directors may exercise any power of the Company to grant rights (including approving the execution of any documents relating to the grant of such rights) to subscribe for ordinary shares or preferred shares in the share capital of the Company ("**Rights**") in accordance with the terms of a Rights Plan.
- 149. For the purposes of effecting an exchange of Rights for ordinary shares or preferred shares in the share capital of the Company (an "Exchange"), the Directors may:
  - (a) resolve to capitalise an amount standing to the credit of the reserves of the Company (including, but not limited to, the share premium account, capital redemption reserve and profit and loss account), whether or not available for distribution, being an amount equal to the nominal value of the ordinary shares or preferred shares which are to be exchanged for the Rights; and
  - (b) apply that sum in paying up in full ordinary shares or preferred shares and allot such shares, credited as fully paid, to those holders of Rights who are entitled to them under an Exchange effected pursuant to the terms of a Rights Plan.
- 150. The common law duties of the Directors to the Company are hereby deemed amended and modified such that the adoption of a Rights Plan and any actions taken thereunder by the Directors (if so approved by the Directors) shall be deemed to constitute an action in the best interests of the Company in all circumstances, and any such action shall be deemed to be immediately confirmed, approved and ratified.

#### **BUSINESS COMBINATION**

- 151. (a) Notwithstanding anything to the contrary contained in these articles, the Company shall not engage in any business combination with any Interested Member for a period of three years following the time that such member became an Interested Member, unless:
  - (i) prior to such time the Directors approved either the business combination or the transaction which resulted in the member becoming an Interested Member;
  - (ii) upon consummation of the transaction which resulted in the member becoming an Interested Member, the Interested Member owned at least 85% of the voting shares of the Company outstanding at the time the transaction commenced, excluding for purposes of determining the voting shares outstanding (but not the outstanding voting shares owned by the Interested Member) those shares owned (A) by persons who are directors and also officers and (B) employee shares plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
  - (iii) at or subsequent to such time the business combination is approved by the Directors and authorised by way of Special Resolution without the Interested Member.
  - (b) The Directors shall have the power and duty to determine, on the basis of information known to them after reasonable inquiry, all facts necessary to determine compliance with this article, including, without limitation, (i) whether a Person is an Interested Member, (ii) the number of shares or other securities beneficially owned by any Person, (iii) whether a Person is an Affiliate or Associate of another, and (iv) the fair market value of the Company's securities or securities of any subsidiary of the Company, and the good faith determination of the Directors on such matters shall be conclusive and binding for all the purposes of this article.
  - (c) As used in this article only, the term:
    - (i) "Affiliate" means a person that directly, or indirectly through one or more intermediaries, controls or is controlled by, or is under common control with, another person.
    - (ii) "Associate", when used to indicate a relationship with any person, means: (A) any company, partnership, unincorporated association or other entity of which such person is a director, officer or partner or is, directly or indirectly, the owner of 20% or more of any class of voting shares; (B) any trust or other estate in which such person has at least a 20% beneficial interest or as to which such person serves as trustee or in a similar fiduciary capacity; and (C) any relative or spouse of such person, or any relative of such spouse, who has the same residence as such person.
    - (iii) "Business combination", when used in reference to any company and any Interested Member of such company, means:
      - (A) any scheme of arrangement, merger or consolidation of the Company or any direct or indirect majority-owned subsidiary of the Company with (1) the Interested Member, or (2) any other company, partnership, unincorporated association or other entity if the scheme of arrangement, merger or consolidation is caused by the Interested Member;
      - (B) any sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), except proportionately as a member of such company, to or with the Interested Member, whether as part of a dissolution or otherwise, of assets of the Company or of any direct or indirect majority-owned subsidiary of the Company which assets have an aggregate market value equal to 10% or more of either the aggregate market value of all the assets of the Company determined on a consolidated basis or the aggregate market value of all the outstanding shares of the Company;
      - (C) any transaction which results in the issuance or transfer by the Company or by any direct or indirect majority-owned subsidiary of the Company of any

shares of the Company or of such subsidiary to the Interested Member, except: (1) pursuant to the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into shares of such company or any such subsidiary which securities were outstanding prior to the time that the Interested Member became such; (2) pursuant to a dividend or distribution paid or made, or the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into shares of such company or any such subsidiary which security is distributed, pro rata to all holders of a class or series of shares of such company to purchase shares made on the same terms to all holders of said shares; or (4) any issuance or transfer of shares by the Company; provided however, that in no case under items (3) and (4) of this subparagraph shall there be an increase in the Interested Member's proportionate share of the shares of any class or series of the Company;

- (D) any transaction involving the Company or any direct or indirect majority-owned subsidiary of the Company which has the effect, directly or indirectly, of increasing the proportionate share of the shares of any class or series, or securities convertible into the shares of any class or series, of the Company or of any such subsidiary which is owned by the Interested Member, except as a result of immaterial changes due to fractional share adjustments or as a result of any purchase or redemption of any shares of shares not caused, directly or indirectly, by the Interested Member; or
- (E) any receipt by the Interested Member of the benefit, directly or indirectly (except proportionately as a member of such company), of any loans, advances, guarantees, pledges or other financial benefits (other than those expressly permitted in subparagraphs (A)-(D) of this paragraph) provided by or through the Company or any direct or indirect majority-owned subsidiary.
- (iv) "Control", including the terms "controlling", "controlled by" and "under common control with", means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person, whether through the ownership of voting shares, by contract or otherwise. A person who is the owner of 20% or more of the outstanding voting shares of any company, partnership, unincorporated association or other entity shall be presumed to have control of such entity, in the absence of proof by a preponderance of the evidence to the contrary. Notwithstanding the foregoing, a presumption of control shall not apply where such person holds voting shares, in good faith and not for the purpose of circumventing this article, as an agent, bank, broker, nominee, custodian or trustee for one or more owners who do not individually or as a group have control of such entity.
- (v) "Interested Member" means any Person, including its Affiliates and Associates (other than the Company and any direct or indirect majority-owned subsidiary of the Company), that is, or was at any time within the three-year period immediately prior to the date in question, the Owner of 15% or more of the outstanding voting shares of the Company; provided, however, that the term "Interested Member" shall not include any person whose ownership of shares in excess of the 15% limitation set forth herein is the result of action taken solely by the Company; provided that such person shall be an Interested Member if thereafter such person acquires additional voting shares of the Company, except as a result of further corporate action not caused, directly or indirectly, by such person. For the purpose of determining whether a person is an Interested Member, the voting shares of the Company deemed to be outstanding shall include shares deemed to be owned by the person through application of (viii) of this subsection but shall not include any other unissued shares of such company which may be issuable pursuant to any agreement, arrangement or understanding, or upon exercise of conversion rights, warrants or options, or otherwise.

- (vi) "Person" means any individual, company, partnership, unincorporated association or other entity.
- (vii) "Shares" means, with respect to any company, capital shares and, with respect to any other entity, any equity interest.
- (viii) "Voting shares" means, with respect to any company, shares of any class or series entitled to vote generally in the election of directors and, with respect to any entity that is not a company, any equity interest entitled to vote generally in the election of the governing body of such entity. Every reference to a percentage of voting shares shall refer to such percentage of the votes of such voting shares.
- (ix) "Owner", including the terms "own" and "owned", when used with respect to any Shares, means a person that individually or with or through any of its Affiliates or Associates:
  - (A) beneficially owns such Shares, directly or indirectly; or
  - (B) has (1) the right to acquire such Shares (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, warrants or options, or otherwise; provided, however, that a person shall not be deemed the Owner of Shares tendered pursuant to a tender or exchange offer made by such person or any of such person's affiliates or associates until such tendered Shares are accepted for purchase or exchange; or (2) the right to vote such shares pursuant to any agreement, arrangement or understanding; provided, however, that a person shall not be deemed the Owner of any Shares because of such person's right to vote such Shares if the agreement, arrangement or understanding to vote such shares arises solely from a revocable proxy or consent given in response to a proxy or consent solicitation made to 10 or more persons; or
  - (C) has any agreement, arrangement or understanding for the purpose of acquiring, holding, voting (except voting pursuant to a revocable proxy or consent as described in item (2) of subparagraph (B) of this paragraph), or disposing of such Shares with any other person that beneficially owns, or whose Affiliates or Associates beneficially own, directly or indirectly, such Shares.

Names, addresses and descriptions of subscribers

J. MCGOWAN-SMYTH For and on behalf of Fand Limited Arthur Cox Building Earlsfort Terrace Dublin 2 J. MCGOWAN-SMYTH For and on behalf of

DIJR Nominees Limited Arthur Cox Building Earlsfort Terrace Dublin 2

J. MCGOWAN-SMYTH For and on behalf of AC Administration Services Limited Arthur Cox Building Earlsfort Terrace Dublin 2

J. MCGOWAN-SMYTH For and on behalf of Arthur Cox Nominees Limited Arthur Cox Building Earlsfort Terrace Dublin 2

J. MCGOWAN-SMYTH For and on behalf of Arthur Cox Registrars Limited Arthur Cox Building Earlsfort Terrace Dublin 2

J. MCGOWAN-SMYTH For and on behalf of Arthur Cox Trust Services Limited Arthur Cox Building Earlsfort Terrace Dublin 2

J. MCGOWAN-SMYTH For and on behalf of Arthur Cox Trustees Limited Arthur Cox Building Earlsfort Terrace Dublin 2 Solicitor

Dated 21 December 2012

Witness to the above signatures:			
Name:	MAIREAD FOLEY		
Address:	ARTHUR COX BUILDING		
	EARLSFORT TERRACE		
	DUBLIN 2		
Occupation:	COMPANY SECRETARY		

Companies Acts 1963 to 2012

# A PUBLIC COMPANY LIMITED BY SHARES

# MEMORANDUM AND ARTICLES OF ASSOCIATION

OF

# MALLINCKRODT PUBLIC LIMITED COMPANY

Arthur Cox Arthur Cox Building Earlsfort Terrace Dublin TRANSITION SERVICES AGREEMENT BY AND BETWEEN COVIDIEN PLC AND MALLINCKRODT PLC DATED AS OF [--], 2013

## TABLE OF CONTENTS

### ARTICLE I DEFINITIONS

#### ARTICLE II

# SERVICES, DURATION AND SERVICES MANAGERS

Section 2.01.	Services	3
Section 2.02.	Duration of Services	3
Section 2.03.	Additional Unspecified Services	3
Section 2.04.	New Services	5
Section 2.05.	Services Not Included	5
Section 2.06.	Transition Services Managers	5
Section 2.07.	Personnel	6
	ARTICLE III	
	COVIDIEN MATERIALS	
Section 3.01.	Corporate Policies	7
Section 3.02.	Limitation on Rights and Obligations with Respect to the Covidien Materials	7
	ARTICLE IV	
	ADDITIONAL ARRANGEMENTS	
Section 4.01.	Software and Software Licenses	8
Section 4.02.	Covidien Computer-Based and Other Resources	9
Section 4.03.	Access to Facilities	10
Section 4.04.	Cooperation	10
Section 4.05.	Data Protection	10
	ARTICLE V	
	COSTS AND DISBURSEMENTS	
Section 5.01.	Costs and Disbursements	10
Section 5.02.	Tax Matters	12
Section 5.03.	No Right to Set-Off	13
	ARTICLE VI	
	STANDARD FOR SERVICE	
Section 6.01.	Standard for Service	13

- Section 6.01.Standard for ServiceSection 6.02.Disclaimer of Warranties
- Section 6.03. Compliance with Laws and Regulations

Page

13

## ARTICLE VII LIMITED LIABILITY AND INDEMNIFICATION

	LIMITED LIABILITY AND INDEMNIFICATION	
Section 7.01.	Consequential and Other Damages	14
Section 7.02.	Limitation of Liability	14
Section 7.03.	Obligation To Reperform; Liabilities	14
Section 7.04.	Release and Recipient Indemnity	15
Section 7.05.	Provider Indemnity	15
Section 7.06.	Indemnification Procedures	15
Section 7.07.	Liability for Payment Obligations	15
Section 7.08.	Exclusion of Other Remedies	15
Section 7.09.	Confirmation	15
	ARTICLE VIII	
	TERM AND TERMINATION	
Section 8.01.	Term and Termination	16
Section 8.02.	Effect of Termination	17
Section 8.03.	Force Majeure	17
	ARTICLE IX	
	GENERAL PROVISIONS	
Section 9.01.	No Agency	18
Section 9.02.	Subcontractors	18
Section 9.02. Section 9.03.	Treatment of Confidential Information	18
Section 9.04.	Further Assurances	19
Section 9.05.	Dispute Resolution	19
Section 9.06.	Notices	19
Section 9.07.	Severability	20
Section 9.08.	Entire Agreement	21
Section 9.09.	No Third-Party Beneficiaries	21
Section 9.10.	Governing Law	21
Section 9.11.	Amendment	21
Section 9.12.	Rules of Construction	21
Section 9.13.	Counterparts	22
Section 9.14.	Assignability	22
Section 9.15.	Public Announcements	23
Section 9.16.	Non-Recourse	23

SCHEDULE A Covidien Services SCHEDULE B Mallinckrodt Services EXHIBIT I Services Managers

ii

A-1 B-1

I-1

#### TRANSITION SERVICES AGREEMENT

This TRANSITION SERVICES AGREEMENT, dated as of [--], 2013 (this "<u>Agreement</u>"), is by and between Covidien plc, an Irish public limited company ("<u>Covidien</u>"), and Mallinckrodt plc, an Irish public limited company ("<u>Mallinckrodt</u>"). Unless otherwise defined in this Agreement, all capitalized terms used in this Agreement shall have the meaning set forth in the Separation and Distribution Agreement, dated as of the date hereof, by and between Covidien and Mallinckrodt (as amended, modified or supplemented from time to time in accordance with its terms, the "<u>Separation Agreement</u>").

#### RECITALS

WHEREAS, the board of directors of Covidien has determined that it is in the best interests of Covidien and its shareholders that the Mallinckrodt Business be operated by a newly incorporated publicly traded company;

WHEREAS, Covidien and Mallinckrodt have entered into the Separation Agreement;

WHEREAS, in order to facilitate and provide for an orderly transition under the Separation Agreement, the Parties (as defined herein) desire to enter into this Agreement to set forth the terms and conditions pursuant to which each of the Parties shall provide to the other the Services (as defined herein) for a transitional period; and

WHEREAS, the Separation Agreement requires execution and delivery of this Agreement by Covidien and Mallinckrodt on or prior to the Distribution Date.

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained in this Agreement, the Parties, intending to be legally bound, hereby agree as follows:

## ARTICLE I DEFINITIONS

The following capitalized terms used in this Agreement shall have the meanings set forth below:

"Additional Services" shall have the meaning set forth in Section 2.03(a).

"Agreement" shall have the meaning set forth in the Preamble.

"Confidential Information" shall have the meaning set forth in Section 9.03(a).

"Covidien" shall have the meaning set forth in the Preamble.

"Covidien Business" shall mean the businesses and operations of the Covidien Group other than the Mallinckrodt Business.

"Covidien Local Service Manager" shall have the meaning set forth in Section 2.06(a).

"Covidien Materials" shall have the meaning set forth in Section 3.01(a).

"Covidien Services" shall have the meaning set forth in Section 2.01.

"Covidien Services Manager" shall have the meaning set forth in Section 2.06(a).

- "Governmental Requirements" shall have the meaning set forth in the Tax Matters Agreement.
- "Interest Payment" shall have the meaning set forth in Section 5.01(d).
- "Mallinckrodt" shall have the meaning set forth in the Preamble.
- "Mallinckrodt Local Service Manager" shall have the meaning set forth in Section 2.06(b).
- "Mallinckrodt Services" shall have the meaning set forth in Section 2.01.
- "Mallinckrodt Services Manager" shall have the meaning set forth in Section 2.06(b)
- "New Services" shall have the meaning set forth in Section 2.04(a).

"Party" shall mean Covidien and Mallinckrodt individually, and "Parties" means Covidien and Mallinckrodt collectively, and, in each case, their permitted successors and assigns.

- "Provider" shall mean the Party or its Subsidiary or Affiliate providing a Service under this Agreement.
- "Provider Indemnified Party" shall have the meaning set forth in Section 7.04.
- "Recipient" shall mean the Party or its Subsidiary or Affiliate to whom a Service under this Agreement is being provided.
- "Recipient Indemnified Party" shall have the meaning set forth in Section 7.05.
- "Reimbursement Charges" shall have the meaning set forth in Section 5.01(c).
- "Schedule(s)" shall have the meaning set forth in Section 2.02.
- "Separation Agreement" shall have the meaning set forth in the Preamble.
- "Service Baseline Period" shall have the meaning set forth in Section 2.03(c).
- "Service Charges" shall have the meaning set forth in Section 5.01(a).
- "Service Extension" shall have the meaning set forth in Section 8.01(c).
- "Service Increases" shall have the meaning set forth in Section 2.03(b).
- "Services" shall have the meaning set forth in Section 2.01.

"Taxes" shall have the meaning set forth in the Tax Matters Agreement.

"Transfer Taxes" shall have the meaning set forth in Section 5.02(a).

"<u>VAT</u>" shall have the meaning set forth in <u>Section 5.02(a)</u>.

### ARTICLE II SERVICES, DURATION AND SERVICES MANAGERS

Section 2.01. <u>Services</u>. Subject to the terms and conditions of this Agreement, (a) Covidien shall provide or cause to be provided to the Mallinckrodt Group the services listed on <u>Schedule A</u> to this Agreement (the "<u>Covidien Services</u>") and (b) Mallinckrodt shall provide or cause to be provided to the Covidien Group the services listed on <u>Schedule B</u> to this Agreement (the "<u>Mallinckrodt Services</u>," and, collectively with the Covidien Services, any Additional Services, any Service Increases and any New Services, the "<u>Services</u>"). For the avoidance of doubt, Services provided in different regions or countries (as indicated by such Services being listed on different subparts of the Schedules hereto) shall be considered separate Services hereunder, notwithstanding that such Services may be similar in nature. All of the Services shall be for the sole use and benefit of the respective Recipient and its respective Party.

Section 2.02. <u>Duration of Services</u>. Subject to the terms of this Agreement, each of Covidien and Mallinckrodt shall provide or cause to be provided to the respective Recipients each Service until the earlier to occur of, with respect to each such Service, (i) the expiration of the term for such Service (or, subject to the terms of <u>Section 8.01(c)</u>, the expiration of any Service Extension) as set forth on <u>Schedule A</u> or <u>Schedule B</u> (each a "<u>Schedule</u>", and collectively, the "<u>Schedules</u>") or (ii) the date on which such Service is terminated under <u>Section 8.01(b)</u>.

Section 2.03. Additional Unspecified Services. (a) After the date of this Agreement, if Mallinckrodt or Covidien (i) identifies a service that (x) the Covidien Group provided to the Mallinckrodt Group prior to the Distribution Date that Mallinckrodt reasonably needs in order for the Mallinckrodt Business to continue to operate in substantially the same manner in which the Mallinckrodt Business operated prior to the Distribution Date, and such service was not included on Schedule A (other than because the Parties agreed such service shall not be provided), or (y) the Mallinckrodt Group provided to the Covidien Group prior to the Distribution Date that Covidien reasonably needs in order for the Covidien Business to continue to operate in substantially the same manner in which the Covidien Business to continue to operate in substantially the same manner in which the Covidien Business operated prior to the Distribution Date, and such service was not included on Schedule B (other than because the Parties agreed such service shall not be provided), and (ii) provides written notice to the other Party within ten (10) days following the date of the filing by Mallinckrodt of its first Annual Report on Form 10-K with the U.S. Securities and Exchange Commission requesting such additional services, then such other Party shall use its commercially reasonable efforts to provide any Additional Service if it does not, in its reasonable judgment, have adequate resources to provide such Additional Service or if the provision of such Additional Services if the Parties are unable to reach agreement on the terms thereof (including with respect to Service Charges

therefor). In connection with any request for Additional Services in accordance with this <u>Section 2.03(a)</u>, the Covidien Services Manager and the Mallinckrodt Services Manager shall in good faith negotiate the terms of a supplement to the applicable Schedule, which terms shall be consistent with the terms of, and the pricing methodology used for, similar Services provided under this Agreement. Upon the mutual written agreement of the Parties, the supplement to the applicable Schedule shall describe in reasonable detail the nature, scope, service period(s), termination provisions and other terms applicable to such Additional Services in a manner similar to that in which the Services are described in the existing Schedules. Each supplement to the applicable Schedule, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such agreement and the Additional Services set forth therein shall be deemed "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement.

(b) After the date of this Agreement, if (i) a Recipient requests to increase, relative to historical levels prior to the Distribution Date, the volume, amount, level or frequency, as applicable, of any Service provided by such Provider and (ii) such increase is reasonably determined by the Recipient as necessary for the Recipient to operate its businesses (such increases, the "<u>Service Increases</u>"), then such Provider shall consider such request in good faith; <u>provided</u>, <u>however</u>, that no Party shall be obligated to provide any Service Increase, including because, after good-faith negotiations between the Parties, the Parties fail to reach an agreement with respect to the terms thereof (including with respect to Service Charges therefor). In connection with any request for Service Increases in accordance with this <u>Section 2.03(b)</u>, the Covidien Services Manager and the Mallinckrodt Services Manager shall in good faith negotiate the terms of an amendment to the applicable Schedule, which amendment shall be consistent with the terms of, and the pricing methodology used for, the applicable Service. Each amended Schedule, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such agreement and the Service Increases set forth therein shall be deemed a part of the "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement.

(c) Notwithstanding the foregoing clauses (a) and (b), and without limiting the remainder of this clause (c), the Provider shall not be obligated to perform or to cause to be performed any Service in a volume or quantity in any fiscal year that exceeds the highest volumes or quantities of analogous services provided to Covidien's applicable functional group or Subsidiary during fiscal year 2012 (without reference to the transactions contemplated by the Separation Agreement) (the "<u>Service Baseline Period</u>"). If the Recipient requests that the Provider perform or cause to be performed any Service in a volume or quantity that exceeds the highest volumes or quantities of analogous services that were provided to Covidien or its applicable functional group or Subsidiary during the Service Baseline Period, then: (i) if such higher volume or quantity results from fluctuations occurring in the ordinary course of business of the Recipient, the Provider shall use commercially reasonable efforts to provide such requested higher volume or quantity; and (ii) if such higher volume or quantity results from any other source, including an acquisition, merger, purchase or other business combination by the Recipient, the Parties shall cooperate and act in good faith to determine whether the Provider shall provide such requested higher volume or quantity. If the Parties agree that the Provider shall provide the requested higher volume or quantity, then Covidien and Mallinckrodt shall document such terms in an amendment to the applicable Schedule, which amendment shall be consistent with the terms of,

and the pricing methodology used for, the applicable Service. Each amended subsection of Schedule A hereto, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such agreement and the volume or quantity increases set forth therein shall be deemed a part of the "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement.

Section 2.04. <u>New Services</u>. (a) From time to time during the term of this Agreement, either Party may request the other Party to provide additional or different services which such other Party is not expressly obligated to provide under this Agreement (excluding, for the avoidance of doubt, any Additional Services or Service Increases, the "<u>New Services</u>"). The Party receiving such request shall consider such request in good faith; <u>provided</u>, <u>however</u>, that no Party shall be obligated to provide any New Services, including because, after negotiations between the Parties pursuant to <u>Section 2.04(b)</u>, the Parties fail to reach an agreement with respect to the terms (including the Service Charges) applicable to the provision of such New Services.

(b) In connection with any request for New Services in accordance with <u>Section 2.04(a)</u>, the Covidien Services Manager and the Mallinckrodt Services Manager shall in good faith (i) negotiate the applicable Service Charge and the terms of a supplement to the applicable Schedule, which supplement shall describe in reasonable detail the nature, scope, service period(s), termination provisions and other terms applicable to such New Services, and (ii) determine any costs and expenses, including any start-up costs and expenses, that would be incurred by the Provider in connection with the provision of such New Services, which costs and expenses shall be borne solely by the Recipient. Each supplement to the applicable Schedule, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such agreement and the New Services set forth therein shall be deemed "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement.

Section 2.05. <u>Services Not Included</u>. It is not the intent of the Provider to render, nor of the Recipient to receive from the Provider, professional advice or opinions, whether with regard to Tax, legal, treasury, finance, employment or other business and financial matters, technical advice, whether with regard to information technology or other matters, or the handling of or addressing environmental matters; the Recipient shall not rely on, or construe, any Service rendered by or on behalf of the Provider as such professional advice or opinions or technical advice; and the Recipient shall seek all third-party professional advice and opinions or technical advice as it may desire or need.

Section 2.06. <u>Transition Services Managers</u>. (a) Covidien hereby appoints and designates the individual holding the Covidien position set forth on <u>Exhibit I</u> to act as its initial services manager (the "<u>Covidien Services Manager</u>"), who will be directly responsible for coordinating and managing the delivery of the Covidien Services and have authority to act on Covidien's behalf with respect to matters relating to the provision of Services under this Agreement. The Covidien Services Manager will work with the personnel of the Covidien Group to periodically address issues and matters raised by Mallinckrodt relating to the provision of Services under this Agreement. Notwithstanding the requirements of <u>Section 9.06</u>, all communications from Mallinckrodt to Covidien pursuant to this Agreement regarding routine matters involving a Service shall be made through the individual specified as the local service manager (the "<u>Covidien Local</u> <u>Service Manager</u>") with respect to such Service on the applicable Schedule or such other individual

as may be specified by the Covidien Services Manager in writing and delivered to Mallinckrodt by email or facsimile transmission with receipt confirmed. Covidien shall notify Mallinckrodt of the appointment of a different Covidien Services Manager or Covidien Local Service Manager(s), if necessary, in accordance with <u>Section 9.06</u>.

(b) Mallinckrodt hereby appoints and designates the individual holding the Mallinckrodt position set forth on Exhibit I to act as its initial services manager (the "<u>Mallinckrodt Services Manager</u>"), who will be directly responsible for coordinating and managing the delivery of Mallinckrodt Services and have authority to act on Mallinckrodt's behalf with respect to matters relating to this Agreement. The Mallinckrodt Services Manager will work with the personnel of the Mallinckrodt Group to periodically address issues and matters raised by Covidien relating to this Agreement. Notwithstanding the requirements of <u>Section 9.06</u>, all communications from Covidien to Mallinckrodt Local Service Manager") with respect to such Service on the applicable Schedule or as specified by the Mallinckrodt Services Manager in writing and delivered to Covidien by email or facsimile transmission with receipt confirmed. Mallinckrodt shall notify Covidien of the appointment of a different Mallinckrodt Services Manager or Mallinckrodt Local Service Manager(s), if necessary, in accordance with <u>Section 9.06</u>.

Section 2.07. <u>Personnel</u>. (a) The Provider of any Service will make available to the Recipient of such Service such personnel as may be necessary to provide such Service on the understanding that such personnel shall remain employed and/or engaged by the Provider. The Provider will have the right, in its reasonable discretion, to (i) designate which personnel it will assign to perform such Service, and (ii) remove and replace such personnel at any time; <u>provided</u>, <u>however</u>, that any such removal or replacement shall not be the basis for any increase in any Service Charge or Reimbursement Charge payable hereunder or relieve the Provider of its obligation to provide any Service hereunder; and <u>provided</u>, <u>further</u>, that the Provider will use its commercially reasonable efforts to limit the disruption to the Recipient in the transition of the Services to different personnel.

(b) In the event that the provision of any Service by the Provider requires the cooperation and services of the personnel of the Recipient, the Recipient will make available to the Provider such personnel (who shall be appropriately qualified for purposes of so supporting the provision of such Service by the Provider) as may be necessary for the Provider to provide such Service on the understanding that such personnel shall remain employed and/or engaged by the Recipient. The Recipient will have the right, in its reasonable discretion, to (i) designate which personnel it will make available to the Provider in connection with the provision of such Service, and (ii) remove and replace such personnel at any time; <u>provided</u>, <u>however</u>, that any resulting increase in costs to the Provider shall be borne by the Recipient and any adverse effect to the provision of such Service by the Provider shall not be deemed a breach of this Agreement; and <u>provided</u>, <u>further</u>, that the Recipient will use its commercially reasonable efforts to limit the disruption to the Provider in the transition of such personnel. If the Provider, in its reasonable discretion and following discussions with the Recipient, requests the Recipient to remove and/or replace any such personnel from their roles in respect of the Services being provided by the Provider, the Recipient shall comply with such request.

(c) No Provider shall be liable under this Agreement for any Liabilities incurred by the Recipient Indemnified Parties that are primarily attributable to, or that are a consequence of, any actions or inactions of the personnel of the Recipient, except for any such actions or inactions undertaken pursuant to the direction of the Provider.

(d) Nothing in this Agreement shall grant the Provider, or its employees, agents and third-party providers that are performing the Services, the right directly or indirectly to control or direct the operations of the Recipient or any member of its Group. Such employees, agents and third-party providers shall not be required to report to the management of the Recipient nor be deemed to be under the management or direction of the Recipient. The Recipient acknowledges and agrees that, except as may be expressly set forth herein as a Service (including any Additional Services, Service Increases or New Services) or otherwise expressly set forth in the Separation Agreement, another Ancillary Agreement or any other applicable agreement, no Provider or any member of its Group shall be obligated to provide, or cause to be provided, any service or goods to any Recipient or any member of its Group.

## ARTICLE III COVIDIEN MATERIALS

Section 3.01. <u>Corporate Policies</u>. (a) Subject to the terms and conditions of this Agreement, Covidien grants to Mallinckrodt a non-exclusive, royalty-free, fully paid-up, worldwide license to create or have created materials based on Covidien's corporate policies and manuals (the "<u>Covidien Materials</u>") for distribution to employees of Mallinckrodt and use such materials in the operation of the Mallinckrodt Business in substantially the same manner as the Covidien Materials were used by Covidien prior to the Distribution. It is understood and agreed that, to the maximum extent permitted by applicable Law, Covidien makes no representation or warranty, express or implied, as to the accuracy or completeness of any of the Covidien Materials, as to whether the Covidien Materials comply with Law, as to the non-infringement of any of the Covidien Materials or as to the suitability of any of the Covidien Materials for use by Mallinckrodt in respect of its business, or otherwise.

(b) Notwithstanding the foregoing, the text of any materials created by or for Mallinckrodt, and related to, or based upon, any of the Covidien Materials, may not contain any references to Covidien (or any of Covidien's marks, names, trade dress, logos or other source or business identifiers, including the Covidien Name and Covidien Marks), Covidien's publications, Covidien's personnel (including senior management), Covidien's management structures or any other indication (other than the verbatim or paraphrased reproduction of the content) that such materials are based upon any of the Covidien Materials.

Section 3.02. Limitation on Rights and Obligations with Respect to the Covidien Materials. (a) Covidien shall have no obligation to (i) notify Mallinckrodt of any changes or proposed changes to any of the Covidien Materials, (ii) include Mallinckrodt in any consideration of proposed changes to any of the Covidien Materials, (iii) provide draft changes of any of the Covidien Materials to Mallinckrodt for review and/or comment or (iv) provide Mallinckrodt with any updated materials relating to any of the Covidien Materials. Mallinckrodt acknowledges and agrees that, except as expressly set forth above, Covidien reserves all rights (including all Intellectual Property rights) in, to and under the Covidien Materials and no rights with respect to

ownership or use, except as otherwise expressly provided in this Agreement, shall vest in Mallinckrodt. The Parties acknowledge and agree that, subject to the exceptions specified in <u>Section 9.03</u>, the Covidien Materials are the Confidential Information of Covidien. Mallinckrodt shall use at least the same degree of care to prevent and restrain the unauthorized use or disclosure of any confidential materials created by or for Mallinckrodt that are based upon any of the Covidien Materials as it uses for its other confidential information of a like nature, but in no event less than a reasonable degree of care. Mallinckrodt will allow Covidien reasonable access to personnel and information as reasonably necessary to determine Mallinckrodt's compliance with the provisions set forth above; <u>provided</u>, <u>however</u>, such access shall not unreasonably interfere with any of the business or operations of Mallinckrodt. Subject to <u>Section 9.05</u>, in the event that Covidien may terminate Mallinckrodt has not materially complied with some or all of its obligations with respect to any or all of the Covidien Materials, Covidien may terminate Mallinckrodt's rights with respect to such Covidien Materials upon written notice to Mallinckrodt and, in such case, Covidien shall be entitled to require such Covidien Materials to be returned to Covidien or destroyed and any materials created by or for Mallinckrodt that are based upon such Covidien Materials to be destroyed (with such destruction certified by Mallinckrodt in writing to Covidien promptly after such termination).

(b) If Mallinckrodt determines to cease to avail itself of any of the Covidien Materials, Covidien and Mallinckrodt shall cooperate in good faith to take reasonable and appropriate actions to arrange for the return to Covidien or destruction of such Covidien Materials and to protect Covidien's rights and interests in such Covidien Materials.

#### ARTICLE IV ADDITIONAL ARRANGEMENTS

Section 4.01. <u>Software and Software Licenses</u>. (a) If and to the extent requested by Mallinckrodt, Covidien shall use commercially reasonable efforts to assist Mallinckrodt in its efforts to obtain licenses (or other appropriate rights) to use, duplicate and distribute, as necessary and applicable, certain computer software necessary for Covidien to provide, and Mallinckrodt to receive, Covidien Services; provided, however, that Covidien shall not be required to pay any fees or other payments or incur any obligations or liabilities to enable Mallinckrodt to obtain any such license or rights (except and to the extent that Mallinckrodt advances such fees or payments to Covidien); provided, further, that Covidien shall not be required to seek broader rights or more favorable terms for Mallinckrodt than those applicable to Covidien or Mallinckrodt shall bear only those costs that relate solely and directly to obtaining such licenses (or other appropriate rights) in the ordinary course. The Parties acknowledge and agree that there can be no assurance that Covidien's efforts will be successful or that Mallinckrodt will be able to obtain such licenses or rights on acceptable terms or at all and, where Covidien enjoys rights under any enterprise or site license or similar license, the Parties acknowledge that such license typically precludes partial transfers or assignments or operation of a service bureau on behalf of unaffiliated entities. In the event that Mallinckrodt is unable to obtain such software licenses, the Parties shall work together using commercially reasonable efforts to obtain an alternative software license to allow Covidien to provide, and Mallinckrodt to receive, such Covidien Services, and the Parties shall negotiate in good faith an amendment to the applicable Schedule to reflect any such new arrangement.

(b) If and to the extent requested by Covidien, Mallinckrodt shall use commercially reasonable efforts to assist Covidien in its efforts to obtain licenses (or other appropriate rights) to use, duplicate and distribute, as necessary and applicable, certain computer software necessary for Mallinckrodt to provide, and Covidien to receive, Mallinckrodt Services; provided, however, that Mallinckrodt shall not be required to pay any fees or other payments or incur any obligations or liabilities to enable Covidien to obtain any such license or rights (except and to the extent that Covidien advances such fees or payments to Mallinckrodt); provided, further, that Mallinckrodt shall not be required to seek broader rights or more favorable terms for Covidien than those applicable to Covidien or Mallinckrodt, as the case may be, prior to the date of this Agreement or as may be applicable to Mallinckrodt from time to time hereafter; and, provided, further, that Covidien shall bear only those costs that relate solely and directly to obtaining such licenses (or other appropriate rights) in the ordinary course. The Parties acknowledge and agree that there can be no assurance that Mallinckrodt's efforts will be successful or that Covidien will be able to obtain such licenses or rights on acceptable terms or at all and, where Mallinckrodt enjoys rights under any enterprise or site license or similar license, the Parties acknowledge that such licenses typically precludes partial transfers or assignments or operation of a service bureau on behalf of unaffiliated entities. In the event that Covidien is unable to obtain such software licenses to allow Mallinckrodt to provide, and Covidien to receive, such Mallinckrodt Services, and the Parties shall negotiate in good faith an amendment to the applicable Schedule to reflect any such new arrangement, which amended Schedule shall not require Covidien to pay for any fees, expenses or costs relating to the software license that Covidien was unable to obtain pursuant to the provisions of

(c) In the event that there are any costs associated with obtaining software licenses in accordance with <u>Section 4.01</u> that (i) would not be payable in the ordinary course, including in the form of a "transfer fee" or other similar fees or expenses payable by the Recipient or the Provider, and (ii) would not have been payable by the Recipient or the Provider absent the need for a consent or waiver in connection with the license that the Recipient is seeking to obtain, such costs shall be borne by the Recipient.

Section 4.02. <u>Covidien Computer-Based and Other Resources</u>. (a) From and after the date of this Agreement, Mallinckrodt and its Affiliates shall cause all of their personnel having access to the Covidien Intranet or such other computer software, networks, hardware, technology or computer based resources pursuant to the Separation Agreement, or any Ancillary Agreement, or in connection with performance, receipt or delivery of a Service, to comply with all security guidelines (including physical security, network access, internet security, confidentiality and personal data security guidelines) of Covidien and its Affiliates (of which Covidien provides Mallinckrodt written notice). Mallinckrodt shall ensure that the access contemplated by this <u>Section 4.02</u> shall be used by such personnel only for the purposes contemplated by, and subject to the terms of, this Agreement. Except as expressly provided in the Separation Agreement, any other Ancillary Agreement or any other applicable agreement or as required in connection with the performance or delivery of any Services, each of the Parties and its Affiliates shall cease using (and shall cause their employees to cease using) the services made available by the other Party and its Affiliates prior to the date of this Agreement.

Section 4.03. Access to Facilities. (a) Mallinckrodt shall, and shall cause its Subsidiaries to, allow Covidien and its Representatives reasonable access to the facilities of Mallinckrodt necessary for Covidien to fulfill its obligations under this Agreement.

(b) Covidien shall, and shall cause its Subsidiaries to, allow Mallinckrodt and its Representatives reasonable access to the facilities of Covidien necessary for Mallinckrodt to fulfill its obligations under this Agreement.

(c) Notwithstanding the other rights of access of the Parties under this Agreement, each Party shall, and shall cause its Subsidiaries to, afford the other Party, its Subsidiaries and Representatives, following not less than five (5) business days' prior written notice from the other Party, reasonable access during normal business hours to the facilities, information, systems, infrastructure, and personnel of the relevant Providers as reasonably necessary for the other Party to verify the adequacy of internal controls over information technology, reporting of financial data and related processes employed in connection with the Services, including in connection with verifying compliance with Section 404 of the Sarbanes-Oxley Act of 2002; provided, however, such access shall not unreasonably interfere with any of the business or operations of such Party or its Subsidiaries.

(d) Except as otherwise permitted by the other Party in writing, each Party shall permit only its authorized Representatives, contractors, invitees or licensees to access the other Party's facilities.

Section 4.04. <u>Cooperation</u>. It is understood that it will require the significant efforts of both Parties to implement this Agreement and to ensure performance of this Agreement by the Parties at the agreed-upon levels in accordance with all of the terms and conditions of this Agreement. The Parties will cooperate, acting in good faith and using commercially reasonable efforts, to effect a smooth and orderly transition of the Services provided under this Agreement from the Provider to the Recipient (including repairs and maintenance Services and the assignment or transfer of the rights and obligations under any third-party contracts relating to the Services); provided, however, that this Section 4.04 shall not require either Party to incur any out-of-pocket costs or expenses.

Section 4.05. <u>Data Protection</u>. The Provider shall only process personal data which it may receive from the Recipient, while carrying out its duties under this Agreement: (a) in such a manner as is necessary to carry out those duties; (b) in accordance with the instructions of the Recipient; and (c) using appropriate technical and organizational measures to prevent the unauthorised or unlawful processing of such personal data and/or the accidental loss or destruction of, or damage to, such personal data.

#### ARTICLE V COSTS AND DISBURSEMENTS

Section 5.01. <u>Costs and Disbursements</u>. (a) Except as otherwise provided in this Agreement, a Recipient of Services shall pay to the Provider of such Services a monthly fee for the Services (or category of Services, as applicable) (each fee constituting a "<u>Service Charge</u>" and, collectively, "<u>Service Charges</u>") as listed on the Schedules hereto. With respect to each

Service or category of Services, the applicable Schedule shall set forth (i) the Recipient that will be invoiced the Service Charge for such Service or category of Services and (ii) the Provider that will be paid such Service Charge.

(b) The amount of the Service Charge for each Service shall increase three percent (3%) annually on each anniversary of this Agreement (including during the term of any Service Extension). In addition, during the term of this Agreement, the amount of a Service Charge for any Services (or category of Services, as applicable) may increase to the extent of: (i) any increases mutually agreed to by the Parties, (ii) any Service Charges applicable to any Additional Services, Service Increases or New Services, and (iii) any increase in the rates or charges imposed by any unaffiliated third-party provider that is providing Services. Together with any monthly invoice for Service Charges and Reimbursement Charges, the Provider shall provide the Recipient with documentation to support the calculation of such Service Charges or any Reimbursement Charges.

(c) The Recipient shall reimburse the Provider for reasonable out-of-pocket costs and expenses incurred by the Provider or its Affiliates in connection with providing the Services (including necessary travel-related expenses) (each such cost or expense, a "<u>Reimbursement Charge</u>" and, collectively, "<u>Reimbursement Charges</u>"); <u>provided</u>, <u>however</u>, that any such cost or expense that is materially inconsistent with historical practice between the Parties for any Service (including business travel and related expenses) shall require advance approval of the Recipient. Any authorized travel-related expenses incurred in performing the Services shall be incurred and charged to the Recipient in accordance with the Provider's then-applicable business travel policies made known to the Recipient.

(d) The Service Charges and Reimbursement Charges due and payable hereunder shall be invoiced and paid in the currency expressly applicable to such Service in the relevant Schedule hereto. The Recipient shall pay the amount of each monthly invoice by wire transfer (or such other method of payment as may be agreed between the Parties) to the Provider within sixty (60) days of the receipt of each such invoice, including appropriate documentation as described herein. In the absence of a timely notice of billing dispute in accordance with the provisions of Article VIII of the Separation Agreement, if the Recipient fails to pay such amount by the due date, the Recipient shall be obligated to pay to the Provider, in addition to the amount due, interest at an annual default interest rate of three percent (3%), or the maximum legal rate, whichever is lower (the "Interest Payment"), accruing from the date the payment was due through the date of actual payment. In the event of any billing dispute, the Recipient shall promptly pay any undisputed amount.

(e) Subject to the confidentiality provisions set forth in <u>Section 9.03</u>, each Party shall, and shall cause their respective Affiliates to, provide, upon ten (10) days' prior written notice from the other Party, any information within such Party's or its Affiliates' possession that the requesting Party reasonably requests in connection with any Services being provided to such requesting Party by an unaffiliated third-party provider, including any applicable invoices, agreements documenting the arrangements between such third-party provider and the Provider and other supporting documentation; provided, however, that each Party shall make no more than one such request during any calendar month.

Section 5.02. <u>Tax Matters</u>. (a) Without limiting any provisions of this Agreement, the Recipient shall be responsible for (i) all excise, sales, use, transfer, stamp, documentary, filing, recordation and other similar Taxes, (ii) any value added, goods and services or similar recoverable indirect Taxes ("<u>VAT</u>") and (iii) any related interest and penalties (collectively, "<u>Transfer Taxes</u>"), in each case imposed or assessed as a result of the provision of Services by the Provider. In particular, but without prejudice to the generality of the foregoing, all amounts payable pursuant to this Agreement are exclusive of amounts in respect of VAT. Where any taxable supply for VAT purposes is made pursuant to this Agreement by the Provider to the Recipient, the Recipient shall either (i) on receipt of a valid VAT invoice from the Provider, pay to the Provider such additional amounts in respect of VAT as are chargeable on the supply of the services; or (ii) where required by legislation to do so, account directly to the relevant Governmental Authority for any such VAT amounts. The Party required to account for Transfer Tax shall provide to the other Party evidence of the remittance of the amount of such Transfer Tax to the relevant Governmental Authority, including, without limitation, copies of any Tax returns remitting such amount. The Provider agrees that it shall take commercially reasonable actions to cooperate with the Recipient in obtaining any refund, return, rebate, or the like of any Transfer Tax. For the avoidance of doubt, any applicable gross receipts-based or net income-based Taxes shall be one by the Provider unless the Provider is required by law to obtain, or allowed to separately invoice for and obtain, reimbursement of such Taxes from the Recipient.

(b) The Recipient shall be entitled to deduct and withhold Taxes required by any Governmental Requirements to be withheld on payments made pursuant to this Agreement. To the extent any amounts are so withheld, the Recipient shall (i) pay, in addition to the amount otherwise due to the Provider under this Agreement, such additional amount as is necessary to ensure that the net amount actually received by the Provider will equal the full amount the Provider would have received had no such deduction or withholding been required, (ii) pay such deducted and withheld amount to the proper Governmental Authority, and (iii) promptly provide to the Provider evidence of such payment to such Governmental Authority. The Provider shall, prior to the date of any payment to be made pursuant to this Agreement, at the request of the Recipient, make commercially reasonable efforts to provide the Recipient any certificate or other documentary evidence (x) required by Governmental Requirements or (y) which the Provider is entitled by Governmental Requirements to provide in order to reduce the amount of any Taxes that may be deducted or withheld from such payment and the Recipient agrees to accept and act in reliance on any such duly and properly executed certificate or other applicable documentary evidence.

(c) If the Provider (i) receives any refund (whether by payment, offset, credit or otherwise) or (ii) utilizes any overpayment of Taxes that are borne by Recipient pursuant to this Agreement, then the Provider shall promptly pay, or cause to be paid, to the Recipient an amount equal to the deficiency or excess, as the case may be, with respect to the amount that the Recipient has borne if the amount of such refund or overpayment (including, for the avoidance of doubt, any interest or other amounts received with respect to such refund or overpayment) had been included originally in the determination of the amounts to be borne by Recipient pursuant to this Agreement, net of any additional Taxes the Provider incurs or will incur as a result of the receipt of such refund or such overpayment.

Section 5.03. <u>No Right to Set-Off</u>. The Recipient shall timely pay the full amount of Service Charges and Reimbursement Charges and shall not set-off, counterclaim or otherwise withhold any amount owed to the Provider under this Agreement on account of any obligation owed by the Provider to the Recipient.

## ARTICLE VI STANDARD FOR SERVICE

#### Section 6.01. Standard for Service.

(a) The Provider agrees (i) to perform the Services with substantially the same nature, quality, standard of care and service levels at which the same or similar services were performed by or on behalf of the Provider prior to the Distribution Date or, if not so previously provided, then substantially similar to that which are applicable to similar services provided to the Provider's Affiliates or other business components; and (ii) upon receipt of written notice from the Recipient identifying any outage, interruption or other failure of any Service, to respond to such outage, interruption or other failure of such Service in a manner that is substantially similar to the manner in which such Provider or its Affiliates responded to any outage, interruption or other failure of the same or similar services prior to the Distribution Date. The Parties acknowledge that an outage, interruption or other failure of any Service shall not be deemed to be a breach of the provisions of this Section 6.01 so long as the applicable Provider complies with the foregoing clause (ii).

(b) Nothing in this Agreement shall require the Provider to perform or cause to be performed any Service to the extent the manner of such performance would constitute a violation of applicable Law or any existing contract or agreement with a third party. If the Provider is or becomes aware of any potential violation on the part of the Provider, the Provider shall promptly send a written notice to the Recipient of any such potential violation. The Parties each agree to cooperate and use commercially reasonable efforts to obtain any necessary third-party consents required under any existing contract or agreement with a third party to allow the Provider to perform or cause to be performed any Service in accordance with the standards set forth in this <u>Section 6.01</u>. Any costs and expenses incurred by either Party in connection with obtaining any such third-party consent that is required to allow the Provider to perform or cause to be performed. If, with respect to a Service, the Parties, despite the use of such commercially reasonable efforts, are unable to obtain a required third-party consent or the performance of such Service by the Provider would continue to constitute a violation of applicable Laws, the Provider shall use commercially reasonable efforts in good faith to provide such Services in a manner as closely as possible to the standards described in this <u>Section 6.01</u> that would apply absent the exception provided for in the first sentence of this <u>Section 6.01(b)</u>.

Section 6.02. <u>Disclaimer of Warranties</u>. EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, THE PARTIES ACKNOWLEDGE AND AGREE THAT THE SERVICES ARE PROVIDED AS-IS, THAT EACH RECIPIENT ASSUMES ALL RISKS

AND LIABILITY ARISING FROM OR RELATING TO ITS USE OF AND RELIANCE UPON THE SERVICES AND EACH PROVIDER, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, MAKES NO REPRESENTATION OR WARRANTY WITH RESPECT THERETO. EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, EACH PROVIDER HEREBY EXPRESSLY DISCLAIMS ALL REPRESENTATIONS AND WARRANTIES REGARDING THE SERVICES, WHETHER EXPRESS OR IMPLIED, EITHER IN FACT OR BY OPERATION OF LAW, BY STATUTE OR OTHERWISE, INCLUDING ANY REPRESENTATION OR WARRANTY IN REGARD TO QUALITY, PERFORMANCE, NONINFRINGEMENT, COMMERCIAL UTILITY, MERCHANTABILITY OR FITNESS OF ANY SERVICE FOR A PARTICULAR PURPOSE.

Section 6.03. <u>Compliance with Laws and Regulations</u>. Each Party shall be responsible for its own compliance and its subcontractors' compliance with any and all Laws applicable to its performance under this Agreement. No Party will knowingly take any action in violation of any such applicable Law that results in liability being imposed on the other Party.

#### ARTICLE VII LIMITED LIABILITY AND INDEMNIFICATION

Section 7.01. <u>Consequential and Other Damages</u>. Notwithstanding anything to the contrary contained in the Separation Agreement or this Agreement, the Provider shall not be liable to the Recipient or any of its Affiliates or Representatives, whether in contract, tort (including negligence and strict liability) or otherwise, at law or equity, for any special, indirect, incidental, punitive or consequential damages whatsoever (including lost profits or damages calculated on multiples of earnings approaches), which in any way arise out of, relate to or are a consequence of, the performance or nonperformance by the Provider (including any Affiliates and Representatives of the Provider and any unaffiliated third-party providers, in each case, providing the applicable Services) under this Agreement or the provision of, or failure to provide, any Services under this Agreement, including with respect to loss of profits, business interruptions or claims of customers.

Section 7.02. <u>Limitation of Liability</u>. The Liabilities of each Provider and its Affiliates and Representatives, collectively, under this Agreement for any act or failure to act in connection herewith (including the performance or breach of this Agreement), or from the sale, delivery, provision or use of any Services provided under or contemplated by this Agreement, whether in contract, tort (including negligence and strict liability) or otherwise, at law or equity, shall not exceed the total aggregate Service Charges (excluding any Reimbursement Charges) actually paid to such Provider by the Recipient pursuant to this Agreement. The foregoing limitations on Liability in this <u>Section 7.02</u> shall not apply to any breach of <u>Section 9.03</u>.

Section 7.03. <u>Obligation To Reperform; Liabilities</u>. In the event of any breach of this Agreement by any Provider with respect to the provision of any Services (with respect to which the Provider can reasonably be expected to re-perform in a commercially reasonable manner), the Provider shall (a) promptly correct in all material respects such error, defect or breach or re-perform in all material respects such Services at the request of the Recipient and at the sole cost and expense of the Provider and (b) subject to the limitations set forth in <u>Sections 7.01</u> and <u>7.02</u>,

reimburse the Recipient and its Affiliates and Representatives for Liabilities attributable to such breach by the Provider. The remedy set forth in this <u>Section 7.03</u> shall be the sole and exclusive remedy of the Recipient for any such breach of this Agreement. Any request for re-performance in accordance with this <u>Section 7.03</u> by the Recipient must be in writing and specify in reasonable detail the particular error, defect or breach, and such request must be made no more than one (1) month from the date such error, defect or breach becomes apparent or should have reasonably become apparent to the Recipient.

Section 7.04. <u>Release and Recipient Indemnity</u>. Subject to <u>Section 7.01</u>, each Recipient hereby releases the applicable Provider and its Affiliates and Representatives (each, a "<u>Provider Indemnified Party</u>"), and each Recipient hereby agrees to indemnify, defend and hold harmless each such Provider Indemnified Party from and against any and all Liabilities arising from, relating to or in connection with: (a) the use of any Services by such Recipient or any of its Affiliates, Representatives or other Persons using such Services; or (b) the sale, delivery, provision or use of any Services provided under or contemplated by this Agreement, in the case of each of clause (a) and (b), except to the extent that such Liabilities arise out of, relate to or are a consequence of the applicable Provider Indemnified Party's bad faith, gross negligence or willful misconduct.

Section 7.05. <u>Provider Indemnity</u>. Subject to <u>Section 7.01</u>, each Provider hereby agrees to indemnify, defend and hold harmless the applicable Recipient and its Affiliates and Representatives (each a "<u>Recipient Indemnified Party</u>"), from and against any and all Liabilities arising from, relating to or in connection with: (a) the use of any Services by such Recipient or any of its Affiliates, Representatives or other Persons using such Services; or (b) the sale, delivery, provision or use of any Services provided under or contemplated by this Agreement, in the case of each of clause (a) and (b), to the extent that such Liabilities arise out of, relate to or are a consequence of the applicable Provider's bad faith, gross negligence or willful misconduct.

Section 7.06. <u>Indemnification Procedures</u>. The provisions of Sections 4.5 and 4.6 of the Separation Agreement shall govern claims for indemnification under this Agreement.

Section 7.07. <u>Liability for Payment Obligations</u>. Nothing in this <u>Article VII</u> shall be deemed to eliminate or limit, in any respect, Covidien's or Mallinckrodt's express obligation in this Agreement to pay Service Charges and Reimbursement Charges for Services rendered in accordance with this Agreement.

Section 7.08. Exclusion of Other Remedies. The provisions of Sections 7.03, 7.04 and 7.05 of this Agreement shall, to the maximum extent permitted by applicable Law, be the sole and exclusive remedies of the Provider Indemnified Parties and the Recipient Indemnified Parties, as applicable, for any claim, loss, damage, expense or liability, whether arising from statute, principle of common or civil law, principles of strict liability, tort, contract or otherwise under this Agreement.

Section 7.09. Confirmation. Neither Party excludes responsibility for any liability which cannot be excluded pursuant to applicable Law.

#### ARTICLE VIII TERM AND TERMINATION

Section 8.01. <u>Term and Termination</u>. (a) This Agreement shall commence immediately upon the Distribution Date and shall terminate upon the earlier to occur of: (i) the last date on which either Party is obligated to provide any Service to the other Party in accordance with the terms of this Agreement or (ii) the mutual written agreement of the Parties to terminate this Agreement in its entirety.

(b) Without prejudice to a Recipient's rights with respect to a Force Majeure, a Recipient may from time to time terminate this Agreement with respect to the entirety of any individual Service but not a portion thereof:

(i) for any reason or no reason, upon providing at least sixty (60) days' prior written notice to the Provider; <u>provided</u>, <u>however</u>, that the Recipient shall pay to the Provider the necessary and reasonable documented out-of-pocket costs incurred in connection with the wind down of such Service other than any employee severance and relocation expenses, but including unamortized license fees and costs for equipment used to provide such Service, contractual obligations under agreements used to provide such Service, any breakage or termination fees and any other termination costs payable by the Provider with respect to any resources or pursuant to any other third-party agreements that were used by the Provider to provide such Service (or an equitably allocated portion thereof, in the case of any such equipment, resources or agreements that also were used for purposes other than providing Services); or

(ii) if the Provider of such Service has failed to perform any of its material obligations under this Agreement with respect to such Service, and such failure shall continue to exist thirty (30) days after receipt by the Provider of written notice of such failure from the Recipient.

In the event that any Service is terminated other than at the end of a month, the Service Charge associated with such Service shall be pro-rated appropriately. The Parties acknowledge that there may be interdependencies among the Services being provided under this Agreement that may not be identified on the applicable Schedules and agree that, if the Provider's ability to provide a particular Service in accordance with this Agreement is materially and adversely affected by the termination of another Service in accordance with <u>Section 8.01(b)(i)</u>, then the Parties shall negotiate in good faith to amend the Schedule relating to such affected continuing Service, which amendment shall be consistent with the terms of, and the pricing methodology used for, comparable Services.

(c) In connection with the termination of any Service, if the Recipient reasonably determines that it will require such Service to continue beyond the date on which such Service is scheduled to terminate, the Recipient may request that the Provider extend such Service (any such extension, a "<u>Service Extension</u>") for a specified period beyond the scheduled termination of such Service (which period shall in no event be longer than one hundred and eighty (180) days) by written notice to the Provider no less than sixty (60) days prior to the date of such scheduled termination, and Provider shall consider any such request in good faith; <u>provided</u>, <u>however</u>, that no Party shall be obligated to agree to any Service Extension, including because, after good-faith negotiations between the Parties, the Parties fail to reach an agreement with respect to the terms thereof; <u>provided</u>, <u>further</u>,

however, that (i) there shall be no more than one (1) Service Extension with respect to each Service and (ii) the Provider shall not be obligated to provide such Service Extension if a third-party consent is required and cannot be obtained by the Provider. Unless otherwise agreed by Provider and Recipient, the Service Charge applicable to any such Service Extension shall be one hundred and twenty percent (120%) of the Service Charge applicable to such Service immediately prior to the Service Extension. In connection with any request for Service Extensions in accordance with this <u>Section 8.01(c)</u>, the Covidien Services Manager and the Mallinckrodt Services Manager shall in good faith (x) negotiate the terms of an amendment to the applicable Schedule, which amendment shall be consistent with the terms of the applicable Service, and (y) determine the costs and expenses (other than Service Charges), if any, that would be incurred by the Provider or the Recipient, as the case may be, in connection with the provision of such Service Extension, which costs and expenses shall be borne solely by the Party requesting the Service Extension. Each amended Schedule to implement a Service Extension, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such agreement and any Services provided pursuant to such Service Extensions shall be deemed "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement.

Section 8.02. Effect of Termination. Upon termination of any Service pursuant to this Agreement, the Provider of the terminated Service will have no further obligation to provide the terminated Service, and the relevant Recipient will have no obligation to pay any future Service Charges relating to any such Service; provided, however, that the Recipient shall remain obligated to the relevant Provider for the (i) Service Charges and Reimbursement Charges owed and payable in respect of Services provided prior to the effective date of termination and (ii) any applicable charges described in Section 8.01(b)(i), which charges shall be payable only in the event that the Recipient terminates any Service pursuant to Section 8.01(b)(i). In connection with the termination of any Service, the provisions of this Agreement not relating solely to such terminated Service shall survive any such termination, and in connection with a termination of this Agreement, Article I, Article VII (including liability in respect of any indemnifiable Liabilities under this Agreement arising or occurring on or prior to the date of termination), Article VIII, Article IX, all confidentiality obligations under this Agreement and liability for all due and unpaid Service Charges and Reimbursement Charges payable pursuant to Section 8.01(b)(i), shall continue to survive indefinitely.

Section 8.03. Force Majeure. (a) Neither Party (nor any Person acting on its behalf) shall have any liability or responsibility for failure to fulfill any obligation (other than a payment obligation) under this Agreement so long as and to the extent to which the fulfillment of such obligation is prevented, frustrated, hindered or delayed as a consequence of a Force Majeure; provided, however, that (i) such Party (or such Person) shall have exercised commercially reasonable efforts to minimize the effect of such Force Majeure on its obligations; and (ii) the nature, quality and standard of care that the Provider shall provide in delivering a Service after a Force Majeure shall be substantially the same as the nature, quality and standard of care that the Provider store of a Sore Majeure, the Party whose performance is affected thereby shall give notice of suspension as soon as reasonably practicable to the other stating the date and extent of such suspension and the cause thereof, and such Party shall resume the performance of such obligations as soon as reasonably practicable after the removal of such cause.

(b) During the period of a Force Majeure, the Recipient shall be entitled to permanently terminate such Service(s) (and shall be relieved of the obligation to pay Service Charges for such Services(s) throughout the duration of such Force Majeure) if a Force Majeure shall continue to exist for more than fifteen (15) consecutive days, it being understood that Recipient shall not be required to provide any advance notice of such termination to Provider or pay any charges in connection therewith.

#### ARTICLE IX GENERAL PROVISIONS

Section 9.01. <u>No Agency</u>. Nothing in this Agreement shall be deemed in any way or for any purpose to constitute any Party an agent of an unaffiliated party in the conduct of such other party's business. A Provider of any Service under this Agreement shall act as an independent contractor and not as the agent of the Recipient in performing such Service, maintaining control over its employees, its subcontractors and their employees and complying with all withholding of income at source requirements, whether federal, national, state, local or foreign.

Section 9.02. <u>Subcontractors</u>. A Provider may hire or engage one or more subcontractors to perform any or all of its obligations under this Agreement; <u>provided</u>, <u>however</u>, that (i) such Provider shall use the same degree of care in selecting any such subcontractor as it would if such contractor was being retained to provide similar services to the Provider and (ii) such Provider shall in all cases remain primarily responsible for all of its obligations under this Agreement with respect to the scope of the Services, the standard for services as set forth in <u>Article VI</u> and the content of the Services provided to the Recipient.

### Section 9.03. Treatment of Confidential Information.

(a) The Parties shall not, and shall cause all other persons providing Services or having access to information of the other Party that is confidential or proprietary ("Confidential Information") not to, disclose to any other person or use, except for purposes of this Agreement, any Confidential Information of the other Party; provided, however, that the Confidential Information may be used by such Party to the extent that such Confidential Information has been (i) in the public domain through no fault of such Party or any member of such Group or any of their respective Representatives, (ii) later lawfully acquired from other sources by such Party (or any member of such Party's Group) which sources are not themselves bound by a confidentiality obligation, or (iii) independently generated without reference to any Confidential Information of the other Party; provided, further, that each Party may disclose Confidential Information of the other Party, to the extent not prohibited by applicable Law: (i) to its Representatives on a need-to-know basis in connection with the performance of such Party's obligations under this Agreement; (ii) in any report, statement, testimony or other submission required to be made to any Governmental Authority having jurisdiction over the disclosing Party; or (iii) in order to comply with applicable Law, or in response to any summons, subpoena or other legal process or formal or informal investigative demand issued to the disclosing Party in the course of any litigation, investigation or administrative proceeding. In the event that a Party becomes legally compelled (based on advice of counsel) by deposition, interrogatory, request for documents subpoena, civil investigative demand or similar judicial or administrative process to disclose any Confidential Information of the other Party, such disclosing Party shall provide the other Party with prompt prior written notice of such requirement, and, to the extent reasonably practicable, cooperate with the other Party (at such other Party's expense) to obtain a protective order or similar remedy to cause such Confidential Information not to be disclosed, including interposing all available objections thereto, such as objections based on settlement privilege. In the event that such protective order or other similar remedy is not obtained, the disclosing Party shall furnish only that portion of the Confidential Information that has been legally compelled, and shall exercise its commercially

reasonable efforts (at such other Party's expense) to obtain assurance that confidential treatment will be accorded such Confidential Information.

(b) Each Party shall, and shall cause its Representatives to, protect the Confidential Information of the other Party by using the same degree of care to prevent the unauthorized disclosure of such as the Party uses to protect its own confidential information of a like nature, but in any event no less than a reasonable degree of care.

(c) Each Party shall be liable for any failure by its respective Representatives to comply with the restrictions on use and disclosure of Confidential Information contained in this Agreement.

(d) Each Party shall comply with all applicable local, state, national, federal and foreign privacy and data protection Laws that are or that may in the future be applicable to the provision of Services under this Agreement.

Section 9.04. <u>Further Assurances</u>. Each Party covenants and agrees that, without any additional consideration, it shall execute and deliver any further legal instruments and perform any acts that are or may become necessary to effectuate this Agreement.

Section 9.05. <u>Dispute Resolution</u>. Any Dispute shall be resolved in accordance with the procedures set forth in Article VIII of the Separation Agreement, which shall be the sole and exclusive procedures for the resolution of any such Dispute unless otherwise specified herein or in Article VIII of the Separation Agreement.

Section 9.06. <u>Notices</u>. Except with respect to routine communications by the Covidien Services Manager, Mallinckrodt Services Manager, Covidien Local Services Manager and Mallinckrodt Local Services Manager under <u>Section 2.06</u>, all notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by facsimile or electronic transmission with receipt confirmed (followed by delivery of an original via overnight courier service) or by registered or certified mail (postage prepaid, return receipt requested) to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this <u>Section 9.06</u>):

(i) if to Covidien:

Covidien plc 1st Floor, 20 on Hatch Lower Hatch Street Dublin 2 Ireland Attn: Chief Financial Officer Facsimile: +352-266-379-92

with copies to:

Covidien plc 15 Hampshire Street Mansfield, MA 02048 Attn: General Counsel Facsimile: (508) 261-8544

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Attention: Adam O. Emmerich Benjamin M. Roth Facsimile: (212) 403-2000

(ii) if to Mallinckrodt:

Mallinckrodt plc Damastown, Mulhuddart Dublin 15 Ireland Attn: Chief Financial Officer Facsimile: +352-266-379-92

with copies to:

Mallinckrodt plc 675 James S. McDonnell Blvd. Hazelwood, MO 63042 Attn: General Counsel Facsimile: (314) 654-5366

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Attention: Adam O. Emmerich Benjamin M. Roth Facsimile: (212) 403-2000

Section 9.07. <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced under any Law or as a matter of public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated by this Agreement is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as

closely as possible in a mutually acceptable manner in order that the transactions contemplated by this Agreement be consummated as originally contemplated to the greatest extent possible.

Section 9.08. Entire Agreement. This Agreement, together with the documents referenced herein (including the Separation Agreement and any other Ancillary Agreements) constitutes the entire agreement between the parties with respect to the subject matter hereof, supersede all prior written and oral and all contemporaneous oral agreements, negotiations, discussions, writings, understandings, commitments and conversations with respect to such subject matter and there are no agreements or understandings between the parties other than those set forth or referred to herein or therein.

Section 9.09. <u>No Third-Party Beneficiaries</u>. Except as provided in <u>Article VII</u> with respect to Provider Indemnified Parties and Recipient Indemnified Parties, this Agreement is for the sole benefit of the Parties and their permitted successors and assigns and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person, including any union or any employee or former employee of Covidien or Mallinckrodt, any legal or equitable right, benefit or remedy of any nature whatsoever, including any rights of employment for any specified period, under or by reason of this Agreement.

Section 9.10. <u>Governing Law</u>. This Agreement (and any claims or disputes arising out of or related to this Agreement or to the transactions contemplated by this Agreement or to the inducement of any Party to enter into this Agreement or the transactions contemplated by this Agreement, whether for breach of contract, tortious conduct or otherwise and whether predicated on common law, statute or otherwise) shall in all respects be governed by, and construed in accordance with, the Laws of the State of New York, including all matters of construction, validity and performance, in each case without reference to any conflict of Law rules that might lead to the application of the Laws of any other jurisdiction (other than Section 5-1401 and Section 5-1402 of the General Obligations Law of the State of New York).

Section 9.11. <u>Amendment</u>. No provision of this Agreement, including any Schedules to this Agreement, may be amended, supplemented or modified except by a written instrument making specific reference to this Agreement or any such Schedules to this Agreement, as applicable, signed by all the Parties.

Section 9.12. <u>Rules of Construction</u>. Interpretation of this Agreement shall be governed by the following rules of construction: (a) words in the singular shall be held to include the plural and vice versa, and words of one gender shall be held to include the other gender as the context requires; (b) references to the terms Article, Section, paragraph and Schedule are references to the Articles, Sections, paragraphs and Schedules of this Agreement unless otherwise specified; (c) references to "\$" shall mean U.S. dollars; (d) the word "including" and words of similar import when used in this Agreement shall mean "including without limitation," unless otherwise specified; (e) the word "or" shall not be exclusive; (f) references to "written" or "in writing" include in electronic form; (g) provisions shall apply, when appropriate, to successive events and transactions; (h) the headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement; (i) Covidien and Mallinckrodt have each participated in the negotiation and drafting of this Agreement and if an ambiguity or question of interpretation should arise, this Agreement shall be construed as if

drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or burdening either Party by virtue of the authorship of any of the provisions in this Agreement or any interim drafts of this Agreement; (j) a reference to any Person includes such Person's successors and permitted assigns; (k) any reference to "days" means calendar days unless business days are expressly specified; and (l) when calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded, and if the last day of such period is not a business day, the period shall end on the next succeeding business day.

Section 9.13. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and by each Party in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Agreement by facsimile or portable document format (PDF) shall be as effective as delivery of a manually executed counterpart of this Agreement.

Section 9.14. <u>Assignability</u>. This Agreement shall not be assigned by operation of Law or otherwise without the prior written consent of Covidien and Mallinekrodt, except that each Party may:

(a) assign all of its rights and obligations under this Agreement to any of its Subsidiaries; <u>provided</u>, that in connection with any such assignment, the assigning Party provides a guarantee to the non-assigning Party (in a form reasonably agreed upon) for any liability or obligation of the assignee under this Agreement;

(b) in connection with the divestiture of any Subsidiary or business of such Party that is a Recipient to an acquiror that is not a competitor of the Provider, assign to the acquiror of such Subsidiary or business its rights and obligations as a Recipient with respect to the Services provided to such divested Subsidiary or business under this Agreement; <u>provided</u>, that (i) in connection with any such assignment, the assigning Party provides a guarantee to the non-assigning Party (in a form reasonably agreed upon) for any liability or obligation of the assignee under this Agreement, (ii) any and all costs and expenses incurred by either Party in connection with such assignment (including in connection with clause (iii) of this proviso) shall be borne solely by the assigning Party, and (iii) the Parties shall in good faith negotiate any amendments to this Agreement, including the Schedules hereto, that may be necessary or appropriate in order to assign such Services; and

(c) in connection with the divestiture of any Subsidiary or business of such Party that is a Recipient to an acquiror that is a competitor of the Provider, assign to the acquiror of such Subsidiary or business its rights and obligations as a Recipient with respect to the Services provided to such divested Subsidiary or business under this Agreement; provided, that (i) in connection with any such assignment, the assigning Party provides a guarantee to the non-assigning Party (in a form reasonably agreed upon) for any liability or obligation of the assignee under this Agreement, (ii) any and all costs and expenses incurred by either Party in connection with such assignment (including in connection with clause (iii) of this proviso) shall be borne solely by the assigning Party, (iii) the Parties shall in good faith negotiate any amendments to this Agreement, including the Schedules hereto, that may be necessary or appropriate in order to ensure that such

assignment will not (x) materially and adversely affect the businesses and operations of each of the Parties and their respective Affiliates or (y) create a competitive disadvantage for the Provider with respect to an acquiror that is a competitor, and (iv) no Party shall be obligated to provide any such assigned Services to an acquiror that is a competitor of such assigned Services to such acquiror would disrupt the operation of such Party's businesses or create a competitive disadvantage for such Party with respect to such acquirer.

Section 9.15. <u>Public Announcements</u>. From and after the Distribution Date, the Parties shall consult with each other before issuing, and give each other the opportunity to review and comment upon, that portion of any press release or other public statements that relates to the transactions contemplated by this Agreement, and shall not issue any such press release or make any such public statement prior to such consultation, except (a) as may be required by applicable Law, court process or by obligations pursuant to any listing agreement with any national securities exchange or national securities quotation system; or (b) as otherwise set forth in the Separation Agreement.

Section 9.16. <u>Non-Recourse</u>. No past, present or future director, officer, employee, incorporator, member, partner, shareholder, Affiliate, agent, attorney or representative of either Covidien or Mallinckrodt or their Affiliates shall have any liability for any obligations or liabilities of Covidien or Mallinckrodt, respectively, under this Agreement or for any claims based on, in respect of, or by reason of, the transactions contemplated by this Agreement.

[The remainder of this page is intentionally left blank.]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed on the date first written above by their respective duly authorized officers.

# **COVIDIEN PLC**

By: Name:

Title:

# MALLINCKRODT PLC

By:

Name: Title:

# TAX MATTERS AGREEMENT

This Tax Matters Agreement (this "**Agreement**") is entered into as of [—], 2013 between Covidien plc, a corporation organized under the laws of Ireland ("**Covidien**"), and Mallinckrodt plc, a corporation organized under the laws of Ireland ("**Mallinckrodt**" and, together with Covidien, the "**Parties**"). Capitalized terms used in this Agreement and not otherwise defined herein shall have the meanings ascribed to such terms in the Separation and Distribution Agreement, dated as of the date hereof, between Covidien and Mallinckrodt (the "**Distribution Agreement**").

# RECITALS

WHEREAS, the board of directors of Covidien has determined that it is in the best interests of Covidien and its shareholders that the Mallinckrodt Business be operated by a newly incorporated publicly traded company;

WHEREAS, Mallinckrodt has been incorporated for these purposes and has not engaged in activities except those incidental to its formation and in preparation for the Distribution;

WHEREAS, Covidien will effect the restructuring transactions described in the Plan of Reorganization (as defined in the Distribution Agreement) for the purpose of aggregating the Mallinckrodt Business in the Mallinckrodt Group prior to the Distribution (collectively, the "**Reorganization**");

WHEREAS, Covidien currently intends that, on the Distribution Date, it will make a distribution in specie of the Mallinckrodt Business to the holders of Covidien Ordinary Shares on the Record Date ("Qualifying Covidien Shareholders"), effected by (i) the transfer of Covidien's entire legal and beneficial interest in the issued share capital of the Mallinckrodt Holding Companies to Mallinckrodt; and (ii) Mallinckrodt issuing Mallinckrodt Ordinary Shares directly to Qualifying Covidien Shareholders on a pro rata basis in return, as more fully described in the Distribution Agreement (the "Distribution");

WHEREAS, the Parties intend that the Distribution will qualify as a non-taxable transaction pursuant to Section 355 of the Code; and

WHEREAS, the Parties desire to set forth their rights and obligations with respect to Taxes due for periods before and after the Distribution Date.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto agree as follows:

# ARTICLE I

# **DEFINITIONS**

1.01 GENERAL. As used in this Agreement, the following terms shall have the following meanings:

"2007 TCE TSA" shall mean the Tax Sharing Agreement entered into as of June 29th, 2007 by and among Tyco International Ltd., Covidien, and Electronics Ltd.

"2007 TCE TSA Tax Contest" shall mean any Tax Contest which is subject to the provisions of, or the administration and settlement of which is otherwise governed by or described in, the 2007 TCE TSA.

"Active Business" shall mean, with respect to the Distribution, the business conducted by the relevant ATOB Entities as of the Distribution Date, or, with respect to any other relevant transaction described in the Plan of Reorganization, the business conducted by the relevant ATOB Entities as of the date of such transaction.

"Adjustment Request" shall mean any formal or informal claim or request filed with any Governmental Entity, or with any administrative agency or court, for the adjustment, refund, or credit of Taxes, including (i) any amended Tax Return claiming adjustment to the Taxes as reported on the Tax Return or, if applicable, as previously adjusted, (ii) any claim for equitable recoupment or other offset, and (iii) any claim for refund or credit of Taxes previously paid.

"Affiliate" shall have the meaning set forth in the Distribution Agreement.

"Agreement" shall have the meaning set forth in the preamble hereto.

"Ancillary Agreement" shall have the meaning set forth in the Distribution Agreement.

"ATOB Entities" shall mean the Section 355 ATOB Entities, the Belgium ATOB Entities, the Poland ATOB Entities, and the Spain ATOB Entities.

"Audit Management Party" shall have the meaning set forth in the 2007 TCE TSA.

"Belgium ATOB Entities" shall mean the entities listed on Schedule A.

"Belgium Restricted Transfer Entities" shall mean the entities listed on Schedule B.

"Business Day" shall mean any day except a Saturday, Sunday or a day on which a commercial bank in New York, New York or Dublin, Ireland is authorized or required to close.

"Canada Restricted Transfer Entities" shall mean the entities listed on Schedule C.

"Controlling Party" shall mean, with respect to a Tax Contest, the Party entitled to control such Tax Contest pursuant to Sections 6.02 and 6.03 of this Agreement.

"Covidien" shall have the meaning set forth in the preamble hereto.

"Covidien Businesses" shall mean the businesses and operations of the Covidien Group other than the Mallinckrodt Business.

"Covidien Controlled Tax Contests" shall have the meaning set forth in Article 6.02.

"Covidien Group" shall mean Covidien and its Affiliates, as well as any entity that becomes an Affiliate of Covidien after the date hereof, excluding any entity that is a member of the Mallinckrodt Group.

"Covidien Ordinary Shares" shall have the meaning set forth in the Distribution Agreement.

"Covidien Separation Tax" shall mean (i) any Separation Tax to the extent that the liability for such Tax does not exceed the amount listed on Schedule I with respect to such Tax and (ii) to the extent that the liability for such Separation Tax exceeds the amount listed on Schedule I with respect to such Tax, the portion of such excess for which neither Mallinckrodt nor any member of the Mallinckrodt Group is obligated under applicable Law to pay; provided, that, for all purposes of this Agreement, the Distribution Agreement and each Ancillary Agreement, Covidien shall be treated as having paid any such Tax to the extent such Tax was paid or otherwise satisfied on or prior to the Distribution Date regardless of which Party or members of its Group paid or otherwise satisfied such Tax.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"**Current Taxes**" shall mean any and all Specified Taxes, other than Separation Taxes and Unanticipated Separation Taxes, which are imposed on or with respect to a taxable period or portion thereof which (i) includes the Distribution Date, (ii) ended on or before the Distribution Date but with respect to which the date prescribed by law for the filing of the applicable Tax Return is after the Distribution Date (without taking into account any applicable extensions or any Tax Returns attributable to estimated, quarterly, or other similar payments or prepayments of Taxes), or (iii) begins after the Distribution Date.

"Dispute" shall have the meaning set forth in Article 9.01.

"Distribution" shall have the meaning set forth in the recitals.

"Distribution Agreement" shall have the meaning set forth in the preamble hereto.

"Distribution Date" shall have the meaning set forth in the Distribution Agreement.

"Employee Matters Agreement" shall mean the Employee Matters Agreement dated as of the date hereof by and among Covidien and Mallinckrodt.

"Employment Tax" shall mean those Liabilities (as defined in the Distribution Agreement) for Taxes which are allocable pursuant to the provisions of the Employee Matters Agreement.

"Final Determination" shall mean the final resolution of liability for any Tax, which resolution may be for a specific issue or adjustment or for a taxable period,

(i) by an acceptance on an IRS Form 870 or 870-AD (or any successor forms thereto), or by a comparable form or agreement pursuant to the laws of a state, local, or non-United States taxing jurisdiction, except that acceptance on an IRS Form 870 or 870-AD or comparable form or agreement shall not constitute a Final Determination to the extent that such form or agreement reserves (whether by its terms or by operation of Law) the right of the taxpayer to file a claim for refund or the right of the Taxing Authority to assert a further deficiency in respect of such issue or adjustment or for such taxable period (as the case may be);

(ii) by a decision, judgment, decree, or other order of a court of competent jurisdiction which is or has become final and unappealable;

(iii) by a closing agreement or accepted offer in compromise pursuant to Sections 7121 or 7122 of the Code, or a comparable agreement pursuant to the laws of a state, local, or non-United States jurisdiction;

(iv) by any allowance of a refund or credit in respect of an overpayment of a Tax, but only after the expiration of all periods during which such refund may be recovered (including by way of offset) or, where such periods are undefined or indefinite, in accordance with ordinary course limitation periods, by the jurisdiction imposing such Tax;

(v) by a final settlement resulting from a treaty-based competent authority determination; or

(vi) by any other final disposition, including by reason of the expiration of the applicable statute of limitations or by mutual agreement of the Parties.

"Germany Restricted Transfer Entities" shall mean the entities listed on Schedule D.

"Governmental Entity" shall mean shall mean any United States federal, state, local or non-United States court, government (or political subdivision thereof), department, commission, board, bureau, agency, official or other regulatory, administrative or governmental authority.

4

"Group" shall mean either the Mallinckrodt Group or the Covidien Group, as the context requires.

"Identified Tax Return" shall mean any Tax Return reporting or otherwise relating to, addressing, or describing any Income Tax, Separation Tax, or Unanticipated Separation Tax, whether directly or indirectly.

"Income Tax" shall mean any federal, state, local or Non-United States Tax determined by reference to income, gains, net worth, gross receipts, or any Taxes imposed in lieu of such a Tax.

"Indemnified Party" shall have the meaning set forth in Article 5.02.

"Indemnifying Party" shall have the meaning set forth in Article 5.02.

"IRS" shall mean the United States Internal Revenue Service.

"Law" shall mean any United States or non-United States federal, national, supranational, state, provincial, local or similar statute, law, ordinance, regulation, rule, code, administrative pronouncement, order, requirement or rule of law (including common law), or any tax treaty.

"Local Separation Agreement" shall mean each of the asset transfer agreements, share transfer agreements, business transfer agreements, certificates of demerger and merger and other agreements and instruments that provide for or effect the separation of the Mallinckrodt Business from the Covidien Business, as contemplated by the Plan of Reorganization.

"Mallinckrodt" shall have the meaning set forth in the preamble hereof.

"Mallinckrodt Assumed Tax Rate" shall mean the highest marginal income Tax rate, as determined in Mallinckrodt's reasonable discretion and reasonably satisfactory to Covidien, taking into account all potentially applicable Taxes (federal, state, local, and non-United States), applicable to the applicable member or members of the Mallinckrodt Group.

"Mallinckrodt Business" shall have the meaning set forth in the Distribution Agreement.

"Mallinckrodt Controlled Tax Contests" shall have the meaning set forth in Article 6.03.

"Mallinckrodt Group" shall mean (i) Mallinckrodt and its Affiliates, as determined immediately after the Distribution Date, (ii) any entity which (A) was an Affiliate of Covidien or an Affiliate of a member of the Mallinckrodt Group, (B) conducted solely or predominantly the Mallinckrodt Business, and (C) is no longer an Affiliate of Covidien as of the Distribution Date, as well as (iii) any entity that becomes an Affiliate of Mallinckrodt after the date hereof.

"Mallinckrodt Historic Tax Liability" shall mean any liability for Specified Taxes (including, for the avoidance of doubt, a liability imposed pursuant to Section 1.1502-6 of the Treasury Regulations or any other similar provision of state, local, or foreign Law) where Mallinckrodt or any member of the Mallinckrodt Group is obligated under applicable Law to (x) pay such Specified Taxes or (y) file a Tax Return with respect to such Specified Taxes, in each case other than a liability for (i) Current Taxes; (ii) Separation Taxes; (iii) Unanticipated Separation Taxes; (iv) Income Taxes imposed by the United States federal government if and to the extent (A) the entity on which such Income Tax is imposed was, during the relevant taxable period or portion thereof; and (v) Income Taxes imposed by the government of any state of the United States if and to the extent (A) the entity on which such Income Tax is imposed, or other similar group (as defined for purposes of such state's Tax law) and (B) neither Mallinckrodt nor any member of the Mallinckrodt Group was the "parent," "common parent," "principal," "named," "key," or other similar company or entity with respect to such consolidated, unitary, combined, or other similar group (as defined for purpose of such state's Tax law) during such relevant taxable period or portion thereof, a member of a "consolidated group" (as defined in Section 1.1502-1(h) of the Treasury Regulations) and (B) neither Mallinckrodt nor any member of the Mallinckrodt Group was the "common parent" of such consolidated group (as

such term is used Section 1504 of the Code and the Treasury Regulations promulgated under Section 1502 of the Code) during such relevant taxable period or portion thereof.

"Mallinckrodt Holding Companies" shall mean Mallinckrodt International Finance S.A. and Mallinckrodt Belgium BVBA.

"Mallinckrodt Ordinary Shares" shall have the meaning set forth in the Distribution Agreement.

"Mallinckrodt Refund Amount" shall mean a Refund of any Tax listed on Schedule K to the extent that the aggregate amount of such Tax which is Refunded does not exceed the amount listed on Schedule K with respect to such Tax.

"Mallinckrodt Separation Tax" shall mean any Separation Tax to the extent that the liability for such Tax is not a Covidien Separation Tax.

"Mallinckrodt Tax Liability Cap" shall mean an amount equal to two hundred million United States dollars (US\$200,000,000).

"Netherlands Restricted Transfer Entities" shall mean the entities listed on Schedule E.

"Non-Controlling Party" shall mean, with respect to a Tax Contest, the Party that is not entitled to control such Tax Contest pursuant to Articles 6.02 and 6.03 of this Agreement.

"Non-United States Taxes" shall mean all Taxes imposed by any jurisdiction other than the United States, or any political subdivision thereof.

"Other Tax" shall mean any Tax imposed by any Governmental Entity other than any (i) Income Taxes, (ii) Employment Taxes, (iii) Separation Taxes, (iv) Unanticipated Separation Taxes, and (v) any interest, penalties, additions to tax, or additional amounts in respect of (i) through (iv) inclusive.

"Parties" shall have the meaning set forth in the preamble hereto.

"Past Practices" shall have the meaning set forth in Article 3.05.

"**Person**" shall mean an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization or a governmental entity or any department, agency or political subdivision thereof, without regard to whether any entity is treated as disregarded for United States federal income Tax purposes.

"Plan of Reorganization" shall have the meaning set forth in the Distribution Agreement.

"Poland ATOB Entities" shall mean the entities listed on Schedule F.

"Post-Distribution Period" shall mean any taxable year or other taxable period beginning after the Distribution Date.

"Pre-Distribution Period" shall mean any taxable year or other taxable period that ends on or before the Distribution Date.

"Preliminary Tax Advisor" shall have the meaning set forth in Article 8.01.

"Prime Rate" shall have the meaning set forth in the Distribution Agreement.

"**Privilege**" shall mean any privilege that may be asserted pursuant to applicable law, including, any privilege arising pursuant to or relating to the attorneyclient relationship (including the attorney-client and work product privileges), the accountant-client privilege and any privilege relating to internal evaluation processes.

"Pro Forma Returns" shall have the meaning set forth in Article 5.03(c).

"Prohibited Acts" shall have the meaning set forth in Article 4.02.

"Proposed Acquisition Transaction" shall mean a transaction or series of related transactions (or any agreement, understanding, arrangement or substantial negotiations, within the meaning of Section 355(e) of the Code and the Treasury Regulations promulgated thereunder, to enter into a transaction or series of related transactions), whether such transaction is supported by Mallinckrodt management or shareholders, is a hostile acquisition, or otherwise, as a result of which Mallinckrodt (or any successor thereto) would merge or consolidate with any other Person or as a result of which any Person or any group of related Persons would (directly or indirectly) acquire, or have the right to acquire (through an option or otherwise) from Mallinckrodt (or any successor thereto) and/or one or more holders of Mallinckrodt Ordinary Shares, respectively, any amount of stock of Mallinckrodt, that would, when combined with any other changes in ownership of the stock of Mallinckrodt pertinent for purposes of Section 355(e) of the Code and the Treasury Regulations promulgated thereunder, comprise more than thirty-five percent (35%) of (i) the value of all outstanding shares of Mallinckrodt as of the date of such transaction, or in the case of a series of transactions, the date of the last transaction of such series, or (ii) the total combined voting power of all shares of voting stock of Mallinckrodt as of the date of the such transaction, or in the case of a series of transactions, the date of the last transaction of such series. Notwithstanding the foregoing, a Proposed Acquisition Transaction shall not include (i) the adoption by Mallinckrodt of a shareholder rights plan or (ii) issuances by Mallinckrodt that satisfy Safe Harbor VIII (relating to acquisition in connection with a person's performance of services) or Safe Harbor IX (relating to acquisitions by a retirement plan of an employer) of Treasury Regulation Section 1.355-7(d). For purposes of determining whether a transaction constitutes an indirect acquisition, any recapitalization resulting in a shift of voting power or any redemption of shares of stock (including any redemption of Mallinckrodt equity pursuant to the exception in Article 4.02(a)(viii)) shall be treated as an indirect acquisition of stock by the nonexchanging shareholders. This definition and the application thereof is intended to monitor compliance with Section 355(e) of the Code and the Treasury Regulations promulgated thereunder and shall be interpreted accordingly.

"Qualifying Covidien Shareholders" shall have the meaning set forth in the recitals.

"**Reasonable Basis**" shall mean reasonable basis within the meaning of Section 6662(d)(2)(B)(ii)(II) of the Code and the Treasury Regulations promulgated thereunder (or such other level of confidence required by the Code at that time to avoid the imposition of penalties).

"Record Date" shall have the meaning set forth in the Distribution Agreement.

"**Refund**" shall mean any refund, reimbursement, offset, credit, or other similar benefit in respect of Taxes (including any overpayment of Taxes that can be refunded or, alternatively, applied against future Taxes payable) together with any interest paid on or with respect to such refund of Taxes; provided, however, that the amount of any refund of Taxes shall be net of any Taxes imposed by any Taxing Authority on the receipt of the refund, including any Taxes imposed by way of withholding or offset.

"Reorganization" shall have the meaning set forth in the recitals.

"Responsible Party" shall mean, with respect to any Tax Return, the Party having responsibility for preparing and filing such Tax Return pursuant to this Agreement.

# "Restricted Period" shall mean:

(i) with respect to the Section 355 Entities, the Section 355 ATOB Entities, the Canada Restricted Transfer Entities, and the Poland ATOB Entities, the period which begins with the Distribution Date and ends two (2) years thereafter;

(ii) with respect to the Belgium Restricted Transfer Entities, the Belgium ATOB Entities, and the Netherlands Restricted Transfer Entities, the period which begins with the Distribution Date and ends three (3) years thereafter; and

(iii) with respect to the Germany Restricted Transfer Entities and the Spain ATOB Transfer Entities, the period which begins with the Distribution Date and ends five (5) years thereafter.

"**Restricted Transfer Entities**" shall mean the Section 355 Entities, Belgium Restricted Transfer Entities, the Canada Restricted Transfer Entities, and the Netherlands Restricted Transfer Entities.

"**Ruling**" shall mean (i) the private letter ruling issued by the IRS to Covidien on March 1, 2013, (ii) the Canadian ruling issued by the Canada Revenue Agency to Covidien on April 25, 2013, (iii) the Irish rulings issued by Tom Connor of the Irish Revenue Commissioners to Arthur Cox, acting on behalf of Covidien, on March 15, 2013 and to PricewaterhouseCoopers LLP, acting on behalf of

Covidien, on March 12, 2013 and April 18, 2013, or (iv) any other ruling issued by a Taxing Authority, in each case in connection with the Reorganization and/or Distribution.

"**Ruling Request**" shall mean (i) the request for rulings submitted by Covidien to the IRS on August 30, 2012, as supplemented and amended on each of December 12, 2012; January 23, 2013; and February 28, 2013, (ii) the request for rulings submitted by Covidien to the Canada Revenue Agency on August 21, 2012, as supplemented and amended, (iii) the request for rulings submitted by Arthur Cox, acting on behalf of Covidien, to the Irish Revenue Commissioners on November 16, 2012, and by PricewaterhouseCoopers LLP, acting on behalf of Covidien, to the Irish Revenue Commissioners on November 23, 2012, or (iv) any other ruling request submitted to a Taxing Authority, in each case including the exhibits attached thereto, and all related supplements.

"Section 355 ATOB Entities" shall mean the entities listed on Schedule G.

"Section 355 Entities" shall mean the entities listed on Schedule H.

"Separation Taxes" shall mean those Taxes listed on Schedule I, in each case, without regard to (x) the amounts shown on Schedule I and (y) whether such Taxes arose, resulted or were incurred, or were paid or otherwise satisfied, prior to, on, or after the Distribution Date, arising as a result of (A) the Distribution or (B) the Reorganization or any transaction associated therewith as described in any Ruling, the Distribution Agreement, or any Local Separation Agreement, except for (i) any Tax resulting from a breach by any Party of any covenant in this Agreement or any Ancillary Agreement, and (ii) any Tax attributable to a Prohibited Act.

"Spain ATOB Entities" shall mean the entities listed on Schedule J.

"Specified Taxes" shall mean all Taxes other than Employment Taxes and Other Taxes.

"Straddle Period" shall mean any taxable year or other taxable period that begins on or before the Distribution Date and ends after the Distribution Date.

"Subsidiary" shall have the meaning set forth in the Distribution Agreement.

"**Tax**" or "**Taxes**" shall mean (i) all taxes, charges, fees, duties, levies, imposts, rates or other assessments or governmental charges of any kind imposed by any federal, state, local or non-United States Governmental Entity, including, without limitation, income, gross receipts, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, custom duties, property, sales, use, license, capital stock, transfer, franchise, registration, payroll, withholding, social security, unemployment, disability, value added, alternative or add-on minimum or other taxes, whether disputed or not, and including any interest, penalties, charges or additions attributable thereto, (ii) liability for the

payment of any amount of the type described in clause (i) above arising as a result of being (or having been) a member of any group or being (or having been) included or required to be included in any Tax Return related thereto, and (iii) liability for the payment of any amount of the type described in clauses (i) or (ii) above as a result of any express or implied obligation to indemnify or otherwise assume or succeed to the liability of any other Person.

"Tax Advisor" shall have the meaning set forth in Article 9.01.

"**Tax Attribute**" shall mean net operating losses, capital losses, investment tax credit carryovers, earnings and profits, foreign tax credit carryovers, overall foreign losses, previously taxed income, separate limitation losses and any other losses, deductions, credits or other comparable items that could affect a Tax liability for a past or future taxable period.

"**Tax Benefit Amount**" shall mean, with respect to the payment of a liability by a Party or any of its Subsidiaries, an amount equal to the reduction in Taxes due and payable during the Tax Benefit Period resulting from the payment of such liability, as determined at any relevant time (including, for the avoidance of doubt, at any time during the Tax Benefit Period), which for each taxable period during the Tax Benefit Period, shall equal the sum of:

(a) the excess (if any) of (i) the amount of Taxes that the Party and its Subsidiaries would have owed in such taxable period had there been no payment of or event giving rise to such liability (without taking into account any carryforwards or carrybacks of any deductions, credits, losses or other Tax Attributes to such taxable period), over (ii) the amount of Taxes that the Party and its Subsidiaries would have owed in such taxable period after taking into account such payment (without taking into account any carryforwards or carrybacks of any deductions, credits, losses or other Tax Attributes to such taxable period other than such deductions, credits, losses, or other Tax Attributes, if any, arising as a result of the payment of such liability); and

(b) the excess (if any) of (i) the amount of the Refund that would be realized by the Party and its Subsidiaries with respect to such taxable period as a result of the carryback of deductions, credits, losses or other Tax Attributes attributable to such payment to such taxable period (without taking into account any other carryforwards or carrybacks of any deductions, credits, losses or other Tax Attributes to such taxable period), over (ii) the amount of the Refund that the Party and its Subsidiaries would have been entitled to realize with respect to such taxable period (without taking into account any carryforwards or carrybacks of any deductions, credits, losses or other Tax Attributes to such taxable period (without taking into account any carryforwards or carrybacks of any deductions, credits, losses or other Tax Attributes to such taxable period (without taking into account any carryforwards or carrybacks of any deductions, credits, losses or other Tax Attributes to such taxable period).

The Tax Benefit Amount shall be computed based on the actual tax rates applicable to the Party and its Subsidiaries during the applicable taxable period.

For the avoidance of doubt, for purposes of this Agreement, the Distribution Agreement and each Ancillary Agreement, the Tax Benefit Amount shall be determined taking into account the application of Article 5.03(d).

"**Tax Benefit Adjusted Amount**" shall mean, with respect to a liability, the amount equal to the amount of such liability reduced by the Tax Benefit Amount, if any, with respect to Mallinckrodt and its Subsidiaries, in respect of the payment of such liability.

"**Tax Benefit Period**" shall mean, with respect to the payment of a liability, the sequential series of taxable periods beginning with the taxable period that includes the date which is one (1) year prior to the date on which such liability is considered to have been paid or satisfied for applicable Tax purposes and ending with the taxable period that includes the date which is five (5) years after the date on which such liability is considered to have been paid or satisfied for applicable Tax purposes.

"Tax Certificates" shall mean any certificates of officers of Covidien and Mallinckrodt, provided to Skadden, Arps, Slate, Meagher & Flom LLP, PricewaterhouseCoopers LLP, Arthur Cox, or any other law or accounting firm in connection with any Tax Opinion issued in connection with the Reorganization or Distribution.

"Tax Contest" shall have the meaning set forth in Article 6.01.

"Tax Counsel" shall mean a tax counsel or accountant of recognized national standing reasonably acceptable to Covidien.

"Taxing Authority" shall mean any Governmental Entity having jurisdiction over the assessment, determination, collection or imposition of any Tax.

"Tax Law" shall mean the law of any governmental entity or political subdivision thereof relating to any Tax.

"Tax Materials" shall have the meaning set forth in Article 4.01(a).

"Tax Opinion" shall mean any written opinion of Skadden, Arps, Slate, Meagher & Flom LLP, PricewaterhouseCoopers LLP, Arthur Cox, or any other law or accounting firm, regarding certain tax consequences of certain transactions executed as part of the Reorganization and the Distribution.

"Tax Records" shall have the meaning set forth in Article 8.01.

"**Tax-Related Losses**" shall mean (i) all accounting, legal and other professional fees, and court costs incurred in connection with such Taxes, as well as any other out-of-pocket costs incurred in connection with such Taxes; and (ii) all costs, expenses and damages associated with stockholder litigation or controversies and any amount paid by Covidien (or its Affiliate) or Mallinckrodt (or its Affiliate) in respect of the liability of shareholders, whether paid to shareholders or to the IRS or any other Taxing Authority, in each case, resulting from the failure of the Distribution, the Reorganization or any transaction associated therewith to be tax-free or otherwise have the tax treatment described in any Tax Opinion or Ruling.

"**Tax Return**" shall mean any return, report, certificate, form or similar statement or document (including any related supporting information or schedule attached thereto and any information return, amended tax return, claim for refund or declaration of estimated tax) supplied to or filed with, or required to be supplied to or filed with, a Governmental Entity, or any bill for or notice related to ad valorem or other similar Taxes received from a Governmental Entity, in each case, in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.

"Treasury Regulations" shall mean the regulations promulgated from time to time under the Code as in effect for the relevant tax period.

"Unanticipated Separation Taxes" shall mean Taxes arising as a result of (i) the Distribution, or (ii) the Reorganization or any transaction associated therewith as described in any Ruling, the Distribution Agreement, or any Local Separation Agreement, in each case where such Tax is not a Separation Tax, except for (A) any Tax resulting from a breach by any Party of any covenant in this Agreement or any Ancillary Agreement, and (B) any Tax attributable to a Prohibited Act.

"Unqualified Tax Opinion" shall mean an unqualified "will" opinion of Tax Counsel on which Mallinckrodt and Covidien may rely to the effect that the Prohibited Act will not result in Unanticipated Separation Taxes or any incremental liability for Separation Taxes. Any such opinion must assume that the Distribution, Reorganization, and any transaction associated therewith would have been tax-free or had the tax treatment described in any applicable Tax Opinion or Ruling if such transaction did not occur.

"US GAAP" means United States generally accepted accounting principles.

1.02 INTERPRETATION. For all purposes of this Agreement: (i) the terms defined in this Agreement include the plural as well as the singular; (ii) all references in this Agreement to "Preamble", "Recitals", "Articles", "Sections" and other subdivisions are to the designated Preamble, Recitals, Articles, Sections and other subdivisions of the body of this Agreement; (iii) pronouns of either gender or neuter include, as appropriate, the other pronoun forms; (iv) the words "herein," "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not

to any particular Article, Section or other subdivision; (v) "or" is not exclusive; (vi) "including" shall be deemed to be followed by "but not limited to"; and (vii) any definition of or reference to any statute shall be construed as referring also to any rules and regulations promulgated thereunder.

# ARTICLE II

# PAYMENTS AND TAX REFUNDS

2.01 COVIDIEN LIABILITY. Covidien shall pay and be responsible for:

(a) any Specified Taxes not allocated to Mallinckrodt pursuant to Article 2.02; and

(b) any Covidien Separation Taxes.

2.02 MALLINCKRODT LIABILITY. Mallinckrodt shall pay and be responsible for:

(a) any Current Taxes where Mallinckrodt or any member of the Mallinckrodt Group is obligated under applicable Law to (i) pay such Taxes or (ii) file a Tax Return with respect to such Taxes;

(b) any Mallinckrodt Historic Tax Liability;

(c) 20% of any Unanticipated Separation Taxes; and

(d) any Mallinckrodt Separation Taxes.

# 2.03 MALLINCKRODT TAX LIABILITY CAP.

(a) Notwithstanding anything in this Agreement to the contrary, Mallinckrodt shall not be liable for Taxes pursuant to Article 2.02(b), 2.02(c), or 2.02(d) to the extent that the aggregate Tax Benefit Adjusted Amount in respect of liability for Taxes allocated to Mallinckrodt pursuant to Article 2.02(b), together with the aggregate Tax Benefit Adjusted Amount in respect of liability for Taxes allocated to Mallinckrodt pursuant to Article 2.02(c) and 2.02(d), exceeds the Mallinckrodt Tax Liability Cap.

(b) If and to the extent the Tax Benefit Adjusted Amount in respect of any liability for Mallinckrodt Historic Tax Liabilities, Unanticipated Separation Taxes, or Mallinckrodt Separation Taxes changes subsequent to any determination made with respect to Article 2.01 or 2.02 including, without limitation, as a result of a carryforward or a carryback of a deduction, credit, loss or other Tax attribute which affects the Tax Benefit Adjusted Amount in respect of such a liability, then the amount of all Taxes for which each Party is responsible pursuant to Articles 2.01 and 2.02 shall be redetermined taking into account the effect of such modified Tax Benefit Adjusted Amount.

(c) If and to the extent Mallinckrodt or any member of the Mallinckrodt Group receives any Refund of Unanticipated Separation Taxes or Mallinckrodt Separation Taxes pursuant to Article 2.05, then the amount of all Taxes for which each Party is responsible pursuant to Articles 2.01 and 2.02 shall be redetermined taking into account the effect of such Refund.

# 2.04 CERTAIN EMPLOYMENT AND OTHER TAXES.

(a) ALLOCATION OF EMPLOYMENT TAXES. Except as otherwise addressed in Article V hereof, liability for Employment Taxes shall be determined pursuant to the Employee Matters Agreement.

(b) ALLOCATION OF OTHER TAXES. Except as otherwise addressed in Article V hereof, liability for Other Taxes shall be determined pursuant to the Distribution Agreement.

#### 2.05 TAX REFUNDS.

(a) Mallinckrodt shall be entitled to

(i) any Refund of Current Taxes received by any member of the Mallinckrodt Group or the Covidien Group, where Mallinckrodt or any member of the Mallinckrodt Group is obligated under applicable Law to (i) pay such Taxes or (ii) file a Tax Return with respect to such Taxes;

(ii) any Mallinckrodt Refund Amounts;

(iii) 20% of any Refund of any Unanticipated Separation Taxes received by any member of the Mallinckrodt Group or the Covidien Group; <u>provided</u> that, if the proportionate amount of such Tax that was borne by Mallinckrodt was lower than 20%, whether by reason of the application of Article 2.03 or otherwise, Mallinckrodt will be entitled to such lower proportionate amount of such Refund of Unanticipated Separation Taxes; and

(iv) any Refund of Mallinckrodt Separation Taxes received by any member of the Mallinckrodt Group or the Covidien Group to the extent that liability for such Taxes was actually borne by Mallinckrodt; <u>provided</u> that, solely for purposes of this Article 2.05(a)(iii), any Refund in respect of a Separation Tax shall be treated first as a Refund of a Mallinckrodt Separation Tax if and only to the extent Mallinckrodt bore such Tax, taking into account the application of Article 2.03, and thereafter as a Refund of a Covidien Separation Tax.

(b) Covidien shall be entitled to all Refunds related to Specified Taxes received by any member of the Mallinckrodt Group or the Covidien Group other than those to which Mallinckrodt is entitled pursuant to Article 2.05(a).

(c) Each Party shall pay to the other Party any Refund received by such Party or any member of such Party's Group that is allocable to the other Party pursuant to this Article 2.05 no later than five (5) Business Days after the receipt of such Refund. For purposes of this Article 2.05(c), any Refund that arises as a result of an offset, credit, or other similar benefit in respect of Taxes other than a receipt of cash shall be deemed to be received on the earlier of (i) the date on which a Tax Return is filed claiming such offset, credit, or other similar benefit and (ii) the date on which payment of the Tax which would have otherwise been paid absent such offset, credit, or other similar benefit is due (determined without taking into account any applicable extensions).

2.06 PRIOR AGREEMENTS. Except as set forth in this Agreement and in consideration of the mutual indemnities and other obligations of this Agreement, any and all prior Tax sharing or allocation agreements or practices between any member of the Covidien Group and any member of the Mallinckrodt Group shall be terminated with respect to the Mallinckrodt Group and the Covidien Group as of the Distribution Date. No member of either the Mallinckrodt Group or the Covidien Group shall have any continuing rights or obligations under any such agreement thereunder.

# ARTICLE III

### **PREPARATION AND FILING OF TAX RETURNS**

3.01 COVIDIEN'S RESPONSIBILITY. Covidien shall prepare and file when due (taking into account any applicable extensions), or shall cause to be prepared and filed, all Tax Returns Covidien or any member of the Covidien Group is obligated to file pursuant to applicable Tax Law.

3.02 MALLINCKRODT'S RESPONSIBILITY. Mallinckrodt shall prepare and file when due (taking into account any applicable extensions), or shall cause to be prepared and filed, all Tax Returns Mallinckrodt or any member of the Mallinckrodt Group is obligated to file pursuant to applicable Tax Law, other than those which Covidien is responsible to file pursuant to Article 3.01.

3.03 RIGHT TO REVIEW TAX RETURNS. With respect to any Identified Tax Return relating to any Pre-Distribution Period or Straddle Period for which Mallinckrodt is the Responsible Party, Mallinckrodt shall deliver such Identified Tax Return and related workpapers to Covidien for approval twenty (20) Business Days prior to the due date of the relevant Identified Tax Return. Mallinckrodt shall provide Covidien no less than ten (10) Business Days to analyze and comment on such Identified Tax Return and shall modify such Identified Tax Return before filing to include Covidien's reasonable comments. Mallinckrodt shall not, and shall not permit any member of the Mallinckrodt Group to, file any such Identified Tax Return without the prior written consent of Covidien, such consent to be exercised in Covidien's sole discretion.

3.04 COOPERATION. The Parties shall provide, and shall cause their Affiliates to provide, assistance and cooperation to one another in accordance with Article VII with respect to the preparation and filing of Tax Returns, including providing information required to be provided in Article VIII.

3.05 TAX REPORTING PRACTICES. Except as provided in Article 3.06 or pursuant to the prior written consent of Covidien, such consent to be exercised in Covidien's sole discretion, with respect to any Tax Return for any taxable period that begins on or before the second anniversary of the Distribution Date with respect to which Mallinckrodt is the Responsible Party, such Tax Return shall be prepared in a manner (i) consistent with past practices, accounting methods, elections and conventions ("**Past Practices**") used with respect to the Tax Returns in question (unless there is no Reasonable Basis for the use of such Past Practices), and to the extent any items are not covered by Past Practices (or in the event that there is no Reasonable Basis for the use of such Past Practices), in

accordance with reasonable Tax accounting practices selected by Mallinckrodt; and (ii) that, to the extent consistent with clause (i), minimizes the overall amount of Taxes due and payable on such Tax Return for all of the Parties by cooperating in making such elections or applications for group or other relief or allowances available in the taxing jurisdiction in which such Tax Return is filed. Mallinckrodt shall not take any action inconsistent with the assumptions (including items of income, gain, deduction, loss and credit) made in determining all estimated or advance payments of Taxes on or prior to the Distribution Date. In addition, Mallinckrodt shall not be permitted, and shall not permit any member of the Mallinckrodt Group, to make a change in any of its methods of accounting for tax purposes until all applicable statutes of limitations for all Pre-Distribution Periods and Straddle Periods have expired.

3.06 REPORTING OF REORGANIZATION. The Tax treatment of any step in or portion of the Reorganization shall be reported on each applicable Tax Return consistently with the treatment thereof in any Ruling Request, Tax Opinion, Ruling, and Local Separation Agreement, taking into account the jurisdiction in which such Tax Returns are filed, unless there is no Reasonable Basis for such Tax treatment. In the event that a Party shall determine that there is no Reasonable Basis for such Tax treatment, such Party shall notify the other Party no later than twenty (20) Business Days prior to filing the relevant Tax Return and the Parties shall attempt in good faith to agree on the manner in which the relevant portion of the Reorganization shall be reported.

### 3.07 PAYMENT OF TAXES.

(a) With respect to any Tax Return required to be filed pursuant to this Agreement, the Responsible Party shall remit or cause to be remitted to the applicable Governmental Entity in a timely manner any Taxes due in respect of any such Tax Return.

(b) In the case of any Tax Return for which the Party that is not the Responsible Party is obligated pursuant to this Agreement to pay all or a portion of the Taxes reported as due on such Tax Return, the Responsible Party shall notify the other Party, in writing, of its obligation to pay such Taxes and, in reasonably sufficient detail, its calculation of the amount due by such other Party and the Party receiving such notice shall pay such amount to the Responsible Party upon the later of five (5) Business Days prior to the date on which such payment is due and fifteen (15) Business Days after the receipt of such notice.

# 3.08 AMENDED RETURNS AND CARRYBACKS.

(a) Mallinckrodt shall not, and shall not permit any member of the Mallinckrodt Group to, file or allow to be filed any Adjustment Request for any Pre-Distribution Period or Straddle Period without the prior written consent of Covidien, such consent to be exercised in Covidien's sole discretion.

(b) Mallinckrodt shall, and shall cause each member of the Mallinckrodt Group to, make any available elections to waive the right to carry back any Tax Attribute from a taxable period or portion thereof ending after the Distribution Date to a taxable period or portion thereof ending on or before the Distribution Date.

(c) Mallinckrodt shall not, and shall cause each member of the Mallinckrodt Group not to, without the prior written consent of Covidien, make any affirmative election to carry back any Tax Attribute from a taxable period or portion thereof ending after the Distribution Date to a taxable period or portion thereof ending on or before the Distribution Date, such consent to be exercised in Covidien's sole discretion.

(d) Receipt of consent by Mallinckrodt or a member of the Mallinckrodt Group from Covidien pursuant to the provisions of this Article 3.08 shall not limit or modify Mallinckrodt's continuing indemnification obligation pursuant to Article V.

# 3.09 TAX ATTRIBUTES.

(a) Mallinckrodt shall make its own determination as to the existence and the amount of the Tax Attributes to which it is entitled after the Distribution Date; provided, however, that such determination shall be made in a manner that is (a) consistent with Past Practices; (b) in accordance with the rules prescribed by applicable Law, including the Code and the Treasury Regulations; (c) consistent with the Rulings, the Tax Certificates, and the Tax Opinions; (d) reasonably determined by Mallinckrodt to minimize the aggregate cash Tax liability of the Parties for all Pre-Distribution Tax Periods and the portion of all Straddle Tax Periods ending on the Distribution Date; and (e) with respect to any determination relating to the existence or availability of net operating losses, consented to in writing by Covidien, such consent to be exercised in Covidien's sole and absolute discretion.

(b) Upon the reasonable request of Mallinckrodt, Covidien shall provide Mallinckrodt with any reasonably available Tax Records relating to the determination of Tax Attributes if and only to the extent such Tax Records exist on the Distribution Date. Nothing in this Agreement, including this Article 3.09(b), shall require Covidien to make any determinations or otherwise create any Tax Records with respect to Tax Attributes or the determination thereof.

### ARTICLE IV

## **REPRESENTATIONS AND COVENANTS**

# 4.01 COMPLIANCE WITH THE RULINGS AND TAX OPINIONS.

(a) Covidien, on behalf of itself and all other members of the Covidien Group, hereby represents and warrants that (i) it has examined (A) the Rulings, (B) the Tax Opinions, (C) the Ruling Requests, (D) the Tax Certificates and (E) any other materials delivered or deliverable in connection with the issuance of the Rulings and the rendering of Tax Opinions (collectively, the "**Tax Materials**") and (ii) the facts presented and representations made therein, to the extent descriptive of or otherwise relating to Covidien or any member of the Covidien Group or the Covidien Businesses, were, at the time presented or represented and from such time until and including the Distribution

Date, true, correct, and complete in all material respects. Covidien, on behalf of itself and all other members of the Covidien Group, hereby confirms and agrees to comply with any and all covenants and agreements in the Tax Materials applicable to Covidien or any member of the Covidien Group or the Covidien Businesses.

(b) Mallinckrodt, on behalf of itself and all other members of the Mallinckrodt Group, hereby represents and warrants that (i) it has examined the Tax Materials and (ii) the facts presented and representations made therein, to the extent descriptive of or otherwise relating to Mallinckrodt or any member of the Mallinckrodt Group or the Mallinckrodt Business, were, at the time presented or represented and from such time until and including the Distribution Date, true, correct, and complete in all material respects. Mallinckrodt, on behalf of itself and all other members of the Mallinckrodt Group, hereby confirms and agrees to comply with any and all covenants and agreements in the Tax Materials applicable to Mallinckrodt or any member of the Mallinckrodt Group or the Mallinckrodt Business.

4.02 CONSENT REQUIREMENT FOR MAJOR TRANSACTIONS. Mallinckrodt, on behalf of itself and all other members of the Mallinckrodt Group, hereby covenants and agrees that no member of the Mallinckrodt Group will take or permit to be taken:

(a) within the applicable Restricted Period, any of the following actions:

(i) any Proposed Acquisition Transaction, or approval of any Proposed Acquisition Transaction for any purpose;

(ii) any merger, scheme of arrangement, or consolidation with any other Person or liquidation or partial liquidation; or any approval or allowance of any merger, scheme of arrangement, consolidation, liquidation, or partial liquidation of any of the Restricted Transfer Entities or the ATOB Entities;

(iii) any approval or allowance of the discontinuance, cessation, or sale or other transfer (to an Affiliate or otherwise) of, or a material change in, any Active Business;

(iv) any approval or allowance of the sale, transfer, issuance, or other disposition (to an Affiliate or otherwise), directly or indirectly, of any share of, or other equity interest or an instrument convertible into an equity interest in, any of the ATOB Entities, Belgium Restricted Transfer Entities, Canada Restricted Transfer Entities, German Restricted Transfer Entities, or Netherlands Restricted Transfer Entities;

(v) any sale, transfer, or other disposition of more than 35 percent (35%) of its consolidated gross or net assets, or approval or allowance of the sale, transfer, or other disposition (to an Affiliate or

otherwise) of more than 35 percent (35%) of the consolidated gross or net assets of any of the Restricted Transfer Entities (in each case, excluding sales in the ordinary course of business and measured based on fair market values as of the date of the applicable Distribution or other transaction);

(vi) any amendment to its certificate of incorporation (or other organizational documents), or any other action or approval or allowance of the taking of any action, whether through a stockholder vote or otherwise, affecting the voting rights of the stock of Mallinckrodt or any of the Restricted Transfer Entities;

(vii) any issuance of shares of a new class of nonvoting stock or approval or allowance of any of the Restricted Transfer Entities to issue shares of a new class of nonvoting stock;

(viii) any purchase, directly or through any Affiliate, of any of its outstanding stock after the Distributions, other than through stock purchases meeting the requirements of Section 4.05(1)(b) of Revenue Procedure 96-30;

(ix) any approval or allowance of an extraordinary contribution to any of the Restricted Transfer Entities (or any successor thereto) by its shareholder or shareholders (or any successor(s) thereto); or

(x) any sale, transfer or other disposition, or any approval or allowance of the sale, transfer, or other disposition, of ten percent (10%) or more of the fair market value of the property, determined at the time of the Distribution, of any Canada Restricted Transfer Entity, as determined pursuant to Section 55(3)(a) of the Canadian Income Tax Act of 1985, as amended and any regulations promulgated thereunder; or

(b) any action at any time that could jeopardize, directly or indirectly, any of the conclusions contained in any Ruling or any Tax Opinion (collectively, the "**Prohibited Acts**").

Notwithstanding the foregoing, Mallinckrodt or a member of the Mallinckrodt Group may take any of the Prohibited Acts if Mallinckrodt either (i) obtains an Unqualified Tax Opinion in form and substance reasonably satisfactory to Covidien or (ii) obtains the prior written consent of Covidien waiving the requirement that Mallinckrodt obtain an Unqualified Tax Opinion, such waiver to be provided in Covidien's sole and absolute discretion. Covidien's evaluation of an Unqualified Tax Opinion may consider, among other factors, the appropriateness of any underlying assumptions, representations, and covenants made in connection with such opinion. Mallinckrodt shall bear all costs and expenses of securing any such Unqualified Tax Opinion and shall reimburse Covidien for all reasonable out-of-pocket expenses that Covidien or its Subsidiaries may incur in good faith in seeking to obtain or evaluate any such Unqualified Tax Opinion. Neither the delivery of an Unqualified Tax Opinion nor Covidien's waiver of Mallinckrodt's obligation to deliver an Unqualified Tax Opinion shall limit or modify Mallinckrodt's continuing indemnification obligation pursuant to Article V.

4.03 COVIDIEN COVENANTS. Notwithstanding anything else to the contrary contained in this Agreement or any other agreement, Covidien, on behalf of itself and all other members of the Covidien Group, hereby confirms and agrees that neither Covidien nor any member of the Covidien Group will take or permit to be taken any action at any time that would likely jeopardize, directly or indirectly, any of the conclusions contained in any Ruling or Tax Opinion.

# ARTICLE V

# **INDEMNITY OBLIGATIONS**

#### 5.01 INDEMNITY OBLIGATIONS.

(a) Covidien shall indemnify and hold harmless Mallinckrodt from and against, and will reimburse Mallinckrodt for, (i) all liability for Taxes allocated to Covidien pursuant to Article II, (ii) all Taxes and Tax-Related Losses arising out of, based upon, or relating or attributable to any breach of or inaccuracy in any representation, covenant, or obligation of any member of the Covidien Group pursuant to this Agreement, and (iii) the amount of any Refund received by any member of the Covidien Group which is allocated to Mallinckrodt pursuant to Article 2.05(a).

(b) Without regard to whether any action is permitted or consented to hereunder and notwithstanding anything else to the contrary contained herein, Mallinckrodt shall indemnify and hold harmless Covidien from and against, and will reimburse Covidien for, (i) all liability for Taxes allocated to Mallinckrodt pursuant to Article II, (ii) all Taxes and Tax-Related Losses arising out of, based upon, or relating or attributable to any breach of or inaccuracy in any representation, covenant, or obligation of any member of the Mallinckrodt Group pursuant to this Agreement, (iii) all Taxes and Tax-Related Losses arising out of, based upon, or relating or attributable to any Prohibited Act by Mallinckrodt or any member of the Mallinckrodt Group, regardless of whether (A) Covidien consented to such Prohibited Act, or (B) Mallinckrodt obtained an Unqualified Tax Opinion, and (iv) the amount of any Refund received by any member of the Mallinckrodt Group which is allocated to Covidien pursuant to Article 2.05(b).

(c) To the extent that any Tax or Tax-Related Loss is subject to indemnity pursuant to both Articles 5.01(a) and 5.01(b), responsibility for such Tax or Tax-Related Loss shall be shared by Covidien and Mallinckrodt according to relative fault.

#### 5.02 INDEMNIFICATION PAYMENTS.

(a) Except as otherwise provided in this Agreement, if either Party (the "Indemnified Party") is required to pay to a Taxing Authority a Tax or to another Party an indemnification payment in respect of a Tax that another Party (the "Indemnifying Party") is liable for under this Agreement, including as the result of a Final Determination, the Indemnified Party shall notify the

Indemnifying Party, in writing, of its obligation to pay such Taxes and, in reasonably sufficient detail, its calculation of the amount due by such Indemnifying Party to the Indemnified Party, including any Tax-Related Losses attributable thereto. The Indemnifying Party shall pay such amount, including any Tax-Related Losses attributable thereto, to the Indemnified Party no later than the later of (i) five (5) Business Days prior to the date on which such payment is due to the applicable Taxing Authority or (ii) fifteen (15) Business Days after the receipt of notice from the other Party.

(b) If, as a result of any change or redetermination made with respect to Article 2.01 or 2.02, including, without limitation, pursuant to Article 2.03(b), any amount previously allocated to and borne by one Party pursuant to the provisions of Article II is thereafter allocated to the other Party, then, no later than five (5) Business Days after such change or redetermination, such other Party shall pay to such Party the amount previously borne by such Party which is allocated to such other Party as a result of such change or redetermination.

### 5.03 PAYMENTS NET OF TAX BENEFITS.

(a) All amounts required to be paid by one Party to another pursuant to this Agreement, the Distribution Agreement or any Ancillary Agreement shall be reduced by the Tax Benefit Amount, if any, with respect to the Indemnified Party or its Subsidiaries in respect of the indemnified liability.

(b) If and to the extent any Tax Benefit Amount arises in respect of an indemnified liability which has not already reduced any payment made to the Indemnified Party or its Subsidiaries pursuant to this Agreement or otherwise been paid to the Indemnifying party then, no later than five (5) Business Days after the filing of a Tax Return with respect to the applicable taxable period in which such Tax Benefit Amount arose, such Indemnified Party shall pay to the Indemnifying Party the amount of any such Tax Benefit Amount not previously taken into account. For the avoidance of doubt, in the event that a deduction, credit, loss or other Tax attribute, or any portion thereof, arising as a result of such payment results in a Tax Benefit Amount in a taxable period or portion thereof prior to or later than the taxable period during which such payment is considered to have been made for applicable Tax purposes, then this Article 5.03 shall continue to apply for all taxable periods included in the applicable Tax Benefit Period.

(c) With respect to each taxable period that is included in a Tax Benefit Period in respect of

(i) any liability for which Mallinckrodt or any member of the Mallinckrodt Group is required to indemnify Covidien or any member of the Covidien Group pursuant to this Agreement, the Distribution Agreement, or any Ancillary Agreement, or

(ii) any liability for a Tax for which Mallinckrodt or any member of the Mallinckrodt Group is liable pursuant to Articles 2.02(b), 2.02(c), or 2.02(d) of this Agreement (in each case, without regard to Article 2.03),

Mallinckrodt and each member of the Mallinckrodt Group shall, with respect to each Tax Return required to be filed, or actually filed, with respect to each such taxable period, provide to Covidien pro forma copies (the "**Pro Forma Returns**") of each such Tax Return prepared in accordance with the principles set forth in the definition of Tax Benefit Amount. Such pro forma Tax Return shall determine the Tax Benefit Amount, if any, realized by Mallinckrodt or any member of the Mallinckrodt Group in respect of such liability or Tax. Mallinckrodt

shall deliver such pro forma Tax Returns to Covidien no later than forty-five (45) calendar days following the earlier of the date on which such Tax Return was filed or required to be filed.

(d) If Mallinckrodt fails to provide to Covidien any Pro Forma Return with respect to a liability for which it is required to provide such a return pursuant to Article 5.03(c), then, notwithstanding anything else to the contrary in this Agreement, including the definition of Tax Benefit Amount, the Tax Benefit Amount in respect of such liability shall, if and to the extent such liability is or may potentially be deductible, creditable, or otherwise potentially available to offset or reduce any amount of Taxes in any jurisdiction (as determined in Covidien's reasonable discretion), for all purposes of this Agreement and each Ancillary Agreement be equal to the product of (a) the absolute value of the amount of such liability and (b) the Mallinckrodt Assumed Tax Rate.

#### 5.04 PAYMENT MECHANICS.

(a) Subject to Article 10.02, all payments under this Agreement shall be made by Covidien directly to Mallinckrodt and by Mallinckrodt directly to Covidien; provided, however, that if the Parties mutually agree with respect to any such indemnification payment, any member of the Covidien Group, on the one hand, may make such indemnification payment to any member of the Mallinckrodt Group, on the other hand, and vice versa. All indemnification payments shall be treated in the manner described in Article 5.05.

(b) Any late payment made by one Party to another Party pursuant to this Agreement shall be subject to interest at a rate per annum equal to the then effective Prime Rate plus 5% (or the maximum legal rate, whichever is lower), calculated for the actual number of days elapsed, and accrued from the date on which such payment was due up to the date of the actual receipt of payment.

(c) In the case of any payment of Taxes made by a Responsible Party or Indemnified Party pursuant to this Agreement for which such Responsible Party or Indemnified Party, as the case may be, has received a payment from the other Party, such Responsible Party or Indemnified Party shall provide to the other Party a copy of any official government receipt received with respect to the payment of such Taxes to the applicable Taxing Authority (or, if no such official governmental receipts are available, executed bank payment forms or other reasonable evidence of payment).

5.05 TREATMENT OF PAYMENTS. The Parties agree that any payment made among the Parties pursuant to this Agreement shall be treated, to the extent permitted by law, for all United States federal income Tax purposes as either (i) a non-taxable contribution by Covidien to Mallinckrodt, or (ii) a distribution by Mallinckrodt to Covidien, in each case, made immediately prior to the Distribution.

# ARTICLE VI

# TAX CONTESTS

6.01 NOTICE. Each Party shall promptly notify the other Party in writing upon receipt by such Party or any member of its Group of a written communication from any Governmental Entity with respect to any pending or threatened audit, claim, dispute, suit, action, proposed assessment or other proceeding (a "**Tax Contest**") concerning any Taxes for which the other Party may be liable pursuant to this Agreement.

6.02 CONTROL OF CONTESTS BY COVIDIEN. Covidien shall have the sole responsibility and right to control the prosecution of any Tax Contest, including the exclusive right to communicate with agents of the applicable Governmental Entity

and to control, resolve, settle, or agree to any deficiency, claim, or adjustment proposed, asserted, or assessed in connection with or as a result of any such Tax Contest, other than Mallinckrodt Controlled Tax Contests (collectively, "Covidien Controlled Tax Contests").

6.03 CONTROL OF CONTESTS BY MALLINCKRODT. Mallinckrodt shall have the full responsibility and right to control the prosecution of any Tax Contest, including the exclusive right to communicate with agents of the applicable Governmental Entity and to control, resolve, settle, or agree to any deficiency, claim, or adjustment proposed, asserted, or assessed in connection with or as a result of any such Tax Contest, involving any Tax Return filed by Mallinckrodt or any member of the Mallinckrodt Group for any Post-Distribution Period, and any Pre-Distribution or Straddle Period Tax Return filed by Mallinckrodt or any member of the Mallinckrodt Group relating exclusively to Non-United States Taxes for any taxable year or taxable period beginning after June 29, 2007 (collectively, "**Mallinckrodt Controlled Tax Contests**"); <u>provided</u>, that in no event shall any 2007 TCE TSA Tax Contest constitute a Mallinckrodt Controlled Tax Contest; <u>provided</u>, <u>further</u>, that Mallinckrodt Shall not resolve, settle or agree to any deficiency, claim or adjustment proposed, asserted or assessed in connection with or as a result of any Mallinckrodt Controlled Tax Contest for any Pre-Distribution Period or Straddle Period without the prior written consent of Covidien, such consent to be exercised in Covidien's sole discretion.

6.04 2007 TCE TSA CONTEST INFORMATION UPDATE. Notwithstanding anything to the contrary in this Agreement, Covidien shall (i) provide to Mallinckrodt any factual information Covidien receives from the Audit Management Party regarding any 2007 TCE TSA Tax Contest relating to Taxes for which Mallinckrodt may be liable pursuant to Article 2.02, and (ii) provide notice to Mallinckrodt of any pending or threatened 2007 TCE TSA Tax Contest of which it becomes aware relating to Taxes for which Mallinckrodt may be liable pursuant to Article 2.02, and (ii) provide notice to Mallinckrodt of any pending or threatened 2007 TCE TSA Tax Contest of which it becomes aware relating to Taxes for which Mallinckrodt may be liable pursuant to Article 2.02 reasonably promptly after receipt of notice pursuant to the 2007 TCE TSA. Such notice shall contain factual information (to the extent known) describing any asserted Tax liability in reasonable detail and shall be accompanied by copies of any notice and other documents received from any Taxing Authority in respect of any such matters. Mallinckrodt shall not have any right to any amount paid to Covidien or any member of the Covidien Group pursuant to the 2007 TCE TSA regardless of whether such payment relates to an amount for which Mallinckrodt is liable pursuant to the terms of this Agreement or any Ancillary Agreement. Mallinckrodt shall take or refrain from taking, and shall cause each member of the Mallinckrodt Group to take or refrain from taking, any and all actions reasonably requested by Covidien that would preserve, exercise, or contravene, as the case may be, Covidien's rights and obligations under the 2007 TCE TSA.

6.05 OBLIGATION OF CONTINUED NOTICE. During the pendency of any Tax Contest or threatened Tax Contest, other than a 2007 TCE TSA Tax Contest, each of the Parties shall provide prompt notice to the other Party of any written communication received by it or a member of its respective Group from a Taxing Authority regarding any Tax Contest for which it is indemnified by the other Party hereunder or for which it may be required to indemnify the other Party hereunder. Such notice shall attach copies of the pertinent portion of any written communication from a Taxing Authority and contain factual information (to the extent known) describing any

asserted Tax liability in reasonable detail and shall be accompanied by copies of any notice and other documents received from any Taxing Authority in respect of any such matters. Such notice shall be provided in a reasonably timely fashion; provided, however, that in the event that timely notice is not provided, a Party shall be relieved of its obligation to indemnify the other Party only to the extent that such delay results in actual increased costs or actual prejudice to such other Party.

6.06 SETTLEMENT RIGHTS. Unless waived by the Parties in writing, in connection with any potential adjustment in a Tax Contest, other than a 2007 TCE TSA Tax Contest, as a result of which adjustment the Non-Controlling Party may reasonably be expected to become liable to make any indemnification payment to the Controlling Party under this Agreement: (i) the Controlling Party shall keep the Non-Controlling Party informed in a timely manner of all actions taken or proposed to be taken by the Controlling Party with respect to such potential adjustment in such Tax Contest; (ii) the Controlling Party shall timely provide the Non-Controlling Party copies of any written materials relating to such potential adjustment in such Tax Contest received from any Tax Authority; (iii) the Controlling Party shall timely provide the Non-Controlling Party with copies of any correspondence or filings submitted to any Tax Authority or judicial authority in connection with such potential adjustment in such Tax Contest; and (iv) the Controlling Party shall defend such Tax Contest diligently and in good faith; provided, however, that nothing in this Article 6.06 shall affect Covidien's right to control, resolve, settle, or agree to any deficiency, claim, or adjustment proposed, asserted, or assessed in connection with or as a result of any Covidien Controlled Tax Contest, or consent to the resolution, settlement or agreement of any deficiency, claim or adjustment proposed, asserted or assessed in connection with or as a result of any such Mallinckrodt Controlled Tax Contest, in Covidien's sole and absolute discretion. The failure of the Controlling Party to take any action specified in the preceding sentence with respect to the Non-Controlling Party shall not relieve the Non-Controlling Party from any other liability or obligation which it may have to the Controlling Party.

### ARTICLE VII

# **COOPERATION**

7.01 GENERAL. Each Party shall fully cooperate, and shall cause all members of such Party's Group to fully cooperate, with the other Party in connection with the preparation and filing of any Tax Return or the conduct of any Tax Contest (including, where appropriate or necessary, providing a power of attorney) concerning any issues or any other matter contemplated pursuant to this Agreement. Each Party shall make its employees and facilities available on a mutually convenient basis to facilitate such cooperation.

7.02 CONSISTENT TREATMENT. Unless and until there has been a Final Determination to the contrary, each Party agrees not to take any position on any Tax Return, in connection with any Tax Contest or otherwise that is inconsistent with (a)

the treatment of payments between the Covidien Group and the Mallinckrodt Group as set forth in Article 5.05, (b) the Rulings, (c) the Tax Opinions, or (d) the Tax treatment of any transaction included in the Reorganization.

### ARTICLE VIII

# **RETENTION OF RECORDS; ACCESS**

8.01 RETENTION OF RECORDS. For so long as the contents thereof may become material in the administration of any matter under applicable Tax law, but in any event until the later of (i) the expiration of any applicable statutes of limitation and (ii) seven years after the Distribution Date, the Parties shall retain records, documents, accounting data and other information (including computer data) necessary for the preparation and filing of all Tax Returns (collectively, "**Tax Records**") in respect of Taxes of any member of either the Covidien Group or the Mallinckrodt Group for any Pre-Distribution Period, Straddle Period, or Post-Distribution Period or for any Tax Contests relating to such Tax Returns. At any time after the Distribution Date that the Covidien Group proposes to destroy such material or information, it shall first notify the Mallinckrodt Group in writing and the Mallinckrodt Group proposes to destroy such materials or information, it shall first notify the Covidien Group in writing and the Mallinckrodt Group proposes to destroy such material or information proposed to be destroyed. At any time after the Distribution Date that the Covidien Group proposes to destroy such materials or information proposed to be destroyed. At any time after the Distribution Date that the Covidien Group proposes to destroy such materials or information proposed to be destroyed. At any time after the Distribution Date that the Mallinckrodt Group proposes to destroy such material or information proposed to be destroyed.

## 8.02 ACCESS TO TAX RECORDS.

(a) GENERAL RULE. The Parties and their respective Affiliates shall make available to each other for inspection and copying during normal business hours upon reasonable notice all Tax Records (and, for the avoidance of doubt, any pertinent underlying data accessed or stored on any computer program or information technology system) in their possession and shall permit the other Party and its Affiliates, authorized agents and representatives and any representative of a Taxing Authority or other Tax auditor direct access, during normal business hours upon reasonable notice to any computer program or information technology system used to access or store any Tax Records, in each case to the extent reasonably required by the other Party in connection with the preparation of Tax Returns or financial accounting statements, audits, litigation, or the resolution of items pursuant to this Agreement. The Party seeking access to the records of the other Party shall bear all costs and expenses associated with such access, including any professional fees.

8.03 PRESERVATION OF PRIVILEGE. No member of the Mallinckrodt Group shall provide access to, copies of, or otherwise disclose to any Person any documentation relating to Taxes existing as of the date hereof to which Privilege may reasonably be asserted without the prior written consent of Covidien, such consent not to be unreasonably withheld.

## ARTICLE IX

# **DISPUTE RESOLUTION**

9.01 The Parties mutually desire that friendly collaboration will continue between them. Accordingly, they will try, and they will cause their respective Group members to try, to resolve in an amicable manner all disagreements and misunderstandings connected with their respective rights and obligations under this Agreement, including any amendments hereto. In furtherance thereof, in the event of any dispute or disagreement (a "**Dispute**") between any member of the Covidien Group and any member of the Mallinckrodt Group as to the interpretation of any provision of this Agreement or the performance of obligations hereunder, the Tax departments of the Parties shall negotiate in good faith to resolve the Dispute. If such good faith negotiations do not resolve the Dispute, then the matter, upon written request of either Party, will be referred to the persons at each Party holding the title of General Counsel (or such other chief legal officer at such Party) for resolution. If such Dispute is not resolved within ninety (90) Business Days following the date on which the senior managers receive notification, the Parties to such Dispute shall each separately retain an independent, nationally recognized law or accounting firm (each, a "**Preliminary Tax Advisor**") and, together, the "**Preliminary Tax Advisors**"), which Preliminary Tax **Advisor**") on behalf of the Parties to the Dispute to act as an arbitrator in order to resolve the Dispute. The Tax Advisor's determination as to any Dispute shall be made in accordance with the terms of this Agreement and shall be final and binding on the Parties and not subject to collateral attack for any reason (other than manifest error). All fees and expenses of the Preliminary Tax Advisor shall be borne by the Party that engaged such advisor and all of the fees and expenses of the Tax Advisor shall be shared equally by each of the Parties to the Dispute.

# ARTICLE X

### **MISCELLANEOUS PROVISIONS**

10.01 CONFLICTING AGREEMENTS. In the event of any inconsistency between this Agreement and any Schedule hereto, the Schedule shall prevail. In the event and to the extent that there shall be a conflict between the provisions of this Agreement and the provisions of the Distribution Agreement or any Ancillary Agreement, this Agreement shall control with respect to the subject matter thereof.

10.02 ASSIGNABILITY. This Agreement shall be binding upon and inure to the benefit of the Parties hereto, and their respective successors and permitted assigns. Except as otherwise provided for in this Agreement, this Agreement shall not be assignable, in whole or in part, directly or indirectly, by either Party without the express written consent of the other Party, and any attempt to assign any rights or obligations arising under this Agreement without such consent shall be void. A Party hereto may assign its respective rights or delegate its respective obligations under this Agreement to any Affiliate of such Party; provided, however, that in connection with each such assignment or delegation, the assigning Party provides a guarantee to the non-assigning Party for any liability or obligation assigned or delegated pursuant to this Section 10.02; provided, further, that Mallinckrodt shall only be entitled to assign its rights or delegate its obligations under this Agreement with the prior written consent of Covidien.

10.03 NO FIDUCIARY RELATIONSHIP. The duties and obligations of the Parties, and their respective successors and permitted assigns, contained herein are the extent of the duties and obligations contemplated by this Agreement; nothing in this Agreement is intended to create a fiduciary relationship between the Parties hereto, or any of their successors and permitted assigns, or create any relationship or obligations other than those explicitly described.

10.04 APPLICATION TO PRESENT AND FUTURE SUBSIDIARIES. This Agreement is being entered into by Covidien and Mallinckrodt on behalf of themselves and the members of their respective Group. This Agreement shall constitute a direct obligation of each such Party and shall be deemed to have been readopted and affirmed on behalf of any entity that becomes a Subsidiary of Covidien or Mallinckrodt in the future.

10.05 FURTHER ASSURANCES. Subject to the provisions hereof, the Parties hereto shall make, execute, acknowledge and deliver such other instruments and documents, and take all such other actions, as may be reasonably required in order to effectuate the purposes of this Agreement and to consummate the transactions contemplated hereby.

10.06 SURVIVAL. Notwithstanding any other provision of this Agreement to the contrary, all representations, covenants and obligations contained in this Agreement shall survive until the expiration of the applicable statute of limitations with respect to any such matter (including extensions thereof).

10.07 NOTICES. All notices, requests, claims, demands or other communications under this Agreement shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by facsimile or electronic transmission with receipt confirmed (followed by delivery of an original via overnight courier service) or by registered or certified mail (postage prepaid, return receipt requested) to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this Article 10.07):

If to Covidien, to: Covidien plc 1st Floor, 20 on Hatch Lower Hatch Street Dublin 2 Ireland Attn: General Counsel Facsimile: +352-266-379-92

and
Covidien 15 Hampshire Street Mansfield, MA 02048 Attn: Eric Green Facsimile: (508) 261-8544
with a copy to:
Skadden, Arps, Slate, Meagher & Flom LLP 4 Times Square New York, NY 10036 Attn: Sally Thurston Facsimile: (917) 777-4140
If to Mallinckrodt, to:
Mallinckrodt plc Damastown, Mulhuddart Dublin 15 Ireland Attn: General Counsel Facsimile: +353-1-438-1798
and
Mallinckrodt 675 James S. McDonnell Blvd. Hazelwood, MO 63042 Attn: Vice President of Taxation Facsimile: +353-1-438-1798
with a copy to:
[] [] [] Attn: [] Facsimile: []

Any Party may, by notice to the other Party, change the address to which such notices are to be given.

10.08 NO CIRCUMVENTION. The Parties agree not to directly or indirectly take any actions, act in concert with any Person who takes an action, or cause or allow any member of any such Party's Group to take any actions (including the failure

to take a reasonable action) such that the resulting effect is to materially undermine the effectiveness of any of the provisions of this Agreement, the Distribution Agreement or any other Ancillary Agreement (including adversely affecting the rights or ability of any Party to successfully pursue indemnification or payment pursuant to the provisions of this Agreement).

10.09 NO DUPLICATION; NO DOUBLE RECOVERY. Nothing in this Agreement is intended to confer to or impose upon any Party a duplicative right, entitlement, obligation, or recovery with respect to any matter arising out of the same facts and circumstances.

10.10 DISTRIBUTION AGREEMENT. To the extent not inconsistent with any specific term of this Agreement, the provisions of the Distribution Agreement shall apply in relevant part to this Agreement, including Article IX Termination; 10.1 Counterparts; Entire Agreement; Corporate Power; 10.2 Governing Law; Submissions to Jurisdiction; Waiver of Jury Trial; 10.4 Third-Party Beneficiaries; 10.6 Severability; 10.7 Force Majeure; 10.9 Expenses; 10.10 Headings; 10.12 Waivers; 10.13 Specific Performance; 10.14 Amendments; and 10.15 Interpretation.

\* \*

\*

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the Parties hereto have duly executed this Agreement as of the day and year first above written.

COVIDIEN PLC

By: Name: Title:

MALLINCKRODT PLC

By:

Name: Title:

EMPLOYEE MATTERS AGREEMENT BY AND BETWEEN COVIDIEN PLC AND MALLINCKRODT PLC DATED AS OF [—]

# TABLE OF CONTENTS

ARTICLE I GENE Section 1.1	RAL PRINCIPLES FOR ALLOCATION OF LIABILITIES General Principles		1 1		
Section 1.2	Service Credit		2		
Section 1.3	Transition Services		3		
Section 1.4	No Duplication or Acceleration of Benefits		3		
Section 1.5	No Expansion of Participation		3		
Section 1.6	Non-U.S. Regulatory Compliance		3		
ARTICLE II DEFINITIONS					
Section 2.1	Definitions		4		
Section 2.2	Interpretation		9		
ARTICLE III ASSIGNMENT OF EMPLOYEES					
Section 3.1	Active Employees		9 9		
Section 3.2	Employee Records		10		
Section 3.3	Non-Solicitation		11		
ARTICLE IV INCENTIVE COMPENSATION PLANS					
Section 4.1	General Principles		11		
Section 4.2	Treatment of Stock Options		12		
Section 4.3	Restricted Units		13		
Section 4.4	Performance Units		14		
Section 4.5	Liabilities for Settlement of Awards		15		
Section 4.6	Annual Incentive Plan Payments		16		
Section 4.7	Equity Plan Approval		16		
ARTICLE V U.S. QUALIFIED RETIREMENT PLANS					
Section 5.1	Pension Plans		16		
Section 5.2	Defined Contribution Plans		17		
Section 5.3	Employee Benefit Plan Governance Structure		18		
ARTICLE VI U.S. NON-QUALIFIED DEFERRED COMPENSATION PLANS					
Section 6.1	Mallinckrodt Non-Qualified Deferred Compensation Plan		19		
Section 6.2	Covidien Non-Qualified Deferred Compensation Plan		19		
Section 6.3	Participation; Distributions		20		
ARTICLE VII U.S. HEALTH, WELFARE AND FRINGE BENEFIT PLANS					
Section 7.1	Health Plans		20		
Section 7.2	Section 125 Plans		21		
Section 7.3	Fringe Benefits		21		
Section 7.4	Workers' Compensation		21		
Section 7.5	Indemnification		21		
Section 7.6	Termination of Participation		21		
ARTICLE VIII NC	ARTICLE VIII NON-U.S. EMPLOYEES				

-i-

# ARTICLE IX GENERAL PROVISIONS

<b>`</b>						
	Section 9.1	Preservation of Rights to Amend				
	Section 9.2	Fiduciary Matters				
	Section 9.3	Entire Agreement				
	Section 9.4	Binding Effect; No Third-Party Beneficiaries; Assignment				
	Section 9.5	Amendment				
	Section 9.6	Remedies Cumulative				
	Section 9.7	Notices				
	Section 9.8	Counterparts				
	Section 9.9	Severability				
	Section 9.10	Governing Law				
	Section 9.11	Dispute Resolution				
	Section 9.12	Performance				

- Section 9.13 Construction
- Section 9.14 Effect if Distribution Does Not Occur

-ii-

### **EMPLOYEE MATTERS AGREEMENT**

THIS EMPLOYEE MATTERS AGREEMENT, made and entered into effective as of [--] (this "<u>Agreement</u>"), is by and between Covidien plc, an Irish public limited company ("<u>Covidien</u>"), and Mallinckrodt plc, an Irish public limited company ("<u>Mallinckrodt</u>"). Covidien and Mallinckrodt are also referred to in this Agreement individually as, a "<u>Party</u>" and collectively, as the "<u>Parties</u>." Capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in Article II.

## RECITALS

WHEREAS, Covidien has determined that it would be appropriate, desirable and in the best interests of Covidien and its shareholders to separate the Mallinekrodt Business from Covidien;

WHEREAS, Covidien and Mallinckrodt have entered into the Separation and Distribution Agreement, dated [ ], 2013 (the "Separation Agreement"), in connection with the separation of the Mallinckrodt Business from Covidien and the Distribution of Mallinckrodt Ordinary Shares to shareholders of Covidien;

WHEREAS, the Separation Agreement also provides for the execution and delivery of certain other agreements, including this Agreement, in order to facilitate and provide for the separation of Mallinckrodt and its Subsidiaries from Covidien; and

WHEREAS, in order to ensure an orderly transition under the Separation Agreement, it will be necessary for the Parties to allocate between them certain Assets and Liabilities with respect to certain employee compensation and benefit plans and programs, and to address certain other employment matters.

NOW, THEREFORE, in consideration of the foregoing and the covenants and agreements set forth below and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties hereby agree as follows:

# ARTICLE I

# GENERAL PRINCIPLES FOR ALLOCATION OF LIABILITIES

Section 1.1 <u>General Principles</u>. (a) Except as otherwise provided in this Agreement, the Separation Agreement or any Ancillary Agreement, effective as of the Distribution Date or earlier, one or more members of the Mallinckrodt Group (as determined by Mallinckrodt) shall assume or continue the sponsorship of, and shall pay, perform and discharge, and no Covidien Entity shall have any Liability with respect to or under, the following agreements, obligations and Liabilities, and Mallinckrodt shall indemnify each Covidien Entity, and the officers, directors, and employees of each Covidien Entity, and hold them harmless with respect to such agreements, obligations and Liabilities:

(i) any and all wages, salaries, incentive compensation (as the same may be modified by this Agreement), commissions, bonuses, and any other employee

compensation or benefits payable to or on behalf of any Mallinckrodt Group Employees after the Distribution Date, without regard to when such wages, salaries, incentive compensation, commissions, bonuses, or other employee compensation or benefits are or may have been earned;

(ii) any and all immigration-related, visa, work application or similar rights, obligations and Liabilities related to any Mallinckrodt Group Employees; and

(iii) any and all Liabilities and obligations whatsoever with respect to claims made by or with respect to any Mallinckrodt Group Employees in connection with any employee benefit plan, program or policy not retained or assumed by any Covidien Entity pursuant to this Agreement, the Separation Agreement or any Ancillary Agreement, including any such Liabilities relating to actions or omissions of or by any Mallinckrodt Entity or any officer, director, employee or agent thereof prior to, on or after the Distribution Date.

(b) Except as otherwise provided in this Agreement, effective as of the Effective Time, no Mallinckrodt Entity shall have any Liability for, and Covidien shall indemnify each Mallinckrodt Entity, and the officers, directors, and employees of each Mallinckrodt Entity, and hold them harmless with respect to any and all Liabilities and obligations whatsoever with respect to, claims made by or with respect to any Covidien Group Employees or Former Covidien Group Employees in connection with any employee benefit plan, program or policy not retained or assumed by any Mallinckrodt Entity pursuant to this Agreement, including such Liabilities relating to actions or omissions of or by any Covidien Entity or any officer, director, employee or agent thereof on, prior to or after the Distribution Date.

# Section 1.2 Service Credit.

(a) <u>Service for Eligibility, Vesting, and Benefit Purposes</u>. Except as otherwise provided in any other provision of this Agreement, the Mallinckrodt Benefit Plans shall, and Mallinckrodt shall cause each Mallinckrodt Entity to, recognize each Mallinckrodt Group Employee's and each Former Mallinckrodt Group Employee's full service with any Covidien Entity on or prior to the Effective Time, to the same extent such service would be credited if it had been performed for a Mallinckrodt Entity, for purposes of eligibility, vesting and determination of level of benefits under any such Mallinckrodt Benefit Plan.

(b) <u>Evidence of Prior Service</u>. Notwithstanding anything to the contrary, but subject to applicable Law, upon reasonable request by either Party (the "<u>Requesting Party</u>"), the other Party (the "<u>Providing Party</u>") will provide to the Requesting Party copies of any records available to the Providing Party to document the service, plan participation and membership of former Employees of the Providing Party who are then Employees of the Requesting Party, and will cooperate with the Requesting Party to resolve any discrepancies or obtain any missing data for purposes of determining benefit eligibility, participation, vesting and calculation of benefits with respect to any such Employee.

-2-

(c) <u>Accrued Time Off</u>. Mallinckrodt shall recognize and assume all Liability for all unused vacation, holiday, sick leave, flex days, personal days and paid-time off and other time-off benefits with respect to Mallinckrodt Group Employees which accrued on or prior to the Effective Time, and Mallinckrodt shall credit each Mallinckrodt Group Employee with such accrual.

(d) <u>Leaves of Absence</u>. Mallinckrodt will continue to apply the appropriate leave of absence policies applicable to inactive Mallinckrodt Group Employees who are on an approved leave of absence as of the Effective Time. Leaves of absence taken by Mallinckrodt Group Employees prior to the Effective Time shall be deemed to have been taken as employees of a Mallinckrodt Entity.

Section 1.3 <u>Transition Services</u>. The Parties acknowledge that the Covidien Group or the Mallinckrodt Group may provide administrative services for certain of the other Party's benefit plans, programs or arrangements for a transitional period under the terms of a Transition Services Agreement. The Parties hereby agree to enter into any agreement necessary to implement a Transition Services Agreement, including but not limited to a business associate agreement (if required by HIPAA or other applicable health information privacy Laws).

Section 1.4 <u>No Duplication or Acceleration of Benefits</u>. Notwithstanding anything to the contrary in this Agreement, the Separation Agreement or any Ancillary Agreement, no participant in the Mallinckrodt 401(k) Plan, Mallinckrodt Health Plans or any other Mallinckrodt Benefit Plan shall receive benefits to the extent that receipt of such benefits would result in duplication of benefits provided by the corresponding Covidien Benefit Plan or any other plan, program or arrangement sponsored or maintained by a Covidien Entity. Furthermore, unless expressly provided for in this Agreement, the Separation Agreement or in any Ancillary Agreement or required by applicable Law, no provision in this Agreement shall be construed to create any right to accelerate vesting or entitlements under any compensation or Benefit Plan, program or arrangement sponsored or maintained by a Covidien Entity on the part of any Employee.

Section 1.5 <u>No Expansion of Participation</u>. Unless otherwise expressly provided in this Agreement, as otherwise determined or agreed to by Covidien and Mallinckrodt, as required by applicable Law, or as explicitly set forth in a Mallinckrodt Benefit Plan, a Mallinckrodt Group Employee shall be entitled to participate in the Mallinckrodt Benefit Plans on the Distribution Date only to the extent that such Mallinckrodt Group Employee was entitled to participate in the corresponding Covidien Benefit Plan as in effect immediately prior to the Distribution Date (to the extent such Mallinckrodt Group Employee is not currently participating in the respective Mallinckrodt Benefit Plan immediately prior to the Distribution Date), it being understood that this Agreement does not expand (a) the number of Mallinckrodt Group Employees entitled to participate in any Mallinckrodt Benefit Plan or (b) the participation rights of Mallinckrodt Group Employees in any Mallinckrodt Benefit Plans beyond the rights of such Mallinckrodt Group Employees under the corresponding Covidien Benefit Plans, in each case, following the Effective Time.

Section 1.6 Non-U.S. Regulatory Compliance. Covidien shall have the authority to adjust the treatment described in this Agreement with respect to Mallinckrodt Group

-3-

Employees who are located outside of the U.S. in order to ensure compliance with the applicable laws or regulations of countries outside the U.S. or to preserve the tax benefits provided under local tax law or regulation prior to the Distribution.

#### ARTICLE II DEFINITIONS

Section 2.1 Definitions. As used in this Agreement, the following terms shall have the meanings set forth in this Section 2.1:

"Adjusted Covidien Option" shall have the meaning set forth in Section 4.2.

"Adjusted Covidien PSU" shall have the meaning set forth in Section 4.4.

"Adjusted Covidien RSU" shall have the meaning set forth in Section 4.3.

"Affiliate" shall have the meaning set forth in the Separation Agreement.

"Agreement" shall mean this Employee Matters Agreement, together with all Schedules hereto and all amendments, modifications, and changes hereto entered into pursuant to Section 9.5.

"Ancillary Agreements" shall have the meaning set forth in the Separation Agreement.

"Assets" shall have the meaning set forth in the Separation Agreement.

"Benefit Management Records" shall have the meaning set forth in Section 3.2(b).

"Benefit Plan" shall mean any contract, agreement, policy, practice, program, plan, trust, commitment or arrangement providing for benefits, perquisites or compensation of any nature from an employer to any Employee, or to any family member, dependent, or beneficiary of any such Employee, including pension plans, thrift plans, supplemental pension plans and welfare plans, and contracts, agreements, policies, practices, programs, plans, trusts, commitments and arrangements providing for terms of employment, fringe benefits, severance benefits, change in control protections or benefits, travel and accident, life, accidental death and dismemberment, disability and accident insurance, tuition reimbursement, travel reimbursement, vacation, sick, personal or bereavement days, leaves of absences and holidays; provided, however, the term "Benefit Plan" does not include any government sponsored benefits, such as workers' compensation, unemployment or any similar plans, programs or policies.

"<u>COBRA</u>" shall mean the U.S. Consolidated Omnibus Budget Reconciliation Act of 1985, as codified at Section 601 *et seq.* of ERISA and at Section 4980B of the Code.

"Code" shall have the meaning set forth in the Separation Agreement.

"Covidien" shall have the meaning set forth in the preamble to this Agreement.

-4-

"Covidien Adjusted Exercise Price" shall have the meaning set forth in Section 4.2(a)(i).

"Covidien Benefit Plan" shall mean any Benefit Plan sponsored or maintained by a Covidien Entity immediately prior to the Effective Time.

"Covidien Entity" shall mean any member of the Covidien Group.

"Covidien Equity Plan" shall mean any equity compensation plan sponsored or maintained by Covidien immediately prior to the Distribution Date, including the Covidien Stock and Incentive Plan, the Covidien Employee Stock Purchase Plan, and the Covidien Savings Related Share Plan.

"Covidien 401(k) Plan" shall mean the Covidien Retirement Savings and Investment Plan.

"Covidien Group" shall have the meaning set forth in the Separation Agreement.

"Covidien Group Employee" shall have the meaning set forth in Section 3.1(b).

"Covidien Health Plan" shall mean the Covidien Health & Welfare Benefits Plan.

"<u>Covidien Non-qualified Plans</u>" shall mean the Covidien Supplemental Savings and Retirement Plan, the Covidien Supplemental Executive Retirement Plan, The Kendall Company Senior Executive Supplemental Retirement Plan, the Batts Inc. Supplemental Retirement and Death Benefit Agreement and the Batts Inc. Nonqualified Deferred Compensation Plan.

"Covidien Options" shall mean exercisable and non-exercisable stock options to purchase Covidien Ordinary Shares granted pursuant to the Covidien Stock and Incentive Plan or a predecessor plan.

"Covidien Ordinary Shares" shall mean the ordinary shares, par value \$0.20 per share, of Covidien.

"Covidien Pension Plan" shall mean the Kendall Pension Plan.

"<u>Covidien Post-Distribution Stock Value</u>" shall mean the volume weighted average per share price of Covidien Ordinary Shares trading on an exdividend basis on the NYSE during Regular Trading Hours on the Distribution Date.

"Covidien Pre-Distribution Stock Value" shall mean the volume weighted average per share price of Covidien Ordinary Shares trading "regular way with due bills" on the NYSE during Regular Trading Hours on the Distribution Date.

"<u>Covidien Price Ratio</u>" shall mean the quotient obtained by dividing the Covidien Post-Distribution Stock Value by the Covidien Pre-Distribution Stock Value, rounded to the nearest one ten-thousandth.

"Covidien PSUs" shall mean performance units granted under the Covidien Stock and Incentive Plan or a predecessor plan.

-5-

"Covidien Retained Savings Plans" shall have the meaning set forth in Section 5.2(b).

"Covidien RSUs" shall mean restricted units granted under the Covidien Stock and Incentive Plan or a predecessor plan.

"<u>Covidien Share Ratio</u>" shall mean the quotient obtained by dividing the Covidien Pre-Distribution Stock Value by the Covidien Post-Distribution Stock Value, rounded to the nearest one-ten-thousandth.

"Distribution" shall have the meaning set forth in the Separation Agreement.

"Distribution Date" shall have the meaning set forth in the Separation Agreement.

"Effective Time" shall have the meaning set forth in the Separation Agreement.

"Employee" shall mean any Covidien Group Employee, Former Covidien Group Employee, Mallinekrodt Group Employee or Former Mallinekrodt Group Employee.

"ERISA" shall mean the U.S. Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder.

"Former Covidien Group Employee" shall mean a former employee of the Covidien Group whose employment with the Covidien Group was terminated before the Effective Time (and who is not actively employed by the Covidien Group as of the Effective Time).

"Former Employees" shall mean Former Covidien Group Employees and Former Mallinckrodt Group Employees.

"Former Mallinckrodt Group Employee" shall mean a former employee of the Mallinckrodt Business whose employment was terminated before the Effective Time (and who is not actively employed by the Mallinckrodt Group immediately following the Effective Time).

"<u>Government Entity</u>" shall mean any instrumentality, subdivision, court, administrative agency, commission, official or other authority of any country, state, province, prefect, municipality, locality or other government or political subdivision thereof, or any quasi-governmental or private body exercising any regulatory, taxing, importing or other governmental or quasi-governmental authority.

"HIPAA" shall mean the U.S. Health Insurance Portability and Accountability Act of 1996, as amended, and the regulations promulgated thereunder.

"IRS" shall mean the U.S. Internal Revenue Service.

"Law" shall have the meaning set forth in the Separation Agreement.

"Liabilities" shall have the meaning set forth in the Separation Agreement.

-6-

"Mallinckrodt" shall have the meaning set forth in the preamble to this Agreement.

"Mallinckrodt Annual Incentive Plan" shall have the meaning set forth in Section 4.6.

"Mallinckrodt Benefit Plan" shall mean any Benefit Plan sponsored or maintained by a Mallinckrodt Entity.

"Mallinckrodt Business" shall have the meaning set forth in the Separation Agreement.

"Mallinckrodt Deferred Compensation Plans" shall have the meaning set forth in Section 6.1(a).

"Mallinckrodt Entity" shall mean any member of the Mallinckrodt Group.

"Mallinckrodt Exercise Price" shall have the meaning set forth in Section 4.2(b).

"Mallinckrodt 401(k) Plan" shall mean the Mallinckrodt Pharmaceuticals Retirement Savings and Investment Plan.

"Mallinckrodt Group" shall have the meaning set forth in the Separation Agreement.

"Mallinckrodt Group Employee" shall have the meaning set forth in Section 3.1(a).

"Mallinckrodt Health Plan" shall mean the Mallinckrodt Pharmaceuticals Health & Welfare Benefits Plan.

"<u>Mallinckrodt New Equity Plan</u>" shall mean the Mallinckrodt Pharmaceuticals Stock and Incentive Plan adopted by Mallinckrodt prior to the Effective Time and approved by Mallinckrodt's shareholders, under which the Mallinckrodt equity-based awards described in Article IV shall be issued.

"Mallinckrodt Options" shall have the meaning set forth in Section 4.2.

"Mallinckrodt Ordinary Shares" shall mean the ordinary shares, par value \$0.20 per share, of Mallinckrodt.

"Mallinckrodt Pension Plans" shall have the meaning set forth in Section 5.1(a).

"<u>Mallinckrodt Price Ratio</u>" shall mean the quotient obtained by dividing the Mallinckrodt Stock Value by the Covidien Pre-Distribution Stock Value rounded to the nearest one ten-thousandth.

-7-

"Mallinckrodt PRC Nationals" shall have the meaning set forth in Section 4.1(d).

"Mallinckrodt PSUs" shall have the meaning set forth in Section 4.4.

"Mallinckrodt RSUs" shall have the meaning set forth in Section 4.3.

"Mallinckrodt Savings Plans" shall have the meaning set forth in Section 5.2(a).

"<u>Mallinckrodt Share Ratio</u>" shall mean the quotient obtained by dividing the Covidien Pre-Distribution Stock Value by the Mallinckrodt Stock Value, rounded to the nearest one ten-thousandth.

"Mallinckrodt Stock Value" shall mean the volume weighted average per share price of Mallinckrodt Ordinary Shares trading on the "when issued" market on the NYSE during Regular Trading Hours on the Distribution Date.

"Non-Covidien Option Holders" shall mean individuals who hold Covidien Options as of the Effective Time who are not Employees.

"<u>NYSE</u>" shall mean the New York Stock Exchange.

"Party" or "Parties" shall have the meaning set forth in the preamble to this Agreement.

"Person" shall have the meaning set forth in the Separation Agreement.

"Providing Party" shall have the meaning set forth in Section 1.2(b).

"Regular Trading Hours" shall mean the period beginning at 9:30 A.M., New York City time and ending at 4:00 P.M., New York City time.

"SAYE" shall have the meaning set forth in Section 4.1(d).

"Separation Agreement" shall have the meaning set forth in the recitals to this Agreement.

"Subsidiary" shall have the meaning set forth in the Separation Agreement.

"Transition Services Agreement" shall have the meaning set forth in the Separation Agreement.

"U.S." means the United States of America.

-8-

Section 2.2 Interpretation. The provisions of Section 11.15 of the Separation Agreement are hereby incorporated by reference.

## ARTICLE III ASSIGNMENT OF EMPLOYEES

### Section 3.1 Active Employees.

(a) <u>Mallinckrodt Group Employees</u>. Effective no later than immediately prior to the Effective Time, the applicable Covidien Entity shall have taken such actions as are necessary to ensure that each individual who is intended to be an employee of the Mallinckrodt Group immediately following the Effective Time (collectively, the "<u>Mallinckrodt Group Employees</u>") is employed by a Mallinckrodt Entity. Each of the Parties agrees to execute, and to seek to have the applicable employees execute, such documentation, if any, as may be necessary to reflect such assignment and/or transfer.

(b) <u>Covidien Group Employees</u>. Effective no later than immediately prior to the Effective Time, the applicable Covidien Entity shall have taken such actions as are necessary to ensure that each individual who is intended to be an employee of the Covidien Group immediately following the Effective Time (collectively, the "<u>Covidien Group Employees</u>") is employed by a Covidien Entity. Each of the Parties agrees to execute, and to seek to have the applicable employees execute, such documentation, if any, as may be necessary to reflect such assignment and/or transfer.

(c) <u>At-Will Status</u>. Notwithstanding the above or any other provision of this Agreement, nothing in this Agreement shall create any obligation on the part of any Covidien Entity or any Mallinckrodt Entity to (i) continue the employment of any Employee or permit the return from a leave of absence for any period following the date of this Agreement or the Distribution Date (except as required by applicable Law) or (ii) change the employment status of any Employee from "at-will," to the extent such Employee is an "at-will" employee under applicable Law.

(d) <u>Severance</u>. The Parties acknowledge and agree that the Distribution and the assignment, transfer or continuation of the employment of Employees as contemplated by this <u>Section 3.1</u> shall not be deemed an involuntary termination of employment entitling any Mallinckrodt Group Employee or Covidien Group Employee to severance payments or benefits.

(e) <u>Not a Change of Control/Change in Control</u>. The Parties acknowledge and agree that neither the consummation of the Distribution nor any transaction in connection with the Distribution shall be deemed a "change of control," "change in control," or term of similar import for purposes of any Benefit Plan sponsored or maintained by any Covidien Entity or Mallinckrodt Entity.

(f) <u>Confidentiality</u>. The provisions of this <u>Section 3.1</u> shall be in addition to, and not in derogation of, the provisions of Article IV and Article VII of the Separation Agreement. Except as otherwise set forth in this Agreement, all records and data relating to Employees shall, in each case, be subject to the confidentiality provisions of the Separation Agreement and any other applicable agreement and applicable Law.

-9-

## Section 3.2 Employee Records.

(a) <u>Sharing of Information</u>. Subject to any limitations imposed by applicable Law, Covidien and Mallinckrodt (acting directly or through members of the Covidien Group or the Mallinckrodt Group, respectively) shall provide to the other and their respective authorized agents and vendors all information necessary for the Parties to perform their respective duties under this Agreement. The Parties also hereby agree to enter into any business associate arrangement that may be required for the sharing of any Information pursuant to this Agreement to comply with the requirements of HIPAA.

(b) *Transfer of Personnel Records and Authorization*. Subject to any limitation imposed by applicable Law and to the extent that it has not done so before the Distribution Date, on the Distribution Date, Covidien shall transfer and assign to Mallinckrodt all personnel records, all immigration documents, including I-9 forms and work authorizations, all payroll deduction authorizations and elections, whether voluntary or mandated by Law, including but not limited to W-4 forms, W-8BEN forms and deductions for benefits under the applicable Mallinckrodt Benefit Plan and all absence management records, Family and Medical Leave Act records, insurance beneficiary designations, flexible spending account enrollment confirmations, attendance, and return to work information ("Benefit Management Records") relating to participants in Mallinckrodt Benefit Plans. Subject to any limitations imposed by applicable Law, Covidien, however, may retain originals of, copies of, or access to, personnel records, immigration records, payroll forms and Benefit Management Records as long as necessary to comply with its obligations under applicable Law or to provide services to Mallinckrodt (acting on its behalf pursuant to any Transition Services Agreement entered into by and between the Parties). Immigration records will, if and as appropriate, become a part of Mallinckrodt's public access file. Mallinckrodt will use personnel records, payroll forms and Benefit Management. It is understood that following the Distribution Date, Covidien records so transferred and assigned may be maintained by Mallinckrodt (acting directly or through one of its Subsidiaries) pursuant to Mallinckrodt's applicable records retention policy.

(c) <u>Access to Records</u>. To the extent not inconsistent with this Agreement, the Separation Agreement or any applicable privacy protection Laws or regulations, reasonable access to Employee-related records after the Distribution Date will be provided to members of the Covidien Group and members of the Mallinckrodt Group pursuant to the terms and conditions of Section 7.6 of the Separation Agreement.

(d) <u>Maintenance of Records</u>. With respect to retaining, destroying, transferring, sharing, copying and permitting access to all Employee-related information, Covidien and Mallinckrodt shall comply with all applicable Laws, regulations and internal policies, and shall indemnify and hold harmless each other from and against any and all Liability, claims, actions, and damages that arise from a failure (by the indemnifying party or its Subsidiaries or their respective agents) to so comply with all applicable Laws, regulations and internal policies applicable to such information.

-10-

(e) <u>Cooperation</u>. Each Party shall use commercially reasonable efforts to cooperate and work together to unify, consolidate and share (to the extent permissible under applicable privacy/data protection laws) all relevant documents, Board resolutions, government filings, data, payroll, employment and benefit plan Information on regular timetables, make certain that each applicable entity's data and records are correct and updated on a timely basis, and cooperate as needed with respect to (i) any litigation with respect to any employee benefit plan, policy or arrangement contemplated by this Agreement, (ii) an audit of any employee benefit plan, policy or arrangement contemplated by this Agreement by the IRS, U.S. Department of Labor or any other Government Entity, (iii) seeking a determination letter, private letter ruling or advisory opinion from the IRS or U.S. Department of Labor on behalf of any employee benefit plan, policy or arrangement contemplated by this Agreement, and (iv) any filings that are required to be made or supplemented to the IRS, U.S. Pension Benefit Guaranty Corporation, U.S. Department of Labor or any other Government Entity; provided, however, that requests for cooperation must be reasonable and not interfere with daily business operations.

Section 3.3 <u>Non-Solicitation</u>. Each Party agrees that, for a period of two years from the Distribution Date, such party will not solicit for employment any employee of the other Party; <u>provided</u>, <u>however</u>, that this <u>Section 3.3</u> shall not prohibit: (a) generalized solicitations which are not directed to specific individuals or employees of the other party (for the avoidance of doubt, including solicitations resulting from actions initiated by employees through the Covidien or Mallinckrodt internal posting system); (b) solicitations of persons whose employment was involuntarily terminated by the other Party; or (c) solicitations expressly permitted in writing by the Senior Vice President, Human Resources of the Party which employs the person who is to be solicited.

#### ARTICLE IV INCENTIVE COMPENSATION PLANS

#### Section 4.1 General Principles.

(a) Covidien and Mallinckrodt shall take any and all reasonable actions as shall be necessary and appropriate to further the provisions of this Article IV, including, to the extent practicable, providing written notice or similar communication to each Employee who holds one or more equity awards granted under any of the Covidien Equity Plans informing such Employee of (i) the actions contemplated by this Article IV with respect to such awards and (ii) whether (and during what time period) any "blackout" period shall be imposed upon holders of awards granted under any of the Covidien Equity Plans during which time awards may not be exercised or settled, as the case may be.

(b) Following the Distribution, a grantee who has outstanding awards under one or more of the Covidien Equity Plans or replacement awards under the Mallinckrodt New Equity Plan shall be considered to have been employed by the applicable plan sponsor before and after the Distribution for purposes of vesting. For purposes of the equity awards and except as otherwise provided in Section 4.1(d) below, the Distribution shall not result in a termination of employment or service for any Employee, rather the date of termination of employment or service with the applicable plan sponsor following the Distribution shall be the termination date for the purposes of any outstanding equity awards.

-11-

(c) No award described in this Article IV, whether outstanding or to be issued, adjusted, substituted or cancelled by reason of or in connection with the Distribution, shall be adjusted, settled, cancelled, or exercisable, until in the judgment of the administrator of the applicable plan or program such action is consistent with all applicable Laws, including U.S. securities Laws. Neither the period of exercisability nor the term of any award will be extended on account of a period during which such an award is not exercisable pursuant to the preceding sentence.

(d) Notwithstanding anything to the contrary in this Agreement, Covidien Options which are intended to be "approved options" and that were issued pursuant to the Covidien Stock and Incentive Plan HMRC Approved Sub-Plan for the United Kingdom and options to purchase Covidien Ordinary Shares which were offered by invitation pursuant to the Covidien Savings Related Share Plan ("SAYE") will not be adjusted as described below and any Covidien Options, Covidien PSUs or Covidien RSUs held by a Mallinckrodt Group Employee who is a resident of the People's Republic of China ("Mallinckrodt PRC Nationals") will not be adjusted as described below. Notwithstanding the provisions of <u>Section 4.1(b)</u>, the Distribution shall be treated as a termination of employment from a Covidien Entity for Mallinckrodt PRC Nationals and any Mallinckrodt Group Employee in the United Kingdom who hold options to purchase Covidien Ordinary Shares through the SAYE. The applicable terms and conditions of equity awards held by Mallinckrodt PRC Nationals and the terms of the SAYE shall govern the vesting and exercisability of the applicable awards upon the Distribution.

(e) The adjustment or conversion of Covidien Options, Covidien PSUs, Covidien RSUs, Mallinckrodt Options, Mallinckrodt RSUs and Mallinckrodt PSUs shall be effectuated in a manner that is intended to avoid the imposition of any penalty or other taxes on the holders thereof pursuant to Code Section 409A.

### Section 4.2 Treatment of Stock Options.

(a) <u>Covidien Options Held by Covidien Group Employees; Former Employees; Non-Covidien Option Holders</u>. Except as otherwise provided in <u>Section 4.1(d)</u>, each Covidien Option that is outstanding as of the Effective Time that is held by a Covidien Group Employee, a Former Employee or a Non-Covidien Option Holder shall remain an option to purchase Covidien Ordinary Shares and shall be adjusted as described below to reflect the Distribution (each such option, an "<u>Adjusted Covidien Option</u>"). Each Adjusted Covidien Option shall be subject to the same terms and conditions after the Effective Time as the terms and conditions applicable to the corresponding Covidien Option immediately prior to the Effective Time; <u>provided</u>, <u>however</u>, that from and after the Effective Time:

(i) the per-share exercise price of each such Adjusted Covidien Option shall be equal to the product of (i) the per-share exercise price of the corresponding Covidien Option immediately prior to the Effective Time multiplied by (ii) the Covidien Price Ratio, rounded up to the nearest whole cent (the "<u>Covidien Adjusted Exercise Price</u>"); and

-12-

(ii) the number of Covidien Ordinary Shares subject to each such Adjusted Covidien Option shall be equal to the product of (i) the number of Covidien Ordinary Shares subject to the corresponding Covidien Option immediately prior to the Effective Time multiplied by (ii) the Covidien Share Ratio, with any fractional shares rounded down to the nearest whole share.

(b) <u>Covidien Options Held by Mallinckrodt Group Employees</u>. Except as otherwise provided in <u>Section 4.1(d)</u>, each Covidien Option that is outstanding as of the Effective Time that is held by a Mallinckrodt Group Employee who continues to be employed by the Mallinckrodt Group immediately after the Distribution shall be converted into an option to purchase Mallinckrodt Ordinary Shares pursuant to the Mallinckrodt New Equity Plan and shall be adjusted as described below to reflect the Distribution (each such option, a "<u>Mallinckrodt Option</u>"). Each Mallinckrodt Option shall be subject to the same terms and conditions after the Effective Time as the terms and conditions applicable to the corresponding Covidien Option immediately prior to the Effective Time; <u>provided</u>, however, that from and after the Effective Time:

(i) the per-share exercise price of each such Mallinckrodt Option shall be equal to the product of (A) the per-share exercise price of the corresponding Covidien Option immediately prior to the Effective Time multiplied by (B) the Mallinckrodt Price Ratio, rounded up to the nearest whole cent (the "<u>Mallinckrodt Exercise Price</u>"); and

(ii) the number of Mallinckrodt Ordinary Shares subject to each such Mallinckrodt Option shall be equal to the product of (A) the number of Covidien Ordinary Shares subject to the corresponding Covidien Option immediately prior to the Effective Time multiplied by (B) the Mallinckrodt Share Ratio, with any fractional share rounded down to the nearest whole share.

#### Section 4.3 Restricted Units.

(a) Each award of Covidien RSUs held by a Covidien Group Employee or Former Employee immediately prior to the Effective Time shall be adjusted, effective as of the Effective Time, by multiplying the number of Covidien Ordinary Shares subject to such Covidien RSU award by the Covidien Share Ratio, which product shall be rounded down to the nearest whole number of units with a cash payment for any fractional units to be made as soon as administratively practicable after the Effective Time but in any event no later than September 1, 2013 (each such adjusted Covidien RSU, an "Adjusted Covidien RSU"); provided, however, that for any Covidien Group Employee who has satisfied the requirements for Normal Retirement (as defined in the applicable terms and conditions) as of the Effective Time, such product shall be subject to regular rounding in lieu of a cash payment for fractional units. Each Adjusted Covidien RSU shall be subject to the same terms and conditions after the Effective Time as the terms and conditions applicable to the corresponding Covidien RSU immediately prior to the Effective Time.

-13-

(b) Except as otherwise provided in <u>Section 4.1(d)</u>, each award of Covidien RSUs held by a Mallinckrodt Group Employee immediately prior to the Effective Time who continues to be employed by the Mallinckrodt Group immediately after the Distribution shall be converted to a restricted unit award relating to a number of Mallinckrodt Ordinary Shares (the "<u>Mallinckrodt RSUs</u>") determined by multiplying the number of Covidien Ordinary Shares subject to each Covidien RSU award by the Mallinckrodt Share Ratio, which product shall be rounded down to the nearest whole number of Mallinckrodt RSUs with a cash payment for any fractional units to be made as soon as administratively practicable after the Effective Time but in any event no later than September 1, 2013; <u>provided</u>, <u>however</u>, that for any Mallinckrodt Group Employee who has satisfied the requirements for Normal Retirement (as defined in the applicable terms and conditions) as of the Effective Time, such product shall be subject to regular rounding in lieu of a cash payment for fractional units. Except as otherwise provided herein, each Mallinckrodt RSU shall be subject to the same terms and conditions after the Distribution as the terms and conditions applicable to the corresponding Covidien RSUs immediately prior to the Distribution.

### Section 4.4 Performance Units.

(a) Each award of Covidien PSUs held by a Covidien Group Employee or Former Employee shall be adjusted, effective as of the Effective Time, by multiplying the number of Covidien Ordinary Shares subject to such Covidien PSU award by the Covidien Share Ratio which product shall be rounded down to the nearest whole number of units with a cash payment for any fractional units to be made as soon as administratively practicable after the Effective Time but in any event no later than September 1, 2013 (each such adjusted Covidien PSU, an "Adjusted Covidien PSU"); provided, however, that for any Covidien Group Employee who has satisfied the requirements for Normal Retirement (as defined in the applicable terms and conditions) as of the Effective Time, such product shall be subject to regular rounding in lieu of a cash payment for fractional units. Each Adjusted Covidien PSU shall be subject to the same terms and conditions applicable to the corresponding Covidien PSU immediately prior to the Effective Time, taking into account the adjustments to the performance measures relating to Covidien PSUs granted in fiscal years 2011 and 2012 that were approved by the Compensation Committee of the Covidien board of directors prior to the date hereof and that will be effective as of the Effective Time, as described below. The Covidien PSUs granted in fiscal year 2011 will be adjusted at the Effective Time to provide for the early conclusion of the performance cycle, resulting in such Covidien PSUs being subject to vesting based solely upon continued service through September 27, 2013. The 2012 Covidien PSUs granted in fiscal year 2012 will be adjusted at the Effective Time to provide for calculating total shareholder return or TSR) in order to reflect the changes in Covidien's peer group following the distribution of the pharmaceuticals business.

(b) Except as otherwise provided in Section 4.1(d), each award of Covidien PSUs held by a Mallinckrodt Group Employee immediately prior to the Effective Time who continues to be employed by the Mallinckrodt Group immediately after the Effective Time shall be converted to a performance unit award relating to a number of Mallinckrodt Ordinary Shares determined by multiplying the number of Covidien Ordinary Shares with respect to the Covidien PSU award by the Mallinckrodt Share Ratio, which product shall be rounded down to the nearest

-14-

whole number of units with a cash payment for any fractional units to be made as soon as administratively practicable after the Effective Time but in any event no later than September 1, 2013 (the "<u>Mallinckrodt PSUs</u>"); <u>provided</u>, <u>however</u>, that for any Covidien Group Employee who has satisfied the requirements for Normal Retirement (as defined in the applicable terms and conditions) as of the Effective Time, such product shall be subject to regular rounding in lieu of a cash payment for fractional units. The respective performance period for each award of Mallinckrodt PSUs shall cease as of the Effective Time and the performance multiplier for such award shall be determined by Covidien's total shareholder return throughout the performance period, as determined pursuant to the amended terms and conditions of the applicable award. To the extent that there is any vesting of performance units under the applicable amended terms and conditions for Mallinckrodt PSUs, such vesting shall be subject to the Mallinckrodt Group Employee's continued employment through the last day of the initial three-year performance period as in effective immediately prior to the Effective Time. Other than with respect to the foregoing, Mallinckrodt PSUs shall be subject to the same terms and conditions applicable to the corresponding Covidien PSUs immediately prior to the Distribution.

## Section 4.5 Liabilities for Settlement of Awards.

(a) <u>Settlement of Covidien Options</u>. Covidien shall be responsible for all Liabilities associated with Covidien Options (regardless of the holder of such awards) including any option exercise, share delivery, registration or other obligations related to the exercise of the Covidien Options. Covidien shall be responsible for all Liabilities associated with amounts payable to Covidien Group Employees or Mallinckrodt Group Employees who hold UK approved options or options through the SAYE; <u>provided</u>, <u>however</u>, that Mallinckrodt shall be responsible for paying to each Mallinckrodt Group Employee, through the payroll of the applicable Mallinckrodt Entity, all compensatory payments attributable to the non-conversion of UK approved options and SAYE options in connection with the Distribution upon receipt from Covidien of a list of Mallinckrodt Group Employees eligible to receive such compensatory payment and the amount payable to such Mallinckrodt Group Employee, listed individually, with such amount to be paid no later than the second regularly scheduled pay period that occurs after Covidien provides such list and amounts. Covidien (or any Covidien Entity) shall reimburse to Mallinckrodt (or the applicable Mallinckrodt Entity) the total amount payable to Mallinckrodt Group Employees pursuant to the previous sentence within sixty (60) days after receipt of an invoice from Mallinckrodt requesting reimbursement for such payment.

(b) <u>Settlement of Outstanding Covidien RSUs</u>. Covidien shall be responsible for all Liabilities associated with Covidien RSUs including any share delivery, registration or other obligations related to the settlement of the Covidien RSUs.

(c) <u>Settlement of Outstanding Covidien PSUs</u>. Covidien shall be responsible for all Liabilities associated with Covidien PSUs, including any share delivery, registration or other obligations related to the settlement of Covidien PSUs.

(d) <u>Settlement of Mallinckrodt Options</u>. Mallinckrodt shall be responsible for all Liabilities associated with Mallinckrodt Options (regardless of the holder of such awards) including any option exercise, share delivery, registration or other obligations related to the exercise of the Mallinckrodt Options.

-15-

(e) <u>Settlement of Outstanding Mallinckrodt RSUs</u>. Mallinckrodt shall be responsible for all Liabilities associated with Mallinckrodt RSUs including any share delivery, registration or other obligations related to the settlement of the Mallinckrodt RSUs.

(f) <u>Settlement of Outstanding Mallinckrodt PSUs</u>. Mallinckrodt shall be responsible for all Liabilities associated with Mallinckrodt PSUs, including any share delivery, registration or other obligations related to the settlement of Mallinckrodt PSUs.

## Section 4.6 Annual Incentive Plan Payments.

(a) Not later than the Effective Time, Mallinckrodt shall, or shall cause another Mallinckrodt Entity to, assume or adopt a plan (the "<u>Mallinckrodt</u> <u>Annual Incentive Plan</u>") for the fiscal year in which the Distribution occurs that will contain terms that are identical to the terms provided to Mallinckrodt Group Employees under the Covidien Annual Incentive Plan immediately prior to the Effective Time, subject to Mallinckrodt's right to amend such plan after the Effective Time in accordance with the terms thereof.

(b) For the avoidance of doubt, (i) the Covidien Group shall be solely responsible for funding, paying, and discharging all obligations relating to any annual incentive bonus awards that any Covidien Group Employee is eligible to receive under any Covidien annual incentive plan with respect to payments made beginning at or after the Effective Time, and no Mallinckrodt Entity shall have any obligations with respect thereto; and (ii) the Mallinckrodt Group shall be solely responsible for funding, paying, and discharging all obligations relating to any annual incentive bonus awards that any Mallinckrodt Group Employee is eligible to receive under any Mallinckrodt Group Employee is eligible to receive under any Mallinckrodt Group annual incentive plan or other short-term incentive compensation plan with respect to payments made at or after the Effective Time, including, but not limited to, the Mallinckrodt Annual Incentive Plan, and no Covidien Entity shall have any obligations with respect thereto.

Section 4.7 Equity Plan Approval. Covidien and Mallinckrodt shall take all actions as may be necessary or advisable to adopt and obtain shareholder approval of any stock-based employee benefit plans of Mallinckrodt (and the grants of adjusted awards over Covidien shares by Covidien and of awards over Mallinckrodt shares by Mallinckrodt) in order to satisfy the requirement of Rule 16b-3 under the Exchange Act, and the applicable rules and regulations of the NYSE.

## ARTICLE V U.S. QUALIFIED RETIREMENT PLANS

Section 5.1 Pension Plans.

### (a) Mallinckrodt Pension Plans.

(i) To the extent not completed before the Effective Time, effective as of the Distribution Date, a Mallinckrodt Entity shall assume sponsorship of and be solely

-16-

responsible for the management and administration of, and except as otherwise provided below, be responsible for all Assets and Liabilities under the pension plans listed in <u>Schedule 5.1(a)</u> (the "<u>Mallinckrodt Pension Plans</u>"). Mallinckrodt and Covidien shall reasonably cooperate with each other in order to facilitate all actions contemplated by this <u>Section 5.1(a)</u>. Nothing contained in this Agreement shall alter in any way the right of Mallinckrodt, subsequent to the Distribution Date, to amend or terminate any of the Mallinckrodt Pension Plans in accordance with its terms and applicable Law.

(ii) Effective as of the Distribution Date, a Mallinckrodt Entity shall be solely responsible for the adjudication of any claims filed by Mallinckrodt Group Employees or Former Mallinckrodt Group Employees under a Mallinckrodt Pension Plan including, but not limited to, claims filed before the Distribution Date under such plans as in effect as of the date such claim was filed, provided that (A) the claim relates to Assets or Liabilities assumed by Mallinckrodt under Section 5.1(a)(i); (B) the claim has not been finally adjudicated by Covidien on the day immediately preceding the Distribution Date; and (C) under the applicable claims procedure, the applicable Mallinckrodt Entity's plan administrator or other authorized person or committee will have at least a sixty (60)-day period after the Distribution Date to respond to such claim. Covidien shall be solely responsible for the adjudicate the claim, Covidien shall immediately transfer administration of such claim to Mallinckrodt for final adjudication upon sending its response to the claimant.

(b) <u>Covidien Pension Plan</u>. Following the Distribution Date, a Covidien Entity shall retain sole responsibility for all benefits accrued prior to the Distribution Date, Assets and Liabilities for the Covidien Pension Plan and Mallinckrodt shall have no obligation with respect thereto. Nothing contained in this Agreement shall alter in any way the right of Covidien, subsequent to the Distribution Date, to amend or terminate the Covidien Pension Plan in accordance with its terms and applicable Law.

(c) To the extent it is determined by mutual agreement of the Parties following the Distribution Date that any assets relating to the Mallinckrodt Pension Plans or the Covidien Pension Plan either (1) were not transferred to the master trust established on behalf of the Mallinckrodt Pension Plans or Covidien Pension Plan, respectively, by the Distribution Date or (2) were acquired after the Distribution Date by a Party's master trust and such assets should have been or should be allocated to the other Party's master trust, the Parties shall cooperate to ensure that such assets are allocated to the appropriate Party's master trust as soon as practicable following such determination. The determination of which Party's trust shall be the appropriate trust for assets shall be governed by ERISA and shall be made by the named fiduciaries for the respective plans.

### Section 5.2 Defined Contribution Plans.

## (a) Mallinckrodt Savings Plans.

(i) To the extent not completed before the Effective Time, effective as of the Distribution Date, a Mallinckrodt Entity shall assume sponsorship of and be solely

-17-

responsible for the management and administration of all Assets and Liabilities under the Mallinckrodt 401(k) Plan and any other defined contribution plan in the U.S. covering Mallinckrodt Group Employees or Former Mallinckrodt Group Employees as of the Distribution Date and listed in <u>Schedule 5.2(a)</u> (the "<u>Mallinckrodt Savings Plans</u>"). Nothing contained in this Agreement shall alter in any way the right of Mallinckrodt, subsequent to the Distribution Date, to amend or terminate the Mallinckrodt Savings Plans in accordance with its terms and applicable Law.

(ii) Effective as of the Distribution Date, a Mallinckrodt Entity shall be solely responsible for the adjudication of claims filed by Mallinckrodt Group Employees or Former Mallinckrodt Group Employees under a Mallinckrodt Savings Plan, including, but not limited to, claims filed before the Distribution Date under such plans as in effect on the date such claim was filed provided that (A) the claim relates to Assets or Liabilities assumed by Mallinckrodt under this Section 5.2(a); (B) the claim has not been finally adjudicated by Covidien on the day immediately preceding the Distribution Date; and (C) under the applicable claims procedure, the applicable Mallinckrodt Entity's plan administrator or other authorized person or committee will have at least a sixty (60)-day period after the Distribution Date to respond to such claim. Covidien shall be solely responsible for the adjudicate the claim, Covidien shall be solely response to such claim does not finally adjudicate the claim, Covidien shall immediately transfer administration of such claim to Mallinckrodt for final adjudication upon sending its response to the claimant.

(b) <u>Covidien Retained Savings Plans</u>. Following the Distribution Date, a Covidien Entity shall retain sole responsibility for all benefit obligations incurred prior to the Distribution Date and Liabilities under the Covidien 401(k) Plan and the Covidien Retirement Savings and Investment Plan for Puerto Rico Employees and any other defined contribution plan in the U.S. covering Covidien Group Employees (the "<u>Covidien Retained Savings Plans</u>"). Nothing contained in this Agreement shall alter in any way the right of Covidien, subsequent to the Distribution Date, to amend or terminate a Covidien Retained Savings Plan in accordance with its terms and applicable Law.

Section 5.3 Employee Benefit Plan Governance Structure. To the extent not completed before the Effective Time, effective as of the Distribution Date, a Mallinckrodt Entity shall take all such actions as are necessary to (a) establish an employee benefit plan governance structure that includes, at a minimum, an investment committee and administrative committee authorized to serve as named fiduciaries of any Benefit Plan sponsored or maintained by a Mallinckrodt Entity that is governed by ERISA; (b) appoint members of such investment and administrative committees; and (c) establish a new trust or trusts to hold tax-qualified retirement plan assets as required by ERISA and applicable Law. Effective as of the Effective Time, Mallinckrodt Savings Plans and Mallinckrodt Pension Plans.

-18-

### ARTICLE VI U.S. NON-QUALIFIED DEFERRED COMPENSATION PLANS

#### Section 6.1 Mallinckrodt Non-Qualified Deferred Compensation Plans.

(a) To the extent not completed before the Effective Time, effective as of the Distribution Date, a Mallinckrodt Entity shall assume sponsorship of and be solely responsible for the management, administration and satisfaction of all Assets and Liabilities under any non-qualified deferred compensation plan in the U.S. maintained by Covidien or any Subsidiary of Covidien and each other Person that is controlled directly or indirectly by Covidien (including, to the extent applicable, any member of the Mallinckrodt Group) as of the day prior to the Distribution Date, other than the Covidien Non-qualified Plans (the "<u>Mallinckrodt Deferred Compensation Plans</u>"). This Agreement hereby authorizes the transfer of sponsorship of any Mallinckrodt Deferred Compensation Plan that has not been transferred to a Mallinckrodt Entity by the Distribution Date, with Covidien authorizing the transfer of sponsorship on behalf of the applicable Covidien Entity and Mallinckrodt authorizing the acceptance of plan sponsorship on behalf of the applicable Mallinckrodt Entity. Nothing contained in this Agreement shall alter in any way the right of Mallinckrodt, subsequent to the Distribution Date, to amend or terminate a Mallinckrodt Deferred Compensation Plan in accordance with its terms and applicable Law.

(b) All elections by Mallinckrodt Group Employees, and Former Mallinckrodt Group Employees that were in effect immediately prior to the Distribution Date shall continue in effect from and after the Distribution Date until a new election that by its terms supersedes the prior election is made by such Mallinckrodt Group Employee or Former Mallinckrodt Group Employee in accordance with the terms of the applicable Mallinckrodt Deferred Compensation Plan and consistent with the provisions of Code Section 409A, to the extent applicable.

(c) As of the Distribution Date, a Mallinckrodt Entity shall be solely responsible for the adjudication of claims filed by Mallinckrodt Group Employees or Former Mallinckrodt Group Employees under a Mallinckrodt Deferred Compensation Plan before the Distribution Date, provided that (A) the claim relates to Assets or Liabilities assumed by Mallinckrodt under this <u>Section 6.1</u>; (B) the claim has not been finally adjudicated by Covidien as of the day immediately preceding the Distribution Date; and (C) under the applicable claims procedure Mallinckrodt's plan administrator or other authorized person or committee will have at least a sixty (60)-day period after the Distribution Date to respond to such claim. Covidien shall be solely responsible for the adjudication of any claim that satisfies subsections (A) and (B) but not (C); <u>provided</u>, <u>however</u>, that if Covidien's response to such claim does not finally adjudicate the claim, Covidien shall immediately transfer administration of such claim to Mallinckrodt for final adjudication upon sending its response to the claimant.

(d) Payments to Mallinckrodt Group Employees and Former Mallinckrodt Group Employees under a Mallinckrodt Deferred Compensation Plan shall be made by a Mallinckrodt Entity as determined in the sole discretion of Mallinckrodt.

Section 6.2 <u>Covidien Non-Qualified Deferred Compensation Plan</u>. Following the Distribution Date, Covidien shall retain sole responsibility for the satisfaction of all Liabilities under Covidien Non-qualified Plans and all Liabilities with respect to nonqualified deferred compensation plan benefits for Covidien Group Employees and Former Covidien Group Employees.

-19-

Section 6.3 <u>Participation; Distributions</u>. Covidien and Mallinckrodt acknowledge that none of the transactions contemplated by the Separation Agreement or any Ancillary Agreement will trigger a payment or distribution of compensation under any of the Covidien Non-qualified Plans or Mallinckrodt Deferred Compensation Plans for any participant and, consequently, that the payment or distribution of any compensation to which such participant is entitled under any of the Covidien Non-qualified Plans or Mallinckrodt Deferred Compensation Plans will occur upon such participant's separation from service from the Mallinckrodt Group or at such other time as provided in the applicable Mallinckrodt Deferred Compensation Plan or participant's deferral election.

#### ARTICLE VII

## U.S. HEALTH, WELFARE AND FRINGE BENEFIT PLANS

#### Section 7.1 Health Plans.

(a) Effective as of January 1, 2013, a Mallinckrodt Entity established or caused to be established the Mallinckrodt Health Plan. After the Distribution Date, a Mallinckrodt Entity shall be solely responsible for the management and administration of the Mallinckrodt Health Plan, including compliance with COBRA continuation coverage requirements, and solely responsible for the payment of all employer-related costs associated with establishing, administering and maintaining the Mallinckrodt Health Plan, and for the collection and remittance of participant contributions and premium payments.

Except as provided below, a Mallinckrodt Entity shall be solely responsible for the adjudication of any claims filed by a Mallinckrodt Group Employee or Former Mallinckrodt Group Employee before the Distribution Date under the Mallinckrodt Health Plan. Notwithstanding the previous sentence, a Covidien Entity shall be solely responsible for the adjudication of any claims filed by a Mallinckrodt Group Employee or Former Mallinckrodt Group Employee under the Mallinckrodt Health Plan before the Distribution Date that (A) has not been finally adjudicated by a Covidien Entity on the day immediately preceding the Distribution Date; and (B) under the applicable claims procedure, the applicable Covidien Entity's plan administrator or other authorized person or committee will have a less than sixty (60)-day period after the Distribution Date to respond to such claim. Notwithstanding the previous sentence, if Covidien's response to such claim does not finally adjudicate the claim, Covidien shall immediately upon sending its response to the claimant transfer administration of such claim to Mallinckrodt for final adjudication. Any determination made or settlements entered into by a Covidien Entity prior to the Distribution Date with respect to claims incurred under the Mallinckrodt Health Plan by Mallinckrodt Group Employees or Former Mallinckrodt Group Employees shall be final and binding.

-20-

Section 7.2 <u>Section 125 Plans</u>. Effective as of January 1, 2013, a Mallinckrodt Entity established or caused to be established a Mallinckrodt Section 125 Plan. After the Distribution Date, a Mallinckrodt Entity shall be solely responsible for the management and administration of the Mallinckrodt Section 125 Plan.

Section 7.3 <u>Fringe Benefits</u>. Effective as of the Distribution Date and to the extent it is not part of the Mallinckrodt Health Plan, Mallinckrodt shall be responsible for establishing (as necessary) and maintaining its own U.S. fringe benefit plans, policies and arrangements, including any employee assistance program, educational assistance program, adoption assistance program and any other fringe benefit plans, programs and arrangements. Mallinckrodt shall be solely responsible for the management and administration of and assume financial and administrative Liability and all related obligations and responsibilities with respect to claims for such fringe benefits incurred by Mallinckrodt Group Employees and Former Mallinckrodt Group Employees (but not paid by Covidien) prior to the Distribution Date.

Section 7.4 <u>Workers' Compensation</u>. With respect to claims for workers' compensation in the U.S., (a) the Covidien Group shall be responsible for claims in respect of Covidien Group Employees and Former Covidien Group Employees, whether occurring prior to, on or following the Distribution Date and (b) the Mallinckrodt Group shall be responsible for all claims in respect of Mallinckrodt Group Employees and Former Mallinckrodt Group Employees occurring on or following the Distribution Date. For purposes of this <u>Section 7.4</u>, claims shall be deemed to be incurred upon the occurrence of the injury giving rise to such claim.

Section 7.5 <u>Indemnification</u>. Mallinckrodt agrees to hold Covidien harmless with respect to any Liabilities related to actions taken to establish any Benefit Plans and related third party administrative agreements prior to the Distribution Date.

Section 7.6 <u>Termination of Participation</u>. Except as otherwise provided under this Agreement or in any Transition Services Agreements and to the extent that Mallinckrodt Group Employees have not previously ceased participating in a Covidien Benefit Plan, effective as of the Effective Time, Mallinckrodt Group Employees shall cease participating in any Covidien Benefit Plan and shall, thereafter, be ineligible for benefits under any Covidien Benefit Plan.

## ARTICLE VIII

# NON-U.S. EMPLOYEES

To the extent not completed before the Effective Time, as of the Distribution Date, a Mallinckrodt Entity shall take such steps as are necessary or appropriate to adopt and provide benefit plan coverage to Mallinckrodt Group Employees working in Non-U.S. jurisdictions that is similar to the benefit plan coverage provided by a Covidien Entity immediately prior to the date that such Mallinckrodt Entity provides such benefit plan coverage; provided, however, that given the limited number of Mallinckrodt Group Employees in certain jurisdictions and the practical limitations of establishing similar benefit plan coverage in such jurisdictions, such arrangements may be different than benefit plan coverage provided by a Covidien Entity and may be determined by Mallinckrodt in its sole discretion. Mallinckrodt shall indemnify Covidien for any and all claims made by any Mallinckrodt Group Employee that relates to the transition of employment in Non-U.S. jurisdictions in connection with the Distribution and resulting changes to benefit plan coverage.

-21-

### ARTICLE IX GENERAL PROVISIONS

Section 9.1 <u>Preservation of Rights to Amend</u>. The rights of each Covidien Entity and each Mallinckrodt Entity to amend, waive, or terminate any plan, arrangement, agreement, program, or policy referred to herein shall not be limited in any way by this Agreement.

Section 9.2 <u>Fiduciary Matters</u>. Covidien and Mallinckrodt each acknowledges that actions required to be taken pursuant to this Agreement may be subject to fiduciary duties or standards of conduct under ERISA or other applicable Law, and no Party shall be deemed to be in violation of this Agreement if it fails to comply with any provisions hereof based upon its good-faith determination (as supported by advice from counsel experienced in such matters) that to do so would violate such a fiduciary duty or standard. Each Party shall be responsible for taking such actions as are deemed necessary and appropriate to comply with its own fiduciary responsibilities and shall fully release and indemnify the other Party for any Liabilities caused by the failure to satisfy any such responsibility.

Section 9.3 <u>Entire Agreement</u>. This Agreement, together with the documents referenced herein (including the Separation Agreement, the Ancillary Agreements and the plans and agreements referenced herein), constitutes the entire agreement and understanding among the Parties with respect to the subject matter hereof and supersedes all prior written and oral and all contemporaneous oral agreements and understandings with respect to the subject matter hereof. To the extent any provision of this Agreement conflicts with the provisions of the Separation Agreement, the provisions of this Agreement shall be deemed to control with respect to the subject matter hereof.

Section 9.4 <u>Binding Effect; No Third-Party Beneficiaries; Assignment</u>. This Agreement shall inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns. Except as otherwise expressly provided in this Agreement, this Agreement is solely for the benefit of the Parties and should not be deemed to confer upon any third parties any remedy, claim, Liability, reimbursement, cause of action, or other right in excess of those existing without reference to this Agreement. Nothing in this Agreement is intended to amend any employee benefit plan or affect the applicable plan sponsor's right to amend or terminate any employee benefit plan pursuant to the terms of such plan. The provisions of this Agreement are solely for the benefit of the Parties, and no current or former Employee, officer, director, or independent contractor or any other individual associated therewith shall be regarded for any purpose as a third-party beneficiary of this Agreement. This Agreement may not be assigned by any Party, except with the prior written consent of the other Parties.

Section 9.5 <u>Amendment</u>. No provisions of this Agreement shall be deemed waived, amended, supplemented or modified by any party, unless such waiver, amendment, supplement or modification is in writing and signed by the authorized representative of the party against whom it is sought to enforce such waiver, amendment, supplement or modification.

-22-

Section 9.6 <u>Remedies Cumulative</u>. All rights and remedies existing under this Agreement or the Schedules attached hereto are cumulative to, and not exclusive of, any rights or remedies otherwise available.

Section 9.7 <u>Notices</u>. Unless otherwise expressly provided herein, all notices, claims, certificates, requests, demands and other communications hereunder shall be made or given in accordance with the provisions of Section 11.5 of the Separation Agreement.

Section 9.8 <u>Counterparts</u>. This Agreement, including the Schedules hereto and the other documents referred to herein, may be executed in multiple counterparts, each of which when executed shall be deemed to be an original but all of which together shall constitute one and the same agreement.

Section 9.9 <u>Severability</u>. If any provision of this Agreement or any Ancillary Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof or thereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby. Upon such determination, the parties shall negotiate in good faith in an effort to agree upon such a suitable and equitable provision to effect the original intent of the parties.

Section 9.10 <u>Governing Law</u>. The construction, interpretation and performance of this Agreement shall be governed and construed according to the laws of the State of New York, without regard to conflicts of laws principles (other than Section 5-1401 and Section 5-1402 of the General Obligations Law of the State of New York).

Section 9.11 <u>Dispute Resolution</u>. The dispute resolution procedures set forth in Article VIII of the Separation Agreement shall apply to any dispute, controversy or claim (whether sounding in contract, tort or otherwise) that arises out of or relates to this Agreement, any breach or alleged breach hereof, the transactions contemplated hereby (including all actions taken in furtherance of the transactions contemplated hereby on or prior to the date hereof), or the construction, interpretation, enforceability, or validity hereof.

Section 9.12 <u>Performance</u>. Covidien will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement to be performed by any member of the Covidien Group. Mallinckrodt will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement to be performed by any member of the Covidien Group. Mallinckrodt will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement to be performed by any member of the Mallinckrodt Group. Each party (including its permitted successors and assigns) further agrees that it will (a) give timely notice of the terms, conditions and continuing obligations contained in this <u>Section 9.12</u> to all of the other members of its Group, and (b) cause all of the other members of its Group not to take any action or fail to take any such action inconsistent with such party's obligations under this Agreement or the transactions contemplated hereby.

-23-

Section 9.13 <u>Construction</u>. This Agreement shall be construed as if jointly drafted by the Parties and no rule of construction or strict interpretation shall be applied against any Party.

Section 9.14 Effect if Distribution Does Not Occur. Notwithstanding anything in this Agreement to the contrary, if the Separation Agreement is terminated prior to the Effective Time, this Agreement shall be of no further force and effect.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed in their names by a duly authorized officer as of the date first written above.

COVIDIEN PLC

By: Name: Title:

MALLINCKRODT PLC

By:

Name:

Title:

# Schedule 5.1(a)

# LIST OF MALLINCKRODT PENSION PLANS

Mallinckrodt Inc. Retirement Plan Mallinckrodt Inc. Cash Balance Pension Plan Mallinckrodt St. Louis Union Pension Plan Mallinckrodt Greenville Union Pension Plan Liebel-Flarsheim Salaried Pension Plan Liebel-Flarsheim Union Pension Plan Schedule 5.2(a)

# LIST OF MALLINCKRODT SAVINGS PLANS

Mallinckrodt Pharmaceuticals Savings and Investment Plan CNS Therapeutics 401(k) Savings Plan

## MALLINCKRODT PHARMACEUTICALS STOCK AND INCENTIVE PLAN

# **EFFECTIVE AS OF JULY 1, 2013**

This document constitutes part of a prospectus covering securities that have been registered under the United States Securities Act of 1933, as amended.

## MALLINCKRODT PHARMACEUTICALS STOCK AND INCENTIVE PLAN

#### **EFFECTIVE AS OF JULY 1, 2013**

### ARTICLE I

#### PURPOSE

1.1. *Purpose.* The purposes of this Mallinckrodt Pharmaceuticals Stock and Incentive Plan as amended and restated (the "Plan") are to promote the interests of Mallinckrodt public limited company (and any successor thereto) by (i) aiding in the recruitment and retention of Directors and Employees, (ii) providing incentives to Directors and Employees by means of performance-related incentives to achieve short-term and long-term performance goals, (iii) providing Directors and Employees with an opportunity to participate in the growth and financial success of the Company, and (iv) promoting the growth and success of the Company's business by aligning the financial interests of Directors and Employees with that of the other shareholders of the Company. Toward these objectives, the Plan provides for the grant of Stock Options, Stock Appreciation Rights, Annual Performance Bonuses, Long-Term Performance Awards and Other Stock-Based Awards.

1.2. *Effective Date; Shareholder Approval.* The Plan is effective as of the date of the Separation, as defined below. The Plan was approved by the Board of Directors of Mallinckrodt public limited company on [May 22–23], 2013 and by the Company's shareholders on [—], 2013.

#### ARTICLE II DEFINITIONS

For purposes of the Plan, the following terms have the following meanings, unless another definition is clearly indicated by particular usage and context:

"Acquired Company" means any business, corporation or other entity acquired by the Company or any Subsidiary.

"Acquired Grantee" means the grantee of a stock-based award of an Acquired Company and may include a current or former Director of an Acquired Company.

"Annual Performance Bonus" means an Award of cash or Shares granted under Section 4.4 of the Plan that is paid solely on account of the attainment of a specified performance target in relation to one or more Performance Measures.

"*Award*" means any form of incentive or performance award granted under the Plan, whether singly or in combination, to a Participant by the Committee pursuant to any terms and conditions that the Committee may establish and set forth in the applicable Award Certificate. Awards granted under the Plan may consist of:

- (a) *"Stock Options"* awarded pursuant to Section 4.3;
- (b) "Stock Appreciation Rights" awarded pursuant to Section 4.3;
- (c) "Annual Performance Bonuses" awarded pursuant to Section 4.4;
- (d) "Long-Term Performance Awards" awarded pursuant to Section 4.5;
- (e) "Other Stock-Based Awards" awarded pursuant to Section 4.6;
- (f) "Director Awards" awarded pursuant to Section 4.7; and
- (g) "Substitute Awards" awarded pursuant to Section 4.8.

"Award Certificate" means the document issued, either in writing or an electronic medium, by the Committee or its designee to a Participant evidencing the grant of an Award and which contains, in the same or accompanying document, the terms and conditions applicable to such Award.

"Board" means the Board of Directors of the Company.

*"Cause"* means, as to any Employee who is a party to an employment agreement with the Company or any Subsidiary which contains a definition of *"cause,"* as set forth in such employment agreement and, if there is no applicable employment agreements, means an Employee's or Director's (i) substantial failure or refusal to perform duties and responsibilities of his or her job as required by the Company or Subsidiary, (ii) violation of any fiduciary duty owed to the Company or Subsidiary, (iii) conviction of a misdemeanor (other than a traffic offense) or felony, (iv) dishonesty, (v) theft, (vi) violation of Company or Subsidiary rules or policy, or (vii) other egregious conduct, that has or could have a serious and detrimental impact on the Company or Subsidiary and its employees. The Committee (or the Nominating Committee solely with respect to Director Awards), in its sole and absolute discretion, shall determine Cause.

"Change in Control" means the first to occur of any of the following events:

- (a) any "person" (as defined in Section 13(d) and 14(d) of the Exchange Act, excluding for this purpose, (i) the Company or any Subsidiary or (ii) any employee benefit plan of the Company or any Subsidiary (or any person or entity organized, appointed or established by the Company for or pursuant to the terms of any such plan that acquires beneficial ownership of voting securities of the Company), is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act) directly or indirectly of securities of the Company representing more than 30 percent of the combined voting power of the Company's then outstanding securities; provided, however, that no Change in Control will be deemed to have occurred as a result of a change in ownership percentage resulting solely from an acquisition of securities by the Company; or
- (b) persons who, as of the Effective Date constitute the Board (the "Incumbent Directors") cease for any reason (including without limitation, as a result of a tender offer, proxy contest, merger or similar transaction) to constitute at least a majority thereof, provided that any person becoming a Director of the Company subsequent to the Effective Date shall be considered an Incumbent Director if such person's election or nomination for election was approved by a vote of at least 50 percent of the Incumbent Directors; but provided further, that any such person whose initial assumption of office is in connection with an actual or threatened proxy contest relating to the election of members of the Board or other actual or threatened solicitation of proxies or consents by or on behalf of a "person" (as defined in Section 13(d) and 14(d) of the Exchange Act) other than the Board, including by reason of agreement intended to avoid or settle any such actual or threatened contest or solicitation, shall not be considered an Incumbent Director; or
- (c) consummation of a reorganization, merger or consolidation or sale or other disposition of at least 80 percent by value of the assets of the Company (a "Business Combination"), in each case, unless, following such Business Combination, all or substantially all of the individuals and entities who were the beneficial owners of outstanding voting securities of the Company immediately prior to such Business Combination beneficially own directly or indirectly more than 50 percent of the company resulting from such Business Combination (including, without limitation, a company which, as a result of such transaction, owns the Company or all or substantially all of the Company's assets either directly or through one or more Subsidiaries) in substantially the same proportions as their ownership, immediately prior to such Business Combination, of the outstanding voting securities of the Company; or
- (d) approval by the shareholders of the Company of a complete liquidation or dissolution of the Company.

"Change in Control Termination" means a Participant's involuntary termination of employment that occurs during the twelve (12) month period immediately following a Change in Control. For this purpose, subject to Section 7.11(b)(ii), a Participant's involuntary termination of employment includes only the following:

(a) termination of the Participant's employment by the Company for any reason other than for Cause, Disability or death;

- 2 -

- (b) termination of the Participant's employment by the Participant after one of the following events, provided that the Participant's termination of employment occurs within sixty (60) days after the occurrence of any such event:
  - (i) the Company (1) assigns or causes to be assigned to the Participant duties inconsistent in any material respect with his or her position as in effect immediately prior to the Change in Control; (2) makes or causes to be made any material adverse change in the Participant's position (including titles and reporting relationships and level), authority, duties or responsibilities; or (3) takes or causes to be taken any other action which, in the reasonable judgment of the Participant, would cause him or her to violate his or her ethical or professional obligations, or which results in a significant diminution in such position, authority, duties or responsibilities; or
  - (ii) the Company, without the Participant's consent, (1) requires the Participant to relocate to a principal place of employment more than fifty (50) miles from his or her existing place of employment and which increases the Participant's commute from his or her principal residence by more than fifty (50) miles; or (2) reduces the Participant's base salary, annual bonus, or retirement, welfare, share incentive, perquisite (if any) and other benefits taken as a whole;

provided, however, that an event described in (i) or (ii) above shall permit a Participant's termination of employment to be deemed a Change in Control Termination only if written notice of such event has been provided by the Participant to the Company and the Company failed to cure such action within a fifteen (15) day period following receipt of such notice.

"Code" means the United States Internal Revenue Code of 1986, as amended.

"Committee" means the Compensation and Human Resources Committee of the Board or any successor committee or other committee to which the Compensation and Human Resources Committee delegates its authority under this Plan. The Compensation and Human Resources Committee shall be comprised solely of "non-employee directors" within the meaning of Rule 16b-3(b)(3) under the Exchange Act and two or more persons who are outside directors within the meaning of Section 162(m)(4)(C)(i) of the Code and the applicable regulations.

"Company" means Mallinckrodt public limited company, a company incorporated in Ireland under registered number 522227, or any successor thereto.

"Deferred Stock Unit" means a Unit granted under Section 4.6 or 4.7 to acquire Shares upon Termination of Directorship or Termination of Employment, subject to any restrictions that the Committee, in its discretion, may determine.

"Director" means a member of the Board.

"Disabled" or "Disability" means, subject to Section 7.11(b)(iii), that the Employee has a permanent and total incapacity from engaging in any employment for the Company or Subsidiary for physical or mental reasons. A "Disability" shall be deemed to exist if the Employee is designated with an inactive employment status at the end of a disability or medical leave or if the Employee meets the requirements for disability benefits under (i) the Company's or Subsidiary's long-term disability plan or (ii) the Social Security law then in effect, for Employees who are on the payroll of any United States Subsidiary.

"Dividend Equivalent" means an amount equal to the cash dividend or the fair market value of the share dividend that would be paid on each Share underlying an Award if the Share were duly issued and outstanding on the date on which the dividend is payable.

"Effective Date" means July 1, 2013, unless otherwise provided herein.

"Employee" means any individual who performs services as an officer or employee of the Company or a Subsidiary.

"Exchange Act" means the United States Securities Exchange Act of 1934, as amended.

- 3 -

"*Exercise Price*" means the price of a Share, as fixed by the Committee, which may be purchased under a Stock Option or with respect to which the amount of any payment pursuant to a Stock Appreciation Right is determined.

*"Fair Market Value"* of a Share means the closing sales price on the New York Stock Exchange of a Share on the trading day of the grant or on the date as of which the determination of Fair Market Value is being made or, if no sale is reported for such day, on the next preceding day on which a sale of Shares is reported. Notwithstanding anything to the contrary herein, the Fair Market Value of a Share will in no event be determined to be less than par value.

"GAAP" means United States generally accepted accounting principles.

*"Incentive Stock Option"* means a Stock Option granted under Section 4.3 of the Plan that is intended to meet the requirements of Section 422 of the Code and any related regulations and is designated in the Award Certificate as intended to be an Incentive Stock Option.

"Covered Employee" means an Employee who is a "covered employee" within the meaning of Section 162(m)(3) of the Code or who is reasonably expected to be a "covered employee" at the time the Company would be entitled to claim a tax deduction in respect of an Award but for Section 162(m) of the Code.

"Long-Term Performance Award" means an Award granted under Section 4.5 of the Plan that is paid solely on account of the attainment of a specified performance target in relation to one or more Performance Measures or other performance criteria as selected in the sole discretion of the Committee.

"Nominating Committee" means the Nominating and Governance Committee the Board.

"Nonqualified Stock Option" means any Stock Option granted under Section 4.3 of the Plan that is not an Incentive Stock Option.

"Normal Retirement" means Termination of Employment on or after a Participant has attained age 60, provided that the sum of the Participant's age and years of service with the Company or a Subsidiary is 70 or higher.

"Ordinary Shares" means the ordinary shares of the Company, \$0.20 (U.S.) par value, and such other securities or property as may become subject to Awards pursuant to an adjustment made under Section 5.3 of the Plan.

"Other Stock-Based Award" means an Award granted under Section 4.6 of the Plan and denominated in Shares.

"Participant" means a Director, Employee or Acquired Grantee who has been granted an Award under the Plan.

"Performance Cycle" means, with respect to any Award that vests based on Performance Measures, the period of 12 months or longer over which the level of performance will be assessed. The first Performance Cycle under the Plan will begin on such date as is set by the Committee, in its sole discretion.

*"Performance Measure"* means, with respect to any Annual Performance Bonus or Long-Term Performance Award, the business criteria selected by the Committee to measure the level of performance of the Company during a Performance Cycle. The Committee may select as the Performance Measure any operating and maintenance expense targets or financial goals as interpreted by the Committee, either individually, alternatively or in any combination, applied to either the Company as a whole or to a business unit or Subsidiary, either individually, alternatively or in any combination, and that are absolute or relative to the performance of one or more comparable companies or an index of comparable companies, and are measured during the Performance Cycle provided that (i) as to an Annual Performance Bonus or Long-Term Performance Award granted to a Key Employee, Performance Measures shall be limited to the following criteria and (ii) as to an Annual Performance Bonus or Long-Term Performance Award granted to a Participant who is not a Covered Employee, Performance Measures may include, but not be limited to, the following criteria: (a) cash flow, (b) earnings per share, (c) earnings before interest, taxes and amortization, (d) return on equity, (e) total shareholder return, (f) share price performance, (g) return on capital, (h) return on assets or net assets, (i) revenue, (j) income or net income, (k) operating income or net operating income, (l) operating profit or net operating profit, (m) operating margin or profit margin, (n) return on operating revenue, (o) return on invested capital, (p) market segment share, (q) product release schedules, (r) new product innovation, (s) product cost reduction through advanced technology, (t) brand recognition/acceptance, (u) product ship targets, or (v) customer satisfaction.

- 4 -

"Performance Unit" means a Long-Term Performance Award denominated in Units.

"Plan" means this Mallinckrodt Pharmaceuticals Stock and Incentive Plan, as it may be amended from time to time.

"Premium-Priced Stock Option" means a Stock Option the Exercise Price of which is fixed by the Committee at a price that exceeds the Fair Market Value of a Share on the date of grant.

"Reporting Person" means a Director or an Employee who is subject to the reporting requirements of Section 16(a) of the Exchange Act.

"Restricted Stock" means Shares issued pursuant to Section 4.6 that are subject to any restrictions that the Committee, in its discretion, may impose.

"Restricted Unit" means a Unit granted under Section 4.5 or Section 4.6 to acquire Shares or an equivalent amount in cash, which Unit is subject to any restrictions that the Committee, in its discretion, may impose.

"Securities Act" means the United States Securities Act of 1933, as amended.

*"Separation"* means the separation of Covidien plc's Pharmaceuticals business (a/k/a Mallinckrodt Pharmaceuticals) from Covidien plc in a transaction described in a Form 10 initially filed with the U.S. Securities and Exchange Commission on February 1, 2013, whereby the public shareholders of Covidien plc are issued a stock dividend of Mallinckrodt plc ordinary shares.

"Share" means an Ordinary Share of the Company, and "Shares" shall be construed accordingly.

*"Stock Appreciation Right"* means a right granted under Section 4.3 of the Plan of an amount in cash or Shares equal to any excess of the Fair Market Value of a Share as of the date on which the right is exercised over the Exercise Price.

"Stock Option" means a right granted under Section 4.3 of the Plan to purchase from the Company a stated number of Shares at a specified price. Stock Options awarded under the Plan may be in the form of Incentive Stock Options or Nonqualified Stock Options.

*"Subsidiary*" means (i) a subsidiary company (wherever incorporated) of the Company, as defined by Section 155 of the Companies Act 1963 of Ireland; (ii) any separately organized business unit, whether or not incorporated, of the Company; (iii) any employer that is required to be aggregated with the Company pursuant to Code Section 414 and the regulations promulgated thereunder; and (iv) any service recipient or employer that is within a controlled group of corporations as defined in Code Sections 1563(a)(1), (2) and (3) which includes the Company, where the phrase "at least 50%" is substituted in each place "at least 80%" appears, and any service recipient or employer within trades or businesses under common control as defined in Code Section 414(c)-2, which includes the Company, where the phrase "at least 50%" is substituted in each place "at least 80%" appears, provided, however, that when the relevant determination is to be based upon legitimate business criteria (as described in Treas. Reg. § 1.409A-1(b) (5)(iii)(E) and § 1.409A-1(h)(3)), the phrase "at least 20%" shall be substituted in each place "at least 80%" appears as described above with respect to both a controlled group of corporations and trades or business under common control.

"Target Amount" means the amount of Performance Units that will be paid if the applicable Performance Measure is fully (100%) attained, as determined in the sole discretion of the Committee.

"Target Bonus" means the target Annual Performance Bonus applicable to a Reporting Person in respect of a particular year, as established by the Committee or its delegate.

- 5 -

*"Target Vesting Percentage"* means the percentage of performance-based Restricted Units or Shares of Restricted Stock that will vest if the applicable Performance Measure is fully (100%) attained, as determined in the sole discretion of by the Committee.

*"Termination of Directorship"* means the date of cessation of a Director's membership on the Board for any reason, with or without Cause, as determined in the sole discretion of the Nominating Committee, provided however that if the Director is a member of the Nominating Committee, such determination shall be made by the full Board (excluding such Director).

*"Termination of Employment"* means the date of cessation of an Employee's employment relationship with the Company or a Subsidiary for any reason, with or without Cause, as determined in the sole discretion of the Company.

"Unit" means, for purposes of Performance Units, the potential right to an Award equal to a specified amount denominated in such form as is deemed appropriate in the discretion of the Committee and, for purposes of Restricted Units or Deferred Stock Units, the potential right to acquire one Share.

### ARTICLE III ADMINISTRATION

3.1. Committee. The Plan will be administered by the Committee, except as otherwise provided in Section 4.7.

3.2. Authority of the Committee. The Committee or, to the extent required by applicable law, the Board will have the authority, in its sole and absolute discretion and subject to the terms of the Plan, to:

- (a) Interpret and administer the Plan and any instrument or agreement relating to the Plan;
- (b) Prescribe the rules and regulations that it deems necessary for the proper operation and administration of the Plan, and amend or rescind any existing rules or regulations relating to the Plan;
- (c) Select Employees to receive Awards under the Plan;
- (d) Determine the form of an Award, the number of Shares subject to each Award, all the terms and conditions of an Award, including, without limitation, the conditions on exercise or vesting, the designation of Stock Options as Incentive Stock Options or Nonqualified Stock Options, and the circumstances under which an Award may be settled in cash or Shares or may be cancelled, forfeited or suspended, and the terms of each Award Certificate;
- (e) Determine whether Awards will be granted singly, in combination or in tandem;
- (f) Establish and interpret Performance Measures (or, as applicable, other performance criteria) in connection with Annual Performance Bonuses and Long-Term Performance Awards, evaluate the level of performance over a Performance Cycle and certify the level of performance attained with respect to Performance Measures (or other performance criteria, as applicable);
- (g) Subject to Sections 6.1 and 7.12, waive or amend any terms, conditions, restriction or limitation on an Award, except that the prohibition on the repricing of Stock Options and Stock Appreciation Rights, as described in Section 4.3(g), may not be waived;
- (h) Make any adjustments to the Plan (including but not limited to adjustment of the number of Shares available under the Plan or any Award) and any Award granted under the Plan as shall be appropriate pursuant to Section 5.3;
- (i) Determine and set forth in the applicable Award Certificate the circumstances under which Awards may be deferred and the extent to which a deferral will be credited with Dividend Equivalents and interest thereon;
- (j) Determine and set forth in the applicable Award Certificate whether a Nonqualified Stock Option or Restricted Share may be transferable to family members, a family trust or a family partnership;

- 6 -

- (k) Establish any subplans and make any modifications to the Plan, without amending the Plan, or to Awards made hereunder (including the establishment of terms and conditions in the Award Certificate not otherwise inconsistent with the terms of the Plan) that the Committee may determine to be necessary or advisable for grants made in countries outside the United States to comply with, or to achieve favorable tax treatment under, applicable foreign laws or regulations or tax policies or customs;
- (I) Appoint such agents as it shall deem appropriate for the proper administration of the Plan; and
- (m) Take any and all other actions it deems necessary or advisable for the proper operation or administration of the Plan.

3.3. Effect of Determinations. All determinations of the Committee will be final, binding and conclusive on all persons having an interest in the Plan.

3.4. Delegation of Authority. The Board or, if permitted under applicable corporate law, the Committee, in its discretion and consistent with applicable law and regulations, may delegate to a committee or an officer or group of officers, as it deems to be advisable, the authority to select Employees to receive an Award and to determine the number of Shares under any such Award, subject to any terms and conditions that the Board or the Committee may establish. When the Board or the Committee delegates authority pursuant to the foregoing sentence, it will limit, in its discretion, the number or value of Shares that may be subject to Awards that the delegate may grant. Only the Committee has the authority to grant and administer Awards to Covered Employees and other Reporting Persons or to delegates of the Committee, and to establish and certify Performance Measures.

3.5. *Employment of Advisors*. The Committee may employ attorneys, consultants, accountants and other advisors, the fees and other expenses of which shall be paid by the Company, and the Committee, the Company and the officers and directors of the Company may rely upon the advice, opinions or valuations of the advisors employed.

3.6. *No Liability.* No member of the Committee or any person acting as a delegate of the Committee with respect to the Plan will be liable for any losses resulting from any action, interpretation or construction made in good faith with respect to the Plan or any Award granted under the Plan.

## ARTICLE IV AWARDS

4.1. *Eligibility*. All Participants and Employees are eligible to be designated to receive Awards granted under the Plan, except as otherwise provided in this Article IV.

4.2. Form of Awards. Awards will be in the form determined by the Committee, in its discretion, and will be evidenced by an Award Certificate. Awards may be granted singly or in combination or in tandem with other Awards.

4.3. *Stock Options and Stock Appreciation Rights.* The Committee may grant Stock Options and Stock Appreciation Rights under the Plan to those Employees whom the Committee may from time to time select, in the amounts and pursuant to the other terms and conditions that the Committee, in its discretion, may determine and set forth in the Award Certificate, subject to the provisions below:

- (a) Form. Stock Options granted under the Plan will, at the discretion of the Committee and as set forth in the Award Certificate, be in the form of Incentive Stock Options, Nonqualified Stock Options or a combination of the two. If an Incentive Stock Option and a Nonqualified Stock Option are granted to the same Participant under the Plan at the same time, the form of each will be clearly identified, and they will be deemed to have been granted in separate grants. In no event will the exercise of one Stock Option affect the right to exercise the other Stock Option. Stock Appreciation Rights may be granted either alone or concurrently with Nonqualified Stock Options and the amount of Shares attributable to each Stock Appreciation Right shall be set forth in the applicable Award Certificate on or before the grant date.
- (b) *Exercise Price*. The Committee will set the Exercise Price of Stock Options (other than Premium-Priced Stock Options or certain Incentive Stock Options as described below) or Stock Appreciation Rights granted

- 7 -

under the Plan at a price that is equal to the Fair Market Value of a Share on the date of grant, subject to adjustment as provided in Section 5.3. The Exercise Price of Incentive Stock Options will be equal to or greater than 110 percent of the Fair Market Value of a Share as of the date of grant if the Participant receiving the Incentive Stock Options owns shares possessing more than 10 percent of the total combined voting power of all classes of shares of the Company or any subsidiary or parent corporation of the Company, as defined in Section 424 of the Code. The Exercise Price of a Stock Appreciation Right granted in tandem with a Stock Option will equal the Exercise Price of the related Stock Option. The Committee will set forth the Exercise Price of a Stock Option or Stock Appreciation Right in the Award Certificate or accompanying documentation.

- (c) *Term and Timing of Exercise*. Each Stock Option or Stock Appreciation Right granted under the Plan will be exercisable in whole or in part, subject to the following conditions, unless determined otherwise by the Committee:
  - (i) The term of each Stock Option shall be determined by the Committee and set forth in the applicable Award Certificate, but in no event shall the term of a Stock Option exceed ten (10) years from the date of its grant.
  - A Stock Option or Stock Appreciation Right will become exercisable at such times and in such manner as determined by the Committee and set forth in the applicable Award Certificate.
  - (iii) Unless the applicable Award Certificate provides otherwise, upon the death, Disability, Normal Retirement or a Change in Control Termination of a Participant who has outstanding Stock Options or Stock Appreciation Rights, the unvested Stock Options or Stock Appreciation Rights will fully vest. Unless the applicable Award Certificate or the remainder of this Section 4.3(c) provides otherwise, the Participant's Stock Options and Stock Appreciation Rights will lapse, and will not thereafter be exercisable, upon the earlier of (A) their original expiration date or (B) the date that is three (3) years after the date on which the Participant dies, incurs a Disability or retires due to Normal Retirement.
  - (iv) Unless the applicable Award Certificate provides otherwise, upon the Termination of Employment of a Participant for any reason other than the Participant's death, Disability, Normal Retirement or a Change in Control Termination, if the Participant has attained age 55 and the sum of the Participant's age and years of service with the Company or a Subsidiary is 60 or higher, a pro rata portion of the Participant's Stock Options and Stock Appreciation Rights will vest so that the total number of vested Stock Options or Stock Appreciation Rights held by the Participant at Termination of Employment (including those that have already vested as of such date) will be equal to the total number of Stock Options or Stock Appreciation Rights originally granted to the Participant under the applicable Award multiplied by a fraction, the numerator of which is the period of time (in whole months) that have elapsed since the date of grant, and the denominator of which is the number of months set forth in the applicable Award Certificate that is required to attain full vesting. Unless the Award Certificate provides otherwise, such Participant's Stock Options and Stock Appreciation Rights will lapse, and will not thereafter be exercisable, upon the earlier of (A) their original expiration date or (B) the date that is three (3) years after the date of Termination of Employment.
  - (v) Unless the applicable Award Certificate provides otherwise, upon the Termination of Employment of a Participant that does not meet the requirements of paragraphs (iii) or (iv) above, any unvested Stock Options or Stock Appreciation Rights will be forfeited. Unless the applicable Award Certificate provides otherwise, any Stock Options or Stock Appreciation Rights that are vested as of such Termination of Employment will lapse, and will not thereafter be exercisable, upon the earlier of (A) their original expiration date or (B) the date that is ninety (90) days after the date of such Termination of Employment.
  - (vi) Stock Options and Stock Appreciation Rights of a deceased Participant may be exercised only by the estate of the Participant or by the person given authority to exercise the Stock Options or Stock Appreciation Rights by the Participant's will or by operation of law. If a Stock Option or Stock Appreciation Right is exercised by the executor or administrator of a deceased Participant, or by the person or persons to whom the Stock Option or Stock Appreciation Right has been transferred by the Participant's will or the applicable laws of descent and distribution, the Company will be under no

- 8 -

obligation to deliver Shares or cash until the Company is satisfied that the person exercising the Stock Option or Stock Appreciation Right is the duly appointed executor or administrator of the deceased Participant or the person to whom the Stock Option or Stock Appreciation Right has been transferred by the Participant's will or by applicable laws of descent and distribution.

- (vii) A Stock Appreciation Right granted in tandem with a Stock Option is subject to the same terms and conditions as the related Stock Option and will be exercisable only to the extent that the related Stock Option is exercisable. When either a Stock Option or a Stock Appreciation Right granted in tandem with each other is exercised, the tandem Stock Option or Stock Appreciation Right, as applicable, shall expire.
- (d) Payment of Exercise Price. The Exercise Price of a Stock Option must be paid in full when the Stock Option is exercised. Shares will be issued and delivered only upon receipt of payment. Payment of the Exercise Price may be made in cash or by certified check, bank draft, wire transfer, or postal or express money order, provided that the format is approved by the Company or a designated third-party administrator. The Committee, in its discretion may also allow payment to be made by any of the following methods, as set forth in the applicable Award Certificate:
  - (i) Delivering a properly executed exercise notice to the Company or its agent, together with irrevocable instructions to a broker to deliver to the Company, within the typical settlement cycle for the sale of equity securities on the relevant trading market (or otherwise in accordance with the provisions of Regulation T issued by the Federal Reserve Board), the amount of sale proceeds with respect to the portion of the Shares to be acquired having a Fair Market Value on the date of exercise equal to the sum of the applicable portion of the Exercise Price being so paid;
  - (ii) Subject to any requirements of applicable law and regulations, tendering (actually or by attestation) to the Company or its agent previously acquired Shares that have a Fair Market Value on the day prior to the date of exercise equal to the applicable portion of the Exercise Price being so paid; or
  - (iii) Subject to any requirements of applicable law and regulations, instructing the Company to reduce the number of Shares that would otherwise be issued by such number of Shares as have in the aggregate a Fair Market Value on the date of exercise equal to the applicable portion of the Exercise Price being so paid.
- (e) *Incentive Stock Options*. Incentive Stock Options granted under the Plan will be subject to the following additional conditions, limitations and restrictions:
  - (i) *Eligibility.* Incentive Stock Options may be granted only to Employees of the Company or a Subsidiary that is a subsidiary or parent corporation of the Company within the meaning of Code Section 424.
  - (ii) *Timing of Grant.* No Incentive Stock Option will be granted under the Plan after the 10-year anniversary of the date on which the Plan is adopted by the Board or, if earlier, the date on which the Plan was approved by shareholders.
  - (iii) Amount of Award. Subject to Section 5.3 of the Plan, no more than 10 million Shares may be available for grant in the form of Incentive Stock Options. The aggregate Fair Market Value (as of the date of grant) of the Shares with respect to which the Incentive Stock Options awarded to any Employee first become exercisable during any calendar year may not exceed \$100,000 (U.S.). For purposes of this \$100,000 (U.S.) limit, the Employee's Incentive Stock Options under this Plan and all other plans maintained by the Company and its Subsidiaries will be aggregated. To the extent any Incentive Stock Option would exceed the \$100,000 (U.S.) limit, the Incentive Stock Option will afterwards be treated as a Nonqualified Stock Option to the extent required by the Code and underlying regulations and rulings.
  - (iv) Timing of Exercise. If the Committee exercises its discretion in the Award Certificate to permit an Incentive Stock Option to be exercised by a Participant more than three months after the Participant has ceased being an Employee (or more than 12 months if the Participant is permanently and totally disabled, within the meaning of Code Section 22(e)), the Incentive Stock Option will afterwards be treated as a Nonqualified Stock Option to the extent required by the Code and underlying regulations and rulings. For

- 9 -

purposes of this paragraph (iv), an Employee's employment relationship will be treated as continuing intact while the Employee is on military leave, sick leave or another approved leave of absence if the period of leave does not exceed 90 days, or a longer period to the extent that the Employee's right to reemployment with the Company or a Subsidiary is guaranteed by statute or by contract. If the period of leave exceeds 90 days and the Employee's right to reemployment is not guaranteed by statute or contract, the employment relationship will be deemed to have ceased on the 91st day of the leave.

- (v) Transfer Restrictions. In no event will the Committee permit an Incentive Stock Option to be transferred by an Employee other than by will or the laws of descent and distribution, and any Incentive Stock Option awarded under this Plan will be exercisable only by the Employee during the Employee's lifetime.
- (f) Exercise of Stock Appreciation Rights. Upon exercise of a Participant's Stock Appreciation Rights, the Company will pay cash or Shares or a combination of cash and Shares, in the discretion of the Committee and as described in the Award Certificate. Cash payments will be equal to the excess of the Fair Market Value of a Share on the date of exercise over the Exercise Price, for each Share for which a Stock Appreciation Right was exercised. If Shares are paid for the Stock Appreciation Right, the Participant will receive a number of whole Shares equal to the quotient of the cash payment amount divided by the Fair Market Value of a Share on the date of exercise.
- (g) No Repricing. Except as otherwise provided in Section 5.3, in no event will the Committee decrease the Exercise Price of a Stock Option or Stock Appreciation Right after the date of grant or cancel outstanding Stock Options or Stock Appreciation Rights and issue cash in exchange for such cancellation or grant replacement Stock Options or Stock Appreciation Rights with a lower Exercise Price than that of the replaced Stock Options or Stock Appreciation Rights or other Awards without first obtaining the approval of the holders of a majority of the Shares who are present in person or by proxy at a meeting of the Company's shareholders and entitled to vote.

4.4. Annual Performance Bonuses. The Committee may grant Annual Performance Bonuses under the Plan in the form of cash or Shares to the Reporting Persons that the Committee may from time to time select, in the amounts and pursuant to the terms and conditions that the Committee may determine and set forth in the Award Certificate, subject to the provisions below:

- (a) *Performance Cycles*. Annual Performance Bonuses will be awarded in connection with a twelve (12) month Performance Cycle, which will be the fiscal year of the Company.
- (b) Eligible Participants. Within ninety (90) days after the commencement of a Performance Cycle, the Committee will determine the Reporting Persons who will be eligible to receive an Annual Performance Bonus under the Plan. If an individual becomes a Reporting Person after this ninety (90) day period, the Committee may determine that such Reporting Person is eligible to receive a pro rata Annual Performance Bonus under the Plan.
- (c) Performance Measures; Targets; Award Criteria.
  - (i) Within ninety (90) days after the commencement of the service period to which a Performance Cycle relates, the Committee will fix and establish in writing (A) the Performance Measures that will apply to that Performance Cycle; (B) the Target Bonus which may be earned by each Participant; and (C) subject to subsection (d) below, the criteria for computing the amount that will be paid with respect to each level of attained performance. The Committee will also set forth the minimum level of performance, based on the applicable Performance Measures, that must be attained during the Performance Cycle before any Annual Performance Bonus will be paid and the percentage of the Target Bonus that will become payable upon attainment of various levels of performance that equal or exceed the minimum required level.
  - (ii) The Committee, in its discretion, may, on a case-by-case basis, reduce, but not increase, the amount otherwise payable to any Covered Employee with respect to any given Performance Cycle, provided, however, that no reduction will result in an increase in the amount payable under any Annual Performance Bonus of another Covered Employee.

- 10 -

- (d) Payment, Certification. No Annual Performance Bonus will be paid to any Reporting Person until the Committee certifies in writing the level of performance attained for the Performance Cycle in relation to the applicable Performance Measures. In applying Performance Measures, the Committee (i) shall make adjustments for events listed in Section 5.3 in accordance therewith and (ii) may, in its discretion, exclude the effect of unusual or infrequently occurring items, the cumulative effect of changes in the law, regulations or accounting rules, and other items, all determined in accordance with GAAP (to the extent applicable) and identified in financial statements, notes to the financial statements or discussion and analysis of management; provided that the determination by the Committee that Performance Measures shall be adjusted for items in accordance with this clause (ii) shall be made no later than ninety (90) days after the commencement of any applicable Performance Cycle in respect of Annual Performance Bonuses awarded to Covered Employees.
- (e) Form of Payment. Annual Performance Bonuses will be paid in cash or Shares. All such Performance Bonuses shall be paid no later than the 15th day of the third month following the end of the calendar year (or, if later, following the end of the Company's fiscal year) in which such Performance Bonuses are no longer subject to a substantial risk of forfeiture (as determined for purposes of Section 409A of the Code), except to the extent that a Participant has elected to defer payment under the terms of a duly authorized deferred compensation arrangement, in which case the terms of such arrangement shall govern.
- (f) Section 162(m) of the Code. It is the intent of the Company that Annual Performance Bonuses made to Covered Employees be "performance-based compensation" for purposes of Section 162(m) of the Code, that this Section 4.4 be interpreted in a manner that satisfies the applicable requirements of Section 162(m)(4)(C) of the Code and related regulations, and that the Plan be operated so that the Company may take a full tax deduction for Annual Performance Bonuses. If any provision of this Plan or any Annual Performance Bonus would otherwise frustrate or conflict with this intent, the provision will be interpreted and deemed amended so as to avoid this conflict.
- (g) Acceleration. Each Participant who is eligible to receive an Annual Performance Bonus with respect to a Performance Cycle during which a Change of Control occurs will, except as otherwise provided below, be deemed to have achieved a level of performance, as of the date of Change in Control, that would cause all (100%) of the Participant's Target Bonus to become payable at such times and in such manner as determined in the sole discretion of the Committee. Notwithstanding the previous sentence, if (i) a surviving entity maintains the Performance Cycle in which a Change in Control occurs, or otherwise provides for the payment of an Annual Performance Bonus based on the level of performance attained for such Performance Cycle in relation to the Performance Measures established for such Performance Cycle (including Performance Measures that were adjusted or modified as a result of the Change in Control) and (ii) the Annual Performance Bonus based on the level of performance attained for such Performance Cycle exceeds all (100%) of the Participant's Target Bonus, then each Participant who is eligible to receive an Annual Performance Bonus with respect to such Performance Cycle shall receive an Annual Performance Bonus based on the level of performance attained for such Performance Bonus with respect to such Performance Cycle shall receive an Annual Performance Bonus based on the level of performance attained for such Performance Cycle at such times and in such manner as determined in the sole discretion of the Committee, or successor to the Committee. Notwithstanding the above, the time and manner of any payments made pursuant to this Section 4.4(g) shall comply with Section 4.4(e) above.

4.5. *Long-Term Performance Awards*. The Committee may grant Long-Term Performance Awards under the Plan in the form of Performance Units, Restricted Units or Restricted Stock to any Employee who the Committee may from time to time select, in the amounts and pursuant to the terms and conditions that the Committee may determine and set forth in the Award Certificate, subject to the provisions below:

(a) *Performance Cycles.* Long-Term Performance Awards will be awarded in connection with a Performance Cycle, as determined by the Committee in its discretion, provided, however, that a Performance Cycle may be no shorter than twelve (12) months.

- 11 -

- (b) Eligible Participants. Within ninety (90) days after the commencement of a Performance Cycle, the Committee will determine the Employees who will be eligible to receive a Long-Term Performance Award for the Performance Cycle, provided that the Committee may determine the eligibility of any Employee other than a Covered Employee after the expiration of this ninety (90) day period.
- (c) Performance Measures; Targets; Award Criteria.
  - (i) Within ninety (90) days after the commencement of the service period to which a Performance Cycle relates, the Committee will fix and establish in writing (A) the Performance Measures that will apply to that Performance Cycle; (B) with respect to Performance Units, the Target Amount payable to each Participant; (C) with respect to Restricted Units and Restricted Stock, the Target Vesting Percentage for each Participant; and (D) subject to subsection (d) below, the criteria for computing the amount that will be paid or will vest with respect to each level of attained performance. The Committee will also set forth the minimum level of performance, based on the applicable Performance Measures, that must be attained during the Performance Cycle before any Long-Term Performance Award will be paid or vest, and the percentage of Performance Units that will become payable and the percentage of performance-based Restricted Units or Shares of Restricted Stock that will vest upon attainment of various levels of performance that equal or exceed the minimum required level.
  - (ii) The Committee, in its discretion, may, on a case-by-case basis, reduce, but not increase, the amount of Long-Term Performance Awards otherwise payable to any Covered Employee with respect to any given Performance Cycle, provided, however, that no reduction will result in an increase in the dollar amount or number of Shares payable under any Long-Term Performance Award of another Covered Employee.
- (d) Payment, Certification. Long-Term Performance Awards shall vest and be paid within the sixty (60) day period following the end of the applicable Performance Cycle, and shall only be paid if, within such sixty (60) day period, the Committee certifies in writing the level of performance attained for the Performance Cycle in relation to the applicable Performance Measures. Long-Term Performance Awards awarded to Participants who are not Covered Employees will be based on the Performance Measures, or other applicable performance criteria, and payment formulas that the Committee, in its discretion, may establish for these purposes. These Performance Measures, or other performance criteria, and formulas may be the same as or different than the Performance Measures and formulas that apply to Covered Employees. In applying Performance Measures, the Committee (i) shall make adjustments for events listed in Section 5.3 in accordance therewith and (ii) may, in its discretion, exclude the effect of unusual or infrequently occurring items, the cumulative effect of changes in the law, regulations or accounting rules, and other items, all determined in accordance with GAAP (to the extent applicable) and identified in financial statements, notes to the financial statements or discussion and analysis of management; provided that the determination by the Committee that Performance Measures shall be adjusted for items in accordance with this clause (ii) shall be made no later than ninety (90) days after the commencement of any applicable Performance Cycle in respect of Long-Term Performance Awards awarded to Covered Employees.
- (e) Form of Payment. Long-Term Performance Awards in the form of Performance Units may be paid in cash or full Shares, in the discretion of the Committee, and as set forth in the applicable Award Certificate. Performance-based Restricted Units and Restricted Stock will be paid in full Shares. Payment with respect to any fractional Share will be in cash in an amount based on the Fair Market Value of the Share as of the date the Performance Unit becomes payable. All Long-Term Performance Awards shall be paid no later than the 15th day of the third month following the end of the calendar year (or, if later, following the end of the Company's fiscal year) in which such Long-Term Performance Awards are no longer subject to a substantial risk of forfeiture (within the meaning of Code Section 409A), except to the extent that a Participant has elected to defer payment under the terms of a duly authorized deferred compensation arrangement, in which case the terms of such arrangement shall govern, or as otherwise provided in Section 4.5(g) below.
- (f) Section 162(m) of the Code. It is the intent of the Company that Long-Term Performance Awards made to Covered Employees be "performance-based compensation" for purposes of Section 162(m) of the Code, that this Section 4.5 be interpreted in a manner that satisfies the applicable requirements of Section 162(m)(4)(C) of the Code and related regulations with respect to Long-Term Performance awards made to Covered Employees, and that the Plan be operated so that the Company may take a full tax deduction for Long-Term

- 12 -

Performance Awards. If any provision of this Plan or any Long-Term Performance Award would otherwise frustrate or conflict with this intent, the provision will be interpreted and deemed amended so as to avoid this conflict.

(g) Special Vesting Provisions. Unless the applicable Award Certificate provides otherwise, upon the death, Disability, Normal Retirement or a Change in Control Termination of a Participant who has an outstanding Long-Term Performance Award, the unvested Long-Term Performance Award will fully vest and be paid as if the Participant had continued in active employment with the Company through the date such Long-Term Performance Award would have vested and been paid in the absence of such event. Unless the applicable Award Certificate provides otherwise, upon the Termination of Employment of a Participant for any reason other than the Participant's death, Disability, Normal Retirement or a Change in Control Termination, the unvested Long-Term Performance Award will be forfeited unless the Participant has attained age 55 and the sum of the Participant's age and years of service with the Company or a Subsidiary is 60 or higher, in which case a pro rata portion of the Participant's Long-Term Performance Award would have vested and been paid as if the Participant had continued in active employment with the Company through the date such Long-Term Performance Award would have vested and been paid in the absence of such event; provided that the number of Long-Term Performance Awards held by the Participant which shall vest under those circumstances shall equal the total number of Long-Term Performance Awards in which such Participant would have vested multiplied by a fraction, the numerator of which is the period of time (in whole months) that have elapsed since the date of grant, and the denominator of which is the number of total months set forth in the applicable Award Certificate for such Performance Period.

4.6. *Other Stock-Based Awards*. The Committee may, from time to time, grant Awards (other than Stock Options, Stock Appreciation Rights, Annual Performance Bonuses or Long-Term Performance Awards) to any Employee who the Committee may from time to time select, which Awards consist of, or are denominated in, payable in, valued in whole or in part by reference to, or otherwise related to, Shares. These Awards may include, among other forms, Restricted Stock, Restricted Units, or Deferred Stock Units. The Committee will determine, in its discretion, the terms and conditions that will apply to Awards granted pursuant to this Section 4.6, which terms and conditions will be set forth in the applicable Award Certificate.

- (a) Vesting. Restrictions on Other Stock-Based Awards granted under this Section 4.6 will lapse at such times and in such manner as determined by the Committee and set forth in the applicable Award Certificate. Unless the applicable Award Certificate provides otherwise, if the restrictions on Other Stock-Based Awards have not lapsed or been satisfied as of the Participant's Termination of Employment, the Shares will be forfeited by the Participant if the termination is for any reason other than the Normal Retirement, death or Disability of the Participant or a Change in Control Termination, except that the Award will vest pro rata with respect to the portion of the vesting term set forth in the applicable Award Certificate that the Participant has completed if the Participant has attained age 55 and the sum of the Participant's age and years of service with the Company is 60 or higher. All restrictions on Other Stock-Based Awards granted pursuant to this Section 4.6 will lapse upon the Normal Retirement, death or Disability of the Participant or a Change in Control Termination.
- (b) Grant of Restricted Stock. The Committee may grant Restricted Stock to any Employee, which Shares will be registered in the name of the Participant and held for the Participant by the Company. The Participant will have all rights of a shareholder with respect to the Shares, including the right to vote and to receive dividends or other distributions (subject to Section 4.6(e)), except that the Shares may be subject to a vesting schedule and will be forfeited if the Participant attempts to sell, transfer, assign, pledge or otherwise encumber or dispose of the Shares before the restrictions are satisfied or lapse.
- (c) Grant of Restricted Units. The Committee may grant Restricted Units to any Employee, which Units will be paid in cash or whole Shares or a combination of cash and Shares, in the discretion of the Committee, when the restrictions on the Units lapse and any other conditions set forth in the Award Certificate have been satisfied. For each Restricted Unit that vests, one Share will be paid or an amount in cash equal to the Fair Market Value of a Share as of the date on which the Restricted Unit vests.
- (d) Grant of Deferred Stock Units. The Committee may grant Deferred Stock Units to any Employee, which Units will be paid in whole Shares upon the Employee's Termination of Employment if the restrictions on the Units have lapsed. One Share will be paid for each Deferred Stock Unit that becomes payable.

- 13 -

(e) Dividends and Dividend Equivalents. At the discretion of the Committee and as set forth in the applicable Award Certificate, dividends paid on Shares may be paid immediately or withheld and deferred in the Participant's account. In the event of a payment of dividends on the Ordinary Shares, the Committee may credit Restricted Units with Dividend Equivalents in accordance with terms and conditions established in the discretion of the Committee. Dividend Equivalents will be subject to such vesting terms as is determined by the Committee and may be distributed immediately or withheld and deferred in the Participant's account as determined by the Committee and set forth in the applicable Award Certificate. Deferred Stock Units may, in the discretion of the Committee and as set forth in the Award Certificate, be credited with Dividend Equivalents or additional Deferred Stock Units. The number of any Deferred Stock Units credited to a Participant's account upon the payment of a dividend will be equal to the quotient produced by dividing the cash value of the dividend by the Fair Market Value of one Share as of the date the dividend is paid. The Committee will determine any terms and conditions on deferral of a dividend or Dividend Equivalent, including the rate of interest to be credited on deferral and whether interest will be compounded.

### 4.7. Director Awards.

- (a) Notwithstanding anything herein to the contrary, the Nominating Committee shall have the exclusive authority to issue awards to Directors who are not also employees of the Company or any Subsidiary (Director Awards), which may consist of, but not be limited to, Stock Options, Stock Appreciation Rights, or Other Stock-Based Awards. Each Director Award shall be governed by an Award Certificate approved by the Nominating Committee.
- (b) The Nominating Committee shall have the exclusive authority to administer Director Awards, and shall have the authority set forth in Section 3.2 and the indemnification set forth in Section 7.7, solely as such provisions apply to the Director Awards. All determinations made by the Nominating Committee hereunder shall be final, binding and conclusive.

4.8. Substitute Awards. The Committee may make Awards under the Plan to Acquired Grantees through the assumption of, or in substitution for, outstanding stock-based awards previously granted to such Acquired Grantees. Such assumed or substituted Awards will be subject to the terms and conditions of the original awards made by the Acquired Company, with such adjustments therein as the Committee considers appropriate to give effect to the relevant provisions of any agreement for the acquisition of the Acquired Company. Any grant of Incentive Stock Options pursuant to this Section 4.8 will be made in accordance with Section 424 of the Code and any final regulations published thereunder.

4.9. *Limit on Individual Grants.* Subject to Sections 5.1 and 5.3, no Employee may be granted more than six (6) million Shares over any calendar year pursuant to Awards of Stock Options, Stock Appreciation Rights and performance-based Restricted Stock and Restricted Units, except that an incentive Award of no more than ten (10) million Shares may be made pursuant to Stock Options, Stock Appreciation Rights and performance-based Restricted Stock and Performance-based Restricted Stock and Restricted Units to any person who has been hired within the calendar year as a Covered Employee. The maximum amount that may be paid in cash or Shares pursuant to Annual Performance Bonuses or Long-Term Performance Awards paid in Performance Units to any one Employee is \$15 million (U.S.) for any Performance Cycle of twelve (12) months. For any longer Performance Cycle, this maximum will be adjusted proportionally.

4.10. *Termination for Cause*. Notwithstanding anything to the contrary herein and unless the applicable Award Certificate provides otherwise, if a Participant incurs a Termination of Directorship or Termination of Employment for Cause, then all Stock Options, Stock Appreciation Rights, Annual Performance Bonuses, Long-Term Performance Awards, Restricted Units, Restricted Stock and Other Stock-Based Awards will immediately be cancelled. The exercise of any Stock Option or Stock Appreciation Right or the payment of any Award may be delayed, in the Committee's discretion, in the event that a potential termination for Cause, then the Participant will be required to deliver to the Company (i) Shares (or, in the discretion of the Committee, cash) equal in value to the amount of any profit the Participant realized upon the exercise of an Option or Stock Appreciation Right or Directorship or Termination of Employment for Cause; and (ii) the number of Shares (or, in the discretion of the Committee, the cash value of Shares) the Participant received for Other Stock Based Awards (including Restricted Stock, Restricted Units and Deferred Stock Units) that vested during the period specified in (i) above. Unless the applicable award certificate provides otherwise, if, after a Participant's

- 14 -

Termination of Directorship or Termination of Employment, the Committee determines in its sole discretion that while the Participant was a Company or Subsidiary employee or a Director, such Participant engaged in activity that would have been grounds for a Termination of Directorship or Termination of Employment for Cause, then the Company will immediately cancel all Stock Options, Stock Appreciation Rights, Annual Performance Bonuses, Long-Term Performance Awards, Restricted Units, Restricted Stock and Other Stock-Based Awards and the Participant will be required to deliver to the Company (A) Shares (or, in the discretion of the Committee, cash) equal in value to the amount of any profit the Participant realized upon the exercise of an Option or Stock Appreciate Right during the period that begins twelve (12) months immediately prior to the Participant's Termination of Directorship or Termination of Employment and ends on the date of the Committee's determination that the Participant's conduct would have constituted grounds for a Termination of Directorship or Termination of Employment for Cause; and (B) the number of Shares (or, in the discretion of the Committee, the cash value of said shares) the Participant received for Other Stock Based Awards (including Restricted Stock, Restricted Units and Deferred Stock Units) that vested during the period specified in (A) above.

## ARTICLE V SHARES SUBJECT TO THE PLAN; ADJUSTMENTS

### 5.1. Shares Available.

- (a) The Shares issuable under the Plan will be authorized but unissued Shares, and, to the extent permissible under applicable law, Shares acquired by the Company, any Subsidiary or any other person or entity designated by the Company and held as treasury shares.
- (b) Subject to the counting rules set forth in Section 5.2 and adjustment in accordance with Section 5.3, the total number of Shares with respect to which Awards may be issued under the Plan shall equal ten percent (10%) of the Shares outstanding as of the Effective Date.
- (c) Incentive Stock Options may be granted under the Plan in respect of no more than 10 million Shares.

### 5.2. Counting Rules.

- (a) The total number of Shares with respect to which Awards may be issued under the Plan, as described in Section 5.1(b), shall be reduced by 2.2 Shares per each Share subject to an Award of Restricted Stock, Restricted Units, Deferred Stock Units, Performance Units or Other Stock-Based Awards, or as payment of an Annual Performance Bonus.
- (b) The following Shares related to Awards under the Plan will again be available for issuance under the Plan:
  - (i) Shares related to Awards paid in cash;
  - (ii) Shares related to Awards that expire, are forfeited or cancelled or terminate for any other reason without issuance of Shares and any Shares of Restricted Stock that are returned to the Company upon a Participant's Termination of Employment or, if applicable, a Director's Termination of Directorship (including, for clarity, at a rate of 2.2 Shares per each Share related to such an Award in the form of Restricted Stock, Restricted Units, Deferred Stock Units, Performance Units or Other Stock-Based Awards, or as payment of an Annual Performance Bonus); and
  - (iii) Any Shares issued in connection with Awards that are assumed, converted or substituted as a result of the acquisition of an Acquired Company by the Company or a combination of the Company with another company. Shares available under a stockholder approved plan of an Acquired Company (as appropriately adjusted to reflect the transaction) may be used for Awards under the Plan to individuals who were not employees or directors of the Company or a subsidiary prior to the transaction (subject to the stock exchange's listing requirements)

5.3. *Adjustments*. In the event of a change in the outstanding Shares by reason of a share split, reverse share split, dividend or other distribution (whether in the form of cash, Shares, other securities or other property), extraordinary cash dividend, recapitalization, merger, consolidation, split-up, spin-off, reorganization, combination, repurchase or exchange of Shares or other securities or similar corporate transaction or event, the Committee shall make an appropriate adjustment to prevent dilution or enlargement of the benefits or potential benefits intended to be made available under the Plan. Any adjustment made by the Committee under this Section 5.3 will be conclusive and binding for all purposes under the Plan.

- 15 -

#### 5.4. Change in Control.

- (a) Acceleration. Unless the applicable Award Certificate provides otherwise, (i) all outstanding Stock Options and Stock Appreciation Rights will become exercisable as of the effective date of a Participant's Change in Control Termination if the Awards are not otherwise vested, and all conditions will be waived with respect to outstanding Restricted Stock and Restricted Units (other than Long-Term Performance Awards) and Deferred Stock Units and (ii) each Participant who has been granted a Long-Term Performance Award that is outstanding as of the date of such Participant's Change in Control Termination will be deemed to have achieved a level of performance, as of the Change in Control Termination, that would cause all (100%) of the Participant's Target Amounts to become payable and all restrictions on the Participant's performance-based Restricted Units and Shares of Restricted Stock to lapse. Unless the Committee determines otherwise in its discretion (either when an Award is granted or any time thereafter), in the event that Awards outstanding as of the date of a Change in Control that are payable in Ordinary Shares of the Company will not be substituted with comparable awards payable or redeemable in shares of publicly-traded stock after the Change in Control and (B) each such Award that is a Stock Option will be settled in cash, without the Participant's consent, for an amount equal to the amount that could have been attained upon the exercise of such Award immediately prior to the Change in Control and such Award that is a stock Option will be settled in cash, without the Participant's consent, for an amount equal to the amount that could have been attained upon the exercise of such Award immediately prior to the Change in Control have been attained upon the exercise of such Award immediately prior to the Change in Control have been attained upon the exercise of such Award immediately prior to the Change in Control have been attained upon the exercise of such Award immediately prior to the Change in Co
- (b) Permissive Actions. In addition to the actions described in Section 5.4(a)(A) and (B), in the event of a Change in Control, the Committee may take any one or more of the following actions with respect to any or all outstanding Awards, without the consent of Participants: (i) the Committee may determine that outstanding Stock Options and Stock Appreciation Rights shall be fully vested and exercisable and restrictions on Restricted Stock, Restricted Units, Deferred Stock Units and Other Stock-Based Awards shall lapse as of the date of the Change in Control or such other time (prior to a Participant's Change in Control Termination) as the Committee determines; (ii) the Committee may require that a Participant surrender his or her outstanding Stock Options and Stock Appreciation Rights in exchange for one or more payments by the Company, in cash or Ordinary Shares, as determined by the Committee, in an amount equal to the amount by which the then Fair Market Value of the Shares subject to the Participant's unexercised Stock Options and Stock Appreciation Rights exceeds the Exercise Price, if any, and on such terms as the Committee determines; (iii) after giving Participants an opportunity to exercise any outstanding Stock Options and Stock Appreciation Rights, the Committee may terminate any or all unexercised Stock Options and Stock Appreciation Rights at such time as the Committee deems appropriate; (iv) the Committee may determine that Annual Performance Bonuses and/or Long-Term Performance Awards will be paid out at their target level, in cash or Ordinary Shares as determined by the Committee; or (v) the Committee may determine that Awards that remain outstanding after the Change in Control shall be converted to similar grants of, or assumed by, the surviving corporation (or a parent or subsidiary of the surviving corporation or successor). Such acceleration, surrender, termination, settlement, payment or conversion shall take place as of the date of the Change in Control or such other date as the Committee determines. The Committee may specify how an Award will be treated in the event of a Change in Control either when the Award is granted or at any time thereafter.

5.5. *Fractional Shares*. No fractional Shares will be issued under the Plan. Except as otherwise provided in Section 4.5(e) and unless otherwise provided by the Committee, if a Participant acquires the right to receive a fractional Share under the Plan, the Participant will receive, in lieu of the fractional Share, a cash payment equal to the Fair Market Value of such fractional share on the date of settlement of the related Award.

# ARTICLE VI AMENDMENT AND TERMINATION

6.1. *Amendment*. The Plan may be amended at any time and from time to time by the Board or authorized Board committee without the approval of shareholders of the Company, except that no material revision to the terms of the Plan will be effective until the amendment is approved by the shareholders of the Company. A revision is "material" for this purpose if it materially increases the number of Shares that may be issued under the Plan (other than an increase pursuant to

- 16 -

Section 5.3 of the Plan), expands the types of Awards available under the Plan, materially expands the class of persons eligible to receive Awards under the Plan, materially extends the term of the Plan, reduces the Exercise Price at which Stock Options or Stock Appreciation Rights may be granted, reduces the Exercise Price of outstanding Stock Options or Stock Appreciation Rights, results in the replacement of outstanding Stock Options or Stock Appreciation Rights that have an Exercise Price that is lower than the Exercise Price of the replaced Stock Options or Stock Appreciation Rights that have an Exercise Price that is lower than the Exercise Price of the replaced Stock Options or Stock Appreciation Rights, or other Awards, or is otherwise an amendment requiring shareholder approval pursuant to any law or the rules of any exchange on which the Company's Ordinary Shares are listed for trading. No amendment of the Plan or any outstanding Award Certificate made without the Participant's written consent may adversely affect any right of a Participant with respect to an outstanding Award.

6.2. Termination. The Plan will terminate upon the earlier of the following dates or events to occur:

- (a) The adoption of a resolution of the Board terminating the Plan; or
- (b) The day before the tenth (10<sup>th</sup>) anniversary of the adoption of the Plan by the Company's shareholders as described in Section 1.2.

No Awards will be granted under this Plan after it has terminated. The termination of the Plan, however, will not alter or impair any of the rights or obligations of any person under any Award previously granted under the Plan without such person's consent. After the termination of the Plan, any previously granted Awards will remain in effect and will continue to be governed by the terms of the Plan and the applicable Award Certificate.

## ARTICLE VII GENERAL PROVISIONS

7.1. Nontransferability of Awards. No Award under the Plan will be subject in any manner to alienation, anticipation, sale, assignment, pledge, encumbrance or transfer, and no other persons will otherwise acquire any rights therein, except as provided below.

- (a) Any Award may be transferred by will or by the laws of descent or distribution.
- (b) Unless the applicable Award Certificate provides otherwise, all or any part of a Nonqualified Stock Option or Shares of Restricted Stock may be transferred to a family member without consideration. For purposes of this subsection (b), "family member" includes any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law of the Participant, including adoptive relationships, any person sharing the Participant's household (other than a tenant or employee), a trust in which these persons have more than fifty percent (50%) of the beneficial interest, a foundation in which these persons (or the Participant) control the management of assets, and any other entity in which these persons (or the Participant) own more than fifty percent (50%) of the voting interests.

Any transferred Award will be subject to all of the same terms and conditions as provided in the Plan and the applicable Award Certificate. The Participant or the Participant's estate will remain liable for any withholding tax that may be imposed by any federal, state or local tax authority. The Company may, in its sole discretion, disallow all or a part of any transfer of an Award pursuant to this Subsection 7.1(b) unless and until the Participant makes arrangements satisfactory to the Company for the payment of any withholding tax. The Participant must immediately notify the Company, in the form and manner required by the applicable Award Certificate or as otherwise required by the Company, of any proposed transfer of an Award pursuant to this Subsection 7.1(b). No transfer will be effective until the Company consents to the transfer.

(c) Unless the applicable Award Certificate provides otherwise, any Nonqualified Stock Option transferred by a Participant pursuant to subsection (b) may be exercised by the transferee only to the extent that the Award would have been exercisable by the Participant had no transfer occurred. The transfer of Shares upon exercise of the Award will be conditioned on the payment of any withholding tax.

- 17 -

- (d) Restricted Stock may be freely transferred after the restrictions lapse or are satisfied and the Shares are delivered, provided, however, that Restricted Stock awarded to an affiliate of the Company may be transferred only pursuant to Rule 144 under the Securities Act, or pursuant to an effective registration for resale under the Securities Act. For purposes of this subsection (d), "affiliate" will have the meaning assigned to that term under Rule 144.
- (e) In no event may a Participant transfer an Incentive Stock Option other than by will or the laws of descent and distribution.

7.2. *Withholding of Taxes*. The Committee, in its discretion, may require the satisfaction of a Participant's tax withholding obligations by any of the following methods or any method as it determines to be in accordance with the laws of the jurisdiction in which the Participant resides, has domicile or performs services.

- (a) Stock Options and Stock Appreciation Rights. As a condition to the delivery of Shares pursuant to the exercise of a Stock Option or Stock Appreciation Right, the Committee may require that the Participant, at the time of exercise, pay to the Company by cash, certified check, bank draft, wire transfer or postal or express money order an amount sufficient to satisfy any applicable tax withholding obligations. The Committee may also, in its discretion, accept payment of tax withholding obligations through any of the Exercise Price payment methods described in Section 4.3(d).
- (b) Other Awards Payable in Shares. The Participant shall satisfy the Participant's tax withholding obligations arising in connection with the release of restrictions on Restricted Units, Restricted Stock and Other Stock-Based Awards by payment to the Company in cash or by certified check, bank draft, wire transfer or postal or express money order, provided that the format is approved by the Company or a designated third-party administrator. However, subject to any requirements of applicable law, the Company may also satisfy the Participant's tax withholding obligations by other methods, including selling or withholding Shares that would otherwise be available for delivery.
- (c) *Cash Awards*. The Company may satisfy a Participant's tax withholding obligation arising in connection with the payment of any Award in cash by withholding cash from such payment.

7.3. *No Implied Rights.* The establishment and operation of the Plan, including the eligibility of a Participant to participate in the Plan, will not be construed as conferring any legal or other right upon any Director for any continuation of directorship or any Employee for the continuation of employment through the end of any Performance Cycle or other period. The Company expressly reserves the right, which may be exercised at any time and in the Company's sole discretion, to discharge any individual or treat him or her without regard to the effect that discharge might have upon him or her as a Participant in the Plan.

7.4. No Obligation to Exercise Awards. The grant of a Stock Option or Stock Appreciation Right will impose no obligation upon the Participant to exercise the Award.

7.5. No Rights as Shareholders. A Participant who is granted an Award under the Plan will have no rights as a shareholder of the Company with respect to the Award unless and until certificates for the Shares underlying the Award are registered in the Participant's name and (other than in the case of Restricted Stock) delivered to the Participant. The right of any Participant to receive an Award by virtue of participation in the Plan will be no greater than the right of any unsecured general creditor of the Company.

7.6. *Indemnification of Committee*. The Company will indemnify, to the fullest extent permitted by law, each person made or threatened to be made a party to any civil or criminal action or proceeding by reason of the fact that the person, or the executor or administrator of the person's estate, is or was a member of the Committee or an authorized delegate of the Committee including, for purposes of Director Awards, the Nominating Committee.

7.7. No Required Segregation of Assets. Neither the Company nor any Subsidiary will be required to segregate any assets that may at any time be represented by Awards granted pursuant to the Plan.

- 18 -

7.8. *Nature of Payments*. All Awards made pursuant to the Plan are in consideration of services for the Company or a Subsidiary. Any gain realized pursuant to Awards under the Plan constitutes a special incentive payment to the Participant and will not be taken into account as compensation for purposes of any other employee benefit plan of the Company or a Subsidiary, except as the Committee otherwise provides. The adoption of the Plan will have no effect on Awards made or to be made under any other benefit plan covering an employee of the Company or a Subsidiary or any predecessor or successor of the Company or a Subsidiary.

7.9. Securities Law Compliance. Awards under the Plan are intended to satisfy the requirements of Rule 16b-3 under the Exchange Act. If any provision of this Plan or any grant of an Award would otherwise frustrate or conflict with this intent, that provision will be interpreted and deemed amended so as to avoid conflict. No Participant will be entitled to a grant, exercise, transfer or payment of any Award if the grant, exercise, transfer or payment would violate the provisions of the Sarbanes-Oxley Act of 2002 or any other applicable law.

7.10. *Coordination with Other Plans.* If this Plan provides a level of benefits with respect to Awards that differs from the level of benefits provided under the Mallinckrodt Pharmaceuticals Severance Plan for U.S. Officers and Executives, the Mallinckrodt Pharmaceuticals Change in Control Severance Plan for Certain U.S. Officers and Executives or the Mallinckrodt Pharmaceuticals Severance Plan for U.S. Employees, then the terms of the plan that provides for the more favorable benefit to the Participant shall govern

7.11. Section 409A Compliance. Notwithstanding any other provision of this Plan or an applicable Award Certificate to the contrary, the provisions of this Section 7.11 shall apply to all Awards that were issued or became vested on or after January 1, 2005 and that are subject to Code Section 409A, but only with respect to the portion of such Award that is subject to Code Section 409A.

- (a) General. To the extent the Committee (or Nominating Committee with respect to Director Awards) determines that any Award granted under the Plan is subject to Code Section 409A, the Award Certificate evidencing such Award will incorporate the terms and conditions required by Code Section 409A. To the extent applicable, the Plan and the Award Certificate will be interpreted in accordance with Code Section 409A and the applicable regulations and rulings thereunder. Notwithstanding any other provision of the Plan to the contrary, in the event that the Committee (or Nominating Committee with respect to Director Awards) determines that any Award may be subject to Code Section 409A, the Committee may adopt such amendments to the Plan and/or the applicable Award Certificate or adopt policies and procedures or take any other action or actions, including an action or amendment with retroactive effect, that the Committee (or Nominating Committee with respect to Director Awards) determines is necessary or appropriate to (i) exempt the Award from the application of Code Section 409A or (ii) comply with the requirements of Code Section 409A.
- (b) *Modifications to Defined Terms*. The following modifications to Plan provisions (and, if necessary, applicable Award Certificate provisions) shall apply.
  - (i) Any payment of deferred compensation subject to Code Section 409A that is to be made under an Award other than an Annual Performance Bonus upon the occurrence of a Change in Control or any change in the timing and/or form of such payment as a direct result of a Change in Control (including payments made upon a specified date or event occurring after a Change in Control) shall not be made, or such change in timing and/or form shall not occur, unless such Change in Control is also a "change in ownership or effective control" of the Company within the meaning of Code Section 409A(a)(2)(A)(v) and applicable regulations and rulings thereunder and such payment, or such change in timing and/or form, occurs no later than two (2) years after the date of such change in ownership or effective control of the Company, in each case to the extent required to avoid the recipient of such Award from incurring tax penalties under Code Section 409A in respect of such Award. Notwithstanding the foregoing, if the Committee takes an action pursuant to Section 5.4(b) to accelerate the payment of deferred compensation upon a Change in Control, then any accelerated payment shall occur on a date specified in the applicable Award Certificate, which date shall be no later than ninety (90) days after a "change in ownership or effective control" of the Company. The payment of an Annual Performance Bonus that is to be accelerated pursuant to Subsection 4.4(g) shall occur within thirty (30) days after a "change in ownership or effective control" of the Company.

- 19 -

- (ii) The definition of "Change in Control Termination" in subsection (b) of that definition shall be deleted in its entirety and replaced with the following:
  - "(b) termination of the Participant's employment by the Participant after one of the following events:
    - (i) the Company (1) assigns or causes to be assigned to the Participant duties inconsistent in any material respect with his or her position as in effect immediately prior to the Change in Control; (2) makes or causes to be made any material adverse change in the Participant's position (including titles and reporting relationships and level), authority, duties or responsibilities, or the budget over which the Participant retains authority; or (3) takes or causes to be taken any other action which results in a material diminution in such position, authority, duties or responsibilities or the budget over which the Participant retains authority, duties or responsibilities or the budget over which the Participant retains authority, duties or responsibilities or the budget over which the Participant retains authority, duties or responsibilities or the budget over which the Participant retains authority, duties or responsibilities or the budget over which the Participant retains authority authority, duties or responsibilities or the budget over which the Participant retains authority authority.
    - (ii) the Company, without the Participant's consent, (1) requires the Participant to relocate to a principal place of employment more than fifty (50) miles from his or her existing place of employment, which increases the Participant's commute from his or her principal residence by more than fifty (50) miles; or (2) materially reduces the Participant's base salary, annual bonus, or retirement, welfare, share incentive, perquisite (if any) and other benefits taken as a whole;

provided that an event described in (i) or (ii) above shall permit a Participant's termination of employment to be deemed a Change in Control Termination only if (x) the Participant provides written notice to the Company specifying in reasonable detail the event upon which the Participant is basing his termination within ninety (90) days after the occurrence of such event, (y) the Company fails to cure such event within thirty (30) days after its receipt of such notice, and (z) the Participant terminates his employment within sixty (60) days after the expiration of such cure period."

(iii) The definition of "Disabled" or "Disability"" shall be deleted in its entirety and replaced with the following:

""Disabled" or "Disability" means that the Employee is receiving income replacement benefits for a period of not less than three (3) months under a Company or Subsidiary accident and health plan covering the Employee by reason of any medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than twelve (12) months."

- (iv) A Termination of Directorship or Termination of Employment shall only occur where such Termination of Directorship or Termination of Employment is a "separation from service" within the meaning of Code Section 409A(a)(2)(A)(i) and the applicable regulations and rulings thereunder. For purposes of determining whether a Termination of Directorship has occurred under this Subsection 7.12(b)(iii), services provided in the capacity of an employee or otherwise shall be excluded.
- (c) *Modifications to or Adjustments of Awards*. Any modifications to an Award pursuant to Subsection 3.2(g) or adjustments of an Award pursuant to Subsections 4.8 or 5.3 shall comply with the requirements of Section 409A.
- (d) Specified Employees. Payments to any Participant who is a "specified employee" of deferred compensation that is subject to Code Section 409A(a)(2) and that becomes payable upon, or that is accelerated upon, such Participant's Termination of Employment (as modified by Subsection 7.12(b)(iv)), shall not be made on or before the date which is six (6) months following such Participant's Termination of Employment (or, if earlier, such Participant's death). A specified employee for this purpose shall be determined by the Committee or its delegate in accordance with the provisions of Code Section 409A and the regulations and rulings thereunder.

- 20 -

7.12. Section 457A Compliance. To the extent the Committee (or Nominating Committee with respect to Director Awards) determines that any Award granted under the Plan is subject to Code Section 457A, the Award Certificate evidencing such Award will incorporate the terms and conditions required by Code Section 457A in order to avoid accelerated taxation or tax penalties to the holder thereof in respect of such Award. To the extent applicable, the Plan and the Award Certificate will be interpreted in accordance with Code Section 457A and applicable guidance issued thereunder. Notwithstanding any other provision of the Plan to the contrary, in the event that the Committee (or Nominating Committee with respect to Director Awards) determines that any Award may be subject to Code Section 457A, the Committee may adopt such amendments to the Plan and/or the applicable Award Certificate or adopt policies and procedures or take any other action or actions, including an action or amendment with retroactive effect, that the Committee (or Nominating Committee with respect to Director As7A or (ii) comply with the requirements of Code Section 457A.

7.13. Governing Law, Severability. The Plan and all determinations made and actions taken under the Plan will be governed by the law of Ireland and construed accordingly. If any provision of the Plan is held unlawful or otherwise invalid or unenforceable in whole or in part, the unlawfulness, invalidity or unenforceability will not affect any other parts of the Plan, which parts will remain in full force and effect.

- 21 -

## MALLINCKRODT PHARMACEUTICALS EMPLOYEE STOCK PURCHASE PLAN EFFECTIVE AS OF JULY 1, 2013

### ARTICLE 1

### PURPOSE

This Mallinckrodt Pharmaceuticals Employee Stock Purchase Plan (the "Plan") is created for the purpose of encouraging share ownership by officers and employees of Mallinckrodt public limited company (the "Company") and its subsidiaries so that they may share in growth of the Company by acquiring or increasing their proprietary interest in the Company.

## ARTICLE 2

### ADMINISTRATION OF THE PLAN

The Plan is administered by the Compensation and Human Resources Committee, a committee of the Board of Directors of the Company (the "Committee"). The Committee may delegate its authority and responsibility for plan administration to a committee or an officer or group of officers, as it deems advisable. The interpretation and construction by the Committee, or its delegate, of any provision of the Plan shall be final and binding on all parties. The Committee, or its delegate, may adopt, from time to time, such rules and regulations, as it deems appropriate for carrying out the Plan. No member of the Board of Directors or the Committee, or its delegate, shall be liable for any action or determination made in good faith with respect to the Plan.

### **ARTICLE 3**

### ELIGIBLE EMPLOYEES

The Company will, from time to time, determine which of its employees (including employees of its subsidiaries and divisions) will be eligible to participate in the Plan. All officers who are employees of the Company will be eligible to participate in the Plan. Eligible employees who elect to participate in the Plan shall hereinafter be referred to as "Participants."

### **ARTICLE 4**

#### SHARES TO BE PURCHASED

The shares subject to purchase under the Plan is 5,000,000 shares (subject to adjustment in the event of share splits, share dividends, recapitalization, or similar adjustment in the Company's share capital) of the ordinary share capital of the Company (the "Shares") which will be purchased in accordance with Article 8.

### **ARTICLE 5**

#### PAYROLL DEDUCTIONS

Participants, upon entering the Plan, shall authorize payroll deductions to be made for the purchase of Shares. The maximum deduction shall not, on a per pay period basis, exceed a Participant's base salary or commission (in the case of an employee who receives commission and no base salary) and deductions shall be exclusive of overtime and net withholding and other deductions. The Participant may authorize

increases or decreases in the amount of payroll deductions at any time. In order to effect such a change in the amount of the payroll deductions, the Company must receive notice of such change in the manner specified by the Company and changes will take effect as soon as administratively practicable. The Company will accumulate and hold for the Participant's account the amounts deducted from his/her pay. No interest shall be paid on such amounts. Notwithstanding the foregoing, the Committee may, in its sole discretion, authorize a special bonus payment be made to a Participant and such bonus be designated as an employee contribution. The Company will match such employee contribution, subject to the limit described in the next Article. The bonus may exceed the contribution limits otherwise imposed on the Participant.

### **ARTICLE 6**

#### EMPLOYER CONTRIBUTION

The Company will match a part of the employee contribution by contributing to the Plan an additional percentage of the Participant's payroll deduction. The Committee, from time to time, may increase or decrease the percentage of the Company's contribution to the Participant's payroll deduction if the interests of the Company so require. The Company shall not match any part of a Participant's contribution that exceeds twenty-five thousand dollars (US) (\$25,000.00) during a single calendar year. The matching contributions hereunder are not intended to be entitlement or part of the regular compensation of any Participant. The Company will pay all commissions relating to the purchase of the Shares under the Plan, and the Company will pay all administrative costs associated with the implementation and operation of the Plan.

### ARTICLE 7

### AUTHORIZATION FOR ENTERING THE PLAN

An eligible employee may enter the Plan by enrolling in the Plan and specifying his/her contribution amount in the manner authorized by the Company. Such authorization will take effect as of the next practicable payroll period. Unless a Participant authorizes changes to his/her payroll deductions in accordance with Article 5 or withdraws from the Plan, his/her deductions under the latest authorization on file with the Company shall continue from one payment period to the succeeding payment period as long as the Plan remains in effect.

#### **ARTICLE 8**

#### PURCHASE OF SHARES

All Shares purchased under the Plan shall be purchased on the open market by a broker designated, from time to time, by the Committee. On a monthly basis, as soon as practicable following the month end, the Company shall remit the total of contributions to the broker for the purchase of the Shares. The broker will then execute the purchase order and the Plan Administrator shall allocate Shares (or fraction thereof) to each participant's individual recordkeeping account. In the event the purchase of Shares takes place over a number of days and at different prices, then each participant's allocation shall be adjusted on the basis of the average price per Share over such period.

### ARTICLE 9

## ISSUANCE OF SHARES

The Shares purchased under the Plan shall be held by the Plan Administrator or its nominee. Participants shall receive periodic statements that will evidence all activity in the accounts that have been established on their behalf. Such statements will be issued by the Plan Administrator or its nominee.

## **ARTICLE 10**

#### DIVIDEND REINVESTMENT

Any dividends paid to a Participant for Shares purchased under the Plan shall be paid in cash except where the Participant voluntarily elects to reinvest such dividends in Shares of the Company in accordance with such rules or procedures as may be established by the Company from time to time.

#### **ARTICLE 11**

# SALE OF SHARES PURCHASED UNDER THE PLAN

Each Participant may sell at any time all or any portion of the Shares acquired under the Plan and held by the Plan Administrator for at least three months by notifying the Plan Administrator, who will direct the broker to execute the sale on behalf of the Participant. The Participant shall pay the broker's commission and any other expenses incurred with regard to the sale of the Shares. All such sales of the Shares will be subject to compliance with any applicable federal or state securities, tax, or other laws. Each participant assumes the risk of any fluctuations in the market price of the Shares.

### ARTICLE 12

### WITHDRAWAL FROM THE PLAN

A Participant may cease making contributions to the Plan at any time by changing his/her payroll deduction to zero as described in Article 5. In order to execute a sale of all or part of the Shares purchased under the Plan and held by the Plan Administrator for at least three months, the Participant must contact the Plan Administrator directly. If the Participant desires to withdraw from the Plan by liquidating all or part of his/her shareholder interest, he/she shall receive the proceeds from the sale thereof, minus the commission and other expenses on such sale.

### ARTICLE 13

#### NO TRANSFER OR ASSIGNMENT

A Participant's right to purchase Shares under the Plan through payroll deduction is his/hers alone and may not be transferred or assigned to, or availed of, by any other person.

### ARTICLE 14

## TERMINATION OF EMPLOYEE RIGHTS

All of the employee's rights under the Plan will terminate when he/she ceases to be an eligible employee due to retirement, resignation, death, termination, or any other reason. A notice of withdrawal will be deemed to have been received from a Participant on the day of his/her final payroll deduction. If a Participant's payroll deductions are interrupted by any legal process, a withdrawal notice will be deemed as having been received on the day the interruption occurs.

### **ARTICLE 15**

### TERMINATION AND AMENDMENT TO THE PLAN

The Plan may be terminated at any time by the Company's Board of Directors. Upon such termination, or any other termination of the Plan, all payroll deductions not used to purchase Shares will be refunded. The Board of Directors also reserves the right to amend the Plan, from time to time, in any respect and authorizes the Committee to approve amendments to the Plan on its behalf.

### **ARTICLE 16**

### LOCAL TAX LAWS

If the provisions of the Plan contradict local tax laws, the local tax laws shall prevail.

#### **ARTICLE 17**

#### GOVERNING LAW

This Plan shall be governed by, and construed in accordance with, the laws of Ireland.

### **ARTICLE 18**

#### SEVERABILITY

If any provision of this Plan is held unlawful or otherwise invalid or unenforceable in whole or in part, the unlawfulness, invalidity or unenforceability will not affect any other parts of the Plan, which parts will remain in force and effect.

## MALLINCKRODT PHARMACEUTICALS EMPLOYEE STOCK PURCHASE PLAN PROSPECTUS

This summary of the Mallinckrodt Pharmaceuticals Employee Stock Purchase Plan effective as of July 1, 2013 (the "Plan") provides information about the Plan and your ability to purchase ordinary shares of Mallinckrodt public limited company (the "Shares") under the Plan. Mallinckrodt public limited company, incorporated in Ireland, together with its subsidiaries, is called "Mallinckrodt". The Plan has been and continues to be introduced into certain countries in which Mallinckrodt employees are located, as determined by Mallinckrodt, to encourage Share ownership by you and other Mallinckrodt employees.

## Shares Available under the Plan

The number of Shares currently approved for purchase under the Plan is 5,000,000. If Mallinckrodt's capital structure changes because of a share dividend, a reorganization or similar event, the number of Shares that may be purchased under the Plan will be adjusted appropriately. Ordinary shares will be purchased on the open market by a securities firm designated on behalf of eligible employees who have authorized payroll deductions. Mallinckrodt will pay all commissions related to such purchases. The board of directors may increase the number of Shares available for purchase under the Plan in the future.

## **Material Features of the Plan**

<u>Administration</u>. The Mallinckrodt Board's Compensation and Human Resources Committee, or its delegate (the "Committee") oversees the Plan and has the power to interpret and construe any provision of the Plan and to adopt rules and regulations for carrying out the Plan. The Committee's interpretations and decisions are final and binding on all persons, including Mallinckrodt and you. The Plan may be amended or terminated at any time by the Committee or the Mallinckrodt Board. If the Plan is terminated, any of your payroll deductions not used to purchase Shares as of the date of Plan termination will be refunded to you.

The Committee has appointed a Plan Administrator who handles the day-to-day administration of the Plan.

<u>Participation</u>. No employee has an absolute right to participate in the Plan. Mallinckrodt determines which employees are eligible to participate. If you are an eligible employee, you may join the Plan by authorizing payroll deductions in the manner specified by Mallinckrodt. Your contributions will begin as soon as administratively practicable.

The maximum contribution that may be deducted from your compensation cannot exceed your base salary or, for employees paid on a commission-only basis, your commission during a payroll period (exclusive of overtime and bonuses, and net of withholding and other deductions). Mallinckrodt will match a portion of your contribution by contributing to the Plan an additional percentage of your contribution. The matching amount is currently [TBD] of the amount you contribute, but Mallinckrodt has the authority to change this amount. Mallinckrodt will not match portions of your contribution that exceed twenty five thousand dollars (US) (\$25,000.00) during a single calendar year. You will receive quarterly account statements from the designated Plan Administrator that will show all activity in your account.

Any dividends paid on Shares purchased under the Plan are automatically paid in cash. However, United States employees may voluntarily elect to reinvest such dividends in additional Shares, in which case such additional Shares shall be allocated to the respective Plan account.

Sale or Transfer of Shares. Subject to the short-term trading rule described below, you may sell or transfer the Shares held in your Plan account at any time. Contact the Plan Administrator directly to make arrangements for the sale of Shares or the transfer of Shares to another securities account. When you sell Shares, your sale order will be combined with all other sale orders from Plan participants placed on the date you gave the Plan Administrator your instructions to sell Shares. There may be situations where the Plan Administrator will sell Shares over a number of days. In these situations, your sale price is the average price for all Shares sold. You are responsible for any commission and fees on the sale of Shares. Under Mallinckrodt's short-term trading rules, you will be prohibited from selling or transferring Shares within twelve (12) months of the date you purchased the Shares.

<u>Miscellaneous</u>. You cannot assign or transfer your interest under the Plan. You can, however, have a joint account with someone else. The Plan is not subject to any provisions of the United States Employee Retirement Income Security Act of 1974, as amended, and is not qualified under Section 401(a) of the United States Internal Revenue Code of 1986, as amended (the "Code").

**End of Participation**. You are an active participant in the Plan until you stop making contributions to the Plan. You can stop all contributions by notifying Mallinckrodt through the appropriate procedures that you want to stop

contributing to the Plan. Contributions will be stopped as soon as administratively practicable after receipt of the notice. Your contributions will automatically cease if you go on unpaid leave or when you are no longer an employee of Mallinckrodt. You will no longer be a Plan participant (active or inactive) once all Shares are sold or transferred from your Plan account.

### **Tax Consequences**

The following is intended only as a general summary of the federal income tax consequences to United States participants in the Plan. It does not address all of the tax consequences that may be relevant to you in light of your particular tax circumstances. The summary is based on provisions of the Code and regulations, administrative rulings and judicial decisions now in effect, all of which are subject to change at any time (possibly with retroactive effect) or different interpretations. This summary does not address tax consequences under the laws of any state, locality or foreign jurisdiction and the tax treatment of each participant in the Plan will depend in part upon such participant's particular tax situation. Accordingly, you are urged to consult your own tax advisor as to the specific tax consequences of your participation in the Plan under federal, state, local and other applicable laws.

You will have taxable income equal to the amount of the matching contributions Mallinckrodt makes to the Plan on your behalf. This taxable income will be subject to both federal and state income tax withholding and FICA withholding. Mallinckrodt is entitled to a deduction for federal income tax purposes for its contributions and expenses in connection with the Plan. Any dividends earned by Shares, and any gain or loss realized by you upon the sale of Shares credited to your account, are taxable to you at the applicable tax rate.

If you elect to have shares issued in your name jointly with another, you may incur federal gift taxes. You should speak with your personal tax advisor before establishing a joint account.

### Information about Mallinckrodt

Additional information about the Plan and its administrators may be obtained by contacting the Human Resources Department, Mallinckrodt Pharmaceuticals, 675 James S. McDonnell Blvd., Hazelwood, MO 63042 or via e-mail at EquityAdmin@Mallinckrodt.com.

The following documents are incorporated by reference and shall be deemed to be part of this prospectus.

1. All reports filed pursuant to Section 13(a) or 15(d) of the United States Securities Exchange Act of 1934, as amended (the "Exchange Act");

2. The description of Mallinckrodt's ordinary shares contained in Mallinckrodt's most recent registration statement under the Exchange Act, including any amendment thereto of report filed for the purposes of updating such description; and

3. All documents filed by Mallinckrodt pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Exchange Act subsequent to the date of this document and prior to the filing of a post-effective amendment which indicates that all securities offered herein have been sold or which deregisters all securities then remaining unsold.

This summary incorporates documents by reference that are not presented herein or delivered herewith. These documents and other documents distributed to eligible employees are available, without charge, upon written or oral request from Mallinckrodt Pharmaceuticals, 675 James S. McDonnell Blvd., Hazelwood, MO 63042 Attn: Equity Plan Administration or via e-mail at EquityAdmin@Mallinckrodt.com.

This document constitutes part of a prospectus covering securities that have been registered under the United States Securities Act of 1933.

# MALLINCKRODT PUBLIC LIMITED COMPANY

# RULES

# OF

# MALLINCKRODT SAVINGS-RELATED SHARE PLAN

Approved by a board/shareholder resolution on: [date]

Approved by HM Revenue & Customs on: [date]

HM Revenue & Customs reference no: SRS110732

# CONTENTS

Rule			Page Number
1.	INTER	PRETATION	1
1.	1.1.	Definitions	1
	1.1.	Interpretation	5
2.		ATIONS TO APPLY FOR, AND APPLICATIONS FOR, GRANT OF OPTIONS	5
2.	2.1.	Announcement of intention to issue Invitations by Board or Trustees	5
	2.2.	Persons to whom Invitations must be issued	5
	2.3.	Documents which must accompany Invitation	5
	2.4.	Contents of Invitation	5
	2.5.	Contents of Application Form	6
	2.6.	Number of Plan Shares applied for in Application	7
	2.7.	Making of Applications	7
3.	GRANT OF OPTIONS		7
5.	3.1.	Options granted by Company or Trustees	7
	3.2.	Persons to whom Options must be granted	7
	3.3.	Procedure for grant of Options and Grant Date	7
	3.4.	Contents of Option Certificate	7
	3.5.	Number of Plan Shares over which Options granted	8
	3.6.	Scaling down of Applications	8
	3.7.	Period allowed for grant of Options	8
	3.8.	Duration of Plan	8
	3.9.	Persons to whom Options may be granted	9
	3.10.	Options non-transferable	9
4.	LIMIT	ON AGGREGATE NUMBER OF PLAN SHARES PLACED UNDER OPTION	9
	4.1.	Power to set limit	9
5.	EXERC	CISE PRICE	9
6.	EXERC	CISE OF OPTIONS	9
	6.1.	Earliest date for exercise of Options	9
	6.2.	Latest date for exercise of Options	9
	6.3.	Persons who may exercise Options	9
	6.4.	Material Interest	10
	6.5.	Number of Plan Shares acquired on exercise of Options	10
	6.6.	Options may be exercised in whole or in part	10
	6.7.	Procedure for exercise of Options	10
	6.8.	Issue or transfer of Plan Shares on exercise of Options	10
	6.9.	Amount of repayment under Savings Contract	10
7.	EXERCISE OF OPTIONS IN SPECIAL CIRCUMSTANCES		11
	7.1.	Death	11
	7.2.	Injury, disability, redundancy, retirement etc	11
	7.3.	Specified Age	11
	7.4.	Other special circumstances	11
	7.5.	Office or employment in Group Company	12
	7.6.	Termination of Savings Contract	12
	7.7.	Meaning of ceasing to be in Relevant Employment	12
	7.8.	Interaction of Rules	12
8.	TAKEC	OVER, RECONSTRUCTION, AMALGAMATION OR WINDING-UP OF COMPANY	12
	8.1.	General offer for, or acquisition of, Company	12
	8.2.	Compulsory acquisition of Company	13
	8.3.	Reconstruction or amalgamation of Company	13
	8.4.	Winding-up of Company	13
	8.5.	Shares subject to Options ceasing to be Plan Shares	13
	8.6.	Meaning of "obtains Control of the Company"	14
	8.7.	Notification of Option Holders	14

8.6. 8.7. Notification of Option Holders

-i-

9.	EXCHANGE OF OPTIONS	14
	9.1. Circumstances in which Exchange can occur	14
	9.2. Period allowed for exchange of Options	14
	9.3. Meaning of "equivalent"	14
	9.4. Grant Date of New Option	15
	9.5. Application of Plan to New Option	15
10.	LAPSE OF OPTIONS	15
11.	ADJUSTMENT OF OPTIONS ON REORGANISATION	16
	11.1. Power to adjust Options	16
	11.2. Exercise Price	16
	11.3. Capitalisation of reserves	16
	11.4. HM Revenue & Customs approval	16
	11.5. Notification of Option Holders	16
12.	ISSUE AND AVAILABILITY OF PLAN SHARES	16
	12.1. Rights attaching to Plan Shares	16
	12.2. Availability of Plan Shares	16
13.	RELATIONSHIP OF PLAN TO CONTRACT OF EMPLOYMENT	17
	13.1. Contractual Provisions	17
14.	ADMINISTRATION OF PLAN	17
	14.1. Responsibility for administration	17
	14.2. Grantor's decision final and binding	17
	14.3. Trustees to consult with Board	17
	14.4. Provision of information	17
	14.5. Cost of Plan	18
	14.6. Establishment of separate plans for overseas territories	18
	14.7. Data protection	18 19
15.	15. AMENDMENT OF PLAN	
	15.1. Power to amend Plan	19
	15.2. HM Revenue & Customs approval of amendments	19
	15.3. Rights of existing Option Holders	19
	15.4. Notification of Option Holders	19
16.	NOTICES	19
	16.1. Notice by Grantor	19
	16.2. Deceased Option Holders	19
	16.3. Notice to Grantor	19
	16.4. Option Certificate and Notice of Option	20
17.	GOVERNING LAW AND JURISDICTION	20
	17.1. Plan governed by English law	20
	17.2. English courts to have jurisdiction	20
	17.3. Jurisdiction agreement for benefit of Company	20
	17.4. Option Holder deemed to submit to such jurisdiction	20

-ii-

## 1. INTERPRETATION

# 1.1. Definitions

In this Plan, unless the context otherwise requires, the following words and expressions have the following meanings:

- 1.1.1. Acquiring Company means a company (including a New Holding Company) which obtains Control of the Company in the circumstances referred to in Rule 8.1, 8.2 or 8.3 (reading the reference in Rule 8.3 to "proposes to obtain" as "obtains");
- 1.1.2. Acting In Concert has the meaning given to that expression in The City Code on Takeovers and Mergers in its present form or as amended from time to time;
- 1.1.3. Adoption Date means the date on which the Plan was adopted by the board of Mallinckrodt public limited company, incorporated in Ireland;
- 1.1.4. **Applicant** means an Eligible Employee who applies for the grant of an Option;
- 1.1.5. **Application** means an application for the grant of an Option;
- 1.1.6. **Application Form** means the form referred to in Rule 2.3 on which an application for the grant of an Option is made;
- 1.1.7. Approval Date means the date on which the Plan is approved by HM Revenue & Customs under Schedule 3;
- 1.1.8. **Associated Company** has the meaning given to that expression by paragraph 47 of Schedule 3 or, where the context requires, paragraph 35(4) of Schedule 3;
- 1.1.9. **Board** means the board of directors of the Company or a duly authorised committee thereof;

# 1.1.10. Bonus Date means

- (a) in the case of a three year Savings Contract, the earliest date on which a Standard Bonus would be payable under the Savings Contract; and
- (b) in the case of a five year Savings Contract, the earliest date on which a Standard Bonus or a Maximum Bonus would be payable under the Savings Contract, according to whether, for the purpose of determining the number of Plan Shares over which the Option linked to the Savings Contract was granted, the repayment under the Savings Contract is to be taken as including the Standard Bonus (or no bonus) or the Maximum Bonus, respectively;
- 1.1.11. **Close Company** has the meaning given to that expression by sections 442(a), 446 and 447 of CTA 2010, and paragraph 11(4) of Schedule 3;
- 1.1.12. **Company** means Mallinckrodt public limited company incorporated in Ireland under registered number 522227, being the scheme organiser for the purposes of paragraph 2(2) of Schedule 3;
- 1.1.13. **Consortium** has the meaning given to that word by paragraph 48(2) of Schedule 3;
- 1.1.14. **Constituent Company** means the Company or a company which is a Subsidiary and which has been nominated by the Board to participate in the Plan from time to time;

-1-

- 1.1.15. **Continuous Employment** has the meaning given by the Employment Rights Act 1996:
- 1.1.16. **Control** has the meaning given to that word by section 719 of ITEPA 2003;
- 1.1.17. **CTA 2010** means the Corporation Tax Act 2010;
- 1.1.18. Eligible Employee means an individual who is:
  - (a) an employee (other than a director) of a Constituent Company; or
  - (b) a director of a Constituent Company who is contracted to work at least 25 hours per week for the Group (exclusive of meal breaks);

and who, in either case:

- (i) is not eligible solely by reason that he is a non-executive director of a Constituent Company;
- (ii) has earnings in respect of his office or employment which are (or would be if there were any) general earnings to which section 15 of ITEPA 2003 applies;
- (iii) has at the Grant Date such period of Continuous Employment as a director or employee, not exceeding five years, as the Grantor determines for the purpose of an issue of Invitations;
- (iv) has not given or been given notice to terminate his employment within the Group; and
- (v) does not have at the Grant Date, and has not had during the preceding twelve months, a Material Interest in a Close Company which is the Company or a company which has Control of the Company or a member of a Consortium which owns the Company; or
- (c) a director (other than a non executive director) or employee of a Constituent Company nominated by the Grantor to be an Eligible Employee who is not prohibited from participating in the Plan by sub paragraph (v) above;
- 1.1.19. **Employees' Share Scheme** has the meaning set out in section 1166 of the Companies Act 2006;
- 1.1.20. **Exercise Price** means the amount per Plan Share payable on the exercise of an Option determined in accordance with Rule 5;
- 1.1.21 **Grant Date** means the date on which an Option is granted to an Eligible Employee determined in accordance with Rule 3.3;
- 1.1.22 Grantor means
  - (a) in relation to an Option granted by the Company, the Board; and
  - (b) in relation to an Option granted by the Trustees, the Trustees;
- 1.1.23 **Invitation** means an invitation to apply for the grant of an Option issued under Rule 2.2;
- 1.1.24 **Invitation Date** means the date on which an Invitation is issued;

-2-

- 1.1.25 **ITA 2007** means the Income Tax Act 2007;
- 1.1.26 **ITEPA 2003** means the Income Tax (Earnings and Pensions) Act 2003;
- 1.1.27 **ITTOIA 2005** means the Income Tax (Trading and Other Income) Act 2005;
- 1.1.28 Key Feature means a provision of the Plan which is necessary in order to meet the requirements of Schedule 3;
- 1.1.29 Market Value means
  - (a) if at the relevant time Plan Shares are listed on the New York Stock Exchange (or any other recognised stock exchange within the meaning of section 1005 of ITA 2007) either:
    - (i) the middle market quotation of a Plan Share (as derived from the New York Stock Exchange or the list appropriate to such other exchange or market) for the trading day immediately preceding the Invitation Date; or
    - (ii) if the Grantor so decides, an amount equal to the arithmetic average of the middle market quotations of a Plan Share (derived on the same basis) for the three trading days immediately preceding the Invitation Date; or
    - (iii) if the Grantor so decides, the closing quotation of a Plan Share (as derived from the New York Stock Exchange or the list appropriate to such other exchange or market) for the trading day immediately preceding the Invitation Date

save that in any case, where the middle market quotation of a Plan Share is derived from a recognised stock exchange other than the New York Stock Exchange, the value must be agreed in advance with HM Revenue & Customs Shares & Assets Valuation; or

- (b) if at the relevant time Plan Shares are not so listed, the market value of a Plan Share as determined in accordance with Part VIII of the Taxation of Chargeable Gains Act 1992 and agreed in advance by the Grantor with HM Revenue & Customs Shares & Assets Valuation on the Invitation Date or such earlier date or dates as may be agreed with HM Revenue & Customs;
- 1.1.30 **Material Interest** has the meaning given to that expression by paragraphs 11 and 12 to 16 of Schedule 3;
- 1.1.31 **Maximum Bonus** means the bonus which is payable two years after the Option Holder has made 60 monthly (or the weekly equivalent) savings contributions under a five year Savings Contract;
- 1.1.32 **Minimum Monthly Savings Amount** means in relation to each Invitation, the minimum monthly saving which may be made by an Option Holder as determined by the Board in accordance with paragraph 25(3)(b) of Schedule 3 being not less than £5 (or such other minimum savings amount specified from time to time by HM Revenue & Customs in their Save-As-You-Earn prospectus) nor more than £10 (or such other amount as may be permitted from time to time under paragraph 25(3)(b) of Schedule 3);

-3-

- 1.1.33 **New Holding Company** means a company which obtains Control of the Company where 90% or more of the New Holding Company's ordinary shares are held in substantially the same proportions by substantially the same persons who previously held the Company's ordinary shares;
- 1.1.34 **New Option** means an option granted by way of exchange under Rule 9.1;
- 1.1.35 **New Plan Shares** means the shares subject to a New Option;
- 1.1.36 New York Stock Exchange means the New York Stock Exchange or any successor body;
- 1.1.37 Notice of Exercise means the notice given in respect of the exercise of an Option under Rule 6.7;
- 1.1.38 **Option** means a right to acquire Plan Shares granted under the Plan;
- 1.1.39 **Option Certificate** means the deed or statement under which an Option is granted in accordance with Rule 3.3;
- 1.1.40 **Option Holder** means an individual who holds an Option or, where the context permits, his legal personal representatives;
- 1.1.41 **Plan** means this Mallinckrodt Savings-Related Share Plan as adopted by Mallinckrodt public limited company on [ ] in its present form or as amended from time to time;
- 1.1.42 Plan Shares means ordinary shares in the capital of the Company which satisfy the conditions in paragraphs 17 to 22 of Schedule 3;
- 1.1.43 **Relevant Employment** means employment with any Group Member;
- 1.1.44 **Reorganisation** means any variation in the share capital of the Company, including but without limitation a capitalisation issue, rights issue, rights offer or bonus issue and a sub-division, consolidation or reduction in the capital of the Company but excluding a capitalisation issue in substitution for or as an alternative to a cash dividend;
- 1.1.45 **Rules** mean the rules of the Plan;
- 1.1.46 **Savings Contract** means a contract under a certified contractual savings scheme within the meaning of section 703 of ITTOIA 2005 which has been approved by HM Revenue & Customs for the purpose of Schedule 3;
- 1.1.47 Schedule 3 means Schedule 3 to ITEPA 2003;
- 1.1.48 Specified Age means 60 years;
- 1.1.49 Standard Bonus means the earliest bonus which is payable under a Savings Contract;
- 1.1.50 **Subsidiary** means a company which is a subsidiary of the Company within the meaning of section 1159 of the Companies Act 2006 over which the Company has Control;
- 1.1.51 **Trustees** means the trustees of any trust created by a Group Member which, when taken together with the Plan, constitutes an Employees' Share Scheme.

-4-

## 1.2. Interpretation

In the Plan, unless otherwise specified:

- 1.2.1. the contents and rule headings are inserted for ease of reference only and do not affect the interpretation of the Plan;
- 1.2.2. a reference to a Rule is a reference to a rule of the Plan;
- 1.2.3. save as provided for by law and subject to Rule 16.4 a reference to writing includes any mode of reproducing words in a legible form and reduced to paper or electronic format or communication including, for the avoidance of doubt, correspondence via e-mail;
- 1.2.4. the singular includes the plural and *vice versa* and the masculine includes the feminine;
- 1.2.5. a reference to a statutory provision includes any statutory modification, amendment or re-enactment thereof; and
- 1.2.6. the Interpretation Act 1978 applies to the Plan in the same way as it applies to an enactment.

## 2. INVITATIONS TO APPLY FOR, AND APPLICATIONS FOR, GRANT OF OPTIONS

### 2.1. Announcement of intention to issue Invitations by Board or Trustees

The Board or the Trustees may, in their absolute discretion, from time to time, announce their intention to issue Invitations in accordance with this Rule 2 to Eligible Employees to apply for the grant of Options.

### 2.2. Persons to whom Invitations must be issued

If the Grantor announces its intention to issue Invitations, it shall issue an Invitation to every person who is, or will on the Grant Date be, an Eligible Employee.

# 2.3. Documents which must accompany Invitation

# An Invitation shall be accompanied by:

- 2.3.1. an Application Form to be used by the recipient of the Invitation to apply for the grant of the Option referred to in the Invitation and to apply to enter into a Savings Contract approved by the Grantor for the purpose of that issue of Invitations and linked to the Option; and
- 2.3.2. a copy of the Rules.

## 2.4. Contents of Invitation

### An Invitation shall state:

2.4.1. the date, being not less than 14 nor more than 21 days after the date of issue of the Invitation, by which the recipient of the Invitation must submit an Application;

### -5-

- 2.4.2. the Minimum Monthly Savings Amount under the Savings Contract linked to the Option referred to in the Invitation;
- 2.4.3. the Exercise Price under the Option referred to in the Invitation or the method by which the Exercise Price will be determined and notified to Eligible Employees;
- 2.4.4. the maximum permitted aggregate monthly savings contribution under the Savings Contract linked to the Option referred to in the Invitation taken together with savings contributions by the Applicant under any other savings contract linked to any other Option or option granted under any other SAYE option scheme approved by HM Revenue & Customs under Schedule 3, being the lesser of £250 (or such other amount as may be permitted from time to time under paragraph 25(3)(a) of Schedule 3) and such other amount (being a multiple of £1 and not less than £5 (or such other minimum savings amount specified from time to time by HM Revenue & Customs in their Save-As-You-Earn prospectus)) as the Board may determine for the purpose of that issue of Invitations;
- 2.4.5. whether an Applicant must enter into a three year or a five year Savings Contract or may choose either;
- 2.4.6. whether, for the purpose of determining the number of Plan Shares over which the Option referred to in the Invitation is to be granted, the repayment under the Savings Contract linked to the Option must be taken as including the Maximum Bonus, the Standard Bonus or no bonus or whether the recipient of the Invitation may choose any of these; and
- 2.4.7. the maximum total number of Plan Shares, if any, set by the Board under Rule 4.1 over which Options will be granted in response to that issue of Invitations.
- Subject to this Rule 2, an Invitation shall be in such form as the Grantor may determine from time to time.

### 2.5. Contents of Application Form

- An Application Form shall require an Applicant to state:
- 2.5.1. the monthly savings contribution (being a multiple of £1 and not less than £5 (or such other minimum savings amount specified from time to time by HM Revenue & Customs in their Save-As-You-Earn prospectus)) which he wishes to make under the Savings Contract linked to the Option referred to in the Invitation;
- 2.5.2. that his proposed monthly savings contribution, when added to any monthly savings contributions then being made by him under any other Savings Contract linked to an Option or to an option granted under any other SAYE option scheme approved by HM Revenue & Customs under Schedule 3, will not exceed the maximum permitted aggregate monthly savings contribution specified in the Invitation;
- 2.5.3. where appropriate, whether he wishes to enter into a three or five year Savings Contract, and, in the case of a five year Savings Contract, whether he wishes it to be linked to the Maximum Bonus or the Standard Bonus; and
- 2.5.4. where appropriate, whether, for the purpose of determining the number of Plan Shares over which the Option referred to in the Invitation is to be granted, he wishes the repayment under the Savings Contract linked to the Option to be taken as including a bonus or no bonus;

-6-

and shall authorise the Grantor to enter on the Application Form, on behalf of the Applicant, such monthly savings contribution, not exceeding the maximum stated on the Application Form, as the Grantor determines under Rule 3.6.

Subject to this Rule 2, an Application Form shall be in such form as the Grantor may determine from time to time.

## 2.6. Number of Plan Shares applied for in Application

An Application shall be deemed to be for the grant of an Option over the maximum whole number of Plan Shares which may be acquired at the Exercise Price out of the expected repayment (including any bonus where permitted under Rule 2.4.6 and requested by the Applicant pursuant to Rule 2.5.4) under the Savings Contract linked to the Option at the applicable Bonus Date.

## 2.7. Making of Applications

The recipient of an Invitation who wishes to apply for the grant of the Option referred to in the Invitation shall submit to the Grantor, within the period specified in the Invitation, a duly completed Application Form.

## 3. GRANT OF OPTIONS

# 3.1. Options granted by Company or Trustees

The Company or the Trustees may from time to time grant Options to Eligible Employees.

## 3.2. Persons to whom Options must be granted

The Grantor shall grant the Option referred to in each Invitation in respect of which the Grantor has received a valid Application and, where Rule 3.6.4 applies, which has been selected by lot.

## 3.3. Procedure for grant of Options and Grant Date

The Grantor shall grant an Option by passing a resolution. The Grant Date shall be the date on which the Grantor passes the resolution or such later date as is specified in the resolution and allowed by Rules 3.7 and 3.8. The grant of an Option or Options shall be evidenced by a deed executed by or on behalf of the Grantor. The deed or a statement providing details of the grant shall be issued to each Applicant who has been granted an Option as soon as reasonably practicable following the grant of the Option.

## 3.4. Contents of Option Certificate

An Option Certificate shall state:

- the Grant Date;
- the number of Plan Shares subject to the Option;
- the Exercise Price; and
- the Bonus Date, being the date on which the Option will ordinarily become exercisable.

-7-

Subject thereto, an Option Certificate shall be in such form as the Board may determine from time to time.

# 3.5. Number of Plan Shares over which Options granted

An Option shall be granted over the number of Plan Shares for which the Applicant is deemed under Rule 2.6 or 3.6, as appropriate, to have applied.

## 3.6. Scaling down of Applications

If the Grantor receives Applications for the grant of Options over a number of Plan Shares in excess of any of the limits in Rule 4, it shall, to the extent necessary to eliminate the excess, take the following steps in the following order or such other steps as it may agree in advance with HM Revenue & Customs:

- 3.6.1. first, for the purpose of determining the number of Plan Shares over which the Option referred to in an Invitation is to be granted, it shall take the repayment under the Savings Contract linked to the Option as including the Standard Bonus instead of the Maximum Bonus;
- 3.6.2. secondly, it shall take the repayment under the Savings Contract linked to the Option as including no bonus instead of the Standard Bonus;
- 3.6.3. thirdly, it shall reduce pro rata the excess over £5 (or such other minimum savings amount specified from time to time by HM Revenue & Customs in their Save-As-You-Earn prospectus), or such greater amount as the Grantor may determine, of the monthly savings contribution selected by each Applicant;
- 3.6.4. fourthly, it shall select Applications by lot and each Application shall be deemed to be for a monthly savings contribution of £5 (or such other minimum savings amount specified from time to time by HM Revenue & Customs in their Save-As-You-Earn prospectus) only with the repayment under the Savings Contract linked to the Option taken as including no bonus.

Each Application shall be deemed to have been withdrawn or amended accordingly and the Grantor shall amend each Application Form to reflect any reduction in the bonus or the monthly savings contribution resulting therefrom.

For the purpose of applying this Rule 3.6, if an Applicant has made multiple Applications, the Applications shall be treated as a single Application and the monthly savings contributions applied for in the Applications shall be aggregated.

## 3.7. Period allowed for grant of Options

An Option may be granted only during the period of thirty days beginning on the earliest of the dates referred to in the definition of "Market Value" and used for the purpose of determining the Exercise Price or, if Rule 3.6 applies, during the period of forty two days beginning on the earliest of such dates.

# 3.8. Duration of Plan

An Option may not be granted

- 3.8.1. earlier than the Approval Date; nor
- 3.8.2. more than ten years after the Adoption Date.

-8-

## 3.9. Persons to whom Options may be granted

The Grantor may not grant an Option to an individual who is not an Eligible Employee on the Grant Date.

## 3.10. Options non-transferable

An Option shall be personal to the Eligible Employee to whom it is granted and, subject to Rule 7.1, shall not be capable of being transferred, charged or otherwise alienated and shall lapse immediately if the Option Holder purports to transfer, charge or otherwise alienate the Option.

## 4. LIMIT ON AGGREGATE NUMBER OF PLAN SHARES PLACED UNDER OPTION

### 4.1. Power to set limit

The Board may, in its absolute discretion, from time to time set a maximum limit on the total number of Plan Shares which may be placed under Option under the Plan in response to an issue of Invitations (but no such limit shall invalidate any Option granted prior to such limit being set).

## 5. EXERCISE PRICE

The Exercise Price shall be determined by the Board and may be any price but shall not be less than the higher of:

- (a) eighty percent of the Market Value of a Plan Share; and
- (b) in the case of any Option which will be satisfied by the issue of new shares the nominal value of a Plan Share.

## 6. EXERCISE OF OPTIONS

### 6.1. Earliest date for exercise of Options

Subject to Rules 7 and 8, an Option may not be exercised before the Bonus Date.

# 6.2. Latest date for exercise of Options

Subject to Rule 7.1, an Option may not be exercised more than six months after the Bonus Date and if not exercised by that date shall lapse immediately.

## 6.3. Persons who may exercise Options

Subject to Rule 7, an Option may be exercised only while the Option Holder is in Relevant Employment and if an Option Holder ceases to be in Relevant Employment, any Option granted to him shall lapse immediately. This Rule 6.3 shall apply where the Option Holder ceases to be in Relevant Employment in any circumstances (including, in particular, but not by way of limitation, where the Option Holder is dismissed unfairly, wrongfully, in breach of contract or otherwise).

-9-

## 6.4. Material Interest

An Option may not be exercised if the Option Holder then has, or has had within the preceding twelve months, a Material Interest in a Close Company which is the Company or which is a company which has Control of the Company or which is a member of a Consortium which owns the Company.

# 6.5. Number of Plan Shares acquired on exercise of Options

The number of Plan Shares which may be acquired on the exercise of an Option shall be limited to the maximum whole number which may be acquired at the Exercise Price out of the repayment (including any interest or bonus that has been taken into account in determining the number of Plan Shares over which the Option was granted) received by the Option Holder under the Savings Contract linked to the Option.

## 6.6. Options may be exercised in whole or in part

An Option may, to the extent it has become exercisable, be exercised in whole or in part. If exercised in part, the unexercised part of the Option shall lapse.

## 6.7. Procedure for exercise of Options

- 6.7.1. An Option shall be exercised by the Option Holder delivering to the Grantor a duly completed Notice of Exercise in the form from time to time prescribed by the Grantor, specifying the number of Plan Shares in respect of which the Option is being exercised, and accompanied by evidence of the termination of the Savings Contract linked to the Option, payment in full for the Plan Shares (which shall not exceed the repayment, including any interest or bonus, received by the Option Holder under the linked Savings Contract) and, if available, the Option Certificate. Such payment may be made by the Option Holder or by the bank or building society with which the Savings Contract was made.
- 6.7.2. For the avoidance of doubt, the date of exercise of an Option shall be determined in accordance with Rule 16.3. If payment is made by cheque and the cheque fails to clear the Option shall be deemed never to have been exercised.

## 6.8. Issue or transfer of Plan Shares on exercise of Options

Subject to any necessary consents and to compliance by the Option Holder with the Rules, the Grantor shall, as soon as reasonably practicable and in any event not later than thirty days after the date of exercise of the Option, issue or transfer to the Option Holder, or procure the issue or transfer to the Option Holder of, the number of Plan Shares specified in the Notice of Exercise and shall deliver or procure the delivery to the Option Holder of a definitive share certificate in respect of such Plan Shares.

## 6.9. Amount of repayment under Savings Contract

For the purpose of Rules 6.5 and 6.7, the repayment received under a Savings Contract shall exclude the repayment of any contribution the due date for payment of which falls after any date on which the Option Holder ceases to be in Relevant Employment.

-10-

## 7. EXERCISE OF OPTIONS IN SPECIAL CIRCUMSTANCES

### 7.1. Death

Notwithstanding Rules 6.1, 6.2 and 6.3, if an Option Holder dies before the Bonus Date, his personal representatives shall be entitled to exercise his Options at any time during the twelve month period after his death. If not so exercised, the Options shall lapse immediately.

Notwithstanding Rules 6.2 and 6.3, if an Option Holder dies during the period of six months after the Bonus Date, his personal representatives shall be entitled to exercise his Options at any time during the twelve month period after the Bonus Date. If not so exercised, the Options shall lapse immediately.

## 7.2. Injury, disability, redundancy, retirement etc

Subject to Rule 7.5, notwithstanding Rules 6.1 and 6.3, if an Option Holder ceases to be in Relevant Employment by reason of:

- 7.2.1. injury or disability;
- 7.2.2. redundancy within the meaning of the Employment Rights Act 1996;
- 7.2.3. retirement on or after reaching the Specified Age or any other age at which he is bound to retire under the terms of his contract of employment;
- 7.2.4. his office or employment ceasing to be a Relevant Employment because
  - 7.2.4.1. it is in a company which ceases to be a member of the Group; or
  - 7.2.4.2. it relates to a business or part of a business which is transferred to a person who is not a member of the Group

he shall be entitled to exercise his Options at any time during the period of six months after the date he ceased to be in Relevant Employment except that in the case of cessation of employment by reason of a circumstance within Rules 7.2.1, 7.2.2 or 7.2.3 occurring within the six month period after an event to which Rule 7.2.4 applied he shall be entitled to exercise his Options within the six month period after such cessation of employment.

## 7.3. Specified Age

If an Option Holder continues to be employed after the date on which he reaches the Specified Age, he shall be entitled to exercise his Options at any time during the six month period thereafter. If not so exercised, the Options shall not lapse but shall be exercisable or not, as the case may be, in accordance with the rules of the Plan.

# 7.4. Other special circumstances

If an Option Holder ceases to be in Relevant Employment for a reason other than those referred to in Rules 7.1 and 7.2 and within three years after the Grant Date, the Option shall lapse immediately.

If an Option Holder ceases to be in Relevant Employment for a reason other than those referred to in Rules 7.1 and 7.2 and more than three years after the Grant Date, he shall be entitled to exercise the Option at any time during the six month period thereafter. If not so exercised, the Option shall lapse immediately.

-11-

## 7.5. Office or employment in Group Company

If, at the relevant Bonus Date, an Option Holder holds an office or employment in a company which is not a Constituent Company but which is a member of the Group he shall be entitled to exercise his Options at any time during the six month period thereafter.

# 7.6. Termination of Savings Contract

If an Option Holder gives, or is deemed under the terms of his Savings Contract to have given, notice that he intends to cease paying contributions under his Savings Contract, the Option linked to the Savings Contract shall lapse immediately unless the Option has already become exercisable in accordance with the rules of the Plan.

## 7.7. Meaning of ceasing to be in Relevant Employment

For the purpose of Rules 6.3, 7.2, 7.4, and 10.1.2, an Option Holder shall not be treated as ceasing to be in Relevant Employment until he no longer holds any office or employment with a member of the Group.

# 7.8. Interaction of Rules

- 7.8.1. If an Option has become exercisable under Rule 7.2 or 7.3 and, during the period allowed for the exercise of the Option under Rule 7.2 or 7.3, the Option Holder dies, the period allowed for the exercise of the Option shall be the period allowed by Rule 7.1.
- 7.8.2. If an Option has become exercisable under Rule 8.1, 8.2 or 8.3 and, during the period allowed for the exercise of the Option under Rule 8.1, 8.2 or 8.3, the Option becomes exercisable under Rule 7.1 or 7.2 also, the period allowed for the exercise of the Option shall be the period allowed by Rule 7.1 or 7.2, as applicable.

## 8. TAKEOVER, RECONSTRUCTION, AMALGAMATION OR WINDING-UP OF COMPANY

## 8.1. General offer for, or acquisition of, Company

Notwithstanding Rule 6.1, if a person other than a New Holding Company obtains Control of the Company as a result of:

- 8.1.1. making a general offer to acquire the whole of the issued ordinary share capital of the Company which is made on a condition such that if it is satisfied the person making the offer will have Control of the Company; or
- 8.1.2. making a general offer to acquire all the shares in the Company of the same class as the Plan Shares

(in either case, other than any shares already held by him or a person Acting In Concert with him)

all Options may be exercised, subject to Rule 8.2, at any time during the period of six months beginning with the time when the person making the offer or proposed acquisition (as the case may be) has obtained Control of the Company and any condition subject to which the offer or proposed acquisition is made has been satisfied. If not so exercised, the Options shall lapse at the expiry of the six month period.

-12-

## 8.2. Compulsory acquisition of Company

Notwithstanding Rule 6.1, if a person, other than a New Holding Company, becomes entitled or bound to acquire shares in the Company, under the Companies Acts 1963-2012 of Ireland pursuant to provisions that are accepted by HM Revenue & Customs as being closely comparable to sections 979 and 982 of the Companies Act 2006, all Options may be exercised at any time during the period beginning with the date the person serves a notice under the relevant provisions and ending seven clear days before the date on which the person ceases to be entitled to serve such a notice. If not so exercised, the Options shall cease to be exercisable and shall lapse when the person ceases to be entitled to serve such a notice.

## 8.3. Reconstruction or amalgamation of Company

Notwithstanding Rule 6.1, if a person, other than a New Holding Company, obtains Control of the Company in pursuance of a compromise or arrangement sanctioned by the court under the Companies Acts 1963-2012 of Ireland pursuant to provisions that are accepted by HM Revenue & Customs as being closely comparable to section 899 of the Companies Act 2006:

- 8.3.1. Option Holders may give notice to exercise all Options, at any time during the six month period beginning with the date on which the court sanctions the compromise or arrangement;
- 8.3.2. any Options not so exercised shall cease to be exercisable and shall lapse immediately at the end of such six month period;
- 8.3.3. an Option which has already become exercisable may be exercised before the date specified in Rule 8.3.1. Any Option not so exercised shall cease to be exercisable and shall lapse at the end of the six month period following the date on which the court sanctions the compromise or arrangement.

### 8.4. Winding-up of Company

If notice is given of a resolution for the voluntary winding-up of the Company:

- 8.4.1. Option Holders may exercise all Options, at any time during the six month period beginning with the date on which the Company passes the resolution for the voluntary winding-up of the Company;
- 8.4.2. any Options not so exercised shall cease to be exercisable and shall lapse immediately at the end of such six month period; and
- 8.4.3. an Option which has already become exercisable may be exercised before the date specified in Rule 8.4.1. Any Option not so exercised shall cease to be exercisable and shall lapse immediately at the end of the six month period beginning with the date on which the Company passes the resolution for the voluntary winding up of the Company.

### 8.5. Shares subject to Options ceasing to be Plan Shares

If the shares subject to an Option cease to satisfy the conditions in paragraphs 17 to 22 of Schedule 3:

8.5.1. the Grantor shall, as soon as reasonably practicable, notify HM Revenue & Customs;

-13-

- 8.5.2. the Option shall continue to exist and shall continue to be entitled to exemptions from income tax applying to an SAYE option scheme approved under Schedule 3 subject to any determination by HM Revenue & Customs under paragraph 42 of Schedule 3; and
- 8.5.3. the Plan shall continue to exist but, if HM Revenue & Customs withdraw approval of the Plan under Schedule 3, as a non HM Revenue & Customs approved plan.

# 8.6. Meaning of "obtains Control of the Company"

For the purpose of Rule 8, a person shall be deemed to have obtained Control of the Company if he and others Acting In Concert with him have together obtained Control of it.

## 8.7. Notification of Option Holders

The Grantor shall, as soon as reasonably practicable, notify each Option Holder of the occurrence of any of the events referred to in this Rule and explain how this affects his position under the Plan.

# 9. EXCHANGE OF OPTIONS

## 9.1. Circumstances in which Exchange can occur

If the person referred to in Rules 8.1, 8.2 or 8.3, including a New Holding Company, obtains Control of the Company, an Option Holder may, at any time during the period set out in Rule 9.2, by agreement with the Acquiring Company, release his Option in consideration of the grant to him of a new option which is equivalent to the Option but which relates to shares in:

- 9.1.1. the Acquiring Company; or
- 9.1.2. a company which has Control of the Acquiring Company; or
- 9.1.3. a company which either is, or has Control of, a company which is a member of a Consortium which owns either the Acquiring Company or a company having Control of the Acquiring Company.

# 9.2. Period allowed for exchange of Options

The period referred to in Rule 9.1 is:

- 9.2.1. where Rule 8.1 applies or would apply if the reference in that Rule to "person" was read as "person including a New Holding Company", the period referred to in that Rule;
- 9.2.2. where Rule 8.2 applies, the period during which the Acquiring Company remains so entitled or bound; and
- 9.2.3. where Rule 8.3 applies, the period of six months beginning with the time when the court sanctions the compromise or arrangement.

# 9.3. Meaning of "equivalent"

The New Option shall not be regarded for the purpose of this Rule 9 as equivalent to the Option unless:

9.3.1. the New Plan Shares satisfy the conditions in paragraphs 17 to 22 of Schedule 3; and

-14-

- 9.3.2. the New Option will be exercisable in the same manner as the Option and subject to the provisions of the Plan as it had effect immediately before the release of the Option;
- 9.3.3. the total market value, immediately before the release of the Option, of the Plan Shares which were subject to the Option is as nearly as may be equal to the total market value, immediately after the grant of the New Option, of the New Plan Shares subject to the New Option (market value being determined for this purpose in accordance with Part VIII of the Taxation of Chargeable Gains Act 1992); and
- 9.3.4. the total amount payable by the Option Holder for the acquisition of the New Plan Shares under the New Option is as nearly as may be equal to the total amount that would have been payable by the Option Holder for the acquisition of the Plan Shares under the Option.

## 9.4. Grant Date of New Option

The Grant Date of the New Option shall be deemed to be the same as the Grant Date of the Option.

### 9.5. Application of Plan to New Option

In the application of the Plan to the New Option, where appropriate, references to "Company" and "Plan Shares" shall be read as if they were references to the company to whose shares the New Option relates and the New Plan Shares, respectively, save that, where appropriate, in the definition of "Board" the reference to "Company" shall be read as if it were a reference to Mallinckrodt public limited company. An exchange of Options pursuant to this Rule 9 shall not alter the fact that this Plan remains that of the Company as the original scheme organiser.

## 10. LAPSE OF OPTIONS

An Option shall lapse on the earliest of:

- 10.1.1. subject to Rule 7.1, six months after the Bonus Date;
- 10.1.2. subject to Rules 7.1, 7.2 and 7.4, the Option Holder ceasing to be in Relevant Employment;
- 10.1.3. the date on which it is provided that the Option shall lapse under Rules 7.1, 7.2 and 7.4 and 8.1 to 8.4;
- 10.1.4. the date on which the Option Holder becomes bankrupt or enters into a compromise with his creditors generally;
- 10.1.5. before an Option has become capable of being exercised, the Option Holder giving notice that he intends to stop paying monthly contributions, or being deemed under the terms of the Savings Contract to have given such notice or making an application for the repayment of his aggregate monthly contributions; and
- 10.1.6. the date on which the Option Holder purports to transfer, charge or otherwise alienate the Option.

-15-

## 11. ADJUSTMENT OF OPTIONS ON REORGANISATION

## 11.1. Power to adjust Options

In the event of a Reorganisation, the number of Plan Shares subject to an Option, the description of the Plan Shares, the Exercise Price, or any one or more of these, may be adjusted in such manner as the Board or, where the Trustees are the Grantor, the Trustees and the Board together determine.

### 11.2. Exercise Price

Subject to Rule 11.3, no adjustment shall be made to the Exercise Price which would result in the Plan Shares subject to an Option being issued directly to the Option Holder at a price per Plan Share lower than the nominal value of a Plan Share and, if an adjustment would so result, the Exercise Price shall be the nominal value of a Plan Share.

## 11.3. Capitalisation of reserves

Notwithstanding Rule 11.2, an adjustment may be made which would result in the Plan Shares subject to an Option being issued at a price per Plan Share lower than the nominal value of a Plan Share if and to the extent that the Board is authorised to capitalise from the Company's reserves a sum equal to the amount by which the aggregate nominal value of the Plan Shares subject to the Options which are adjusted exceeds the aggregate adjusted Exercise Price under such Options. If such an adjustment is made, on the subsequent exercise of the Option, the Board shall capitalise such sum and apply the sum in paying up such exceeds.

### 11.4. HM Revenue & Customs approval

An adjustment shall not have effect until the adjustment has been approved by HM Revenue & Customs.

### 11.5. Notification of Option Holders

The Grantor shall, as soon as reasonably practicable, notify each Option Holder of any adjustment made under this Rule 11 and explain how this affects his position under the Plan. The Grantor may call in for endorsement or cancellation and re-issue any Option Certificate in order to take account of such adjustment.

## 12. ISSUE AND AVAILABILITY OF PLAN SHARES

## 12.1. Rights attaching to Plan Shares

All Plan Shares issued in respect of exercise of an Option shall, as to voting, dividend, transfer and other rights, including those arising on a liquidation of the Company, rank equally in all respects and as one class with the Plan Shares in issue at the date of such issue save as regards any rights attaching to such Plan Shares by reference to a record date prior to the date of such issue.

# 12.2. Availability of Plan Shares

The Company shall at all times use its reasonable endeavours to keep available sufficient authorised but unissued Plan Shares to satisfy the exercise of all Options which the Board has determined will be satisfied by the issue of Plan Shares (whether directly to the Option Holder or indirectly via the Trustees).

-16-

## 13. RELATIONSHIP OF PLAN TO CONTRACT OF EMPLOYMENT

## 13.1. Contractual Provisions

Notwithstanding any other provision of the Plan:

- 13.1.1. the Plan shall not form part of any contract of employment between any Group Member and an Eligible Employee;
- 13.1.2. unless expressly so provided in his contract of employment, an Eligible Employee has no right to be granted an Option;
- 13.1.3. the benefit to an Eligible Employee of participation in the Plan (including, in particular but not by way of limitation, any Options held by him) shall not form any part of his remuneration or count as his remuneration for any purpose and, for the purposes of his contract of employment, shall not be pensionable; and
- 13.1.4. if an Eligible Employee ceases to be in Relevant Employment, he shall not be entitled to compensation for the loss of any right or benefit or prospective right or benefit under the Plan (including, in particular but not by way of limitation, any Options held by him which lapse by reason of his ceasing to be in Relevant Employment) whether by way of damages for unfair dismissal, wrongful dismissal, breach of contract or otherwise.

By applying for an Option an Option Holder is deemed to have agreed to the provisions of this Rule 13.

### 14. ADMINISTRATION OF PLAN

### 14.1. Responsibility for administration

The Company, and the Grantor where appropriate, shall be responsible for, and shall have the conduct of, the administration of the Plan. The Grantor may from time to time make, amend or rescind regulations for the administration of the Plan provided that such regulations shall be consistent with the Rules and not cause any of the provisions of Schedule 3 which are relevant to the Plan to cease to be satisfied.

## 14.2. Grantor's decision final and binding

The decision of the Grantor shall be final and binding in all matters relating to the administration of the Plan, including but not limited to the resolution of any dispute concerning, or any inconsistency or ambiguity in the Rules or any document used in connection with the Plan.

## 14.3. Trustees to consult with Board

Where the Trustees have granted, or propose to grant, an Option, the Trustees shall consult with, and take account of the wishes of, the Board before making any determination or exercising any power or discretion under the Plan.

## 14.4. Provision of information

The Trustees and an Option Holder shall provide to the Company as soon as reasonably practicable such information as the Company reasonably requests for the purpose of complying with its obligations under paragraph 45 of Schedule 3.

-17-

## 14.5. Cost of Plan

The cost of introducing and administering the Plan shall be met by the Company. The Company shall be entitled, if it wishes, to charge an appropriate part of such cost to a Subsidiary. The Company shall also be entitled, if it wishes, to charge to a Subsidiary the opportunity cost of issuing Plan Shares to an Option Holder employed by the Subsidiary in relation to his exercise of an Option.

## 14.6. Establishment of separate plans for overseas territories

The Company may establish separate plans to operate in overseas territories or in respect of overseas employees which are on substantially the same terms as the Plan but which make such modifications to the terms as are necessary or expedient to take account of local tax, exchange control or securities laws in any one or more overseas territories (a "Modified Plan"). Rule 4 shall apply so as to limit the number of Plan Shares which may be placed under Option under a Modified Plan and Plan Shares placed under an Option granted under a Modified Plan shall be included for the purpose of the limit set out in Rule 4.

For the avoidance of doubt, such plans shall not be intended to be subject to HM Revenue & Customs approval under Schedule 3 and no modifications made in accordance with this clause shall affect the Plan.

## 14.7. Data protection

The Company, the Trustees and their agents may accumulate, hold and process Eligible Employees' personal data and/or "sensitive personal data" within the meaning of applicable law ("Personal Data"). Personal Data includes, but is not limited to, the information provided to Eligible Employees as part of the Invitation and any changes thereto, other appropriate personal and financial data about the Eligible Employees (e.g., name, home address, telephone number, date of birth, nationality, job title, reason for termination of employment, and national insurance number), and information about the Eligible Employees' participation in the Plan and Plan Shares obtained under the Plan from time to time.

By accepting Options or Plan Shares under this Plan, the Eligible Employees give their explicit consent to their employer's, the Trustees' and the Company's accumulating, transferring and processing Personal Data as necessary or appropriate for Plan administration. Eligible Employees' Personal Data will be retained only as long as is necessary to administer their participation in the Plan. If applicable, by accepting Options or Plan Shares, Eligible Employees also gives their explicit consent to the Company's transfer of Personal Data outside the country in which they work or reside and to the United States of America, where the same level of data protection laws may not apply as in their home country. The legal persons for whom the Eligible Employees' Personal Data are intended (and by whom their Personal Data may be transferred, processed or exchanged) include the Company, its Subsidiaries (or former Subsidiaries as are deemed necessary), the Trustees, their respective agents, and any other person that the Company retains or utilises for compensation planning or Plan administration purposes. The Eligible Employees have the right to request a list of the names and addresses of any potential recipients of their Personal Data and to review and correct their Personal Data by contacting their local Human Resources Representative.

By accepting Options or Plan Shares under this Plan, the Eligible Employees acknowledge their understanding that the transfer of the information outlined here is important to Plan administration. By accepting Options or Plan Shares under this Plan, Eligible Employees acknowledge that they are providing the consents herein on a purely voluntary basis and that, if they do not consent or if they later seek to revoke their consent, it will adversely impact the ability of the Company to administer the Plan but it will not adversely impact their employment status or service with their employer.

-18-

# **15. AMENDMENT OF PLAN**

# 15.1. Power to amend Plan

Subject to Rules 15.2 to 15.3, the Board may from time to time amend the rules of the Plan.

# 15.2. HM Revenue & Customs approval of amendments

Save for an amendment pursuant to Rule 8.5, an amendment to a Key Feature of the Plan shall not have effect at a time when the Plan is approved by HM Revenue & Customs, until the amendment has been approved by HM Revenue & Customs under Schedule 3.

# 15.3. Rights of existing Option Holders

An amendment may not adversely affect the rights of an existing Option Holder except where the amendment has been approved by those existing Option Holders who would be adversely affected by the amendment in such manner as would be required by the Company's articles of association (with appropriate changes) if the Plan Shares subject to those Options which would be so adversely affected had been issued or transferred to them (so that they had become shareholders in the Company) and constituted a separate class of shares.

# 15.4. Notification of Option Holders

The Board shall, as soon as reasonably practicable, notify each Option Holder of any amendment to the Rules under this Rule 15 and explain how it affects his position under the Plan.

# 16. NOTICES

# 16.1. Notice by Grantor

Save as provided for by law and subject to Rule 16.4, any notice, document or other communication given by, or on behalf of, the Grantor or to any person in connection with the Plan shall be deemed to have been duly given if delivered to him at his place of work, if he is in Relevant Employment if sent by e-mail to such e-mail address as may be specified by him from time to time, or sent through the post in a pre-paid envelope to the postal address last known to the Company to be his address and, if so sent, shall be deemed to have been duly given on the date of posting.

# 16.2. Deceased Option Holders

Save as provided for by law and subject to Rule 16.4, any notice, document or other communication so sent to an Option Holder shall be deemed to have been duly given notwithstanding that such Option Holder is then deceased (and whether or not the Grantor has notice of his death) except where his personal representatives have established their title to the satisfaction of the Grantor and supplied to the Grantor an e-mail or postal address to which notices, documents and other communications are to be sent.

# 16.3. Notice to Grantor

Save as provided for by law and subject to Rule 16.4, any notice, document or other communication given to the Grantor in connection with the Plan shall be delivered or sent by post to the Company Secretary at the Company's registered office or such other e-mail or postal address as may from time to time be notified to Option Holders but shall not in any event be duly given unless and until it is actually received at the registered office or such e-mail or postal address and shall be deemed to have been duly given on the date of such receipt.

-19-

#### 16.4. Option Certificate and Notice of Option

For the avoidance of doubt, the Option Certificate and Notice of Option may not be executed or delivered by e-mail or other such similar electronic communication.

# 17. GOVERNING LAW AND JURISDICTION

#### 17.1. Plan governed by English law

The formation, existence, construction, performance, validity and all aspects whatsoever of the Plan, any term of the Plan and any Option granted under it shall be governed by English law.

# 17.2. English courts to have jurisdiction

The English courts shall have jurisdiction to settle any dispute which may arise out of, or in connection with, the Plan.

#### 17.3. Jurisdiction agreement for benefit of Company

The jurisdiction agreement contained in this Rule 17 is made for the benefit of the Company only, which accordingly retains the right to bring proceedings in any other court of competent jurisdiction.

# 17.4. Option Holder deemed to submit to such jurisdiction

By executing and returning the Option Certificate to the Grantor, an Option Holder is deemed to have agreed to submit to such jurisdiction.

-20-

# Subsidiaries of Mallinckrodt plc

Set forth below is a list of subsidiaries that will be transferred by Covidien plc and its subsidiaries to Mallinckrodt plc in connection with the separation. Unless otherwise indicated, all of the subsidiaries listed below will be wholly owned subsidiaries of Mallinckrodt plc and will be owned directly by either Mallinckrodt plc or by wholly owned subsidiaries of Mallinckrodt plc.

Name of Subsidiary	Jurisdiction of Formation
Mallinckrodt Medical Argentina Ltd., Argentinean Branch	Argentina
Mallinckrodt Australia Pty Ltd	Australia
Mallinckrodt Belgium BVBA	Belgium
Carnforth Limited	Bermuda
Mallinckrodt do Brasil, Ltda.	Brazil
Mallinckrodt Canada Holdings ULC	Canada
Mallinckrodt Canada ULC	Canada
Comercializadora Mallinckrodt Chile Limitada	Chile
Mallinckrodt Medical Consulting (Shanghai) Co., Ltd.	China
Mallinckrodt Medical Consulting (Shanghai) Co., Ltd. (Beijing Branch)	China
Mallinckrodt Colombia SAS	Colombia
Mallinckrodt Netherlands Holdings B.V., organizacní složka	Czech Republic
Mallinckrodt Netherlands Holdings B.V. Holland (Denmark Branch)	Denmark
Mallinckrodt Netherlands Holdings B V Finland Branch	Finland
Covidien Imaging France Sarl (FKA Mallinckrodt France S.a.r.l.)	France
Dritte CORSA Verwaltungsgesellschaft GmbH	Germany
Mallinckrodt Deutschland GmbH	Germany
Mallinckrodt Deutschland Holdings GmbH	Germany

Mallinckrodt Hong Kong Limited Mallinckrodt Pharmaceuticals India Private Limited Mallinckrodt Ireland Limited Mallinckrodt Medical Imaging - Ireland Mallinckrodt plc Mallinckrodt Italia Spa Mallinckrodt Japan Co. Ltd. Mallinckrodt Korea Inc. Mallinckrodt Group Sarl Mallinckrodt International Finance SA Mallinckrodt Medical S.A. de C.V. Mallinckrodt Canada Cooperatie U.A. Mallinckrodt Medical B.V. Mallinckrodt Nederland B.V. Mallinckrodt Netherlands Holdings BV Mallinckrodt Petten Holdings B.V. Mallinckrodt Netherlands Holdings B.V. (Norway Branch) Mallinckrodt Panama Distribution, S.A. Mallinckrodt Panama, S.A. Mallinckrodt sp. z o.o. Mallinckrodt Caribbean, Inc. (Puerto Rico Branch) Mallinckrodt Netherlands Holdings B.V. Russian Representative Office Mallinckrodt Netherlands Holdings B.V. Slovakia, organizacná zložka Mallinckrodt (Pty) Ltd

Hong Kong India Ireland Ireland Ireland Italy Japan Korea Luxembourg Luxembourg Mexico Netherlands Netherlands Netherlands Netherlands Netherlands Norway Panama Panama Poland Puerto Rico Russia Slovakia South Africa

Mallinckrodt Spain, S.L.	Spain
Mallinckrodt Sverige AB	Sweden
Covidien Finance GmbH	Switzerland
Mallinckrodt AG	Switzerland
Mallinckrodt Group Sarl, Luxembourg (LU) Neuhausen AM Rheinfall Branch	Switzerland
Mallinckrodt Holdings GmbH	Switzerland
Mallinckrodt Switzerland Limited	Switzerland
Mallinckrodt Hong Kong Limited, Taiwan Representative Office	Taiwan
Mallinckrodt Hong Kong Limited, Thailand Branch	Thailand
Mallinckrodt Saglik Anonim Sirketi	Turkey
Mallinckrodt Chemical Holdings (UK) Ltd.	UK
Mallinckrodt Chemical Limited	UK
Mallinckrodt Medical Argentina Ltd.	UK
Mallinckrodt Medical Holdings (UK) Limited	UK
Mallinckrodt UK Commercial Ltd	UK
Mallinckrodt UK Ltd	UK
MKG Medical UK Ltd	UK
CNS Therapeutics, Inc.	Delaware
Enterpises Holdings, Inc.	Delaware
IMC Exploration Company	Maryland
Lafayette Pharmaceuticals LLC	Delaware
Liebel-Flarsheim Company LLC	Delaware
Ludlow Corporation	Massachusetts
Mallinckrodt Brand Pharmaceuticals, Inc. (DE)	Delaware

Mallinckrodt Caribbean, Inc.
Mallinckrodt Enterprises Holdings, Inc.
Mallinckrodt Enterprises LLC
Mallinckrodt Inc. (DE)
Mallinckrodt LLC
Mallinckrodt US Holdings, Inc. (FKA Kendall Holding Corp.)
Mallinckrodt US Holdings LLC
Mallinckrodt US Pool LLC
Mallinckrodt Veterinary, Inc.
MEH, Inc.

Delaware California Delaware Delaware Nevada Delaware Nevada Delaware Nevada



#### Dear Covidien Shareholder:

, 2013

Previously, we announced plans to spin off our Pharmaceuticals business into a separate, publicly traded company, which we have named Mallinckrodt plc. As two distinct businesses, Covidien and Mallinckrodt will be better positioned to capitalize on significant growth opportunities and provide greater focus on their respective businesses and strategic priorities.

Both of these companies have businesses with industry-leading products and services. Following the separation, Covidien will continue to be a global medical devices and supplies company with products and services designed to enhance the quality of life for patients and improve outcomes for our customers, and Mallinckrodt will be a leading developer, manufacturer and distributor of specialty pharmaceutical products. As independent, publicly owned companies, Covidien and Mallinckrodt each will be able to pursue and focus on its own strategic and operational plans, including setting an optimal level of investment in research and development projects and in the operation and expansion of its businesses and creating a business-appropriate capital structure.

The separation will provide current Covidien shareholders with ownership interests in both Covidien and Mallinckrodt. The distribution is subject to certain conditions, including the continued validity of the private letter ruling received from the United States Internal Revenue Service and the receipt of an opinion of tax counsel confirming that the distribution and certain transactions entered into in connection with the distribution generally will be tax-free to Covidien and its shareholders for U.S. federal income tax purposes, except for cash received in lieu of fractional shares.

As a result of the separation, each Covidien shareholder will receive one ordinary share of Mallinckrodt for every eight Covidien ordinary shares held on June 19, 2013, the record date for the distribution, with cash being paid in lieu of fractional shares. You do not need to take any action to receive ordinary shares of Mallinckrodt to which you are entitled as a Covidien shareholder. You do not need to pay any consideration or surrender or exchange your Covidien ordinary shares.

I encourage you to read the attached information statement, which is being provided to all Covidien shareholders who hold ordinary shares on June 19, 2013. The information statement describes the separation in detail and contains important business and financial information about Mallinckrodt.

I believe the separation is a positive progression for our businesses and our shareholders. We remain committed to working on your behalf to continue to build long-term shareholder value.

Sincerely,

José E. Almeida Chairman of the Board, President and Chief Executive Officer Covidien plc

, 2013

Dear Future Mallinckrodt Shareholder:

On behalf of the entire Mallinckrodt plc team, I welcome you as a future shareholder.

We are a global company that develops, manufactures, markets and distributes both branded and generic specialty pharmaceuticals, active pharmaceutical ingredients and diagnostic imaging agents. Our specialty pharmaceuticals products are sold to major wholesalers and retail drug store chains. We use our active pharmaceutical ingredients products in the manufacture of our generic pharmaceuticals and also sell them to other pharmaceutical companies. We market our global medical imaging products primarily to physicians, technologists and purchasing administrators at hospitals, imaging centers, cardiology clinics and radiopharmacies.

As an independent company, we will be able to pursue our own strategic and operational plans, including setting an optimal level of investment in research and development projects and in the operation and expansion of our businesses and creating a business-appropriate capital structure. We anticipate that this will improve our ability to invest in our business and continue to develop innovative new products, pursue strategic transactions, enhance our market recognition with investors and increase our ability to attract and retain employees by providing compensation more directly tied to our business results. Our focused management team is highly motivated to make a difference in healthcare, as we enhance value for our customers and shareholders.

I encourage you to learn more about us and our strategic initiatives by reading the attached information statement. We have received authorization to list our ordinary shares on the New York Stock Exchange under the symbol "MNK."

We look forward to serving our customers and patients, as well as rewarding our shareholders, as we begin a new and exciting chapter for our company.

Sincerely,

Mark Trudeau President and Chief Executive Officer Mallinckrodt plc Information contained herein is subject to completion or amendment. A Registration Statement on Form 10 relating to these securities has been filed with the United States Securities and Exchange Commission under the United States Securities Exchange Act of 1934, as amended.

# Preliminary and Subject to Completion, dated May 31, 2013 INFORMATION STATEMENT Mallinckrodt plc

This information statement is being furnished in connection with the distribution of ordinary shares of Mallinckrodt plc ("Mallinckrodt"), which will hold the Pharmaceuticals business of Covidien plc ("Covidien"), to Covidien's shareholders.

For every eight ordinary shares of Covidien held of record by you as of the close of business on June 19, 2013, the record date for the distribution (the "record date"), you will receive one ordinary share of Mallinckrodt. You will receive cash in lieu of any fractional ordinary shares of Mallinckrodt which you would have received after application of the above ratio. We expect our ordinary shares to be distributed to you on June 28, 2013. We refer to the date of the distribution of our ordinary shares as the "distribution date." As discussed under "The Separation—Trading Between the Record Date and Distribution Date," if you sell your ordinary shares of Covidien in the "regular-way" market after the record date and before the distribution date, you also will be selling your right to receive ordinary shares of Mallinckrodt in connection with the separation.

The distribution is intended to be tax-free to Covidien shareholders for United States federal income tax purposes, except for cash received in lieu of fractional shares. The distribution is subject to certain conditions, including the continued validity of the private letter ruling received from the U.S. Internal Revenue Service ("IRS") and the receipt of an opinion of tax counsel confirming that the distribution and certain transactions entered into in connection with the distribution generally will be tax-free to Covidien and its shareholders for U.S. federal income tax purposes, except for cash received in lieu of fractional shares.

No further vote of Covidien's shareholders is required in connection with the separation. Therefore, you are not being asked for a proxy, and you are requested not to send us a proxy, in connection with the separation. You do not need to pay any consideration, exchange or surrender your existing ordinary shares of Covidien or take any other action to receive your ordinary shares of Mallinckrodt.

There is no current trading market for our ordinary shares, although we expect that a limited market, commonly known as a "when-issued" trading market, will develop on or shortly before the record date for the distribution, and we expect "regular-way" trading of our ordinary shares to begin on the first trading day following the completion of the separation. We have received authorization to list our ordinary shares on the New York Stock Exchange ("NYSE") under the symbol "MNK."

# In reviewing this information statement, you should carefully consider the matters described under "<u>Risk Factors</u>" beginning on page 20.

Neither the U.S. Securities and Exchange Commission nor any state securities commission has approved or disapproved these securities or determined if this information statement is truthful or complete. Any representation to the contrary is a criminal offense.

This information statement does not constitute an offer to sell or the solicitation of an offer to buy any securities.

This document is not a prospectus within the meaning of Part 5 of the Investment Funds, Companies and Miscellaneous Provisions Act 2005 of Ireland (as amended) or the Prospectus Directive (2003/71/EC). No offer of shares to the public is made, or will be made, that requires the publication of a prospectus pursuant to Irish prospectus law (within the meaning of Part 5 of the Investment Funds, Companies and Miscellaneous Provisions Act 2005 of Ireland, as amended) or the Prospectus Directive (2003/71/EC). This document has not been approved or reviewed by or registered with the Central Bank of Ireland. This document does not constitute investment advice or the provision of investment services within the meaning of the European Communities (Markets in Capital Instruments) Regulations 2007 of Ireland (as amended) or the Markets in Financial Instruments Directive (2004/39/EC). Neither Covidien nor Mallinckrodt is an authorized investment firm within the meaning of the European Communities (Markets Financial Instruments) Regulations 2007 of Ireland (as amended) or the Markets in Financial Instruments Of the recipients of this document should seek independent legal and financial advice in determining their actions in respect of or pursuant to this document.

The date of this information statement is , 2013.

This information statement was first mailed to Covidien shareholders on or about , 2013.

# TABLE OF CONTENTS

	Page
Questions and Answers about the Separation	1
Information Statement Summary	8
Summary Historical and Unaudited Pro Forma Combined Financial Data	17
Risk Factors	20
Cautionary Statement Concerning Forward-Looking Statements	41
Dividends	42
Capitalization	43
Unaudited Pro Forma Condensed Combined Financial Statements	45
Selected Historical Combined Financial Data	51
Business	53
Management's Discussion and Analysis of Financial Condition and Results of Operations	81
Management	111
Compensation Discussion and Analysis	120
Executive Compensation	131
Director Compensation	153
Certain Relationships and Related Person Transactions	154
Security Ownership of Certain Beneficial Owners and Management	155
<u>The Separation</u>	156
Our Relationship with Covidien Following the Distribution	162
Material Tax Consequences	167
Indebtedness	175
Description of Mallinckrodt's Share Capital	177
Where You Can Find More Information	192
Index to Combined Financial Statements	F-1
Relevant Territories	Annex A

# **Presentation of Information**

Except as otherwise indicated or unless the context otherwise requires, the information included in this information statement about Mallinckrodt assumes the completion of all of the transactions referred to in this information statement in connection with the separation. Unless the context otherwise requires, references in this information statement to "Mallinckrodt plc," "Mallinckrodt public limited company," "Mallinckrodt Pharmaceuticals," "Mallinckrodt," "we," "us," "our," "our company" and "the company" refer to Mallinckrodt plc, an Irish public limited company, and its combined subsidiaries. Unless the context otherwise requires, references to Mallinckrodt's historical business and operations refer to the business and operations of Covidien's Pharmaceuticals business as it was historically managed as part of Covidien and its subsidiaries prior to completion of the separation. Unless the context otherwise requires, references in this information statement to "Covidien" refer to Covidien plc, an Irish public limited company, and its consolidated subsidiaries, including the Pharmaceuticals business prior to completion of the separation. References in this information statement to the "separation" refer to the Pharmaceuticals business from Covidien's other businesses and the creation, as a result of the distribution, of an independent, publicly traded company, Mallinckrodt, to hold the assets and liabilities associated with the Pharmaceuticals business after the distribution. References in this information statement to the "distribution" refer to the dividend on Covidien ordinary shares outstanding on the record date that will be satisfied by Mallinckrodt's issuance of its ordinary shares to the persons entitled to receive the dividend. References to "dollars" or "\$" refer to U.S. dollars.

i

#### **Trademarks and Trade Names**

We own or have rights to use the trademarks and trade names that we use in conjunction with the operation of our business. One of the more important trademarks that we own or have rights to use that appears in this information statement is "Mallinckrodt," which is a registered trademark or the subject of pending trademark applications in the U.S. and other jurisdictions. Solely for convenience, we only use the  $^{\text{m}}$  or  $^{\text{m}}$  symbols the first time any trademark or trade name is mentioned. Such references are not intended to indicate in any way that we will not assert, to the fullest extent permitted under applicable law, our rights to our trademarks and trade names. Each trademark or trade name of any other company appearing in this information statement is, to our knowledge, owned by such other company.

# Use of Certain Terms

The following is a list indicating where certain terms that we use in this information statement are defined:

	0		•
acetaminophen	9	Healthcare Reform Act	26
ADHD Adjusted EBITDA	9 17	HEU hydrocodone	12 9
AIP	133	IDNs	22
ANDA	24	IMC	78
API	8	Indenture	175
AUS	79	intrathecal	57
BPCA	67	IPR&D	107
C.A.R.E.S. Alliance	56	IRB	67
Cardinal Health	22	IRS	Cover
CAT	174	IRS ruling	5
cGMP	23	LEU	12
CMDS	9	Maine Board	F-39
CNS Therapeutics	11	Mallinckrodt Baker	53
Code	5	Mallinckrodt SIP	121
Compensation Committee	120	McKesson	22
Computershare	2	MDEP	F-39
Covidien 2012 AIP	123	Medicare Part D	70
Covidien Change in Control Plan	139	methylphenidate	9
Covidien Compensation Committee	120	MIFSA	6
Covidien Retirement Savings Plan	128	Millsboro Site	79
Covidien Severance Plan	139	Mo-99	9
Covidien Supplemental Savings Plan	128	MRI	9
credit facility	175	Mutual	24
CSA	20	MVI	79
CT	9	named executive officers	120
DEA	9	NCE	66
Depomed	64	NDA	56
Deposited Securities	173	notes	175
distribution date	Cover	Novartis	53
DOJ	68	NRC	9
DSAs	106	NYSE	Cover
DTC	2	Octreoscan™	9
Duexis®	53	OIG	71
DWT	171	Optimark™	9
DWT Forms	172	options	133
EPA	30	Optiray™	9
Exalgo®	8	Optivantage™ DH	58
Exchange Act FCPA	32 28	OTFC	84 38
FDA	28	our board	38 9
FFDCA	65	oxycodone Paragraph IV certification	24
Form 10	162	parent	24 F-8
FTS	84	Pennsaid®	83
funded amount	127	performance units	133
GAAP	17	PMDA	73
Gablofen®	8	PRPs	79 79
GBCA	59	PSUs	133
GCP	65	R&D	10
GE	F-39	record date	Cover
General Dynamics	79	Recoupment Policy	129
generic exclusivity	68	REMS	60
GLP	65	Restoril™	53
Government Agencies	79	restricted units	133
GPOs	9	RFID	58

RI/FS	
RLD	
Roxicodone®	
RSUs	
S&P 500	
SEC	
Securities Act	
separation and distribution agreement	
Site	
SOM	
SPECT	
Sumavel® DosePro®	
tax matters agreement	

79	tax opinion	5
67	Tc-99m	9
82	TCE	79
133	Technescan MAG3™	9
39	the board	39
3	TLR	122
39	Tofranil-PM <sup>™</sup>	53
14	TussiCaps™	83
79	Tyco International	35
25	Tyco tax sharing agreement	35
21	U.S. Holder	156
53	Ultra-Technekow™ DTE	9
25		

iv

# QUESTIONS AND ANSWERS ABOUT THE SEPARATION

What is Mallinckrodt and why is Covidien separating its Pharmaceuticals business and distributing Mallinckrodt's ordinary shares?	Mallinckrodt was incorporated in Ireland on January 9, 2013 for the purpose of holding Covidien's Pharmaceuticals business following the separation. The separation of Covidien's Pharmaceuticals business from Covidien and the distribution of Mallinckrodt ordinary shares to Covidien shareholders are intended to provide you with equity investments in two separate companies that will be able to focus on each of their respective businesses. We expect that the separation will result in enhanced long-term performance of each business for the reasons discussed in "The Separation—Background" and "The Separation—Reasons for the Separation."
Why am I receiving this document?	Covidien is delivering this document to you because you were a holder of ordinary shares of Covidien on the record date of June 19, 2013, and are entitled to receive one ordinary share of Mallinckrodt for every eight ordinary shares of Covidien that you held at the close of business on the record date. We will not issue fractional shares in the distribution; you will receive cash in lieu of any fractional ordinary shares of Mallinckrodt which you would have received after application of the above ratio. This document will help you understand how the separation will affect your investment in Covidien and your investment in Mallinckrodt after the separation.
How will the separation work?	Currently, all of Mallinckrodt's issued shares are held beneficially by an Irish corporate services provider. Prior to the transfer by Covidien to us of our business, which will occur prior to the distribution, we will have no business operations. Covidien will transfer its Pharmaceuticals business to us in return for which we will issue shares to Covidien ordinary shareholders, pro rata to their respective holdings. For the purposes of Irish law, this will be treated as Covidien having made a dividend in specie, or a non-cash dividend, to its ordinary shareholders. In connection with these transactions, we will acquire the shares held beneficially by the Irish corporate services provider referred to above for no consideration and cancel these shares. Immediately following the distribution, the persons entitled to receive Mallinckrodt ordinary shares in the distribution will own all of our outstanding ordinary shares.
Why is the separation of Mallinckrodt structured in this manner?	Covidien believes that a distribution of Mallinckrodt ordinary shares that is tax-free to Covidien shareholders for U.S. federal income tax purposes is an efficient way to separate the Pharmaceuticals business of Covidien in a manner that will create long-term value for Covidien, Mallinckrodt and their respective shareholders.
What is the record date for the distribution?	The record date for the distribution will be June 19, 2013.
When will the distribution occur?	We expect the distribution of our ordinary shares to occur on June 28, 2013, to holders of record of ordinary shares of Covidien at the close of business on the record date.

-1-

What do shareholders need to do to participate in the distribution?	Shareholders of Covidien as of the record date will not be required to take any action to receive Mallinckrodt ordinary shares in the distribution, but you are urged to read this entire information statement carefully. No shareholder approval of the distribution is required. You are not being asked for a proxy. You do not need to pay any consideration, exchange or surrender your existing ordinary shares of Covidien or take any other action to receive your ordinary shares of Mallinckrodt.
Will I receive physical certificates representing ordinary shares of Mallinckrodt following the separation?	No. Following the separation, we will not issue physical certificates representing our ordinary shares. If you own ordinary shares of Covidien as of the close of business on the record date, Covidien, with the assistance of Computershare Trust Company N.A. ("Computershare"), the distribution agent, will electronically distribute ordinary shares to you in book-entry form by way of registration in the "direct registration system" (if you hold the shares in your own name as a registered shareholder) or to your bank or brokerage firm on your behalf or through the systems of the Depository Trust Company ("DTC") (if you hold the shares through a bank or brokerage firm that uses DTC). Computershare will mail you a book-entry account statement that reflects your ordinary shares of Mallinckrodt, or your bank or brokerage firm will credit your account for the Mallinckrodt ordinary shares. See "The Separation—When and How You Will Receive Mallinckrodt Ordinary Shares in the Distribution."
How many ordinary shares of Mallinckrodt will I receive the distribution?	<i>in</i> You will receive one ordinary share of Mallinckrodt for every eight ordinary shares of Covidien held at the record date. Based on approximately 458 million Covidien ordinary shares outstanding as of May 24, 2013, a total of approximately 57 million ordinary shares of Mallinckrodt will be distributed. For additional information on the distribution, see "The Separation."
<i>Will Mallinckrodt issue fractional ordinary shares in the distribution?</i>	No. We will not issue fractional shares in the distribution. Fractional shares that Covidien shareholders would otherwise have been entitled to receive will be aggregated and sold in the public market by the distribution agent. The aggregate net cash proceeds of these sales will be distributed ratably to those shareholders who would otherwise have been entitled to receive fractional shares.
What are the conditions to the distribution?	The distribution is subject to the following conditions, among others:
	• the continued validity of the private letter ruling received from the IRS and the receipt of an opinion of tax counsel confirming that the distribution and certain transactions entered into in connection with the distribution generally will be tax-free to Covidien and its shareholders for U.S. federal income tax purposes except for cash received in lieu of fractional shares;
	<ul> <li>no order, injunction or decree issued by any court of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the separation or any of the related transactions being in effect;</li> </ul>

-2-

	• the U.S. Securities and Exchange Commission ("SEC") declaring effective the registration statement of which this information statement forms a part, with no order suspending the effectiveness of the registration statement in effect and no proceedings for such purposes pending before or threatened by the SEC;
	• the mailing of this information statement to the holders of Covidien ordinary shares as of the record date for the distribution; and
	• no other event or development existing or having occurred that, in the judgment of Covidien's board of directors, in its sole discretion, makes it inadvisable to effect the separation and other related transactions.
	As of the date of this information statement, the following additional conditions have been satisfied:
	• the debt financing contemplated to be obtained in connection with the separation, as described in the separation and distribution agreement, having been obtained;
	• the receipt of opinions, in form and substance acceptable to Covidien in its sole discretion and from an independent firm acceptable to Covidien in its sole discretion, with respect to the solvency of each of Covidien and Mallinckrodt; and
	<ul> <li>the approval for listing on the NYSE of our ordinary shares to be delivered in the distribution having been obtained.</li> </ul>
	We cannot assure you that any or all of these conditions will be met. For a complete discussion of all of the conditions to the distribution, see "The Separation—Conditions to the Distribution."
What is the expected date of completion of the separation?	The completion and timing of the separation is dependent upon the satisfaction of a number of conditions. We expect our ordinary shares to be distributed after the close of trading on June 28, 2013 to the holders of record of ordinary shares of Covidien at the close of business on the record date; however, no assurance can be provided as to the timing of the separation or that all conditions to the separation will be met.
Can Covidien decide to cancel the distribution even if all of the conditions have been met?	Yes. The distribution is subject to the satisfaction or waiver of certain conditions. See "The Separation—Conditions to the Distribution." Until the distribution has occurred, Covidien has the right to terminate the distribution, even if all of the conditions are satisfied, if at any time the board of directors of Covidien determines that the distribution is not in the best interests of Covidien and its shareholders or that market conditions or other circumstances are such that it is not advisable at that time to separate the Pharmaceuticals business from the remainder of Covidien.
What if I want to sell my Covidien ordinary shares or my Mallinckrodt ordinary shares?	You should consult with your financial advisors, such as your broker, bank, other nominee or tax advisor.

	If you decide to sell any ordinary shares of Covidien before the distribution date, you should make sure your broker, bank or other nominee understands whether you want to sell your ordinary shares of Covidien with or without your entitlement to Mallinckrodt ordinary shares pursuant to the distribution.
What is "regular-way" and "ex-distribution" trading?	Beginning on or shortly before the record date and continuing up to and through the distribution date, it is expected that there will be two markets in ordinary shares of Covidien: a "regular-way" market and an "ex-distribution" market. Ordinary shares of Covidien that trade in the "regular-way" market will trade with an entitlement to ordinary shares of Mallinckrodt distributed pursuant to the distribution. Shares that trade in the "ex-distribution" market will trade without an entitlement to ordinary shares of Mallinckrodt distributed pursuant to the distribution. Covidien cannot predict the trading prices of its ordinary shares before, on or after the distribution date.
Where will I be able to trade ordinary shares of Mallinckrodt?	We have received authorization to list our ordinary shares on the NYSE under the symbol "MNK." We anticipate that trading in our ordinary shares will begin on a "when-issued" basis on or shortly before the record date and will continue up to and through the distribution date and that "regular-way" trading in our ordinary shares will begin on the first trading day following the completion of the separation. If trading begins on a "when-issued" basis, you may purchase or sell our ordinary shares up to and through the distribution date, but your transaction will not settle until after the distribution date. We cannot predict the trading prices of our ordinary shares before, on or after the distribution date.
What will happen to the listing of Covidien's ordinary shares?	Ordinary shares of Covidien will continue to trade on the NYSE after the distribution.
Will the number of ordinary shares of Covidien that I own change as a result of the distribution?	No. The number of ordinary shares of Covidien that you own will not change as a result of the distribution.
Will the distribution affect the market price of my Covidier ordinary shares?	Yes. As a result of the distribution, Covidien expects the trading price of Covidien ordinary shares immediately following the distribution to be lower than the "regular-way" trading price of such shares immediately prior to the distribution because the trading price will no longer reflect the value of the Pharmaceuticals business held by Mallinckrodt. Covidien believes that over time following the separation, assuming the same market conditions and the realization of the expected benefits of the separation, Covidien ordinary shares and Mallinckrodt ordinary shares should have a higher aggregate market value as compared to what the market value of Covidien ordinary shares would be if the separation did not occur. There can be no assurance, however, that such a higher aggregate market value will be achieved. This means, for example, that the combined trading prices of eight Covidien ordinary shares and one ordinary share of Mallinckrodt after the distribution may be equal to, greater than or less than the trading price of eight Covidien ordinary shares before the distribution.

# -4-

What are the material U.S. federal income tax consequences of the separation?

How will I determine my tax basis for U.S. federal income tax purposes in the Covidien ordinary shares I continue to hold and the Mallinckrodt ordinary shares I receive in the distribution?

What are the material Irish tax consequences of the separation?

The distribution is conditioned on the continued validity of the private letter ruling received by Covidien from the IRS (the "IRS ruling") substantially to the effect that, for U.S. federal income tax purposes, (i) certain transactions to be effected in connection with the separation qualify as transactions under Sections 355 and/or 368(a) of the U.S. Internal Revenue Code of 1986, as amended (the "Code"), and (ii) the distribution qualifies as a transaction under Sections 355 and 368(a)(1)(D) of the Code. This condition requires that the IRS ruling remain in full force and effect and not be modified or amended in any respect adversely affecting the intended tax-free treatment of the distribution and certain related transactions. The distribution is further conditioned on Skadden, Arps, Slate, Meagher & Flom LLP issuing an opinion (the "tax opinion"), in form and substance acceptable to Covidien, which tax opinion will rely on the effectiveness of the IRS ruling, to Covidien, substantially to the effect that, for U.S. federal income tax purposes, the distribution and certain transactions entered into in connection with the distribution will qualify as transactions under Sections 355 and/or 368(a) of the Code. See "The Separation-Conditions to the Distribution." Assuming that the distribution and certain related transactions will qualify as tax-free transactions under Sections 355 and/or 368(a) of the Code, for U.S. federal income tax purposes, except for gain realized on the receipt of cash paid in lieu of fractional shares, no gain or loss generally will be recognized by a Covidien shareholder, and no amount generally will be included in such Covidien shareholder's taxable income, as a result of the separation. You should, however, consult your own tax advisor as to the particular tax consequences to you. The U.S. federal income tax consequences of the separation are described in more detail under "Material Tax Consequences-Material U.S. Federal Income Tax Consequences."

Assuming that the distribution is tax-free to Covidien shareholders, except for cash received in lieu of fractional shares, your tax basis for U.S. federal income tax purposes in the Covidien ordinary shares held by you immediately prior to the distribution will be allocated between such Covidien ordinary shares and the Mallinckrodt ordinary shares received by you in the distribution in proportion to the relative fair market values of each immediately following the distribution. Covidien will provide its shareholders with information to enable them to compute their tax basis in both the Covidien and Mallinckrodt ordinary shares. This information will be posted on Covidien's website, www.covidien.com.

Covidien shareholders that are not resident or ordinarily resident in Ireland for Irish tax purposes and do not hold their shares in connection with a trade or business carried on by such shareholders through an Irish branch or agency will not be subject to Irish tax on chargeable gains on the receipt of new Mallinckrodt ordinary shares or cash in lieu of fractional shares pursuant to the transaction. Other Covidien shareholders will not be subject to Irish tax on chargeable gains on the receipt of new Mallinckrodt ordinary shares pursuant to the distribution but will be subject to Irish tax on chargeable gains on

-5-

	the receipt of any cash in lieu of fractional shares. You should consult your own tax advisor as to the particular tax consequences to you. The Irish tax consequences of the separation are described in more detail under "Material Tax Consequences—Material Irish Tax Consequences."
What will Mallinckrodt's relationship be with Covidien following the separation?	In connection with the separation, we and Covidien will enter into a separation and distribution agreement and various other agreements, including a transition services agreement, a tax matters agreement and an employee matters agreement. These agreements will provide a framework for our relationship with Covidien after the separation and provide for the allocation between us and Covidien of Covidien's assets, employees, liabilities and obligations (including its property, employee benefits, environmental liabilities and tax liabilities) attributable to periods prior to, at and after our separation from Covidien. For additional information regarding the separation and distribution agreement and other transaction agreements, see "Risk Factors—Risks Related to the Separation" and "Our Relationship with Covidien Following the Distribution."
Who will manage Mallinckrodt after the separation?	Led by Mark Trudeau, who will be our President and Chief Executive Officer after the separation, our management team possesses deep knowledge of, and extensive experience in, our industry. Our management has been involved in strategic decisions with respect to Covidien's Pharmaceuticals business and in establishing a vision for the future of Mallinckrodt. For more information regarding our management, see "Management."
Are there risks associated with owning Mallinckrodt ordinary shares?	Yes. Our business is subject to both general and specific risks relating to our business, the industry in which we operate, our ongoing contractual relationships with Covidien and our status as a separate, publicly traded company. There also are risks relating to the separation, certain tax matters, our jurisdiction of incorporation and ownership of our ordinary shares. These risks are described in the "Risk Factors" section of this information statement beginning on page 20. You are encouraged to read that section carefully.
Does Mallinckrodt plan to pay dividends?	We currently intend to retain any earnings to finance research and development, acquisitions and the operation and expansion of our business, and do not anticipate paying any cash dividends for the foreseeable future. As a result, the return on your investment in our ordinary shares will be initially determined by increases and decreases in the market price of our ordinary shares. See "Dividends."
<i>Will Mallinckrodt incur any debt prior to or at the time of the distribution?</i>	Yes. We anticipate having approximately \$920 million of indebtedness upon completion of the separation. In addition, Mallinckrodt International Finance S.A. ("MIFSA"), a wholly owned subsidiary of Covidien that will become our wholly owned subsidiary upon completion of the separation, has entered into a senior unsecured revolving credit facility allowing borrowings of up to \$250 million in the aggregate; we do not anticipate having any indebtedness outstanding under this facility upon completion of the

-6-

separation. See "Description of Material Indebtedness" and "Risk Factors—Risks Related to Our Business."

Who will be the distribution agent, transfer agent, and registrar for the Mallinckrodt ordinary shares?

Computershare will be the distribution agent, transfer agent, and registrar for our ordinary shares. For questions relating to the transfer or mechanics of the distribution, you should contact:

Computershare 250 Royall Street Canton, MA 02021 (877) 498-8861

Where can I find more information about Covidien and Mallinckrodt?

Before the distribution, if you have any questions relating to Covidien's business performance, you should contact:

Covidien plc Investor Relations 15 Hampshire Street Mansfield, MA 02048 (508) 452-4650

After the distribution, our shareholders who have any questions relating to our business performance should contact us at:

Mallinckrodt plc Investor Relations 675 James S. McDonnell Blvd. Hazelwood, MO 63042 (314) 654-6650

-7-

#### INFORMATION STATEMENT SUMMARY

The following is a summary of material information discussed in this information statement. This summary may not contain all of the details concerning the separation or other information that may be important to you. To better understand the separation and our business and financial position, you should carefully review this entire information statement. Except as otherwise indicated or unless the context otherwise requires, the information included in this information statement about Mallinckrodt assumes the completion of all of the transactions referred to in this information statement in connection with the separation. Unless the context otherwise requires, references in this information statement to "Mallinckrodt plc," "Mallinckrodt public limited company," "Mallinckrodt Pharmaceuticals," "Mallinckrodt," "we," "us," "our company" and "the company" refer to Mallinckrodt plc, an Irish public limited company, and its combined subsidiaries. Unless the context otherwise requires, references to therwise requires, references in this information statement to "Covidien and its subsidiaries prior to completion of the separation. Unless the context otherwise requires, references in this information statement to "Covidien" refer to Covidien plc, an Irish public limited company, and its consolidated subsidiaries, including the Pharmaceuticals business prior to completion of the separation. Except as otherwise indicated, references in this information statement to "Iscal 2013, fiscal 2011, fiscal 2010, fiscal 2009 and fiscal 2008 are to Mallinckrodt's fiscal years ending or ended September 27, 2013, September 28, 2012, September 24, 2010, September 25, 2009 and September 26, 2008, respectively.

References in this information statement to our historical assets, liabilities, products, businesses or activities of our business are generally intended to refer to the historical assets, liabilities, products, businesses or activities of the Pharmaceuticals business of Covidien as the business was conducted as part of Covidien and its subsidiaries prior to completion of the separation.

#### **Our Company**

We are a global company that develops, manufactures, markets and distributes both branded and generic specialty pharmaceuticals, active pharmaceutical ingredients ("API") and diagnostic imaging agents. We use our API products in the manufacture of our generic pharmaceuticals and also sell them to other pharmaceutical companies. Our products are found in almost every hospital, standalone diagnostic imaging center or pharmacy in the U.S. and we have a sales presence in approximately 50 countries. Our diverse product portfolio and solid market positions reflect our 145-year history of pharmaceutical excellence with many innovations important for the treatment of pain, the development of the modern U.S. pharmaceuticals industry and the evolution of nuclear and diagnostic imaging.

During fiscal 2012, we generated net sales of approximately \$2.1 billion and net income of \$134.6 million. Approximately 66% of our fiscal 2012 net sales were generated in the U.S. and 34% were generated outside of the U.S.

Upon completion of the separation, we will conduct our business under the name Mallinckrodt Pharmaceuticals through two operating segments:

Our Specialty Pharmaceuticals segment develops, manufactures and sells, through its Brands business, branded drugs, including EXALGO® (hydromorphone HCl) Extended-Release Tablets, which are indicated for the treatment of moderate to severe pain in opioid-tolerant patients requiring continuous around-the-clock opioid analgesia for an extended amount of time ("Exalgo"), and GABLOFEN® (baclofen injection), which are injections indicated for use in the management of severe spasticity of cerebral or spinal origin in patients age four years and above ("Gablofen"). Our Specialty Pharmaceuticals segment has a pipeline of multiple new pain products. We market our branded products in the U.S. to physicians including, for example, pain specialists, anesthesiologists, orthopedic surgeons, rheumatologists and neurologists, who prescribe them for their patients. We develop, manufacture and sell generic drugs, including a variety of products containing U.S. Drug

-8-

Enforcement Administration ("DEA") Schedule II and III controlled substances such as oxycodone, which is usually indicated alone for treatment of moderate to severe pain or in combination with acetaminophen for treatment of moderate to moderately severe pain; hydrocodone, which in combination with acetaminophen is most often indicated for the treatment of moderate to moderately severe pain; and methylphenidate, which is indicated for the treatment of attention deficit hyperactivity disorder ("ADHD"). We sell our generic products to wholesalers, large- and medium-sized retail pharmacy chains, food store chains with pharmacies, mail order pharmacies and through multiple other channels of distribution. Nearly all of our generic products are sourced from our own manufactured API, including controlled substances and acetaminophen, a pain reliever and fever reducer used to treat many conditions such as headache, muscle aches, arthritis, backache, toothaches, colds and fevers. We also manufacture and sell API to other pharmaceutical companies around the world.

Our *Global Medical Imaging* segment develops, manufactures and markets contrast media and delivery systems ("CMDS"). Our contrast media offerings include iodine- and gadolinium-containing injectable products for diagnostic imaging applications such as computed tomography ("CT") and magnetic resonance imaging ("MRI") under brand names including Optiray<sup>™</sup> and Optimark<sup>™</sup>. These diagnostic imaging agents allow radiologists to improve the diagnostic capability of the CT and MRI scanners for certain types of imaging procedures. We package our contrast media in either pre-filled syringes or vials and bottles. Our pre-filled syringes fit into our power injectors, including the Optivantage<sup>™</sup> DH, and allow the radiology staff to have greater throughput while maintaining a high degree of safety for patients. We sell our contrast media to hospitals and hospital groups and have contracts with group purchasing organizations ("GPOs"), primarily in the U.S., that provide access to large groups of hospitals, and to standalone diagnostic imaging agents, such as Technescan MAG3<sup>™</sup>, a nuclear imaging agent that delivers both quantitative and qualitative information used to detect and evaluate a wide variety of renal disorders, primarily in the U.S. and Europe. In addition, we sell technetium-99m ("Tc-99m") for use in our Ultra-Technekow<sup>™</sup> DTE generators in the U.S. and Europe, as well as cold kits that are combined with these imaging agents to show cardiac function and the function of other organ systems. We are the only manufacturer of Tc-99m generators that processes molybdenum-99 ("Mo-99"). We also sell other radiopharmaceuticals, such as Octreoscan<sup>™</sup>, for the detection of certain types of cancer.

#### **Our Competitive Strengths**

#### We believe we have the following strengths:

- *Expertise in the acquisition and importation of highly regulated raw materials, and strong regulatory relationships.* We have expertise in the acquisition and importation of highly regulated raw materials, such as opioids, other controlled substances and radioisotopes. For example, in 2012, we believe we received almost 40% of the DEA's total annual quota for controlled substances that we manufacture. In 2012, our Generics business had an approximate 30% market share of DEA Schedule II and III opioid, oral solid doses, based on IMS Health data. The acquisition of certain raw materials and the processing of them into finished products requires a close collaboration with a wide variety of regulatory authorities including the DEA, U.S. Food and Drug Administration ("FDA"), U.S. Nuclear Regulatory Commission ("NRC"), European Medicines Agency and Irish Medicines Board, among many others. We have a long history of working with regulatory agencies to provide ongoing, reliable access to these highly regulated materials.
- Specialized chemistry, development and formulation expertise which supports a sustainable, robust product pipeline. We have specialized chemistry expertise for the formulation of new drug combinations and reformulation of existing drugs into a wide range of products, such as tablets, capsules, oral liquids and injectable products. In late 2009, we completed a significant upgrade to our formulation pilot plant in Webster Groves, Missouri. This expansion greatly enhanced our pharmaceutical formulation capability, which has resulted in a significant increase in both branded and generic formulations that have been approved by the FDA or that are in various stages of pre-clinical development, clinical development or regulatory review. On our Hazelwood, Missouri campus, we

have a parenteral pilot plant focused on the reformulation of imaging agents for our Global Medical Imaging segment.

- The broadest portfolio of generic products and controlled substance API for pain and a growing pipeline of branded pharmaceutical pain products. Our Generics and API businesses have a strong position in the controlled substance generics market. We believe our Generics and API businesses offer the broadest product line of opioid and other controlled substances (primarily DEA Schedule II and III). Our strong market position is a result of the following:
  - <sup>1</sup> Formulation and manufacturing expertise in controlled substances and complex generics;
  - <sup>i</sup> Our commitment to investment in our research and development ("R&D") infrastructure and capabilities has resulted in a pipeline of generic and branded controlled substances, many of which are long-acting or hard to formulate products, which are under development or pending approval by the FDA. For example, on December 28, 2012, we became the first company to receive approval from the FDA to manufacture and market in the U.S. a generic version of Concerta<sup>®</sup>, a branded pharmaceutical for the treatment of ADHD and a registered trademark of ALZA Corporation. Total gross sales of Concerta and its authorized generic version exceeded \$1.6 billion in the twelve months ended September 30, 2012, according to IMS Health data;
  - <sup>1</sup> Our strong position in controlled substance API and vertical integration from opioid raw materials to finished dosage forms; and
  - <sup>1</sup> U.S. importation restrictions of controlled substance API and finished products.
- *Solid market position in diagnostic imaging agents.* We believe that we are one of the top three participants globally in nuclear radiopharmaceutical products. We are one of only two manufacturers of Tc-99m generators (marketed under the brand name Ultra-Technekow DTE) in North America, one of only three in Europe and the only one on either continent that has its own Mo-99 processing facility, which provides cost and raw material supply advantages. In CMDS, we offer a fully integrated line of contrast media, pre-filled syringes and proprietary power injectors. Our leading contrast media product, Optiray, has been on the market for over 25 years and is differentiated in part by being offered in pre-filled syringes that fit our proprietary power injectors, which enhances clinician safety and reduces risks in medication management.
- Distinctive high-quality manufacturing and distribution skills with vertical integration where there are competitive advantages. Our manufacturing
  and supply chain capabilities enable highly efficient controlled substance tableting, packaging and distribution. Our investments include one of the
  world's largest DEA Schedule C-II vault storage capacities for raw materials, intermediates and finished dosages. In our Global Medical Imaging
  segment, we have the capability to process Mo-99 for use in our Ultra-Technekow DTE (Tc-99m) generators and to manufacture cyclotron-derived
  isotopes such as thallium-201, indium-111, gallium-67, germanium-68 and iodine-123. In addition, we produce the large-volume terminally sterilized
  pre-filled plastic syringes that fit into our power injectors. Where appropriate, we have also pursued selective vertical integration initiatives to ensure
  our manufacturing and supply chain benefit from cost and productivity efficiencies, such as using several of our API products to provide the raw
  materials for some of our generic products.
- *Global commercial reach.* Our Global Medical Imaging segment operates throughout the world and its direct and indirect marketing and selling capabilities are tailored to business and geographic needs. Our Global Medical Imaging sales presence in approximately 50 countries has positioned us well for expansion.
- Strong management team with extensive industry experience. We benefit from having a management team with extensive experience in small, medium and large life sciences firms. Mark Trudeau, who will serve as our President and Chief Executive Officer, has more than 29 years of experience in the pharmaceuticals industry and Matthew Harbaugh, with over 20 years of financial experience, mostly in the life sciences field, will serve as our Senior Vice President and Chief Financial Officer.

#### **Our Strategy**

Our strategy is to enhance growth and build shareholder value by increasing our core technical and commercial capabilities, expanding our product portfolio in pain management and imaging and selectively pursuing growth opportunities in adjacent markets through acquisitions, licensing arrangements and co-promotions.

We are committed to the following goals:

- Grow sales faster than our Specialty Pharmaceuticals market segment. We believe that our R&D investments in our Specialty Pharmaceuticals segment have positioned us to grow sales at a faster rate than the overall market growth rate.
- Expand core product portfolio with new branded and generic products. We intend to continue to focus on marketing our pain drugs (such as
  extended-release opioids and topical anti-inflammatories) and the drugs and pipeline we acquired as a result of our recent acquisition of CNS
  Therapeutics, Inc. ("CNS Therapeutics") (such as Gablofen). We also have a pipeline of several branded pain management products that we intend to
  develop and bring to market. In addition, we believe that we can continue to expand our generic product portfolio of controlled substances,
  particularly in the pain market and the ADHD segment of the controlled substance market, especially those products that are difficult to formulate.
- *Enhance commercial and technical capabilities in branded pharmaceuticals.* We plan on enhancing our branded commercial infrastructure by focusing on a multi-pronged approach of near-term product launches, co-promotions, line extensions and selective acquisitions. Our intention is to increase our branded sales faster than our generic sales to drive margin expansion over the long term.
- Grow into new, adjacent areas through acquisitions and targeted partnerships. Our business development objectives are focused on targeted
  partnerships, as shown by our recent co-promotions and acquisitions (including, most recently, our acquisition of CNS Therapeutics), which we
  believe complement our core competencies and accelerate our organic growth initiatives. Our priority areas include co-promotions and licensing of
  existing product franchises, licensing of novel delivery mechanisms and technologies for existing drugs, expansion into targeted adjacent therapeutic
  markets such as central nervous system drugs, and broader distribution channels in developed and developing geographical markets.
- *Target growth in select markets.* We expect our manufacturing and global distribution and sales to enable our expansion beyond developed markets. We believe that our Specialty Pharmaceuticals segment is positioned for growth into select markets and that it will be able to leverage our Global Medical Imaging segment's presence for expansion.

#### Risks Associated with Mallinckrodt's Business and the Separation

An investment in our ordinary shares is subject to a number of risks, including risks relating to the separation. The following list of risk factors is not exhaustive. Please read the information in the section captioned "Risk Factors" for a more thorough description of these and other risks.

#### **Risks Related to Our Business**

- The DEA regulates the availability of controlled substances that are API, drug products under development and marketed drug products. At times, the procurement and production quotas granted by the DEA may be insufficient to meet our commercial and R&D needs.
- The manufacture of our products is highly exacting and complex, and our business could suffer if we, or our suppliers, encounter manufacturing or supply problems.

-11-

- The global supply of fission-produced Mo-99 is limited. Our inability to obtain and/or to timely transport Mo-99 to our Tc-99m generator production facilities could prevent us from delivering our Ultra-Technekow DTE Tc-99m generators to our customers in the required quantities, within the required timeframe, or at all, which could result in order cancellations and decreased revenues or increased costs if we procure supply from other sources.
- In response to the U.S. National Security Administration's Global Threat Initiative, we are in the process of converting our Mo-99 production operation in the Netherlands from high enriched uranium ("HEU") targets to low enriched uranium ("LEU") targets. There can be no assurance that we will be successful in completing this conversion.
- Our customer concentration may materially adversely affect our financial condition and results of operations.
- Cost-containment efforts of our customers, purchasing groups, third-party payors and governmental organizations could materially adversely affect our net sales and results of operations.
- We may be unable to successfully develop or commercialize new products or adapt to a changing technology and diagnostic treatment landscape and, as a result, our results of operations may suffer.
- We may be unable to protect our intellectual property rights or we may be subject to claims that we infringe on the intellectual property rights of others.
- We face significant competition and may not be able to compete effectively.
- Any acquisitions of technologies, products and businesses may be difficult to integrate, could materially adversely affect our relationships with key customers and/or could result in significant impairment charges.
- We may incur product liability losses and other litigation liability.
- The implementation of healthcare reform in the U.S. may materially adversely affect us.
- Sales of our products are affected by the reimbursement practices of a small number of large public and private insurers. In addition, reimbursement criteria and the use of tender systems outside the U.S. could reduce prices for our products or reduce our market opportunities.
- Our reporting and payment obligations under the Medicare and/or Medicaid rebate program and other governmental purchasing and rebate programs are complex. Any determination of failure to comply with these obligations or those relating to healthcare fraud and abuse laws could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.
- Changes in laws and regulations may materially adversely affect us.
- Global economic conditions could harm us.
- Our global operations expose us to risks and challenges associated with conducting business internationally.
- Currency exchange rate fluctuations could materially adversely affect our business and results of operations.
- Our operations expose us to the risk of material health, safety and environmental liabilities, litigation and violations.
- If we are unable to retain our key personnel, we may be unable to maintain or expand our business.
- Our business depends on the continued effectiveness and availability of our information technology infrastructure, and failures of this infrastructure could harm our operations.

-12-

#### **Risks Related to the Separation**

- We have no recent history operating as an independent company, and our historical and pro forma financial information is not necessarily representative of the results that we would have achieved as a separate, publicly traded company and may not be an accurate indicator of our future results of operations.
- As we build our information technology infrastructure and transition our data to our own systems, we could incur substantial additional costs and experience temporary business interruptions.
- Our accounting and other management systems and resources may not be adequately prepared to meet the financial reporting and other requirements to which we will be subject following the separation.
- We may have received more favorable or less favorable terms from unaffiliated third parties than the terms we will receive in our agreements with Covidien.
- Covidien may fail to perform under various transaction agreements that will be executed as part of the separation or we may fail to have necessary systems and services in place when certain of the transaction agreements expire.
- Potential indemnification liabilities to Covidien pursuant to the separation and distribution agreement could materially adversely affect us.
- We may not achieve some or all of the expected benefits of the separation, and the separation may materially adversely affect our business.
- Challenges in the commercial and credit environment may materially adversely affect our ability to issue debt on acceptable terms and our future access to capital.
- After the separation, we will have indebtedness, which could restrict our ability to pay dividends and have a negative impact on our financing options and liquidity position.
- We may need additional financing in the future to meet our capital needs or to make acquisitions, and such financing may not be available on favorable or acceptable terms, and may be dilutive to existing shareholders.
- No further vote of the Covidien shareholders is required in connection with the distribution. As a result, if the distribution occurs and you do not want to receive our ordinary shares in the distribution, your sole recourse will be to divest yourself of your Covidien ordinary shares prior to the record date.

#### **Risks Related to Tax Matters**

- If the distribution fails to qualify as a tax-free transaction for U.S. federal income tax purposes, then Mallinckrodt, Covidien and Covidien's shareholders could be subject to significant tax liability or tax indemnity obligations.
- We could have significant tax liabilities under our tax matters agreement with Covidien, including for periods during which our subsidiaries and operations were those of Tyco International Ltd.
- Examination and audits by tax authorities, including the IRS, could result in additional tax payments.
- We may not be able to maintain a competitive worldwide effective corporate tax rate.

#### **Risks Related to Our Jurisdiction of Incorporation**

- Legislative action in the U.S. could materially adversely affect us.
- There is no guarantee that the High Court of Ireland approval of the creation of distributable reserves will be forthcoming.
- Irish law differs from the laws in effect in the U.S. and may afford less protection to holders of our securities.
- Irish law imposes restrictions on certain aspects of capital management.

### -13-

#### **Risks Related to Our Ordinary Shares**

- We cannot be certain that an active trading market for our ordinary shares will develop or be sustained after the separation, and following the separation, our share price may fluctuate significantly.
- A number of our ordinary shares are or will be eligible for future sale, which may cause our share price to decline.
- Your percentage of ownership in Mallinckrodt may be diluted.
- Certain provisions in our articles of association, among other things, could prevent or delay an acquisition of Mallinckrodt, which could decrease the trading price of our ordinary shares.

#### **The Separation**

On December 15, 2011, Covidien announced that it intended to separate its Pharmaceuticals business from the remainder of its business. On May 23, 2013, the Covidien board of directors approved the transfer of Covidien's Pharmaceuticals business to Mallinckrodt in return for Mallinckrodt issuing ordinary shares to Covidien shareholders on the basis of one ordinary share of Mallinckrodt for every eight Covidien ordinary shares held on the record date.

Mallinckrodt plc was incorporated in Ireland on January 9, 2013 for the purpose of holding Covidien's Pharmaceuticals business following the separation. Currently, all of our issued shares are held beneficially by an Irish corporate services provider. Immediately prior to the distribution, Covidien will transfer its Pharmaceuticals business to us in return for which we will issue shares to Covidien ordinary shareholders, pro rata to their respective holdings. Prior to the transfer by Covidien to Mallinckrodt plc of our business, we will have no business operations. In connection with these transactions, we will acquire the shares held beneficially by the Irish corporate services provider referred to above for no consideration and cancel these shares. Immediately following the distribution, the persons entitled to receive Mallinckrodt ordinary shares in the distribution will own all of our outstanding ordinary shares.

#### **Our Post-Separation Relationship with Covidien**

In connection with the separation, we and Covidien will enter into a separation and distribution agreement (the "separation and distribution agreement") and various other agreements, including a transition services agreement, a tax matters agreement and an employee matters agreement. These agreements will provide a framework for our relationship with Covidien after the separation and provide for the allocation between us and Covidien of Covidien's assets, employees, liabilities and obligations (including its property, employee benefits, environmental liabilities and tax liabilities) attributable to periods prior to, at and after our separation from Covidien. For additional information regarding the separation and distribution agreement and other transaction agreements, see "Risk Factors—Risks Related to the Separation" and "Our Relationship with Covidien Following the Distribution."

# **Reasons for the Separation**

The Covidien board of directors believes that separating the Pharmaceuticals business from the remainder of Covidien is in the best interests of Covidien and its shareholders for a number of reasons, including that:

The separation will allow each of the Pharmaceuticals business and Covidien's other businesses to focus on its own strategic and operational plans and capital structure without diverting human and financial resources to the other business or being constrained by a board and management that are also responsible for overseeing and furthering the objectives of the other business. The separation will also enhance the success of each business by reducing internal complexity and enabling each of Covidien and Mallinckrodt to avoid management, systemic and other problems that arise by operation of different businesses within the same corporate structure.



- The separation will enable each of Covidien and Mallinckrodt to pursue the capital structure that is most appropriate for its business and business strategy. Each business has different capital requirements that cannot be optimally addressed with a single capital structure. The separation will permit each of Covidien and Mallinckrodt to pursue a different capital structure that results in a more efficient pricing of its equity in the financial markets.
- The separation will allow Covidien and Mallinckrodt to set new investor expectations for their respective businesses and separate financial prospects based on their unique investment identities, including the merits, performance and future prospects of their respective businesses. The separation will also provide investors with two distinct and targeted investment opportunities and provide a more efficient currency for acquisitions.
- The separation will increase the effectiveness of the equity-based compensation programs of both Covidien and Mallinckrodt by tying the value of the equity compensation awarded to employees, officers or directors more directly to the performance of the business for which these individuals provide services.

The Covidien board of directors also considered a number of potentially negative factors in evaluating the separation, including that:

- As a current part of Covidien, we take advantage of certain functions performed by Covidien, such as accounting, tax, legal, human resources and other general and administrative functions. After the separation, Covidien will not perform certain of these functions for us, and, because of our smaller scale as a standalone company, our cost of performing such functions will be higher than the amounts reflected in our historical financial statements, which will cause our profitability to decrease.
- The actions required to separate Covidien's and Mallinckrodt's respective businesses could disrupt our operations.
- Certain costs and liabilities that were otherwise less significant to Covidien as a whole will be more significant for us as a standalone company due to our being smaller than Covidien.
- We will incur costs in connection with the transition to being a standalone public company that may include accounting, tax, legal, and other professional services costs, recruiting and relocation costs associated with hiring key senior management personnel new to Mallinckrodt, costs related to establishing a new brand identity in the marketplace, tax costs and costs to separate information systems.
- We may not achieve the anticipated benefits of the separation for a variety of reasons, including, among others: (a) the separation will require significant amounts of management's time and effort, which may divert management's attention from operating and growing our business; and (b) following the separation, we may be more susceptible to market fluctuations and other adverse events than if we were still a part of Covidien, because our business will be less diversified than Covidien's business.
- In addition, under the terms of the tax matters agreement that we will enter into with Covidien, we will be restricted from taking certain actions that could cause the distribution or certain related transactions to fail to qualify as a tax-free or tax-favored transaction under applicable law for a period of time. During this period, these restrictions may limit our ability to pursue certain strategic transactions and equity issuances or engage in new business or other transactions that might increase the value of our business, over some period of time.
- As a current part of Covidien, we take advantage of Covidien's size and purchasing power in procuring certain goods and services. After the separation, as a standalone company, we may be unable to obtain these goods, services, and technologies at prices or on terms as favorable as those Covidien obtained prior to completion of the separation.

-15-

In determining to pursue the separation, the Covidien board of directors concluded that the potential benefits of the separation outweighed these factors. See "The Separation—Reasons for the Separation" and "Risk Factors."

#### **Corporate Information**

Our principal executive offices are located at 1st Floor, 20 On Hatch, Lower Hatch Street, Dublin 2, Ireland. Our telephone number at this location is +353 (1) 438-1700. Our U.S. headquarters is located at 675 James S. McDonnell Blvd., Hazelwood, MO 63042. Our telephone number at this location is (314) 654-2000. Our website is www.mallinckrodt.com.

#### **Reason for Furnishing this Information Statement**

This information statement is being furnished solely to provide information to the shareholders of Covidien who will receive ordinary shares of Mallinckrodt in the distribution. It is not, and is not to be construed as, an inducement or encouragement to buy or sell any of Mallinckrodt's securities. The information contained in this information statement is believed by Mallinckrodt to be accurate as of the date set forth on its cover. Changes may occur after that date and neither Covidien nor Mallinckrodt will update this information except in the normal course of their respective disclosure obligations and practices.

This document does not constitute a prospectus within the meaning of Part 5 of the Investment Funds, Companies and Miscellaneous Provisions Act 2005 of Ireland (as amended) or the Prospectus Directive (2003/71/EC). No offer of shares to the public is made, or will be made, that requires the publication of a prospectus pursuant to the Irish prospectus law (within the meaning of Part 5 of the Investment Funds, Companies and Miscellaneous Provisions Act 2005 of Ireland, as amended) or the Prospectus Directive (2003/71/EC). This document has not been approved or reviewed by or registered with the Central Bank of Ireland. This document does not constitute investment advice or the provision of investment services within the meaning of the European Communities (Markets in Capital Instruments) Regulations 2007 of Ireland (as amended) or the Markets in Financial Instruments Directive (2004/39/EC). Neither Covidien nor Mallinckrodt is an authorized investment firm within the meaning of the European Communities (Markets Financial Instruments) Regulations 2007 of Ireland (as amended) or the Markets in Financial Instruments Directive (2004/39/EC) and the recipients of this document should seek independent legal and financial advice in determining their actions in respect of or pursuant to this document.

-16-

#### SUMMARY HISTORICAL AND UNAUDITED PRO FORMA COMBINED FINANCIAL DATA

The following table sets forth summary historical financial data for the periods indicated below. The combined income statement data for the six months ended March 29, 2013 and March 30, 2012 and the combined balance sheet data at March 29, 2013 have been derived from our unaudited condensed combined financial statements included elsewhere in this information statement. The summary income statement data for each of the fiscal years in the three-year period ended September 28, 2012 and the summary balance sheet data as of September 28, 2012 and September 30, 2011 have been derived from our audited combined financial statements, which are included elsewhere in this information statement. The summary balance sheet data as of March 30, 2012 and September 24, 2010 have been derived from our unaudited combined financial statements that are not included in this information statement. The summary financial data should be read in conjunction with our combined financial statements and accompanying notes and "Management's Discussion and Analysis of Financial Condition and Results of Operations" included elsewhere in this information statement.

The combined financial statements have been prepared by Covidien to present the historical operating assets, liabilities and related results of operations of its Pharmaceuticals business. The combined financial statements include all assets and liabilities related to the operation of the business and which were subject to oversight and review by management of the Pharmaceuticals business. The combined financial statements do not include certain corporate non-operating assets and liabilities, principally related to changes in the internal capital structure resulting from the internal reorganization of our legal entities to facilitate the separation. These non-operating assets and liabilities do not represent standalone businesses and primarily relate to intercompany transactions.

The following table also presents summary unaudited pro forma data. The pro forma data for the periods ended March 29, 2013 and September 28, 2012 assumes that the separation occurred on October 1, 2011, the first day of fiscal 2012. The pro forma balance sheet assumes that the separation occurred on March 29, 2013. The pro forma adjustments are based upon available information and assumptions that management believes are reasonable. Refer to the notes to the unaudited pro forma condensed combined financial statements and accompanying notes included elsewhere in this information statement for a discussion of adjustments reflected in the pro forma data.

The summary historical and unaudited pro forma data does not necessarily reflect what our results of operations and financial condition would have been had we operated as a separate, publicly traded company during the periods presented. In addition, they are not necessarily indicative of our future results of operations or financial condition.

#### **Non-GAAP Financial Measures**

Adjusted EBITDA represents earnings from net income before interest, income taxes, depreciation and amortization, adjusted to exclude certain items. These items, if applicable, include discontinued operations; other income, net; separation costs; restructuring charges, net; immediately expensed up-front and milestone payments; acquisition related costs; and non-cash impairment charges. We have provided this non-GAAP financial measure because it is used by management, along with financial measures in accordance with accounting principles generally accepted in the U.S. ("GAAP"), to evaluate our operating performance. In addition, we believe it will be used by certain investors to measure our operating results. Management believes that presenting Adjusted EBITDA to investors provides useful information about our performance across reporting periods on a consistent basis by excluding items that we do not believe are indicative of our core operating performance.

Adjusted EBITDA has the following limitations:

- it does not reflect our cash expenditures, or future requirements, for capital expenditures or contractual commitments;
- · it does not reflect changes in, or cash requirements for, our working capital needs;

#### -17-

- it does not reflect interest expense or the cash requirements necessary to service interest or principal payments;
- · it is not adjusted for all non-cash income or expense items that are reflected in our statements of cash flows; and
- other companies in our industry may calculate this measure differently than we do, limiting its usefulness as a comparative measure.

Because of these limitations, Adjusted EBITDA should be considered supplemental to and not a substitute for net income or any other performance measures derived in accordance with GAAP. See our combined financial statements included elsewhere in this information statement for our GAAP results.

	Si	Six Months Ended			Fiscal <sup>(1)</sup>			
(Dollars in Millions)	Pro forma for the Separation	March 29, 2013 <sup>(2)</sup>	March 30, 2012 <sup>(3)</sup>	Pro forma for the Separation 2012	2012 <sup>(4)</sup>	2011 <sup>(5)</sup>	2010 <sup>(6)</sup>	
Combined Statement of Income Data:	· · · · · · · · · · · · · · · · · · ·							
Net sales	\$ 1,089.3	\$1,089.3	\$ 1,026.8	\$ 2,056.2	\$2,056.2	\$2,021.8	\$2,047.6	
Gross profit	507.0	507.0	488.3	964.8	964.8	914.9	932.4	
Operating income <sup>(7)</sup>	116.7	90.3	127.6	260.7	235.2	240.7	240.4	
Income from continuing operations before income taxes	95.3	90.4	128.3	218.5	236.1	243.2	243.2	
Income from continuing operations	70.8	54.3	78.9	166.3	141.3	157.0	145.9	
Combined Balance Sheet Data:								
Total assets	\$ 3,316.5	\$3,118.0	\$ 2,842.7		\$2,874.6	\$2,823.4	\$2,888.3	
Long-term debt	918.7	2.3	9.6		8.9	10.4	11.6	
Parent company equity	1,241.7	2,139.4	1,857.3		1,891.9	1,788.7	1,835.9	
Other Financial Data:								
Adjusted EBITDA <sup>(8)</sup>		\$ 190.2	\$ 208.5		\$ 402.8	\$ 371.8	\$ 366.1	

<sup>(1)</sup> Fiscal 2011 includes 53 weeks, while fiscal 2012 and 2010 each include 52 weeks.

<sup>(2)</sup> The six months ended March 29, 2013 includes \$26.4 million of separation costs and \$7.9 million of restructuring and related charges, net.

<sup>(3)</sup> The six months ended March 30, 2012 includes \$10.9 million of restructuring and related charges, net and \$10.2 million of separation costs.

- <sup>(4)</sup> Fiscal 2012 includes \$25.5 million of separation costs and \$19.2 million of restructuring and related charges, net.
- <sup>(5)</sup> Fiscal 2011 includes \$10.0 million of restructuring and related charges, net and \$2.9 million of separation costs.
- <sup>(6)</sup> Fiscal 2010 includes \$31.3 million of product liability charges and \$11.5 million of restructuring charges, net.
- (7) During the first six months of fiscal 2013 and 2012, Covidien allocated to us general corporate expenses in the amount of \$25.5 million and \$22.7 million, respectively. During fiscal 2012, 2011 and 2010, Covidien allocated general corporate expenses to us in the amount of \$49.2 million, \$56.3 million and \$60.8 million, respectively, which are included in our historical results. General corporate expenses include, but are not limited to, costs related to finance, legal, information technology, human resources, communications, employee benefits and incentives, insurance and stock-based compensation. Effective upon the separation, we will assume responsibility for all of these functions and related costs and anticipate our costs as a standalone entity will be higher than those allocated to us from Covidien. On an annual basis, these operating costs are estimated to be approximately \$40 million higher than the general corporate expenses historically allocated from Covidien to us. In addition, as part of Covidien, we shared in other costs of Covidien, including costs associated with Covidien's international infrastructure. Our portions of these costs were \$20.6 million and \$21.2 million and \$38.1 million, respectively. As a standalone company, we expect the recurring annual costs of our own international infrastructure to approximate the amount of costs incurred during fiscal 2012 as part of Covidien. No pro forma adjustments have been made to reflect the costs and expenses described in this paragraph because they are projected amounts based on judgmental estimates.

-18-

<sup>(8)</sup> The following table provides a reconciliation of our net income to Adjusted EBITDA for the periods presented:

	Six Mont	hs Ended	Fiscal			
	March 29,	March 30,				
(Dollars in Millions)	2013	2012	2012	2011	2010	
Net income	\$ 53.2	\$ 75.2	\$134.6	\$150.7	2010 \$200.6	
Interest expense, net	0.1	—	0.1	0.4	0.6	
Provision for income taxes	36.1	49.4	94.8	86.2	97.3	
Depreciation expense	49.2	51.7	103.6	92.8	90.8	
Amortization expense	17.7	13.5	27.3	27.0	23.4	
Loss (income) from discontinued operations, net of income taxes	1.1	3.7	6.7	6.3	(54.7)	
Other income, net	(0.2)	(0.7)	(1.0)	(2.9)	(3.4)	
Restructuring charges, net	6.6	5.5	11.2	8.4	11.5	
Separation costs	26.4	10.2	25.5	2.9		
Adjusted EBITDA	\$190.2	\$208.5	\$402.8	\$371.8	\$366.1	

-19-

#### **RISK FACTORS**

You should carefully consider the following risks and other information in this information statement in evaluating us and our ordinary shares. Our competitive position, business, financial condition, results of operations and cash flows can be affected by the factors set forth below, any one of which could cause our actual results to vary materially from recent results or from our anticipated future results. The risk factors generally have been separated into five groups: risks related to our business, risks related to the separation, risks relating to tax matters, risks relating to our jurisdiction of incorporation and risks related to our ordinary shares.

#### **Risks Related to Our Business**

We operate in a rapidly changing environment that involves a number of risks, some of which are beyond our control. The following discussion highlights some of these risks and others are discussed elsewhere in this information statement. These and other risks could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# The DEA regulates the availability of controlled substances that are API, drug products under development and marketed drug products. At times, the procurement and manufacturing quotas granted by the DEA may be insufficient to meet our commercial and R&D needs.

The DEA is the federal agency responsible for domestic enforcement of the Controlled Substances Act of 1970 ("CSA"). The CSA classifies drugs and other substances based on identified potential for abuse. Schedule I controlled substances, such as heroin and LSD, have a high abuse potential and have no currently accepted medical use; thus, they cannot be lawfully marketed or sold. Schedule II or III controlled substances include molecules such as oxycodone, hydrocodone and methylphenidate. The manufacture, storage, distribution and sale of these controlled substances are permitted, but highly regulated.

The DEA regulates the availability of API for products under development and marketed drug products that are Schedule II or III by setting annual quotas. Every calendar year, we must apply to the DEA for manufacturing quota to manufacture API and procurement quota to manufacture finished dosage products.

Given that the DEA has discretion to grant or deny our manufacturing and procurement quota requests, the quota the DEA grants may be insufficient to meet our commercial and R&D needs. The initial hydrocodone manufacturing and procurement quota grants we received from the DEA for 2012 were below the amounts we requested and were therefore insufficient to meet customer demand. We subsequently requested supplemental manufacturing and procurement quota in March 2012. In April 2012, the DEA denied our supplemental hydrocodone manufacturing quota request (to manufacture API) but granted the full amount of our hydrocodone procurement quota request (to manufacture finished dosage products). While our Hobart, New York facility had sufficient hydrocodone procurement quota to manufacture finished dosage products, our St. Louis, Missouri facility did not have sufficient hydrocodone bulk API manufacturing quota, which resulted in our inability to fulfill third-party customer requests. Subsequently, the DEA published a revised proposed U.S. aggregate quota for bulk manufacture of hydrocodone, and in August 2012, we filed another supplemental hydrocodone manufacturing quota request. In October 2012, the DEA granted 78% of our requested amount. This hydrocodone bulk API manufacturing quota shortage resulted in lost sales, the amount of which was not significant. See "Business—Regulatory Matters—Drug Enforcement Administration."

Any future delay or refusal by the DEA to grant, in whole or in part, our quota requests could delay or result in stopping the manufacture of our API and marketed drug products, new product launches or the conduct of bioequivalence studies and clinical trials. Such delay or refusal also could require us to allocate marketed drug products among our customers. These factors, along with any delay or refusal by the DEA to provide customers who purchase API from us with sufficient quota, could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### -20-

# The manufacture of our products is highly exacting and complex, and our business could suffer if we, or our suppliers, encounter manufacturing or supply problems.

The manufacture of our products is highly exacting and complex, due in part to strict regulatory and manufacturing requirements. Problems may arise during manufacturing for a variety of reasons including equipment malfunction, failure to follow specific protocols and procedures, defective raw materials and environmental factors. If a batch of finished product fails to meet quality standards during a production run, then that entire batch of product may have to be discarded. These problems could lead to backorders, increased costs (including contractual damages for failure to meet supply requirements), lost revenue, damage to customer relationships, time and expense spent investigating, correcting and preventing the root causes and, depending on the root causes, similar losses with respect to other products. In fiscal 2012, we experienced disruptions in supplying products to our customers due to a number of factors, including facility. These issues resulted in higher than usual backorders and obligations to pay contractual damages for failure to meet supply requirements. During fiscal 2012, our Generics business incurred approximately \$13 million of expenses for such contractual damages, a substantial portion of which was attributable to the issues experienced at this facility. In the event that such problems are not discovered before the product is released to the market, we also could incur product recall and product liability costs. If we incur a product recall or product liability costs involving one of our products, such product could receive reduced market acceptance and thus reduced product demand and could harm our reputation and our ability to market our products in the future. Significant manufacturing problems could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# The global supply of fission-produced Mo-99 is limited. Our inability to obtain and/or to timely transport Mo-99 to our Tc-99m generator production facilities could prevent us from delivering our Ultra-Technekow DTE Tc-99m generators to our customers in the required quantities, within the required timeframe, or at all, which could result in order cancellations and decreased revenues or increased costs if we procure supply from other sources.

Mo-99 is a critical ingredient of our Tc-99m generators. Mo-99 is produced in nuclear research reactors utilizing HEU or LEU targets. These targets, either tubular or flat and of varying sizes, are fabricated from HEU or LEU and, in either case, aluminum. The targets are placed in or near the core of the nuclear reactor where fission reactions occur resulting in the production of Mo-99 and other isotopes. This process, which takes approximately six days, is known as target irradiation. There are currently eight reactors around the world producing the global supply of Mo-99. We have agreements to obtain Mo-99 from three of these reactors and we rely predominantly on two of these reactors for our Mo-99 supply. These reactors are subject to scheduled and unscheduled shutdowns which can have a significant impact on the amount of Mo-99 available for processing. Mo-99 produced at these reactors is then finished at one of five processing sites located throughout the world, including our processing facility located in the Netherlands. At the processing facility, the targets are dissolved and chemically separated. In this process, the Mo-99 is isolated as a radiochemical. Once finished, Mo-99 must be transported to generator facilities where it is loaded into our Tc-99m generators that are sold, in the U.S., principally to nuclear radiopharmacies as well as hospitals and, in Europe and other markets, principally to hospitals, where single unit doses are then prepared. Mo-99 has a 66-hour half-life and decays primarily into Tc-99m, which has a half-life of only six hours. The radiopharmacies or hospitals prepare dosages from the Tc-99m generators for use in single photon emission computed tomography ("SPECT") imaging medical procedures. Given the product's radioactive decay, if we encounter delays in transporting Mo-99 to our generator facilities or if the generator facilities experience delays in loading Mo-99, we may be limited in the amount of Ultra-Technekow DTE generators that we could manufacture, distribute and se

In November 2012, one of the research reactors we use to irradiate targets as part of our Mo-99 processing operation experienced an unscheduled shutdown. The additional Mo-99 we are procuring from alternative sources comes at a higher than normal cost. While we expect the reactor to resume production in June 2013,

-21-

should this shutdown overlap the time period during which another reactor is planned to shut down for routine maintenance, there may be an impact on the amount of available Mo-99, which could result in global shortages, continued increased raw material costs and decreased sales. While we are pursuing additional sources of Mo-99 from potential producers around the world to augment our current supply, it is not certain whether these possible additional sources of Mo-99 will produce commercial quantities of Mo-99 for our business, or that these suppliers, together with our current suppliers, will be able to deliver a sufficient quantity of Mo-99 to meet our needs.

# In response to the U.S. National Security Administration's Global Threat Initiative, we are in the process of converting our Mo-99 production operation in the Netherlands from HEU targets to LEU targets. There can be no assurance that we will be successful in completing this conversion.

We currently use HEU targets for the production of Mo-99. In 2004, the U.S. National Security Administration established its Global Threat Initiative to, as quickly as possible, identify, secure, remove and/or facilitate the disposition of vulnerable, high-risk nuclear and radiological materials around the world. Included as one of the stated initiatives is the conversion by research reactors and isotope production facilities to LEU from HEU. We are in the process of converting our Mo-99 production operation in the Netherlands to LEU targets. However, there is no assurance that we will be successful in completing the conversion.

#### Our customer concentration may materially adversely affect our financial condition and results of operations.

We primarily sell our products to a limited number of wholesale drug distributors and large pharmacy chains. In turn, these wholesale drug distributors and large pharmacy chains supply products to pharmacies, hospitals, governmental agencies and physicians. Sales to two of our distributors that supply our products to many end user customers—Cardinal Health, Inc. ("Cardinal Health") and McKesson Corporation ("McKesson")—each accounted for 10% or more of our total net sales in each of the past three fiscal years. Additionally, AmerisourceBergen Corporation accounted for 10% of our total net sales in fiscal 2011. If we were to lose the business of these distributors, or if these distributors were to experience difficulty in paying us on a timely basis, this could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# Cost-containment efforts of our customers, purchasing groups, third-party payors and governmental organizations could materially adversely affect our net sales and results of operations.

In an effort to reduce cost, many existing and potential customers for our products within the U.S. have become members of GPOs and integrated delivery networks ("IDNs"). GPOs and IDNs negotiate pricing arrangements with healthcare product manufacturers and distributors and offer the negotiated prices to affiliated hospitals and other members. GPOs and IDNs typically award contracts on a category-by-category basis through a competitive bidding process. Bids are generally solicited from multiple manufacturers with the intention of driving down pricing. Due to the highly competitive nature of the GPO and IDN contracting processes, there is no assurance that we will be able to obtain or maintain contracts with major GPOs and IDNs across our product portfolio. Furthermore, the increasing leverage of organized buying groups may reduce market prices for our products, thereby reducing our profitability. While having a contract with a GPO or IDN for a given product can facilitate sales to members of that GPO or IDN, having a contract is no assurance that sales volume of those products will be maintained. GPOs and IDNs increasingly are awarding contracts to multiple suppliers for the same product category. Even when we are the sole contracted supplier of a GPO or IDN for a certain product, members of the GPO or IDN generally are free to purchase from other suppliers. Furthermore, GPO and IDN contracts typically are terminable without cause upon 60 to 90 days' prior notice. Accordingly, although we have contracts with many major GPOs and IDNs, the members of such groups may choose to purchase from our competitors, which could result in a decline in our net sales and results of operations.

-22-

Distributors of our products also have begun to negotiate terms of sale more aggressively in an effort to increase their profitability. Failure to negotiate distribution arrangements having advantageous pricing and other terms of sale could cause us to lose market share to our competitors and could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. Outside the U.S., we have experienced pricing pressure due to the concentration of purchasing power in centralized governmental healthcare authorities and increased efforts by such authorities to lower healthcare costs. We frequently are required to engage in competitive bidding for the sale of our products to governmental purchasing agents. Our failure to offer acceptable prices to these customers could materially adversely affect our net sales and results of operations in these markets.

# We may be unable to successfully develop or commercialize new products or adapt to a changing technology and diagnostic treatment landscape and, as a result, our results of operations may suffer.

Our future results of operations will depend to a significant extent upon our ability to successfully develop and commercialize new products in a timely manner. There are numerous difficulties in developing and commercializing new products, including:

- developing, testing and manufacturing products in compliance with regulatory and quality standards in a timely manner;
- receiving requisite regulatory approvals for such products in a timely manner, or at all;
- the availability, on commercially reasonable terms, of raw materials, including API and other key ingredients;
- developing and commercializing a new product is time-consuming, costly and subject to numerous factors, including legal actions brought by our competitors, that may delay or prevent the development and commercialization of new products;
- unanticipated costs;
- payment of prescription drug user fees to the FDA to defray the costs of review and approval of marketing applications for branded and generic drugs;
- experiencing delays as a result of limited resources at the FDA or other regulatory authorities;
- changing review and approval policies and standards at the FDA or other regulatory authorities; and
- potential delay in the commercializing of generic products by up to 30 months resulting from the listing of patents with the FDA.

As a result of these and other difficulties, products currently in development by us may or may not receive timely regulatory approvals, or approvals at all, as to one or more dosage strengths. This risk particularly exists with respect to the development of proprietary products because of the uncertainties, higher costs and length of time associated with R&D of such products and the inherent unproven market acceptance of such products. In addition, we face heightened risks in connection with our development of extended-release products because of the technical complexities and evolving regulatory and quality requirements related to such products. Moreover, the FDA regulates the facilities, processes and procedures used to manufacture and market pharmaceutical products in the U.S. Manufacturing facilities must be registered with the FDA and all products made in such facilities must be manufactured in accordance with current good manufacturing practice ("cGMP") regulations enforced by the FDA. Compliance with cGMP regulations requires the dedication of substantial resources and requires significant expenditures. The FDA periodically inspects both our facilities and procedures to ensure compliance. The FDA may cause a suspension or withdrawal of product approvals if regulatory standards are not maintained. In the event an approved manufacturing facility for a particular drug is required by the FDA to curtail or cease operations, or otherwise becomes inoperable, obtaining the required FDA authorization to manufacture at the same or a different manufacturing site could result in production delays, which could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

-23-

With respect to generic products for which we are the first developer to have its application accepted for filing by the FDA and which filing includes a certification that the applicable patent(s) are invalid, unenforceable and/or not infringed (known as a "Paragraph IV certification"), our ability to obtain and realize the full benefits of six months of market exclusivity is dependent upon a number of factors, including, for example, being the first to file, the status of any litigation that might be brought against us as a result of our filing, or our not meeting regulatory, manufacturing or quality requirements or standards. If any of our products are not timely approved, or if we are unable to obtain and realize the full benefits of six months of market exclusivity for our products cannot be successfully manufactured or timely commercialized, our results of operations could be materially adversely affected. In addition, we cannot guarantee that any investment we make in developing products will be recouped, even if we are successful in commercializing those products.

Also, new products, including contrast agents, are being developed and existing products are being refined in the field of diagnostic imaging. Our own diagnostic imaging agents compete not only with other similarly administrated imaging agents, but also with imaging agents employed in different and often competing diagnostic modalities. New imaging agents in a given diagnostic modality may be developed that provide benefits superior to the then-dominant agent in that modality, resulting in commercial displacement. Similarly, changing perceptions about comparative efficacy and safety, including, among other things, with respect to comparative radiation exposure, and changing availability of supply may favor one agent over another or one modality over another.

#### We may be unable to protect our intellectual property rights or we may be subject to claims that we infringe on the intellectual property rights of others.

We rely on a combination of patents, trademarks, trade secrets, market exclusivity gained from the regulatory approval process and other intellectual property to support our business strategy. However, our efforts to protect our intellectual property rights may not be sufficient. If we do not obtain sufficient protection for our intellectual property, or if we are unable to effectively enforce our intellectual property rights, our competitiveness could be impaired, which would limit our growth and future revenue.

Our pending patent applications may not result in the issuance of patents, or the patents issued to or licensed by us in the past or in the future may be challenged or circumvented by competitors. Existing patents may be found to be invalid or insufficiently broad to preclude our competitors from using methods or making or selling products similar or identical to those covered by our patents and patent applications. Regulatory agencies may refuse to grant us the market exclusivity that we were anticipating, or may unexpectedly grant market exclusivity rights to other parties. In addition, our ability to obtain and enforce intellectual property rights is limited by the unique laws of each country. In some countries it may be particularly difficult to adequately obtain or enforce intellectual property rights, which could make it easier for competitors to capture market share in such countries by utilizing technologies and product features that are similar or identical to those developed or licensed by us. Competitors also may harm our sales by designing products that mirror the capabilities of our products or technology without infringing our patents. Competitors may diminish the value of our trade secrets by reverse engineering or by independent invention. Additionally, current or former employees may improperly disclose such trade secrets to competitors or other third parties. We may not become aware of any such improper disclosure, and, in the event we do become aware, we may not have an adequate remedy available to us.

We operate in an industry characterized by extensive patent litigation, and we may from time to time be a party to such litigation. In *Tyco Healthcare Group LP, et al. v. Mutual Pharmaceutical Company, Inc.*, we filed a patent infringement suit in the U.S. District Court for the District of New Jersey against Mutual Pharmaceutical Co., Inc., et al. (collectively, "Mutual") on March 20, 2007 pursuant to procedures set out in the Drug Price Competition and Patent Term Restoration Act of 1984, after Mutual submitted an Abbreviated New Drug Application ("ANDA") to the FDA seeking to sell a generic version of our 7.5 mg Restoril sleep aid product. Mutual also filed antitrust and unfair competition counterclaims. The patents at issue have since expired or been found invalid. On January 18, 2013, the trial court issued an opinion and order granting our motion for summary judgment regarding Mutual's antitrust and unfair competition counterclaims. On May 1, 2013, Mutual appealed this decision to the U.S. Court of Appeals for the Federal Circuit.

-24-

The pursuit of or defense against patent infringement, such as the case discussed above, is costly and time-consuming and we may not know the outcomes of such litigation for protracted periods of time. We may be unsuccessful in our efforts to enforce our patent or other intellectual property rights. In addition, patent litigation can result in significant damage awards, including the possibility of treble damages and injunctions. Additionally, we could be forced to stop manufacturing and selling certain products, or we may need to enter into license agreements that require us to make significant royalty or up-front payments in order to continue selling the affected products. We can expect to face additional claims of patent infringement in the future. A successful claim of patent or other intellectual property infringement against us could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### We face significant competition and may not be able to compete effectively.

The industries in which we operate are highly competitive. Competition takes many forms, such as price reductions on products that are comparable to our own, development, acquisition or in-licensing of new products that may be more cost-effective than or have performance superior to our products, and the introduction of generic versions when our proprietary products lose their patent protection or market exclusivity. See "Business—Competition" and "Business—Intellectual Property." Our current or future products could be rendered obsolete or uneconomical as a result of this competition. Our failure to compete effectively could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# Any acquisitions of technologies, products and businesses may be difficult to integrate, could materially adversely affect our relationships with key customers and/or could result in significant impairment charges.

We regularly review potential acquisitions of technologies, products and businesses complementary to our business. Acquisitions typically entail many risks and could result in difficulties in integrating operations, personnel, technologies and products. If we are not able to successfully integrate our acquisitions, we may not obtain the advantages and synergies that the acquisitions were intended to create, which may have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. Moreover, the due diligence that we conduct in conjunction with an acquisition may not sufficiently discover risks and contingent liabilities associated with the acquisition target and, consequently, we may consummate an acquisition for which the risks and contingent liabilities are greater than were projected. In addition, in connection with acquisitions, we could experience disruption in our business, technology and information systems, and our customer or employee base, including diversion of management's attention from our continuing operations. There is also a risk that key employees of companies that we acquire or key employees necessary to successfully commercialize technologies and products that we acquire may seek employment elsewhere, including with our competitors. Furthermore, there may be overlap between our products or customers and the companies which we acquire that may create conflicts in relationships or other commitments detrimental to the integrated businesses. Additionally, the time between our expenditures to acquire new products, technologies or businesses and the subsequent generation of revenues from those acquired products, technologies or businesses (or the timing of revenue recognition related to licensing agreements and/or strategic collaborations) could cause fluctuations in our financial performance from period to period. Finally, if we are unable to successfully integrate products, technologies, businesses or personnel that we acquire, we c

#### We may incur product liability losses and other litigation liability.

We are or may be involved in various legal proceedings and certain government inquiries and investigations, including, but not limited to, patent infringement, product liability, antitrust matters, breach of contract, Medicare and/or Medicaid reimbursements claims, or compliance with laws relating to marketing and sales or controlled substance distribution practices, including those relating to the establishment of suspicious order monitoring ("SOM") programs. Such proceedings, inquiries and investigations may involve claims for, or

the possibility of fines and penalties involving substantial amounts of money or other relief, including but not limited to civil or criminal fines and penalties and exclusion from participation in various government healthcare-related programs. If any of these legal proceedings, inquiries or investigations were to result in an adverse outcome, the impact could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

With respect to product liability and clinical trial risks, in the ordinary course of business we are subject to liability claims and lawsuits, including potential class actions, alleging that our marketed products or products in development have caused, or could cause, serious adverse events or other injury. Any such claim brought against us, with or without merit, could be costly to defend and could result in an increase in our insurance premiums. We retain liability for the first \$2.5 million per claim and purchase, through a combination of primary and umbrella/excess liability policies, \$300 million of coverage beyond the retained liabilities. We believe this coverage level is adequate to meet our current business exposure. However, some claims brought against us might not be covered by our insurance policies. Moreover, where the claim is covered by our insurance, if our insurance coverage is inadequate, we would have to pay the amount of any settlement or judgment that is in excess of our policy limits. We may not be able to obtain insurance on terms acceptable to us or at all since insurance varies in cost and can be difficult to obtain. Our failure to maintain adequate insurance coverage or successfully defend against product liability claims could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### The implementation of healthcare reform in the U.S. may materially adversely affect us.

In March 2010, the Patient Protection and Affordable Care Act, as amended by the Health Care and Education Affordability Reconciliation Act (collectively, the "Healthcare Reform Act"), was enacted into law in the U.S. The Healthcare Reform Act contains a number of provisions that affect coverage and reimbursement of drug products and the medical imaging procedures in which our drug products are used. For example, the Healthcare Reform Act includes a provision that imposes a \$28 billion fee on the branded pharmaceutical industry over nine years starting in 2011 and a \$2.8 billion annual fee on the branded pharmaceutical industry thereafter. To the extent that the market share of our Brands business grows, the portion of this fee that we will be obligated to pay will increase.

There can be no assurance that the Healthcare Reform Act as currently enacted will not materially adversely affect our competitive position, business, financial condition, results of operations and cash flows, nor can we predict with certainty how federal or state legislative or administrative changes relating to healthcare will affect our business.

# Sales of our products are affected by the reimbursement practices of a small number of large public and private insurers. In addition, reimbursement criteria and the use of tender systems outside the U.S. could reduce prices for our products or reduce our market opportunities.

In fiscal 2012, approximately 64% of our gross sales were subject to various forms of rebates and chargebacks. Sales of our products depend, in part, on the extent to which the costs of our products are reimbursed by governmental health administration authorities, private health coverage insurers and other third-party payors. Our potential customers' ability to obtain appropriate reimbursement for products and services from these third-party payors affects the selection of products they purchase and the prices they are willing to pay. In addition, demand for new products may be limited unless we obtain reimbursement approval from governmental and private third-party payors prior to introduction. Reimbursement criteria, which vary by country, are becoming increasingly stringent and require management expertise and significant attention to obtain and maintain qualification for reimbursement.

In addition, a number of markets in which we operate have implemented or may implement tender systems in an effort to lower prices. Under such tender systems, manufacturers submit bids which establish prices for products. The company that wins the tender receives preferential reimbursement for a period of time. Accordingly, the tender system often results in companies underbidding one another by proposing low pricing in

-26-

order to win the tender. Certain other countries may consider implementation of a tender system. Even if a tender system is ultimately not implemented, the anticipation of such could result in price reductions. Failing to win tenders, or the implementation of similar systems in other markets leading to price declines, could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### Our reporting and payment obligations under the Medicare and/or Medicaid rebate program and other governmental purchasing and rebate programs are complex. Any determination of failure to comply with these obligations or those relating to healthcare fraud and abuse laws could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

The regulations regarding reporting and payment obligations with respect to Medicare and/or Medicaid reimbursement and rebates and other governmental programs are complex. Because our processes for these calculations and the judgments used in making these calculations involve subjective decisions and complex methodologies, these calculations are subject to the risk of errors. In addition, they are subject to review and challenge by the applicable governmental agencies, and it is possible that such reviews could result in material adjustments to amounts previously paid.

Any governmental agencies that have commenced, or may commence, an investigation of Mallinckrodt relating to the sales, marketing, pricing, quality or manufacturing of pharmaceutical products could seek to impose, based on a claim of violation of fraud and false claims laws or otherwise, civil and/or criminal sanctions, including fines, penalties and possible exclusion from federal healthcare programs including Medicare and/or Medicaid. Some of the applicable laws may impose liability even in the absence of specific intent to defraud. Furthermore, should there be ambiguity with regard to how to properly calculate and report payments—and even in the absence of any such ambiguity—a governmental authority may take a position contrary to a position we have taken, and may impose civil and/or criminal sanctions. There are two cases pending against us that allege generally that we and numerous other pharmaceuticals companies reported false pricing information in connection with certain drugs that are reimbursable under Medicaid, resulting in overpayment by state Medicaid programs for those drugs. These cases, brought by state Attorneys General in Utah and Louisiana, generally seek monetary damages and attorneys' fees. We are named as a defendant in *State of Utah v. Actavis US, Inc., et al.*, filed May 8, 2008, which is pending in the Third Judicial District, Parish of East Baton Rouge, Louisiana. In May 2013, we agreed to terms of settlement with the Attorney General for the state of Louisiana resolving all claims in *State of Louisiana v. Abbott Laboratories Inc., et al.*, involving alleged reporting of false pricing information in connection with certain drugs that certain drugs that are reimbursable under Medicaid, resulting in overpayment by the state Medicaid, resulting in overpayment for the state of Louisiana v. *Abbott Laboratories Inc., et al.*, filed Nay 8, 2008, which is pending in the 19th Judicial District, Parish of East Baton Rouge, Louisiana. In May 2013, we agreed to terms of settlement with the Atto

#### Changes in laws and regulations may materially adversely affect us.

The development, manufacture, marketing, sale, promotion, and distribution of our products are subject to comprehensive government regulation. Changes in laws and regulations could affect us in various ways. For example, both the federal and state governments have given increased attention to the public health issue of opioid abuse, overdose and diversion. At the federal level, the White House Office of National Drug Control Policy continues to coordinate efforts between the FDA, DEA and other agencies to address this problem. In January 2013, the FDA released draft guidance on incorporating abuse-deterrent characteristics into extended-release opioids and held an Advisory Committee meeting, at which the Advisory Committee recommended to the FDA that it reschedule hydrocodone/acetaminophen combination products from DEA Schedule III to Schedule II. When the FDA finds that a new formulation has abuse-deterrent characteristics, the agency has the authority

-27-

to require that generics also have abuse-deterrent characteristics. One of our ANDAs that is currently under review in the U.S. refers to an NDA that did not have abuse-deterrent characteristics. From a compliance standpoint, the DEA continues to increase its efforts to hold manufacturers, distributors and pharmacies accountable through various enforcement actions as well as the implementation of compliance practices for controlled substances, including SOM activities for Schedule II opioids. In addition, many state legislatures continue to consider various bills intended to reduce opioid abuse, overdose and diversion, for example by establishing prescription drug monitoring programs, mandating prescriber education and prohibiting the substitution of generic versions of opioids that lack abuse-deterrent characteristics for branded products that have them. Future legislation and regulation in the markets that we serve could affect access to healthcare products and services, increase rebates, reduce prices or the rate of price increases for healthcare products and services, change healthcare delivery systems, create new fees and obligations for the pharmaceutical industry, or require additional reporting and disclosure. These and other changes in laws and regulations could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### Global economic conditions could harm us.

Over the course of the last few years, global market and economic conditions have been unprecedented and challenging, with tighter credit conditions and recession in most major economies. Continued concerns about the systemic impact of potential long-term and wide-spread recession (including concerns that certain European countries may default on payments due on their national debt), energy costs, geopolitical issues and the availability and cost of credit have contributed to increased market volatility and diminished expectations for developed and developing economies. These conditions, combined with volatile oil prices, declining business and consumer confidence and increased unemployment, have contributed to volatility of unprecedented levels.

As a result of these market conditions, the cost and availability of credit has been and may continue to be adversely affected by illiquid credit markets and wider credit spreads. Concern about the stability of the markets generally and the strength of counterparties specifically has led many lenders and institutional investors to reduce, and in some cases, cease to provide credit to businesses and consumers. These factors have resulted in a decrease in spending by businesses and consumers alike. Continued turbulence in the U.S. and international markets and economies and prolonged declines in consumer spending may materially adversely affect our liquidity and financial condition as well as our share price.

#### Our global operations expose us to risks and challenges associated with conducting business internationally.

We operate globally with offices or activities in Europe, Africa, Asia, South America, Australia and North America. We face several risks inherent in conducting business internationally, including compliance with international and U.S. laws and regulations that apply to our international operations. These laws and regulations include data privacy requirements, labor relations laws, tax laws, anti-competition regulations, import and trade restrictions, export requirements, U.S. laws such as the Foreign Corrupt Practices Act of 1977 ("FCPA") and local laws which also prohibit corrupt payments to governmental officials or certain payments or remunerations to customers. Given the high level of complexity of these laws, there is a risk that some provisions may be violated, for example inadvertently or through fraudulent or negligent behavior of individual employees, our failure to comply with certain formal documentation requirements or otherwise. Violations of these laws and regulations could result in fines or criminal sanctions against us, our officers or our employees, and prohibitions on the conduct of our business. Any such violations could include prohibitions on our ability to offer our products in one or more countries and could materially damage our reputation, our brand, our international expansion efforts, our ability to attract and retain employees, our business and our results of operations. Our success depends, in part, on our ability to anticipate and prevent or mitigate these risks and manage difficulties as they arise.

-28-

In addition to the foregoing, engaging in international business inherently involves a number of other difficulties and risks, including:

- longer payment cycles in countries like Spain and Italy and difficulties in enforcing agreements and collecting receivables through certain non-U.S. legal systems;
- political and economic instability, including the risks and uncertainty associated with the current concerns regarding the stability of the Eurozone and the related possibility of sovereign defaults in countries such as Spain and Italy, and the possibility that such a default or the exit of one or more member countries from the Eurozone or from the European Union ("E.U.") entirely may lead to difficulties for other members of the E.U.;
- potentially adverse tax consequences, tariffs, customs charges, bureaucratic requirements and trade barriers; and
- failure to successfully implement our new non-U.S. operating structure, and difficulties and costs of staffing and managing non-U.S. operations;

These or other factors or any combination of them may have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### Currency exchange rate fluctuations could materially adversely affect our business and results of operations.

We do business and generate sales in numerous countries outside the U.S. As such, currency exchange rate fluctuations may affect the costs that we incur in such international operations. Some of our operating expenses are incurred in non-U.S. dollar currencies. The appreciation of non-U.S. dollar currencies relative to the U.S. dollar in those countries where we have operations could increase our costs and could harm our results of operations and financial condition. In addition, we report our operating results in U.S. dollars, so the appreciation of the U.S. dollar relative to such other currencies could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### Our operations expose us to the risk of material health, safety and environmental liabilities, litigation and violations.

We are subject to numerous federal, state, local and non-U.S. environmental protection and health and safety laws and regulations governing, among other things:

- the generation, storage, use and transportation of hazardous materials;
- emissions or discharges of substances into the environment;
- investigation and remediation of hazardous substances or materials at various sites;
- · chemical constituents in products and end-of-life disposal, mandatory recycling and take-back programs; and
- the health and safety of our employees.

We may not have been, or we may not at all times be, in full compliance with environmental and health and safety laws and regulations. In the event a regulatory authority concludes that we are not in full compliance with these laws, we could be fined, criminally charged or otherwise sanctioned. Environmental laws are becoming more stringent, including outside the U.S., resulting in increased costs and compliance burdens.

Certain environmental laws assess liability on current or previous owners of real property and current or previous owners or operators of facilities for the costs of investigation, removal or remediation of hazardous substances or materials at such properties or at properties at which parties have disposed of hazardous substances. Liability for investigative, removal and remedial costs under certain federal and state laws is retroactive, strict (i.e., can be imposed regardless of fault) and joint and several. In addition to cleanup actions



brought by governmental authorities, private parties could bring personal injury or other claims due to the presence of, or exposure to, hazardous substances. Certain radiological licenses at certain manufacturing sites owned by us require the establishment of decommissioning programs which will require remediation in accordance with regulatory requirements upon cessation of operations at such sites. We have received notification from the U.S. Environmental Protection Agency (the "EPA") and similar state environmental agencies that conditions at a number of sites where we and others have disposed of hazardous substances require investigation, cleanup and other possible remedial action. These agencies may require that we reimburse the government for its costs incurred at these sites or otherwise pay for the costs of investigation and cleanup of these sites, including by providing compensation for natural resource damage claims arising from such sites.

In the ordinary course of our business planning process, we take into account our known environmental matters as we plan for our future capital and operating expenditures requirements. The ultimate cost of site cleanup and timing of future cash outflows is difficult to predict, given the uncertainties regarding the extent of the required cleanup, the interpretation of applicable laws and regulations, and alternative cleanup methods. We concluded that, as of March 29, 2013, it was probable that we would incur remedial costs in the range of \$144.6 million to \$251.7 million. We concluded that, as of March 29, 2013, the best estimate within this range was \$144.6 million. This amount includes \$94.7 million relating to a site located in Orrington, Maine which will be a liability of a Covidien entity following the separation. For more information, see "Unaudited Pro Forma Condensed Combined Financial Statements," "Business— Environmental" and "Business—Legal Proceedings—Environmental Remediation and Litigation Proceedings." Based upon information known to date, we believe our current capital and operating plans are adequate for costs associated with the investigation, cleanup and potential remedial action for our known environmental matters.

While we have planned for future capital and operating expenditures to comply with environmental laws, our costs of complying with current or future environmental protection and health and safety laws and regulations, or our liabilities arising from past or future releases of, or exposures to, hazardous substances may exceed our estimates or could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. We may also be subject to additional environmental claims for personal injury or cost recovery actions for remediation of facilities in the future based on our past, present or future business activities.

#### If we are unable to retain our key personnel, we may be unable to maintain or expand our business.

Because of the specialized scientific nature of our business, our ability to develop products and to compete with our current and future competitors will remain highly dependent, in large part, upon our ability to attract and retain qualified scientific, technical, regulatory and commercial personnel. The loss of key scientific, technical, regulatory and commercial personnel or the failure to recruit additional key scientific, technical, regulatory and commercial personnel could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. There is intense competition for qualified personnel in the areas of our activities, and we may not be able to continue to attract and retain the qualified personnel necessary for the development of our business.

# Our business depends on the continued effectiveness and availability of our information technology infrastructure, and failures of this infrastructure could harm our operations.

To remain competitive in our industry, we must employ information technologies to support manufacturing processes, quality processes, distribution, R&D and regulatory applications that capture, manage and analyze, in compliance with applicable regulatory requirements, the large streams of data generated in our clinical trials. We rely extensively on technology to allow concurrent work sharing around the world. As with all information technology, our systems are vulnerable to potential damage or interruptions from fires, blackouts, telecommunications failures and other unexpected events, as well as to break-ins, sabotage or intentional acts of vandalism. Given the extensive reliance of our business on technology, any substantial disruption or resulting loss of data that is not avoided or corrected by our backup measures could harm our business, operations and financial condition.

-30-

#### **Risks Related to the Separation**

# We have no recent history operating as an independent company, and our historical and pro forma financial information is not necessarily representative of the results that we would have achieved as a separate, publicly traded company and may not be an accurate indicator of our future results of operations.

The historical information about Mallinckrodt in this information statement refers to our business as operated by and integrated with Covidien. Our historical and pro forma financial information included in this information statement is derived from the consolidated financial statements and accounting records of Covidien. Accordingly, the historical and pro forma financial information included in this information included in this information included in this information statement does not necessarily reflect the financial condition, results of operations or cash flows that we would have achieved as a separate, publicly traded company during the periods presented or those that we will achieve in the future primarily as a result of the factors described below:

- Our business has historically been operated by Covidien as part of its broader corporate organization, rather than as an independent company, particularly in relation to its non-U.S. locations. Covidien or one of its affiliates performed various corporate functions for Mallinckrodt, such as accounting, information technology and finance. Following the separation, Covidien will provide some of these functions to us for a period of time, as described in "Our Relationship with Covidien Following the Distribution." Our historical and pro forma financial results reflect allocations of corporate expenses from Covidien for such functions and are likely to be less than the expenses we would have incurred had we operated as a separate, publicly traded company. In addition, we expect to incur additional annual expenses related to the separation, including with respect to, among other things, directors and officers liability insurance, director fees, reporting fees with the SEC, NYSE listing fees, transfer agent fees, increased auditing and legal fees, which expenses may be significant. We will need to make significant investments to replicate or outsource from other providers certain facilities, systems, infrastructure and personnel to which we will no longer have access after our separation from Covidien. These initiatives to develop our independent ability to operate without access to Covidien's existing operational and administrative infrastructure will be costly to implement. We may not be able to operate our business efficiently or at comparable costs, and our profitability may decline;
- Generally, our working capital and capital for our general corporate purposes have historically been provided as part of the corporate-wide cash management policies of Covidien. Following the completion of the separation, we may need to obtain additional financing from lenders, through public offerings or private placements of debt or equity securities, strategic relationships or other arrangements;
- After the completion of the separation, the cost of capital for our business may be higher than Covidien's cost of capital prior to completion of the separation; and
- Currently, we are able to use Covidien's purchasing power in procuring various goods and services and has shared economies of scope and scale in vendor relationships. As a standalone company, we may be unable to obtain goods and services at the prices and terms obtained prior to completion of the separation, which could decrease our overall profitability.

Other significant changes may occur in our cost structure, management, financing and business operations as a result of operating as a company separate from Covidien. For additional information about the past financial performance of our business and the basis of presentation of the historical combined financial statements and the unaudited pro forma combined financial statements of our business, see "Unaudited Pro Forma Condensed Combined Financial Statements," "Selected Historical Combined Financial Data," "Management's Discussion and Analysis of Financial Condition and Results of Operations" and the historical financial statements and accompanying notes included elsewhere in this information statement.

-31-

# As we build our information technology infrastructure and transition our data to our own systems, we could incur substantial additional costs and experience temporary business interruptions.

After the separation, we will continue to install and implement information technology infrastructure to support our critical business functions, particularly in relation to areas outside the U.S., including systems relating to accounting and reporting, manufacturing process control, customer service, inventory control and distribution. We may incur temporary interruptions in business operations if we cannot transition effectively from Covidien's existing transactional and operational systems and data centers and the transition services that support these functions as we replace these systems. We may not be successful in effectively and efficiently implementing our new systems and transitioning our data, and we may incur substantially higher costs for implementation than currently anticipated. Our failure to avoid operational interruptions as we implement the new systems and replace Covidien's services, or our failure to implement the new systems and replace Covidien's services effectively and efficiently, could disrupt our business and could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# Our accounting and other management systems and resources may not be adequately prepared to meet the financial reporting and other requirements to which we will be subject following the separation.

Our financial results previously were included within the consolidated results of Covidien, and our reporting and control systems were appropriate for those of subsidiaries of a public company. Prior to the effectiveness of our registration statement on Form 10, we are not directly subject to reporting and other requirements of the Securities Exchange Act of 1934, as amended (the "Exchange Act") and Section 404 of the Sarbanes-Oxley Act of 2002. After the distribution, we will be subject to such reporting and other requirements, which will require, among other things, annual management assessments of the effectiveness of our internal controls over financial reporting and a report by our independent registered public accounting firm addressing these assessments. These and other obligations will place significant demands on our management, administrative and operational resources, including accounting and information technology resources.

To comply with these requirements, we anticipate that we will need to upgrade our systems, including computer hardware infrastructure, implement additional financial and management controls, reporting systems and procedures and hire additional accounting, finance and information technology staff. If we are unable to upgrade our financial and management controls, reporting systems, information technology and procedures in a timely and effective fashion, our ability to comply with our financial reporting requirements and other rules that apply to reporting companies could be impaired. Moreover, until we complete the creation of the corporate infrastructure necessary to operate as an independent public company, including hiring of additional staff and establishment of financial reporting information systems, we will be reliant on Covidien for services relating to some of our internal controls over financial reporting. Any failure to achieve and maintain effective internal controls could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# We may have received more favorable or less favorable terms from unaffiliated third parties than the terms we will receive in our agreements with Covidien.

We will enter into agreements with Covidien in connection with the separation, including a separation and distribution agreement, a transition services agreement, a tax matters agreement and an employee matters agreement. Since such agreements were negotiated in the context of a separation, the terms of such agreements may be more favorable or less favorable than the terms that would have resulted from arm's-length negotiations between unaffiliated third parties. See "Our Relationship with Covidien Following the Distribution."

-32-

# Covidien may fail to perform under various transaction agreements that will be executed as part of the separation or we may fail to have necessary systems and services in place when certain of the transaction agreements expire.

In connection with the separation, Mallinckrodt and Covidien will enter into a separation and distribution agreement and will enter into various other agreements, including a transition services agreement, a tax matters agreement and an employee matters agreement. These agreements are discussed in greater detail in "Our Relationship with Covidien Following the Distribution." Certain of these agreements will provide for the performance of services by each company for the benefit of the other for a period of time after the separation. We will rely on Covidien to satisfy its performance and payment obligations under these agreements, including its indemnification obligations, we could incur operational difficulties or losses.

If we do not have in place our own systems and services, or if we do not have agreements with other providers of these services when the transaction or long-term agreements terminate, we may not be able to operate our business effectively and our profitability may decline. We are in the process of creating our own, or engaging third parties to provide, systems and services to replace many of the systems and services Covidien currently provides to us. These systems and services may also be more expensive or less efficient than the systems and services Covidien is expected to provide during the transition period.

#### Potential indemnification liabilities to Covidien pursuant to the separation and distribution agreement could materially adversely affect us.

The separation and distribution agreement with Covidien will provide for, among other things, the principal corporate transactions required to effect the separation, certain conditions to the distribution and provisions governing the relationship between Mallinckrodt and Covidien following the separation. For a description of the separation and distribution agreement, see "Our Relationship with Covidien Following the Distribution—Separation and Distribution Agreement." Among other things, the separation and distribution agreement will provide for indemnification obligations principally designed to place financial responsibility for the obligations and liabilities of our business with us and financial responsibility for the obligations and liabilities. If we are required to indemnify Covidien under the circumstances set forth in the separation and distribution agreement, we may be subject to substantial liabilities.

#### We may not achieve some or all of the expected benefits of the separation, and the separation may materially adversely affect our business.

We may not be able to achieve the full strategic and financial benefits expected to result from the separation, or such benefits may be delayed or not occur at all. The separation is expected to provide the following benefits, among others: (i) the ability of each of Covidien and Mallinckrodt to focus on its own strategic and operational plans and capital structure; (ii) an appropriate capital structure for each of Covidien and Mallinckrodt; (iii) a distinct investment identity allowing investors to evaluate the merits, performance and future prospects of Mallinckrodt separately from Covidien; and (iv) more effective equity-based compensation and currency for acquisitions.

We may not achieve these and other anticipated benefits for a variety of reasons, including, among others: (a) the separation will require significant amounts of management's time and effort, which may divert management's attention from operating and growing our business; (b) following the separation, Mallinckrodt may be more susceptible to market fluctuations and other adverse events than if it were still a part of Covidien; (c) following the separation, our business will be less diversified than Covidien's business prior to completion of the separation; and (d) the actions required to separate Covidien's and Mallinckrodt's respective businesses could disrupt our operations. If we fail to achieve some or all of the benefits expected to result from the separation, or if such benefits are delayed, it could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

-33-

# Challenges in the commercial and credit environment may materially adversely affect our ability to issue debt on acceptable terms and our future access to capital.

Our ability to issue debt or enter into other financing arrangements on acceptable terms could be materially adversely affected if there is a material decline in the demand for our products or in the solvency of our customers or suppliers or if other significantly unfavorable changes in economic conditions occur. In addition, volatility in the world financial markets could increase borrowing costs or affect our ability to access the capital markets, which could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

# After the separation, we will have indebtedness, which could restrict our ability to pay dividends and have a negative impact on our financing options and liquidity position.

Immediately following the separation, we expect to bear a total combined indebtedness for borrowed money of approximately \$920 million. We may also incur additional indebtedness in the future. Our indebtedness may impose restrictions on us that could have material adverse consequences by:

- Limiting our ability to obtain additional financing in the future for working capital, capital expenditures and acquisitions;
- · Limiting our ability to refinance our indebtedness on terms acceptable to us or at all;
- Imposing restrictive covenants on our operations;
- Requiring us to dedicate a significant portion of our cash flows from operations to paying the principal of and interest on our indebtedness, thereby
  reducing funds available for other corporate purposes; and
- Making us more vulnerable to economic downturns and limiting our ability to withstand competitive pressures.

See "Description of Material Indebtedness."

# We may need additional financing in the future to meet our capital needs or to make acquisitions, and such financing may not be available on favorable or acceptable terms, and may be dilutive to existing shareholders.

We may need to seek additional financing for general corporate purposes. For example, we may need to increase our investment in R&D activities or need funds to make acquisitions. We may be unable to obtain any desired additional financing on terms that are favorable or acceptable to us. We currently have an investment grade credit rating from Standard & Poors, one of the two primary credit rating agencies that rates our debt. If we were to lose this investment grade credit rating or adequate funds are not available to us on acceptable terms, we may be unable to fund our expansion, successfully develop or enhance products, or respond to competitive pressures, any of which could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. If we raise additional funds through the issuance of equity securities, our shareholders will experience dilution of their ownership interest.

#### No further vote of the Covidien shareholders is required in connection with the distribution. As a result, if the distribution occurs and you do not want to receive our ordinary shares in the distribution, your sole recourse will be to divest yourself of your Covidien ordinary shares prior to the record date.

No further vote of the Covidien shareholders is required in connection with the distribution. Accordingly, if the distribution occurs and you do not want to receive our ordinary shares in the distribution, your only recourse will be to divest yourself of your Covidien ordinary shares prior to the record date for the distribution.

#### **Risks Related to Tax Matters**

# If the distribution fails to qualify as a tax-free transaction for U.S. federal income tax purposes, then Mallinckrodt, Covidien and Covidien's shareholders could be subject to significant tax liability or tax indemnity obligations.

Covidien has received an IRS ruling substantially to the effect that, for U.S. federal income tax purposes, (i) certain transactions to be effected in connection with the separation qualify as transactions under Sections 355 and/or 368(a) of the Code, and (ii) the distribution qualifies as a transaction under Sections 355 and 368(a)(1)(D) of the Code. In addition to obtaining the IRS ruling, Covidien expects to receive a tax opinion from Skadden, Arps, Slate, Meagher & Flom LLP, in form and substance acceptable to Covidien, which tax opinion will rely on the effectiveness of the IRS ruling, substantially to the effect that, for U.S. federal income tax purposes, the distribution and certain transactions entered into in connection with the distribution will qualify as transactions under Sections 355 and/or 368(a) of the Code. The continued validity of the IRS ruling and Covidien's receipt of the tax opinion is a condition to the completion of the distribution.

The IRS ruling relies and the tax opinion will rely on certain facts and assumptions, certain representations from Covidien and Mallinckrodt regarding the past and future conduct of their respective businesses and other matters, and certain undertakings made by Covidien and Mallinckrodt. Notwithstanding the IRS ruling and tax opinion, the IRS could determine on audit that the distribution should be treated as a taxable transaction if it determines that any of these facts, assumptions, representations or undertakings is not correct or has been violated, or that the distribution should be taxable for other reasons, including as a result of a significant change in stock or asset ownership after the distribution, or if the IRS were to disagree with the conclusions of the tax opinion that are not covered by the IRS ruling. If the distribution is ultimately determined to be taxable, the distribution could be treated as a taxable dividend to you for U.S. federal income tax liabilities or tax indemnification obligations, whether under applicable law or the tax matters agreement that we will enter into with Covidien (the "tax matters agreement"), if it is ultimately determined that certain related transactions undertaken in anticipation of the distribution are taxable.

# We could have significant tax liabilities under our tax matters agreement with Covidien, including for periods during which our subsidiaries and operations were those of Tyco International Ltd.

Our tax returns are subject to examination by various tax authorities, including the IRS. The IRS is examining our U.S. federal income tax returns for periods during which certain of our subsidiaries and operations were those of Covidien. In addition, the IRS continues to examine the U.S. federal income tax returns of Tyco International Ltd. ("Tyco International") for periods during which certain of our subsidiaries and operations were those of Tyco International. Our potential liability under the tax matters agreement with Covidien for any taxes related to periods prior to the distribution (after taking into account certain tax benefits realized by us), including those which are subject to the provisions of the tax sharing agreement by and among Covidien, Tyco International and TE Connectivity Ltd. (the "Tyco tax sharing agreement"), is anticipated to be approximately \$150 million, and will be subject to an overall limitation of \$200 million. For a more detailed description of the tax matters agreement, see "Our Relationship with Covidien Following the Distribution—Tax Matters Agreement."

The resolution of the matters arising during periods in which certain of our subsidiaries and operations were subsidiaries and operations of Covidien will be subject to the provisions of the tax matters agreement with Covidien. Under the tax matters agreement with Covidien, Covidien will have the right to administer, control and settle, in its sole and absolute discretion, all tax audits that do not relate solely to non-U.S. taxes for periods prior to the separation that are not covered by the Tyco tax sharing agreement. The outcome of any such examination, and any associated litigation which might arise, is uncertain and could result in a significant increase in our liability for taxes arising during these periods, subject to the overall \$200 million limitation described above. The timing and outcome of such examination or litigation is highly uncertain and could have a material adverse effect

-35-

on our competitive position, business, financial condition, results of operations and cash flows. Under the tax matters agreement, Covidien will agree to provide to us information it receives related to examinations of tax matters for which we may be liable but we will not otherwise be permitted to control or participate in the settlement or defense or such examinations.

The resolution of the matters arising during periods in which certain of our subsidiaries and operations were subsidiaries and operations of Tyco International will be subject to the provisions of the tax matters agreement with Covidien and the Tyco tax sharing agreement. Under the Tyco tax sharing agreement, Covidien, Tyco International and TE Connectivity Ltd. are responsible for 42%, 27% and 31%, respectively, of U.S. income tax liabilities prior to the 2007 separation of Covidien, Tyco International and TE Connectivity Ltd. We are not a party to the Tyco tax sharing agreement. Under our tax matters agreement with Covidien we will, however, be liable for certain taxes relating to our subsidiaries and operations arising during periods governed by the Tyco tax sharing agreement. Although we will be liable to Covidien for certain taxes arising during periods governed by the Tyco tax sharing agreement, we will not be liable to Tyco International or TE Connectivity Ltd. under the Tyco tax sharing agreement, nor will we share in the receivable that Covidien has from Tyco International or TE Connectivity Ltd. In addition, Covidien will retain all reimbursements from Tyco International or TE Connectivity Ltd. pursuant to the Tyco tax sharing agreement, including reimbursements for taxes that are borne by us pursuant to the tax matters agreement with Covidien.

Under the Tyco tax sharing agreement, Tyco International has the right to administer, control and settle all U.S. income tax audits for periods prior to the separation from Tyco International. In connection with such examinations, tax authorities, including the IRS, have proposed tax adjustments. Tyco International has appealed certain of the proposed tax adjustments and it is our understanding that Tyco International intends to vigorously defend its previously filed tax returns. In the event that Tyco International is unable to resolve these issues in the IRS administrative process, Tyco International will likely contest the adjustments through litigation. The outcome of any such litigation is uncertain and could result in a significant increase in our liability for taxes arising during these periods, subject to the overall \$200 million limitation described above. While we believe that the amounts recorded as income taxes payable related to these adjustments are adequate, the timing and outcome of such litigation is highly uncertain and could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. Under the tax matters agreement, Covidien will agree to provide to us information it receives from Tyco International related to examinations of tax matters for which we may be liable that are governed by the Tyco tax sharing agreement.

#### Examination and audits by tax authorities, including the IRS, could result in additional tax payments.

We provide reserves for potential payments of tax to various tax authorities related to uncertain tax positions. It is Covidien's intention to vigorously defend our prior tax returns. However, the calculation of our tax liabilities involves the application of complex tax regulations to our global operations in many jurisdictions. Therefore, any dispute with a tax authority may result in a payment that is materially different from our current estimate of the tax liabilities associated with these returns. If payment of these amounts ultimately proves to be less than the recorded amounts, the reversal of the reserves generally would result in tax benefits being recognized in the period when we determine the reserves are no longer necessary. If our estimate of tax liabilities proves to be less than the amount for which we are ultimately liable, we would incur additional charges to expense and such charges could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### We may not be able to maintain a competitive worldwide effective corporate tax rate.

We cannot give any assurance as to what our effective tax rate will be after the distribution, because of, among other things, uncertainty regarding the tax policies of the jurisdictions where we operate. Our actual effective tax rate

-36-

may vary from our expectation and that variance may be material. Additionally, the tax laws of Ireland and other jurisdictions could change in the future, and such changes could cause a material change in our effective tax rate.

#### **Risks Related to Our Jurisdiction of Incorporation**

#### Legislative action in the U.S. could materially adversely affect us.

Legislative action may be taken by the U.S. Congress which, if ultimately enacted, could limit the availability of tax benefits or deductions that we currently claim, override tax treaties upon which we rely, or otherwise affect the taxes that the U.S. imposes on our worldwide operations. Such changes could materially adversely affect our effective tax rate and/or require us to take further action, at potentially significant expense, to seek to preserve our effective tax rate. In addition, if proposals were enacted that had the effect of limiting our ability as an Irish company to take advantage of tax treaties with the U.S., we could incur additional tax expense and/or otherwise incur business detriment.

#### There is no guarantee that the High Court of Ireland approval of the creation of distributable reserves will be forthcoming.

While we currently do not intend for the foreseeable future to pay dividends, we may determine to pay dividends in the future, subject to applicable law. Under Irish law, dividends must be paid (and share repurchases must generally be funded) out of "distributable reserves," which we will not have immediately following the distribution. See "Description of Mallinckrodt's Share Capital—Dividends" and "Description of Mallinckrodt's Share Capital—Share Repurchases and Redemptions." Immediately after the separation, we will not have any "distributable reserves" but will have a significant amount of share premium. We intend to undertake an Irish legal process pursuant to which we will convert up to our entire share premium account to "distributable reserves." This process will require the approval of the High Court of Ireland. Although we are not aware of any reason why the High Court of Ireland would not approve the creation of distributable reserves in this manner, the issuance of the required order is a matter for the discretion of the High Court of Ireland and there is no guarantee that such approval will be forthcoming. In the event that distributable reserves of Mallinckrodt are not created, no distributions by way of dividends, share repurchases or otherwise will be permitted under Irish law until such time as we have created sufficient distributable reserves from our operating activities.

#### Irish law differs from the laws in effect in the U.S. and may afford less protection to holders of our securities.

It may not be possible to enforce court judgments obtained in the U.S. against us in Ireland based on the civil liability provisions of the U.S. federal or state securities laws. In addition, there is some uncertainty as to whether the courts of Ireland would recognize or enforce judgments of U.S. courts obtained against us or our directors or officers based on the civil liabilities provisions of the U.S. federal or state securities laws or hear actions against us or those persons based on those laws. We have been advised the U.S. currently does not have a treaty with Ireland providing for the reciprocal recognition and enforcement of judgments in civil and commercial matters. Therefore, a final judgment for the payment of money rendered by any U.S. federal or state court based on civil liability, whether or not based solely on U.S. federal or state securities laws, would not automatically be enforceable in Ireland.

A judgment obtained against us will be enforced by the courts of Ireland if the following general requirements are met (i) U.S. courts must have had jurisdiction in relation to the particular defendant according to Irish conflict of law rules (the submission to jurisdiction by the defendant would satisfy this rule) and (ii) the judgment must be final and conclusive and the decree must be final and unalterable in the court which pronounces it. A judgment can be final and conclusive even if it is subject to appeal or even if an appeal is pending. Where however the effect of lodging an appeal under the applicable law is to stay execution of the judgment, it is possible that in the meantime the judgment may not be actionable in Ireland. It remains to be determined whether final judgment given in default of appearance is final and conclusive. However, Irish courts

may refuse to enforce a judgment of the U.S. courts which meets the above requirements for one of the following reasons: (i) if the judgment is not for a definite sum of money; (ii) if the judgment was obtained by fraud; (iii) the enforcement of the judgment in Ireland would be contrary to natural or constitutional justice; (iv) the judgment is contrary to Irish public policy or involves certain U.S. laws which will not be enforced in Ireland; or (v) jurisdiction cannot be obtained by the Irish courts over the judgment debtors in the enforcement proceedings by personal service in Ireland or outside Ireland under Order 11 of the Ireland Superior Courts Rules.

As an Irish company, we are governed by the Irish Companies Acts, which differ in some material respects from laws generally applicable to U.S. corporations and shareholders, including, among others, differences relating to interested director and officer transactions and shareholder lawsuits. Likewise, the duties of directors and officers of an Irish company generally are owed to the company only. Shareholders of Irish companies generally do not have a personal right of action against directors or officers of the company and may exercise such rights of action on behalf of the company only in limited circumstances. Accordingly, holders of our securities may have more difficulty protecting their interests than would holders of securities of a corporation incorporated in a jurisdiction of the U.S.

#### Irish law imposes restrictions on certain aspects of capital management.

Irish law allows our shareholders to pre-authorize shares to be issued by our board of directors without further shareholder approval for up to a maximum of five years. Our current authorization will therefore lapse approximately five years after the distribution unless renewed by shareholders and we cannot guarantee that such renewal will always be approved. Additionally, subject to specified exceptions, including the opt-out that will be included in our articles of association upon consummation of the distribution, Irish law grants statutory pre-emptive rights to existing shareholders to subscribe for new issuances of shares for cash. This opt-out also expires approximately five years after the distribution unless renewed by further shareholder approval and we cannot guarantee that such renewal of the opt-out from pre-emptive rights will always be approved. We cannot assure you that these Irish legal restrictions will not interfere with our capital management. See "Description of Mallinckrodt's Share Capital—Share Capital" and "Description of Mallinckrodt's Share Capital—Pre-emption Rights, Share Warrants and Share Options."

#### **Risks Related to Our Ordinary Shares**

# We cannot be certain that an active trading market for our ordinary shares will develop or be sustained after the distribution, and following the distribution, our share price may fluctuate significantly.

A public market for our ordinary shares does not currently exist. We anticipate that on or prior to the record date for the distribution, trading of our ordinary shares will begin on a "when-issued" basis and will continue through the distribution date. However, we cannot guarantee that an active trading market will develop or be sustained for our ordinary shares after the distribution. We also cannot predict the effect of the distribution on the trading prices of our ordinary shares or whether the combined market value of our ordinary shares and Covidien's ordinary shares will be less than, equal to or greater than the market value of Covidien's ordinary shares prior to the distribution.

The market price of our ordinary shares may fluctuate significantly due to a number of factors, some of which may be beyond our control, including:

- actual or anticipated fluctuations in our results of operations;
- changes in earnings estimated by securities analysts or our ability to meet those estimates;
- the operating and share price performance of comparable companies;
- changes to the regulatory and legal environment in which we operate; and
- U.S. and worldwide economic conditions.

#### -38-

In addition, when the market price of a company's ordinary shares drops significantly, shareholders often institute securities class action lawsuits against the company. A lawsuit against us could cause us to incur substantial costs and could divert the time and attention of our management and other resources.

#### A number of our ordinary shares are or will be eligible for future sale, which may cause our share price to decline.

Any sales of substantial amounts of our ordinary shares in the public market or the perception that such sales might occur, in connection with the distribution or otherwise, may cause the market price of our ordinary shares to decline. Upon completion of the distribution, based on approximately 458 million Covidien ordinary shares outstanding as of May 24, 2013, we expect that we will have an aggregate of approximately 57 million of our ordinary shares issued and outstanding. These shares will be tradable without restriction or further registration under the U.S. Securities Act of 1933, as amended (the "Securities Act"), unless the shares are owned by one of our "affiliates," as that term is defined in Rule 405 under the Securities Act.

We are unable to predict whether large amounts of our ordinary shares will be sold in the open market following the distribution. We are also unable to predict whether a sufficient number of buyers would be in the market at that time. A portion of Covidien's ordinary shares is held by index funds tied to the Standard & Poor's 500 Index ("S&P 500") or other stock indices. We do not expect that Mallinckrodt will be included in the S&P 500. If Mallinckrodt is not included in the S&P 500 or other stock indices at the time of the distribution, these index funds may be required to sell our ordinary shares that they receive in the distribution, which may cause our share price to decline.

#### Your percentage of ownership in Mallinckrodt may be diluted.

Your percentage ownership in Mallinckrodt may be diluted because of equity issuances for acquisitions, capital market transactions or otherwise, including equity awards that we expect to be granting to our directors, officers and employees. Such issuances may have a dilutive effect on our earnings per share, which could materially adversely affect the market price of our ordinary shares. In addition, Covidien equity awards held by Mallinckrodt employees will convert into Mallinckrodt equity awards in connection with the separation.

In addition, our articles of association entitle our board of directors, without shareholder approval, to cause us to issue preferred shares with such terms as the board may determine. Preferred shares may be preferred as to dividends, rights on a winding up or voting in such manner as our directors may resolve. The preferred shares may also be redeemable at the option of the holder of the preferred shares or at the option of Mallinckrodt, and may be convertible into or exchangeable for shares of any other class or classes of our shares, depending on the terms of such preferred shares. The terms of one or more classes or series of preferred shares could dilute the voting power or reduce the value of our ordinary shares. For example, we could grant the holders of preferred shares the right to elect some number of our directors in all events or on the happening of specified events or the right to veto specified transactions. Similarly, the repurchase or redemption rights or liquidation preferences we could assign to holders of preferred shares could affect the residual value of our ordinary shares. See "Description of Mallinckrodt's Share Capital."

# Certain provisions in our articles of association, among other things, could prevent or delay an acquisition of Mallinckrodt, which could decrease the trading price of our ordinary shares.

Our articles of association contain provisions that could have the effect of deterring coercive takeover practices, inadequate takeover bids and unsolicited offers. These provisions include, amongst others:

 provisions of our articles of association which allow the company's board of directors to adopt a shareholder rights plan (commonly known as a "poison pill") upon such terms and conditions as the board of directors deems expedient and in the best interests of Mallinckrodt, subject to applicable law;

-39-

- a provision of our articles of association which generally prohibits us from engaging in a business combination with an interested shareholder for a period of three years following the date the person became an interested shareholder, subject to certain exceptions;
- rules regarding how shareholders may present proposals or nominate directors for election at shareholder meetings;
- the right of our board of directors to issue preferred shares without shareholder approval in certain circumstances, subject to applicable law; and
- the ability of our board of directors to fill vacancies on our board of directors in certain circumstances.

We believe these provisions will provide some protection to our shareholders from coercive or otherwise unfair takeover tactics. These provisions are not intended to make us immune from takeovers. However, these provisions will apply even if the offer may be considered beneficial by some shareholders and could delay or prevent an acquisition that our board of directors determines is in the best interests of Mallinckrodt and its shareholders. These provisions may also prevent or discourage attempts to remove and replace incumbent directors.

In addition, several mandatory provisions of Irish law could prevent or delay an acquisition of Mallinckrodt. For example, Irish law does not permit shareholders of an Irish public limited company to take action by written consent with less than unanimous consent. We also will be subject to various provisions of Irish law relating to mandatory bids, voluntary bids, requirements to make a cash offer and minimum price requirements, as well as substantial acquisition rules and rules requiring the disclosure of interests in our ordinary shares in certain circumstances. Also, Irish companies, including Mallinckrodt, may only alter their memorandum of association and articles of association with the approval of the holders of at least 75% of the company's shares present and voting in person or by proxy at a general meeting of the company.

For additional information on these and other provisions of our articles of association and Irish law that could be considered to have an anti-takeover effect, see "Description of Mallinckrodt's Share Capital—Anti-Takeover Provisions."

The agreements that we will enter into with Covidien in connection with the separation generally will require Covidien's consent to any assignment by us of our rights and obligations under the agreements. The consent and termination rights set forth in these agreements might discourage, delay or prevent a change of control that shareholders may consider favorable. For a more detailed description of these agreements, see "Our Relationship with Covidien Following the Distribution."

Moreover, an acquisition or further issuance of our ordinary shares after the separation could trigger the application of Section 355(e) of the Code, even if the distribution and certain related transactions undertaken in connection therewith otherwise qualify for tax-free treatment. Under Section 355(e) of the Code, we and/or Covidien could incur tax upon certain transactions undertaken in anticipation of the distribution if 50% or more, by vote or value, of our ordinary shares or Covidien ordinary shares are acquired or issued as part of a plan or series of related transactions that include the separation. The process for determining whether an acquisition or issuance triggering these provisions has occurred is complex, inherently factual and subject to interpretation. Any acquisitions or issuances of our ordinary shares or Covidien ordinary shares within two years after the distribution are presumed to be part of such a plan, although we or Covidien, as applicable, may be able to rebut that presumption. Moreover, under the tax matters agreement that we will enter into with Covidien, we will be restricted from engaging in certain transactions within two years of the distribution which potentially could trigger application of Section 355(e) of the Code. See "Our Relationship with Covidien Following the Distribution—Tax Matters Agreement." During such period, these restrictions may limit the ability that we, or a potential acquirer of Mallinckrodt, have to pursue certain strategic transactions that might increase the value of our ordinary shares.

-40-

#### CAUTIONARY STATEMENT CONCERNING FORWARD-LOOKING STATEMENTS

This information statement and other materials Covidien and Mallinckrodt have filed or will file with the SEC contain, or will contain, certain forward-looking statements regarding business strategies, market potential, future financial performance and other matters. The words "believe," "expect," "anticipate," "project" and similar expressions, among others, generally identify "forward-looking statements," which speak only as of the date the statements were made. The matters discussed in these forward-looking statements are subject to risks, uncertainties and other factors that could cause actual results to differ materially from those projected, anticipated or implied in the forward-looking statements. In particular, information included under "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Business," and "The Separation" contain forward-looking statements. Where, in any forward-looking statement, an expectation or belief as to future results or events is expressed, such expectation or belief is based on the current plans and expectations of Mallinckrodt management and expressed in good faith and believed to have a reasonable basis, but there can be no assurance that the expectation or belief will result or be achieved or accomplished. Factors that could cause actual results or events to differ materially from those anticipated include the matters described under "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Sicussion and Analysis of events to differ materially from those anticipated include the matters described under "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations."

-41-

#### DIVIDENDS

### **Dividend Policy**

We currently intend to retain any earnings to finance R&D, acquisitions and the operation and expansion of our business and do not anticipate paying any cash dividends for the foreseeable future. The recommendation, declaration and payment of any dividends in the future by us will be subject to the sole discretion of our board of directors and will depend upon many factors, including our financial condition, earnings, capital requirements of our operating subsidiaries, covenants associated with certain of our debt obligations, legal requirements, regulatory constraints and other factors deemed relevant by our board of directors. Moreover, if we determine to pay any dividends in the future, there can be no assurance that we will continue to pay such dividends.

#### **Creation of Distributable Reserves**

Under Irish law, we require "distributable reserves" in our unconsolidated balance sheet prepared in accordance with the Irish Companies Acts to enable us to make distributions to our shareholders (including by way of the payment of cash dividends or share repurchases). See "Description of Mallinckrodt's Share Capital—Dividends" and "Description of Mallinckrodt's Share Capital—Share Repurchases and Redemptions."

Immediately following the separation, our unconsolidated balance sheet will not contain any distributable reserves, and "shareholders' equity" in such balance sheet will be comprised entirely of "share capital" (equal to the aggregate par value of our ordinary shares issued in the distribution) and "share premium" (resulting from the issuance of our ordinary shares in the distribution and equal to (a) the aggregate value of Covidien's Pharmaceuticals business at the time of its transfer to us less (b) the share capital). We therefore will not have the ability to pay dividends (or make other forms of distributions) immediately following the distribution. The current nominee shareholders of Mallinckrodt are expected to pass a resolution that would (subject to the approval of the High Court of Ireland) create distributable reserves following the distribution by converting to distributable reserves up to all of the share premium of Mallinckrodt.

The creation of distributable reserves described above was approved by Covidien shareholders at Covidien's 2013 Annual General Meeting on March 20, 2013. We will seek to obtain the approval of the High Court of Ireland, which is required for the creation of distributable reserves to be effective, as soon as practicable following the distribution. The approval of the High Court of Ireland is expected to be obtained within approximately two months of the consummation of the distribution, but is dependent on a number of factors, such as the case load of the High Court of Ireland at the time of our initial application, and court vacations.

Until the High Court of Ireland approval is obtained or distributable reserves are created as a result of the profitable operation of the Mallinckrodt group, we will not have sufficient distributable reserves to make distributions by way of dividends, share repurchases or otherwise. Although we are not aware of any reason why the High Court of Ireland would not approve the creation of distributable reserves, there is no guarantee that such approval will be forthcoming.

-42-

### CAPITALIZATION

The following table sets forth our capitalization as of March 29, 2013 on a historical basis and on a pro forma basis to give effect to the pro forma adjustments included in our unaudited pro forma financial information. The historical information below does not necessarily reflect what our capitalization would have been had we operated as a separate, publicly traded company for the period presented and is not necessarily indicative of our future capitalization. This table should be read in conjunction with our unaudited pro forma condensed combined financial statements and accompanying notes, "Management's Discussion and Analysis of Financial Condition and Results of Operations," and our combined financial statements and accompanying notes included elsewhere in this information statement.

		ch 29, 2013
(Dollars in Millions)	Actual	Pro Forma
Cash and Cash Equivalents <sup>(1)</sup>	\$ —	\$ 168.0
Debt:		
Current maturities of long-term debt:		
7.00% debentures due December 2013	\$ 5.8	\$ —
Capital lease obligation	1.3	1.3
Total current maturities of long-term debt and obligation under capital lease	7.1	1.3
Long-term debt:		
Senior unsecured revolving credit facility <sup>(2)</sup>	_	_
3.50% notes due April 2018 <sup>(3)</sup>	_	299.9
9.50% notes due May 2022	_	10.4
8.00% notes due March 2023	—	8.0
4.75% notes due April 2023 <sup>(3)</sup>	_	598.1
Capital lease obligation	2.3	2.3
Total long-term debt and obligation under capital lease	2.3	918.7
Total debt	9.4	920.0
Equity <sup>(4)</sup> :		
Preferred shares, par value \$0.20 per share	_	_
Ordinary shares, par value \$0.20 per share	_	11.8
Additional paid-in capital	_	1,165.4
Parent company investment	2,068.4	_
Accumulated other comprehensive income	71.0	64.5
Total equity	2,139.4	1,241.7
Total capitalization	\$2,148.8	\$2,161.7

- (1) Historical cash and cash equivalents held by Covidien at the corporate level are not specifically identifiable to us. It is anticipated that, on the distribution date, Mallinckrodt will have approximately \$168 million of cash. The separation and distribution agreement will provide for an adjustment payment to potentially be made following the distribution from Covidien to Mallinckrodt plc, or from Mallinckrodt plc to Covidien. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources" and "The Separation—Separation and Distribution Agreement."
- (2) In March 2013, MIFSA entered into a five-year senior unsecured revolving credit facility with a borrowing capacity up to \$250 million. MIFSA will not be permitted to draw upon the credit facility until certain conditions are met, including the completion of the distribution. Mallinckrodt plc will guarantee the credit facility on an unsecured and unsubordinated basis upon completion of the distribution.

-43-

- (3) In April 2013, MIFSA issued \$900 million of senior unsecured notes in a private offering, consisting of \$300 million of 3.50% senior unsecured notes due April 2018 and \$600 million of 4.75% senior unsecured notes due April 2023. Mallinckrodt plc will guarantee the notes on an unsecured and unsubordinated basis upon completion of the distribution. The amounts presented are net of the related discount associated with each series of notes.
- (4) Shareholders' equity assumes 58,781,008 ordinary shares of Mallinckrodt outstanding, based on the number of Covidien shares outstanding on March 29, 2013 and an expected distribution ratio of one ordinary share of Mallinckrodt for every eight Covidien ordinary shares.

Pro forma financial information reflecting our post-distribution capitalization will be included in an amendment to this information statement.

### UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

The following unaudited pro forma condensed combined financial statements have been derived from the historical combined financial statements of the Pharmaceuticals business of Covidien included elsewhere in this information statement. The unaudited pro forma condensed combined income statements assume that the separation from Covidien occurred on October 1, 2011, the first day of fiscal 2012. The unaudited pro forma condensed combined balance sheet assumes that the separation from Covidien occurred on March 29, 2013. These financial statements have been adjusted to reflect the following:

- the transfer by Covidien to us of various corporate non-operating assets and liabilities historically managed by Covidien and its subsidiaries that are
  not related to our business and not included in our historical combined balance sheet and the transfer of certain of our assets and liabilities which will
  be retained by Covidien;
- the distribution of our ordinary shares to Covidien's shareholders and the elimination of historical parent company investment; and
- our anticipated capital structure.

The assumptions used and pro forma adjustments derived from such assumptions are described in the accompanying notes, which should be read in conjunction with the unaudited pro forma condensed combined financial data. The assumptions used and pro forma adjustments derived from such assumptions are based on currently available information. Management believes such assumptions are reasonable.

The following unaudited pro forma condensed combined financial statements should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our combined financial statements and accompanying notes included elsewhere in this information statement. The unaudited pro forma condensed combined financial statements have been presented for informational purposes only. These unaudited pro forma condensed combined financial statements of operations or financial condition had the distribution and related transactions been completed on the dates assumed. Also, they may not reflect the results of operations or financial condition that would have been obtained if we had operated as a separate, publicly traded company during such periods. In addition, they are not necessarily indicative of our future results of operations or financial condition.

During the six months ended March 29, 2013 and fiscal 2012, Covidien allocated general corporate expenses to us in the amount of \$25.5 million and \$49.2 million, respectively. General corporate expenses include, but are not limited to, costs related to finance, legal, information technology, human resources, communications, employee benefits and incentives, insurance and stock-based compensation, which are included in our historical results. Effective upon the separation, we will assume responsibility for all of these functions and related costs and anticipate our costs as a standalone company will be higher than those allocated to us from Covidien. On an annual basis, these operating costs are estimated to be approximately \$40 million higher than the general corporate expenses historically allocated from Covidien to us. In addition, as part of Covidien, we shared in other costs of Covidien, including costs associated with Covidien's international infrastructure. Our portions of these costs were \$20.6 million and \$44.9 million during the six months ended March 29, 2013 and fiscal 2012, respectively. As a standalone company, we expect the recurring annual costs of our own international infrastructure to approximate the amount of costs incurred during fiscal 2012 as part of Covidien. No pro forma adjustments have been made to our financial statements to reflect the additional costs and expenses described in this paragraph because they are projected amounts based on judgmental estimates.

The following pro forma statements do not reflect any impact of the adjustment payment to potentially be made following the distribution from Covidien to us, or from us to Covidien, as the amount of such adjustment payment at the distribution date is not currently determinable and would represent a financial projection. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources" and "The Separation—Separation and Distribution Agreement."

Covidien's debt and the related interest expense have not been allocated to us for any of the periods presented since we are not the legal obligor of the debt and Covidien's borrowings were not directly attributable to our business. Covidien does not intend to use any of the proceeds from our contemplated debt offering to repay any of its indebtedness.

-45-

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF INCOME Six Months Ended March 29, 2013 (in millions, except per share data)

		Pro Forma			
NT ( 1	Historical	Adjustments	Pro Forma		
Net sales	\$1,089.3	\$ —	\$1,089.3		
Cost of sales	582.3		582.3		
Gross profit	507.0	—	507.0		
Selling, general and administrative expenses	307.5		307.5		
Research and development expenses	77.6		77.6		
Separation costs	26.4	(26.4) (a	a) —		
Restructuring charges, net	6.6	—	6.6		
Gain on divestiture	(1.4)		(1.4)		
Operating income	90.3	26.4	116.7		
Other income, net	0.2		0.2		
Interest expense	(0.2)	(21.5) (1	o) (21.7)		
Interest income	0.1		0.1		
Income from continuing operations before income taxes	90.4	4.9	95.3		
Provision for income taxes	36.1	(11.6) (	c) 24.5		
Income from continuing operations	\$ 54.3	\$ 16.5	\$ 70.8		
Pro forma earnings per share from continuing operations:					
Basic			\$ 1.20 (d)		
Diluted			\$ 1.20 (e)		
Pro forma weighted-average shares outstanding:					
Basic			59.0 (d)		
Diluted			59.0 (e)		

See Notes to Unaudited Pro Forma Condensed Combined Financial Statements.

-46-

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF INCOME Fiscal Year Ended September 28, 2012 (in millions, except per share data)

Net sales       \$2,056.2       \$ -       \$2,056.2         Cost of sales       1,091.4       -       1,091.4         Gross profit       964.8       -       964.8         Selling, general and administrative expenses       551.7       -       551.7         Research and development expenses       144.1       -       144.1         Separation costs       25.5       (25.5)       (a)       -         Restructuring charges, net       11.2       -       11.2       -         Gain on divestiture       (2.9)       -       (2.9)       -       (2.9)         Operating income       235.2       25.5       260.7       0         Other income, net       1.0       -       1.0       1.0         Interest income       0.4       -       0.4       0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.62         Income from continuing operations:       \$       141.3       \$       25.0       \$       5.6.3         Pro forma earnings per share from continuing operations:       \$       2.7.7       (d)         D		Historical	Pro Forma Adjustments	Pro Forma
Gross profit         964.8         -         964.8           Selling, general and administrative expenses         551.7         -         551.7           Research and development expenses         144.1         -         144.1           Separation costs         25.5         (25.5)         (a)         -           Restructuring charges, net         11.2         -         11.2           Gain on divestiture         (2.9)         -         (2.9)           Operating income         235.2         25.5         260.7           Other income, net         1.0         -         1.0           Interest expense         (0.5)         (43.1)         (b)         (43.6)           Interest income         0.4         -         0.4           Income from continuing operations before income taxes         236.1         (17.6)         218.5           Provision for income taxes         94.8         (42.6)         (c)         52.2           Income from continuing operations         \$ 141.3         \$ 25.0         \$ 166.3           Pro forma earnings per share from continuing operations:         \$ 2.77         (d)           Basic         \$ 2.77         (d)         \$ 2.76         (e)           Prot forma weighted-av	Net sales			
Selling, general and administrative expenses       551.7        551.7         Research and development expenses       144.1        144.1         Separation costs       25.5       (25.5)       (a)          Restructuring charges, net       11.2        11.2         Gain on divestiture       (2.9)        (2.9)         Operating income       235.2       25.5       260.7         Other income, net       1.0        1.0         Interest expense       (0.5)       (43.1)       (b)       (43.6)         Interest income       0.4        0.4          Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       \$ 141.3       \$ 25.0       \$ 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.77       (d)       \$ 2.76       (e)         Pro forma weighted-average shares outstanding:       \$ 0.1       (d)       \$ 2.76       (e)	Cost of sales	1,091.4	—	1,091.4
Research and development expenses       144.1        144.1         Separation costs       25.5       (25.5)       (a)          Restructuring charges, net       11.2        11.2         Gain on divestiture       (2.9)        (2.9)         Operating income       235.2       25.5       260.7         Other income, net       1.0        1.0         Interest expense       (0.5)       (43.1)       (b)       (43.6)         Interest income       0.4        0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       § 141.3       § 25.0       § 16.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.76       (e)         Diluted       \$ 2.77       (d)         Pro forma weighted-average shares outstanding:       \$ 0.1       (d)	Gross profit	964.8		964.8
Separation costs       25.5 $(25.5)$ (a)          Restructuring charges, net       11.2        11.2         Gain on divestiture $(2.9)$ $(2.9)$ Operating income       235.2       25.5       260.7         Other income, net       1.0        1.0         Interest expense $(0.5)$ $(43.1)$ $(b)$ $(43.6)$ Interest income $0.4$ $0.4$ Income from continuing operations before income taxes       236.1 $(17.6)$ 218.5         Provision for income taxes       94.8 $(42.6)$ $(c)$ $52.2$ Income from continuing operations       § 141.3       § 25.0       § 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77 $(d)$ Basic       \$ 2.76 $(e)$ Pro forma weighted-average shares outstanding:       \$ 60.1 $(d)$	Selling, general and administrative expenses	551.7	—	551.7
Restructuring charges, net       11.2       -       11.2         Gain on divestiture       (2.9)       -       (2.9)         Operating income       235.2       25.5       260.7         Other income, net       1.0       -       1.0         Interest expense       (0.5)       (43.1)       (b)       (43.6)         Interest income       0.4       -       0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       \$ 141.3       \$ 25.0       \$ 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.76       (c)         Pro forma weighted-average shares outstanding:       \$ 60.1       (d)	Research and development expenses	144.1	—	144.1
Gain on divestiture       (2.9)       —       (2.9)         Operating income       235.2       25.5       260.7         Other income, net       1.0       —       1.0         Interest expense       (0.5)       (43.1)       (b)       (43.6)         Interest income       0.4       —       0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       § 141.3       § 25.0       § 166.3         Pro forma earnings per share from continuing operations:       § 2.77       (d)         Basic       \$ 2.76       (e)         Pro forma weighted-average shares outstanding:       5       5       60.1       (d)	Separation costs	25.5	(25.5)	(a) —
Operating income         235.2         25.5         260.7           Other income, net         1.0         —         1.0           Interest expense         (0.5)         (43.1)         (b)         (43.6)           Interest income         0.4         —         0.4           Income from continuing operations before income taxes         236.1         (17.6)         218.5           Provision for income taxes         94.8         (42.6)         (c)         52.2           Income from continuing operations         § 141.3         § 25.0         § 166.3           Pro forma earnings per share from continuing operations:         § 2.77         (d)           Diluted         § 2.76         (e)           Pro forma weighted-average shares outstanding:         60.1         (d)	Restructuring charges, net	11.2	—	11.2
Other income, net       1.0       -       1.0         Interest expense       (0.5)       (43.1)       (b)       (43.6)         Interest income       0.4       -       0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       § 141.3       § 25.0       § 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.76       (e)         Pro forma weighted-average shares outstanding:       \$ 60.1       (d)	Gain on divestiture	(2.9)		(2.9)
Interest expense       (0.5)       (43.1)       (b)       (43.6)         Interest income       0.4        0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       \$ 141.3       \$ 25.0       \$ 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.76       (e)         Pro forma weighted-average shares outstanding:       \$ 60.1       (d)	Operating income	235.2	25.5	260.7
Interest income       0.4       —       0.4         Income from continuing operations before income taxes       236.1       (17.6)       218.5         Provision for income taxes       94.8       (42.6)       (c)       52.2         Income from continuing operations       \$ 141.3       \$ 25.0       \$ 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.76       (e)         Pro forma weighted-average shares outstanding:       \$ 60.1       (d)	Other income, net	1.0	—	1.0
Income from continuing operations before income taxes236.1(17.6)218.5Provision for income taxes94.8(42.6)(c)52.2Income from continuing operations\$ 141.3\$ 25.0\$ 166.3Pro forma earnings per share from continuing operations: Basic\$ 2.77(d)Diluted\$ 2.76(e)Pro forma weighted-average shares outstanding: Basic60.1(d)	Interest expense	(0.5)	(43.1)	(b) (43.6)
Provision for income taxes 94.8 (42.6) (c) 52.2 Income from continuing operations \$ 141.3 \$ 25.0 \$ 166.3 Pro forma earnings per share from continuing operations: Basic Diluted \$ 2.77 (d) Diluted \$ 2.76 (e) Pro forma weighted-average shares outstanding: Basic 60.1 (d)	Interest income	0.4	—	0.4
Income from continuing operations       \$ 141.3       \$ 25.0       \$ 166.3         Pro forma earnings per share from continuing operations:       \$ 2.77       (d)         Basic       \$ 2.76       (e)         Pro forma weighted-average shares outstanding:       \$ 2.76       (e)         Basic       \$ 60.1       (d)	Income from continuing operations before income taxes	236.1	(17.6)	218.5
Pro forma earnings per share from continuing operations:       S       2.77       (d)         Basic       \$       2.76       (e)         Pro forma weighted-average shares outstanding:       Basic       60.1       (d)	Provision for income taxes	94.8	(42.6)	(c) 52.2
Basic\$ 2.77(d)Diluted\$ 2.76(e)Pro forma weighted-average shares outstanding: Basic60.1(d)	Income from continuing operations	\$ 141.3	\$ 25.0	\$ 166.3
Basic\$ 2.77(d)Diluted\$ 2.76(e)Pro forma weighted-average shares outstanding: Basic60.1(d)	Pro forma earnings per share from continuing operations:			
Pro forma weighted-average shares outstanding: Basic 60.1 (d)				\$ 2.77 (d)
Basic 60.1 (d)	Diluted			\$ 2.76 (e)
	Pro forma weighted-average shares outstanding:			
Diluted 60.2 (e)	Basic			60.1 (d)
	Diluted			60.2 (e)

See Notes to Unaudited Pro Forma Condensed Combined Financial Statements.

-47-

### THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET At March 29, 2013 (in millions, except share data)

	Historical	Pro Forma Adjustments		Pro Forma
Assets				
Current Assets:				
Cash and cash equivalents	\$ —	\$ 168.0	(f)	\$ 168.0
Accounts receivable trade, less allowance for doubtful accounts	364.3		(g)	364.3
Inventories	461.4	—		461.4
Prepaid expenses and other current assets	166.3	(2.8)	(h)	163.5
Total current assets	992.0	165.2		1,157.2
Property, plant and equipment, net	969.3			969.3
Goodwill	532.0			532.0
Intangible assets, net	439.8			439.8
Other assets	184.9	33.3	(h)(i)(j)	218.2
Total Assets	\$3,118.0	\$ 198.5		\$3,316.5
Liabilities and Shareholders' Equity				
Current Liabilities:				
Current maturities of long-term debt	\$ 7.1	\$ (5.8)	(f)	\$ 1.3
Accounts payable	100.9	—		100.9
Accrued and other current liabilities			(j)(k)(l)	
	286.8	(8.7)	(m)	278.1
Total current liabilities	394.8	(14.5)		380.3
Long-term debt	2.3	916.4	(f)	918.7
Pension and postretirement benefits	153.4	6.9	(j)	160.3
Environmental liabilities	134.4	(91.6)	(m)	42.8
Other liabilities	293.7	279.0	(m)(n)	572.7
Total Liabilities	978.6	1,096.2		2,074.8
Equity:				
Preferred shares, \$0.20 par value, 500,000,000 authorized; none issued on a pro forma basis				
Ordinary shares, \$0.20 par value, 500,000,000 authorized; 58,781,008 issued and outstanding				
on a pro forma basis	—	11.8	(0)	11.8
Additional paid-in capital	_	1,165.4	(p)	1,165.4
Parent company investment	2,068.4	(2,068.4)	(p)(q)	_
Accumulated other comprehensive income	71.0	(6.5)	(r)	64.5
Total Equity	2,139.4	(897.7)		1,241.7
Total Liabilities and Equity	\$3,118.0	\$ 198.5		\$3,316.5

See Notes to Unaudited Pro Forma Condensed Combined Financial Statements.

-48-

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

- (a) Reflects the removal of separation costs directly related to the separation that were incurred during the historical period. These costs were primarily for legal, tax, accounting and other professional fees.
- (b) Reflects the estimated increase in interest expense in connection with debt we expect to assume prior to or at the time of separation. The pro forma impact was primarily based on the incurrence of \$300 million of 3.50% notes and \$600 million of 4.75% notes.
- (c) Reflects the tax effects of the pro forma adjustments at the applicable statutory income tax rates. Also represents a \$16.9 million and \$44.0 million decrease in income tax expense for the six months ended March 29, 2013 and fiscal 2012, respectively, due to changes in the internal capital structure resulting from the reorganization of our legal entities to facilitate the separation.
- (d) Pro forma basic earnings per share and pro forma weighted-average basic shares outstanding for the six months ended March 29, 2013 and fiscal 2012, respectively, reflect the estimated number of ordinary shares we expect to have outstanding upon completion of the distribution based on the number of Covidien ordinary shares outstanding on March 29, 2013 and September 28, 2012, respectively, adjusted for an assumed distribution ratio of one ordinary shares.
- (e) Pro forma diluted earnings per share and pro forma weighted-average diluted shares outstanding reflect the estimated number of ordinary shares we expect to have outstanding upon completion of the distribution and reflect the potential issuance of ordinary shares under Covidien equity plans in which our employees participate based on the distribution ratio. While the actual dilutive impact in the future may differ from these estimates, we believe this estimate reflects a reasonable approximation of the dilutive impact of Covidien equity plans.
- (f) Primarily reflects the issuance of \$900 million of debt. It is anticipated that, on the distribution date, Mallinckrodt will have approximately \$168 million of cash, which will include a portion of the net proceeds of such debt issuance. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Adjustment Amount" and "The Separation—Separation and Distribution Agreement."
- (g) Upon separation, certain accounts receivable that cannot be segregated by business line will be retained by Covidien.
- (h) Represents the net transfer of \$2.8 million of current deferred tax assets and \$2.9 million of non-current deferred tax assets to Covidien as a result of the reorganization of our legal entities to facilitate the separation.
- (i) Reflects the capitalization of \$9.7 million of debt issuance costs incurred in connection with the issuance of \$900 million of debt.
- (j) Reflects a \$6.9 million increase to pension and postretirement benefits and a \$0.6 million increase to accrued and other current liabilities for pension liabilities that are expected to be transferred to us and a \$26.5 million increase to other assets for the transfer of investments held in a rabbi trust, the assets of which may be used to pay retirement benefits.
- (k) Represents the net transfer of \$3.9 million of current income taxes payable and \$0.7 million of current deferred tax liabilities to Covidien as a result of the reorganization of our legal entities to facilitate the separation.
- (1) Represents the settlement of a \$1.3 million payable associated with an interest rate lock contract discussed in (r).
- (m) Reflects the removal of \$95.0 million of environmental liabilities, of which \$3.4 million is included within accrued and other current liabilities, and the removal of the related \$35.6 million deferred tax asset. These

-49-

environmental liabilities primarily related to a site located in Orrington, Maine and were historically managed by us; however, they will be liabilities of a Covidien entity following the separation.

- (n) Represents \$114.8 million tax liabilities for uncertain tax benefits related to unresolved tax matters that will be transferred to us in connection with the separation, as set forth in the tax matters agreement that we expect to enter into with Covidien. As discussed in "Our Relationship with Covidien Following the Distribution—Tax Matters Agreement," the tax matters agreement will govern the rights and obligations of Mallinckrodt and Covidien for certain tax liabilities with respect to periods or portions thereof ending on or before the date of the distribution. The actual amounts that we may be required to accrue or pay under the tax matters agreement will depend upon a number of factors, including the outcome of the unresolved tax matters. Also reflects a \$128.6 million increase to deferred tax liabilities primarily resulting from the reorganization of our legal entities to facilitate the separation.
- (o) Represents the issuance of approximately 58.8 million ordinary shares at a par value of \$0.20 per share. Our number of ordinary shares is based on the number of Covidien ordinary shares outstanding on March 29, 2013 and an expected distribution ratio of one ordinary share of Mallinckrodt for every eight Covidien ordinary shares.
- (p) Represents the reclassification of Covidien's net investment in us to additional paid-in capital, after considering the effects of the pro forma adjustments described in (q).
- (q) Represents a net reduction to parent company investment as a result of the following:
  - Retention of cash by Covidien and incremental debt assumed from Covidien, net of related issuance costs, which are described in (f) and (i);
  - Assumption of pension liabilities and transfer of related investments held in a rabbi trust, both of which are described in (j);
  - Removal of liabilities described in (l) and (m);
  - Assumption of net tax liabilities described in (h), (k), (m) and (n); and
  - Issuance of Mallinckrodt ordinary shares described in (o).
- (r) In March and April 2013, MIFSA entered into a series of forward interest rate lock contracts to hedge the risk of variability in the market interest rates prior to the issuance of the fixed rate debt. The interest rate locks that terminated subsequent to March 29, 2013 resulted in a loss of \$3.6 million, which is reflected as an adjustment to accumulated other comprehensive income. Also reflects a \$2.9 million adjustment to accumulated other comprehensive income for losses associated with the assumption of pension liabilities described in (j).

-50-

### SELECTED HISTORICAL COMBINED FINANCIAL DATA

The following table sets forth selected financial data for the Pharmaceuticals business of Covidien. The combined statement of income data for the six months ended March 29, 2013 and March 30, 2012 and the combined balance sheet data at March 29, 2013 have been derived from the unaudited condensed combined financial statements included elsewhere in this information statement. The combined statement of income data for fiscal 2012, 2011 and 2010 and the combined balance sheet data as of September 28, 2012 and September 30, 2011 are derived from our audited combined financial statements included elsewhere in this information statement. The combined statement of income data at March 30, 2012, September 24, 2010, September 25, 2009 and September 26, 2008 are derived from our unaudited combined financial statements that are not included in this information statement. The unaudited combined financial statements have been prepared on the same basis as the audited combined financial statements and, in the opinion of management, include all adjustments, consisting only of normal recurring adjustments, necessary for a fair presentation of the information set forth herein.

The selected historical combined financial data presented below should be read in conjunction with our combined financial statements and accompanying notes, "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our unaudited pro forma condensed combined financial statements and accompanying notes included elsewhere in this information statement. Our historical financial data may not be indicative of the results of operations or financial condition that would have been obtained if we had operated as a separate, publicly traded company during the periods presented or of our future performance as an independent company.

Six Mon	ths Ended			Fiscal <sup>(1)</sup>		
March 29, 2013 <sup>(2)</sup>	March 30, 2012 <sup>(3)</sup>	2012 <sup>(4)</sup>	2011 <sup>(5)</sup>	2010(6)	2009(7)(9)	2008(8)(9)
\$ 1,089.3	\$ 1,026.8	\$ 2,056.2	\$ 2,021.8	\$ 2,047.6	\$ 2,429.5	\$ 2,199.8
507.0	488.3	964.8	914.9	932.4	1,296.3	1,023.9
77.6	72.3	144.1	141.5	119.1	155.2	109.2
90.3	127.6	235.2	240.7	240.4	508.5	363.6
90.4	128.3	236.1	243.2	243.2	512.0	366.8
54.3	78.9	141.3	157.0	145.9	315.5	239.0
\$ 3,118.0	\$ 2,842.7	\$ 2,874.6	\$ 2,823.4	\$ 2,888.3	\$ 3,166.9	\$ 3,120.9
2.3	9.6	8.9	10.4	11.6	13.6	14.8
2,139.4	1,857.3	1,891.9	1,788.7	1,835.9	2,016.4	2,128.6
	March 29, 2013 <sup>(2)</sup> \$ 1,089.3 507.0 77.6 90.3 90.4 54.3 \$ 3,118.0 2.3	<u>2013(2)</u> \$ 1,089.3 \$ 1,026.8 507.0 488.3 77.6 72.3 90.3 127.6 90.4 128.3 54.3 78.9 \$ 3,118.0 \$ 2,842.7 2.3 9.6	March 29, 2013 <sup>(2)</sup> March 30, 2012 <sup>(3)</sup> 2012 <sup>(4)</sup> \$ 1,089.3         \$ 1,026.8         \$ 2,056.2           507.0         488.3         964.8           77.6         72.3         144.1           90.3         127.6         235.2           90.4         128.3         236.1           54.3         78.9         141.3           \$ 3,118.0         \$ 2,842.7         \$ 2,874.6           2.3         9.6         8.9	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$

<sup>(1)</sup> Fiscal 2011 includes 53 weeks. All other fiscal years presented include 52 weeks.

<sup>(2)</sup> The six months ended March 29, 2013 includes \$26.4 million of separation costs and \$7.9 million of restructuring and related charges, net.

(3) The six months ended March 30, 2012 includes \$10.9 million of restructuring and related charges, net and \$10.2 million of separation costs.

<sup>(4)</sup> Fiscal 2012 includes \$25.5 million of separation costs and \$19.2 million of restructuring and related charges, net.

<sup>(5)</sup> Fiscal 2011 includes \$10.0 million of restructuring and related charges, net and \$2.9 million of separation costs.

<sup>(6)</sup> Fiscal 2010 includes \$31.3 million of product liability charges and \$11.5 million of restructuring charges, net.

(7) Fiscal 2009 includes a \$71.2 million charge for the estimated additional cost to remediate environmental matters at a site located in Orrington, Maine and \$27.8 million of product liability charges, net of insurance recoveries. Fiscal 2009 also

-51-

includes a \$35.3 million charge related to upfront fees and milestone payments related to a product acquisition and licensing arrangements, which was included in R&D expenses, and \$26.7 million of restructuring charges, net.

- <sup>(8)</sup> Fiscal 2008 includes \$6.1 million of restructuring charges, net.
- <sup>(9)</sup> Includes \$354.5 million and \$56.9 million of sales of oxycodone hydrocodone extended-release tablets in fiscal 2009 and 2008, respectively. These tablets were sold under a license agreement that began in the fourth quarter of fiscal 2008 and ended in the second quarter of fiscal 2009.
- (10) During the first six months of fiscal 2013 and 2012, Covidien allocated to us general corporate expenses in the amount of \$25.5 million and \$22.7 million, respectively. During fiscal 2012, 2011, 2010, 2009 and 2008, Covidien allocated to us general corporate expenses in the amount of \$49.2 million, \$56.3 million, \$60.8 million, \$60.6 million and \$65.3 million, respectively. General corporate expenses include, but are not limited to, costs related to finance, legal, information technology, human resources, communications, employee benefits and incentives, insurance and stock-based compensation. Effective with the separation, we will assume responsibility for all of these functions and related costs and anticipate our costs as a standalone entity will be higher than those allocated to us from Covidien. On an annual basis, these operating costs are estimated to be approximately \$40 million higher than the general corporate expenses historically allocated from Covidien to us.

#### BUSINESS

#### Overview

We are a global company that develops, manufactures, markets and distributes both branded and generic specialty pharmaceuticals, API and diagnostic imaging agents. We use our API products in the manufacture of our generic pharmaceuticals and also sell them to other pharmaceutical companies. Our products are found in almost every hospital, standalone diagnostic imaging center or pharmacy in the U.S. and we have a sales presence in approximately 50 countries. Our diverse product portfolio and solid market positions reflect our 145-year history of pharmaceutical excellence with many innovations important for the treatment of pain, the development of the modern U.S. pharmaceuticals industry and the evolution of nuclear and diagnostic imaging.

We believe that our extensive commercial reach and chemistry expertise, coupled with our ability to deal with the highly regulated and technical nature of our business, have created compelling competitive advantages that we anticipate will sustain future revenue growth. We expect our investments in operating improvements to lead to cost efficiencies and continued margin expansion.

#### **History and Development**

Our Specialty Pharmaceuticals segment can trace its development from the founding of G. Mallinckrodt & Co. in 1867 (predecessor of today's API business). We expanded from the controlled substance API business into controlled substance generics in the mid-1990s to become the 12th largest U.S. generic pharmaceuticals business in 2012 as measured by prescription volume. We started our Brands product portfolio in 2001 with the acquisition of a suite of products, including RESTORIL<sup>™</sup> (temazepam) capsules ("Restoril") and TOFRANIL-PM<sup>™</sup> (imipramine pamoate) capsules ("Tofranil-PM"), from Novartis International AG ("Novartis"). Restoril is indicated for the short-term treatment of insomnia (generally seven to ten days), while Tofranil-PM is indicated for the relief of symptoms of depression. By 2010, we more than doubled our branded pharmaceuticals sales force to over 200 representatives and shifted our focus to pain management. We have since developed the business and are now providing physicians and patients with a comprehensive suite of pain management products, including our Exalgo 32 mg strength extended-release tablets (which were approved by the FDA in August 2012) and our co-promotions of Sumavel<sup>®</sup> DosePro<sup>®</sup> and Duexis<sup>®</sup>. Most recently, in October 2012, we acquired CNS Therapeutics, a specialty pharmaceutical company focused on developing and commercializing products for site-specific administration to the central nervous system to treat neurological disorders and intractable chronic pain.

Our Global Medical Imaging segment traces its start from a series of innovations by Mallinckrodt and its predecessors, including the introduction of barium in 1916, and of iodeikon as the first contrast agent for gall bladder imaging in 1920. In 1989, we launched our non-ionic iodinated contrast media product, Optiray, which remains our largest product and is used in conjunction with CT imaging technology. We further expanded our contrast media product line into the MRI contrast segment with the launch of Optimark in 2000. These products are associated with our CMDS business. We entered the nuclear imaging business in 1966 and started manufacturing and distributing our Ultra-Technekow DTE technetium generators. We subsequently launched a number of cold kits and other radioisotopes to expand our Nuclear Imaging product line. In 1994, we launched Octreoscan, the first molecular imaging agent to diagnose cancer. Finally in 2008, we launched a generic version of Cardiolite<sup>®</sup> (sestamibi), a leading branded cardiac imaging agent, which allowed us to fundamentally change the competitive dynamics for technetium generators and improve the overall profitability of our Global Medical Imaging segment.

In 2010, we divested our nuclear radiopharmacies in the U.S., which allowed us to focus our efforts on manufacturing and stabilizing our Mo-99 supply. Also, in 2010, we divested our Specialty Chemicals business (formerly known as "Mallinckrodt Baker") to focus our businesses more on pharmaceuticals.

-53-

Mallinckrodt plc was incorporated in Ireland on January 9, 2013 for the purpose of holding Covidien's Pharmaceuticals business following the separation. Prior to the transfer by Covidien to us of our business, which will occur immediately prior to the distribution, Mallinckrodt plc will have no business operations.

### **Our Competitive Strengths**

We believe we have the following strengths:

- *Expertise in the acquisition and importation of highly regulated raw materials, and strong regulatory relationships.* We have expertise in the acquisition and importation of highly regulated raw materials, such as opioids, other controlled substances and radioisotopes. For example, in 2012, we believe we received almost 40% of the DEA's total annual quota for controlled substances that we manufacture. In 2012, our Generics business had an approximate 30% market share of DEA Schedule II and III opioid, oral solid doses, based on IMS Health data. The acquisition of certain raw materials and the processing of them into finished products requires a close collaboration with a wide variety of regulatory authorities including the DEA, FDA, NRC, European Medicines Agency and Irish Medicines Board, among many others. We have a long history of working closely with regulatory agencies to ensure ongoing, reliable access to these highly regulated materials.
- Specialized chemistry, development and formulation expertise which supports a sustainable, robust product pipeline. We have specialized chemistry expertise in the formulation of new drug combinations and reformulation of existing drugs into a wide range of products, such as tablets, capsules, oral liquids and injectable products. In late 2009, we completed a significant upgrade to our formulation pilot plant in Webster Groves, Missouri. This expansion greatly enhanced our pharmaceutical formulation capability, which has resulted in a significant increase in both branded and generic formulations that have been approved by the FDA or that are in various stages of pre-clinical development, clinical development or regulatory review. On our Hazelwood, Missouri campus, we have a parenteral pilot plant focused on the reformulation of imaging agents for our Global Medical Imaging segment.
- The broadest portfolio of generic products and controlled substance API for pain and a growing pipeline of branded pharmaceutical pain products. Our Generics and API businesses have a strong position in the controlled substance generics market. We believe our Generics and API businesses offer the broadest product line of opioid and other controlled substances (primarily DEA Schedule II and III). Our strong market position is a result of the following:
  - <sup>1</sup> Formulation and manufacturing expertise in controlled substances and complex generics;
  - <sup>1</sup>Our commitment to investment in our R&D infrastructure and capabilities has resulted in a pipeline of generic and branded controlled substances, many of which are long-acting or hard to formulate products, which are under development or pending approval by the FDA. For example, on December 28, 2012, we became the first company to receive approval from the FDA to manufacture and market in the U.S. a generic version of Concerta. Total gross sales of Concerta and its authorized generic version exceeded \$1.6 billion in the twelve months ended September 30, 2012, according to IMS Health data;
  - <sup>1</sup> Our strong position in controlled substance API and vertical integration from opioid raw materials to finished dosage forms; and
  - <sup>1</sup> U.S. importation restrictions of controlled substance API and finished products.
  - Solid market position in diagnostic imaging agents. We believe that we are one of the top three participants globally in nuclear radiopharmaceutical products. We are one of only two manufacturers of Tc-99m generators (marketed under the brand name Ultra-Technekow DTE) in North America, one of only three in Europe and the only one on either continent that has its own Mo-99 processing facility,

-54-

which provides cost and raw material supply advantages. In CMDS, we offer a fully integrated line of contrast media, pre-filled syringes and proprietary power injectors. Our leading contrast media product, Optiray, has been on the market for over 25 years and is differentiated in part by being offered in pre-filled syringes that fit our proprietary power injectors, which enhances clinician safety and reduces risks in medication management.

- Distinctive high-quality manufacturing and distribution skills with vertical integration where there are competitive advantages. Our manufacturing
  and supply chain capabilities enable highly efficient controlled substance tableting, packaging and distribution. Our investments include one of the
  world's largest DEA Schedule C-II vault storage capacities for raw materials, intermediates and finished dosages. In our Global Medical Imaging
  segment, we have the capability to process Mo-99 for use in our Ultra-Technekow DTE (Tc-99m) generators and to manufacture cyclotron-derived
  isotopes such as thallium-201, indium-111, gallium-67, germanium-68 and iodine-123. In addition, we produce the large-volume terminally sterilized
  pre-filled plastic syringes that fit into our power injectors. Where appropriate, we have also pursued selective vertical integration initiatives to ensure
  our manufacturing and supply chain benefit from cost and productivity efficiencies, such as using several of our API products to provide the raw
  materials for some of our generic products.
- *Global commercial reach*. Our Global Medical Imaging segment operates throughout the world and its direct and indirect marketing and selling capabilities are tailored to business and geographic needs. Our Global Medical Imaging sales presence in approximately 50 countries has positioned us for expansion.
- Strong management team with extensive industry experience. We benefit from having a management team with extensive experience in small, medium and large life sciences firms. Mark Trudeau, who will serve as our President and Chief Executive Officer, has more than 29 years of experience in the pharmaceuticals industry. Prior to joining Covidien in January 2012, Mr. Trudeau served as Chief Executive Officer of Bayer Healthcare LLC USA, the U.S. healthcare business of Bayer AG and as President of Bayer HealthCare Pharmaceuticals U.S. Region. Mr. Trudeau also served on the Board of the Pharmaceutical Researchers and Manufacturers of America, the National Pharmaceutical Council and as a Trustee of the HealthCare Institute of New Jersey. Matthew Harbaugh will serve as our Senior Vice President and Chief Financial Officer. Mr. Harbaugh has worked in Covidien's Pharmaceuticals business since joining Covidien in 2007 and has over 20 years of financial experience, mostly in the life sciences field. Additional members of the senior management team include Steve Carchedi, who will be our Senior Vice President and President, Commercial Operations (North America); Thomas Berry, who will be our Senior Vice President and Chief Human Resources Officer; Meredith Fischer, who will be our Senior Vice President, communications and Public Affairs; and Steve Merrick, who will be our Senior Vice President and President and President and President, Commercial Operations (International) who have 29, 35, 22, 28, 11 and 21 years, respectively, of experience in life sciences fields.

While we have set forth our competitive strengths above, our business involves numerous risks and uncertainties which may prevent us from executing our strategies. These risks include, among others, risks relating to: DEA regulation of the availability of controlled substances that are API, drug products under development and marketed drug products; the highly exacting and complex nature of our manufacturing processes; the limited global supply of fission-produced Mo-99 for use in our Ultra-Technekow DTE generators; our customer concentration; cost-containment efforts of our customers, purchasing groups, third-party payors and governmental organizations; developing or commercializing new products or adapting to a changing technology and diagnostic treatment landscape; protecting our intellectual property rights or being subject to claims that we infringe on the intellectual property rights of others; and significant competition. For a more complete description of the risks associated with our business, see "Risk Factors."

-55-

#### **Our Businesses and Product Strategies**

Information with respect to our Specialty Pharmaceuticals and Global Medical Imaging operating segments is included below and in note 13 to our interim unaudited condensed combined financial statements and note 21 to our annual combined financial statements.

### **Specialty Pharmaceuticals**

Our Specialty Pharmaceuticals segment has two major components: (1) Brands, which we believe will continue to be a growth area for our business, and (2) Generics and API, which we expect will continue to grow and generate significant cash.

Our Brands business markets branded pain drugs, including Exalgo, to physicians. In addition, we have an organic pipeline of branded pain products that are either in clinical trials or awaiting approval from the FDA. We also provide generic drugs, including a variety of product formulations containing hydrocodone, oxycodone, methylphenidate and several other controlled substances. We have a pipeline of controlled substance generic products either in development or awaiting approval from the FDA. Our API business provides bulk API products, including opioids and acetaminophen, to a wide variety of pharmaceutical companies, many of which are direct competitors of our Brands and Generics businesses. In addition, we use our API for internal manufacturing of our finished dosage products. In fiscal 2012, our Specialty Pharmaceuticals segment accounted for 50% of net sales from our operating segments. We expect this segment will represent a larger percentage of our sales over the long term.

We are committed to responsible prescribing, dispensing, use and storage of opioid analgesics to avoid misuse, abuse, addiction, diversion and overdose. In 2010, we started the Collaborating & Acting Responsibly to Ensure Safety Alliance (the "C.A.R.E.S. Alliance"), which offers free non-branded tools and materials to patients, pharmacists and physicians to foster the safe use of opioid pain medications. The C.A.R.E.S. Alliance sponsors drug take back programs among other initiatives. In addition to educational efforts, we work closely with our major distributors to monitor suspicious controlled substance orders and take active steps to limit potential diversion.

#### **Brands**

We started our Brands product portfolio in 2001 with the acquisition of a suite of products, including Restoril and Tofranil-PM, from Novartis. In 2010, we decided to focus on pain management and launched our then newly acquired pain product, Exalgo. We subsequently gained approval for a 32 mg dosage strength of Exalgo in August 2012. In addition, we have filed a New Drug Application ("NDA") for a product in development, known as MNK-395, a diclofenac topical solution in a metered-dose pump. In March 2013, the FDA requested additional information before this application can be considered for approval. In order to comply with this request, we are in the process of repeating a pharmacokinetic study. We anticipate that we will be able to submit the results from this study to the FDA in the third quarter of calendar 2013. Our development pipeline contains two extended-release formulations of controlled substance analgesics, which are in the late stages of clinical development. These two development products are combination products formulated with abuse-deterrent characteristics to address unmet needs in the market. Our strategy is to advance these pipeline products and bring them to market to expand the size and profitability of our Brands business. Moreover, we plan to enhance our branded commercial infrastructure by focusing on a multi-pronged approach of product launches, co-promotions, line extensions and selective acquisitions. Our intention is to increase our branded sales faster than our generic sales to drive margin expansion over the long term.

We promote our branded products directly to physicians (including, for example, pain specialists, anesthesiologists and orthopedic surgeons) with our own direct sales force of over 200 sales representatives. We also use our Brands sales force to co-promote two other products, Sumavel DosePro from Zogenix, Inc. and Duexis from Horizon Pharma, Inc. Sumavel DosePro is a sumatriptan injection that utilizes a needle-free

-56-

delivery system to treat adults who have been diagnosed with acute migraine or cluster headaches. Duexis is a combination of non-steroid anti-inflammatory drug, ibuprofen and H<sub>2</sub>-receptor antagonist famotidine indicated for the relief of signs and symptoms of rheumatoid arthritis and osteoarthritis and administered to decrease the risk of upper gastrointestinal ulcers, which in clinical trials was defined as a gastric and/or duodenal ulcer in patients who are taking ibuprofen for those indications. In addition, we market our branded products directly to managed care organizations to gain access to drug formularies and allow patients access to these medications. Our products are purchased by wholesalers and retail pharmacy chains (among others) and are eventually dispensed by prescription to patients.

The following is a description of select products in our Brands product portfolio:

- *Exalgo* was acquired in June 2009. Exalgo extended-release tablets Class II (8 mg, 12 mg and 16 mg) were approved by the FDA in March 2010 for the treatment of moderate to severe pain in opioid-tolerant patients requiring continuous around-the-clock opioid analgesia for an extended amount of time. We launched these three dosage strengths of Exalgo in late April 2010. Exalgo is the only long-acting once-daily form of hydromorphone on the U.S. market and has shown significant prescription growth since product launch. In August 2012, the FDA approved a 32 mg dosage strength extended-release tablet of Exalgo that further expanded the patient population that Exalgo can effectively treat with a single daily dose. Exalgo was granted marketing exclusivity in the U.S. as a prescription medicine until March 2013. There are two Orange Book-listed patents for this product, both of which expire in July 2014. A third party, pursuant to agreements with us, will have the right to sell Exalgo tablets in the 8 mg, 12 mg and 16 mg dosage strengths beginning in November 2013 and the 32 mg dosage strength beginning in May 2014.
- *Gablofen* was acquired on October 1, 2012 with the acquisition of CNS Therapeutics. Gablofen is indicated for use in management of severe spasticity of cerebral or spinal origin in patients age four years and above. Gablofen is provided in three dosage strengths and in vials (which are generally believed to be safer and more convenient than ampules for clinicians to use), as well as in pre-filled syringes. This pharmaceutical is delivered to the patient via intrathecal administration, *i.e.*, an injection into the sheath around the spinal cord. Along with the acquisition of CNS Therapeutics came a developmental pipeline of additional presentations and strengths of Gablofen, as well as pain products for intrathecal administration.

#### Generics and API

We market our API products to other pharmaceutical companies around the world, many of which are competitors of our Brands and Generics businesses. Additionally, we use our API for internal manufacturing of our finished dosage products. We are among the largest manufacturers of bulk acetaminophen in the world and the only producer of acetaminophen outside of Asia. We manufacture controlled substances under strict DEA quota restrictions and in 2012 we received approximately 40% of the total DEA quota provided to the U.S. market for the controlled substances we manufacture. We believe that our strong market position in the API business and allocation of opioid raw materials from the DEA is a competitive advantage for our API business and in turn for our Generics and Brands businesses. The strategy for our API business is based on manufacturing large volumes of high-quality product and customized product offerings, responsive technical services and timely delivery to our customers.

We believe our Generics and API businesses represent the broadest product line of opioid and other controlled substances (primarily DEA Schedule II and III). Our Generics and API businesses have a strong position in the controlled substance generics market with products, including hydrocodone, hydrocodone-containing tablets, oxycodone and oxycodone-containing tablets, all of which are significant products in the overall pain products segment, as well as methylphenidate and other controlled substance products. Historically, our primary competition has been other U.S. participants due to importation restrictions on controlled substance API and finished products. Our commitment to investment in our R&D infrastructure and capabilities has

resulted in a pipeline of generic controlled substances, many of which are long-acting or hard to formulate products, which are under development or pending approval by the FDA. For example, we were the first company to receive approval from the FDA to manufacture and market a generic version of Concerta, a branded pharmaceutical for the treatment of ADHD. Total gross sales of Concerta and its authorized generic version exceeded \$1.6 billion in the twelve months ended September 30, 2012, according to IMS Health data. An authorized generic version is a version of a branded drug authorized by the holder of the NDA to be marketed under a different label and sold at a lower price. The other method for obtaining approval to produce a generic version is to submit an ANDA and have it approved by the FDA, as we did with respect to Concerta.

We market our generic products principally to drug wholesalers, large- and medium-size retail pharmacy chains, food store chains with pharmacies, pharmaceutical benefit managers that have mail order pharmacies, and hospital buying groups.

The following is a list of significant products and product families in our Generics and API product portfolio:

- Acetaminophen (API) products (represent 11%, 11% and 10% of our total net sales in fiscal 2012, 2011 and 2010, respectively)
- Hydrocodone (API) and hydrocodone-containing tablets
- Oxycodone (API) and oxycodone-containing tablets
- Methylphenidate HCI

#### **Global Medical Imaging**

Our Global Medical Imaging segment develops, manufactures and markets products in two areas: (1) Contrast Media and Delivery Systems used in CT and MRI imaging, and (2) Nuclear Imaging, which provides radiopharmaceuticals used in SPECT imaging for myocardial perfusion cardiac imaging and bone scans. In fiscal 2012, our Global Medical Imaging segment accounted for 50% of net sales from our operating segments. We believe our Global Medical Imaging segment provides a platform for growth outside the U.S. and significant cash generation.

#### **Contrast Media and Delivery Systems**

Our contrast media include the brands Optiray for CT and Optimark for MRI, which are packaged in pre-filled syringes, vials and bottles. Our delivery systems include power injectors to allow delivery of contrast media into the patient, coordination of the timing of the injection with the CT or MRI scanner and delivery of the contrast media at a specific rate and volume. Our CMDS product strategy is based on differentiating our Optiray and Optimark brands with pre-filled syringes as opposed to vials or bulk containers that must be transferred to a syringe for injection. Pre-filled syringes offer a safer alternative to self-filled doses and offer risk reduction benefits that address The Joint Commission (formerly the Joint Commission on Accreditation of Healthcare Organizations) and U.S. Pharmacopeia <797> guidelines. In addition, our pre-filled syringes are color coded and pre-labeled for easier medication management. Our delivery systems are marketed under the brand Optivantage<sup>™</sup> DH for CT, Optistar<sup>™</sup> for MRI and Illumena<sup>™</sup> for cardiac catheterization laboratories. All of our injectors can accept both pre-filled syringes and our disposable syringes for use with saline and/or contrast media. We sell our CMDS products to hospitals and imaging centers through GPOs and otherwise.

The following are significant products in our CMDS product portfolio:

• **Optiray** (ioversol injection) is a low osmolar, lower viscosity and nonionic organically bound solution of iodine with a broad range of indications in CT imaging procedures (including, for example, peripheral and coronary arteriography, angiography and venography). Optiray is available in a Radio Frequency Identification ("RFID")-enabled Ultraject pre-filled syringe that, when combined with a RFID-enabled Optivantage Dual-Head CT Contrast Delivery System ("Optivantage DH")—a medical device used to synchronize the injection of contrast media with the CT scanner—provides a safer and

more efficient method of delivering contrast media. Sales of our Optiray product represent 17%, 19% and 17% of our total net sales in fiscal 2012, 2011 and 2010, respectively. Optiray has been on the market for approximately 25 years. The high capital intensity in manufacturing API for Optiray products and our significant scale have contributed to the longevity of this product.

**Optimark** (gadoversetamide injection) is a non-ionic extracellular Gadolinium-Based Contrast Agent ("GBCA") indicated for use with MRI in patients where abnormal vascularity of the brain or liver is suspected. It is the only GBCA approved by the FDA for administration by power injector and is available in pre-filled syringes to help reduce medication errors and improve patient safety.

#### Nuclear Imaging

Our Nuclear Imaging business manufactures radioactive isotopes for the diagnosis and treatment of disease. Our nuclear radiopharmaceutical product offering includes both "hot" radioisotopes (primarily Tc-99m, used in approximately 80% of nuclear medicine imaging procedures) and "cold" kits (tagging agents that are paired with "hot" radioisotopes for diagnostic procedures). We have significant expertise in managing the highly regulated nature of the radioactive materials used to manufacture the isotope generators and the short half-life of isotopes, which precludes stockpiling and requires exacting execution along all aspects of the supply chain. We believe that our investment in Tc-99m generators in North America and Europe, our own Mo-99 processing facility and a very well-coordinated logistics network provides us with a significant competitive advantage. Our strategy for our Nuclear Imaging business is focused on bolstering the Tc-99m/Mo-99 supply chain through supplier diversification and our investments in new generator manufacturing lines. For example, in the Spring of 2010, we entered into an agreement to obtain Mo-99 from the Maria nuclear research reactor in Poland. The Maria agreement complements our other agreements to obtain Mo-99 from the High Flux Reactor in the Netherlands and the BR2 reactor in Belgium. In addition, we are able to purchase finished Mo-99 from other suppliers in the marketplace with whom we do not have long-term supply agreements. Going forward, we will continue to seek further diversification of our supplier base.

We intend to ultimately eliminate the use of HEU in favor of using LEU. We currently use HEU targets for the production of Mo-99. In 2004, the U.S. National Security Administration established its Global Threat Initiative to, as quickly as possible, identify, secure, remove and/or facilitate the disposition of vulnerable, high-risk nuclear and radiological materials around the world. Included as one of the stated initiatives is the conversion by research reactors and isotope production facilities to LEU from HEU. We are in the process of converting our Mo-99 production operation in the Netherlands to LEU targets. For a discussion of how Mo-99 is used in our business, see "—Raw Materials" and "Risk Factors." We primarily market our nuclear radiopharmaceutical products to nuclear radiopharmacies in the U.S. and to hospitals in Europe.

The following are significant products in our Nuclear Imaging product portfolio:

- *Ultra-Technekow DTE* is a dry-ship, top eluting Tc-99m radioisotope generator that provides an on-site isotope source of Tc-99m solution that is combined by a nuclear pharmacist with various "cold kit" targeting agents to prepare an individualized radiopharmaceutical dose. The prepared Tc-99m radiopharmaceutical is used in procedures using SPECT. SPECT radiopharmaceutical scans account for approximately 85% of all radiopharmaceutical scans and are used in a number of applications, including myocardial perfusion imaging and bone scans. Tc-99m is a decay product of Mo-99, the parent isotope contained in the Tc-99m generator. We are one of only a limited number of manufacturers of Tc-99m generators in North America and in Europe and the only one on either continent that has its own Mo-99 processing facility, which provides significant cost and raw material supply advantages.
- Octreoscan (kit for the preparation of indium In-111 pentetreotide) is a unique molecular imaging agent used for the localization of primary and metastatic neuroendocrine tumors bearing somatostatin receptors. The product was approved by the FDA in June 1994 and is sold primarily in the U.S. and Europe. There are three Orange Book-listed patents for the drug product and usage in detection of neuroendocrine tumors. The last patent expires in July 2015.

-59-

#### **Industry Overview and Trends**

We believe our businesses are well positioned in attractive markets based on a broadening of access to healthcare globally, increased demand for pharmaceutical products from emerging markets and the medical industry's continued focus on diagnostic imaging for the early diagnosis of diseases.

We expect that the specialty pharmaceuticals market in the U.S. will likely grow in the mid-to-high single digits in the near-term, with the most successful companies being focused on innovation in single molecule therapeutics. With respect to branded drugs, most disease areas are addressed by products of a small group of companies that can create extensions of existing brands. Pain management represents the largest therapeutic prescription market in the U.S., with pain medications accounting for approximately one out of every ten dispensed prescriptions in 2011. Pain management is a time-tested therapeutic area, and pain products have been available on the U.S. market since the 1920s.

We believe our experience satisfying the regulatory requirements relating to raw materials for nuclear radiopharmaceuticals provides competitive advantages versus other potential competitors. Currently, imaging tends to be concentrated in developed markets due to its high capital-intensity requirements. However, there are opportunities for growth in emerging markets as governments build out their healthcare infrastructure.

#### Competition

#### Specialty Pharmaceuticals

The pharmaceutical industry is highly competitive. Our Specialty Pharmaceuticals products compete with products manufactured by many other companies in highly competitive markets primarily throughout the U.S. Our competitors vary depending upon therapeutic and product categories. Major competitors of our Specialty Pharmaceuticals business segment include Actavis, Inc. (formerly Watson Pharmaceuticals, Inc.), Endo Health Solutions Inc., Johnson & Johnson, Johnson Matthey ple, Mylan Inc., Noramco, Inc., Pfizer Inc., Purdue Pharma L.P., and Teva Pharmaceutical Ltd., among others. Our secure sources of raw opioid material, vertically integrated manufacturing capabilities, broad offerings of API controlled substances and acetaminophen, comprehensive generic controlled substance product line and established relationships with retail pharmacies enable us to compete effectively with larger generics manufacturers. In addition, we believe that our experience with the FDA, DEA and Risk Evaluation and Mitigation Strategies ("REMS") provides us the knowledge to successfully operate in this highly competitive and highly regulated environment.

In our Brands business, we compete principally through our targeted product development and acquisition and in-licensing strategies. The competitive landscape in the acquisition and in-licensing of pharmaceutical products has intensified in recent years as there has been a reduction in the number of compounds available and an increase in the number of companies and the collective resources bidding on available assets. In addition to product development and acquisitions, other competitive factors in the pharmaceutical industry include product efficacy, safety, ease of use, price, demonstrated cost-effectiveness, marketing effectiveness, service, reliability of supply, reputation and access to technical information.

The highly competitive environment of our Brands business requires us to continually seek out technological innovations and to market our products effectively. Some of our current branded products not only face competition from other brands, but also from generic versions. Generic versions are generally significantly less expensive than branded versions, and, where available, may be required in preference to the branded version under third-party reimbursement programs or substituted by pharmacies. If competitors introduce new products, delivery systems or processes with therapeutic or cost advantages, our products can be subject to progressive price reductions, decreased sales volume or both. Most new products that we introduce must compete with other products already on the market or products that are later developed by competitors. Manufacturers of generic pharmaceuticals typically invest far less in R&D than research-based pharmaceutical companies and therefore can price their products significantly lower than branded products. Accordingly, when a branded product loses its

-60-

market exclusivity, it normally faces intense price competition from generic forms of the product. To successfully compete for business with managed care and pharmacy benefits management organizations, we must often demonstrate that our branded products offer not only medical benefits but also cost advantages as compared with other forms of care.

In our Generics business, we face intense competition from other generic drug manufacturers, brand-name pharmaceutical companies through authorized generics, existing branded equivalents and manufacturers of therapeutically similar drugs. In the market for generic pharmaceuticals, the competition varies depending on the specific product category and dosage strength. One of our key advantages in this market is our vertical integration—the production of our own API for most of our generic products. Among the large generic controlled substance providers, we are the only generic manufacturer that has its own controlled substance API manufacturing capability.

We believe that our competitive advantages in the generic pharmaceuticals business include our ability to introduce new generic versions of brand-name drug products, our formulation expertise and drug delivery technology, our access to controlled substance API, our quality and cost-effective production, our customer service and the breadth of our generic product line.

As a result of consolidation among wholesale distributors and rapid growth of large retail drug store chains, a small number of large wholesale distributors and retail drug store chains control a significant share of the market, and the number of independent drug stores and small drug store chains has decreased. This has resulted in customers gaining more purchasing power. Consequently, there is heightened competition among generic drug producers for the business of this smaller and more selective customer base.

In our API business, we believe that our competitive advantages include our manufacturing capabilities in controlled substances that enable high-speed, high-volume tableting, packaging and distribution. Additionally, we believe we offer customers reliability of supply and broad-based technical customer service.

Newly introduced generic products with limited or no other generic competition are typically sold at higher selling prices. As competition from other generic products increases, selling prices for all participants typically decline. Consequently, the maintenance of profitable operations in generic pharmaceuticals depends, in part, on our ability to select, develop and timely launch new generic products and to manufacture such new products in a cost efficient, high-quality manner. New drugs and future developments in improved and/or advanced drug delivery technologies or other therapeutic techniques may provide therapeutic or cost advantages to competing products.

## **Global Medical Imaging**

We compete primarily on the ability of our products to capture market share. While we believe that the number of procedures using contrast media will grow in emerging markets due in part to increasing access to healthcare, we expect that our ability to compete with other providers of contrast media will be impacted by pricing pressures. We believe that our key product characteristics, such as proven efficacy, reliability and safety, coupled with our core competencies such as our efficient manufacturing processes, established distribution network, field sales organization and customer service, are important factors that distinguish us from our competitors.

The market for imaging agents is highly competitive. Major competitors in our Global Medical Imaging segment include, among others:

- for contrast imaging agents: GE Healthcare, a division of General Electric Company, Bracco Imaging S.p.A., Bayer AG and Guerbet Group;
- for delivery systems: Nemoto & Co, Ltd.; for CMDS: Bayer AG and Bracco Imaging S.p.A.;
- for radiopharmaceutical generators sold in the U.S.: Lantheus Medical Imaging, Inc.;

-61-

- · for radiopharmaceutical generators sold in Europe: GE Healthcare and IBA Group; and
- for radiopharmaceutical SPECT cold kits: Lantheus Medical Imaging, Inc., GE Healthcare, Bracco Imaging S.p.A. and IBA Group.

Unlike most of our competition, we offer a full line of CMDS and radiopharmaceutical products. Our broad product portfolio allows us to be a complete source for most imaging agent needs.

Our current or future products could be rendered obsolete or uneconomical as a result of the competition described above and the factors described in "— Intellectual Property" below. Our failure to compete effectively could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

## **Intellectual Property**

We own or license a number of patents in the U.S. and other countries covering certain products and have also developed brand names and trademarks for other products. Generally, our Brands business relies upon patent protection to ensure market exclusivity for the life of the patent. We consider the overall protection of our patents, trademarks and license rights to be of material value and act to protect these rights from infringement. However, our business is not materially dependent upon any single patent, trademark or license or any group of patents, trademarks or licenses.

In the branded pharmaceutical industry, the majority of an innovative product's commercial value is usually realized during the period in which the product has market exclusivity. In the U.S. and some other countries, when market exclusivity expires and generic versions of a product are approved and marketed, there often are very substantial and rapid declines in the branded product's sales. The rate of this decline varies by country and by therapeutic category; however, following patent expiration, branded products often continue to have some market viability based upon the goodwill of the product name, which typically benefits from trademark protection or is based on the difficulties associated with replicating the product formulation or bioavailability.

An innovator product's market exclusivity is generally determined by two forms of intellectual property: patent rights held by the innovator company and any regulatory forms of exclusivity to which the innovator is entitled.

Patents are a key determinant of market exclusivity for most branded pharmaceuticals. Patents provide the innovator with the right to exclude others from practicing an invention related to the product. Patents may cover, among other things, the active ingredient(s), various uses of a drug product, pharmaceutical formulations, drug delivery mechanisms, and processes for (or intermediates useful in) the manufacture of products. Protection for individual products extends for varying periods in accordance with the expiration dates of patents in the various countries. The protection afforded, which may also vary from country to country, depends upon the type of patent, its scope of coverage, and the availability of meaningful legal remedies in the country.

Market exclusivity is also sometimes influenced by regulatory intellectual property rights. Many developed countries provide certain non-patent incentives for the development of pharmaceuticals. For example, the U.S., E.U. and Japan each provides for a minimum period of time after the approval of certain new drugs during which the regulatory agency may not rely upon the innovator's data to approve a competitor's generic copy. Regulatory intellectual property rights are also available in certain markets as incentives for research on new indications, orphan drugs (*i.e.*, drugs that demonstrate promise for the diagnosis and/or treatment of rare diseases or conditions) and medicines that may be useful in treating pediatric patients. Regulatory intellectual property rights are independent of any patent rights and can be particularly important when a drug lacks broad patent protection. However, most regulatory forms of exclusivity do not prevent a competitor from gaining regulatory approval prior to the expiration of regulatory data exclusivity on the basis of the competitor's own safety and efficacy data on its drug, even when that drug is identical to that marketed by the innovator.

-62-

We estimate the likely market exclusivity period for each of our branded products on a case-by-case basis. It is not possible to predict with certainty the length of market exclusivity for any of our branded products because of the complex interaction between patent and regulatory forms of exclusivity, the relative success or lack thereof by potential competitors' experience in product development and inherent uncertainties concerning patent litigation. There can be no assurance that a particular product will enjoy market exclusivity for the full period of time that we currently estimate or that the exclusivity will be limited to the estimate.

In addition to patents and regulatory forms of exclusivity, we also market products with trademarks. Trademarks have no effect on market exclusivity for a product, but are considered to have marketing value. Trademark protection continues in some countries as long as used; in other countries, as long as registered. Registrations of such trademarks are for fixed terms and subject to renewal as provided by the laws of the particular country.

### **Research and Development**

We devote significant resources to the research and development of products and proprietary drug delivery technologies. We incurred R&D expenses of \$144.1 million, \$141.5 million and \$119.1 million in fiscal 2012, 2011 and 2010, respectively. Our R&D group comprises a number of highly experienced, trained and skilled individuals, with nearly 25% holding Ph.D. degrees.

In our Brands business, we invest significantly into the research and development of our branded products, and plan on increasing such investment in the future. A number of our branded products are protected by patents and have enjoyed market exclusivity. Our R&D strategy focuses on branded product development in the area of pain and other central nervous system areas, such as spasticity. We are presently developing a number of branded products, some of which utilize novel drug-delivery systems, through a combination of internal and collaborative programs. As of April 30, 2013, we had one NDA under review in the U.S.

As noted above, we market our products to pain specialists, anesthesiologists, neurologists and other physician specialists. In targeting future R&D spending, we would consider new products that can be sold to these physician specialists.

In our Generics business, we are presently developing a number of generic products through a combination of internal and collaborative programs. From a product development perspective, we are focused on controlled substances and difficult-to-replicate pharmacokinetic profiles. In addition, we are focused on process improvements to increase yields and reduce costs. As of April 30, 2013, we had five ANDAs awaiting review in the U.S. Our Generics R&D is focused on developing ANDA products that are DEA-controlled substances and difficult to replicate formulations that we believe will provide sustainable growth. Our API R&D is focused on process improvement to our core products to increase manufacturing yields and reduce our costs. We also selectively add API products to our portfolio where we believe we have created a unique, cost-effective and competitive manufacturing process. While we patent some of these API process improvements, many more are kept as trade secrets.

Our main focus for our Global Medical Imaging segment is to enhance our CMDS products by having them communicate directly with hospital information systems, by developing specific devices to target emerging markets and by expanding our nuclear imaging portfolio. To this end, we are improving our CMDS platform by enhancing our RFID technology, pre-filled syringes and data management to provide a seamless transmission of key procedure and product information directly to the user's information systems. In addition, we are designing injectors to meet the needs of emerging markets. In our Nuclear Imaging business, we are expanding our portfolio by developing additional radioisotopes and radiopharmaceuticals while utilizing existing cyclotron and radiopharmaceutical product capacity.

-63-

## Select Products in Development

Our pipeline portfolio contains various products and product candidates that are the reformulation of existing molecules for the treatment of pain and close adjacencies. The following are our most promising pipeline products:

- **MNK-395** is a new 2% formulation of diclofenac topical solution indicated for the treatment of osteoarthritis of the knee. This new formulation was studied using a twice-daily administration and is dispensed for topical usage by a new metered dose pump bottle. The NDA for MNK-395 was submitted in June 2012. In March 2013, the FDA requested additional information before this application can be considered for approval. In order to comply with this request, we are in the process of repeating a pharmacokinetic study. We anticipate that we will be able to submit the results from this study to the FDA in the third quarter of calendar 2013.
- MNK-795 is a novel reformulation of existing controlled substance analgesic combination products that may be indicated for acute, moderate to severe pain. MNK-795 was formulated as a low dose product to fulfill an unmet clinical need in the market and also has certain abuse-deterrent characteristics. The MNK-795 NDA was submitted to the FDA, in May 2013, and we are awaiting official certification of acceptance for the filing. The formulation uses the patented Depomed, Inc. ("Depomed") Acuform<sup>™</sup> drug-delivery technology licensed in 2009.
- MNK-155 is a novel reformulation of a different combination of controlled substance analgesic products that may be indicated for acute, moderate to severe pain. MNK-155 was formulated as a low-dose product to fulfill an unmet clinical need in the market and also has certain abuse-deterrent characteristics. MNK-155 entered Phase III clinical development in the first half of fiscal 2013. The formulation uses the patented Depomed Acuform drug-delivery technology licensed in 2009.
- Intrathecal Product Development—Our acquisition of CNS Therapeutics in 2012 provided us with an R&D pipeline of additional formulations/presentations of Gablofen for the management of severe spasticity, which are at various stages of development. In addition to Gablofen line extensions, we also have several pain products in development for intrathecal administration (*i.e.*, an injection into the sheath around the spinal cord), which would provide an alternative to products that are only available today through compounding pharmacies. Additionally, this R&D pipeline may present opportunities for development of certain products that would be eligible to receive orphan status from the FDA.

## Key Areas of Study

Our R&D scientists have developed expertise in a number of platform technologies, including:

- Formulation of oral solids in novel ways to mimic patented delivery systems;
- Formulation of parenteral products to provide sustained blood levels of select small molecules;
- Linker technology to attach small molecules to radioisotopes; and
- Abuse-deterrent characteristics for oral solids in both immediate-release as well as extended-release to limit abuse and misuse of controlled substances.

While many of these programs are in pre-clinical development, we anticipate that some of these will form the basis of novel products in the future. However, there is no guarantee that any of the studies underway will lead to the development of a product or whether or when such product will be further developed, launched and become commercially viable.

### **Pilot Plants**

To facilitate our development efforts we have two pilot plants where we can test and scale our manufacturing processes for new products without impacting our core manufacturing plants. The two pilot plants are for generic and branded oral formulation product development and for generic or branded parenteral product

-64-

development. In late 2009, we completed a significant upgrade to our formulation pilot plant in Webster Groves, Missouri. This expansion greatly enhanced our pharmaceutical formulation capability, which has resulted in a significant increase in both branded and generic formulations that have been approved by the FDA or that are in various stages of pre-clinical development, clinical development or regulatory review. On our Hazelwood, Missouri campus, we have a parenteral pilot plant focused on the reformulation of imaging agents for our Global Medical Imaging segment.

#### **Quality Assurance Requirements**

The FDA enforces regulations to ensure that the methods used in, and the facilities and controls used for, the manufacture, processing, packaging and holding of drugs and medical devices conform to cGMP. The cGMP regulations the FDA enforces are comprehensive and cover all aspects of manufacturing operations, from receipt of raw materials to finished product distribution, to ensure that the finished product meets all the identity, strength, quality and purity characteristics required of them. The cGMP regulations for devices, called the Quality System Regulations, are also comprehensive and cover all aspects of device manufacture, from pre-production design validation to installation and servicing, insofar as they bear upon the safe and effective use of the device and whether the device otherwise meets the requirements of U.S. Federal Food, Drug and Cosmetic Act (the "FFDCA"). Other regulatory authorities have their own cGMP rules. Ensuring compliance requires a continuous commitment of time, money and effort in all operational areas.

The FDA conducts pre-approval inspections of facilities engaged in the development, manufacture, processing, packing, testing and holding of the drugs subject to NDAs and ANDAs. If the FDA concludes that the facilities to be used do not or did not meet cGMP, good laboratory practice ("GLP") or good clinical practice ("GCP") requirements, it will not approve the application. Corrective actions to remedy the deficiencies must be performed and are usually verified in a subsequent inspection. In addition, manufacturers of both pharmaceutical products and API used to formulate the drug also ordinarily undergo a pre-approval inspection, although the inspection can be waived when the manufacturer has had a passing cGMP inspection in the immediate past. Failure of any facility to pass a pre-approval inspection will result in delayed approval and could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

The FDA also conducts periodic inspections of drug and device facilities to assess their cGMP status. If the FDA were to find serious cGMP noncompliance during such an inspection, it could take regulatory actions that could materially adversely affect our business, results of operations, financial condition and cash flows. Additionally, imported API and other components needed to manufacture our products could be rejected by U.S. Customs and Border Protection, usually after conferring with the FDA. In the case of domestic facilities, the FDA could initiate product seizures or, in some instances, require product recalls and seek to enjoin a product's manufacture and distribution. In certain circumstances, violations could support civil penalties and criminal prosecutions. In addition, if the FDA concludes that a company is not in compliance with cGMP requirements, sanctions may be imposed that include preventing that company from receiving the necessary licenses to export its products and classifying that company as an "unacceptable supplier," thereby disqualifying that company from selling products to federal agencies.

### **Regulatory Matters**

#### United States

In general, drug manufacturers operate in a highly regulated environment. In the U.S., we must comply with laws, regulations, guidance documents and standards promulgated by the FDA, the Department of Health and Human Services, the DEA, the EPA, the NRC, the Customs Service and state boards of pharmacy.

The FDA's authority to regulate the safety and efficacy of pharmaceuticals comes from the FFDCA. In addition to reviewing NDAs, for branded drugs, and ANDAs, for generic drugs, the FDA has the authority to ensure that pharmaceuticals introduced into interstate commerce are neither "adulterated" nor "misbranded."

Adulterated means that the product may cause or has caused injury to patients when used as intended because it fails to comply with current cGMP. Misbranded means that the labels of, or promotional materials for, the product contain false or misleading information. Failure to comply with applicable FDA and other federal and state regulations could result in product recalls or seizures, partial or complete suspension of manufacturing and/or distribution, refusal to approve pending NDAs or ANDAs, monetary fines, civil penalties and/or criminal prosecution.

In order to market and sell a new prescription drug product in the U.S., a drug manufacturer must file with the FDA an NDA that shows the safety and effectiveness of (a) a new chemical entity that serves as the API, known as a 505(b)(1) NDA; or (b) a product that has significant differences from an already approved one (*e.g.*, different dosage strengths or route of administration), known as a 505(b)(2) NDA. Alternatively, in order to market and sell a generic version of an already approved drug product, a drug manufacturer must file an ANDA that shows that the generic version is "therapeutically equivalent" (*i.e.*, behaves almost the same when taken by a patient) to the branded drug product and therefore is substitutable.

*NDA Process*. The path leading to FDA approval of an NDA for a new chemical entity ("NCE") begins when the drug product is merely a chemical formulation in the laboratory. In general, the process involves the following steps:

- Completion of formulation, laboratory and animal testing in accordance with GLP that fully characterizes the drug product from a pre-clinical perspective and provides preliminary evidence that the drug product is safe to test in human beings;
- Filing with the FDA an Investigational New Drug Application that will permit the conduct of clinical trials (*i.e.*, testing in human beings under adequate and well-controlled conditions);
- Designing and conducting clinical trials to show the safety and efficacy of the drug product in accordance with GCP;
- Submitting the NDA for FDA review, which provides a complete characterization of the drug product;
- Satisfactory completion of FDA pre-approval inspections regarding the conduct of the clinical trials and the manufacturing processes at the designated facility in accordance with cGMP;
- If applicable, satisfactory completion of an FDA Advisory Committee meeting in which the Agency requests help from outside experts in evaluating the NDA;
- Final FDA approval of the full prescribing information, labeling and packaging of the drug product; and
- Ongoing monitoring and reporting of adverse events related to the drug product, implementation of a REMS program, if applicable, and conduct of required Phase IV studies.

Clinical trials are typically conducted in four sequential phases, although they may overlap. The four phases are as follows:

- Phase I trials are typically small (less than 100 healthy volunteers) and are designed to determine the toxicity and maximum safe dose of the drug product.
- Phase II trials usually involve 100 to 300 participants and are designed to determine whether the drug product produces any clinically significant effects in patients with the intended disease or condition. If the results of these trials show promise, then a larger Phase III trial may be conducted.
- Phase III trials are often multi-institution studies that involve a large number of participants and are designed to show efficacy. Phase III (and some Phase II) trials are designed to be pivotal, or confirmatory trials. The goal of a pivotal trial is to establish the safety and efficacy of a drug product by eliminating biases and increasing statistical power.

In some cases, the FDA requires Phase IV trials, which are usually performed after the NDA has been approved. Such post-marketing surveillance is intended to obtain more information about the risks of harm, benefits and optimal use of the drug product by observing the results of the drug product in a large number of patients.

A drug manufacturer may conduct clinical trials either in the U.S. or outside the U.S., but in all cases must comply with GCP, which includes (a) a legally effective informed consent process when enrolling participants; (b) an independent review by an Institutional Review Board ("IRB") to minimize and manage the risks of harm to participants; and (c) ongoing monitoring and reporting of adverse events related to the drug product.

The path leading to FDA approval of an NDA for a drug product that has significant differences from an already approved one is somewhat shorter. The FDA requires a drug manufacturer to submit data from either already published reports or newly conducted studies that show the safety and efficacy of those differences.

Under the U.S. Prescription Drug User Fee Act, the FDA has the authority to collect fees from drug manufacturers who submit NDAs for review and approval. These user fees help the FDA fund the drug approval process. Currently, the user fee rate has been set at \$1,958,800 for a 505(b)(1) NDA and \$979,400 for an NDA not requiring clinical data, generally a 505(b)(2) NDA. We expense these fees as they are incurred. The average review time for an NDA is approximately six to ten months.

In addition, a drug manufacturer may decide to conduct a clinical trial of a drug product on pediatric patients in order to obtain a form of marketing exclusivity as permitted under the Best Pharmaceuticals for Children Act (the "BPCA"). Alternatively, the FDA may require a drug manufacturer, using its authority under the Pediatric Research Equity Act, to conduct a pediatric clinical trial. The goal of conducting pediatric clinical trials is to gather data on how drug products should best be administered to this patient population.

ANDA Process. The path leading to FDA approval of an ANDA is much different from that of an NDA. By statute, the FDA waives the requirement for a drug manufacturer to complete pre-clinical studies and clinical trials and instead focuses on data from bioequivalence studies. The term "bioequivalence studies" generally involves comparing the absorption rate and concentration levels of a generic drug in the human body to that of the branded drug or Reference Listed Drug (the "RLD"). In the event that the generic drug behaves in the same manner in the human body as the RLD, the two drug products are considered bioequivalent. The FDA considers a generic drug "therapeutically equivalent" and therefore substitutable (*i.e.*, the generic drug will produce the same clinical effect and safety profile as the RLD) if it also contains the same active ingredients, dosage form, route of administration and strength.

At present, the average review time for an ANDA is approximately 27 months. In 2010, the FDA's Office of Generic Drugs reported a backlog of over 2,000 ANDAs. To address this problem, U.S. Congress has granted the FDA authority to collect, for the first time, user fees from generic drug manufacturers who submit ANDAs for review and approval. Monies collected under the Generic Drug User Fee Act will help the FDA fund the drug approval process. For fiscal 2013, the user fee rate is set at \$51,520 for an ANDA and \$25,760 for a prior approval supplement to an ANDA. In addition, the FDA also will collect from generic drug manufacturers a one-time backlog fee, a one-time Drug Master File first reference fee, and separate annual manufacturing facility fees for API and finished drug products. These fees are expensed as incurred. The FDA anticipates that the approval process timeframe will not begin to improve until fiscal 2015.

Aside from the backlog described above, the timing of FDA approval of ANDAs depends on other factors, including whether an ANDA holder has challenged any listed patents to the RLD and whether the RLD is entitled to one or more periods of marketing exclusivity under the FFDCA (such as pediatric exclusivity under the BPCA). In general, the FDA will not approve (but will continue to review) an ANDA in which the RLD holder has sued, within 45 days of receiving notice of the ANDA filing, the ANDA holder for patent infringement until either the litigation has been resolved or 30 months has elapsed, whichever is later.

-67-

For all pharmaceuticals sold in the U.S., the FDA also regulates sales and marketing to ensure that drug product claims made by manufacturers are neither false nor misleading. Manufacturers are required to submit copies of all product-specific promotional materials to the FDA's Office of Prescription Drug Promotion prior to their first use by sales representatives. In general, such advertising does not require FDA prior approval, although most manufacturers submit their direct-to-consumer advertising to the FDA for its prior review. Failure to implement a robust internal company review process and comply with FDA regulations regarding advertising and promotion increases the risk of enforcement action by either the FDA or the U.S. Department of Justice ("DOJ").

For both NDAs and ANDAs, the manufacture, marketing and selling of certain drug products may be limited by quota grants for controlled substances by the DEA. See "—Drug Enforcement Administration" below.

Patent and Non-Patent Exclusivity Periods. A sponsor of an NDA is required to identify in its application any patent that claims the drug or a use of the drug subject to the application. Upon NDA approval, the FDA lists these patents in a publication referred to as the Orange Book. Any person that files a Section 505(b)(2) NDA, the type of NDA that relies upon the data in the application for which the patents are listed, or an ANDA to secure approval of a generic version of a previous drug, must make a certification in respect to listed patents. The FDA may not approve such an application for the drug until expiration of the listed patents unless (1) the generic applicant certifies that the listed patents are invalid, unenforceable or not infringed by the proposed generic drug and gives notice to the holder of the NDA for the RLD of the bases upon which the patents are challenged, and (2) the holder of the RLD does not sue the later applicant for patent infringement within 45 days of receipt of notice. If an infringement suit is filed, the FDA may not approve the later application until the earliest of: (i) 30 months after receipt of the notice by the holder of the NDA for the RLD; (ii) entry of an appellate court judgment holding the patent invalid, unenforceable or not infringed; (iii) such time as the court may order; or (iv) the expiration of the patent.

One of the key motivators for challenging patents is the 180-day market exclusivity period ("generic exclusivity") vis-à-vis other generic applicants granted to the developer of a generic version of a product that is the first to make a Paragraph IV certification and that prevails in litigation with the manufacturer of the branded product over the applicable patent(s) or is not sued. For a variety of reasons, there are situations in which a company may not be able to take advantage of an award of generic exclusivity. The determination of when generic exclusivity begins and ends is very complicated.

The holder of the NDA for the RLD may also be entitled to certain non-patent exclusivity during which the FDA cannot approve an application for a competing generic product or 505(b)(2) NDA product. Generally, if the RLD is a new chemical entity, the FDA may not accept for filing any application that references the innovator's NDA for five years from the approval of the innovator's NDA. However, this five-year period is shortened to four years where a filer's ANDA includes a Paragraph IV certification. In other cases, where the innovator has provided certain clinical study information, the FDA may accept for filing, but may not approve, an application that references the innovator's NDA for a period of three years from the approval of the innovator's NDA.

Certain additional periods of exclusivity may be available if the RLD is indicated for use in a rare disease or condition or is studied for pediatric indications.

*Risk Evaluation and Mitigation Strategies*. For certain drug products or classes, such as transmucosal immediate-release fentanyl products and extendedrelease and long-acting opioids, the FDA has the authority to require the manufacturer to provide a REMS that is intended to ensure that the benefits of a drug product (or class of drug products) outweigh the risks of harm. The FDA may require that a REMS include elements to ensure safe use to mitigate a specific serious risk of harm (*e.g.*, prescribers must have particular training or experience or the drug product must be dispensed in certain healthcare settings). The FDA has the authority to impose civil penalties on or take other enforcement action against any drug manufacturer who fails to properly implement an approved REMS program. Separately, a drug manufacturer cannot use an approved REMS program to delay generic competition.

-68-

In December 2011, the FDA approved a single, class-wide REMS program for transmucosal immediate-release fentanyl products (called the "TIRF REMS Access Program") in order to ease the burden on the healthcare system. TIRF products are opioids used to manage pain in adults with cancer who routinely take other opioid pain medicines around-the-clock. We were part of the original industry working group that collaborated to develop and implement this REMS program. The goals of this REMS program are to ensure patient access to important medications and mitigate the risk of misuse, abuse, addiction, overdose and serious complications due to medication errors by: (a) prescribing and dispensing only to appropriate patients, including use only in opioid-tolerant patients; (b) preventing inappropriate conversion between fentanyl products; (c) preventing accidental exposure to children and others for whom such products were not prescribed; and (d) educating prescribers, pharmacists and patients on the potential for misuse, abuse, addiction and overdose. This program started in March 2012 and requires manufacturers, distributors, prescribers, dispensers and patients to enroll in a real-time database that maintains a closed-distribution system.

In February 2009, the FDA requested that drug manufacturers help develop a single, shared REMS for extended-release and long-acting opioid products that contain fentanyl, hydromorphone, methadone, morphine, oxycodone and oxymorphone. In April 2009, the FDA announced that the "REMS would be intended to ensure that the benefits of these drugs continue to outweigh the risks associated with: (1) use of high doses of long-acting opioids and extended-release opioid products in non-opioid-tolerant and inappropriately selected individuals; (2) abuse; (3) misuse; and (4) overdose, both accidental and intentional." We were part of the original industry working group that collaborated to develop and implement this REMS program. Upon FDA approval of Exalgo in March 2010, we implemented the product-specific REMS program that was developed internally while continuing to collaborate on the class-wide REMS program. In July 2012, the FDA approved a class-wide REMS program (called the "Extended-Release and Long-Acting Opioid Analgesics REMS") that affected more than 30 extended-release and long-acting opioid analgesics (both branded and generic products). This REMS program requires drug manufacturers to make available training on appropriate prescribing practices for healthcare professionals who prescribe these opioid analgesics and to distribute educational materials on their safe use to prescribers and patients.

As part of our ongoing commitment to the responsible prescribing, dispensing and safe use of prescription opioids beyond the FDA's REMS requirements, we launched the C.A.R.E.S. Alliance in September 2010. For a discussion of the C.A.R.E.S. Alliance, see "—Our Business and Product Strategies—Specialty Pharmaceuticals."

Drug Enforcement Administration. The DEA is the federal agency responsible for domestic enforcement of the CSA. The CSA classifies drugs and other substances based on identified potential for abuse. Schedule I controlled substances, such as heroin and LSD, have a high abuse potential and have no currently accepted medical use; thus, they cannot be lawfully marketed or sold. Opioids, such as oxycodone, oxymorphone, morphine, fentanyl and hydrocodone, are either Schedule II or III controlled substances. Consequently, the manufacture, storage, distribution and sale of these substances are highly regulated.

The DEA regulates the availability of API, products under development, and marketed drug products that are Schedule II or III by setting annual quotas. Every year, we must apply to the DEA for manufacturing quota to manufacture API and procurement quota to manufacture finished dosage products.

Given that the DEA has discretion to grant or deny our manufacturing and procurement quota requests, the quota the DEA grants may be insufficient to meet our commercial and R&D needs. The initial hydrocodone manufacturing and procurement quota grants we received from the DEA for 2012 were below the amounts we requested and were therefore insufficient to meet customer demand. We subsequently requested supplemental manufacturing and procurement quota request (to manufacture API) but the DEA granted the full amount of our hydrocodone procurement quota request (to manufacture finished dosage products). While our Hobart, New York facility had sufficient hydrocodone bulk API manufacturing quota, which resulted in our

-69-

inability to fulfill third-party customer requests. Subsequently, the DEA published a revised proposed U.S. aggregate quota for bulk manufacture of hydrocodone, and in August 2012, we filed another supplemental hydrocodone manufacturing quota request. In October 2012, the DEA granted 78% of our requested amount. This hydrocodone bulk API manufacturing quota shortage resulted in lost sales, the amount of which was not significant.

Any future delay or refusal by the DEA to grant, in whole or in part, our quota requests could delay or result in stopping the manufacture of our marketed drug products, new product launches or the conduct of bioequivalence studies and clinical trials.

DEA regulations make it extremely difficult for a manufacturer in the U.S. to import finished dosage forms of controlled substances manufactured outside the U.S. These rules reflect a broader enforcement approach by the DEA to regulate the manufacture, distribution and dispensing of legally produced controlled substances. Accordingly, drug manufacturers who market and sell finished dosage forms of controlled substances in the U.S. typically manufacture or have them manufactured in the U.S.

The DEA also requires drug manufacturers to design and implement a system that identifies suspicious orders of controlled substances (such as those of unusual size, those that deviate substantially from a normal pattern and those of unusual frequency) prior to completion of the sale. A compliant SOM system includes well-defined due diligence, "know your customer" efforts and order monitoring.

To meet its responsibilities, the DEA conducts periodic inspections of registered establishments that handle controlled substances. Annual registration is required for any facility that manufactures, tests, distributes, dispenses, imports or exports any controlled substance. The facilities must have the security, control and accounting mechanisms required by the DEA to prevent loss and diversion. Failure to maintain compliance, particularly as manifested in loss or diversion, can result in regulatory action that could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. The DEA may seek civil penalties, refuse to renew necessary registrations, or initiate proceedings to revoke those registrations. In certain circumstances, violations could lead to criminal proceedings.

Individual states also regulate controlled substances, and we, as well as our third-party API suppliers and manufacturers, are subject to such regulation by several states with respect to the manufacture and distribution of these products.

We and, to our knowledge, our third-party API suppliers, dosage form manufacturers, distributors and researchers have all necessary registrations, and we believe all registrants operate in conformity with applicable registration requirements, under controlled substance laws.

Government Benefit Programs. Statutory and regulatory requirements for Medicaid, Medicare, Tricare and other government healthcare programs govern provider reimbursement levels, including requiring that all pharmaceutical companies pay rebates to individual states based on a percentage of their net sales arising from Medicaid program-reimbursed products. The federal and/or state governments may continue to enact measures in the future aimed at containing or reducing payment levels for prescription pharmaceuticals paid for in whole or in part with government funds. We cannot predict the nature of such measures or their impact on our profitability and cash flows. These efforts could have material adverse consequences for the pharmaceutical industry as a whole and, consequently, also for us. However, we believe we have provided for our best estimate of potential refunds based on current information available.

From time to time, legislative changes are made to government healthcare programs that impact our business. For example, the Medicare Prescription Drug Improvement and Modernization Act of 2003 created a new prescription drug coverage program for people with Medicare through a new system of private market drug benefit plans. This law provides a prescription drug benefit to seniors and individuals with disabilities in the Medicare program ("Medicare Part D"). Congress continues to examine various Medicare policy proposals that may result in pressure on the prices of prescription drugs in the Medicare program.

In addition, the Healthcare Reform Act provides for major changes to the U.S. healthcare system. While some provisions of the Healthcare Reform Act have already taken effect, most of the provisions to expand access to healthcare coverage will not be implemented until 2014 and beyond. Since much of the implementation is yet to take place, there are still many challenges and uncertainties ahead. Such a comprehensive reform measure will require expanded implementation efforts on the part of federal and state agencies embarking on rule-making to develop the specific components of their new authority. We intend to monitor closely the implementation of the Healthcare Reform Act and related legislative and regulatory developments.

The Healthcare Reform Act will result in a transformation of the delivery and payment for healthcare services in the U.S. The combination of these measures is expected to expand health insurance coverage by an estimated 32 million people in the U.S. In addition, there are significant health insurance reforms in the U.S. that are expected to improve patients' ability to obtain and maintain health insurance. Such measures include: the elimination of lifetime caps and no rescission of policies or denial of coverage due to preexisting conditions.

Our estimate of the overall impact of the Healthcare Reform Act reflects a number of uncertainties. However, we believe that the impact to our business will be largely attributable to changes in the Medicare Part D coverage gap, the imposition of an annual fee on branded prescription pharmaceutical manufacturers, and increased rebates in the Medicaid Fee-For-Service Program and Medicaid Managed Care plans. There are a number of other provisions in the legislation that collectively are expected to have a small impact, including originator average manufacturers' price for new formulations and the expansion of 340B pricing to new entities. The various elements of the Healthcare Reform Act adversely impacted net sales by approximately \$12 million in fiscal 2012 and \$13 million in fiscal 2011.

### Healthcare Fraud and Abuse Laws

We are subject to various federal, state and local laws targeting fraud and abuse in the healthcare industry. For example, in the U.S., there are federal and state anti-kickback laws that prohibit the payment or receipt of kickbacks, bribes or other remuneration intended to induce the purchase or recommendation of healthcare products and services or reward past purchases or recommendations, including the U.S. Anti-Kickback Statute and similar state statutes, the U.S. Federal Sunshine Law and other parts of the Healthcare Reform Act, the False Claims Act and the Health Insurance Portability and Accountability Act of 1996. Violations of these laws can lead to civil and criminal penalties, including fines, imprisonment and exclusion from participation in federal healthcare programs. These laws are potentially applicable to us as both a manufacturer and a supplier of products reimbursed by federal healthcare programs. These laws also apply to hospitals, physicians and other potential purchasers of our products. In addition, some states in the U.S. have enacted compliance and reporting requirements aimed at drug manufacturers.

We are also subject to the FCPA and similar worldwide anti-bribery laws in non-U.S. jurisdictions which generally prohibit companies and their intermediaries from making improper payments to non-U.S. officials for the purpose of obtaining or retaining business. Because of the predominance of government-sponsored healthcare systems around the world, most of our customer relationships outside of the U.S. are with governmental entities and are therefore subject to such anti-bribery laws. Our policies mandate compliance with these anti-bribery laws. We operate in many parts of the world that have experienced governmental corruption to some degree, and in certain circumstances strict compliance with anti-bribery laws may conflict with local customs and practices. Despite our training and compliance programs, our internal control policies and procedures may not protect us from reckless or criminal acts committed by our employees or agents.

#### **Compliance** Programs

In order to systematically and comprehensively mitigate the risks of non-compliance with regulatory requirements described above, we believe we have developed a robust Compliance Program based on the April 2003 Office of the Inspector General ("OIG") Compliance Program Guidance for Pharmaceutical Manufacturers,

-71-

the U.S. Federal Sentencing Guidelines, the Pharmaceutical Research and Manufacturers of America Code on Interactions with Healthcare Professionals, the Code of Ethics of the Advanced Medical Technology Association, the United Kingdom Anti-Bribery guidance, and other relevant government guidances and national or regional industry codes of behavior. We conduct ongoing compliance training programs for all employees. We also maintain a 24-hour ethics and compliance reporting hotline.

As part of our Compliance Program, we have implemented internal cross-functional processes to review and approve all product-specific promotional materials, presentations and external communications to address the risk of misbranding or mislabeling our products through our promotional efforts. For example, we have established programs to monitor promotional speaker activities and field sales representatives. Specifically, we have developed a "ride along" program for field sales representatives similar to those included in recent Corporate Integrity Agreements from the OIG in order to obtain first-hand observations of how these approved materials are used.

We have also implemented a comprehensive controlled substances compliance program, including anti-diversion efforts that go beyond the DEA's SOM requirements. For example, we regularly assist federal, state and local law enforcement and prosecutors in the U.S. by providing information and testimony on our products and placebos for use by the DEA and other law enforcement agencies in investigations and at trial. As part of this program, we also work with some of our customers to help develop and implement what we believe are best practices for SOM and other anti-diversion activities.

We believe our Compliance Program design also addresses our FDA, healthcare anti-kickback and anti-fraud, and anti-bribery-related activities.

#### **Outside the United States**

Outside the U.S., we must comply with laws, guidelines and standards promulgated by other regulatory authorities that regulate the development, testing, manufacturing, marketing and selling of pharmaceuticals, including, but not limited to, Health Canada, the Medicines and Healthcare Products Regulatory Agency in the United Kingdom (U.K.), the Irish Medicines Board, the European Medicines Agency and member states of the E.U., the State Food and Drug Administration in China, the Therapeutic Goods Administration in Australia, the New Zealand Medicines and Medical Devices Safety Authority, the Ministry of Health and Welfare in Japan, the European Pharmacopoeia of the Council of Europe and the International Conference on Harmonization. Although international harmonization efforts continue, many laws, guidelines and standards differ by region or country.

We currently market our products in Canada, in various countries in the E.U., and in the Latin American and Asia-Pacific regions. The approval requirements and process vary by country, and the time required to obtain marketing authorization may vary from that required for FDA approval. Certain drug products and variations in drug product lines also must meet country-specific and other local regulatory requirements.

The examples below highlight some of the differences in the approval process in other regions or countries outside the U.S.:

*European Union*. Marketing authorizations are obtained either pursuant to a centralized or decentralized procedure. The centralized procedure, which provides for a single marketing authorization valid for all E.U. member states, is mandatory for the approval of certain drug products and is optional for novel drug products that are in the interest of patient health. Under the centralized procedure, a single marketing authorization application is submitted for review to the European Medicines Agency, which makes a recommendation on the application to the European Commission. The final determination as to whether or not to approve the application rests with the European Commission. The decentralized procedure provides for concurrent mutual recognition of national approval decisions and is available for products that are not subject to the centralized procedure.

-72-

The E.U. has also adopted directives and other laws that govern the labeling, marketing, advertising, supply, distribution, and drug safety monitoring and reporting of drug products. Such directives set regulatory standards throughout the E.U. and permit member states to supplement such standards with additional requirements.

European governments also regulate drug prices through the control of national healthcare systems that fund a large part of such costs to patients. As a result, patients are unlikely to take a drug product that is not reimbursed by their government. Many European governments regulate the pricing of a new drug product at launch through direct price controls or reference pricing. Recently, many individual countries also have imposed additional cost-containment measures on drug products. Such differences in national pricing regimes may create price differentials between E.U. member states. Many European governments also advocate generic substitution by requiring or permitting prescribers or pharmacists to substitute a different company's generic version of a brand drug product that was prescribed.

*Japan*. The Pharmaceutical and Medical Devices Agency ("PMDA") is responsible for reviewing marketing authorizations of drug products. The PMDA may require bridging studies (a clinical trial with a smaller sub-population than the original clinical trials) to demonstrate that clinical trial data obtained in trials conducted outside of Japan are applicable to Japanese patients. After completing a comprehensive review, the PMDA reports its findings to the Ministry of Health, Labour and Welfare, which either approves or denies the application.

Japan's national health insurance system maintains a Drug Price List that specifies which drug products are eligible for reimbursement and the Ministry of Health, Labour and Welfare sets pricing for such drug products. In general, the Japanese government introduces a round of price cuts every other year and mandates price reductions for specific drug products. However, new drug products that are judged innovative or useful, indicated for pediatric use, or target orphan diseases may be eligible for premium prices. In addition, the Japanese government also has advocated the prescribing and use of generic drugs, where available.

*Emerging Markets*. Many emerging markets continue to evolve their regulatory review and oversight processes. At present, such countries typically require prior regulatory approval or marketing authorization from large, developed markets (such as the U.S.) before they will initiate or complete their review. Some countries also require the applicant to conduct local clinical trials as a condition of marketing authorization.

Many emerging markets continue to implement measures to control drug product prices, such as implementing direct price controls or advocating the prescribing and use of generic drugs.

#### **Raw Materials**

We contract with various third-party manufacturers and suppliers to provide us with raw materials used in our products, finished goods and certain services.

The active ingredients in the majority of our current pharmaceutical products and products in development, including oxycodone, oxymorphone, morphine, fentanyl, methylphenidate and hydrocodone, are listed by the DEA as Schedule II or III substances under the CSA. Consequently, their manufacture, shipment, storage, sale and use are subject to a high degree of regulation.

Furthermore, the DEA limits the availability of the active ingredients used in many of our current products and products in development, as well as the production of these products. As discussed in "—Drug Enforcement Administration," we must annually apply to the DEA for procurement and production quotas in order to obtain and produce these substances. Moreover, the DEA has complete discretion to adjust these quotas from time to time during the calendar year. As a result, our procurement and production quotas may not be sufficient to meet commercial demand or to conduct bioequivalence studies and clinical trials. Any delay or refusal by the DEA in granting, in whole or in part, our quota requests for controlled substances could delay or result in the stoppage of the manufacture of our pharmaceutical products, our clinical trials or product launches and could require us to allocate product among our customers, all of which could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

-73-

Our radiopharmaceutical product offering includes "hot" radioisotopes including Mo-99, a critical ingredient of our Ultra-Technekow DTE Tc-99m generators. Mo-99 is produced in nuclear research reactors utilizing HEU or LEU targets. These targets, either tubular or flat and of varying sizes, are fabricated from HEU or LEU and, in either case, aluminum. The targets are placed in or near the core of the nuclear reactor where fission reactions occur resulting in the production of Mo-99 and other isotopes. This process, which takes approximately six days, is known as target irradiation. There are currently eight reactors around the world producing the global supply of Mo-99. We have agreements to obtain Mo-99 from three of these reactors and we rely predominantly on two of these reactors for our Mo-99 supply. These reactors are subject to scheduled and unscheduled shutdowns which can have a significant impact on the amount of Mo-99 available for processing. Mo-99 produced at these reactors is then finished at one of five processing sites located throughout the world, including our processing facility located in the Netherlands. At the processing facility, the targets are dissolved and chemically separated. In this process, the Mo-99 is isolated as a radiochemical. We transport finished Mo-99 from our processing facility in the Netherlands to our facility in Maryland Heights, Missouri, where it, together with Mo-99 received from other third-party processors, is loaded into our Tc-99m generators. Mo-99 has a 66 hour half-life and degrades into, among other things, Tc-99m, which has a half-life of only six hours. The radiopharmacies or hospitals prepare dosages from the Tc-99m generators for use in SPECT imaging medical procedures.

If, for any reason, we are unable to obtain sufficient quantities of any of the finished goods or raw materials or components required for our products, it could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

### Long-Lived and Total Assets

Our long-lived assets, which are primarily composed of property, plant and equipment, by geographic area are set forth below:

		Fiscal	
(Dollars in Millions)	2012	2011	2010
U.S.	\$847.7	\$802.0	\$802.9
Europe, Middle East and Africa (including \$45.5, \$48.9 and \$49.6 in Ireland)	72.2	81.3	85.6
Other	52.1	48.1	50.6
	\$972.0	\$931.4	\$939.1

Our total assets by segment are as follows:

		Fiscal	
(Dollars in Millions)	2012	2011	2010
Specialty Pharmaceuticals	\$1,547.3	\$1,458.5	\$1,477.3
Global Medical Imaging	1,085.7	1,103.6	1,144.9
Corporate <sup>(1)</sup>	241.6	261.3	266.1
	\$2,874.6	\$2,823.4	\$2,888.3

<sup>(1)</sup> Consists of assets used in managing our total business and not allocated to any one segment.

#### Environmental

We are subject to numerous federal, state, local and non-U.S. environmental protection and health and safety laws and regulations. Our operations, like those of other pharmaceutical companies, involve the use of substances regulated under environmental laws, primarily in manufacturing processes. We cannot assure you that we have been or will be in full compliance with environmental and health and safety laws and regulations at all

## -74-

times. If we violate these laws, we could be fined, criminally charged or otherwise sanctioned by regulators. We believe that our operations currently comply in all material respects with applicable environmental laws and regulations.

Certain environmental laws assess strict (i.e., regardless of fault) and joint and several liability on current or previous owners of real property and current or previous owners or operators of facilities for the costs of investigation, removal or remediation of hazardous substances or materials at such properties or at properties at which parties have disposed of hazardous substances. In addition to cleanup actions brought by governmental authorities, private parties could bring personal injury or other claims due to the presence of, or exposure to, hazardous substances.

In addition, from time to time, we have received notification from the EPA and from state environmental agencies in the U.S. that conditions at a number of sites where we and others disposed of hazardous substances require investigation, cleanup and other possible remedial actions. These agencies may require that we reimburse the government for costs incurred at these sites or otherwise pay for the cost of investigation and cleanup of these sites including compensation for damage to natural resources. We have projects underway at a number of current and former manufacturing facilities to investigate and remediate environmental contamination resulting from past operations. These projects relate to a variety of activities, including decontamination and decommissioning of radioactive materials and investigation and removal of hazardous substances from soil and groundwater. These projects involve both investigation and remediation expenses and capital expenditures.

We provide for expenses associated with environmental remediation obligations once we determine that a potential environmental liability at a particular site is probable and the amount can be reasonably estimated. We regularly assess current information and developments as the investigations and remediation activities proceed and adjust accruals, as necessary, to provide for the expected impact of these environmental matters.

The ultimate cost of investigation and cleanup at disposal sites and manufacturing facilities is difficult to predict given uncertainties regarding the extent of the required cleanup, the interpretation of applicable laws and regulations and alternative cleanup methods. Based upon our experience, current information available and applicable laws, we believe that it is probable that we will incur investigation and remedial costs, including asset retirement obligations, of \$192.2 million, of which \$10.2 million is included in accrued and other current liabilities, \$134.4 million is included in environmental liabilities and \$47.6 million is included in other liabilities on our combined balance sheet at March 29, 2013. This amount includes \$94.7 million relating to a site located in Orrington, Maine which will be a liability of a Covidien entity following the separation. For more information on our pro forma adjustments, see "Unaudited Pro Forma Condensed Combined Financial Statements." Note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement provide additional information regarding environmental matters, including asset retirement obligations. All accruals have been recorded without giving effect to any possible future insurance proceeds.

Environmental laws are complex, change frequently and generally have become more stringent over time. While we have planned for future capital and operating expenditures to comply with these laws and to address liabilities arising from past or future releases of, or exposures to, hazardous substances, we cannot assure you that our costs of complying with current or future environmental protection, health and safety laws and regulations will not exceed our estimates or have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows. Further, we cannot assure you that we will not be subject to additional environmental claims for personal injury or cleanup in the future based on our past, present or future business activities. While it is not feasible to predict the outcome of all pending environmental matters, it is reasonably probable that there will be a need for future provisions for environmental costs that, in management's opinion, are not likely to have a material adverse effect on our financial condition, but could be material to the results of operations in any one accounting period.

-75-

Certain radiological licenses at certain manufacturing sites owned by us require the establishment of decommissioning programs which will require remediation in accordance with regulatory requirements upon cessation of operations at these sites.

#### Manufacturing

We have 10 manufacturing sites, including seven located in the U.S., as well as sites in Canada, Ireland and the Netherlands, which handle production, assembly, quality assurance testing, packaging and sterilization of our products. We estimate that our manufacturing production by region in fiscal 2012 (as measured by cost of production) was approximately: U.S.–78%, Europe, Middle East and Africa–14% and Other–8%.

### Sales, Marketing and Distribution

We maintain distribution centers in over 20 countries. Products generally are delivered to these distribution centers from our manufacturing facilities and then subsequently delivered to the customer. In some instances, product, such as nuclear medicine, is delivered directly from our manufacturing facility to the customer. We contract with a wide range of transport providers to deliver our products by road, rail, sea and air.

### Customers

We market our branded and generic products and CMDS to physicians, pharmacists, pharmacy buyers, radiologists and radiology technicians. We distribute these products to major drug wholesalers, retail pharmacy chains, hospital networks and governmental agencies. In addition, we contract with GPOs and managed care organizations to improve access to our products.

We utilize our API to manufacture our own products. In addition, we sell and distribute API directly or through distributors to other pharmaceutical companies. In the U.S., we market and distribute our nuclear imaging products to radiopharmacies which, in turn, supply hospitals and standalone imaging centers with patient-customized doses. Outside the U.S., we market and distribute our nuclear imaging products to hospitals.

We often negotiate with parties that enter into supply contracts for the benefit of their member facilities, including GPOs, IDNs, large and medium size retail pharmacy chains, nuclear pharmacy chains, wholesalers, and solely outside the U.S., with governments through a tender process.

Cardinal Health, a distributor, represented 19%, 19% and 15% of our net sales in fiscal 2012, 2011 and 2010, respectively. McKesson, also a distributor, represented 14%, 13% and 11% of our net sales in fiscal 2012, 2011 and 2010, respectively. AmerisourceBergen Corporation, also a distributor, represented 9%, 10% and 8% of our net sales in fiscal 2012, 2011 and 2010, respectively. No other customer accounted for 10% or more of our net sales in the past three fiscal years.

-76-

Net sales by geographic area, based on the location of the entity that records the transaction, are shown in the following table:

		Fiscal	
(Dollars in Millions)	2012	2011	2010
U.S.:			
Specialty Pharmaceuticals	\$ 880.6	\$ 784.8	\$ 756.3
Global Medical Imaging	466.8	505.8	621.5
	1,347.4	1,290.6	1,377.8
Europe, Middle East and Africa <sup>(1)</sup> :			
Specialty Pharmaceuticals	108.7	93.4	89.7
Global Medical Imaging	302.3	326.3	304.1
	411.0	419.7	393.8
Other:			
Specialty Pharmaceuticals	15.9	31.2	23.0
Global Medical Imaging	227.7	227.9	202.5
	243.6	259.1	225.5
Total:			
Specialty Pharmaceuticals	1,005.2	909.4	869.0
Global Medical Imaging	996.8	1,060.0	1,128.1
Net sales of operating segments	2,002.0	1,969.4	1,997.1
Net sales to related parties <sup>(2)</sup> :	54.2	52.4	50.5
	\$2,056.2	\$2,021.8	\$2,047.6

<sup>(1)</sup> There were no sales recorded in Ireland.

<sup>(2)</sup> Represents products that were sold to other Covidien businesses.

## Backlog

At September 28, 2012, the backlog of firm orders was less than 1% of net sales. We anticipate that substantially all of the backlog as of September 28, 2012 will be shipped during fiscal 2013.

## Seasonality

There are no significant seasonal aspects to our business; however, DEA quotas are allocated in each calendar year to companies and may impact our sales until the DEA grants additional quotas, if any.

### Employees

At September 28, 2012, we had approximately 5,300 employees, approximately 4,800 of which are based in the U.S.

## Properties

Our offices in the U.S. are located in a facility in Hazelwood, Missouri, which we own. As of September 28, 2012, we owned a total of 33 facilities in nine countries. Our owned facilities consist of approximately 2.7 million square feet, and our leased facilities consist of approximately 0.7 million square feet. We have ten manufacturing sites, six of which are used by our Global Medical Imaging segment, three of which are used by our Specialty Pharmaceuticals segment and one of which is shared by both segments. We have a manufacturing site in each of Canada, Ireland and the Netherlands and seven manufacturing sites in the U.S. We believe all of these facilities are well-maintained and suitable for the operations conducted in them.

-77-

#### Legal Proceedings

We are subject to various legal proceedings and claims, including patent infringement claims, product liability matters, environmental matters, employment disputes, contractual disputes and other commercial disputes, including those described below and in note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement. We believe that these legal proceedings and claims likely will be resolved over an extended period of time. Although it is not feasible to predict the outcome of these matters, management is of the opinion that their ultimate resolution should not have a material adverse effect on our competitive position, business, financial condition, cash flows or results of operations.

#### **Governmental Proceedings**

On January 7, 2009, we received a subpoena from the U.S. Attorney's Office for the Northern District of California requesting production of documents relating to the sales and marketing of our Tofranil-PM, Restoril and Magnacet products. We are complying as required by the terms of the subpoena.

On November 30, 2011 and October 22, 2012, we received subpoenas from the DEA requesting production of documents relating to our SOM program. We are complying as required by the terms of the subpoenas.

### Patent/Antitrust Litigation

*Tyco Healthcare Group LP, et al. v. Mutual Pharmaceutical Company, Inc.* We filed a patent infringement suit in the U.S. District Court for the District of New Jersey against Mutual Pharmaceutical Co., Inc., et al. on March 20, 2007 pursuant to procedures set out in the Drug Price Competition and Patent Term Restoration Act of 1984, after Mutual submitted an ANDA to the FDA seeking to sell a generic version of our 7.5 mg Restoril sleep aid product. Mutual also filed antitrust and unfair competition counterclaims. The patents at issue have since expired or been found invalid. On January 18, 2013, the trial court issued an opinion and order granting our motion for summary judgment regarding Mutual's antitrust and unfair competition counterclaims. On May 1, 2013, Mutual appealed this decision to the U.S. Court of Appeals for the Federal Circuit.

#### **Pricing Litigation**

Two cases are pending against us that allege generally that we and numerous other pharmaceuticals companies reported false pricing information in connection with certain drugs that are reimbursable under Medicaid, resulting in overpayment by state Medicaid programs for those drugs. These cases, brought by state Attorneys General in Utah and Louisiana, generally seek monetary damages and attorneys' fees. We are named as a defendant in *State of Utah v. Actavis US, Inc., et al.*, filed May 8, 2008, which is pending in the Third Judicial Circuit of Salt Lake County, Utah and in *State of Louisiana v. Abbott Laboratories Inc., et al.*, filed November 3, 2010, which was pending in the 19th Judicial District, Parish of East Baton Rouge, Louisiana. In May 2013, we agreed to terms of settlement with the Attorney General for the state of Louisiana resolving all claims in *State of Louisiana v. Abbott Laboratories Inc., et al.*, involving alleged reporting of false pricing information in connection with certain drugs that are reimbursable under Medicaid, resulting in overpayment by the state Medicaid program for those drugs. We intend to contest the Utah case and explore other options as appropriate.

#### **Environmental Remediation and Litigation Proceedings**

We are involved in various stages of investigation and cleanup related to environmental remediation matters at a number of sites, including those described below.

Crab Orchard National Wildlife Refuge Superfund Site, near Marion, Illinois. Mallinckrodt US LLC, an entity included in our combined financial statements, is a successor in interest to International Minerals and Chemicals Corporation ("IMC"). Between 1967 and 1982, IMC leased portions of the Additional and

-78-

Uncharacterized Sites ("AUS") Operable Unit at the Crab Orchard Superfund Site (the "Site") from the government and manufactured various explosives for use in mining and other operations. In March 2002, the DOJ, the U.S. Department of the Interior and the EPA (together, the "Government Agencies") issued a special notice letter to General Dynamics Ordnance and Tactical Systems, Inc. ("General Dynamics"), one of the other potentially responsible parties ("PRPs") at the Site, to compel General Dynamics to perform the remedial investigation and feasibility study ("RI/FS") for the AUS Operable Unit. General Dynamics negotiated an Administrative Order on Consent with the Government Agencies to conduct an extensive RI/FS at the Site under the direction of the U.S. Fish and Wildlife Service. General Dynamics asserted in August 2004 that Mallinckrodt US LLC is jointly and severally liable, along with approximately eight other lessees and operators at the AUS Operable Unit, for alleged contamination of soils and groundwater resulting from historic operations and has threatened to file a contribution claim against Mallinckrodt US LLC and other parties for recovery of its costs incurred in connection with the RI/FS activities being conducted at the AUS Operable Unit. Mallinckrodt US LLC and other PRPs who received demand letters from General Dynamics have explored settlement alternatives, but have not reached settlement to date. Mallinckrodt US LLC and other PRPs are awaiting completion of the RI/FS by General Dynamics before the initiation of formal PRP negotiations to address resolution of these alleged claims.

*Mallinckrodt Veterinary, Inc., Millsboro, Delaware.* Mallinckrodt Veterinary, Inc. ("MVI"), an entity included in our combined financial statements, previously operated a plant in Millsboro, Delaware (the "Millsboro Site") that manufactured various animal healthcare products. In 2005, the Delaware Department of Natural Resources and Environmental Control found trichloroethylene ("TCE") in the Millsboro public water supply at levels that exceeded the federal drinking water standards. Further investigation to identify the TCE plume in the ground water indicated that the plume has extended to property owned by a third party near the Millsboro Site. We and other former owners assumed responsibility for the Millsboro Site cleanup under the Alternative Superfund Program administered by the EPA. We and other PRPs entered into an Administrative Order on Consent with the EPA on May 10, 2010 which was subsequently amended in November 2010 and January 2011 to investigate the potential source of TCE contamination and to evaluate options to abate, mitigate and/or eliminate the release or threat of release of hazardous substances at the Millsboro Site. We, along with other parties, continue to conduct the studies and prepare remediation plans in accordance with the amended Administrative Order on Consent.

Coldwater Creek, St. Louis County, Missouri. Mallinckrodt is one of several companies named as defendants in four tort complaints (*McClurg, et al. v. MI* Holdings, Inc., et al., filed February 28, 2012; Adams, et al. v. MI Holdings, Inc., et al., filed April 10, 2012, Steinman v. MI Holdings, Inc., et al., filed October 23, 2012 and Schneider, et al. v. MI Holdings, Inc., et al., filed April 19, 2013) with numerous plaintiffs pending in the U.S. District Court for the Eastern District of Missouri. These cases allege personal injury for alleged exposure to radiological substances present in Coldwater Creek in Missouri. Plaintiffs lived in various locations in St. Louis County, Missouri near Coldwater Creek. Radiological residues which may have been present in the creek have been remediated by the U.S. Army Corps of Engineers. We believe that we have meritorious defenses to these complaints and are vigorously defending against them.

Orrington, Maine and Penobscot River and Bay. Note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement provide information regarding investigation and remediation of a site located in Orrington, Maine and the lawsuit styled Maine People's Alliance and Natural Resources Defense Council, Inc. v. HoltraChem Manufacturing Company, LLC and Mallinckrodt US LLC regarding an investigation being conducted in the Penobscot River and Bay. The liability for such remediation has been included in our combined financial statements since the liability had historically been included in the Pharmaceuticals business of Covidien as it was historically managed as part of Covidien and its subsidiaries prior to completion of the separation. However, the entity with the liability for such investigation and remediation will not be transferred to Mallinckrodt as part of the separation. Accordingly, this will be a liability of a Covidien entity following the separation.

-79-

## **Products Liability Litigation**

We are one of four manufacturers of GBCAs, such as our Optimark product, involved in litigation alleging that administration of these agents causes development of nephrogenic systemic fibrosis in a small number of patients with advanced renal impairment. In May 2013, we agreed to terms of settlement with the plaintiffs in four previously disclosed lawsuits involving our Optimark product. These settlements resolved cases that were included in federal multi-district litigation pending in the U.S. District Court for the Northern District of Ohio (*In re Gadolinium-Based Contrast Agents Product Liability Litigation*, which was established on February 27, 2008) and cases in various state courts.

Beginning with lawsuits brought in July 1976, we have also been named as a defendant in personal injury lawsuits based on alleged exposure to asbestoscontaining materials. A majority of the cases involve product liability claims, based principally on allegations of past distribution of products containing asbestos. A limited number of the cases allege premises liability, based on claims that individuals were exposed to asbestos while on our property. Each case typically names dozens of corporate defendants in addition to us. The complaints generally seek monetary damages for personal injury or bodily injury resulting from alleged exposure to products containing asbestos. Our involvement in asbestos cases has been limited because we did not mine or produce asbestos. Furthermore, in our experience, a large percentage of these claims have never been substantiated and have been dismissed by the courts. We have not suffered an adverse verdict in a trial court proceeding related to asbestos claims, and intend to continue to defend these lawsuits. When appropriate, we settle claims; however, amounts paid to settle and defend all asbestos claims have been immaterial to date, and are not expected to be material in the future. As of April 30, 2013, there were approximately 11,600 asbestos-related cases pending against us.

#### **Other Matters**

We are a defendant in a number of other pending legal proceedings incidental to present and former operations, acquisitions and dispositions. We do not expect the outcome of these proceedings, either individually or in the aggregate, to have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

## MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following information should be read in conjunction with our audited and unaudited combined financial statements and accompanying notes, and our unaudited pro forma combined financial statements and accompanying notes included elsewhere in this information statement. The following discussion contains forward-looking statements that involve risks, uncertainties and assumptions. Our actual results could differ materially from those discussed in these forward-looking statements as a result of many factors, including, but not limited to, those discussed under headings "Risk Factors" and "Cautionary Statement Concerning Forward-Looking Statements."

#### Separation from Covidien

In December 2011, Covidien announced a plan to spin off its Pharmaceuticals business into a separate, publicly traded company. Upon separation, Mallinckrodt plc will be the parent company which will own the Pharmaceuticals business. The Pharmaceuticals business of Covidien, presented herein, represents a combined reporting entity comprising the assets and liabilities used in managing and operating Covidien's Pharmaceuticals business, including subsidiaries, branches and operations that have been carved out that relate to Covidien's Pharmaceuticals business. Certain subsidiaries have disposed of some of the operations previously owned. Where appropriate, these operations have been reflected as discontinued operations in our combined financial statements. Divestitures of product lines not representing businesses have been reflected in operating income.

Our combined financial statements have been prepared on a standalone basis in U.S. dollars, in accordance with GAAP and reflect our business as it was historically managed as part of Covidien and its subsidiaries prior to completion of the separation. These combined financial statements may not be indicative of our future performance and do not necessarily reflect what our combined results of operations, financial condition and cash flows would have been had we operated as a separate, publicly traded company during the periods presented, particularly since many changes will occur in our operations and capitalization as a result of our separation from Covidien.

Our combined financial statements include expense allocations for certain functions provided by Covidien, including, but not limited to, general corporate expenses related to finance, legal, information technology, human resources, communications, employee benefits and incentives, insurance and stock-based compensation. Management believes such allocations are reasonable; however, they may not be indicative of the actual expenses we would have incurred had we been operating as a separate, publicly traded company for the periods presented. Note 1 to our interim unaudited condensed combined financial statements and note 1 to our annual combined financial statements provide further information regarding allocated expenses. Following the separation, we will perform these functions using our own resources or purchased services. For an interim period, however, some of these functions will continue to be provided by Covidien under a transition services agreement, particularly in relation to areas outside the U.S. The terms and prices on which such services are rendered may differ from the terms and prices in effect prior to completion of the separation. We also may incur additional costs associated with being a separate, publicly traded company. These additional anticipated costs are not reflected in our historical combined financial statements. On an annual basis, we estimate these operating costs will be approximately \$40 million higher than the general corporate expenses historically allocated from Covidien to us.

### Overview

We are a global company that develops, manufactures, markets and distributes both branded and generic specialty pharmaceuticals, API and diagnostic imaging agents. Our products are found in almost every hospital, standalone diagnostic imaging center or pharmacy in the U.S. and we have a sales presence in approximately 50 countries. We believe our extensive commercial reach and chemistry expertise, coupled with our ability to

deal with the highly regulated and technical nature of our business, have created compelling competitive advantages that we anticipate will sustain future revenue growth. We expect our investments in operating improvements to lead to cost efficiencies and continued margin expansion.

We operate our business through two segments:

- Specialty Pharmaceuticals produces and markets Brands, Generics and API; and
- Global Medical Imaging develops, manufactures and markets CMDS and Nuclear Imaging products.

## **Healthcare Reform**

In 2010, the Healthcare Reform Act was enacted into law in the U.S. This legislation imposes a \$28 billion fee on the branded pharmaceutical industry over nine years starting in 2011 and a \$2.8 billion annual fee on the branded pharmaceutical industry thereafter. The amount of the fee payable by each company is based upon market share. Our share of the fee was not significant in fiscal 2012 and 2011. In addition, beginning in 2011, the law requires pharmaceutical manufacturers to pay a 50% discount to Medicare Part D beneficiaries when they are in the Medicare Part D coverage gap (also known as the "doughnut hole"). The impact of this provision on both fiscal 2012 and fiscal 2011 net sales was insignificant. The law also increased mandated Medicaid rebates, which reduced net sales by \$11.2 million and \$13.1 million in fiscal 2012 and 2011, respectively.

The legislation also includes a provision that imposes a 2.3% excise tax on the sale of certain medical devices by a manufacturer, producer or importer of such devices in the U.S. starting after December 31, 2012. This assessment did not have a significant impact on our results of operations.

## **Product Launches**

On December 28, 2012, we received approval from the FDA to manufacture a generic version of CONCERTA (Methylphenidate HCl) extended-release Tablets USP for the treatment of ADHD in 27 mg, 36 mg and 54 mg dosages. We believe we hold a 180-day exclusivity period for each of the 27 mg, 36 mg and 54 mg dosage strengths, which begins upon the commercial launch of each dosage strength. We launched the 27 mg dosage strength upon FDA approval during the first quarter of fiscal 2013 and launched the 36 mg and 54 mg dosage strengths during the second quarter of fiscal 2013. In February 2013, we submitted a supplement to our approved ANDA for an 18 mg dosage strength. Sales of Methylphenidate HCl products were \$70.9 million during the first six months of fiscal 2013. While sales of these products are subject to our receipt of sufficient quota from the DEA, we currently expect sales of Methylphenidate HCl products to be at least \$125 million in fiscal 2013. However, sales of these products may subsequently decline in fiscal 2014, depending on a number of factors, including expiration of the exclusivity period.

#### Acquisitions

In October 2012, we acquired CNS Therapeutics, a specialty pharmaceutical company focused on developing and commercializing products for sitespecific administration to the central nervous system to treat neurological disorders and intractable chronic pain, for total consideration of \$95.0 million. The total consideration was comprised of an upfront cash payment of \$88.1 million (net of cash acquired) and the fair value of contingent consideration of \$6.9 million. This contingent consideration, which could potentially total a maximum of \$9.0 million, is primarily based on whether the FDA approves another dosage form of Gablofen on or before December 31, 2016. The acquisition of CNS Therapeutics expanded our branded pharmaceuticals portfolio and supports our strategy of leveraging our therapeutic expertise and core capabilities in manufacturing, regulatory and commercialization to serve patients.

In August 2012, we paid \$13.2 million under an agreement to acquire all of the rights to Roxicodone<sup>®</sup> from Xanodyne Pharmaceuticals, Inc., which was capitalized as an intangible asset. Roxicodone is an immediate-

release oral formulation of oxycodone hydrochloride indicated for the management of moderate to severe pain where the use of an opioid analgesic is appropriate. Roxicodone is the RLD for one of our generic products and is important to our product pipeline. There are no ongoing royalty payments under this agreement.

In June 2009, we acquired the rights to market and distribute the pain management drug Exalgo in the U.S. for an upfront cash payment of \$10.0 million, which was included in R&D expenses during fiscal 2009. Under the license arrangement, we are obligated to make additional payments up to \$73.0 million based upon the successful completion of specified development and regulatory milestones. During fiscal 2009, \$10.0 million of such milestone payments were made and included in R&D expenses. During fiscal 2010, the FDA approved the Exalgo NDA for the 8 mg, 12 mg and 16 mg tablet dosage forms, resulting in additional payments of \$55.0 million, which were capitalized as an intangible asset. In addition, during fiscal 2012, we received FDA approval to market a 32 mg tablet dosage form. We are also required to pay royalties on sales of the product. During fiscal 2012, 2011 and 2010, we paid royalties of \$16.1 million, \$5.5 million and \$4.4 million, respectively. In addition, during the first six months of fiscal 2013 and 2012, we paid royalties of \$11.4 million and \$6.7 million, respectively.

## License Agreements

In October 2009, we licensed worldwide rights to utilize Depomed's Acuform gastric retentive drug delivery technology for the exclusive development of four products. Under this license agreement, we paid Depomed upfront and development payments of \$5.3 million during fiscal 2009. In addition to these payments, we may be obligated to pay up to \$64 million in additional development milestone payments. We will also pay Depomed a royalty on sales of products developed under this license agreement. During fiscal 2012 and 2010, an insignificant amount of milestone payments were expensed as incurred since regulatory approval had not yet been received. No milestone payments were made in fiscal 2011. In addition, no royalties have been paid through the first six months of fiscal 2013.

In June 2009, we entered into a license agreement which granted us rights to market and distribute PENNSAID® (diclofenac sodium topical solution) 1.5% w/w ("Pennsaid") and MNK-395, product candidates for the treatment of osteoarthritis for the knee(s). This license arrangement included an upfront cash payment of \$10.0 million, which was included in R&D expenses during fiscal 2009. We are also responsible for all future development activities and expenses. In addition, we may be required to make additional payments up to \$120 million based upon the successful completion of specified regulatory and sales milestones, and are required to pay royalties on sales of the products. During fiscal 2010, upon FDA approval of the Pennsaid NDA, we made a milestone payment of \$15.0 million, which was capitalized as an intangible asset. During fiscal 2012, we paid royalties of \$7.5 million associated with this product. The amount of royalties paid during fiscal 2011 and 2010 were insignificant. Royalties paid during the first six months of fiscal 2013 and 2012 were \$1.7 million and \$2.6 million, respectively. We submitted an NDA for MNK-395 in June 2012. In March 2013, the FDA requested additional information before the application can be considered for approval. In order to comply with this request, we are in the process of repeating a pharmacokinetic study. We anticipate that we will be able to submit the results from this study to the FDA in the third quarter of calendar 2013.

### Divestitures

During fiscal 2011, we sold the rights to market TussiCaps<sup>TM</sup>, which are hydrocodone bitartrate and chlorpheniramine maleate extended-release capsules for use as a cough suppressant, for an upfront cash payment of \$11.5 million. As a result of this transaction, we recorded an \$11.1 million gain. The purchaser also may be obligated to make contingent payments to us of up to \$11.5 million from December 31, 2011 through September 30, 2015, payable in equal quarterly installments until such time as a new competitive generic product is introduced into the market. In addition, we would receive a \$1.0 million contingent payment if certain sales targets are achieved over the same time period. We received \$2.9 million of contingent payments during fiscal 2012 and an additional \$1.4 million during the first six months of fiscal 2013.

During fiscal 2010, we sold our nuclear radiopharmacies in the U.S. for net cash proceeds of \$13.0 million. As a result of this transaction, we recorded a \$3.9 million net gain. In connection with this sale, we also entered into a supply agreement, under which the purchaser committed to annual purchase volumes through December 31, 2014.

## **Nuclear Imaging**

In November 2012, one of the research reactors we use to irradiate targets as part of our Mo-99 processing operation experienced an unscheduled shutdown. While we have been able to receive increased target irradiations at two other reactors and have purchased additional Mo-99 from other sources to continue meeting customer orders in the near term, the additional Mo-99 we are procuring from alternative sources comes at a higher than normal cost. While we expect the reactor to resume production in June 2013, should this shutdown overlap the time period during which another reactor is planned to shut down for routine maintenance, there may be an impact on the amount of available Mo-99, which could result in global shortages, continued increased raw material costs and decreased sales. We will continue to work closely with reactor operators and other processors to provide maximum available coverage to meet our customer needs.

## **Business Factors Influencing the Results of Operations**

### Fiscal Year

We report our results based on a "52-53 week" year ending on the last Friday of September. Fiscal 2012 and 2010 consisted of 52 weeks and ended on September 28, 2012 and September 24, 2010, respectively. Fiscal 2011 ended on September 30, 2011 and consisted of 53 weeks.

#### New Products

In March 2010, Exalgo extended-release tablets (8 mg, 12 mg and 16 mg) were approved by the FDA for the treatment of moderate to severe pain in opioid-tolerant patients requiring continuous around-the-clock opioid analgesia for an extended amount of time. We launched these three tablet strengths of Exalgo in late April 2010. Beginning in November 2013, a third party will have the right, pursuant to an agreement with us, to sell Exalgo tablets in the 8 mg, 12 mg and 16 mg dosages. In addition, our patents for these dosages expire in July 2014.

In August 2012, the FDA approved a 32 mg tablet of Exalgo which will further expand the patient population that Exalgo can effectively treat with a single daily dose. Exalgo was granted marketing exclusivity in the U.S. as a prescription medicine through March 2013 and is protected by two Orange Book-listed patents for a method of treating moderate to severe pain. Beginning in May 2014, a third party will have the right, pursuant to an agreement with us, to sell Exalgo tablets in the 32 mg dosage strength.

Sales of Exalgo were \$91.9 million in fiscal 2012, which we expect to increase in fiscal 2013. In addition, we expect sales of Exalgo to decrease in fiscal 2014 (compared with fiscal 2013) when a third party enters the market pursuant to the agreement referred to above.

We launched Pennsaid into the U.S. market in late April 2010. Pennsaid was granted marketing exclusivity in the U.S. as a prescription medicine until November 2012 and is protected by an Orange Book-listed patent for the method of use of topical diclofenac on the knee and a second topical medication on the same knee which expires in July 2029.

In February 2010, we launched an oral transmucosal fentanyl citrate ("OTFC") in the U.S. market, which is offered in 200, 400, 600, 800, 1,200 and 1,600 micrograms. OTFC is a generic alternative to the branded ACTIQ<sup>®</sup>, a trademark of Cephalon, Inc. or its affiliates.

In February 2011, we launched a fentanyl transdermal system ("FTS") patch in the U.S. market, which is offered in 25 mcg/hr, 50 mcg/hr, 75 mcg/hr and 100 mcg/hr strengths. It is a transdermal formulation of fentanyl

-84-

that is delivered slowly into the body through a patch worn on the skin. FTS is a generic alternative to the branded Duragesic® patch, a trademark of Johnson & Johnson or its affiliates.

Net sales of new products discussed above were \$101.9 million and \$90.3 million during the first six months of fiscal 2013 and 2012, respectively, and were \$191.6 million, \$114.5 million and \$42.3 million in fiscal 2012, 2011 and 2010, respectively.

### **Restructuring Initiatives**

We continue to look for opportunities to improve our cost structure and achieve operating excellence and efficiencies. Our recent initiatives have primarily been part of the 2011 restructuring program. We launched an initiative that closed a manufacturing facility in Chesterfield, U.K. The manufacturing facility produced API products and we transferred these processes to another manufacturing site creating operating and logistic efficiencies. In addition, we announced a comprehensive initiative to renovate, upgrade and modernize key manufacturing operations at our St. Louis manufacturing facility. We began to realize benefits from this initiative in fiscal 2012. During the first six months of fiscal 2013 and 2012, we incurred net restructuring and related costs of \$7.9 million and \$10.9 million, respectively, which include accelerated depreciation costs of \$1.3 million and \$5.4 million during the first six months of fiscal 2012, 2011 and 2010, we incurred net restructuring and related costs of \$19.2 million, \$10.0 million and \$11.5 million, respectively, which include accelerated depreciation costs of \$8.0 million and \$1.6 million during fiscal 2012 and 2011, respectively. The restructuring charges incurred during all of these periods primarily related to severance and employee benefit costs.

### **Research and Development Investment**

During the first six months of fiscal 2013, R&D expenses increased \$5.3 million, compared with the first six months of fiscal 2012. In addition, R&D expenses increased \$22.4 million in fiscal 2011 compared with fiscal 2010 and increased \$2.6 million in fiscal 2012, compared with fiscal 2011. We expect to continue to invest in internal R&D activities, as well as enter into license agreements to supplement our internal R&D initiatives. We intend to initially focus our R&D investments in the specialty pharmaceuticals area where we believe we have the greatest opportunity for growth and profitability. Accordingly, we plan to increase R&D expenditures to support our Brands business.

*Specialty Pharmaceuticals.* We devote significant resources to the R&D of our branded products. A number of our branded products are protected by patents and have enjoyed market exclusivity. Our R&D strategy focuses on branded product development in the area of pain and other central nervous system areas, such as spasticity. We are presently developing a number of branded products, some of which utilize novel drug-delivery systems, through a combination of internal and collaborative programs. As of April 30, 2013, we had one NDA under review in the U.S.

We are presently developing a number of generic products through a combination of internal and collaborative programs. From a product development perspective, we are focused on controlled substances and difficult-to-replicate pharmacokinetic profiles. In addition, we are focused on process improvements to increase yields and reduce costs. As of April 30, 2013, we had five ANDAs awaiting review in the U.S.

*Global Medical Imaging*. Our main focus for our Global Medical Imaging segment is to enhance our CMDS in terms of communicating with hospital information systems and developing specific devices targeting emerging markets. In our Nuclear Imaging business, we are expanding our portfolio of radioisotopes and better utilizing existing capacity.

## Legal Charges

During fiscal 2012, we recorded a legal charge of \$4.3 million to settle a long-standing commercial dispute and charges of \$3.1 million related to product liability litigation, including legal fees. In addition, during fiscal

-85-

2011 and 2010, we incurred legal charges of \$7.8 million and \$31.3 million, respectively, related to product liability litigation and related legal fees. All of the above charges are included in selling, general and administrative expenses.

### **Results of Operations**

### Six Months Ended March 29, 2013 Compared with Six Months Ended March 30, 2012

## Net Sales

Net sales by geographic area are as follows:

	Six Mont	ths Ended			
(Dollars in Millions)	March 29, 2013	March 30, 2012	Percentage Change	Currency Impact	Operational Growth <sup>(1)</sup>
U.S.	\$ 749.1	\$ 670.1	11.8%	%	11.8%
Europe, Middle East and Africa	197.9	210.7	(6.1)	(0.9)	(5.2)
Other	142.3	146.0	(2.5)	(2.7)	0.2
	\$1,089.3	\$1,026.8	6.1	(0.6)	6.7

<sup>1)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Net sales in the first six months of fiscal 2013 increased \$62.5 million, or 6.1%, to \$1,089.3 million, compared with \$1,026.8 million in the first six months of fiscal 2012. In the first six months of fiscal 2013, net sales in the U.S. increased \$79.0 million, or 11.8%, and net sales outside the U.S. decreased \$16.5 million, or 4.6%. The increase in net sales was primarily driven by increased sales within our Specialty Pharmaceuticals segment resulting from the launch of Methylphenidate HCl, increased sales of our Exalgo-branded products and the impact of the CNS Therapeutics. These increases in net sales were partially offset by decreased sales of contrast media and delivery systems products within our Global Medical Imaging segment. Additional information regarding changes in our net sales is provided in "—Business Segment Results."

### **Operating Income**

*Gross profit.* Gross profit for the first six months of fiscal 2013 increased \$18.7 million, or 3.8%, to \$507.0 million, compared with \$488.3 million in the first six months of fiscal 2012. Gross margin was 46.5% in the first six months of fiscal 2013, compared with 47.6% in the first six months of fiscal 2012. The increase in gross profit was primarily the result of higher net sales in the first six months of fiscal 2013 and favorable product mix resulting from increased sales of our higher margin pharmaceutical products. These factors were partially offset by increased manufacturing costs and increased raw material costs, primarily attributable to the unscheduled shutdown of a nuclear reactor that supplies us with Mo-99. The increase in raw material costs also resulted in a decrease in gross margin in the first six months of fiscal 2013.

Selling, general and administrative expenses. Selling, general and administrative expenses for the first six months of fiscal 2013 were \$307.5 million, compared with \$274.1 million for the first six months of fiscal 2012, an increase of \$33.4 million, or 12.2%. The increase in selling, general and administrative expenses primarily resulted from \$28.5 million of costs incurred in the first six months of fiscal 2013 to build out our corporate infrastructure and higher legal costs. Selling, general and administrative expenses were 28.2% of net sales for the first six months of fiscal 2013, compared with 26.7% for the first six months of fiscal 2012.

*Research and development expenses.* R&D expenses increased \$5.3 million, or 7.3%, to \$77.6 million in the first six months of fiscal 2013, compared with \$72.3 million in the first six months of fiscal 2012. The increase in R&D expenses is attributable to increased development activities related to our product pipeline. As a percentage of our net sales, R&D expenses were 7.1% and 7.0% in the first six months of fiscal 2012, respectively.

-86-

Separation costs. During the first six months of fiscal 2013 and 2012, we incurred separation costs of \$26.4 and \$10.2 million, respectively, primarily related to legal, accounting, tax and other professional fees. We expect to continue to incur costs related to the separation throughout fiscal 2013 and beyond.

*Restructuring and related charges, net.* During the first six months of fiscal 2013, we recorded \$7.9 million of net restructuring and related charges, of which \$1.3 million related to accelerated depreciation and was included in cost of sales. The remaining \$6.6 million primarily related to severance and employee benefits costs incurred within our Specialty Pharmaceuticals segment. During the first six months of fiscal 2012, we recorded net restructuring and related charges of \$10.9 million, of which \$5.4 million related to accelerated depreciation and was included in cost of sales. The remaining \$5.5 million primarily related to severance and employee benefit costs incurred across both segments.

*Gain on divestitures.* As discussed under "—Divestitures," during the first six months of both fiscal 2013 and 2012, we recorded a \$1.4 million gain related to the sale of the rights to market TussiCaps extended-release capsules in fiscal 2011.

### Non-Operating Items

Interest expense and interest income. During the first six months of fiscal 2013, net interest expense was \$0.1 million. We expect our annual interest expense to increase approximately \$43 million primarily as a result of the financing arrangements that MIFSA has entered into in connection with our separation from Covidien. See "Unaudited Pro Forma Condensed Combined Financial Statements—The Pharmaceuticals Business of Covidien plc Unaudited Pro Forma Condensed Combined Financial Statements."

Other income, net. During the first six months of fiscal 2013 and 2012, we recorded other income, net of \$0.2 million and \$0.7 million, respectively, which represents miscellaneous items, none of which are material.

*Income tax expense.* Income tax expense was \$36.1 million and \$49.4 million on income from continuing operations before income taxes of \$90.4 million and \$128.3 million for the first six months of fiscal 2013 and 2012, respectively. Our effective tax rate was 39.9% and 38.5% for the first six months of fiscal 2013 and 2012, respectively. The increase in the effective tax rate for the first six months of fiscal 2013, compared with the first six months of fiscal 2012, primarily resulted from increased sales of Methylphenidate HCl in higher-tax jurisdictions and the non-deductibility of certain professional fees incurred in connection with the separation. These increases to our effective tax rate were partially offset by the retroactive re-enactment of the U.S. R&D tax credit in the current period. Our pro forma adjusted tax rate is 25.7% for the first six months of fiscal 2013. See "Unaudited Pro Forma Condensed Combined Financial Statements—The Pharmaceuticals Business of Covidien plc Unaudited Pro Forma Condensed Combined Statement of Income."

## Fiscal Year Ended September 28, 2012 Compared with Fiscal Year Ended September 30, 2011

## Net Sales

Net sales by geographic area are as follows:

	Fis	scal	Percentage	Currency	Operational
(Dollars in Millions)	2012	2011	Change	Impact	Growth <sup>(1)</sup>
U.S.	\$1,350.2	\$1,293.8	4.4%	%	4.4%
Europe, Middle East and Africa	411.0	419.7	(2.1)	(5.6)	3.5
Other	295.0	308.3	(4.3)	(2.4)	(1.9)
	\$2,056.2	\$2,021.8	1.7	(1.5)	3.2

(1) Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

# -87-

Our net sales for fiscal 2012 increased \$34.4 million, or 1.7%, to \$2,056.2 million, compared with \$2,021.8 million in fiscal 2011. In fiscal 2012, net sales in the U.S. increased \$56.4 million, or 4.4%, and net sales outside the U.S. decreased \$22.0 million, or 3.0%. The overall increase in net sales was primarily driven by a \$50.7 million increase in sales of our Exalgo-branded products within our Specialty Pharmaceuticals segment, partially offset by a \$22.7 million decrease in sales of our Optiray contrast product within our Global Medical Imaging segment. Additional information regarding changes in our net sales is provided in "—Business Segment Results."

## **Operating Income**

*Gross profit.* Gross profit for fiscal 2012 increased \$49.9 million, or 5.5%, to \$964.8 million, compared with \$914.9 million in fiscal 2011. The increase in gross profit was primarily a result of overall higher net sales. Gross margin was 46.9% in fiscal 2012, compared with 45.3% in fiscal 2011. The increase in gross margin was primarily attributable to a more favorable product mix resulting from increased sales of our higher margin branded pharmaceutical products.

Selling, general and administrative expenses. Selling, general and administrative expenses for fiscal 2012 increased \$19.2 million, or 3.6%, to \$551.7 million, compared with \$532.5 million in fiscal 2011. The increase in selling, general and administrative expenses primarily resulted from higher legal and benefit costs. Selling, general and administrative expenses were 26.8% of net sales for fiscal 2012, compared with 26.3% of net sales for fiscal 2011.

*Research and development expenses.* R&D expenses increased \$2.6 million to \$144.1 million in fiscal 2012, compared with \$141.5 million in fiscal 2011. The increase primarily resulted from additional spending on our MNK-795 and MNK-155 branded products that are under development within our Specialty Pharmaceuticals segment and higher salary and benefit costs. As a percentage of our net sales, R&D expenses were 7.0% in both fiscal 2012 and 2011.

Separation costs. During fiscal 2012 and 2011, we incurred separation costs of \$25.5 million and \$2.9 million, respectively, primarily related to tax, accounting and other professional fees.

*Restructuring and related charges, net.* During fiscal 2012, we recorded \$19.2 million of net restructuring and related charges, of which \$8.0 million related to accelerated depreciation and were included in cost of sales. The accelerated depreciation resulted from the decision to shut down our plant in Chesterfield, U.K. The remaining \$11.2 million primarily related to severance and employee benefit costs due to a reduction in work force. During fiscal 2011, we recorded net restructuring and related charges of \$10.0 million, of which \$1.6 million related to accelerated depreciation and was included in cost of sales. The remaining \$8.4 million primarily related to severance and employee benefit costs incurred within our Specialty Pharmaceuticals segment.

*Gain on divestitures.* As discussed under "—Divestitures," during fiscal 2011, we recorded an \$11.1 million gain on the sale of the rights to market TussiCaps extended-release capsules. We recorded an additional \$2.9 million gain related to this sale during fiscal 2012.

#### Non-Operating Items

Interest expense and interest income. During fiscal 2012 and 2011, interest expense, net of interest income, was \$0.1 million and \$0.4 million, respectively.

*Other income, net.* During fiscal 2012 and 2011, we recorded other income of \$1.0 million and \$2.9 million, respectively. These amounts primarily represent royalty payments from a subsidiary of Covidien for use of certain of our trademarks and technology.

-88-

*Income tax expense*. Income tax expense was \$94.8 million and \$86.2 million on income from continuing operations before income taxes of \$236.1 million and \$243.2 million for fiscal 2012 and 2011, respectively. Our effective tax rate was 40.2% and 35.4% for fiscal 2012 and 2011, respectively. The increase in the effective tax rate for fiscal 2012, compared with fiscal 2011, resulted primarily from a decrease in earnings in lower-tax jurisdictions. The expiration of the U.S. R&D tax credit as of December 31, 2011 and the retroactive reenactment of the 2010 R&D tax credit during fiscal 2012, our effective tax rate would have been approximately 0.7% lower. In addition, in fiscal 2011, we reached a settlement with certain non-U.S. taxing authorities that favorably benefitted our fiscal 2011 effective tax rate.

### Fiscal Year Ended September 30, 2011 Compared with Fiscal Year Ended September 24, 2010

## Net Sales

Net sales by geographic area are as follows:

	Fis	scal	Percentage	Currency	Operational
(Dollars in Millions)	2011	2010	Change	Impact	Ġrowth <sup>(1)</sup>
U.S.	\$1,293.8	\$1,380.5	(6.3)%	%	(6.3)%
Europe, Middle East and Africa	419.7	393.8	6.6	3.2	3.4
Other	308.3	273.3	12.8	5.0	7.8
	\$2,021.8	\$2,047.6	(1.3)	1.2	(2.5)

<sup>1)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Our net sales for fiscal 2011 decreased \$25.8 million, or 1.3%, to \$2,021.8 million, compared with \$2,047.6 million in fiscal 2010. In fiscal 2011, net sales in the U.S. decreased \$86.7 million, or 6.3%, and net sales outside the U.S. increased \$60.9 million, or 9.1%. The overall decrease in net sales was primarily driven by a decline in Nuclear Imaging net sales within our Global Medical Imaging segment resulting from the divestiture of our nuclear radiopharmacies in the U.S. in May 2010, largely offset by increased sales of our Specialty Pharmaceuticals segment. Additional information regarding changes in our net sales is provided in "—Business Segment Results."

### **Operating Income**

*Gross profit*. Gross profit for fiscal 2011 decreased \$17.5 million, or 1.9%, to \$914.9 million, compared with \$932.4 million in fiscal 2010, primarily as a result of our lower overall net sales. Gross profit margins were 45.3% in fiscal 2011, compared with 45.5% in fiscal 2010. The decrease in gross profit margin was primarily attributable to a \$14.7 million increase in royalties largely associated with certain products within our Specialty Pharmaceuticals segment.

Selling, general and administrative expenses. Selling, general and administrative expenses for fiscal 2011 decreased \$32.8 million, or 5.8%, to \$532.5 million, compared with \$565.3 million in fiscal 2010. The decrease in selling, general and administrative expenses primarily resulted from decreased legal and benefit costs. These decreases were partially offset by an increase in selling and marketing expenses to support our Exalgo and Pennsaid product launches. Selling, general and administrative expenses were 26.3% of net sales for fiscal 2011, compared with 27.6% of net sales for fiscal 2010.

Research and development expenses. R&D expenses increased \$22.4 million to \$141.5 million in fiscal 2011, compared with \$119.1 million in fiscal 2010. This increase primarily resulted from additional spending on

-89-

our MNK-795, MNK-155 and MNK-395 branded products that are under development within our Specialty Pharmaceuticals segment and the reformulation of existing products within our Global Medical Imaging segment. As a percentage of our net sales, R&D expenses were 7.0% for fiscal 2011, compared with 5.8% for fiscal 2010.

Separation costs. During fiscal 2011, we recorded \$2.9 million of separation costs.

*Restructuring and related charges, net.* During fiscal 2011, we recorded net restructuring and related charges of \$10.0 million, of which \$1.6 million related to accelerated depreciation and was included in cost of sales. The remaining \$8.4 million primarily related to severance and employee benefit costs incurred within our Specialty Pharmaceuticals segment. During fiscal 2010, we recorded \$11.5 million of net restructuring charges, which primarily related to severance and employee benefit costs incurred within our Global Medical Imaging segment.

*Gain on divestitures*. During fiscal 2011, we recorded an \$11.1 million gain on the sale of the rights to market TussiCaps extended-release capsules. During fiscal 2010, we recorded a \$3.9 million gain on the sale of our nuclear radiopharmacies in the U.S.

#### Non-Operating Items

Interest expense and interest income. During fiscal 2011 and 2010, interest expense, net of interest income, was \$0.4 million and \$0.6 million, respectively.

Other income, net. During fiscal 2011 and 2010, we recorded other income, net of \$2.9 million and \$3.4 million, respectively. These amounts represent royalty payments from a subsidiary of Covidien for use of certain of our trademarks and technology.

*Income tax expense*. Income tax expense was \$86.2 million and \$97.3 million on income from continuing operations before income taxes of \$243.2 million for both fiscal 2011 and 2010, respectively. Our effective tax rate was 35.4% and 40.0% for fiscal 2011 and 2010, respectively. The decrease in the effective tax rate for fiscal 2011, compared with fiscal 2010, resulted primarily from a favorable settlement reached with certain non-U.S. taxing authorities and the release of certain U.S. and non-U.S. uncertain tax positions due to expiration of statutory limitation periods. In addition, the decrease in the effective tax rate resulted from an increase in earnings in lower-tax jurisdictions, the retroactive reenactment in December 2010 of the U.S. R&D tax credit as described above and the implementation of our tax planning strategies.

Discontinued operations. During fiscal 2010, we sold our Specialty Chemicals business (formerly known as "Mallinckrodt Baker"), because its products and customer bases were not aligned with our long-term strategic objectives. This business met the discontinued operations criteria, and accordingly is included in discontinued operations.

We received net cash proceeds of \$273.3 million and recorded a \$20.4 million pre-tax gain on the sale of Mallinckrodt Baker in fiscal 2010. Included within this gain is a \$17.7 million pre-tax charge associated with indemnification obligations to the purchaser. In addition, we paid \$30.0 million into an escrow account as collateral for these indemnification obligations. Additional information regarding these indemnification obligations is included in "—Commitments and Contingencies—Guarantees."

During fiscal 2011, we recorded a \$9.1 million pre-tax loss on the sale of Mallinckrodt Baker, primarily for pension settlements related to its employees. In addition, during fiscal 2012, we recorded an additional \$6.7 million loss, primarily related to the indemnification obligations discussed above.

-90-

### **Business Segment Results**

The businesses included within our Specialty Pharmaceuticals and our Global Medical Imaging segments are described below:

### Specialty Pharmaceuticals

- Brands—includes branded pharmaceuticals for pain and spasticity.
- *Generics and API*—produces generic pharmaceutical products, medicinal opioids, synthetic controlled substances, acetaminophen and addiction treatment.

## **Global Medical Imaging**

- Contrast Media and Delivery Systems—develops, manufactures and markets contrast media for diagnostic imaging applications, and power injectors
  to allow delivery of contrast media.
- Nuclear Imaging—manufactures and markets radioactive isotopes and associated pharmaceuticals used for the diagnosis and treatment of disease.

Management measures and evaluates our operating segments based on segment net sales and operating income. Management excludes corporate expenses from segment operating income. In addition, management evaluates the operating results of the segments excluding certain amounts that management considers to be non-recurring or non-operational. These items include revenues and expenses associated with related party sales of products to other Covidien businesses, intangible asset amortization, net restructuring and related charges, and separation costs. Although these amounts are excluded from segment operating income, as applicable, they are included in reported combined operating income and accordingly, are included in our discussion of our combined results of operations.

## Six Months Ended March 29, 2013 Compared with Six Months Ended March 30, 2012

## Net Sales

Net sales by segment are shown in the following table:

	Six Mont	hs Ended			
(Dollars in Millions)	March 29, 2013	March 30, 2012	Percentage Change	Currency Impact	Operational Growth <sup>(1)</sup>
Specialty Pharmaceuticals	\$ 604.6	\$ 491.6	23.0%	(0.1)%	23.1%
Global Medical Imaging	458.8	507.3	(9.6)	(1.0)	(8.6)
Net sales of operating segments	1,063.4	998.9	6.5	(0.5)	7.0
Net sales to related parties <sup>(2)</sup>	25.9	27.9	(7.2)	—	(7.2)
Net sales	\$1,089.3	\$1,026.8	6.1	(0.6)	6.7

<sup>(1)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

<sup>(2)</sup> Represents products that were sold to other Covidien businesses.

Specialty Pharmaceuticals. Net sales for the first six months of fiscal 2013 increased \$113.0 million, or 23.0%, to \$604.6 million, compared with \$491.6 million for the first six months of fiscal 2012. The increase in net sales was primarily driven by \$70.9 million of sales from the launch of Methylphenidate HCl; an \$18.9 million increase in net sales of Exalgo-branded products, which was aided by the launch of a new dosage in August 2012; a \$15.1 million increase in sales of Oxycodone (API) and oxycodone-containing tablets; and \$13.3 million of sales of intrathecal products resulting from the acquisition of CNS Therapeutics. These increases in net sales were partially offset by a \$13.0 million decrease in sales of other Brands products, primarily

-91-

Pennsaid. We expect our Brands business to grow in fiscal 2013, compared with fiscal 2012, as a result of increases in sales of Exalgo and intrathecal products. Our Generics and API business is also expected to grow in fiscal 2013, compared with fiscal 2012, due to the launch of Methylphenidate HCl in fiscal 2013.

Net sales for Specialty Pharmaceuticals by geography are as follows:

	Six Mon	ths Ended			
(Dollars in Millions)	March 29, 2013	March 30, 2012	Percentage Change	Currency Impact	Operational Growth <sup>(1)</sup>
U.S.	\$ 547.9	\$ 434.2	26.2%	%	26.2%
Europe, Middle East and Africa	48.6	51.1	(4.9)	0.3	(5.2)
Other	8.1	6.3	28.6	(14.7)	43.3
	\$ 604.6	\$ 491.6	23.0	(0.1)	23.1

(1) Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Net sales for Specialty Pharmaceuticals by key products are as follows:

	Six N	Six Months Ended		
(Dollars in Millions)	March 29, 2013	March 30, 2012	Percentage Change	
Acetaminophen (API) products	\$ 104.7	\$ 107.9	(3.0)%	
Oxycodone (API) and oxycodone-containing tablets	85.2	70.1	21.5	
Methylphenidate HC1	70.9	_	NM <sup>(2)</sup>	
Hydrocodone (API) and hydrocodone-containing tablets	69.0	71.5	(3.5)	
Other controlled substances	57.0	54.7	4.2	
Other	123.4	112.2	10.0	
Generics and API	510.2	416.4	22.5	
Exalgo	58.0	39.1	48.3	
Intrathecal products	13.3	_	NM <sup>(2)</sup>	
Other	23.1	36.1	(36.0)	
Brands	94.4	75.2	25.5	
Specialty Pharmaceuticals	\$ 604.6	\$ 491.6	23.0	

<sup>(2)</sup> Not meaningful.

*Global Medical Imaging.* Net sales for the first six months of fiscal 2013 decreased \$48.5 million, or 9.6%, to \$458.8 million compared with \$507.3 million for the first six months of fiscal 2012. This decrease was largely due to a \$40.7 million decrease in sales of CMDS products. The decrease in sales of CMDS products primarily resulted from lower Optiray sales due to the renegotiation of a customer contract in the U.S. market and continued weakness in the U.S. A one-time order in the comparative prior year period also contributed to the CMDS sales decline. We expect CMDS to continue to experience weakness resulting from a decreasing number of procedures in developed markets and pricing pressure.

-92-

Net sales for Global Medical Imaging by geography are as follows:

Six Mont	ths Ended			
March 29, 2013	March 30, 2012	Percentage Change	Currency Impact	Operational Growth <sup>(1)</sup>
\$ 199.6	\$ 234.4	(14.8)%	<u> </u>	(14.8)%
149.3	159.6	(6.5)	(1.4)	(5.1)
109.9	113.3	(3.0)	(2.3)	(0.7)
\$ 458.8	\$ 507.3	(9.6)	(1.0)	(8.6)
	March 29, 2013 \$ 199.6 149.3 109.9	2013         2012           \$ 199.6         \$ 234.4           149.3         159.6           109.9         113.3           \$ 458.8         \$ 507.3	March 29, 2013         March 30, 2012         Percentage Change           \$ 199.6         \$ 234.4         (14.8)%           149.3         159.6         (6.5)           109.9         113.3         (3.0)           \$ 458.8         \$ 507.3         (9.6)	March 29, 2013         March 30, 2012         Percentage Change         Currency Impact           \$ 199.6         \$ 234.4         (14.8)%        %           149.3         159.6         (6.5)         (1.4)           109.9         113.3         (3.0)         (2.3)           \$ 458.8         \$ 507.3         (9.6)         (1.0)

<sup>(1)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Net sales for Global Medical Imaging by key products are as follows:

	Six Mont	hs Ended		
(Dollars in Millions)	March 29, 2013	March 30, 2012	Percentage Change	
Optiray	\$ 154.5	\$ 175.3	(11.9)%	
Optimark	22.3	22.6	(1.3)	
Other	62.8	82.4	(23.8)	
Contrast Media and Delivery Systems	239.6	280.3	(14.5)	
Ultra-Technekow DTE	96.9	99.0	(2.1)	
Octreoscan	41.1	39.9	3.0	
Other	81.2	88.1	(7.8)	
Nuclear Imaging	219.2	227.0	(3.4)	
Global Medical Imaging	\$ 458.8	\$ 507.3	(9.6)	

## **Operating Income**

Operating income by segment and as a percentage of segment net sales for the first six months of fiscal 2013 and 2012 is shown in the following table:

		Six Months Ended				
(Dollars in Millions)		March 29, 2013		30,		
Specialty Pharmaceuticals	\$140.0	23.2%	\$ 77.4	15.7%		
Global Medical Imaging	68.0	14.8	111.4	22.0		
Segment operating income	208.0	19.6	188.8	18.9		
Unallocated amounts:						
Corporate and allocated expenses	(65.7)		(26.6)			
Intangible asset amortization	(17.7)		(13.5)			
Restructuring and related charges, net	(7.9)		(10.9)			
Separation costs	(26.4)		(10.2)			
Total operating income	\$ 90.3		\$127.6			

Specialty Pharmaceuticals. Operating income for the first six months of fiscal 2013 increased \$62.6 million to \$140.0 million, compared with \$77.4 million for the first six months of fiscal 2012. Our operating margin increased to 23.2% for the first six months of fiscal 2013, compared with 15.7% for the first six months of fiscal

-93-

2012. The increase in operating income and margin was primarily due to increased sales of higher margin products, namely Methylphenidate HCl, and favorable pricing. We anticipate operating income for our Specialty Pharmaceuticals segment to increase in fiscal 2013, compared with fiscal 2012, as a result of these sales increases.

*Global Medical Imaging.* Operating income for the first six months of fiscal 2013 decreased \$43.4 million to \$68.0 million, compared with \$111.4 million for the first six months of fiscal 2012. Our operating margin decreased to 14.8% for the first six months of fiscal 2013, compared with 22.0% for the first six months of fiscal 2012. The decrease in operating income was attributable to lower net sales discussed above and increased manufacturing and raw material costs, partially offset by a decrease in selling, general and administrative expenses. Our operating margin was most significantly impacted by higher raw material costs from the unscheduled shutdown of a nuclear reactor that supplies us with Mo-99. We expect operating income for our Global Medical Imaging segment to decline in fiscal 2013, compared to fiscal 2012, due to negative market trends, including a decrease in the number of procedures performed in developed markets and pricing pressure. In addition, we may continue to experience increased raw materials costs, partially as a result of the unscheduled shutdown of one of the reactors that supplies us with Mo-99, as discussed under "—Nuclear Imaging."

*Corporate and allocated expenses.* Corporate and allocated expenses were \$65.7 million and \$26.6 million for the first six months of fiscal 2013 and 2012, respectively. These amounts include allocations of \$25.5 million and \$22.7 million during the first six months of fiscal 2013 and 2012, respectively, for certain functions provided by Covidien, as described under "—Separation from Covidien." These expenses have been allocated to us on the basis of direct usage when identifiable, with the remainder allocated on the basis of operating expenses, headcount or other measures. Excluding the \$2.8 million increase in the amount of allocated expenses, the remaining \$36.3 million increase in corporate expenses in the first six months of fiscal 2013, compared with the corresponding prior year period, primarily resulted from \$28.5 million of costs incurred to build out our corporate infrastructure in first six months of fiscal 2013 compared with \$0.1 million in the first six months of fiscal 2012.

## Fiscal Year Ended September 28, 2012 Compared with Fiscal Year Ended September 30, 2011

### Net Sales

Net sales by segment are shown in the following table:

	Fiscal		Percentage	Currency	Operational
(Dollars in Millions)	2012	2011	Change	Impact	Growth <sup>(1)</sup>
Specialty Pharmaceuticals	\$1,005.2	\$ 909.4	10.5%	(0.2)%	10.7%
Global Medical Imaging	996.8	1,060.0	(6.0)	(2.8)	(3.2)
Net sales of operating segments	2,002.0	1,969.4	1.7	(1.5)	3.2
Net sales to related parties <sup>(2)</sup>	54.2	52.4	3.4	—	3.4
Net sales	\$2,056.2	\$2,021.8	1.7	(1.5)	3.2

(1) Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

<sup>(2)</sup> Represents products that were sold to other Covidien businesses.

Specialty Pharmaceuticals. Net sales for fiscal 2012 increased \$95.8 million, or 10.5%, to \$1,005.2 million, compared with \$909.4 million in fiscal 2011. The increase in net sales was primarily driven by increased sales of our Exalgo and Pennsaid branded products. This increase was partially offset by the impact of the extra selling week in fiscal 2011 and a decrease in sales of oxycodone immediate-release tablets.

-94-

Net sales for Specialty Pharmaceuticals by geography are as follows:

	F	Fiscal		Currency	Operational
(Dollars in Millions)	2012	2011	Change	Impact	Growth <sup>(1)</sup>
U.S.	\$ 880.6	\$784.8	12.2%	%	12.2%
Europe, Middle East and Africa	108.7	93.4	16.4	(2.1)	18.5
Other	15.9	31.2	(49.0)	0.8	(49.8)
	\$1,005.2	\$909.4	10.5	(0.2)	10.7

<sup>(1)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Net sales for Specialty Pharmaceuticals by key products are as follows:

(Dollars in Millions)	Fi	scal 2011	Percentage Change
Acetaminophen (API) products	\$ 217.7	\$222.2	(2.0)%
Oxycodone (API) and oxycodone-containing tablets	144.1	154.1	(6.5)
Hydrocodone (API) and hydrocodone-containing tablets	130.5	116.9	11.6
Other controlled substances	111.7	107.9	3.5
Other	244.8	223.6	9.5
Generics and API	848.8	824.7	2.9
Exalgo	91.9	41.2	123.1
Other	64.5	43.5	48.3
Brands	156.4	84.7	84.7
Specialty Pharmaceuticals	\$1,005.2	\$909.4	10.5

*Global Medical Imaging*. Net sales for fiscal 2012 decreased \$63.2 million, or 6.0%, to \$996.8 million, compared with \$1,060.0 million in fiscal 2011. This decrease was largely due to decreased sales of CMDS, primarily resulting from lower sales of Optiray due to the renegotiation of a customer contract in the U.S. market and discontinuance of a product, combined with unfavorable currency exchange rate fluctuations and other market-related challenges. In addition, fiscal 2012 sales growth was negatively impacted by the extra selling week in fiscal 2011.

Net sales for Global Medical Imaging by geography are as follows:

	F	Fiscal		Currency	Operational
(Dollars in Millions)	2012	2011	Change	Impact	Ġrowth <sup>(1)</sup>
U.S.	\$466.8	\$ 505.8	(7.7)%	— %	(7.7)%
Europe, Middle East and Africa	302.3	326.3	(7.4)	(6.7)	(0.7)
Other	227.7	227.9	(0.1)	(3.5)	3.4
	\$996.8	\$1,060.0	(6.0)	(2.8)	(3.2)

<sup>(1)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

-95-

Net sales for Global Medical Imaging by key products are as follows:

		Fiscal	Percentage
lars in Millions)	2012	2011	Change
tiray	\$352.2	\$ 374.9	(6.1)%
timark	48.0	50.3	(4.6)
er	141.8	170.3	(16.7)
Contrast Media and Delivery Systems	542.0	595.5	(9.0)
ra-Technekow DTE	202.5	200.3	1.1
reoscan	78.7	76.9	2.3
er	173.6	187.3	(7.3)
Nuclear Imaging	454.8	464.5	(2.1)
bal Medical Imaging	\$996.8	\$1,060.0	(6.0)
er Contrast Media and Delivery Systems ra-Technekow DTE reoscan eer Nuclear Imaging	141.8 542.0 202.5 78.7	170.3 595.5 200.3 76.9 187.3 464.5	(16 (9 1 2 (7 (2

# **Operating Income**

Operating income by segment and as a percentage of segment net sales for fiscal 2012 and 2011 is shown in the following table:

		Fiscal			
(Dollars in Millions)	2012		2011		
Specialty Pharmaceuticals	\$162.8	16.2%	\$121.5	13.4%	
Global Medical Imaging	214.3	21.5	232.4	21.9	
Segment operating income	377.1	18.8	353.9	18.0	
Unallocated amounts:					
Corporate and allocated expenses	(69.9)		(73.3)		
Intangible asset amortization	(27.3)		(27.0)		
Restructuring and related charges, net	(19.2)		(10.0)		
Separation costs	(25.5)		(2.9)		
Total operating income	\$235.2		\$240.7		

Specialty Pharmaceuticals. Operating income for fiscal 2012 increased \$41.3 million to \$162.8 million, compared with \$121.5 million for fiscal 2011. Our operating margin was 16.2% for fiscal 2012, compared with 13.4% for fiscal 2011. The increase in operating income and margin was primarily due to favorable product mix resulting from increased sales of our higher margin branded products.

*Global Medical Imaging*. Operating income for fiscal 2012 decreased \$18.1 million to \$214.3 million, compared with \$232.4 million for fiscal 2011. Our operating margin was 21.5% for fiscal 2012, compared with 21.9% for fiscal 2011. The decrease in operating income and margin was primarily due to lower pricing and volume from renegotiated contracts with certain customer groups, which resulted in a switch to a dual source contract from a single source contract.

*Corporate and allocated expenses.* Corporate and allocated expenses were \$69.9 million and \$73.3 million for fiscal 2012 and 2011, respectively. These amounts include allocations of \$49.2 million and \$56.3 million during fiscal 2012 and 2011, respectively, for certain functions provided by Covidien, as described under "—Separation from Covidien." Excluding the \$7.1 million decrease in the amount of allocated expenses, the remaining \$3.7 million increase in corporate expenses in fiscal 2012, compared with fiscal 2011, primarily resulted from \$10.7 million of costs incurred to build out our corporate infrastructure, partially offset by lower environmental and asbestos-related costs.

# -96-

### Fiscal Year Ended September 30, 2011 Compared with Fiscal Year Ended September 24, 2010

Net Sales

Net sales by segment are show in the following table:

	Fiscal		Percentage	Currency	Operational
(Dollars in Millions)	2011	2010	Change	Impact	Ĝrowth <sup>(1)</sup>
Specialty Pharmaceuticals	\$ 909.4	\$ 869.0	4.6%	0.5%	4.1%
Global Medical Imaging	1,060.0	1,128.1	(6.0)	1.9	(7.9)
Net sales of operating segments	1,969.4	1,997.1	(1.4)	1.3	(2.7)
Net sales to related parties <sup>(2)</sup>	52.4	50.5	3.8		3.8
Net sales	\$2,021.8	\$2,047.6	(1.3)	1.2	(2.5)

(1) Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

<sup>(2)</sup> Represents products that were sold to other Covidien businesses.

Specialty Pharmaceuticals. Net sales for fiscal 2011 increased \$40.4 million, or 4.6%, to \$909.4 million, compared with \$869.0 million in fiscal 2010. This increase was driven primarily by increased sales of generic pharmaceuticals, primarily the fentanyl patch and lozenge which are included within Other in the table of key products and product families below, and increased sales of acetaminophen within API. Increased sales of our Exalgo and Pennsaid branded products were more than offset by the decline in sales of our older branded products due to generic competition. Net sales in fiscal 2011 also benefitted from the extra selling week, which favorably impacted both product groups.

Net sales for Specialty Pharmaceuticals by geography are as follows:

	Fis	cal	Percentage	Currency	Operational
(Dollars in Millions)	2011	2010	Change	Impact	Ġrowth <sup>(3)</sup>
U.S.	\$784.8	\$756.3	3.8%	%	3.8%
Europe, Middle East and Africa	93.4	89.7	4.1	2.9	1.2
Other	31.2	23.0	35.7	10.0	25.7
	\$909.4	\$869.0	4.6	0.5	4.1

<sup>(3)</sup> Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Net sales for Specialty Pharmaceuticals by key products and product families are as follows:

	Fis	cal	Percentage
(Dollars in Millions)	2011	2010	Change
Acetaminophen (API) products	\$222.2	\$203.6	9.1%
Oxycodone (API) and oxycodone-containing tablets	154.1	170.2	(9.5)
Hydrocodone (API) and hydrocodone-containing tablets	116.9	116.7	0.2
Other controlled substances	107.9	106.8	1.0
Other	223.6	184.5	21.2
Generics and API	824.7	781.8	5.5
Exalgo	41.2	24.8	66.1
Other	43.5	62.4	(30.3)
Brands	84.7	87.2	(2.9)
Specialty Pharmaceuticals	\$909.4	\$869.0	4.6

-97-

*Global Medical Imaging.* Net sales for fiscal 2011 decreased \$68.1 million, or 6.0%, to \$1,060.0 million, compared with \$1,128.1 million in fiscal 2010. This decrease was driven primarily by a decline in Nuclear Imaging net sales resulting from the divestiture of our nuclear radiopharmacies within the U.S. during fiscal 2010. This decrease was partially offset by increased sales of Ultra-Technekow DTE generators. Net sales in fiscal 2011 also benefitted from the extra selling week.

Net sales for Global Medical Imaging by geography are as follows:

	Fis	scal	Percentage	Currency	Operational
(Dollars in Millions)	2011	2010	Change	Impact	Ĝrowth <sup>(1)</sup>
U.S.	\$ 505.8	\$ 621.5	(18.6)%	— %	(18.6)%
Europe, Middle East and Africa	326.3	304.1	7.3	3.3	4.0
Other	227.9	202.5	12.5	5.6	6.9
	\$1,060.0	\$1,128.1	(6.0)	1.9	(7.9)

(1) Operational growth is a non-GAAP financial measure which should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP. See "—Management's Use of Non-GAAP Measures."

Net sales for Global Medical Imaging by key products are as follows:

	Fis	cal	Percentage
(Dollars in Millions)	2011	2010	Change
Optiray	\$ 374.9	\$ 357.7	4.8%
Optimark	50.3	48.8	3.1
Other	170.3	202.6	(15.9)
Contrast Media and Delivery Systems	595.5	609.1	(2.2)
Ultra-Technekow DTE	200.3	176.2	13.7
Octreoscan	76.9	65.2	17.9
Other	187.3	277.6	(32.5)
Nuclear Imaging	464.5	519.0	(10.5)
Global Medical Imaging	\$1,060.0	\$1,128.1	(6.0)

### **Operating Income**

Operating income by segment and as a percentage of segment net sales for fiscal 2011 and 2010 is shown in the following table:

Fiscal			al	
(Dollars in Millions)	201	l	2010	
Specialty Pharmaceuticals	\$121.5	13.4%	\$139.6	16.1%
Global Medical Imaging	232.4	21.9	221.5	19.6
Segment operating income	353.9	18.0	361.1	18.1
Unallocated amounts:				
Corporate and allocated expenses	(73.3)		(85.8)	
Intangible asset amortization	(27.0)		(23.4)	
Restructuring and related charges, net	(10.0)		(11.5)	
Separation costs	(2.9)			
Total operating income	\$240.7		\$240.4	

-98-

Specialty Pharmaceuticals. Operating income for fiscal 2011 decreased \$18.1 million to \$121.5 million, compared with \$139.6 million for fiscal 2010. Our operating margin was 13.4% for fiscal 2011, compared with 16.1% for fiscal 2010. The decrease in operating income and margin was primarily due to a \$14.9 million increase in R&D expenses and increased selling and marketing expenses to support product launches, partially offset by an \$11.1 million gain on the sale of the rights to market TussiCaps and decreased benefit costs.

*Global Medical Imaging*. Operating income for fiscal 2011 increased \$10.9 million to \$232.4 million, compared with \$221.5 million for fiscal 2010. Our operating margin was 21.9% for fiscal 2011, compared with 19.6% for fiscal 2010. The increase in operating income was primarily due to a decrease in legal costs, partially offset by increased R&D expenses.

*Corporate and allocated expenses.* Corporate and allocated expenses were \$73.3 million and \$85.8 million for fiscal 2011 and 2010, respectively. These amounts include allocations of \$56.3 million and \$60.8 million during fiscal 2011 and 2010, respectively, for certain functions provided by Covidien, as described in "—Separation from Covidien." Excluding the \$4.5 million decline in allocated expenses, the remaining \$8.0 million decrease in corporate expenses primarily resulted from lower legal and environmental costs and decreased equity-based compensation expense.

## Liquidity and Capital Resources

Significant factors driving our liquidity position include cash flows generated from operating activities, capital expenditures and cash paid in connection with acquisitions and license agreements. Historically, we have typically generated and expect to continue to generate positive cash flow from operations. As part of Covidien, our cash is swept regularly by Covidien at its discretion. Covidien also funds our operating and investing activities as needed. Cash flows related to financing activities reflect changes in Covidien's investments in us. Transfers of cash to and from Covidien are reflected as a component of parent company investment within parent company equity on our combined balance sheets. As discussed further under "—Capitalization," we have not reported cash or cash equivalents on our combined balance sheets for the periods presented.

Subsequent to the separation, we will no longer participate in cash management and funding arrangements with Covidien. Our ability to fund our capital needs will be affected by our ongoing ability to generate cash from operations and access to capital markets. We believe that our future cash from operations, borrowing capacity under the credit facility that we will be able to draw upon after the distribution and access to capital markets will provide adequate resources to fund our working capital needs, capital expenditures and strategic investments.

In March 2013, MIFSA entered into a \$250 million five-year senior unsecured revolving credit facility that matures in June 2018. Borrowings under the credit facility will initially bear interest at LIBOR plus 1.50% per annum (subject to adjustment based upon a ratings-based pricing grid). Our credit facility agreement contains customary covenants, including a financial maintenance covenant that limits our ratio of debt to earnings before interest, income taxes, depreciation and amortization, as adjusted for certain items, and another financial maintenance covenant that requires our ratio of earnings before interest, income taxes, depreciation and amortization, as adjusted for certain items, to interest expense to exceed certain thresholds. MIFSA will not be permitted to draw upon the credit facility until certain conditions are met, including completion of the distribution. In April 2013 MIFSA issued debt in the amount of approximately \$900 million through a notes offering, consisting of \$300 million of 3.50% senior notes due April 2018 and \$600 million of 4.75% senior notes due April 2023.

It is anticipated that, on the distribution date, we will have approximately \$168 million of cash. The separation and distribution agreement will provide for an adjustment payment to potentially be made following the distribution from Covidien to us, or from us to Covidien. The purpose of the adjustment payment is to compensate Mallinckrodt or Covidien, as applicable, to the extent that the aggregate of our cash, indebtedness and specified working capital accounts as of the distribution date, as well as the capital expenditures made with

respect to our business during fiscal 2013 through the distribution date, deviates from a target. The target will be calculated pursuant to a formula that will be set forth in the separation and distribution agreement, which was determined assuming that the distribution date is June 28, 2013, that our business is conducted in the ordinary course through that date and that we will have approximately \$168 million of cash upon completion of the distribution. The actual amount of cash that we will have after giving effect to any adjustment payment, however, may be more or less than \$168 million. The separation and distribution agreement will also provide that an adjustment payment will only be payable if the amount of the adjustment payment exceeds \$20 million (in which case the entire amount will be paid).

In fiscal 2013, we expect our total capital expenditures to be in the range of \$140 million to \$160 million, which includes \$20 million of non-recurring capital expenditures to build out our corporate infrastructure and information technology systems. While we intend to fund these capital expenditures with cash generated from operations, we also will have \$250 million of borrowing capacity under the credit facility after certain conditions are met, including the completion of the distribution. At September 28, 2012, we had capital expenditure commitments of \$3.8 million.

A summary of our cash flows from operating, investing and financing activities is provided in the following table:

	Six Months Ended		Fiscal		
(Dollars in Millions)	March 29, 2013	March 30, 2012	2012	2011	2010
Net cash (used in) provided by continuing:					
Operating activities	\$ (7.8)	\$ 107.3	\$ 255.8	\$ 370.2	\$ 379.4
Investing activities	\$ (165.0)	\$ (62.2)	\$(152.2)	\$(112.6)	\$ 114.3
Financing activities	\$ 172.8	\$ (45.1)	\$(103.6)	\$(257.6)	\$(505.2)

### **Operating** Activities

Net cash used in operating activities of \$7.8 million for the first six months of fiscal 2013 was primarily attributable to a \$136.3 million outflow from net investments in working capital, partially offset by income from continuing operations, as adjusted for non-cash items. The working capital outflow was primarily driven by a \$74.4 million increase in accounts receivable, a \$41.8 million decrease in accrued and other liabilities and a \$23.1 million increase in inventory, partially offset by a \$27.3 million increase in income taxes payable, which was recorded in parent company investment. The increase in accounts receivable was attributable to sales growth primarily from the launch of Methylphenidate HCl. The decrease in accrued and other liabilities resulted largely from a \$37.5 million voluntary contribution to our pension plans and the annual payout of cash bonuses for performance in the prior fiscal year.

Net cash provided by operating activities of \$107.3 million for the first six months of fiscal 2012 was primarily attributable to income from continuing operations, as adjusted for non-cash items, partially offset by a \$41.7 million outflow from net investments in working capital. The working capital outflow was primarily driven by a \$49.2 million decrease in accrued and other liabilities and a \$32.8 million increase in inventory, partially offset by a \$39.1 million increase in income taxes payable, which was recorded in parent company investment. The decrease in accrued and other liabilities resulted largely from decreases in pension and environmental liabilities, as well as the payment of annual cash bonuses for performance in fiscal 2011 that were paid in fiscal 2012.

Net cash provided by operating activities of \$255.8 million in fiscal 2012 was primarily attributable to income from continuing operations, as adjusted for depreciation and amortization, partially offset by a \$25.4 million outflow from net investments in working capital. The working capital outflow was primarily driven by a \$62.8 million increase in inventory and a \$54.2 million decrease in accrued and other liabilities, partially offset by a \$79.4 million increase in income taxes payable, the latter of which was recorded in parent company investment. A build-up of inventory in advance of a planned plant closure contributed to the increase in inventory, while environmental payments contributed to the decrease in accrued and other liabilities.

-100-

Net cash provided by operating activities of \$370.2 million in fiscal 2011 was primarily attributable to income from continuing operations, as adjusted for depreciation and amortization, deferred income taxes and an increase in working capital of \$58.1 million. The increase in working capital was primarily driven by a \$36.0 million increase in income taxes payable, which was recorded in parent company investment.

Net cash provided by operating activities of \$379.4 million in fiscal 2010 was primarily attributable to income from continuing operations, as adjusted for depreciation and amortization and an increase in working capital of \$111.6 million. The increase in working capital was primarily driven by a \$99.5 million increase in income taxes payable, which was recorded in parent company investment.

#### **Investing** Activities

Net cash used in investing activities increased \$102.8 million to \$165.0 million for the first six months of fiscal 2013, compared with \$62.2 million for the first six months of fiscal 2012. This increase primarily resulted from an \$88.1 million payment made in fiscal 2013 to acquire CNS Therapeutics and a \$13.4 million increase in capital expenditures resulting from investments made in connection with the separation.

Net cash used in investing activities increased \$39.6 million to \$152.2 million in fiscal 2012, compared with \$112.6 million in fiscal 2011. This increase primarily resulted from a \$23.8 million increase in capital expenditures and a \$13.2 million payment made in fiscal 2012 to acquire rights to Roxicodone.

Net cash used in investing activities of \$112.6 million in fiscal 2011 was primarily due to capital expenditures of \$120.4 million, partially offset by net proceeds from divestitures.

Net cash provided by investing activities of \$114.3 million in fiscal 2010 primarily resulted from net cash proceeds of \$273.3 million from the divestiture of Mallinckrodt Baker, partially offset by capital expenditures of \$103.5 million and cash paid to acquire Exalgo and license Pennsaid.

#### **Financing** Activities

Net cash provided by financing activities was \$172.8 million for the first six months of fiscal 2013, compared with net cash used in financing activities of \$45.1 million for the first six months of fiscal 2012. The \$217.9 million increase in cash provided by financing activities resulted from an increase in net transfers from Covidien. Net transfers from Covidien were higher during the first six months of fiscal 2013 due to a decrease in operating cash flow and an increase in cash used in investing activities, primarily for the acquisition of CNS Therapeutics.

Net cash used in financing activities decreased \$154.0 million to \$103.6 million in fiscal 2012, compared with \$257.6 million in fiscal 2011. This resulted from a decrease in net transfers to Covidien. Net transfers to Covidien were lower in fiscal 2012 due to a decrease in operating cash flow and an increase in capital expenditures.

Net cash used in financing activities decreased \$247.6 million to \$257.6 million in fiscal 2011, compared with \$505.2 million in fiscal 2010. This resulted from a decrease in net transfers to Covidien. Net transfers to Covidien were higher in fiscal 2010 due to the transfer of the proceeds received from the sale of Mallinckrodt Baker.

#### Capitalization

The cash and cash equivalents held by Covidien at the corporate level are not specifically identifiable to us. Accordingly, cash and cash equivalents have not been allocated to us for any of the periods presented. In addition, Covidien's debt and the related interest expense have not been allocated to us for any of the periods presented since we are not the legal obligor of the debt and Covidien's borrowings were not directly attributable to our business. Debt incurred by us directly is included in our combined financial statements and totaled \$9.4 million at March 29, 2013.

#### Dividends

We currently intend to retain any earnings to finance R&D, acquisitions and the operation and expansion of our business and do not anticipate paying any cash dividends for the foreseeable future. The recommendation, declaration and payment of any dividends in the future by us will be subject to the sole discretion of our board of directors and will depend upon many factors, including our financial condition, earnings, capital requirements of our operating subsidiaries, covenants associated with certain of our debt obligations, legal requirements, regulatory constraints and other factors deemed relevant by our board of directors. Moreover, if we determine to pay any dividends in the future, there can be no assurance that we will continue to pay such dividends. For more information, see "Dividends."

# **Commitments and Contingencies**

# **Contractual Obligations**

The following table summarizes our contractual obligations as of September 28, 2012.

	Payments Due By Period				
		Less than			More than
(Dollars in Millions)	Total	1 year	1-3 years	3-5 years	5 years
Long-term debt obligations <sup>(1)</sup>	\$ 6.4	\$ 0.4	\$ 6.0	\$ —	\$ —
Capital lease obligations <sup>(1)</sup>	4.6	1.4	2.8	0.4	—
Operating leases <sup>(2)</sup>	54.3	11.3	18.2	12.1	12.7
Purchase obligations <sup>(3)</sup>	137.1	70.1	45.8	21.2	—
Total contractual cash obligations	\$202.4	\$ 83.2	\$ 72.8	\$ 33.7	\$ 12.7

<sup>(1)</sup> Interest on debt and capital lease obligations are projected for future periods using interest rates in effect as of September 28, 2012. Certain of these projected interest payments may differ in the future based on changes in market interest rates.

<sup>(2)</sup> Amounts exclude lease arrangements that we may enter into with Covidien at separation.

(3) Purchase obligations consist of commitments for purchases of goods and services made in the normal course of business to meet operational and capital requirements.

The table above excludes obligations that result from financing arrangements that MIFSA entered into in March and April 2013. In addition, the table above does not include other liabilities of \$504.3 million, primarily consisting of obligations under our pension and postretirement benefit plans, unrecognized tax benefits for uncertain tax positions and related accrued interest and penalties, environmental liabilities and asset retirement obligations, because the timing of their future cash outflow is uncertain. The most significant of these liabilities are discussed below.

As of September 28, 2012, we had net unfunded pension and postretirement benefit obligations of \$101.2 million and \$80.3 million, respectively. However, during the first six months of fiscal 2013, Covidien contributed \$37.5 million to our pension plans. While the timing and amounts of long-term funding requirements for pension and postretirement obligations are uncertain, in fiscal 2013, we expect an additional \$12.8 million will be contributed to our pension and postretirement benefit plans (1) by Covidien to the extent that the contribution occurs prior to completion of the separation and/or (2) by us to the extent that the contribution occurs after the separation.

We are involved in various stages of investigation and cleanup related to environmental remediation matters at a number of sites. These projects relate to a variety of activities, including decontamination and decommissioning of radioactive materials and removal of solvents, metals and other hazardous substances from soil and groundwater. The ultimate cost of cleanup and timing of future cash outlays is difficult to predict given uncertainties regarding the extent of the required cleanup, the interpretation of applicable laws and regulations and alternative cleanup methods. As of September 28, 2012, we believe that it is probable that we will incur

-102-

investigation and remedial costs, including asset retirement obligations, of approximately \$197.9 million, of which \$15.2 million is included in accrued and other current liabilities, \$136.5 million is included in environmental liabilities and \$46.2 million is included in other liabilities on our combined balance sheet at September 28, 2012. This amount includes \$95.8 million at September 28, 2012 relating to a site located in Orrington, Maine which will be a liability of a Covidien entity following the separation. Note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement provides additional information regarding environmental matters, including asset retirement obligations.

#### Legal Proceedings

We are subject to various legal proceedings and claims, including patent infringement claims, product liability matters, environmental matters, employment disputes, contractual disputes and other commercial disputes, including those described in "Business—Legal Proceedings" and in note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement. We believe that these legal proceedings and claims likely will be resolved over an extended period of time. Although it is not feasible to predict the outcome of these matters, management is of the opinion that their ultimate resolution should not have a material adverse effect on our financial condition, results of operations and cash flows.

#### Guarantees

In disposing of assets or businesses, we often provide representations, warranties and indemnities to cover various risks, including unknown damage to the assets, environmental risks involved in the sale of real estate, liability to investigate and remediate environmental contamination at waste disposal sites and manufacturing facilities, and unidentified tax liabilities and legal fees related to periods prior to disposition. Except as discussed below, we generally do not have the ability to estimate the potential liability from such indemnities because they relate to unknown conditions. However, we have no reason to believe that these uncertainties would have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

In connection with the sale of Mallinckrodt Baker in fiscal 2010, we agreed to indemnify the purchaser with respect to various matters, including certain environmental, health, safety, tax and other matters. The indemnification obligations relating to certain environmental, health and safety matters have a term of 17 years, while some of the other indemnification obligations have an indefinite term. The amount of the liability relating to all of these indemnification obligations included in other liabilities on our combined balance sheet at March 29, 2013 was \$22.4 million, of which \$18.3 million related to environmental, health and safety matters. The value of the environmental, health and safety indemnity was measured based on the probability-weighted present value of the costs expected to be incurred to address environmental, health and safety claims made under the indemnity. As of March 29, 2013, the maximum future payments we could be required to make under all of these indemnification obligations was \$75.7 million. We were required to pay \$30.0 million into an escrow account as collateral for all of these indemnification obligations to the purchaser, of which \$23.7 million remained in other assets on the condensed combined balance sheet at March 29, 2013.

We have recorded liabilities for known indemnification obligations included as part of environmental liabilities, which are discussed in note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements. In addition, we are liable for product performance; however, in the opinion of management, such obligations will not have a material adverse effect on our financial condition, results of operations and cash flows.

In addition, the separation and distribution agreement will provide for cross-indemnities principally designed to place financial responsibility for the obligations and liabilities of our business with us and financial responsibility for the obligations and liabilities of Covidien's remaining business with Covidien, among other

-103-

indemnities. Specifically, each of Covidien and Mallinckrodt will indemnify, defend and hold harmless the other party, its subsidiaries and their respective directors, officers, employees and agents against any losses arising out of or resulting from:

- the liabilities that each such party assumed or retained pursuant to the separation and distribution agreement (which, in the case of Mallinckrodt, would include the Mallinckrodt Liabilities (as defined below) and, in the case of Covidien, would include the Excluded Liabilities (as defined below)); and
- any breach by such party of the separation and distribution agreement or the other transaction agreements.

Also, we will indemnify, defend and hold harmless Covidien, its subsidiaries and their respective directors, officers, employees and agents from and against any losses arising out of or resulting from:

- except to the extent it relates to an Excluded Liability, the operation of our business;
- except to the extent it relates to an Excluded Liability, any guarantee, indemnification obligation, letter of credit reimbursement obligation, surety bond or other credit support agreement, arrangement, commitment or understanding for the benefit of Mallinckrodt or its subsidiaries by Covidien or any of its subsidiaries that survives following the distribution; and
- any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, with respect to all information contained in the Form 10 (as defined below), this information statement (as amended or supplemented), the offering memorandum for the April 2013 notes offering or any other disclosure document that describes the separation or the distribution or Mallinckrodt and its subsidiaries or primarily relates to the transactions contemplated by the separation and distribution agreement.

In addition, Covidien will indemnify, defend and hold harmless Mallinckrodt, its subsidiaries and their respective directors, officers, employees and agents from and against any losses arising out of or resulting from:

- Covidien's businesses other than the Pharmaceuticals business (except to the extent it relates to a Mallinckrodt Liability and other than the conduct of business, operations or activities for the benefit of Mallinckrodt or its subsidiaries pursuant to the separation and distribution agreement and the other transaction agreements); and
- the investigation and remediation of sites in Orrington, Maine and Penobscot River and Bay (as described in note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement).

The separation and distribution agreement also will specify procedures with respect to claims subject to indemnification and related matters.

### **Off-Balance Sheet Arrangements**

We are required to provide the NRC financial assurance demonstrating our ability to cover the cost of decommissioning our Maryland Heights, Missouri radiopharmaceuticals production facility upon closure, though we do not intend to close this facility. We have provided this financial assurance in the form of a \$58.0 million surety bond. In addition, as of March 29, 2013, we had a \$21.1 million letter of credit to guarantee decommissioning costs associated with our St. Louis, Missouri plant.

As of March 29, 2013, we had various other letters of credit and guarantee and surety bonds totaling \$20.5 million. In addition, at March 29, 2013, Covidien had outstanding letters of credit and guarantee and surety bonds totaling \$132.1 million, which supported multiple Covidien businesses, including our business.

-104-

# Concentration of Credit and Other Risks

Financial instruments that potentially subject us to concentrations of credit risk primarily consist of accounts receivable. We do not require collateral from customers; however, concentrations of credit risk with respect to trade receivables are generally limited due to our large number of customers and their diversity across geographic areas. A portion of our trade accounts receivable outside the U.S., however, include sales to government-owned or supported healthcare systems in several countries, which are subject to payment delays. Payment is dependent upon the financial stability of those countries' national governments. Deteriorating credit and economic conditions in parts of Western Europe, particularly in Spain and Italy, may continue to increase the average length of time it takes us to collect our accounts receivable in certain regions within these countries.

We routinely evaluate all government receivables for potential collection risks associated with the availability of government funding and reimbursement practices. While our accounts receivable, net of allowance for doubtful accounts in Greece, is insignificant, during fiscal 2012, we recorded a \$4.4 million charge to write down our outstanding accounts receivables in Greece. We have not incurred significant losses on any other government receivables; however, if the financial condition of customers or the countries' healthcare systems continue to deteriorate such that their ability to make payments is uncertain, additional allowances may be required in future periods.

Our accounts receivable, net of the allowance for doubtful accounts, in Spain and Italy at the end of each period are as follows:

(Dollars in Millions)	March 29, 2013	September 28, 2012	September 30, 2011	
Spain	\$ 15.7	\$ 15.0	\$ 26.6	
Italy	13.9	12.5	14.7	

Net sales to customers in Spain and Italy totaled \$26.3 million and \$29.0 million for the six months ended March 29, 2013 and March 30, 2012, respectively, and \$55.0 million, \$60.2 million and \$58.7 million for fiscal 2012, 2011 and 2010, respectively.

The following table shows net sales attributable to distributors that accounted for 10% or more of our total net sales:

	Six Mont	Six Months Ended			
	March 29, 2013	March 30, 2012	2012	2011	2010
Cardinal Health, Inc.	20%	18%	19%	19%	15%
McKesson Corporation	16%	12%	14%	13%	11%
AmerisourceBergen Corporation	7%	8%	9%	10%	8%

The following table shows accounts receivable attributable to distributors that accounted for 10% or more of our gross accounts receivable at the end of each period:

	March 29, 2013	September 28, 2012	September 30, 2011
Cardinal Health, Inc.	20%	19%	19%
McKesson Corporation	23%	20%	16%
AmerisourceBergen Corporation	9%	10%	12%

The following table shows net sales attributable to products that accounted for 10% or more of our total net sales:

	Six Month	s Ended	Fiscal				
	March 29, 2013	March 30, 2012	2012	2011	2010		
Optiray (CMDS)	14%	17%	17%	19%	17%		
Acetaminophen products (API)	10%	11%	11%	11%	10%		

Mo-99 is a key raw material in our Ultra-Technekow DTE technetium generators that are sold by our Global Medical Imaging segment. There are only eight suppliers of this raw material worldwide. We have agreements to obtain Mo-99 from three nuclear research reactors and we rely predominantly on two of these reactors for our Mo-99 supply. Accordingly, a disruption in the commercial supply or a significant increase in the cost of this material from these sources could have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

#### Management's Use of Non-GAAP Measures

Operational growth, a non-GAAP financial measure, measures the change in sales between current and prior year periods using a constant currency, the exchange rate in effect during the applicable prior year period. We have provided this non-GAAP financial measure because we believe it provides meaningful information regarding our results on a consistent and comparable basis for the periods presented. Management uses this non-GAAP financial measure, in addition to GAAP financial measures, to evaluate our operating results. It is also one of the performance metrics that determines management incentive compensation. This non-GAAP financial measure should be considered supplemental to and not a substitute for our reported financial results prepared in accordance with GAAP.

#### **Critical Accounting Policies and Estimates**

The preparation of our combined financial statements in conformity with accounting principles generally accepted in the U.S. requires management to use judgment in making estimates and assumptions that affect the reported amounts of assets, liabilities, revenue and expenses, and related disclosure of contingent assets and liabilities. The following accounting policies are based on, among other things, judgments and assumptions made by management that include inherent risks and uncertainties. Management's estimates are based on the relevant information available at the end of each period.

#### **Revenue Recognition**

We recognize revenue for product sales when title and risk of loss have transferred from us to the buyer, which may be upon shipment or upon delivery to the customer site, based on contract terms or legal requirements in non-U.S. jurisdictions. We sell products direct to retail pharmacies and end user customers and through distributors who resell the products to retail pharmacies, institutions and end user customers. We establish contracts with wholesalers, chain stores, government agencies, institutions, managed care organizations and GPOs that provide for rebates, sales incentives, Distribution Service Agreements ("DSAs") fees, fees for services and administration fees. Direct rebates and fees are paid based on direct customer's purchases from us, including DSA fees paid to wholesalers under our DSAs. Indirect rebates and fees are paid based on products purchased from a wholesaler under a contract with us. We enter into agreements with some indirect customers to establish contract pricing for certain products. These indirect customers then independently select a wholesaler from which to purchase the products at these contracted prices. Alternatively, we may enter into agreements with wholesalers at a contract price to offer our products to other indirect customers. Under either arrangement, we provide credit to the wholesaler for any difference between the contracted price with the indirect customer and the wholesaler's invoice price. Such credit is called a chargeback.

When we recognize net sales, we simultaneously record an adjustment to revenue for estimated chargebacks, rebates, product returns and other sales deductions. These provisions are estimated based upon: historical experience, estimated future trends, estimated customer inventory levels, current contracted sales terms with customers, level of utilization of our products and other competitive factors. We adjust reserves for rebates and chargebacks, product returns and other sales deductions to reflect differences between estimated and actual experience. Such adjustments impact the amount of sales we recognize in the period of adjustment.

Sales return reserves for new products are estimated and primarily based on our historical sales return experience with similar products, such as those within the same product line or those within the same or similar

-106-

therapeutic category. In limited circumstances, where the new product is not an extension of an existing product line or where we have no historical experience with products in a similar therapeutic category (such that we cannot reliably estimate expected returns), we would defer recognition of revenue until the right of return no longer exists or until we have developed sufficient historical experience to estimate sales returns. When establishing sales return reserves for new products, we also consider estimated levels of inventory in the distribution channel and projected demand.

The following table reflects activity in our sales reserve accounts (dollars in millions):

	Rebates and Chargebacks	Product Returns	Other Sales Deductions	Total
Balance at September 25, 2009	\$ 207.0	\$ 21.4	\$ 11.4	\$ 239.8
Provisions	1,164.3	36.2	57.3	1,257.8
Payments or credits	(1,166.0)	(25.1)	(56.8)	(1,247.9)
Balance at September 24, 2010	205.3	32.5	11.9	249.7
Provisions	1,218.8	40.5	47.1	1,306.4
Payments or credits	(1,200.1)	(39.1)	(45.7)	(1,284.9)
Balance at September 30, 2011	224.0	33.9	13.3	271.2
Provisions	1,085.9	30.0	41.9	1,157.8
Payments or credits	(1,077.7)	(29.2)	(42.3)	(1,149.2)
Balance at September 28, 2012	\$ 232.2	\$ 34.7	\$ 12.9	\$ 279.8

### Goodwill and Other Intangible Assets

*Goodwill*—In performing goodwill assessments, management relies on a number of factors including operating results, business plans, economic projections, anticipated future cash flows, and transactions and market place data. There are inherent uncertainties related to these factors and judgment in applying them to the analysis of goodwill impairment. Since judgment is involved in performing goodwill valuation analyses, there is risk that the carrying value of our goodwill may be overstated or understated. We calculate our goodwill valuations using an income approach based on the present value of future cash flows of each reporting unit. This approach incorporates many assumptions including future growth rates, discount factors and income tax rates. Changes in economic and operating conditions impacting these assumptions could result in goodwill impairment in future periods.

We test goodwill during the fourth quarter of each year for impairment, or more frequently if certain indicators are present or changes in circumstances suggest that impairment may exist. We utilize a two-step approach. The first step requires a comparison of the carrying value of the reporting units to the fair value of these units. We estimate the fair value of our reporting units through internal analyses and valuation, using an income approach based on the present value of future cash flows. If the carrying value of a reporting unit exceeds its fair value, we will perform the second step of the goodwill impairment to measure the amount of impairment loss, if any. The second step of the goodwill impairment test compares the implied fair value of a reporting unit's goodwill with its carrying value. To determine the implied fair value of goodwill, we allocate the fair value of a reporting unit to all of the assets and liabilities of that unit, including intangible assets, as if the reporting unit had been acquired in a business combination. Any excess of the value of a reporting unit over the amounts assigned to its assets and liabilities represents the implied fair value of goodwill. The results of our annual goodwill impairment test for fiscal 2012 showed that the fair value of each of our reporting units significantly exceeded their respective carrying values.

Other Intangible Assets—Intangible assets include completed technology, licenses, trademarks and in-process research and development ("IPR&D"). We record intangible assets at cost and amortize certain of such assets using the straight-line method over five to thirty years. We review intangible assets for impairment by

-107-

comparing the fair value of the assets, estimated using an income approach, with their carrying value. If the carrying value exceeds the fair value of the intangible asset, the amount recognized for impairment is equal to the difference between the carrying value of the asset and the present value of future cash flows. We assess the remaining useful life and the recoverability of finite-lived intangible assets whenever events or circumstances indicate that the carrying value of an asset may not be recoverable. Indefinite-lived intangible assets are tested for impairment at least annually.

#### Contingencies

We are involved, both as a plaintiff and a defendant, in various legal proceedings that arise in the ordinary course of business, including, without limitation, patent infringement, product liability and environmental matters, as further discussed in note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements. Accruals recorded for various contingencies, including legal proceedings, self-insurance and other claims, are based on judgment, the probability of losses and, where applicable, the consideration of opinions of internal and/or external legal counsel, internal and/or external technical consultants and actuarially determined estimates. When a range is established but a best estimate cannot be made, we record the minimum loss contingency amount. These estimates are often initially developed substantially earlier than the ultimate loss is known, and the estimates are reevaluated each accounting period, as additional information is available. When we are initially unable to develop a best estimate of loss, we record the minimum amount of loss, which could be zero. As information becomes known, additional loss provision is recorded when either a best estimate can be made or the minimum loss amount is increased. When events result in an expectation of a more favorable outcome than previously expected, our best estimate is changed to a lower amount. We record receivables from third-party insurers up to the amount of the related liability when we have determined that existing insurance policies will provide reimbursement. In making this determination, we consider applicable deductibles, policy limits and the historical payment experience of the insurance carriers. Receivables are not netted against the related liabilities for financial statement presentation.

#### Pension and Postretirement Benefits

Our pension expense and obligations are developed from actuarial valuations. Two critical assumptions in determining pension expense and obligations are the discount rate and expected long-term return on plan assets. We evaluate these assumptions at least annually. Other assumptions reflect demographic factors such as retirement, mortality and turnover and are evaluated periodically and updated to reflect our actual experience. Actual results may differ from actuarial assumptions. The discount rate is used to calculate the present value of the expected future cash flows for benefit obligations under our pension plans. For our U.S. plans, we use a broad population of Moody's AA-rated corporate bonds to determine the discount rate assumption. All bonds are non-callable, denominated in U.S. dollars and have a minimum amount outstanding of \$250 million. This population of bonds was used to generate a yield curve and associated spot rate curve, to discount the projected benefit payments for the U.S. plans. The discount rate is the single level rate that produces the same result as the spot rate curve. For our non-U.S. plans, the discount rate is generally determined by reviewing country- and region-specific government and corporate bond interest rates. A decrease in the discount rate increases the present value of pension benefit obligations and increases pension expense. A 50 basis point decrease in the discount rate would increase our present value of pension obligations by approximately \$34.0 million.

We consider the current and expected asset allocations of our pension plans, as well as historical and expected long-term rates of return on those types of plan assets, in determining the expected long-term return on plan assets. In determining the expected return on pension plan assets, we consider the relative weighting of plan assets by class and individual asset class performance expectations as provided by external advisors in reaching our conclusions on appropriate assumptions. The investment strategy for the pension plans has been governed by Covidien. Covidien's overall investment objective is to obtain a long-term return on plan assets that is consistent with the level of investment risk that is considered appropriate. Investment risks and returns are reviewed regularly against benchmarks to ensure objectives are being met. A 50 basis point decrease in the expected long-term return on plan assets would increase our annual pension expense by approximately \$1.9 million.

-108-

#### Income Taxes

In determining income for financial statement purposes, we must make certain estimates and judgments. These estimates and judgments affect the calculation of certain tax liabilities and the determination of the recoverability of certain of the deferred tax assets, which arise from temporary differences between the tax and financial statement recognition of revenue and expense.

Deferred tax assets are reduced by a valuation allowance if, based on the weight of available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized. In evaluating our ability to recover our deferred tax assets, we consider all available positive and negative evidence including our past operating results, the existence of cumulative losses in the most recent years and our forecast of future taxable income. In estimating future taxable income, we develop assumptions including the amount of future state, federal and international pre-tax operating income, the reversal of temporary differences, and the implementation of feasible and prudent tax planning strategies. These assumptions require significant judgment about the forecasts of future taxable income and are consistent with the plans and estimates we use to manage the underlying businesses.

We determine whether it is more likely than not that a tax position will be sustained upon examination. The tax benefit of any tax position that meets the more-likely-than-not recognition threshold is calculated as the largest amount that is more than 50% likely of being realized upon resolution of the uncertainty. To the extent a full benefit is not realized on the uncertain tax position, an income tax liability is established. We adjust these liabilities as a result of changing facts and circumstances; however, due to the complexity of some of these uncertainties, the ultimate resolution may result in a payment that is materially different from our current estimate of the tax liabilities. A significant portion of our potential tax liabilities are recorded in non-current income taxes payable, which is included in other liabilities on our combined balance sheets as payment is not expected within one year.

The calculation of our tax liabilities involves dealing with uncertainties in the application of complex tax regulations in a multitude of jurisdictions across our global operations. Changes in tax laws and rates could affect recorded deferred tax assets and liabilities in the future. Management is not aware of any such changes, however, which would have a material adverse effect on our competitive position, business, financial condition, results of operations and cash flows.

We believe that we will generate sufficient future taxable income in the appropriate jurisdictions to realize the tax benefits related to the net deferred tax assets on our combined balance sheets. However, any reduction in future taxable income, including any future restructuring activities, may require that we record an additional valuation allowance against our deferred tax assets. An increase in the valuation allowance would result in additional income tax expense in such period and could have a significant impact on our future earnings. Our income tax expense recorded in the future may also be reduced to the extent of decreases in our valuation allowances.

#### Quantitative and Qualitative Disclosure about Market Risk

Our operations include activities in the U.S. and countries outside of the U.S. These operations expose us to a variety of market risks, including the effects of changes in interest rates and currency exchange rates. We monitor and manage these financial exposures as an integral part of our overall risk management program.

### Interest Rate Risk

MIFSA has entered into a new revolving credit facility that bears interest at a floating rate. As a result, we will be exposed to fluctuations in interest rates to the extent of our borrowings under the revolving credit facility. Our long-term debt portfolio is expected to primarily consist of fixed-rate instruments.

-109-

# Currency Risk

We are exposed to currency exchange rate fluctuations that affect transactions not denominated in the functional currency of our U.S. and non-U.S. operations. We may from time to time use financial derivatives, which may include forward currency exchange contracts and currency options, to hedge this risk. However, gains and losses on these contracts would be offset by the gains or losses on the revaluation or settlement of the underlying transaction. We do not use derivative financial instruments to hedge investments in non-U.S. subsidiaries since such investments are long-term in nature.

### MANAGEMENT

### **Executive Officers Following the Separation**

Upon completion of the separation, none of our executive officers will be executive officers or employees of Covidien. The following table sets forth information regarding individuals who are expected to serve as our executive officers, including their positions after the separation.

Name	Age	Position
Mark Trudeau	51	President, Chief Executive Officer and Director
Matthew Harbaugh	42	Senior Vice President and Chief Financial Officer
Thomas Berry	62	Senior Vice President, Product Supply
Peter Edwards	51	Senior Vice President and General Counsel
Steve Carchedi	51	Senior Vice President and President, Commercial Operations (North
		America)
Meredith Fischer	60	Senior Vice President, Communications and Public Affairs
Stephen Merrick	52	Senior Vice President and President, Commercial Operations
		(International)
Ian Watkins	50	Senior Vice President and Chief Human Resources Officer

Mr. Trudeau will be named President and Chief Executive Officer of Mallinckrodt and is expected to serve on our board of directors. Mr. Trudeau joined the Pharmaceuticals segment of Covidien in February 2012 as a Senior Vice President and President of its Pharmaceuticals business. He joined Covidien from Bayer HealthCare Pharmaceuticals LLC USA, the U.S. healthcare business of Bayer AG, where he served as Chief Executive Officer. He simultaneously served as President of Bayer HealthCare Pharmaceuticals, the U.S. organization of Bayer's global pharmaceuticals business. In addition, he served as Interim President of the global specialty medicine business unit from January to August 2010. Prior to joining Bayer in 2009, Mr. Trudeau headed the Immunoscience Division at Bristol-Myers Squibb. During his 10-plus years at Bristol-Myers Squibb, he served in multiple senior roles, including President of the Asia/Pacific region, President and General Manager of Canada and General Manager/Managing Director in the United Kingdom. Mr. Trudeau was also with Abbott Laboratories, serving in a variety of executive positions, from 1988 to 1998. Mr. Trudeau holds a Bachelor's degree in chemical engineering and a M.B.A., both from the University of Michigan. Having worked as the President of Covidien's Pharmaceuticals business for over a year, Mr. Trudeau is familiar with all aspects of our business.

Mr. Harbaugh will be named Senior Vice President and Chief Financial Officer of Mallinckrodt. Mr. Harbaugh currently serves as Vice President, Finance of Covidien's Pharmaceuticals business, a position he has held since July 2008. He also served as Interim President of Covidien's Pharmaceuticals business from November 2010 to January 2012. Mr. Harbaugh joined Covidien's Pharmaceuticals business in August 2007 as its Vice President and Controller, Global Finance for the Global Medical Imaging business. Mr. Harbaugh was a Lead Finance Executive with Cerberus Capital Management, L.P. from April 2007 until August 2007. Mr. Harbaugh worked for Monsanto from 1997 to 2007 serving in senior U.S. roles in treasury, investor relations, financial planning and analysis and strategy in addition to two international assignments in Canada and Argentina.

Mr. Berry will be named Senior Vice President, Product Supply of Mallinckrodt. Mr. Berry currently serves as Vice President, Product Supply of Covidien's Pharmaceuticals business, a position he has held since February 2010. Mr. Berry was Senior Vice President of Global Manufacturing for the Fort Dodge Animal Health division of Wyeth Pharmaceuticals from October 2006 until February 2010.

-111-

Mr. Edwards will be named Senior Vice President and General Counsel of Mallinckrodt. Mr. Edwards joined Covidien's Pharmaceuticals business in May 2010 as Vice President and General Counsel. Mr. Edwards joined Covidien from the Solvay Group in Brussels, Belgium, where he served as Executive Vice President and General Counsel for the global pharmaceuticals business from June 2007 until April 2010.

Mr. Carchedi will be named Senior Vice President and President of Commercial Operations (North America) of Mallinckrodt. Mr. Carchedi joined Covidien's Pharmaceuticals business in October 2012 as Vice President and President of Commercial Operations (North America). Mr. Carchedi served from May 2010 to May 2012 as Chief Marketing Officer of General Electric Healthcare where he was responsible for leading worldwide marketing for GE's Medical Diagnostics business. From April 2009 to May 2010, Mr. Carchedi served as Senior Vice President in charge of the specialty pharmaceuticals business at Endo Pharmaceuticals. From May 2008 to April 2009, Mr. Carchedi served as Senior Vice President, Commercial Operations at Enzon Pharmaceuticals.

Ms. Fischer will be named Senior Vice President, Communications and Public Affairs of Mallinckrodt. Ms. Fischer joined Covidien's Pharmaceuticals business in February 2013 as Vice President, Communications and Public Affairs of Covidien's Pharmaceuticals business. Ms. Fischer was employed by Bayer Corporation from December 2001 until February 2013, where she served as Vice President of Communications and Public Policy for Bayer HealthCare and Bayer HealthCare Pharmaceuticals, North America. In that role, she supported Bayer HealthCare's U.S. pharmaceutical and animal health divisions and the company's global medical care and consumer care businesses.

Mr. Merrick will be named Senior Vice President and President of Commercial Operations (International) of Mallinckrodt. Mr. Merrick joined Covidien's Pharmaceuticals business in February 2013 as Vice President and President of Commercial Operations (International). Mr. Merrick was employed by Bristol-Myers Squibb Company, where he served as Vice President, Strategic Projects – Intercontinental Region from September 2012 until February 2013, President and General Manager – Brazil from December 2009 until September 2012 and as Vice President – Distributor Markets and Geographic Optimization from November 2007 until December 2009.

Mr. Watkins will be named Senior Vice President and Chief Human Resources Officer of Mallinckrodt. Mr. Watkins joined Covidien's Pharmaceuticals business in September 2012 as the Chief Human Resources Officer. Mr. Watkins served as Vice President, Global Human Resources at Synthes, Inc. from June 2007 to September 2012, which was recently acquired by Johnson & Johnson. Mr. Watkins served as Senior Vice President, Human Resources from 2003 to 2006 for Andrx Corporation, which is now part of Watson/Actavis.

#### **Board of Directors Following the Separation**

The following table sets forth information with respect to those persons who are expected to serve on our board of directors following the completion of the separation. We may name additional directors prior to completion of the separation.

Name	Age	Title
Melvin D. Booth	67	Chairman of the Board
Mark C. Trudeau	51	President, Chief Executive Officer and Director
David R. Carlucci	58	Director
J. Martin Carroll	63	Director
Diane H. Gulyas	56	Director
Nancy S. Lurker	55	Director
JoAnn A. Reed	57	Director
Kneeland C. Youngblood, M.D.	57	Director
Joseph A. Zaccagnino	66	Director

-112-

Mr. Booth has been a director of Catalent Pharma Solutions since 2010, a director of PRA International since 2004 and a director of eResearch Technologies since 2012. Mr. Booth has also been a strategic advisor in life sciences to Genstar Capital (a private equity firm) since 2005. Mr. Booth's previous public company board experience includes serving as Lead Director of Millipore, a life science research company, from 2004 to 2010, and as a member of the boards of MedImmune from 1998 to 2005 and of Human Genome Sciences from 1995 to 1998. Mr. Booth was President of MedImmune from 1998 until his retirement at the end of 2003. Mr. Booth was President of Human Genome Sciences from 1995 to 1998. He held a variety of domestic and international positions with Syntex from 1981 to 1995, including serving as President of its U.S. pharmaceuticals business. Mr. Booth has been active in U.S. pharmaceutical industry organizations and is a past Chairman of the Pharmaceuticals Manufacturers Association of Canada. Mr. Booth received a B.S. degree in accounting from Northwest Missouri State University where he was also awarded an honorary Doctor of Science degree. He is also a Certified Public Accountant. Mr. Booth's qualifications to serve on our board include his significant experience in leadership positions at pharmaceutical companies.

Mr. Carlucci was President and Chief Operating Officer of IMS Health from October 2002 until January 2005, when he was named Chief Executive Officer and President. He became Chairman and Chief Executive Officer the following year. Mr. Carlucci retired from IMS Health in December 2010. Mr. Carlucci held several senior executive level positions at IBM from 1976 to 2002, including operations and management positions in the U.S., Canada, Latin America and Asia Pacific. Mr. Carlucci has been a director and Chairman of the Human Resources and Compensation Committee for MasterCard International since 2006. Mr. Carlucci also served as a member of the advisory board of Mitsui USA, one of the world's most diversified comprehensive trading, investment and service companies. Mr. Carlucci received a B.A. in political science from the University of Rochester. Mr. Carlucci's qualifications to serve on our board include his significant experience as an executive and/or board member of publicly traded and private companies.

Mr. Carroll served as President and Chief Executive Officer of Boehringer Ingelheim Corporation and of Boehringer Pharmaceuticals, Inc. from 2003 until 2012. Mr. Carroll currently serves as the head of corporate strategy and development for Boehringer Ingelheim's U.S. operations and remains a director of Boehringer Ingelheim Corporation. Mr. Carroll joined the organization in 2002 as President of Boehringer Pharmaceuticals, Inc. Mr. Carroll worked at Merck & Company, Inc. from 1976 to 2001. From 1972 to 1976, Mr. Carroll served in the United States Air Force where he attained the rank of Captain. Mr. Carroll also serves as a director of Vivus, Inc. Mr. Carroll received a B.A. in accounting & economics from the College of the Holy Cross and a M.B.A. from Babson College. Mr. Carroll's qualifications to serve on our board include his significant experience in leadership positions at pharmaceutical companies.

Ms. Gulyas has worked at E. I. du Pont de Nemours and Company since 1978 and has been the President of DuPont's Performance Polymers division since 2009. She is also the Vice Chairman of the DuPont-Teijin Films global joint venture. From 2009 until 2012, Ms. Gulyas served as a director and as a member of the Finance Committee of Navistar International Corporation, a leading manufacture of commercial trucks, buses, RVs, defense vehicles and engines. Ms. Gulyas received her B.S. in chemical engineering from the University of Notre Dame. Ms. Gulyas' qualifications to serve on our board include her extensive executive experience with chemical and manufacturing companies.

Ms. Lurker has been serving as a director and Chief Executive Officer of PDI Inc. since 2008. Prior to joining PDI, Ms. Lurker served as Senior Vice President and Chief Marketing Officer of Novartis Pharmaceuticals Corporation from 2006 to 2008. Prior to that, she was President and Chief Executive Officer of ImpactRx, Inc. from 2003 to 2006. From 1998 to 2003, Ms. Lurker served as Group Vice President – Global Primary Care Products for Pharmacia Corporation. She was also a member of Pharmacia's U.S. Executive Management Committee from 1998 to 2003. Ms. Lurker began her career at Bristol-Myers Squibb, where she worked for 14 years. Ms. Lurker also has served as a director of Auxilium Corporation since 2011. Ms. Lurker served as a director of ConjuChem Biotechnologies, Inc. from 2006 and as a director of Elan Corporation from 2005 to 2006. Ms. Lurker received a B.S. magna cum laude in biology from Seattle Pacific University and a

-113-

M.B.A. from the University of Evansville. Ms. Lurker's qualifications to serve on our board include her significant experience in leadership positions at pharmaceutical companies.

Ms. Reed is a healthcare services consultant. Ms. Reed served as an advisor to the Chief Executive Officer of Medco Health Solutions from April 2008 to April 2009. From 2002 to March 2008, Ms. Reed served as Senior Vice President, Finance and Chief Financial Officer of Medco Health Solutions. From 1992 to 2002, she served as Senior Vice President, Finance of Medco Health Solutions. She joined Medco Containment Services, Inc. in 1988. Ms. Reed has been a director of American Tower Corporation since 2007, a director of Waters Corporation since 2006 and a trustee of St. Mary's College of Notre Dame since 2006. Ms. Reed received a B.B.A. in business administration from St. Mary's College. She received her M.B.A. in finance and international marketing cum laude from Fordham University. Ms. Reed's qualifications to serve on our board include her experience as a healthcare services consultant and her financial expertise experience and knowledge of financial statements, corporate finance and accounting matters.

Dr. Youngblood is a founding partner of Pharos Capital Group, a private equity firm that focuses on providing growth and expansion capital/buyouts in healthcare, business services and opportunistic investments. Dr. Youngblood served as a director of iStar Financial from 1998 to 2001, a director of Starwood Hotels and Resorts from 2001 to 2012, a director of Burger King Corporation from 2004 to 2010 and a director of the Gap Inc. from 2006 to 2012. Dr. Youngblood has been serving as a director of Energy Future Holdings Corp, an electric utility provider, since 2007. Dr. Youngblood is a physician by training, with over 15 years of experience in emergency medicine. He is also a member of the Council on Foreign Relations. Dr. Youngblood earned a B.A. in politics from Princeton University and an M.D. from the University of Texas Southwestern Medical School. Dr. Youngblood's qualifications to serve on our board include his extensive experience in healthcare practice, policy and business.

Mr. Zaccagnino, who has been a director of Covidien since its spin-off from Tyco International in 2007 and serves on its Compliance and Transactions Committees and as Chairman of the Nominating and Governance Committee. Mr. Zaccagnino has served as President, Chief Executive Officer and director of Yale New Haven Health System and its flagship Yale-New Haven Hospital from 1991 until his retirement in 2005. He has also served as a director of NewAlliance Bancshares, Inc. from 1991 until it was acquired in 2010. Mr. Zaccagnino has served on the board of the National Committee for Quality Healthcare from 1995 until 2005, and was elected Chairman of the Board in 2003. From 1999 until 2006 he served as a director and from 2004 to 2006 as Chairman of the Board of VHA Inc., a provider member cooperative of community owned health systems and their physicians which provides supply chain and group purchasing services through their subsidiaries, Novation and Provista. Mr. Zaccagnino received a B.S. (business administration) from the University of Connecticut and a M.P.H. (healthcare management) from Yale University School of Medicine. Mr. Zaccagnino's qualifications to serve on our board include his broad healthcare management and governance experience and his knowledge of healthcare policy and regulation, patient care delivery and financing and of clinical research and medical technology assessment, all of which will provide our board with unique insights and a keen perspective on the complexities of the healthcare sector and on the priorities of and challenges facing our company and the purchasers of our products.

At the time of completion of the separation, we expect that our board of directors will consist of the directors set forth above. At any meeting of shareholders for the election of directors at which a quorum is present, directors will be elected by the affirmative vote of a majority of the votes cast and will serve for one-year terms. Any nominee for director who does not receive a majority of the votes cast will not be elected to the board, except as described in "Description of Mallinckrodt's Share Capital—Election of Directors."

#### **Independence of Directors**

A majority of our board of directors will be comprised of directors who are "independent" as defined by the rules of the NYSE and the corporate governance guidelines to be adopted by the board. The criteria to be adopted by our board to assist it in making determinations regarding the independence of its members, summarized

below, are consistent with the NYSE listing standards regarding director independence. To be considered independent, the board will have to determine that a director does not have a material relationship, directly or indirectly, with Mallinckrodt. In assessing independence, the board will consider all relevant facts and circumstances. In particular, when assessing the materiality of a director's relationship with the company, the board will consider the issue not just from the standpoint of the director, but also from that of the persons or organizations with which the director has an affiliation. A director will not be considered independent if he or she, at the time of determination:

- is, or has been within the prior three years, an employee of Mallinckrodt or its subsidiaries;
- has an immediate family member who is, or has been within the prior three years, an executive officer of Mallinckrodt or its subsidiaries;
- is a current partner or employee of our auditor;
- has an immediate family member who is a current partner of our auditor or who is an employee of our auditor and personally works on our audit;
- has been, or has an immediate family member who has been, within the prior three years, a partner or employee of our auditor who personally worked on our audit during that time;
- is, or has an immediate family member who is, or has been within the prior three years, employed as an executive officer of a public company that has or had on the compensation committee of its board an executive officer of Mallinekrodt (during the same period of time);
- has, or has an immediate family member who has, received more than \$120,000 in direct compensation from Mallinckrodt, other than director and committee fees or other forms of deferred compensation for prior service (provided such compensation is not contingent in any way on continued service), in any 12-month period within the prior three years;
- is a current employee, or has an immediate family member who is a current executive officer, of a company that has made payments to, or received payments from, Mallinckrodt for property or services in an amount which, in any of the prior three fiscal years, exceeds the greater of \$1 million or 2% of such other company's consolidated gross revenues; or
- is, or his or her spouse is, an executive officer, director or trustee of a charitable organization to which Mallinckrodt's contributions, not including our
  matching of charitable contributions by employees, exceed, in any single fiscal year within the prior three years, the greater of \$1 million or 2% of
  such organization's total charitable receipts during that year.

The board will consider the independence of its members in light of these independence criteria. Based on these considerations, we expect that each of our directors, other than Mr. Trudeau, will satisfy the criteria. Each independent director is expected to notify the Chair of the Nominating and Governance Committee, as soon as reasonably practicable, of changes in his or her personal circumstances that may affect the board's evaluation of his or her independence.

#### **Director Nominations Process**

The Nominating and Governance Committee will be responsible for developing the general criteria, subject to approval by the full board, for use in identifying, evaluating and selecting qualified candidates for election or re-election to the board. The Nominating and Governance Committee will periodically review with the board the appropriate skills and characteristics required of board members in the context of the then-current make-up of the board. Final approval of director candidates will be determined by the full board, and invitations to join the board will be extended by the Chairman of the board on behalf of the entire board.

The Nominating and Governance Committee, in accordance with our corporate governance guidelines, will seek to create a board that is strong in its collective knowledge and has a diversity of backgrounds, skills and

-115-

experience with respect to accounting and finance, management and leadership, vision and strategy, business operations, business judgment, industry knowledge, corporate governance and global markets. When the Committee reviews a potential new candidate, the Committee will look specifically at the candidate's qualifications in light of the needs of the board and Mallinckrodt at that time, given the then-current mix of director attributes.

Our Corporate Governance Guidelines will provide that:

- directors should be individuals of the highest ethical character and integrity;
- directors should have demonstrated management ability at senior levels in successful organizations, including as the chief executive officer of a
  public company or as the leader of a large, multifaceted organization, including government, educational and other non-profit organizations;
- each director should have the ability to provide wise, informed and thoughtful counsel to senior management on a range of issues and be able to
  express independent opinions, while at the same time working as a member of a team;
- directors should be free from any conflict of interest or business or personal relationship that would interfere with the duty of loyalty owed to the company; and
- directors should be independent of any particular constituency and be able to represent all shareholders of the company.

The Committee will assess independence and also monitor compliance by the members of the board with the requisite qualifications under NYSE listing standards for populating the Audit, Compensation and Human Resources and Nominating and Governance Committees. Directors may not serve on more than four public company boards of directors (including Mallinckrodt) or, if the director is employed as chief executive officer of a publicly traded company, no more than three public company boards of directors (including Mallinckrodt). No person may stand for election as a director after reaching age 72.

Our articles of association will contain provisions that address the process by which a shareholder may nominate an individual to stand for election to the board of directors and establish certain qualifications for service as a director. The Nominating and Governance Committee's charter will include procedures by which the Committee will consider nominations submitted by shareholders.

The Nominating and Governance Committee will consider suggestions for director candidates from board members and, in its discretion, may employ a third-party search firm to assist in identifying candidates for director. In evaluating candidates for director, the Committee will use the guidelines described above, and will evaluate shareholder candidates in the same manner as candidates proposed from all other sources.

## **Committees of the Board of Directors**

Effective upon the completion of the separation, our board of directors will have the following standing committees: an Audit Committee, a Compensation and Human Resources Committee, a Nominating and Governance Committee and a Compliance Committee. Our board of directors will adopt a written charter for each of these committees, which will be posted on our website, www.mallinckrodt.com.

#### Audit Committee

The Audit Committee will monitor the integrity of our financial statements, the independence and qualifications of the independent auditors, the performance of our internal auditors and independent auditors, our compliance with certain legal and regulatory requirements and the effectiveness of our internal controls. The Audit Committee will be responsible for selecting, retaining, evaluating, setting the remuneration of and, if appropriate, recommending the termination of our independent auditors. The members of the Audit Committee

-116-

are expected to be Ms. Reed, Mr. Booth and Ms. Gulyas, each of whom is expected to be determined by the board to be independent under SEC rules and NYSE listing standards applicable to audit committee members. Additionally, at least one member of the Audit Committee is expected to be an audit committee financial expert under SEC rules and the NYSE listing standards applicable to audit committees. Ms. Reed is expected to serve as the Chair of the Audit Committee.

### **Compensation and Human Resources Committee**

The Compensation and Human Resources Committee will review and approve compensation and benefits policies and objectives, determine whether our officers and employees are compensated according to those objectives and carry out the board's responsibilities relating to the compensation of our executives. The members of the Compensation and Human Resources Committee are expected to be Mr. Carlucci, Ms. Gulyas and Ms. Lurker, each of whom is expected to be determined by the board to be independent under SEC rules and NYSE listing standards applicable to compensation committee members. Mr. Carlucci is expected to serve as the Chair of the Compensation and Human Resources Committee.

#### Nominating and Governance Committee

The Nominating and Governance Committee will be responsible for identifying individuals qualified to become board members, recommending to the board the director nominees for election at the Annual General Meeting, developing and recommending to the board a set of corporate governance guidelines, and taking a general leadership role in our corporate governance. The members of the Nominating and Governance Committee are expected to be Mr. Zaccagnino, Mr. Carroll and Dr. Youngblood, each of whom is expected to be determined by the board to be independent under NYSE listing standards. Mr. Zaccagnino is expected to serve as the Chair of the Nominating and Governance Committee.

#### **Compliance** Committee

The Compliance Committee will assist the board in fulfilling its oversight responsibility with respect to regulatory, healthcare compliance and public policy issues that affect us. The members of the Compliance Committee are expected to be Mr. Carroll, Dr. Youngblood and Mr. Zaccagnino, each of whom is expected to be determined by the board to be independent under NYSE listing standards. Mr. Carroll is expected to serve as the Chair of the Compliance Committee.

#### **Compensation Committee Interlocks and Insider Participation**

During fiscal 2012, Mallinckrodt was not an independent company, and did not have a compensation committee or any other committee serving a similar function. Decisions as to the compensation of those who currently serve as our executive officers were made by Covidien, as described in "Compensation Discussion and Analysis."

### **Board Leadership Structure**

At completion of the separation, the positions of Chairman of the Board and Chief Executive Officer will be held by separate people. The Chairman of the Board will provide leadership to the board and work with the board to define its structure and activities in the fulfillment of its responsibilities. The Chairman of the Board will set the board agendas with board and management input, facilitate communication among directors, provide an appropriate information flow to the board and preside at meetings of the board of directors and shareholders. The Chairman of the Board will work with other board members to provide strong, independent oversight of the company's management and affairs. Future modification of the board leadership structure will be made at the sole discretion of our board of directors. A more detailed description of the role and responsibilities of the Chairman of the Board will be set forth in our Corporate Governance Guidelines.

#### **Corporate Governance Guidelines**

Our board will adopt governance guidelines designed to assist the company and our board in implementing effective corporate governance practices. The governance guidelines will be reviewed regularly by the Nominating and Governance Committee in light of changing circumstances in order to continue serving our best interests and the best interests of our shareholders.

#### **Code of Ethics**

We will adopt a Guide to Business Conduct, which will apply to all of our employees, officers and directors and will meet the requirements of a "code of ethics" as defined by SEC regulations. The Guide to Business Conduct also will meet the requirements of a code of business conduct and ethics under the listing standards of the NYSE. The Guide to Business Conduct will be posted on our website, www.mallinckrodt.com. We will disclose any material amendments to the Guide to Business Conduct, as well as any waivers for executive officers or directors, on our website.

#### **Board Risk Oversight**

Our board of directors will oversee an enterprise-wide approach to risk management designed to support the achievement of organizational objectives, including strategic objectives, to improve long-term organizational performance and enhance shareholder value. A fundamental part of risk management is not only understanding the risks we face and what steps management is taking to manage those risks, but also understanding what level of risk is appropriate for us. The involvement of the full board of directors in setting our business strategy is a key part of its assessment of management's appetite for risk and the determination of what constitutes an appropriate level of risk for the company. In this process, risk is assessed throughout the business, focusing on three primary areas of risk: financial risk, legal/compliance risk and operational/strategic risk.

While the board of directors will have the ultimate oversight responsibility for the risk management process, various committees of the board also will have responsibility for risk management. In particular, the Audit Committee will focus on financial risk, including internal controls, and will receive an annual risk assessment report from our internal auditors. Our Compliance Committee will assist the board of directors in fulfilling its oversight responsibility with respect to regulatory, healthcare compliance and public policy issues that affect us and work closely with our legal and regulatory groups. In addition, in setting compensation, the Compensation and Human Resources Committee will strive to create incentives that encourage a level of risk-taking behavior consistent with our business strategy.

### **Communications with the Board of Directors**

The board will establish a process for interested parties to communicate with members of the board. If you have a concern, question or complaint regarding our compliance with any policy or law, or would otherwise like to contact the board, you will be able to reach the board via email. A direct link to this email address will be found on our website. You also will be able to submit communications in writing to a special address or by phone to a toll-free number that will be published on our website. You will be able to submit inquiries anonymously and confidentially. All concerns and inquiries will be received and reviewed promptly by the Office of the General Counsel. Any significant concerns relating to accounting, internal controls or audit matters will be reviewed with the Audit Committee.

All concerns will be addressed by the Office of the General Counsel unless otherwise instructed by the Audit Committee or the Chairman of the Board. The status of all outstanding concerns will be reported to the Chairman of the Board and the Audit Committee on a quarterly basis, and any concern that is determined to (1) pose an immediate threat to the company or (2) concern a senior company official (any executive officer or any direct report to the President and Chief Executive Officer) will be immediately communicated to the Chair of

-118-

the Audit Committee. The Chairman of the Board or the Audit Committee will determine if certain matters should be presented to the full board and will be able to direct the retention of outside counsel or other advisors in connection with any concern addressed to them. Our Guide to Business Conduct will prohibit any employee from retaliating against anyone for raising or helping to resolve an integrity question.

#### **Application of Non-U.S. Corporate Governance Codes**

Our corporate governance guidelines and general approach to corporate governance as reflected in our memorandum and articles of association and our internal policies and procedures are guided by U.S. practice and applicable federal securities laws and regulations and NYSE requirements. Although we are an Irish public limited company, we are not subject to the listing rules of the Irish Stock Exchange or the listing rules of the U.K. Listing Authority and we are therefore not subject to, nor have we adopted, the U.K. Corporate Governance Code or any other non-statutory Irish or U.K. governance standards or guidelines. While there are many similarities and overlaps between the U.S. corporate governance standards applied by us and the U.K. Corporate Governance Code and other Irish/U.K. governance standards or guidelines, there are differences, in particular relating to the extent of the authorization to issue share capital and effect share repurchases that may be granted to the board and the criteria for determining the independence of directors.

-119-

## COMPENSATION DISCUSSION AND ANALYSIS

The Pharmaceuticals business is currently part of Covidien and not an independent company and the Mallinckrodt Compensation and Human Resources Committee (our "Compensation Committee") has not yet been formed. Decisions as to the past compensation of those who currently serve as executive officers of the Pharmaceuticals business, and who will serve as our named executive officers upon the separation, have been made by Covidien. This Compensation Discussion and Analysis discusses these historical compensation practices and describes certain aspects of our anticipated compensation structure for our named executive officers following the separation. While we have discussed our anticipated programs and policies with Covidien and the Compensation and Human Resources Committee of the Covidien board of directors ("Covidien Compensation Committee"), our Compensation Committee may decide to change such policies and programs following the completion of the separation.

For purposes of the following Compensation Discussion and Analysis and executive compensation disclosures, the individuals listed below are referred to collectively as our "named executive officers." They are our President and Chief Executive Officer, our Chief Financial Officer and our other three most highly compensated executive officers, based on fiscal 2012 compensation from Covidien.

- *Mark Trudeau, Mallinckrodt President and Chief Executive Officer*. Prior to completion of the separation, Mr. Trudeau served as President of Covidien's Pharmaceuticals business.
- *Matthew Harbaugh, Mallinckrodt Senior Vice President and Chief Financial Officer*. Prior to completion of the separation, Mr. Harbaugh served as Vice President, Finance of Covidien's Pharmaceuticals business.
- Thomas Berry, Mallinckrodt Senior Vice President, Product Supply. Prior to completion of the separation, Mr. Berry served as Vice President, Operations of Covidien's Pharmaceuticals business.
- David Silver, Mallinckrodt Senior Vice President, Portfolio Management, Strategy and Business Development and Licensing\*. Prior to completion of the separation, Mr. Silver served as Vice President, Strategy and Portfolio Management of Covidien's Pharmaceuticals business.
- *Peter Edwards, Mallinckrodt Senior Vice President and General Counsel.* Prior to completion of the separation, Mr. Edwards served as Vice President and General Counsel of Covidien's Pharmaceuticals business.

Additional information about our expected senior executive team following the separation is set forth in "Management—Executive Officers Following the Separation." Initially, our compensation policies will be largely the same as those adopted by Covidien. Our Compensation Committee will review these policies and, it is expected, will make adjustments to support our strategies and to remain competitive in the marketplace.

The following sections of this Compensation Discussion and Analysis describe Covidien's compensation philosophy, policies and practices as they applied to our named executive officers listed above during fiscal 2012.

### Introduction

### Historically

Covidien and the Covidien Compensation Committee have established a compensation philosophy that is designed to attract, retain and motivate its executive officers. The core principles of that compensation philosophy are as follows:

- Compensation should strongly align the interests of executive officers and shareholders.
- Compensation should support effective governance.

\* Mr. Silver will be terminating his employment with the Pharmaceuticals business prior to the completion of the separation. None of the other executive officers of the Pharmaceuticals business had total compensation in excess of \$100,000 in fiscal 2012.

-120-

- Compensation should be based on a total rewards perspective with an explicit role for each element.
- Compensation should be competitive, but not excessive, in order to attract and retain talented executive officers who can achieve Covidien's long-term strategic goals and create shareholder value.
- Compensation should support Covidien's business strategy in the areas of customer focus, globalization, operational excellence and innovation, as well as Covidien's talent strategy.
- The reward elements should be balanced, with an emphasis on performance-based compensation.
- Compensation goals and practices should be transparent and easy to communicate, both internally and externally.
- Target setting is a key activity and should be done in a rigorous manner resulting in targets that reflect stretch, yet are achievable.

There are three major components to Covidien's executive compensation program: base salary, annual incentive compensation, and long-term incentive awards. All of these components are designed to work together to drive a complementary set of behaviors and outcomes.

Base salary. Base salary is intended to reflect the market value of the executive officer's role, with differentiation for individual capability and experience.

Annual incentive compensation. Annual incentive compensation in the form of a market-competitive, performance-based cash bonus is designed to focus executive officers on pre-set objectives each year and drive specific behaviors that foster short- and long-term growth and profitability.

Long-term incentive compensation. Long-term incentive compensation, which consists of awards of stock options, restricted units and performance units, is designed to recognize executive officers for their contributions to Covidien, to highlight the strategic significance of each executive's role, to promote retention and to align the interests of executive officers with the interests of shareholders in long-term growth and stock performance, rewarding executive officers for shareholder value creation.

## **Going Forward**

*Base salary*. Our Compensation Committee will establish the base salary for named executive officers after the separation. We expect any adjustments to base salary will be reflective of factors such as each named executive officer's post-separation level of responsibility and market data for similar positions at companies in our peer group. A discussion of our peer group is contained in "—How Executive Pay Decisions Are Made—Going Forward—Peer Group."

Annual incentive compensation. In connection with the separation, we will adopt an annual incentive plan with terms that are expected to be similar to those of Covidien's annual incentive plan. Following the separation, our Compensation Committee will establish performance goals and target bonus opportunities for our named executive officers that are consistent with the then-current market practices and competitive market levels and that are based on our peer group.

Long-term incentive awards. Prior to completion of the separation, we will adopt, subject to the approval of our current shareholders, the Mallinckrodt Pharmaceuticals Stock and Incentive Plan ("Mallinckrodt SIP"), which will be substantially similar to the Covidien Stock and Incentive Plan. The Mallinckrodt SIP will permit us to grant stock options, stock appreciation rights, restricted stock, restricted units, performance units, other share-based awards and cash awards. The values of long-term incentive compensation awards issued to named executive officers following the separation are expected to be set based on each named executive officer's post-separation level of responsibility and market data for similar positions at companies in our peer group, which is set forth under "—How Executive Pay Decision Are Made—Going Forward—Peer Group."

-121-

# How Executive Pay Decisions Are Made

# Historically

As noted above, during fiscal 2012, the named executive officers participated in Covidien's executive compensation programs. In determining executive compensation packages for fiscal 2012, Covidien and the Covidien Compensation Committee sought to strike an appropriate balance between fixed and variable compensation and between short- and long-term compensation. Because Covidien believes that making a significant portion of its named executive officers' compensation variable and long-term supports its pay-for-performance executive compensation philosophy, the majority of compensation is provided in the form of long-term incentive compensation (*i.e.*, equity awards). Covidien believes this encourages strategies and levels of risk-taking that correlate with the long-term best interests of Covidien and its shareholders. Covidien emphasizes share-based compensation, in combination with executive share ownership guidelines, to promote long-term incentive compensation program includes a mix of vehicles to mitigate the risk of over-emphasis on any one element and includes a cap on performance units. Covidien equity awards include claw-back provisions which apply to certain monetary gains on equity grants realized by executives whose employment is terminated for cause. Finally, in assessing the contributions of a particular named executive officer, Covidien and the Covidien Compensation Committee look not only to results-oriented performance, but also to how those results were achieved—whether the decisions and actions leading to the results were consistent with Covidien values—and the long-term impact of those decisions. Based on these principles, Covidien and, where applicable, the Covidien Compensation Committee established the compensation payable to the named executive officers as described below.

Covidien utilizes a Talent and Leadership Review ("TLR") process to manage its talent and organizational capability with the goal of maximizing organizational excellence and business success. As part of the TLR process, each employee's manager, in conjunction with a human resources representative, assigns to each employee a rating on two discrete dimensions: leadership competencies and results. For fiscal 2012, three possible ratings could be assigned in each of these two dimensions: exceptional, effective and not yet effective. These performance ratings impact base salary decisions, as well as decisions regarding the individual award target established for the employee pursuant to the annual incentive plan and the value of long-term incentive compensation awards.

Mr. Trudeau, who is a named executive officer of Covidien for fiscal 2012 and who is expected to be a named executive officer of Mallinckrodt, commenced employment with Covidien during fiscal 2012. As a new hire, Mr. Trudeau's compensation was not established through the TLR process—as a Covidien named executive officer, his compensation was set by the Covidien Compensation Committee. To establish the compensation payable to Mr. Trudeau, the Covidien Compensation Committee considered a market study prepared by its independent compensation consultant, Steven Hall & Partners. This market study included information regarding base salary, annual cash incentive awards and the value of equity awards and compiled data derived from a number of sources, including the 2011 Radford Global Technology Survey, the 2011 Towers Watson U.S. General Industry Executive Database, the 2011 Hewitt U.S. General Industry/Retail Total Compensation Measurement, the Towers Watson 2010/2011 Survey Report on Top Management Compensation, and the 2011 U.S. Mercer Benchmark Database—Executive. The Covidien Compensation Committee's independent compensation consultant weighted each of these surveys based on company revenue and industry in order to utilize survey data for companies that replicate, in particular, the revenue generated by Covidien's Pharmaceuticals business. Given the anticipated separation and the hiring of Mr. Trudeau to serve as our President and Chief Executive Officer upon separation, the Covidien Compensation Committee then established Mr. Trudeau's compensation based on the results of that process and with the explicit intent to allow for increases in such compensation upon his becoming our President and Chief Executive Officer and to provide our Compensation Committee latitude in establishing Mr. Trudeau's compensation.

-122-

Mr. Harbaugh served as interim-President of Covidien's Pharmaceuticals business from December 2, 2010 through February 1, 2012. At the time he was appointed interim-President, the Covidien Compensation Committee established a supplemental compensation package to reflect his increased responsibilities. This supplemental compensation was provided to Mr. Harbaugh after his compensation for serving as Vice President, Finance of Covidien's Pharmaceuticals business was set through the TLR process and only for the period of time that he performed additional services as interim-President.

The following discusses the decision-making criteria for each component of compensation.

*Base Salary*. With respect to named executive officers other than Mr. Trudeau, base salary for fiscal 2012 was established through the TLR process and was based on individual performance and an assessment of the value of the individual to Covidien, particularly given the pending separation. Mr. Harbaugh received an additional monthly allowance of \$10,000 in his base salary as part of the supplemental compensation package approved by the Covidien Compensation Committee. The Covidien Compensation Committee established Mr. Trudeau's base salary in connection with his hiring utilizing the process discussed above.

Annual Incentive Compensation. During fiscal 2012, each named executive officer participated in the Covidien 2012 Annual Incentive Plan ("Covidien 2012 AIP"), which is a component of the Covidien 2007 Stock and Incentive Plan. At the beginning of the fiscal year, the Covidien Compensation Committee established performance measures and goals, which included various core financial and strategic focus metrics, performance targets for each metric, including minimum threshold performance requirements to earn an award, and maximum performance scores. As discussed under the heading "2012 Annual Incentive Awards" below, each named executive officer had core financial metrics of sales growth and operating income of Covidien's Pharmaceuticals business and each named executive officer other than Mr. Trudeau had a strategic focus metric which was based on core competencies and individual performance goals, while Mr. Trudeau's strategic focus metric was based on the gross margin of Covidien's Pharmaceuticals business. Covidien set individual award targets, expressed as a percentage of base salary, for each named executive officer, other than Messrs. Trudeau and Harbaugh, based on the executive's level of responsibility and performance review through the TLR process. The individual award target for Mr. Trudeau was set by the Covidien Compensation Committee in connection with his hiring utilizing the process discussed above, while the individual award target for Mr. Harbaugh was set as part of the supplemental compensation package approved by the Covidien Compensation Committee.

After the close of the fiscal year, the Covidien Compensation Committee received a report from management regarding the performance of Covidien's Pharmaceuticals business against the pre-established performance goals. Awards were based on each named executive officer's individual award target percentage and the performance of Covidien's Pharmaceuticals business relative to the specific performance goals, as certified by the Covidien Compensation Committee, and, with respect to named executive officers other than Mr. Trudeau, considering attainment of each officer's individual performance goals.

*Long-Term Incentive Compensation*. During fiscal 2012, named executive officers were eligible to receive long-term incentive compensation awards for each named executive officer other than Mr. Trudeau, the Covidien Compensation Committee considered the recommendations of Covidien's management, which were based upon individual performance, including TLR performance ratings, the officer's total compensation and mix of compensation for the previous fiscal year, the resulting compensation mix projected for fiscal 2012, the officer's level of responsibility and previous equity grants. With respect to Mr. Trudeau, the Covidien Compensation committee available of the long-term incentive award, which it then adjusted to reflect both a prorated amount based upon the number of days during fiscal 2012 that Mr. Trudeau was employed by Covidien as well as amounts that he forfeited upon leaving his prior employer. With respect to Mr. Harbaugh, as part of the supplemental compensation package, the Covidien Compensation Committee approved the accumulation of an additional \$40,000 for each month that he served as interim-President, with such accumulated amount to be delivered in the form of a long-term incentive compensation award at the time of the next following annual equity award cycle and at the time he ceased

-123-

serving as interim-President. During fiscal 2012, in addition to the annual long-term incentive compensation award which was determined through the TLR process, the Covidien Compensation Committee awarded to Mr. Harbaugh supplemental long-term incentive compensation awards with a total value reflective of his entire period of service as interim-President.

### **Going Forward**

The executive compensation programs that we initially adopt will be similar to those in place at Covidien immediately prior to completion of the separation. Following the separation, our Compensation Committee will continue to consider and develop our compensation structure, practices, and procedures in order to effectively meet our business needs and goals.

Compensation Consultant. Our Compensation Committee will engage an independent compensation consultant to assist it with the review and development of our compensation structure, practices and procedures.

*Peer Group.* The Covidien Compensation Committee utilized a Mallinckrodt peer group, which it developed with the assistance of its independent compensation consultant, Steven Hall & Partners, to set compensation payable to individuals hired in connection with the separation and who Covidien retained to serve as employees of Mallinckrodt following the separation. Steven Hall & Partners did not provide any other services to the Covidien Compensation Committee identified our peer group based on similar criteria used for selecting the Covidien peer group, namely that the company is in the same industry (for this purpose, the pharmaceuticals industry) and has revenue of between one-half and two times the revenue generated by Covidien's Pharmaceuticals business. The companies listed below comprise the Mallinckrodt peer group utilized by the Covidien Compensation Committee.

- Actavis, Inc. (formerly Watson Pharmaceuticals, Inc.)
- Endo Health Solutions, Inc.
- Forest Laboratories, Inc.
- Hospira, Inc.

- Medicis Pharmaceutical Corp.
- Par Pharmaceutical Companies, Inc.
- Perrigo Company
- Valeant Pharmaceuticals International, Inc.
- Warner Chilcott Ltd.

Going forward, we expect that our Compensation Committee will review the peer group utilized by Covidien and determine, with the assistance of its independent compensation consultant, whether to continue with such peer group or modify it as it deems appropriate.

#### 2012 Annual Incentive Awards

### **Historically**

Covidien's payment of fiscal 2012 annual incentive awards to the named executive officers was subject to the achievement of core financial and strategic focus metrics established pursuant to the Covidien 2012 AIP. For fiscal 2012, there were two core financial metrics which were weighted 35% each and which accounted for, in the aggregate, 70% of the performance multiplier. The strategic focus metric accounted for the remaining 30% of the performance multiplier. The following describes the core financial and strategic focus metrics applicable to each named executive officer for fiscal 2012 as well as the process employed by Covidien to calculate the performance multiplier and final payouts to named executive officers under the Covidien 2012 AIP.

Core Financial Metrics. The two core financial metrics for fiscal 2012 were operating income and sales growth of Covidien's Pharmaceuticals business.

Strategic Focus Metric. The strategic focus metric for Mr. Trudeau was gross margin of Covidien's Pharmaceuticals business. The strategic focus metric for the other named executive officers consisted of core competencies established by Covidien and individual performance goals approved by the manager of each named executive officer according to the process described below.

-124-

At the start of fiscal 2012, Covidien established six core competencies, which are company-wide initiatives that Covidien utilizes to assess a portion of certain employees' performance during fiscal 2012. Also at the start of fiscal 2012, each named executive officer's manager established goals for the departments over which such executive has responsibility. After these departmental goals were established, each named executive officer proposed his own individual performance goals which supported and furthered the various departmental goals and assigned to each goal a particular weighting, with the total weighting equaling 100%. Each named executive officer's manager then reviewed and approved the individual performance goals and weightings proposed by the named executive officer after either adjusting the goals and/or weightings to further refine the objective of supporting the departmental goals or accepting the goals and/or weightings proposed by the named executive officer.

The following chart summarizes the Covidien 2012 AIP design, including the performance targets and performance scores for the core financial metrics for each named executive officer as well as the performance target and performance score for the strategic focus metric for Mr. Trudeau. Please refer to the discussion that immediately follows this chart for more detail regarding the calculation of the performance scores for the strategic focus metric for named executive officers other than Mr. Trudeau, as well as the final payout under the Covidien 2012 AIP for each named executive officer.

## Fiscal 2012 Annual Incentive Plan Design Summary

Executive Officer	Performance Metric	Weight	Performance Target <sup>(1)</sup> (dollars i		Performance Results in millions)		Performance Multiplier	Weighted Performance Score
Mark Trudeau	Operating Income			,				
	(Pharmaceuticals)	35%	\$	350	\$	360	1.283x	45%
	Sales Growth	250/		2 (0/		2.20/	1 201	420/
	(Pharmaceuticals) Gross Margin	35%		2.6%		3.3%	1.201x	42%
	(Pharmaceuticals)	30%		43.6%		45.6%	2x	60%
	(1 nurmaceuteuts)	5070		45.070		45.070		
Performance Multiplier and Score Total							1.47x	147%
Matthew Harbaugh	Operating Income							
Thomas Berry	(Pharmaceuticals)	35%	\$	350	\$	360	1.283x	N/A
David Silver	Sales Growth							
Peter Edwards	(Pharmaceuticals)	35%		2.6%		3.3%	1.201x	N/A
Performance Multiplier for Core Financial Metric					1.242x	N/A		

The performance metrics used for compensation purposes include non-GAAP financial measures which exclude the effects of anticipated one-time, generally non-recurring items which the Covidien Compensation Committee believes may mask the underlying operating results and/or business trends of the business segment. The categories of these anticipated extraordinary items are identified at the beginning of the fiscal year when the performance measure is approved and, for the Covidien 2012 AIP, included certain restructuring charges, revenue adjustments related to businesses exited or sold, acquisitions, goodwill or other intangible asset impairment charges, shareholder and other litigation charges and certain legacy tax matters.

For the Covidien 2012 AIP, the performance targets were calculated as follows:

- Operating income is the operating income of Covidien's Pharmaceuticals business, calculated using the currency exchange rate applied in setting Covidien's Pharmaceuticals business's annual operating plan in order to eliminate the effect of currency exchange rate fluctuations.
- Sales growth is the total change in net trade sales for fiscal 2012 in U.S. dollars, calculated using fiscal 2011 currency exchange rates divided by fiscal 2011 net trade sales.
- Gross margin is gross margin dollars divided by net sales dollars, where gross margin dollars is calculated by adjusting sales primarily for product costs, variances in plant, freight costs, royalties, warehousing, inventory adjustments and currency exchange rate fluctuations.

-125-

The operating income and sales growth of Covidien's Pharmaceuticals business exceeded the Covidien 2012 AIP target performance level, while gross margin exceeded the maximum performance level. Payout under the Covidien 2012 AIP to Mr. Trudeau was made at 147% of target performance level (*i.e.*, by application of a performance multiplier of 1.47).

With respect to the other named executive officers, a preliminary payout under the Covidien 2012 AIP was determined solely on the results of the core financial metrics—that is, it was based on a performance multiplier of 1.242, which represents an equal weighting of the 1.283 and 1.201 performance multipliers for the operating income and sales growth core financial metrics, respectively. Accordingly, 70% of the fiscal 2012 payout for named executive officers other than Mr. Trudeau was based on a performance multiplier of 1.242, while the remaining 30% was based upon the performance multiplier for the strategic focus metric, determined through the process described below.

As stated above, the strategic focus metric for named executive officers other than Mr. Trudeau consisted of core competencies established by Covidien and individual performance goals approved by each named executive officer's manager. For purposes of the following discussion regarding the calculation of the strategic focus metric, named executive officer refers to all of the named executive officers other than Mr. Trudeau.

For fiscal 2012, Covidien established the following six core competencies:

- Adaptability
- Creative Problem Solving
- Cross-Cultural Respect

- Customer Focus
- Drive for Results
- Interpersonal Relationships

For fiscal 2012, the individual performance goals approved for named executive officers by his respective manager were as follows:

### Mr. Harbaugh

- Meet or Exceed Budget Commitments
- Drive Global Customer Focus

### Mr. Berry

- Achieve Target Objectives for Manufacturing Dashboard
- Reduced Operations Cost
- Complete Facility Restructuring

## Mr. Silver

- Support the Spin-Off Transaction
- Coordinate and Prepare Portfolio Business Review and Strategic
   Plan
- Shepherd Pipeline Products to Registration

## Mr. Edwards

- Build-Out Legal Team
- Support Transition of New President
- Sustainable Productivity

- Sustainable Productivity
- Develop Global Leaders and Capabilities
- Achieve Key Project Milestones
- Improve Supply and Operations Process
- Oversee Market Research and Commercial Analytics
- Effectively Manage the Specialty Pharmaceuticals Generics Business
- Support the Spin-Off Transaction
- Promote Diversity Initiatives
- Maintain High-Quality Services During Spin

-126-

Immediately after the conclusion of fiscal 2012, each named executive officer's manager conducted a performance evaluation for such executive officer by assessing the executive officer's performance during fiscal 2012 against each of the six core competencies and each of the respective individual performance goals. During this process, each named executive officer's manager categorized the respective executive officer's performance as either exceeding, achieving, partially achieving or not achieving the stated objective. Each of these categories was assigned a numerical score, with the numerical scores assigned to each of the six core competencies being equally weighted and the numerical scores assigned to each of the underlying individual performance goals being weighted according to the predetermined weighting approved by the named executive officer's manager at the beginning of fiscal 2012. Once the scores for the core competencies and individual performance goals were calculated, they were weighted equally to determine a preliminary performance multiplier for the strategic focus metric component of the Covidien 2012 AIP. At that time, Covidien also calculated a preliminary payout for each named executive officer based on both the core financial metrics and the strategic focus metric. Each named executive officer's manager then reviewed the preliminary payout and adjusted, if appropriate, the amount of the payout based on individual performance, additional contributions by the named executive officer that were not captured within the parameters of the core competencies or individual performance goals, and the amount of the payout calculated solely based on the core financial metrics in order to align more closely the final payout with the financial performance of Covidien's Pharmaceuticals business.

The following chart lists the performance multiplier for the core financial metrics only, the payout based only on the performance multiplier for the core financial metrics and the strategic focus metric, the preliminary payout amount determined by application of the performance multiplier for both the core financial metrics and the strategic focus metric and the final payout made to each named executive officer. The chart also lists, for Mr. Trudeau, the performance multiplier applicable to his payout and his final payout amount.

Executive Officer	Performance Multiplier for CFM Only	Pe Mu	t Based on CFM erformance ittiplier Only ided Amount")	Performance Multiplier for CFM and SFM	CF Pe	ninary Payout Based on M and SFM erformance Multiplier	2012 Annual ntive Payout
Mark Trudeau	N/A		N/A	1.47x		N/A	\$ 507,252
Matthew Harbaugh	1.242x	\$	205,547	1.17x	\$	193,539	\$ 205,543
Thomas Berry	1.242x	\$	199,705	1.17x	\$	188,039	\$ 188,039
David Silver	1.242x	\$	148,632	1.17x	\$	139,949	\$ 149,998
Peter Edwards	1.242x	\$	181,825	1.32x	\$	193,167	\$ 181,825

Pursuant to the terms of the Covidien 2012 AIP, the amount allocated to making payouts under such plan (the "funded amount") was determined based upon the performance multiplier for the core financial metrics only. Accordingly, if the performance multiplier for both the core financial metrics and the strategic focus metric resulted in a preliminary payout for a named executive officer that exceeded the funded amount, and the named executive officer's manager desired to provide the named executive officer with that higher payout amount, such manager was required to reallocate amounts from preliminary payouts for other employees in order to provide the higher payout to the named executive officer. However, if the performance multiplier for both the core financial metrics and the strategic focus metric resulted in a preliminary payout for a named executive officer that was less than the funded amount, and the named executive officer's manager desired to provide the named executive officer with a payout that was equal to the funded amount, such manager was required to adjust the preliminary payout up to the funded amount, but did not have to reallocate amounts from preliminary payouts for other employees to do so. Each respective manager for Messrs. Harbaugh, Silver and Edwards adjusted the payout amounts to provide a final payout that was close or equal to the funded amount.

### **Going Forward**

Our Compensation Committee will develop a process for establishing financial and non-financial performance goals that initially will be similar to that of Covidien.

-127-

#### **Retention Benefits**

Covidien implemented a retention program for key employees of its Pharmaceuticals business, including the named executive officers. At the time of implementation, Covidien was considering a sale or spin-off transaction for its Pharmaceuticals business and deemed the retention of these key employees as being essential to the ultimate consummation of a transaction and the smooth transition of such business to a purchaser or an independent company, as applicable. While Covidien has entered into retention agreements with each named executive officer, we expect that we will assume each agreement in connection with the separation and will be responsible for satisfying any obligations with respect to the retention benefits provided therein. For more information about these retention benefits, see "Executive Compensation—Potential Payments Upon Termination—Covidien Retention Agreements."

### **Other Benefits**

## Historically

Each of the benefits described below was chosen to support Covidien's philosophy of providing a total rewards perspective to compensating its employees. Collectively, these benefits are intended to be competitive with Covidien's peer companies.

*Retirement Benefits*. Covidien maintains six defined benefit pension plans for the benefit of U.S. employees associated with its Pharmaceuticals business. These pension plans have been frozen with respect to all future benefit accruals and will be sponsored and maintained by us immediately upon the separation. No named executive officer is eligible to participate in any of these defined benefit plans because all such plans were frozen before each executive officer commenced employment with Covidien. However, the named executive officers are eligible to participate in the Covidien Retirement Savings and Investment Plan ("Covidien Retirement Savings Plan"), Covidien's 401(k) plan, which is available to all eligible U.S. employees, and the Covidien Supplemental Savings and Retirement Plan ("Covidien Supplemental Savings Plan"), Covidien's non-qualified deferred compensation plan in which executive officers and other senior employees may participate. For more information regarding the Covidien Supplemental Savings Plan, see "Executive Compensation—Non-Qualified Deferred Compensation."

*Health and Welfare Benefits.* The health and welfare benefits Covidien provides to the named executive officers are offered to all eligible U.S.-based employees and include medical, dental, prescription drug, vision, life insurance, accidental death and dismemberment, business travel accident, personal and family accident, flexible spending accounts, short- and long-term disability coverage and an employee assistance program.

*Perquisites*. Although Covidien does not have a perquisite program, it maintains an executive physical program which offers comprehensive and coordinated annual physical examinations to certain senior-level employees. This program is available to Mr. Trudeau, but not the other named executive officers.

*Employee Stock Purchase Plan.* Covidien maintains a broad-based employee stock purchase plan that provides eligible employees, including the named executive officers, with the opportunity to purchase Covidien ordinary shares. Eligible employees authorize payroll deductions to be made for the purchase of Covidien ordinary shares and Covidien provides a 15% matching contribution on up to \$25,000 of an employee's payroll deductions in any calendar year. All shares are purchased on the open market by a designated broker.

Severance Benefits. Covidien maintains an executive severance plan which provides benefits to Covidien senior executives upon an involuntary termination of employment for any reason other than cause, permanent disability or death. Severance benefits, in the form of base salary continuation, bonus and health benefits are generally payable for 18 months (24 months for Covidien's President and Chief Executive Officer) following termination of employment. For fiscal 2012, under the Covidien executive severance plan, Mr. Trudeau was eligible for 18 months of severance benefits while the other named executive officers were eligible for 12 months of severance benefits. Receipt of these benefits is conditioned upon the named executive officer signing a release of any claims against Covidien.

-128-

*Change in Control Benefits.* Covidien maintains a change in control plan which provides benefits to certain Covidien senior executives upon an involuntary termination of employment or good reason resignation that occurs during a period shortly before and continuing after a change in control (a double trigger arrangement). Benefits are generally payable following termination of employment in a lump-sum cash payment equal to two times (2.99 times for Covidien's President and Chief Executive Officer) the sum of the executive's base salary and the average of the executive's bonus for the previous three fiscal years. Additional benefits provided upon a change in control termination include full vesting of outstanding equity awards, continued subsidy for health plan premiums for a 24-month period (36 months for Covidien's President and Chief Executive Officer) and outplacement services. For fiscal 2012, Mr. Trudeau was eligible for change in control severance benefits under the Covidien Change in Control Plan, but the other named executive offers were not eligible for such benefits. Receipt of change in control severance benefits is conditioned upon the executive signing a release of any claims against Covidien.

### **Going Forward**

We expect to maintain the various benefits mentioned above immediately following the separation. Going forward, we expect that our Compensation Committee will consider and determine whether to adopt, modify or terminate any of these benefits.

#### **Executive Compensation Recoupment Policy**

### Historically

Covidien maintains an Executive Compensation Recoupment Policy ("Recoupment Policy") which requires that Covidien recoup portions of incentive compensation paid to its executive officers if there is a restatement of Covidien's financial statements due to material noncompliance with financial reporting requirements under applicable securities laws or regulations and the amount of incentive compensation that was awarded to an executive officer during the three fiscal years immediately preceding the date of the restatement (or such other period as required under applicable securities laws or regulations) is higher than the amount of incentive compensation that would have been awarded to the executive officer had the financial results subject to the restatement been properly reported. For this purpose, incentive compensation includes annual incentive compensation, certain long-term incentive awards, and any other compensation determined to be incentive compensation pursuant to regulations to be issued by the SEC. In addition, Covidien's equity awards are subject to a claw-back provision, pursuant to which Covidien may recover the amount of any profit the named executive officer realized upon the exercise of options or vesting of other equity awards during the 12-month period that occurs immediately prior to the executive officer's involuntary termination of employment for cause.

#### **Going Forward**

Our executive compensation recoupment policy, including the inclusion of any claw-back provisions in equity awards, will be developed in consultation with our Compensation Committee, taking into account market practice and any applicable laws.

#### **Share Ownership Guidelines**

#### **Historically**

To reinforce the alignment of management and shareholder interests, the Covidien Compensation Committee adopted share ownership guidelines. Under these guidelines, Covidien named executive officers are expected to hold Covidien equity with a value expressed as a multiple of base salary as follows:

> President and Chief Executive Officer Other Named Executive Officers

5 times base salary 3 times base salary

In determining an executive's ownership, shares held directly as well as shares underlying restricted units and their accompanying dividend equivalent units are included. Shares underlying unexercised stock options and unvested performance units and their accompanying dividend equivalent units are not included in the calculation.

-129-

Executives are required to achieve the requisite ownership position within five years of first becoming subject to the share ownership guidelines. Covidien's Insider Trading Policy prohibits employees, including named executive officers, from engaging in transactions in puts, calls, cashless collars, options or similar rights and obligations involving Covidien securities, other than the exercise of a Covidien-issued stock option.

### **Going Forward**

We expect our share ownership guidelines for named executive officers and insider trading policy to be developed taking into account market practice and any applicable laws.

### **Deductibility of Executive Compensation**

### **Historically**

The Covidien Compensation Committee has generally intended to structure Covidien's executive compensation in a manner designed to qualify for deductibility under Section 162(m) of the Code when consistent with Covidien's overall compensation program objectives, while also maintaining maximum flexibility in the design of Covidien compensation programs and in making appropriate payments to named executive officers.

### **Going Forward**

We expect our Compensation Committee to adopt a similar practice with respect to minimizing the adverse effect of Section 162(m) of the Code (once applicable) on the deductibility of compensation expense following the separation.

### **Compensation Risk Assessment**

### Historically

At the Covidien Compensation Committee's direction, representatives of Covidien's human resources and legal departments conducted a risk assessment of Covidien's compensation policies and practices during fiscal 2012. This risk assessment consisted of a review of cash and equity compensation provided to Covidien employees, including named executive officers, with a focus on compensation payable to senior executives and incentive compensation plans which provide variable compensation to other employees based upon Covidien and individual performance. The Covidien Compensation Committee and its independent compensation consultant reviewed the findings of this assessment and agreed with the conclusion that Covidien's compensation programs are designed with the appropriate balance of risk and reward in relation to Covidien's overall business strategy and do not create risk that is reasonably likely to have a material adverse effect on Covidien. The following characteristics of Covidien's compensation programs support this finding:

- The use of different types of compensation vehicles that provide a balance of short- and long-term incentives with fixed and variable components;
- The cap on awards to limit windfalls;
- The practice of looking beyond results-oriented performance in assessing the contributions of a particular executive;
- The share ownership guidelines;
- The executive compensation recoupment policy;
- The claw-back policy for equity awards; and
- The ability of the Covidien Compensation Committee to reduce incentive payouts if deemed appropriate.

#### **Going Forward**

Our Compensation Committee will take into account risk-management practices and risk-taking incentives as it considers and develops our employee and executive compensation programs and it will adopt a risk assessment process relating to compensation policies and practices initially similar to that in place at Covidien.

-130-

# EXECUTIVE COMPENSATION

## **Summary Compensation**

The information included in the Summary Compensation Table below reflects compensation earned during fiscal 2012 by individuals who we expect to serve as our executive officers and who could serve as our named executive officers upon the separation. As we continue to build-out the infrastructure necessary to continue as a standalone publicly traded company after the separation, we expect to retain the services of other individuals who also could serve as its named executive officers upon the separation. We refer to the individuals listed in the table below collectively as the "named executive officers." For a more complete understanding of the table, please read the narrative disclosures that follow the table.

# SUMMARY COMPENSATION TABLE

Name and <u>Principal Position*</u> (A) Mark Trudeau, President and Chief Executive Officer	Fiscal Year (B) 2012	Salary (\$) (C) \$420,000	Bonus (\$) (D) \$225,000	Stock Awards (\$) (E) \$945,965	Option Awards (\$) (F) \$623,096	Inc	on-Equity entive Plan npensation (\$) (G) 507,252	Change in Pension Value and Non-Qualified Deferred Compensation Earnings (S) (H)	II Other pensation (\$) (I) 109,730	Total (\$) (J) \$2,831,044
Matthew Harbaugh, Senior Vice President and Chief Financial Officer	2012	\$334,723	—	\$428,537	\$364,707	\$	205,543	_	\$ 34,295	\$1,367,804
Thomas Berry, Senior Vice President, Product Supply	2012	\$317,910	\$ 76,947	\$145,637	\$ 79,458	\$	188,039	—	\$ 28,220	\$ 836,211
David Silver, Senior Vice President, Portfolio Management, Strategy, and Business Development and Licensing*	2012	\$296,881	_	\$211,517	\$115,335	\$	149,998	_	\$ 21,985	\$ 795,716
Peter Edwards, Senior Vice President and General Counsel	2012	\$322,827	—	\$149,465	\$ 81,535	\$	181,825	—	\$ 23,522	\$ 759,174

\* Mr. Silver will be terminating his employment with the Pharmaceuticals business prior to the completion of the separation. None of the other executive officers of the Pharmaceuticals business had total compensation in excess of \$100,000 in fiscal 2012.

The discussion below sets forth a description of the elements of compensation reported in the columns of the Summary Compensation Table.

### Salary (Column C)

With respect to Mr. Harbaugh, the Covidien Compensation Committee approved an additional \$10,000 per month in base salary as part of his supplemental compensation package for serving as interim-President. Amounts reported in this column for Mr. Harbaugh reflect \$289,723 paid as base salary and \$45,000 paid as the additional monthly allowance.

#### Bonus (Column D)

This column reflects a one-time sign-on bonus paid to Mr. Trudeau in connection with his commencement of employment with Covidien on February 1, 2012 and a retention bonus paid to Mr. Berry in connection with the retention program Covidien implemented for key employees of its Pharmaceuticals business.

### -131-

## Stock Awards (Column E) and Option Awards (Column F)

These columns represent the aggregate grant date fair value, computed in accordance with Accounting Standards Codification 718 *Compensation—Stock Compensation*, of restricted unit, performance unit and stock option awards issued to each named executive officer during fiscal 2012. Further information regarding the 2012 awards is included in the Fiscal 2012 Grants of Plan-Based Awards Table and the Outstanding Equity Awards at 2012 Fiscal Year End Table.

In the case of performance unit awards issued to named executive officers (other than Mr. Trudeau) as part of Covidien's 2012 annual equity award, the grant date fair value is based on the probable outcome of the market-based performance conditions, calculated based on the application of a Monte Carlo simulation model. The actual amounts which vest are determined at the end of the three-year performance cycle and are based on total shareholder return for Covidien as compared to total shareholder return of companies comprising a healthcare industry index. Depending upon whether or to what extent the performance conditions are met, twice as many performance units may vest, or none may vest at all. Amounts in these columns do not correspond to the actual value that may be recognized by the named executive officers, which may be higher or lower based on a number of factors, including Covidien's performance, stock price fluctuations and applicable vesting. For additional information relating to assumptions made in the valuation for current year awards reflected in these columns, see note 17 to the annual combined financial statements included elsewhere in this information statement.

### Non-Equity Incentive Plan Compensation (Column G)

The amounts reported in Column G represent annual incentive cash awards paid to the named executive officers under the Covidien 2012 AIP. The amount of Mr. Harbaugh's annual incentive cash award was calculated by taking into account the additional monthly allowance and increased target bonus opportunity percentage that was provided to him as part of his supplemental compensation package, but only for the period during which he served as interim-President. For information regarding the calculation of these awards, see "Compensation Discussion and Analysis."

## Change in Pension Value and Non-Qualified Deferred Compensation Earnings (Column H)

No named executive officer is eligible to participate in a Mallinckrodt or Covidien defined benefit pension plan because all such plans were frozen before each executive officer commenced employment with Covidien.

-132-

## All Other Compensation (Column I)

The amounts reported in Column I represent the aggregate dollar amount for each named executive officer for employer contributions to the Covidien Retirement Savings Plan, employer credits to the Covidien Supplemental Savings Plan, relocation benefits, and tax reimbursements attributable to relocation benefits. The following table shows the specific amounts included in Column I of the Summary Compensation Table for fiscal 2012. For a more complete understanding of the table, please read the narrative disclosures that follow the table.

### ALL OTHER COMPENSATION

Name and Principal Position (A)	Cor to I	Covidien Contributions to Retirement Savings Plan (B)		Covidien Credits to pplemental <u>vings Plan</u> (C)	Relocation Benefits (D)	Tax Reimbursements on <u>Relocation Benefits</u> (E)		Total (F)	
Mark Trudeau, President and Chief Executive Officer	\$	7,500	\$	2,975	\$ 65,599	\$	33,656	\$109,730	
Matthew Harbaugh, Senior Vice President and Chief Financial Officer	\$	9,066	\$	25,229	—		—	\$ 34,295	
Thomas Berry, Senior Vice President, Product Supply	\$	15,041	\$	13,179	—		—	\$ 28,220	
David Silver, Senior Vice President, Portfolio Management, Strategy, and Business Development and Licensing	\$	10,939	\$	11,046	_			\$ 21,985	
Peter Edwards, Senior Vice President and General Counsel	\$	10,610	\$	12,911	_			\$ 23,522	

### **Relocation Benefits (Column D)**

This column reflects relocation benefits paid by Covidien during fiscal 2012.

### Tax Reimbursements on Relocation Benefits (Column E)

This column reflects reimbursements for taxes associated with relocation benefits paid by Covidien during fiscal 2012.

### **Grants of Plan-Based Awards**

The following table provides information concerning the annual incentive cash awards and equity incentive awards granted to each of the named executive officers in fiscal 2012.

- "AIP" is the annual incentive cash award payable pursuant to the Covidien 2012 AIP.
- "PSUs" are restricted unit awards subject to performance-based vesting, which we refer to as "performance units."
- "RSUs" are restricted unit awards subject to time-based vesting, which we refer to as "restricted units."
- "Options" are nonqualified stock options subject to time-based vesting.

For a more complete understanding of the table, please read the related narrative.

-133-

### FISCAL 2012 GRANTS OF PLAN-BASED AWARDS

			Estimated Future Payouts Under Non-Equity Incentive Plan Awards		Estimated Future Payouts Under Equity Incentive Plan Awards		All other Stock Awards: Number of Shares of	All other Option Awards: Number of Securities	Exercise or Base Price of	Grant Date Fair Value of Stock			
Name (A)	Grant Date (B)	Date of Committee Action	Threshold (\$) (C)	Target (\$) (D)	Maximum (\$) (E)	Threshold (#) (F)	Target 	Maximum (#) (H)	Stock or Units (#) (I)	Underlying Options (#) (J)	Option Awards (\$/Sh) (K)	Av	Option wards (\$) (L)
Mark Trudeau AIP RSUs Options	02/1/2012 02/1/2012	11/28/2011 11/28/2011	\$ 260,000	\$520,000	\$1,040,000				18,115	51,900	\$ 52.22		945,965 623,096
Matthew Harbaugh AIP PSUs RSUs Supplemental RSUs Options Supplemental Options	12/1/2011 12/1/2011 02/1/2012 12/1/2011 02/1/2012	11/16/2011 11/16/2011 11/16/2011	\$ 82,411	\$164,821	\$ 329,643	775	1,550	3,100	5,942 1,101	30,555 5,010	\$ 46.45 \$ 52.22	\$	95,037 311,458 57,495 309,672 55,035
Thomas Berry AIP PSUs RSUs Options	12/1/2011 12/1/2011 12/1/2011	11/16/2011 11/16/2011 11/16/2011	\$ 80,410	\$160,819	\$ 321,638	862	1,723	3,446	861	7,840	\$ 46.45	\$ \$ \$	105,644 39,993 79,458
David Silver AIP PSUs RSUs Options	12/1/2011 12/1/2011 12/1/2011	11/16/2011 11/16/2011 11/16/2011	\$ 59,853	\$119,707	\$ 239,414	1,251	2,502	5,004	1,251	11,380	\$ 46.45	\$	153,408 58,109 115,335
Peter Edwards AIP PSUs RSUs Options	12/1/2011 12/1/2011 12/1/2011	11/16/2011 11/16/2011 11/16/2011	\$ 73,210	\$146,421	\$ 292,841	884	1,768	3,536	884	8,045	\$ 46.45	\$	108,403 41,062 81,535

## Non-Equity Incentive Plan Awards (Columns C through E)

The amounts reported in Columns C through E reflect threshold, target and maximum award amounts for fiscal 2012 pursuant to the Covidien 2012 AIP, which is an element of the Covidien 2007 Stock and Incentive Plan. With respect to Mr. Harbaugh, the supplemental compensation package approved by the Covidien Compensation Committee set his target bonus percentage under the 2012 AIP at 65% of base salary (which, for this purpose, *includes* the additional monthly allowance awarded as part of his supplemental compensation package) for the period of time during fiscal 2012 that he served as interim-President and at 50% of base salary (which, for this purpose, *excludes* the additional monthly allowance awarded as part of his supplemental compensation package) for the period of time during fiscal 2012, that he served as interim-President and at 50% of base salary (which, for this purpose, *excludes* the additional monthly allowance awarded as part of his supplemental compensation package) for the remainder of fiscal 2012. Mr. Harbaugh served as interim-President for four out of twelve months in fiscal 2012, resulting in his fiscal 2012 bonus being calculated by applying an effective target bonus percentage of 55% (*i.e.*, the weighted-average of a 65% target bonus percentage for one-third of fiscal 2012 and a 50% target bonus percentage for the remaining two-thirds of fiscal 2012). Accordingly, the threshold, target and maximum award amounts reported in Columns C through E for Mr. Harbaugh represent the respective potential payments when applying the effective target bonus percentage (*i.e.*, 55%) for fiscal 2012 to a weighted-average of his base salary during fiscal 2012. The actual amounts earned by each named executive officer pursuant to such awards are set forth in Column G of the Summary Compensation Table.

#### Equity Incentive Plan Awards (Columns F through H)

The amounts reported in Columns F through H reflect threshold, target and maximum award amounts for the fiscal 2012—2014 performance cycle pursuant to performance unit awards issued as part of Covidien's fiscal 2012

-134-

annual equity awards. The actual amounts, if any, earned by each named executive officer pursuant to such awards are determined by the Covidien Compensation Committee at the end of the three-year performance cycle and are based upon total shareholder return for Covidien as compared to the total shareholder return of companies comprising a healthcare industry index (*i.e.*, relative total shareholder return). Threshold, target and maximum award amounts are payable upon achievement of relative total shareholder return in the 25<sup>th</sup>, 50<sup>th</sup> and 75<sup>th</sup> percentile, respectively. Dividend equivalent units will be credited on performance unit awards only if, and to the extent that, dividends are payable on ordinary shares, and will vest only if the applicable performance criteria are satisfied.

### Stock Awards and Option Awards (Columns I and J)

The amounts reported in Column I and Column J reflect the number of shares underlying restricted unit awards and stock option awards, respectively, that were granted as part of Covidien's fiscal 2012 annual equity awards, which vest one-quarter annually beginning on the first anniversary of the grant date. Dividend equivalent units will be credited on restricted unit awards only if, and to the extent that, dividends are payable on ordinary shares, and will vest according to the same schedule as the underlying restricted units.

*Mr. Harbaugh.* With respect to Mr. Harbaugh, amounts reported in these columns for the December 1, 2011 restricted unit and stock option awards reflect Mr. Harbaugh's annual equity incentive award and an additional 775 restricted units and 7,050 stock options, each valued at \$71,451, that were awarded as part of his supplemental compensation package for serving as interim-President through November 2011. Due to an administrative error, it was discovered shortly after the issuance of the annual equity awards that this December 2011 award issued to Mr. Harbaugh understated the value of the award that he should have received. Amounts reported in these columns for the February 1, 2012 restricted unit and stock option awards reflect the difference between what Mr. Harbaugh received in December 2011 and what he should have received absent the administrative error plus an additional 766 restricted units and 3,485 stock options, valued at \$40,001 and \$38,283, respectively, that were awarded as part of his supplemental compensation package for serving as interim-President ended upon Mr. Trudeau's commencement of employment with Covidien on February 1, 2012.

#### Grant Date Fair Value (Column L)

In the case of performance unit awards issued as part of Covidien's 2012 annual equity awards, the grant date fair value is based on the probable outcome of the market-based performance conditions, calculated based on the application of a Monte Carlo simulation model. Depending upon whether or to what extent the respective performance conditions are met, the number of shares for which the performance units are settled may range from zero to 200%.



## **Outstanding Equity Awards at Fiscal Year End**

The following table provides information regarding outstanding stock option awards and unvested restricted unit awards and, if applicable, performance unit awards held by each named executive officer as of September 28, 2012. Restricted unit awards and performance unit awards listed in the table include dividend equivalent units credited on such awards. Dividend equivalent units vest according to the same schedule as the underlying restricted unit award or, in the case of performance unit awards, if the applicable performance criteria are satisfied. For a more complete understanding of the table, please read the footnotes that follow the table. Unless otherwise specified, the market value of outstanding stock awards in the table below is calculated by multiplying the number of unvested restricted or performance units by \$59.42, the closing price of Covidien shares on September 28, 2012.

### **OUTSTANDING EQUITY AWARDS AT 2012 FISCAL YEAR END**

		<b>Option</b> A	Awards		Stock Awards							
<u>Name</u> (A) Mark Trudeau	Number of Securities Underlying Unexercised Options (#) <u>Exercisable</u> (B)	Number of Securities Underlying Unexercised Options (#) <u>Unexercisable</u> (C) 51,900(1)	Option Exercise Price (\$) (D) \$ 52,2200	Option Expiration Date (E) 01/31/2022	Number of Shares or Units of Stock That Have Not Vested (#) (F) 18,261(10)	Market Value of Shares or Units of Stock That Have Not Vested (\$) (G) \$1,085,069	Equity Incentive Plan Awards: Number of Unearned Shares, Units or Other Rights That Have Not Vested (#) (H)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$) (1) \$ 0				
Matthew Harbaugh	1,800 3,155 2,295 2,708 0 0	0 3,155 <sup>(2)</sup> 4,590 <sup>(3)</sup> 8,127 <sup>(4)</sup> 30,555 <sup>(5)</sup> 5,010 <sup>(6)</sup>	\$40.2600 \$34.1500 \$47.6000 \$42.9400 \$46.4500 \$52.2200	09/03/2017 11/30/2018 11/30/2019 11/30/2020 11/30/2021 01/31/2022	$\begin{array}{c} 488^{(11)}\\ 591^{(12)}\\ 965^{(13)}\\ 6,014^{(14)}\\ 1,109^{(15)}\end{array}$	\$ 28,997 \$ 35,117 \$ 57,340 \$ 357,352 \$ 65,897	2,364 <sup>(19)</sup> 2,576 <sup>(20)</sup> 3,138 <sup>(21)</sup>	\$ 142,218 \$ 153,066 \$ 186,460				
Thomas Berry	2,265 2,457 0	2,265 <sup>(7)</sup> 7,373 <sup>(4)</sup> 7,840 <sup>(5)</sup>	\$50.4800 \$42.9400 \$46.4500	03/31/2020 11/30/2020 11/30/2021	580 <sup>(16)</sup> 876 <sup>(13)</sup> 871 <sup>(14)</sup>	\$ 34,464 \$ 52,052 \$ 51,755	2,336 <sup>(20)</sup> 3,488 <sup>(21)</sup>	\$ 138,805 \$ 207,257				
David Silver	0 0 0 0 0	$1,617^{(2)} \\ 565^{(8)} \\ 4,910^{(3)} \\ 8,037^{(4)} \\ 11,380^{(5)}$	\$34.1500 \$32.3600 \$47.6000 \$42.9400 \$46.4500	11/30/2018 04/30/2019 11/30/2019 11/30/2020 11/30/2021	$\begin{array}{c} 250^{(11)} \\ 174^{(17)} \\ 632^{(12)} \\ 956^{(13)} \\ 1,266^{(14)} \end{array}$	<ul> <li>\$ 14,855</li> <li>\$ 10,339</li> <li>\$ 37,553</li> <li>\$ 56,806</li> <li>\$ 75,226</li> </ul>	2,528 <sup>(19)</sup> 2,548 <sup>(20)</sup> 5,064 <sup>(21)</sup>	\$ 152,084 \$ 151,402 \$ 300,903				
Peter Edwards	0 0 0	2,895 <sup>(9)</sup> 7,617 <sup>(4)</sup> 8,045 <sup>(5)</sup>	\$41.2400 \$42.9400 \$46.4500	05/31/2020 11/30/2020 11/30/2021	740 <sup>(18)</sup> 905 <sup>(13)</sup> 894 <sup>(14)</sup>	\$ 43,971 \$ 53,775 \$ 53,121	2,414 <sup>(20)</sup> 3,578 <sup>(21)</sup>	\$ 143,440 \$ 212,605				

Unless otherwise specified, stock option and restricted unit awards vest one-quarter annually, beginning on the first anniversary of the grant date.

<sup>(1)</sup> Represents stock options granted on February 1, 2012 to Mr. Trudeau in connection with his commencement of employment with Covidien.

<sup>(2)</sup> Represents stock options granted on December 1, 2008.

<sup>(3)</sup> Represents stock options granted on December 1, 2009.

<sup>(4)</sup> Represents stock options granted on December 1, 2010.

<sup>(5)</sup> Represents stock options granted on December 1, 2011.

(6) Represents stock options granted on February 1, 2012 to Mr. Harbaugh as a supplemental award. For more information about this award, see the Fiscal 2012 Grants of Plan-Based Awards Table and related narrative under the "Stock Awards and Option Awards (Columns I and J)" heading.

-136-

- <sup>(7)</sup> Represents stock options granted on April 1, 2010 to Mr. Berry in connection with his commencement of employment with Covidien.
- <sup>(8)</sup> Represents stock options granted on May 1, 2009 to Mr. Silver in connection with a promotion.
- (9) Represents stock options granted on June 1, 2010 to Mr. Edwards in connection with his commencement of employment with Covidien.
  (10) Represents restricted units granted on Expressive 1, 2012 to Mr. Trudeev in connection with his commencement of employment as Brasida
- (10) Represents restricted units granted on February 1, 2012 to Mr. Trudeau in connection with his commencement of employment as President of Covidien's Pharmaceuticals business; 6,756 of which vest one-third annually, beginning on the first anniversary of the grant date and 11,505 of which vest one-quarter annually, beginning on the first anniversary of the grant date.
- <sup>(11)</sup> Represents restricted units granted on December 1, 2008.
- <sup>(12)</sup> Represents restricted units granted on December 1, 2009.
- <sup>(13)</sup> Represents restricted units granted on December 1, 2010.
- <sup>(14)</sup> Represents restricted units granted on December 1, 2011.
- <sup>(15)</sup> Represents restricted units granted on February 1, 2012 to Mr. Harbaugh as a supplemental award. For more information about this award, see the Fiscal
- 2012 Grants of Plan-Based Awards Table and related narrative under the "Stock Awards and Option Awards (Columns I and J)" heading.
- (16) Represents restricted units granted on April 1, 2010 to Mr. Berry in connection with his commencement of employment with Covidien.
  (17) Represents restricted units granted on May 1, 2000 to Mr. Silver is connection with a promotion.
- <sup>(17)</sup> Represents restricted units granted on May 1, 2009 to Mr. Silver in connection with a promotion.
- <sup>(18)</sup> Represents restricted units granted on June 1, 2010 to Mr. Edwards in connection with his commencement of employment with Covidien.
- (19) Represents performance units granted on December 1, 2009 that vested on October 4, 2012, shortly after the end of the fiscal 2010—2012 performance cycle. The amounts reported in Column I and J are based on actual achievement, which was two hundred percent (200%) of target, and are valued by using the closing price of Covidien stock on the vesting date, which was \$60.16.
- (20) Represents performance units granted on December 1, 2010 that vest at the end of the fiscal 2011—2013 performance cycle if the applicable performance criteria have been satisfied. The amounts reported in this column are based on achievement of maximum performance through the end of fiscal 2012.
- (21) Represents performance units granted on December 1, 2011 that vest at the end of the fiscal 2012—2014 performance cycle if the applicable performance criteria have been satisfied. The amounts reported in this column are based on achievement of maximum performance through the end of fiscal 2012.

### **Option Exercises and Stock Vested**

The following table provides information regarding the number of Covidien stock options that were exercised by named executive officers during fiscal 2012 and the value realized from the exercise of such awards. The table also provides information regarding the vesting of restricted unit and performance unit awards during fiscal 2012.

# FISCAL 2012 OPTION EXERCISES AND STOCK VESTED

	Opti	on Awards	Stock Awards			
Name (A)	Number of Shares Acquired on Exercise (#) (B)		lue Realized n Exercise (\$) (C)	Number of Shares Acquired on Vesting (#) (D)		lue Realized on Vesting (\$) (E)
Mark Trudeau	0	\$	0	0	\$	0
Matthew Harbaugh	0	\$	0	2,894	\$	130,228
Thomas Berry	0	\$	0	577	\$	29,180
David Silver	12,751	\$	216,474	1,967	\$	90,820
Peter Edwards	5,433	\$	73,896	667	\$	32,539

### **Pension Benefits**

No named executive officer is eligible to participate in a Mallinckrodt or Covidien defined benefit pension plan because all such plans were frozen before each executive officer commenced employment with Covidien.



#### **Non-Qualified Deferred Compensation**

The following table provides information with respect to fiscal 2012 non-qualified deferred compensation for each named executive officer. For more information regarding information contained in the table and the material terms of Covidien's non-qualified deferred compensation plan, please read the related narrative and footnotes that follow the table.

## FISCAL 2012 NON-QUALIFIED DEFERRED COMPENSATION

Name (A)	Con	xecutive atributions Last FY (\$) (B)	Con	Covidien htributions Last FY (\$) (C)	Aggregate Earnings in Last FY (S) (D)	Aggregate Withdrawals/ Distributions (\$) (E)	Aggregate Balance at Last FYE (\$) (F)
Mark Trudeau	\$	51,000	\$	2,975	\$10,775		\$ 64,750
Matthew Harbaugh	\$	0	\$	25,229	\$ 8,033	—	\$ 53,407
Thomas Berry	\$	87,638	\$	13,179	\$15,268	—	\$127,078
David Silver	\$	0	\$	11,046	\$19,168	—	\$ 92,647
Peter Edwards	\$	0	\$	12,911	\$ 1,388	_	\$ 14,300

#### Executive Contributions in Last Fiscal Year (Column B)

Of the amounts reported in this column, the following amounts reflect deferrals from fiscal 2012 base salary that also are reported in Column C (Salary) of the Summary Compensation Table: Mr. Trudeau, \$51,000 and Mr. Berry, \$9,537. The remaining amount in this column for Mr. Berry relates to the deferral of Covidien 2011 Annual Incentive Plan bonus payments paid in fiscal 2012.

#### Covidien Contributions in Last Fiscal Year (Column C)

The amounts reported in Column C are included in Column I of the Summary Compensation Table for fiscal 2012.

#### Aggregate Earnings in Last Fiscal Year (Column D)

The amounts reported in Column D include earnings credited to the named executive officer's account in the Covidien Supplemental Savings Plan. Earnings on amounts credited to the Covidien Supplemental Savings Plan are determined by investment selections made by each named executive officer in investment alternatives that generally mirror investment choices offered under the Covidien Retirement Savings Plan.

Under the Covidien Supplemental Savings Plan, participants, including the named executive officers, may defer up to 50% of their base salary and 100% of their annual bonus. Covidien provides matching credits based on the participant's deferred base salary and bonus at the same rate such participant is eligible to receive matching contributions under the Covidien Retirement Savings Plan (the Covidien 401(k) plan) and employer credits on any cash compensation (*i.e.*, base and bonus) that the participant earns during a calendar year in excess of applicable IRS limits (\$245,000 for 2011 and \$250,000 for 2012). Under the Covidien Retirement Savings Plan, Covidien makes an automatic contribution of three percent (3%) of an employee's eligible pay, irrespective of whether the employee contributes to such plan. Additionally, Covidien matches fifty cents (\$0.50) for every one dollar (\$1.00) employees contribute, up to the first six percent (6%) of eligible pay. Participants are fully vested in matching and employer credits (including earnings on such credits) upon completion of two years of service. The Covidien Supplemental Savings Plan is a non-qualified deferred compensation plan that is maintained as an unfunded "top-hat" plan and is designed to comply with Section 409A of the Code. Amounts credited to the Covidien Supplemental Savings Plan as participant deferrals or employer credits may also be credited with earnings (or losses) based upon investment selections made by each participant from investments that generally mirror

-138-

investments offered under the Covidien Retirement Savings Plan. Participants may elect whether they will receive a distribution of their Covidien Supplemental Savings Plan account balances upon termination of employment or at a specified date. Distributions can be made in a lump sum or in up to 15 annual installments.

### **Potential Payments Upon Termination**

## **Covidien Severance Plan**

During fiscal 2012, for all of the named executive officers in the table below, severance benefits were payable pursuant to the Covidien Severance Plan for U.S. Officers and Executives (the "Covidien Severance Plan"). Under the Covidien Severance Plan, benefits are payable to eligible executives, including named executive officers, upon an involuntary termination of employment for any reason other than cause, permanent disability or death. With respect to the named executive officers, severance benefits consist of:

- continuation of base salary for a period of 12 months (18 months for Mr. Trudeau);
- payment, over a 12-month period, of one times the average of the named executive officer's bonus for the previous three fiscal years (1.5 times the average of the previous three fiscal year bonuses paid over an 18-month period for Mr. Trudeau);
- continuation of health and dental benefits at active employee rates for up to 12 months (18 months for Mr. Trudeau);
- 12 months accelerated vesting of unvested stock options and 12 months to exercise vested stock options (unless a longer period is provided in the applicable award agreement);
- payment of a pro-rata portion of the named executive officer's annual incentive cash award for the fiscal year in which the applicable employment termination date occurs; and
- outplacement services, in Covidien's discretion, for up to 12 months.

Upon a termination of employment other than for cause, including an involuntary termination of employment where the named executive officer becomes eligible for severance benefits, named executive officers forfeit all unvested restricted unit awards and performance unit awards and any stock options which do not vest within 12 months after the applicable employment termination date.

### Covidien Change in Control Plan

For Mr. Trudeau, change in control severance benefits are payable pursuant to the Covidien Change in Control Severance Plan for Certain U.S. Officers and Executives (the "Covidien Change in Control Plan"). Under the Covidien Change in Control Plan, benefits are payable to eligible senior executives only if the executive experienced an involuntary termination of employment or good reason resignation during a period that begins 60 days before and ends two years after a change in control. No other named executive officer was eligible for change in control benefits under the Covidien Change in Control Plan during fiscal 2012. However, named executive officers other than Mr. Trudeau were eligible for severance benefits under the Covidien Severance Plan in the event of an involuntary termination of employment following a change in control. Also, as described below under "—Other Termination Benefits," the terms of the Covidien 2012 AIP and outstanding equity awards issued pursuant to the Covidien equity plan provide for certain benefits upon an involuntary termination of employment following a change in control. All named executive officers were eligible for these benefits during fiscal 2012. For purposes of the following, we list the benefits that would be provided upon an involuntary termination of employment after a change in control for all named executive officers other than Mr. Trudeau. For Mr. Trudeau, we list the benefits that would be provided if he became eligible for benefits pursuant to the Covidien Change in Control Plan. With respect to named executive officers, the change in control benefits consist of:

- continuation of base salary for a period of 12 months (a single lump-sum payment equal to 24 months of base salary for Mr. Trudeau pursuant to the Covidien Change in Control Plan);
- payment, over a 12-month period, of one times the average of the named executive officer's bonus for the previous three fiscal years (a single lumpsum payment equal to two times the average of his bonus for the previous three fiscal years for Mr. Trudeau under the Covidien Change in Control Plan);
- continuation of health and dental benefits at active employee rates for a period of up to 12 months (18 months for Mr. Trudeau plus a lump-sum payment equal to six months of the employer portion of the applicable premium under the Covidien Change in Control Plan);
- full vesting of unvested stock options and 12 months to exercise vested stock options (unless a longer period is provided in the applicable option agreement) and full vesting of unvested restricted unit awards and performance unit awards;
- payment of a pro-rata portion of the annual incentive plan bonus for the fiscal year during which the applicable employment termination date occurs; and
- outplacement services, in Covidien's discretion, for up to 12 months.

The payment of benefits under the Covidien Severance Plan and the Covidien Change in Control Plan is conditioned upon the named executive officer executing a release of claims against Covidien and is subject to the terms of the Non-Competition, Non-Solicitation, and Confidentiality Agreement by and between the named executive officer and Covidien, under which the named executive officer agreed not to disclose confidential information at any time and not to compete with Covidien or solicit Covidien employees or customers for a period of one year following termination of employment. Covidien may cancel benefits that are payable or seek to recover benefits previously paid if the named executive officer does not comply with these provisions or violates the release of claims. Payments may be delayed until six months after termination of employment if necessary to comply with Section 409A of the Code.

Upon a termination of employment for cause, named executive officers are not eligible for severance benefits under the Covidien Severance Plan or the Covidien Change in Control Plan and forfeit all unvested stock options, restricted unit and performance unit awards. In addition, the stock option, restricted unit and performance unit awards include a "claw-back" feature pursuant to which Covidien may recover the amount realized by the named executive officer upon the vesting of any stock award during the 12-month period that occurs immediately prior to the officer's involuntary termination for cause.

### Other Termination Benefits

The terms of the Covidien Annual Incentive Plan and Covidien 2007 Stock and Incentive Plan provide for certain benefits upon a named executive officer's termination of employment due to death, disability or retirement. For this purpose, normal retirement occurs where an executive officer terminates employment after attaining age 60 and the sum of the executive's age and years of service equals at least 70 and early retirement occurs where an executive officer terminates employment after attaining age 55 and the sum of the executive's age plus years of service equals at least 60. Under the Covidien Annual Incentive Plan, named executive officers are eligible to receive a pro-rated annual incentive cash award based on the number of days that the executive officer was employed by Covidien during the fiscal year upon death, disability or normal or early retirement. Under the Covidien 2007 Stock and Incentive Plan, named executive officers are eligible to receive full vesting of stock options, restricted units and performance units upon death, disability, normal retirement or an involuntary termination of employment after a change in control and pro-rated vesting of such awards upon early retirement, based on the number of whole months that the executive officer was employed by Covidien during the applicable vesting period. As of the end of fiscal 2012, Messrs. Berry and Silver had satisfied the requirements for early retirement.

-140-

### **Covidien Retention Agreements**

The following describes the retention benefits that Covidien has agreed to provide to each named executive officer as part of its retention program.

*Mr. Trudeau*. The retention agreement that Covidien entered into with Mr. Trudeau provides for benefits in the event of a sale of Covidien's Pharmaceuticals business, including a sale bonus, a sale price bonus and an enhanced severance benefit. The sale bonus, which is payable upon a sale, equals the sum of Mr. Trudeau's then-current base salary and the average of his annual incentive bonus for the previous three fiscal years. If Mr. Trudeau has not received three annual incentive bonus payments, the average of his annual incentive bonus amounts equals the average of all actual bonuses paid to him pursuant to the Covidien Annual Incentive Plan. The sale price bonus is payable only if the sale proceeds received by Covidien exceed a threshold amount and is capped at \$1 million. The enhanced severance benefit, which is payable if, in connection with a sale, Covidien involuntarily terminates Mr. Trudeau's employment, the purchaser does not offer Mr. Trudeau a position after consummation of the sale, or Mr. Trudeau resigns from employment for good reason within 12 months after consummation of a sale, equals the severance Mr. Trudeau would be entitled to under the Covidien executive severance plan plus 1.5 times the sum of Mr. Trudeau's then-current base salary and the average of Mr. Trudeau's annual incentive bonus for the previous three fiscal years.

*Messrs. Harbaugh and Silver*. The retention agreements that Covidien entered into with Messrs. Harbaugh and Silver provide for benefits in the event of a sale or, in the alternative, a spin-off of Covidien's Pharmaceuticals business. In the event of a sale, Messrs. Harbaugh and Silver are eligible to receive a retention bonus and a sale price bonus; in the event of a spin-off, each is eligible to receive a spin bonus or termination bonus. The retention bonus, which is payable on the six-month anniversary of a sale if the respective executive remains continuously employed by the purchaser through such date or, if before such anniversary date, the purchaser involuntarily terminates the executive's employment, the purchaser does not offer the executive a comparable position after consummation of the sale, the executive resigns from employment for good reason, or the executive dies or becomes permanently disabled, equals \$750,000 for Mr. Harbaugh and \$1 million for Mr. Silver. The sale price bonus is payable only if the sale proceeds received by Covidien exceed a threshold amount and is capped at \$500,000 for each executive. The spin bonus, which is payable on the six-month anniversary of the completion of the separation if the respective executive remains continuously employed by us through such anniversary date, equals \$139,755 for Mr. Harbaugh and \$145,256 for Mr. Silver. The termination bonus, which is payable if, before the six-month anniversary of the completion of the separation if the executive's employment, the executive resigns from employment, the executive dies or becomes permanently disabled, equals \$145,256 for Mr. Silver. The termination bonus, which is payable if, before the six-month anniversary of the completion of the separation if the executive's employment, the executive resigns from employment for good reason, or the executive dies or becomes permanently disabled, equals \$150,000 for Mr. Silver. The termination bonus, which is payable if, before the six-month anniversary of the completion of

*Mr. Berry*. The retention agreement that Covidien entered into with Mr. Berry provides for a retention bonus, a sale bonus and benefits in the event of a spin-off of Covidien's Pharmaceuticals business, which include a spin bonus or termination bonus. The retention bonus consists of two payments of \$76,947 each, with the first payment having been made on the one-year anniversary of the retention agreement's effective date (this payment was made on August 1, 2012) and with the second payment being payable on the 18-month anniversary of the retention agreement's effective date (*i.e.*, on February 1, 2013). The retention agreement includes a claw-back feature which requires that Mr. Berry repay any amounts paid pursuant to the retention agreement if he voluntarily terminates employment before a sale or spin-off. The sale bonus, which is payable on the six-month anniversary of a sale if Mr. Berry remains continuously employed by the purchaser through such date or, if before such anniversary date, the purchaser involuntarily terminates Mr. Berry's employment, the purchaser does not offer Mr. Berry a comparable position after consummation of the sale, Mr. Berry resigns from employment for good reason, or Mr. Berry dies or becomes permanently disabled, equals \$307,788, but is reduced by the amount of the retention bonus paid to Mr. Berry. The spin bonus, which is payable on the six-month anniversary date, equals \$153,894. The termination bonus, which is payable if, before the six-month anniversary of the completion of the separation, if Mr. Berry remains continuously employed by us through such anniversary date, equals \$153,894. The termination bonus, which is payable if, before the six-month anniversary of the completion of the separation, we involuntarily terminate Mr. Berry's employment, Mr. Berry resigns from employment for good reason, or Mr. Berry is employment, Mr. Berry resigns from employment for good reason, or Mr. Berry is employment, Mr. Berry resigns from employment for good reason, or Mr. Berry is empl

-141-

*Mr. Edwards.* The retention agreement that Covidien entered into with Mr. Edwards provides for benefits in the event of a sale or, in the alternative, a spinoff of Covidien's Pharmaceuticals business. In the event of a sale, Mr. Edwards is eligible to receive a sale bonus; in the event of a spin-off, Mr. Edwards is eligible to receive a spin bonus or termination bonus. The sale bonus, which is payable on the six-month anniversary of a sale if Mr. Edwards remains continuously employed by the purchaser through such date or, if before such anniversary date, the purchaser involuntarily terminates Mr. Edwards' employment, the purchaser does not offer to Mr. Edwards a comparable position after consummation of the sale, Mr. Edwards resigns from employment for good reason, or Mr. Edwards dies or becomes permanently disabled, equals \$500,000. The spin bonus, which is payable on the six-month anniversary of the completion of the separation if Mr. Edwards remains continuously employed by us through such anniversary date, equals \$157,951. The termination bonus, which is payable if, before the six-month anniversary of the completion of the separation, we involuntarily terminate Mr. Edwards' employment, Mr. Edwards resigns from employment for good reason, or Mr. Edwards dies or becomes permanently disabled, equals \$500,000.

All of the retention agreements discussed above require the forfeiture of retention benefits in the event that Covidien (or the purchaser or Mallinckrodt, as applicable) terminates the named executive officer's employment for cause. The retention agreements also subject the payment of retention benefits to the named executive officer complying with the Covidien Guide to Business Conduct (or successor guide to business conduct), preserving confidentiality on the terms and conditions of any transaction or the status of any negotiations relating to any transaction, and cooperating with efforts surrounding a sale or spin-off transaction.

For purposes of the Covidien Severance Plan, the Covidien Change in Control Plan and the retention agreements, "cause" means substantial failure or refusal of the named executive officer to perform the duties and responsibilities of his job as required by Covidien, violation of any fiduciary duty owed to Covidien, conviction of a felony or misdemeanor, dishonesty, theft, violation of Covidien rules or policy, including a violation of the Covidien Guide to Business Conduct, or other egregious conduct that has or could have a serious and detrimental impact on Covidien and its employees.

For purposes of the Covidien Change in Control Plan and the retention agreements, "good reason" means any retirement or termination of employment by the named executive officer that is not initiated by Covidien and that is caused by any one or more of the following events, in each case, without the named executive officer's written consent: (i) assignment to the named executive officer of any duties inconsistent in any material respect with the named executive officer's authority, duties or responsibilities as in effect immediately prior to the change in control or effective date of the retention agreement, as applicable; (ii) a material diminution in the authority, duties or responsibilities of the supervisor to whom the named executive officer is required to report as in effect immediately prior to the change in control or effective date of the retention agreement, as applicable; (iii) a material change in the geographic location at which the named executive officer must perform services to a location which is more than 50 miles from the named executive officer's principal place of business immediately preceding the change in control or effective date of the retention agreement, as applicable; (iv) a material reduction in the named executive officer's compensation and benefits, taken as a whole, as in effect immediately prior to the change in control or effective date of the retention agreement, as applicable; (iv) a material reduction in the named executive officer services of business immediately price to the Covidien Change in Control Plan, Covidien's failure to obtain a satisfactory agreement from any successor to assume and agree to perform Covidien's obligations to the named executive officer runder such plan; or (vi) a material diminution in the budget over which the named executive officer retains authority. Additionally, "good reason" will only exist if the named executive officer provides written notice stating the good reason event, Covidien does not cure such event, and the named executive officer

### **Potential Payments Upon Termination Table**

The table below reflects the amount of compensation that would become payable to each named executive officer under the Covidien Severance Plan and, with respect to Mr. Trudeau, the Covidien Change in Control Plan, if the named executive officer's employment had terminated on September 28, 2012, the last day of

Covidien's 2012 fiscal year, given the named executive's service level as of such date and, if applicable, based on Covidien's closing stock price as of that date, which was \$59.42. These benefits are in addition to benefits available before the occurrence of a termination of employment, including under then-exercisable stock options and benefits available generally to salaried employees, such as distributions under the Covidien Retirement Savings Plan.

The actual amounts that would be paid upon a named executive officer's termination of employment or in connection with a change in control can be determined only at the time of any such event. Due to a number of factors that may affect the amount of any benefits provided upon the events discussed below, actual amounts paid or distributed may be higher or lower than indicated in the table. Factors that could affect these amounts include the timing during the year of any such event, Covidien's stock price, the executive's age and years of service, the attained level of achievement for performance units, and any additional agreements or arrangements entered into in connection with any change in control or termination of employment. For a more complete understanding of the table, please read the narrative that follows the table.

## POTENTIAL PAYMENTS UPON TERMINATION

Name and Termination Scenario (A)	Cash Severance (B)	Bonus (C)	Option Awards (D)	Stock Awards (E)	Ben	Velfare efits and <u>placement</u> (F)	Total (G)
Mark Trudeau							
Involuntary Termination (other than for cause)	\$1,735,878	\$507,252	\$ 93,420	—	\$	43,659	\$2,390,684(1)
Death or Disability		\$507,252	\$373,680	\$1,085,009		—	\$1,976,416(1)
Change in Control Termination	\$2,314,504	\$507,252	\$373,680	\$1,085,009	\$	49,296	\$4,340,216(1)
Matthew Harbaugh							
Involuntary Termination (other than for cause)	\$ 458,014	\$205,543	\$259,577	\$ 0	\$	37,756	\$ 960,890
Death or Disability		\$205,543	\$700,284	\$1,026,447			\$1,932,274
Change in Control Termination	\$ 458,014	\$205,543	\$700,284	\$1,026,447	\$	37,756	\$2,428,044
Thomas Berry							
Involuntary Termination (other than for cause)	\$ 461,798	\$188,039	\$ 76,049	_	\$	37,756	\$ 763,642
Voluntary Termination (early retirement)			\$ 35,445	\$ 102,896			\$ 138,341
Death or Disability		\$188,039	\$243,441	\$ 484,332			\$ 915,813
Change in Control Termination	\$ 461,798	\$188,039	\$243,441	\$ 484,332	\$	37,756	\$1,415,366
David Silver							
Involuntary Termination (other than for cause)	\$ 418,667	\$149,998	\$166,218	\$ 283,957	\$	37,756	\$1,056,597
Voluntary Termination (early retirement)	_	\$149,998	\$ 90,616	\$ 283,957			\$ 524,571
Death or Disability		\$149,998	\$394,235	\$ 799,168		_	\$1,343,401
Change in Control Termination	\$ 418,667	\$149,998	\$394,235	\$ 799,168	\$	37,756	\$1,799,825
Peter Edwards							
Involuntary Termination (other than for cause)	\$ 507,204	\$181,825	\$ 94,232	_	\$	37,756	\$ 821,017
Death or Disability	—	\$181,825	\$282,503	\$ 506,912			\$ 971,240
Change in Control Termination	\$ 507,204	\$181,825	\$282,503	\$ 506,912	\$	37,756	\$1,516,200

Also includes \$7,500 in employer contributions to the Covidien Retirement Savings Plan and \$2,975 in employer credits to the Covidien Supplemental Savings Plan that will become fully vested upon an involuntary termination of employment (other than for cause), death or disability or a change in control termination. All other named executive officers are fully vested in employer contributions and credits.

### Cash Severance (Column B)

*Involuntary Termination (other than for cause).* For all named executive officers other than Mr. Trudeau, the cash severance amount in this scenario represents continuation of the named executive officer's base salary, as of September 28, 2012, for a 12-month severance period plus the average of the named executive officer's annual incentive cash awards for the previous three fiscal years (*i.e.*, fiscal 2011, 2010 and 2009), payable during

-143-

the 12-month severance period and on Covidien's normal payroll schedule. With respect to Messrs. Berry and Edwards, who commenced employment with Covidien during fiscal 2010 and who received a pro-rated annual incentive bonus for such year, the average of their respective annual incentive cash awards has been adjusted to reflect the period of time that they were employed by Covidien through the end of fiscal 2012. For Mr. Trudeau, the amount represents continuation of his base salary, as of September 28, 2012, for an 18-month severance period, plus an amount equal to 1.5 times his annual incentive cash award for fiscal 2012, payable during the 18-month severance period and on Covidien's normal payroll schedule. If Mr. Trudeau's involuntary termination of employment (other than for cause) was in connection with a sale of Covidien's Pharmaceuticals business, such an event would have increased the cash severance payable to Mr. Trudeau to \$3,471,756, and resulted in a total potential payment of \$4,126,562. While all of the other amounts payable under this scenario and listed in columns C, D and F would have remained the same, upon a sale of Covidien's Pharmaceuticals business, Mr. Trudeau would be eligible for a sale bonus and a sale price bonus. For more information about the enhanced severance benefit and the sale bonus and the sale price bonus, please read the section above for Mr. Trudeau under "—Covidien Retention Agreements."

*Change in Control Termination.* For Mr. Trudeau, who is the only named executive officer eligible for benefits under the Covidien Change in Control Plan, the amount in this scenario represents a lump-sum payment equal to two times his base salary as of September 28, 2012 plus his annual incentive cash award for fiscal 2012. For all other named executive officers, we assume that such executive officers experience an involuntary termination of employment (other than for cause) after the change in control which renders them eligible for benefits under the Covidien Severance Plan. Accordingly, the cash severance amount for named executive officers other than Mr. Trudeau in this scenario equals the cash severance amount set forth under the "Involuntary Termination (other than for cause)" scenario.

### Bonus (Column C)

Involuntary Termination (other than for cause). In the case of an involuntary termination of employment (other than for cause), executive officers are entitled to a pro-rata payment of the annual incentive cash award based on the number of days they were employed by Covidien during the fiscal year. Because we have assumed that the applicable terminations of employment occurred on the last day of Covidien's 2012 fiscal year, the amounts reported in Column C for this scenario represent the full annual incentive cash award payable to each named executive officer for fiscal 2012.

*Voluntary Termination (early retirement).* Because Messrs. Berry and Silver have satisfied the requirements for early retirement under the Covidien 2012 AIP, in the event of a voluntary termination of employment, each is entitled to a pro-rata payment of the annual incentive cash award based on the number of days that they, respectively, were employed by Covidien during the fiscal year. Because we have assumed that the applicable terminations of employment occurred on the last day of Covidien's 2012 fiscal year, the amounts reported in Column C for this scenario represent the full annual incentive cash award payable to Messrs. Silver and Berry, respectively, for fiscal 2012.

Death or Disability and Change in Control Termination. The bonus amount represents the pro-rata payment of the annual incentive cash award based on the number of days that the named executive officer was employed with Covidien's Pharmaceuticals business during the fiscal year. Because we have assumed that the applicable termination of employment occurred on the last day of our 2012 fiscal year, the amounts reported in Column C for this scenario represent the full annual incentive cash award payable to each named executive officer for fiscal 2012.

#### **Option Awards (Column D)**

Involuntary Termination (other than for cause). For all named executive officers, the option award amount represents the value as of September 28, 2012 of outstanding options held by the named executive officer that would have vested during the 12-month period that immediately follows September 28, 2012 (*i.e.*, from September 28, 2012 to September 28, 2013).

*Voluntary Termination (early retirement).* As of September 28, 2012, Messrs. Berry and Silver satisfied the requirements for early retirement under Covidien's equity plan. The amounts reported in Column D for this scenario represent the value attributable to the portion of the following stock option awards that would have vested on September 28, 2012 had Messrs. Berry and Silver voluntarily terminated employment on such date: for Mr. Berry, the April 2010 and December 2010 option awards; and for Mr. Silver, the December 2008, May 2009, December 2009 and December 2010 option awards. Messrs. Berry and Silver did not satisfy the requirements for early retirement with respect to the December 2011 option award because such award requires that the employee retire at least 12 months after the grant date to receive early retirement. Because the assumed employment termination date (September 28, 2012) is less than 12 months after the December 2011 grant date, neither Mr. Berry nor Mr. Silver was entitled to pro-rata vesting of the December 2011 option award as of the last day of fiscal 2012.

Death or Disability and Change in Control Termination. The option award amount represents the full vesting of unvested stock options held by the named executive officer as of September 28, 2012.

### Stock Awards (Column E)

*Involuntary Termination (other than for cause).* The amounts reported in Column E for this scenario represent the value of the performance unit awards issued in December 2009 to Messrs. Harbaugh and Silver (but not to the other named executive officers because they were not employed by Covidien at the time such award was granted), which vested on October 4, 2012 and which the executive officer would have been entitled to receive upon an involuntary termination of employment on the last day of the fiscal year. For purposes of this scenario, the amount reported for the December 2009 performance unit award is based on the actual number of shares that vested after the conclusion of the fiscal 2010—2012 performance cycle and the actual value attained upon vesting. With respect to Messrs. Berry and Silver, who, as of September 28, 2012, satisfied the requirements for early retirement under the Covidien equity plan, the amount reported in Column E for this scenario includes the value attributable to the portion of the following restricted unit and performance unit awards issued in April 2010 and December 2010 and the performance unit award issued in December 2010; and for Mr. Silver, the restricted unit awards issued in December 2009, December 2009 and December 2011 restricted unit and performance unit awards because such awards require that the employee retire at least 12 months after the grant date to receive early retirement. Because the assumed employment termination date (September 28, 2012) is less than 12 months after the December 2011 grant date, neither Mr. Berry nor Mr. Silver were entitled to pro-rata vesting of the December 2011 restricted unit and performance unit awards.

*Voluntary Termination (early retirement)*. For Messrs. Berry and Silver, the stock award amount represents the pro-rata vesting of restricted unit and performance unit awards, as described above under "Involuntary Termination (other than for cause)."

Death or Disability and Change in Control Termination. The amounts reported in Column E for this scenario represent the value that would have been attained upon the full vesting of all unvested restricted unit and performance unit awards held by the named executive officer as of September 28, 2012. For purposes of this scenario, amounts attributable to performance unit awards are based on the following assumptions: (1) for the December 2009 award, the actual number of shares that vested after the conclusion of the fiscal 2010—2012 performance cycle and based on the value attained upon vesting; and (2) for the December 2010 and December 2011 awards, the number of shares that would have vested based on achievement of maximum performance through the end of fiscal 2012.

### Welfare Benefits and Outplacement Services (Column F)

The welfare benefits amount represents the employer portion of the premium paid on behalf of the named executive officer for continued coverage under the Covidien medical, dental and vision plans during the

-145-

applicable severance period. Amounts for calendar year 2012 and 2013 are based on actual rates determined by Covidien for the respective plan in such years, while the rates for subsequent years, where applicable, are assumed based on the historic percentage increase in rates for such coverage. Although payable in Covidien's discretion, for purposes of this column, we assumed that Covidien would pay \$25,000 on behalf of each named executive officer for outplacement services upon an involuntary termination (other than for cause) and a change in control termination.

### Mallinckrodt Pharmaceuticals Stock and Incentive Plan

Prior to the completion of this offering, we expect to adopt the Mallinckrodt Pharmaceuticals Stock and Incentive Plan, or Mallinckrodt SIP.

### Purpose

The purpose of the Mallinckrodt SIP is to assist in the recruitment and retention of directors and key employees, provide incentives to such individuals in consideration of their services to Mallinckrodt, promote the growth and success of our business by aligning the interests of such individuals with those of our shareholders, and provide such individuals with an opportunity to participate in Mallinckrodt's growth and financial success. The Mallinckrodt SIP is expected to provide for the assumption of awards pursuant to the adjustment of awards granted under current plans of Covidien and its subsidiaries.

The following description of the material terms of the Mallinckrodt SIP is qualified in its entirety by the terms and conditions of the plan document, the form of which is included as Exhibit 10.10 hereto.

## Plan Administration

The Mallinckrodt SIP is administered by the Compensation and Human Resources Committee except with respect to director awards, which are administered by the Nominating and Governance Committee. The Compensation and Human Resources Committee or, to the extent required by applicable law, the board of directors, has broad discretion and authority under the Mallinckrodt SIP including the authority to:

- interpret and administer the Mallinckrodt SIP;
- prescribe the rules and regulations that it deems necessary for the proper operation and administration of the Mallinckrodt SIP;
- select employees to receive awards and determine the form of awards, the number of ordinary shares subject to an award and the terms and conditions of each award;
- waive or amend any terms, conditions, restrictions or limitations on an award and/or vest awards upon a participant's termination of employment, except that the Mallinckrodt SIP's prohibition on the repricing of stock options and stock appreciation rights cannot be waived; and
- delegate its duties and appoint agents to help administer the Mallinckrodt SIP.

## Eligibility

Each of our employees providing services to us or any of our affiliates who is selected by the Compensation and Human Resources Committee or its delegate is eligible to receive an award under the Mallinckrodt SIP, and each of our non-employee Directors selected by the Nominating and Governance Committee is eligible to receive an award under the Mallinckrodt SIP.

#### Shares Available

Subject to the share counting rules described below and any adjustment in accordance with the terms of the Mallinckrodt SIP, the total number of ordinary shares with respect to which awards may be issued under the Mallinckrodt SIP is equal to ten percent (10%) of the ordinary shares outstanding on the effective date of the plan.

-146-

Share Counting Rules. When ordinary shares are issued pursuant to a grant of full value awards (*i.e.*, restricted stock, restricted units, deferred stock units or performance units) or as payment of an annual performance bonus or other stock-based awards (which includes awards other than stock options, stock appreciation rights, annual performance bonuses or long-term performance awards), the total number of ordinary shares remaining available for grant is decreased by a margin of 2.2 per ordinary share issued. In determining the number of shares that remain available under the Mallinckrodt SIP, shares related to awards paid in cash do not count against the Mallinckrodt SIP's share limit. In addition, shares of restricted stock that are returned to us upon a participant's termination of employment or, if applicable, a director's termination of directorship and shares related to awards that expire or are forfeited or cancelled, or terminate for any other reason without issuance of shares, are added back to the share limit at a rate of 2.2 shares per each share subject to the expired, forfeited, cancelled or terminated award.

*Limit on Individual Grants.* Subject to adjustment in accordance with the terms of the Mallinckrodt SIP, no employee may be granted more than six (6) million ordinary shares over any calendar year pursuant to stock options, stock appreciation rights and performance-based restricted stock and restricted units, except that an incentive award of no more than ten (10) million ordinary shares may be made pursuant to stock options, stock appreciation rights and performance-based restricted stock and restricted units to any performance based nestricted stock and restricted units to any performance bonuses or long-term performance awards paid in performance units to any one employee is \$15 million (U.S.) for any performance cycle of twelve (12) months. For any longer performance cycle, this maximum will be adjusted proportionately.

#### Stock Options and Stock Appreciation Rights

Stock options awarded under the Mallinckrodt SIP may be in the form of nonqualified stock options or incentive stock options or a combination of the two. Stock appreciation rights may be awarded either alone or in tandem with stock options. Stock appreciation rights will be paid in cash or ordinary shares, or a combination of cash and ordinary shares. Unless otherwise determined by the Compensation and Human Resources Committee or as required by law, stock options and stock appreciation rights granted under the Mallinckrodt SIP are subject to the following terms and conditions:

- *Exercise Price*. The Compensation and Human Resources Committee will set the exercise price at the time of grant, which will be no less than the fair market value of an ordinary share as of the date of grant. Under the Mallinckrodt SIP, fair market value is the closing sales price of an ordinary share of Mallinckrodt stock as reported on the NYSE on the date for which fair market value is being determined which, in the case of establishing the exercise price of an option, is the grant date.
- No Repricing. The exercise price may not be decreased after the grant date, other than in connection with required Mallinckrodt SIP adjustments such as recapitalizations, unless our shareholders specifically approve the repricing.
- Vesting. Stock options and stock appreciation rights will vest at such time and in the manner as determined at the time of grant by the Compensation and Human Resources Committee. Unless otherwise provided in the award certificate, stock options and stock appreciation rights will immediately vest upon the normal retirement, death or disability of a participant, or upon a termination of employment without cause or resignation for good reason after a change in control.
- *Post-Termination Exercise.* Subject to the term of the award, any vested stock option or stock appreciation right that has not already been exercised will remain exercisable for a period of three years after termination of employment because of early or normal retirement, death or disability, and any vested stock option or stock appreciation right that has not already been exercised will remain exercisable for a period of 90 days after termination of employment for any other reason except for a termination for cause.

-147-

#### Performance-Based Awards

The Mallinckrodt SIP provides for performance-based awards in the form of: (1) annual performance bonuses that may be granted in the form of cash or ordinary shares; and (2) long-term performance awards in the form of performance units that may be paid in cash or shares, or performance-based restricted units or restricted stock awards that are paid in shares. The Compensation and Human Resources Committee, in its discretion, will fix the amount, terms and conditions of annual performance bonuses and long-term performance awards, subject to the following:

- *Performance Cycles.* Annual performance bonuses will be awarded in connection with a 12-month performance cycle, which will coincide with our fiscal year. Long-term performance awards will be awarded in connection with a performance cycle that will not be shorter than 12 months. The annual performance bonus amount and the number of shares or units that are earned will be determined by the level of performance attained in relation to the applicable performance measures, as certified by the Compensation and Human Resources Committee following completion of the performance period.
- Target Awards and Award Criteria. The Compensation and Human Resources Committee will set a target amount or target number of shares or units for each participant receiving an annual performance bonus or long-term performance award within 90 days after the start of a performance cycle. At that time, the Compensation and Human Resources Committee will also establish criteria for these awards, including the minimum level of performance that must be attained before any annual performance bonuses and long-term performance awards will be paid or vest and the annual performance bonus amounts and the number of shares or units that will become payable upon attainment of various levels of performance. The Compensation and Human Resources Committee may select as the performance measure(s) any operating and maintenance expense targets or financial goals as interpreted by the Compensation and Human Resources Committee, either individually, alternatively or in any combination, applied to either the Company as a whole or to a business unit or subsidiary, either individually, alternatively or in any combination, and that are absolute or relative to the performance of one or more comparable companies or an index of comparable companies, and are measured during the performance cycle, provided that, as to an annual performance bonus or long-term performance award granted to a covered employee (which is defined in the Mallinckrodt SIP as being a "covered employee" for purposes of Code Section 162(m)), performance measures are limited to the following criteria, and with respect to such awards granted to an employee other than a covered employee, performance measures may include, but not be limited to the following: (a) cash flow, (b) earnings per share, (c) earnings before interest, taxes and amortization, (d) return on equity, (e) total shareholder return, (f) share price performance, (g) return on capital, (h) return on assets or net assets, (i) revenue, (j) income or net income, (k) operating income or net operating income, (1) operating profit or net operating profit, (m) operating margin or profit margin, (n) return on operating revenue. (o) return on invested capital, (p) market segment share, (q) product release schedules, (r) new product innovation, (s) product cost reduction through advanced technology, (t) brand recognition/acceptance, (u) product ship targets, or (v) customer satisfaction. Financial performance measures may take into account such adjustments as the Compensation and Human Resources Committee may specify, including the exclusion of unusual or infrequently occurring items; provided, however, that such adjustments shall not impact a covered employee unless the Compensation and Human Resources Committee determines to make such adjustments no more than ninety (90) days after the commencement of the applicable performance cycle.
- *Dividends and Dividend Equivalents.* At the discretion of the Compensation and Human Resources Committee, dividends paid on shares may be paid immediately or withheld and deferred in the participant's account. In the event of a payment of dividends on ordinary shares, the Compensation and Human Resources Committee may credit long-term performance awards with dividend equivalent units, which may be distributed immediately, withheld and deferred in the participant's account or credited in the form of additional share units.

-148-

### Restricted Stock, Restricted Units and Deferred Stock Units

Restricted stock, restricted units and deferred stock units may be awarded under the Mallinckrodt SIP to any employee selected by the Compensation and Human Resources Committee. Restricted units and deferred stock units may be settled in shares or cash. The Compensation and Human Resources Committee has the discretion to fix the terms and conditions applicable to awards of restricted stock, restricted units and deferred stock units.

- Vesting. Unless the award certificate provides otherwise, any restrictions on restricted stock, restricted units or deferred stock units will vest in equal annual installments over a four-year period after the grant date. Unless the award certificate provides otherwise, any restrictions on restricted stock, restricted units or deferred stock units that have not vested or been satisfied on the date of a participant's termination of employment will immediately vest in full or in part upon early or normal retirement, death or disability of the participant or certain terminations of employment following a change in control. Upon a termination of employment for any other reason, any unvested restricted units, deferred stock units or shares of restricted stock will be forfeited.
- Dividends and Dividend Equivalents. At the discretion of the Compensation and Human Resources Committee, dividends paid on shares may be paid
  immediately or withheld and deferred in the participant's account. In the event of a payment of dividends on ordinary shares, the Compensation and
  Human Resources Committee may credit restricted units and deferred stock units with dividend equivalent units, which may be distributed
  immediately, withheld and deferred in the participant's account or credited in the form of additional share units. The Compensation and Human
  Resources Committee has issued restricted unit awards which are credited with dividend equivalent units, the payment of which is deferred until the
  underlying restricted units vest. The Compensation and Human Resources Committee expects to continue this practice going forward.

### **Director** Awards

The Nominating and Governance Committee has the exclusive authority to issue awards to directors, which may consist of, but not be limited to, restricted stock, restricted units, deferred stock units, stock options, stock appreciation rights or other stock-based awards. Each director award is governed by an award certificate that is approved by the Nominating and Governance Committee.

### **Other Stock-Based Awards**

The Compensation and Human Resources Committee may grant other share-based awards under the Mallinckrodt SIP that consist of, or are denominated in, ordinary shares. These awards may include phantom or hypothetical shares. The Compensation and Human Resources Committee has broad discretion to determine the terms and conditions that will apply to other stock-based awards.

#### Substitute Awards

The Compensation and Human Resources Committee may make awards to grantees of an acquired company through the assumption of, or in substitution for, outstanding stock-based awards previously granted to the grantees. The assumed or substituted awards will be subject to the terms and conditions of the original awards made by the acquired company, with any adjustments that the Compensation and Human Resources Committee considers necessary to comply with applicable law or appropriate to give effect to the relevant provisions of any agreement for the acquired company.

#### Adjustments

The kind or maximum number of ordinary shares available for issuance under the Mallinckrodt SIP, the individual and aggregate maximums that may be issued under each form of award, the number of ordinary shares

underlying outstanding awards and the exercise price applicable to outstanding stock options and stock appreciation rights shall be appropriately adjusted by the Compensation and Human Resources Committee upon any stock split, reverse stock split, dividend or other distribution, extraordinary cash dividend, recapitalization, merger, consolidation, split-up, spin-off, reorganization, combination, repurchase or exchange of ordinary shares or other securities, or similar corporate transaction or event, to prevent dilution or enlargement of the benefits intended to be made available under the Mallinckrodt SIP.

### Change in Control

All outstanding stock options, stock appreciation rights and long-term performance awards will become exercisable and all outstanding restricted stock, restricted units and deferred stock units will vest if there is a change in control *and* the change in control results in a termination without cause, resignation for good reason or substitution of the awards for awards not payable in publicly-traded stock. Each participant who has been granted an annual performance bonus or long-term performance award that is outstanding as of the date of a change in control will be deemed to have achieved a level of performance that, as of the change in control, would cause all of the participant's target amount to become payable, unless the successor entity maintains the annual performance plan and the actual level of performance achieved would result in an annual performance bonus that exceeds the participant's target amount, in which case bonuses based on actual performance shall be paid.

### **Restrictions on Transfer of Awards**

No award issued under the Mallinckrodt SIP may be alienated, anticipated, sold, assigned, pledged, encumbered or transferred, except that (a) awards may be transferred by will or by the laws of descent or distribution; (b) unless the award certificate provides otherwise, stock options may be transferred to a family member (as described in the Mallinckrodt SIP) without consideration; and (c) restricted stock may be freely transferable after the restrictions lapse or are satisfied and the shares are delivered.

#### Amendment and Termination

The Mallinckrodt SIP may be amended or terminated by our board of directors at any time without shareholder approval, except that any material revision to the terms of the Mallinckrodt SIP requires shareholder approval before it can be effective. A revision is "material" for this purpose if it materially increases the number of ordinary shares that may be issued under the plan, other than an increase pursuant to an "adjustment" as described above, materially expands the class of persons eligible to receive awards, materially extends the term of the plan, reduces the exercise price at which stock options or stock appreciation rights may be granted, reduces the exercise price of outstanding stock options or stock appreciation rights, results in the replacement of outstanding stock options or stock appreciation rights with cash, stock options or stock appreciation rights that have a lower exercise price, or any other awards, or is otherwise an amendment requiring shareholder approval pursuant to any law or the rules of any exchange on which our ordinary shares are listed for trading. If not earlier terminated, the Mallinckrodt SIP will terminate on the day before the tenth anniversary of the adoption by the board of directors of the Mallinckrodt SIP. No awards may be granted under the Mallinckrodt SIP after it is terminated, but any previously granted awards will remain in effect until they expire.

#### Code Section 162(m)

Code Section 162(m) generally limits a company's annual deduction for compensation in excess of \$1 million paid to certain executive officers (these executive officers are referred to in the Mallinckrodt SIP as "covered employees"). Compensation paid to covered employees is not subject to the deduction limitation, however, if it is considered "qualified performance-based compensation" within the meaning of Code Section 162(m). Awards of stock options, stock appreciation rights, annual performance bonuses, performance units, performance-based restricted units and performance-based restricted stock under the Mallinckrodt SIP can, but are not required to, satisfy this standard under Code Section 162(m).

## Summary of Federal Income Tax Consequences of Awards

The following is a brief summary of the principal United States federal income tax consequences of the grant, exercise and disposition of stock options, stock appreciation rights, restricted stock, restricted units and deferred stock units under the Mallinckrodt SIP, based on advice received from our counsel regarding current U.S. federal income tax laws. This summary is not intended to be exhaustive and, among other things, does not describe state, local or foreign tax consequences. Because the federal income tax rules governing awards and related payments are complex, subject to frequent change, and depend on individual circumstances, participants should consult their tax advisors before exercising options or other awards or disposing of stock acquired pursuant to awards.

*Nonqualified Stock Options and Stock Appreciation Rights.* A participant will not recognize any income at the time a nonqualified stock option or stock appreciation right is granted, nor will we be entitled to a deduction at that time. When a nonqualified stock option is exercised, the participant will recognize ordinary income in an amount equal to the excess of the fair market value of the ordinary shares received as of the date of exercise over the exercise price. When a stock appreciation right is exercised, the participant will recognize ordinary income in an amount equal to the cash received or, if the stock appreciation right is paid in ordinary shares, the fair market value of the ordinary shares received as of the date of exercise. Payroll taxes are required to be withheld from the participant on the amount of ordinary income recognized by the participant. We generally will be entitled to a tax deduction with respect to a nonqualified stock option or stock appreciation right at the same time and in the same amount as the participant recognizes income. The participant's subsequent sale of the ordinary shares plus the ordinary income the participant recognized with respect to the shares, and these capital gains will be taxable as long-term capital gains if the participant held the shares for more than one year following exercise.

*Incentive Stock Options.* A participant will not recognize any income at the time an incentive stock option ("ISO") is granted. Nor will a participant recognize any income at the time an ISO is exercised. However, the excess of the fair market value of the ordinary shares on the date of exercise over the exercise price paid will be a preference item that could create liability under the alternative minimum tax. If a participant disposes of ordinary shares acquired upon exercise of an ISO after the later of two years after the date of grant of the ISO or one year after the date of exercise of the ISO (the "holding period"), the gain, if any, will be long-term capital gain eligible for favorable tax rates. If the participant disposes of such ordinary shares before the end of the holding period, the participant generally will recognize ordinary income in the year of the disposition equal to the excess of the lesser of (i) the fair market value of the ordinary shares on the date of exercise or (ii) the amount received for the ordinary shares, over the exercise price paid. The balance of the gain or loss, if any, will be short-or long-term capital gain or loss, depending on how long the ordinary shares were held by the participant prior to disposition. We are not entitled to a deduction as a result of the grant or exercise of an ISO unless a participant recognizes ordinary income as a result of a disposition, in which case we will be entitled to a deduction at the same time and in the same amount as the participant recognizes ordinary income.

*Restricted Stock.* Unless a participant makes an election to accelerate recognition of the income to the date of grant (as described below), the participant will not recognize income, and Mallinckrodt will not be allowed a tax deduction, at the time a restricted stock award is granted. When the restrictions lapse, the participant will recognize ordinary income equal to the fair market value of the common stock as of that date (less any amount paid for the stock) and Mallinckrodt will be allowed a corresponding federal income tax deduction. The participant's subsequent sale of the ordinary shares will give rise to capital gain or loss equal to the difference between the sale price and the ordinary income the participant recognized with respect to the shares, and any capital gains will be taxable as long-term gains if the participant held the shares for more than one year following the date on which restrictions lapsed. If the participant files an election under Section 83(b) of the Code within 30 days of the date of grant of the restricted stock, the participant will recognize ordinary income as of the date of grant equal to the fair market value of the stock as of that date (less any amount paid for the stock) and

-151-

Mallinckrodt will be allowed a corresponding federal income tax deduction. The participant's subsequent sale of the ordinary shares will give rise to capital gain or loss equal to the difference between the sale price and the ordinary income the participant recognized with respect to the shares, and any capital gains will be taxable as long-term gains if the participant held the shares for more than one year following the date of grant. However, if the stock is later forfeited, the participant will not be able to recover the tax previously paid pursuant to a Section 83(b) election.

*Restricted Units/Deferred Stock Units.* A participant will not recognize any income at the time a restricted unit or deferred stock unit is granted, nor will we be entitled to a deduction at that time. Instead, the value of shares delivered on or after the vesting of restricted units or deferred stock units generally will be taxable to the recipient as ordinary income when shares are delivered to the participant. The amount of the income recognized will be the fair market value of the shares on the date shares are delivered. We will generally receive a deduction for federal income tax purposes in an amount equal to the amount of compensation included in the participant's income. The participant's subsequent sale of the ordinary shares will give rise to capital gain or loss equal to the difference between the sale price and the ordinary income the participant recognized with respect to the shares, and any capital gains will be taxable as long-term gains if the participant held the shares for more than one year following the date on which they were delivered.

-152-

## DIRECTOR COMPENSATION

It is anticipated that our board of directors will approve compensation for non-employee directors effective as of the distribution that will consist of an equity award, annual cash retainer and, for some positions, a supplemental cash retainer.

The cash retainers will be paid in four quarterly installments at the end of each quarter. The annual cash retainer for all directors is \$100,000, with the nonexecutive Chairman receiving a supplemental cash retainer of \$50,000, the chairs of the Audit Committee and the Compensation and Human Resources Committee each receiving a supplemental cash retainer of \$20,000, the chairs of the Compliance Committee and the Nominating and Governance Committee each receiving a supplemental cash retainer of \$10,000 and each member of a committee required by NYSE rules (excluding committee chairs) receiving a supplemental cash retainer of \$5,000.

In addition, at the time of our Annual General Meeting, each non-employee director will be granted restricted units with a value of \$180,000 and the nonexecutive Chairman will be granted additional restricted units with a value of \$90,000. These awards fully vest on the date of the following Annual General Meeting and any dividends that we pay between the grant and vesting dates will be credited as dividend equivalent units and will be paid out when the underlying restricted units vest and shares are issued.

Directors are also reimbursed for reasonable out-of-pocket expenses incurred in attending meetings of our board of directors, committee meetings and shareholder meetings. Directors will be provided with private aircraft in order to travel to and from such meetings.

It is anticipated that any new directors joining our board of directors (including our initial directors) will receive a prorated cash retainer and a prorated annual equity grant. A prorated annual equity grant will not be granted to any new director who commences serving less than three months prior to the vesting date.

-153-

## CERTAIN RELATIONSHIPS AND RELATED PERSON TRANSACTIONS

Our board's Nominating and Governance Committee will be responsible for the review and, if appropriate, approval or ratification of "related-person transactions" involving us or our subsidiaries and related persons in accordance with the related-person transactions policy to be adopted by the board. Under SEC rules, a related person is a director, nominee for director, executive officer or a beneficial owner of 5% or more of our ordinary shares, and their immediate family members.

Our personnel in the legal and finance departments will review transactions involving related persons. If they determine that a related person could have a material interest in such a transaction, the transaction will be reviewed by the Nominating and Governance Committee. The Nominating and Governance Committee will determine whether the related person has a material interest in a transaction and may, in its discretion, approve, ratify or take other action with respect to the transaction. The Nominating and Governance Committee will review all material facts related to the transaction and take into account, among other factors it deems appropriate, whether the transaction is on terms no less favorable to us than terms generally available to an unaffiliated third party under the same or similar circumstances, the extent of the related person's interest in the transaction and, if applicable, the availability of other sources of comparable products or services.

We engage in transactions with other Covidien businesses. Those transactions are described in more detail in note 2 to our interim unaudited condensed combined financial statements and note 11 to our annual combined financial statements included elsewhere in this information statement.

For a discussion of certain agreements we will enter into with Covidien in connection with the separation, see "Our Relationship with Covidien Following the Distribution."

-154-

## SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

Before the separation, all of the outstanding shares of Mallinckrodt will be owned beneficially by an Irish corporate services provider. The following table sets forth information, immediately following the completion of the distribution calculated as of May 24, 2013, based upon the distribution of one ordinary share of Mallinckrodt for every eight ordinary shares of Covidien, regarding: (1) each person known to us who would beneficially own more than 5% of our ordinary shares, (2) each of our expected directors and named executive officers and (3) all of our expected directors and executive officers as a group. The address of each director and executive officer shown in the table below is c/o Mallinckrodt, 675 James S. McDonnell Blvd., Hazelwood, MO 63042.

	Beneficial Ownership of Our Ordinary	Percent of
Name of Beneficial Owner	Shares	Class
Directors and Named Executive Officers		
Melvin D. Booth	0	*
Mark C. Trudeau	5,520(1)	*
David R. Carlucci	0	*
J. Martin Carroll	0	*
Diane H. Gulyas	0	*
Nancy S. Lurker	0	*
JoAnn A. Reed	0	*
Kneeland C. Youngblood, M.D.	0	*
Joseph A. Zaccagnino	3,414(2)	*
Matthew Harbaugh	3,904(3)	*
Thomas Berry	1,415(4)	*
David Silver	509(5)	*
Peter Edwards	872(6)	*
All directors and executive officers as a group (17 persons)	17,103(7)	*
Other Beneficial Owners		
BlackRock, Inc.,		

40 East 52nd Street New York, NY 10022 2,956,700<sup>(8)</sup>

\* Represents less than 1% of outstanding ordinary shares.

 Includes 3,817 restricted units and 1,622 ordinary shares issuable upon the exercise of stock options presently exercisable or exercisable within 60 days of May 24, 2013.

(2) Includes 1,843 restricted units.

(3) Includes 1,124 restricted units and 1,788 ordinary shares issuable upon the exercise of stock options presently exercisable or exercisable within 60 days of May 24, 2013.

(4) Includes 131 restricted units and 1,284 ordinary shares issuable upon the exercise of stock options presently exercisable or exercisable within 60 days of May 24, 2013.

5.17%

(5) Includes 509 restricted units.

(6) Includes 511 restricted units and 361 ordinary shares issuable upon the exercise of stock options presently exercisable or exercisable within 60 days of May 24, 2013.

(7) Includes, for executive officers not specifically named in the table, an aggregate of 1,458 restricted units.

(8) Based on information contained in a Schedule 13G filed by BlackRock, Inc. with the SEC on January 30, 2013. BlackRock, Inc. reports it has sole voting and investment power with respect to these shares.

-155-

## THE SEPARATION

## Background

On December 15, 2011, Covidien announced that it intended to separate its Pharmaceuticals business from the remainder of its business. Covidien also announced that it anticipated that the transaction will be in the form of a distribution that will be tax-free to U.S. shareholders of new publicly traded stock in the new pharmaceuticals company.

Mallinckrodt plc was incorporated in Ireland on January 9, 2013 for the purpose of holding Covidien's Pharmaceuticals business following the separation.

On March 13, 2012, Covidien's shareholders approved an amendment to Covidien's articles of association to give the Covidien board of directors authority to declare dividends in specie, or non-cash dividends. The distribution constitutes a dividend in specie for the purposes of Irish law.

On May 23, the Covidien board of directors approved the transfer of its Pharmaceuticals business to Mallinckrodt in return for which Mallinckrodt will issue ordinary shares on the basis of one of our ordinary shares for every eight Covidien ordinary shares held on the record date, subject to the satisfaction of the conditions to the distribution.

Currently, all of our issued shares are held beneficially by an Irish corporate services provider (which is not a subsidiary of Covidien). Immediately prior to the distribution, Covidien will transfer its Pharmaceuticals business to us in return for which we will issue shares to Covidien ordinary shareholders, pro rata to their respective holdings. Prior to the transfer by Covidien to Mallinckrodt plc of our business, we will have no business operations.

On June 28, 2013, the expected distribution date, each person who held Covidien ordinary shares at the close of business on the record date will receive one ordinary share of Mallinckrodt for every eight Covidien ordinary shares held at the close of business on the record date, as described below. You will receive cash in lieu of any fractional ordinary shares of Mallinckrodt which you would have received after the application of the above ratio. Immediately following the distribution, the persons entitled to receive Mallinckrodt ordinary shares in the distribution will own all of our outstanding ordinary shares. You will not be required to make any payment, surrender or exchange your Covidien ordinary shares or take any other action to receive your ordinary shares of Mallinckrodt in the distribution. In connection with these transactions, we will acquire the shares held beneficially by the Irish corporate services provider referred to above for no consideration and cancel these shares.

The distribution of our ordinary shares as described in this information statement is subject to the satisfaction or waiver of certain conditions. For a more detailed description of these conditions, see "-Conditions to the Distribution."

### **Reasons for the Separation**

The Covidien board of directors determined that the separation of the Pharmaceuticals business from the medical devices and medical supplies businesses would be in the best interests of Covidien and its shareholders and approved the plan of separation. A wide variety of factors were considered by the Covidien board of directors in evaluating the separation. Among other things, the Covidien board of directors considered the following potential benefits of the separation:

*Enhanced business focus*. The separation will allow each of the Pharmaceuticals business and Covidien's other businesses to focus on its own strategic and operational plans and capital structure without diverting human and financial resources to the other business or being constrained by a board

-156-

and management that are also responsible for overseeing and furthering the objectives of the other business. The separation will also enhance the success of each business by reducing internal complexity and enabling each of Covidien and Mallinckrodt to avoid management, systemic and other problems that arise by operation of different businesses within the same corporate structure.

- Business-appropriate capital structure. The separation will enable each of Covidien and Mallinckrodt to pursue the capital structure that is most appropriate for its business and business strategy. Each business has different capital requirements that cannot be optimally addressed with a single capital structure. The separation will permit each of Covidien and Mallinckrodt to pursue a different capital structure that results in a more efficient pricing of its equity in the financial markets.
- Distinct investment identity. The separation will allow Covidien and Mallinckrodt to set new investor expectations for their respective businesses and separate financial prospects based on their unique investment identities, including the merits, performance and future prospects of their respective businesses. The separation will also provide investors with two distinct and targeted investment opportunities and provide a more efficient currency for acquisitions.
- Effectiveness of equity-based compensation. The separation will increase the effectiveness of the equity-based compensation programs of both Covidien and Mallinckrodt by tying the value of the equity compensation awarded to employees, officers or directors more directly to the performance of the business for which these individuals provide services.

Although we believe the above anticipated benefits will be realized, neither Mallinckrodt nor Covidien can assure you that, following the separation, any of the benefits described above or otherwise will be realized to the extent anticipated or at all.

The Covidien board of directors also considered a number of potentially negative factors in evaluating the separation, including the following:

- Loss of synergies and increased costs. As a current part of Covidien, we take advantage of certain functions performed by Covidien, such as
  accounting, tax, legal, human resources and other general and administrative functions. After the separation, Covidien will not perform certain of
  these functions for us, and, because of our smaller scale as a standalone company, our cost of performing such functions will be higher than the
  amounts reflected in our historical financial statements, which will cause our profitability to decrease.
- Disruptions to the business as a result of the separation. The actions required to separate Covidien's and Mallinckrodt's respective businesses could disrupt our operations.
- Increased significance of certain costs and liabilities. Certain costs and liabilities that were otherwise less significant to Covidien as a whole will be more significant for us as a standalone company due to our being smaller than Covidien.
- One-time costs of the separation. We will incur costs in connection with the transition to being a standalone public company that may include accounting, tax, legal, and other professional services costs, recruiting and relocation costs associated with hiring key senior management personnel new to Mallinckrodt, costs related to establishing a new brand identity in the marketplace, tax costs and costs to separate information systems.
- Inability to realize anticipated benefits of the separation. We may not achieve the anticipated benefits of the separation for a variety of reasons, including, among others: (a) the separation will require significant amounts of management's time and effort, which may divert management's attention from operating and growing our business; and (b) following the separation, we may be more susceptible to market fluctuations and other adverse events than if we were still a part of Covidien, because our business will be less diversified than Covidien's business.

-157-

- Limitations placed upon us as a result of the tax matters agreement. In addition, under the terms of the tax matters agreement that we will enter into
  with Covidien, we will be restricted from taking certain actions that could cause the distribution or certain related transactions to fail to qualify as a
  tax-free or tax-favored transaction under applicable law for a period of time. During this period, these restrictions may limit our ability to pursue
  certain strategic transactions and equity issuances or engage in new business or other transactions that might increase the value of our business, over
  some period of time.
- Loss of joint purchasing power. As a current part of Covidien, we take advantage of Covidien's size and purchasing power in procuring certain goods and services. After the separation, as a standalone company, we may be unable to obtain these goods, services, and technologies at prices or on terms as favorable as those Covidien obtained prior to completion of the separation.

In determining to pursue the separation, the Covidien board of directors concluded that the potential benefits of the separation outweighed these factors.

#### When and How You Will Receive Mallinckrodt Ordinary Shares in the Distribution

With the assistance of Computershare, we expect to issue our ordinary shares on June 28, 2013, the distribution date, to all holders of outstanding ordinary shares of Covidien on June 19, the record date. Computershare, which currently serves as the transfer agent and registrar for Covidien's ordinary shares, will serve as the distribution agent in connection with the distribution and the transfer agent and registrar for our ordinary shares.

If you own ordinary shares of Covidien as of the close of business on the record date, Covidien, with the assistance of Computershare, will electronically distribute ordinary shares to you in book-entry form by way of registration in the "direct registration system" (if you hold the shares in your own name as a registered shareholder) or to your bank or brokerage firm on your behalf or through the systems of DTC (if you hold the shares through a bank or brokerage firm that uses DTC).

Direct registration form refers to a method of recording share ownership when no physical share certificates are issued to shareholders, as is the case in this distribution. If you are a registered shareholder, Computershare will then mail you a direct registration account statement that reflects your ordinary shares of Mallinckrodt.

Most Covidien shareholders hold their ordinary shares through a bank or brokerage firm. In such cases, the bank or brokerage firm would be said to hold the shares in "street name" and ownership would be recorded on the bank or brokerage firm's books. If you hold your Covidien ordinary shares through a bank or brokerage firm, your bank or brokerage firm will credit your account for the ordinary shares of Mallinckrodt that you are entitled to receive in the distribution. If you have any questions concerning the mechanics of having shares held in "street name," we encourage you to contact your bank or brokerage firm.

If you sell ordinary shares of Covidien in the "regular-way" market up to and including the distribution date, you will be selling your right to receive ordinary shares of Mallinckrodt in the distribution.

### **Transferability of Shares You Receive**

Our ordinary shares distributed to holders in connection with the distribution will be transferable without registration under the Securities Act, except for shares received by persons who may be deemed to be our affiliates. Persons who may be deemed to be our affiliates after the distribution generally include individuals or entities that control, are controlled by or are under common control with us, which may include certain of our executive officers, directors or principal shareholders. Securities held by our affiliates will be subject to resale restrictions under the Securities Act. Our affiliates will be permitted to sell Mallinckrodt ordinary shares only pursuant to an effective registration statement or an exemption from the registration requirements of the Securities Act, such as the exemption afforded by Rule 144 under the Securities Act.

-158-

## The Number of Ordinary Shares of Mallinckrodt You Will Receive

For every eight Covidien ordinary shares that you own at the close of business on June 19, 2013, the record date, you will receive one ordinary share of Mallinckrodt on the distribution date. Covidien will not distribute any fractional shares to its shareholders. Instead, the transfer agent will aggregate fractional shares into whole shares, sell the whole shares in the open market at prevailing market prices and distribute the aggregate cash proceeds (net of discounts and commissions) of the sales pro rata (based on the fractional share such holder would otherwise be entitled to receive) to each holder who otherwise would have been entitled to receive a fractional share in the distribution. The transfer agent, in its sole discretion, without any influence by Covidien or us, will determine when, how, through which broker-dealer and at what price to sell the whole shares. Any broker-dealer used by the transfer agent will not be an affiliate of either Covidien or us. Neither we nor Covidien will be able to guarantee any minimum sale price in connection with the sale of these shares. Recipients of cash in lieu of fractional shares will not be entitled to any interest on the amounts of payment made in lieu of fractional shares.

The aggregate net cash proceeds of these sales will be taxable for U.S. federal income tax purposes. For an explanation of the material U.S. federal income tax consequences of the distribution, see "Material Tax Consequences—Material U.S. Federal Income Tax Consequences." We estimate that it will take approximately two weeks from the distribution date for the distribution agent to complete the distributions of the aggregate net cash proceeds. If you are the registered holder of ordinary shares of Covidien, you will receive a check from the distribution agent in an amount equal to your pro-rata share of the aggregate net cash proceeds of the sales. If you hold your Covidien ordinary shares through a bank or brokerage firm, your bank or brokerage firm will receive, on your behalf, your pro-rata share of the aggregate net cash proceeds of the sales and will electronically credit your account for your share of such proceeds.

## **Results of the Distribution**

After our separation from Covidien, Mallinckrodt will be a separate, publicly traded company. The actual number of shares to be distributed will be determined after June 19, 2013, the record date for the distribution. The distribution will not affect the number of outstanding ordinary shares of Covidien. No fractional ordinary shares of Mallinckrodt will be distributed.

In connection with the separation, we and Covidien will enter into a separation and distribution agreement and various other agreements, including a transition services agreement, a tax matters agreement and an employee matters agreement. These agreements will effect the separation, provide a framework for our relationship with Covidien after the separation and provide for the allocation between us and Covidien of Covidien's assets, employees, liabilities and obligations (including its property, employee benefits, environmental liabilities and tax liabilities) attributable to periods prior to, at and after our separation from Covidien. For a more detailed description of these agreements, see "Our Relationship with Covidien Following the Distribution."

## Market for Mallinckrodt Ordinary Shares

There is currently no public trading market for our ordinary shares. We intend to apply for authorization to list our ordinary shares on the New York Stock Exchange under the symbol "MNK." We have not and will not set the initial price of our ordinary shares. The initial price will be established by the public markets.

We cannot predict the price at which our ordinary shares will trade after the distribution. In fact, the combined trading prices, after the separation, of our ordinary shares that each Covidien shareholder will receive in the distribution and the ordinary shares of Covidien held at the record date may not equal the "regular-way" trading price of a Covidien share immediately prior to completion of the separation. The price at which our ordinary shares trade may fluctuate significantly, particularly until an orderly public market develops. Trading

-159-

prices for our ordinary shares will be determined in the public markets and may be influenced by many factors. See "Risk Factors—Risks Related to Our Ordinary Shares—A number of Mallinckrodt's ordinary shares are or will be eligible for future sale, which may cause Mallinckrodt's share price to decline."

### Trading Between the Record Date and Distribution Date

Beginning on or shortly before the record date and continuing up to and including the distribution date, Covidien expects that there will be two markets in Covidien ordinary shares: a "regular-way" market and an "ex-distribution" market. Covidien ordinary shares that trade on the "regular-way" market will trade with an entitlement to our ordinary shares distributed pursuant to the distribution. Covidien ordinary shares that trade on the "ex-distribution" market will trade without an entitlement to our ordinary shares distributed pursuant to the distribution. Therefore, if you sell ordinary shares of Covidien in the "regular-way" market up to and including through the distribution date, you will be selling your right to receive our ordinary shares in the distribution. If you own Covidien ordinary shares at the close of business on the record date and sell those shares on the "ex-distribution" market up to and including through the distribution date, you will receive pursuant to your ownership as of the record date of Covidien ordinary shares.

Furthermore, beginning on or shortly before the record date and continuing up to and including the distribution date, we expect that there will be a "whenissued" market in our ordinary shares. "When-issued" trading refers to a sale or purchase made conditionally because the security has been authorized but not yet issued. The "when-issued" trading market will be a market for our ordinary shares that will be distributed to holders of Covidien ordinary shares on the distribution date. If you owned Covidien ordinary shares at the close of business on the record date, you would be entitled to our ordinary shares distributed pursuant to the distribution. You may trade this entitlement to our ordinary shares, without the Covidien ordinary shares you own, on the "when-issued" market. On the first trading day following the distribution date, "when-issued" trading with respect to our ordinary shares will end, and "regular-way" trading will begin.

#### **Conditions to the Distribution**

We expect that the distribution will be effective on June 28, 2013, which is the distribution date, provided that the following conditions have been satisfied (or waived by Covidien in its sole discretion):

- the continued validity of the IRS ruling, which remains in full force and effect and has not been modified or amended in any respect adversely affecting the intended tax-free treatment of the distribution and certain related transactions;
- the receipt of the tax opinion dated as of the distribution date from Skadden, Arps, Slate, Meagher & Flom LLP, in form and substance acceptable to Covidien, which tax opinion will rely on the effectiveness of the IRS ruling, substantially to the effect that, for U.S. federal income tax purposes, the distribution and certain related transactions will qualify as transactions under Sections 355 and/or 368(a) of the Code;
- the internal restructuring transactions and the transfer of assets and liabilities to Mallinckrodt contemplated by the separation and distribution agreement to be completed prior to the distribution shall have been completed;
- the transaction agreements relating to the separation shall have been duly executed and delivered by the parties;
- no order, injunction or decree issued by any court of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the separation or any of the related transactions being in effect;
- any actions and filings necessary or appropriate under applicable U.S. federal, U.S. state or other securities laws shall have been taken and, where applicable, have become effective or been accepted by the applicable governmental authority;

-160-

- any governmental approvals necessary to consummate the separation and related transactions will have been obtained and be in full force and effect;
- the separation shall not violate or result in a breach of applicable law or any material contract of Covidien or Mallinckrodt or any of their respective subsidiaries;
- the SEC declaring effective the registration statement of which this information statement forms a part, with no order suspending the effectiveness of the registration statement in effect and no proceedings for such purposes pending before or threatened by the SEC;
- the mailing of this information statement to the holders of Covidien ordinary shares as of the record date for the distribution; and
- no other event or development existing or having occurred that, in the judgment of Covidien's board of directors, in its sole discretion, makes it inadvisable to effect the separation and other related transactions.

As of the date of this information statement, the following additional conditions have been satisfied:

- the debt financing contemplated to be obtained in connection with the separation, as described in the separation and distribution agreement, having been obtained;
- the receipt of opinions, in form and substance acceptable to Covidien in its sole discretion and from an independent firm acceptable to Covidien in its sole discretion, with respect to the solvency of each of Covidien and Mallinckrodt; and
- the approval for listing on the NYSE of our ordinary shares to be delivered in the distribution having been obtained.

Covidien will have the sole and absolute discretion to determine (and change) the terms of, and whether to proceed with, the distribution and, to the extent it determines to so proceed, to determine the record date, the distribution date and the distribution ratio. Covidien does not intend to notify its shareholders of any modifications to the terms of the separation that, in the judgment of its board of directors, are not material. For example, the Covidien board of directors might consider material such matters as significant changes to the distribution ratio, the assets to be contributed or the liabilities to be assumed in the separation. To the extent that the Covidien board of directors determines that any modifications by Covidien materially change the material terms of the distribution, Covidien will notify Covidien shareholders in a manner reasonably calculated to inform them about the modification as may be required by law, by, for example, publishing a press release, filing a current report on Form 8-K or circulating a supplement to this information statement.

-161-

## OUR RELATIONSHIP WITH COVIDIEN FOLLOWING THE DISTRIBUTION

Following the separation, we and Covidien will operate as separate, independent public companies. In connection with the separation, we and Covidien will enter into certain agreements to provide a framework for our relationship with Covidien after the separation and provide for the allocation between us and Covidien of Covidien's assets, employees, liabilities and obligations (including its property, employee benefits, environmental liabilities) attributable to periods prior to, at and after our separation from Covidien. The following is a summary of the terms of the material agreements that we intend to enter into with Covidien in connection with the separation.

The material agreements described below will be or have been filed as exhibits to the registration statement on Form 10 of which this information statement is a part (the "Form 10"). The summaries of each of these agreements set forth the terms of the agreements that we believe are material. These summaries are qualified in their entireties by reference to the full text of the applicable agreements, which are incorporated by reference into this information statement.

### Separation and Distribution Agreement

The separation and distribution agreement will set forth the agreements between us and Covidien regarding the principal corporate transactions required to effect our separation from Covidien and other agreements governing our relationship with Covidien.

The separation and distribution agreement will identify assets to be transferred, liabilities to be assumed and contracts to be assigned to each of us and Covidien as part of the separation, and it will provide for when and how these transfers, assumptions and assignments will occur. In particular, the separation and distribution agreement will provide, among other things, that, subject to the terms and conditions contained therein:

- certain assets related to the businesses and operations of Covidien's Pharmaceuticals business (and certain legacy businesses and operations of Mallinckrodt entities), which we refer to as the Mallinckrodt Assets, will be transferred to us or one of our subsidiaries;
- certain liabilities (including whether accrued, contingent or otherwise) arising out of or resulting from the Mallinckrodt Assets, and other liabilities
  related to the businesses and operations of Covidien's Pharmaceuticals business (and certain legacy businesses and operations of Mallinckrodt
  entities), which we refer to as the Mallinckrodt Liabilities, will be retained by or transferred to us or one of our subsidiaries;
- all of the assets and liabilities (including whether accrued, contingent or otherwise) other than the Mallinckrodt Assets and Mallinckrodt Liabilities (such assets and liabilities, other than the Mallinckrodt Assets and the Mallinckrodt Liabilities, are referred to as the Excluded Assets and Excluded Liabilities, respectively) will be retained by or transferred to Covidien or one of its subsidiaries; and
- certain shared contracts will be assigned, in part to us or our applicable subsidiaries or be appropriately amended.

Except as may expressly be set forth in the separation and distribution agreement or any other transaction agreements, all assets will be transferred on an "as is," "where is" basis and the respective transferees will bear the economic and legal risks that (1) any conveyance will prove to be insufficient to vest in the transferee good title, free and clear of any security interest, and (2) any necessary consents or governmental approvals are not obtained or any requirements of laws or judgments are not complied with. In general, each party to the separation and distribution agreement will assume liability for all pending, threatened and unasserted legal matters related to its own business or its assumed or retained liabilities and will indemnify the other party for any liability to the

-162-

extent arising out of or resulting from such assumed or retained legal matters. In addition, the separation and distribution agreement will provide for crossindemnities principally designed to place financial responsibility for the obligations and liabilities of our business with us and financial responsibility for the obligations and liabilities of Covidien's remaining business with Covidien, among other indemnities. Specifically, each of Covidien and Mallinckrodt will indemnify, defend and hold harmless the other party, its subsidiaries and their respective directors, officers, employees and agents against any losses arising out of or resulting from:

- the liabilities that each such party assumed or retained pursuant to the separation and distribution agreement (which, in the case of Mallinckrodt, would include the Mallinckrodt Liabilities and, in the case of Covidien, would include the Excluded Liabilities); and
- any breach by such party of the separation and distribution agreement or the other transaction agreements.

Also, we will indemnify, defend and hold harmless Covidien, its subsidiaries and their respective directors, officers, employees and agents from and against any losses arising out of or resulting from:

- the operation of our business;
- except to the extent it relates to an Excluded Liability, any guarantee, indemnification obligation, letter of credit reimbursement obligation, surety bond or other credit support agreement, arrangement, commitment or understanding for the benefit of Mallinckrodt or its subsidiaries by Covidien or any of its subsidiaries that survives following the distribution; and
- any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, with respect to all information contained in the Form 10 (as defined below), this information statement (as amended or supplemented), the offering memorandum for the April 2013 notes offering or any other disclosure document that describes the separation or the distribution or Mallinckrodt and its subsidiaries or primarily relates to the transactions contemplated by the separation and distribution agreement.

In addition, Covidien will indemnify, defend and hold harmless Mallinckrodt, its subsidiaries and their respective directors, officers, employees and agents from and against any losses arising out of or resulting from:

- Covidien's business other than the Pharmaceuticals business (except to the extent it relates to a Mallinckrodt Liability and other than the conduct of business, operations or activities for the benefit of Mallinckrodt or its subsidiaries pursuant to the separation and distribution agreement, the transition services agreement, the tax matters agreement or the employee matters agreement); and
- the investigation and remediation of sites in Orrington, Maine and Penobscot River and Bay (as described in note 12 to our interim unaudited condensed combined financial statements and note 20 to our annual combined financial statements included elsewhere in this information statement).

The separation and distribution agreement also will specify procedures with respect to claims subject to indemnification and related matters.

To the extent that any transfers contemplated by the separation and distribution agreement have not been consummated on or prior to the distribution date, the parties will agree to cooperate to effect such transfers as promptly as practicable following the distribution date. In addition, each of the parties will agree to cooperate with the other party and use commercially reasonable efforts to take or to cause to be taken all actions, and to do, or to cause to be done, all things reasonably necessary under applicable law or contractual obligations to consummate and make effective the transactions contemplated by the separation and distribution agreement and the other transaction agreements.

-163-

The separation and distribution agreement also will govern the rights and obligations of Covidien and us regarding the distribution. The separation and distribution agreement will provide that Covidien's obligation to complete the distribution is subject to several conditions that must be satisfied (or waived by Covidien in its sole discretion), which are described in "The Separation—Conditions to the Distribution." We will cooperate with Covidien to accomplish the distribution and will, at Covidien's direction, promptly take any and all actions necessary or desirable to effect the distribution.

The separation and distribution agreement will also provide for an adjustment payment to potentially be made following the distribution from Covidien to us, or from us to Covidien. The purpose of the adjustment payment is to compensate Mallinckrodt or Covidien, as applicable, to the extent that the aggregate of our cash, indebtedness and specified working capital accounts as of the distribution date, as well as the capital expenditures made with respect to our business during fiscal 2013 through the distribution date, deviates from a target. The target will be calculated pursuant to a formula that will be set forth in the separation and distribution agreement, which was determined assuming that the distribution date is June 28, 2013, that our business is conducted in the ordinary course through that date and that we will have approximately \$168 million of cash upon completion of the distribution agreement will also provide that an adjustment payment will only be payable if the amount of the adjustment payment exceeds \$20 million (in which case the entire amount will be paid).

Under the separation and distribution agreement, following the separation, we and Covidien will be obligated to provide each other access to information in certain circumstances. The separation and distribution agreement also will impose obligations with respect to retention of information and confidentiality.

The separation and distribution agreement will provide for the allocation among the parties of rights and obligations under existing insurance policies with respect to occurrences prior to completion of the separation and will set forth procedures for the administration of insured claims. In addition, the separation and distribution agreement will allocate between the parties the right to proceeds and the obligation to incur certain deductibles under certain insurance policies.

The separation and distribution agreement may be terminated and the distribution may be amended, modified or abandoned at any time prior to the distribution by Covidien.

#### **Transition Services Agreement**

We and Covidien will enter into a transition services agreement in connection with the separation pursuant to which we and Covidien and our respective affiliates will provide each other, on an interim, transitional basis, various services, including, but not limited to, treasury administration, employee benefits administration, information technology services, non-exclusive distribution and importation services for our products in certain countries outside the United States, regulatory, general administrative services and other support services. The agreed-upon charges for such services are generally intended to allow the servicing party to recover all out-of-pocket costs and expenses and a predetermined profit equal to a mark-up of such out-of-pocket expenses. The party receiving each transition service will be provided with reasonable information that supports the charges for such transition service by the party providing the service.

The services generally will commence on the distribution date and terminate up to 24 months following the distribution date. The receiving party may terminate certain specified services by giving prior written notice to the provider of such services and paying specified wind-down charges.

Subject to certain exceptions, the liabilities of each party providing services under the transition services agreement will generally be limited to the aggregate charges (excluding any third-party costs and expenses



included in such charges) actually paid to such party by the other party pursuant to the transition services agreement. The transition services agreement also will provide that the provider of a service will not be liable to the recipient of such service for any special, indirect, incidental or consequential damages.

### **Tax Matters Agreement**

In connection with the separation, we will enter into a tax matters agreement with Covidien that generally will govern Covidien's and our respective rights, responsibilities and obligations after the distribution with respect to certain taxes, including ordinary course of business taxes and taxes, if any, incurred as a result of any failure of the distribution of our shares to qualify as a tax-free distribution for U.S. federal income tax purposes within the meaning of Section 355 of the Code or other applicable tax law or any failure of certain internal transactions undertaken in anticipation of the distribution to qualify for tax-free or tax-favored treatment under the applicable tax law. The agreement will also assign rights and responsibilities for administrative matters, such as the filing of returns, payment of taxes due, retention of records, tax reporting practices and conduct of audits, examinations or similar proceedings. In addition, the agreement provides for cooperation and information sharing with respect to tax matters.

Under the tax matters agreement, we expect, with certain exceptions, that we will generally be responsible for the payment of:

- All taxes attributable to us or our subsidiaries for taxable periods beginning on or after September 29, 2012; and
- To the extent that our liability for such taxes (after taking into account certain tax benefits realized by us) does not, in the aggregate, exceed \$200 million, taxes attributable to the following:
  - Taxes attributable to us or our subsidiaries for taxable periods beginning before September 29, 2012;
  - Certain taxes related to the separation; and
  - 20% of certain taxes arising from a failure of the distribution or any internal transaction undertaken in anticipation of the distribution, to qualify for tax-free or tax-favored treatment under applicable tax law through no fault of us or Covidien.

Our potential liability for any taxes related to periods prior to the distribution (after taking into account certain tax benefits realized by us), including those which are subject to the provisions of the Tyco tax sharing agreement, is anticipated to be approximately \$150 million. The tax matters agreement also will contain restrictions on our ability to take actions without Covidien's consent that could cause the distribution or certain internal transactions undertaken in anticipation of the distribution to fail to qualify as tax-free or tax-favored transactions under applicable tax law, including entering into, approving or allowing any transaction that results in a change in ownership of more than 35% of our shares; any merger, consolidation, scheme of arrangement, liquidation or partial liquidation, or any approval or allowance of such transaction with respect to certain of our subsidiaries; the cessation or transfer of certain business activities; the sale, issuance or other disposition of any equity interest in certain of our subsidiaries; a sale or other disposition of a substantial portion of our assets or a substantial portion of the assets of certain of our subsidiaries; extraordinary distributions by or to certain of our subsidiaries; or engaging in certain internal transactions. These restrictions will all apply for the two-year period after the distribution and in some cases will apply for periods as long as five years following the distribution.

Moreover, the tax matters agreement generally will provide that a party thereto is responsible for any taxes imposed on any other party thereto as a result of the failure of the distribution or the internal transactions to qualify as tax-free or tax-favored transactions under the Code or other applicable tax law if such failure is attributable to certain post-distribution actions taken by or in respect of the responsible party or its shareholders, regardless of whether the actions occur more than two years after the distribution, or Covidien consents to such actions. Any such taxes for which we are liable as a result of our actions or the actions of our shareholders will not be subject to the \$200 million limitation described above.

-165-

### **Employee Matters Agreement**

Mallinckrodt and Covidien will enter into an employee matters agreement in connection with the separation to allocate assets, liabilities, and responsibilities and obligations relating to employment matters, employee compensation, employee benefit plans, programs, arrangements and agreements and other related matters. In connection with the separation, Covidien has or currently intends to transfer the employment of employees who will continue in employment with Mallinckrodt after the separation to an entity that will be within the Mallinckrodt controlled group after the separation. Also, Mallinckrodt has either assumed sponsorship of or adopted various employee benefit plans, including United States tax-qualified retirement plans, non-qualified deferred compensation plans and health and welfare benefit plans, that provide benefits to eligible current and former United States employees of Covidien's Pharmaceuticals business.

It is currently contemplated that the employee matters agreement will allocate responsibility to Mallinckrodt to continue to provide employee benefits to employees of Covidien's Pharmaceuticals business (i.e., Mallinckrodt employees) upon the separation and to assume the responsibility for any assets and liabilities associated with the plans or programs providing such employee benefits. It is currently anticipated that Mallinckrodt employees who are resident outside of the United States or who otherwise are subject to non-U.S. law and their related benefits and obligations shall be treated in the same manner as the Mallinckrodt employees who are residents of the United States; provided, however, that all actions taken with respect to non-U.S. Mallinckrodt employees in connection with the separation will be accomplished in accordance with applicable law and custom in each of the applicable jurisdictions. In addition, it is currently contemplated that outstanding Covidien equity awards held by active employees of Covidien's Pharmaceuticals business upon the separation will be converted into Mallinckrodt equity awards in connection with the distribution, except in certain non-U.S. jurisdictions where such conversion would either have an adverse tax impact or be subject to local exchange control requirements that make such conversion impracticable or impossible. The mechanics of this conversion will be set forth in the employee matters agreement. Finally, it is currently anticipated that the employee matters agreement and any Mallinckrodt employee benefit plan, program, agreement or arrangement assumed or adopted in anticipation of the separation; and (ii) the distribution and the continuation of employee benefit plan, program, agreement or arrangement.

-166-

### MATERIAL TAX CONSEQUENCES

## Material U.S. Federal Income Tax Consequences

The following is a summary of the material U.S. federal income tax consequences to Covidien and to the holders of Covidien ordinary shares in connection with the distribution. This summary is based on the Code, the Treasury Regulations promulgated thereunder and judicial and administrative interpretations thereof, in each case as in effect and available as of the date of this information statement and all of which are subject to differing interpretations that may change at any time, possibly with retroactive effect. Any such change could affect the tax consequences described below.

Except as specifically described below, this summary is limited to holders of Covidien ordinary shares that are U.S. Holders (as defined below). For purposes of this summary, a U.S. Holder is a beneficial owner of Covidien ordinary shares that is, for U.S. federal income tax purposes:

- an individual who is a citizen or resident of the U.S.;
- a corporation, or other entity taxable as a corporation for U.S. federal income tax purposes, created or organized under the laws of the U.S. or any state or political subdivision thereof;
- an estate, the income of which is subject to U.S. federal income taxation regardless of its source; or
- a trust, if (i) a court within the U.S. is able to exercise primary jurisdiction over its administration and one or more U.S. persons have authority to control all of its substantial decisions, or (ii) it has a valid election in place under applicable Treasury Regulations to be treated as a U.S. person.

This summary does not discuss all tax considerations that may be relevant to Covidien shareholders in light of their particular circumstances, nor does it address the consequences to Covidien shareholders subject to special treatment under the U.S. federal income tax laws, such as:

- dealers or traders in securities or currencies;
  - tax-exempt entities;
  - cooperatives;
  - banks, trusts, financial institutions or insurance companies;
  - persons who acquired Covidien ordinary shares pursuant to the exercise of employee share options or otherwise as compensation;
  - persons who own, or are deemed to own, at least 10 percent or more, by voting power or value, of the Covidien ordinary shares;
  - holders owning Covidien ordinary shares as part of a position in a straddle or as part of a hedging, conversion or other risk reduction transaction for U.S. federal income tax purposes;
  - certain former citizens or long-term residents of the U.S.;
  - holders who are subject to alternative minimum tax; or
  - persons that own Covidien ordinary shares through partnerships (including entities treated as partnerships for U.S. federal income tax purposes) or other pass-through entities.

This summary does not address the U.S. federal income tax consequences to Covidien shareholders who do not hold Covidien ordinary shares as capital assets. Moreover, this summary does not address any state, local or non-U.S. tax consequences or any estate, gift or other non-income tax consequences.

If a partnership (or any other entity treated as a partnership for U.S. federal income tax purposes) holds Covidien ordinary shares, the tax treatment of a partner in that partnership generally will depend on the status of

-167-

the partner and the activities of the partnership. Such a partner or partnership should consult its own tax advisor as to the tax consequences of the separation.

## HOLDERS OF COVIDIEN ORDINARY SHARES SHOULD CONSULT THEIR OWN TAX ADVISORS WITH RESPECT TO THE SPECIFIC U.S. FEDERAL, STATE AND LOCAL AND NON-U.S. TAX CONSEQUENCES OF THE DISTRIBUTION IN LIGHT OF THEIR PARTICULAR CIRCUMSTANCES AND THE EFECT OF POSSIBLE CHANGES IN LAW THAT MIGHT AFFECT THE TAX CONSEQUENCES DESCRIBED HEREIN.

Covidien has received an IRS ruling substantially to the effect that, for U.S. federal income tax purposes, (i) certain transactions to be effected in connection with the distribution qualify as transactions under Sections 355 and/or 368(a) of the Code, and (ii) the distribution qualifies as a transaction under Sections 355 and 368(a)(1)(D) of the Code.

In addition to obtaining the IRS ruling, Covidien expects to receive the tax opinion from Skadden, Arps, Slate, Meagher & Flom LLP, in form and substance acceptable to Covidien, which tax opinion will rely on the effectiveness of the IRS ruling, substantially to the effect that, for U.S. federal income tax purposes, the distribution and certain related transactions will qualify as transactions under Sections 355 and/or 368(a) of the Code. The continued validity of the IRS ruling and the receipt by Covidien of the tax opinion are conditions to the distribution.

Assuming that the distribution qualifies under Sections 355 and 368(a)(1)(D) of the Code, for U.S. federal income tax purposes:

- no gain or loss will be recognized by Covidien on the distribution;
- no gain or loss will be recognized by, or be includible in the income of, a holder of Covidien ordinary shares upon receipt of our ordinary shares in the distribution;
- each Covidien shareholder's basis in the Covidien ordinary shares and the Mallinckrodt ordinary shares following the distribution will equal the aggregate basis of the Covidien ordinary shares that such holder held immediately before the distribution, allocated between the Covidien ordinary shares and the Mallinckrodt ordinary shares in proportion to their relative fair market values at the time of the distribution;
- each Covidien shareholder's holding period in the Mallinckrodt ordinary shares received in the distribution will include the holding period of the Covidien ordinary shares with respect to which the distribution is made, provided that such holder holds such Covidien ordinary shares as a capital asset on the date of the distribution; and
- a Covidien shareholder who receives cash in lieu of fractional Mallinckrodt ordinary shares will recognize gain or loss measured by the difference between the basis of the fraction of a share that the shareholder would have received and the amount of cash received in lieu thereof. Any gain or loss will be treated as a capital gain or loss, provided any fractional shares would have been held as capital assets on the date of the distribution.

Although the IRS ruling is generally binding on the IRS, the IRS ruling is based on certain facts and assumptions, and certain representations and undertakings from Covidien and Mallinckrodt that certain necessary conditions to obtain tax-free treatment under the Code have been satisfied. Furthermore, the IRS did not rule on whether the distribution satisfies certain critical requirements necessary to obtain tax-free treatment under the Code. Specifically, the IRS did not rule that the distribution is effected for valid business purposes, that the distribution does not constitute a device for the distribution of earnings and profits, or that the distribution is not part of a plan described in Section 355(e) of the Code (as discussed below). Instead, Covidien represented to the IRS that there are valid business purposes for the distribution, the distribution is not being used as a device for

-168-

the distribution of earnings and profits, and the distribution is not part of a plan described in Section 355(e) of the Code. In connection with obtaining the IRS ruling, Covidien expects to receive a tax opinion. The tax opinion will be expressed as of the date issued and will not cover subsequent periods, and the tax opinion will rely on the effectiveness of the IRS ruling. As a result, the tax opinion is not expected to be issued until after the date of this information statement. An opinion of counsel represents the counsel's best legal judgment based on current law and is not binding on the IRS or any court. We cannot assure you that the IRS will agree with the conclusions expected to be set forth in the tax opinion, and it is possible that the IRS or another tax authority could adopt a position contrary to one or all of those conclusions and that a court could sustain that contrary position. If any of the facts, representations, assumptions or undertakings described or made in connection with the IRS ruling or the tax opinion are not correct, are incomplete or have been violated, the IRS ruling could be revoked retroactively or modified by the IRS, and Covidien's ability to rely on the tax opinion could be jeopardized. Covidien and Mallinckrodt are not aware of any facts or circumstances, however, that would cause these facts, representations or assumptions to be untrue or incomplete, or that would cause any of these undertakings to fail to be complied with, in any material respect.

If, notwithstanding the conclusions included in the IRS ruling and the conclusions we expect to be included in the tax opinion, it is ultimately determined that the distribution does not qualify as a tax-free transaction for U.S. federal income tax purposes under Sections 355 and 368(a)(1)(D) of the Code, Covidien or we could incur significant U.S. federal income tax liabilities attributable to certain related transactions undertaken in anticipation of the distribution. In addition, if the distribution was not to qualify as a tax-free transaction for U.S. federal income tax purposes under Sections 355 and 368(a)(1)(D) of the Code, each Covidien shareholder that receives our ordinary shares in the distribution could be treated as receiving a taxable distribution in an amount equal to the fair market value of our ordinary shares that were distributed to the shareholder, which generally would be taxed as a dividend to the extent of the shareholder's pro-rata share of Covidien's current and accumulated earnings and profits, then treated as a non-taxable return of capital to the extent of the shareholder's basis in its Covidien ordinary shares and finally treated as capital gain from the sale or exchange of its Covidien ordinary shares.

Even if the distribution and the related transactions otherwise qualify for tax-free treatment under Sections 355 and/or 368(a) of the Code, corporate-level taxable gain under Section 355(e) of the Code may result if fifty percent or more, by vote or value, of our ordinary shares or Covidien ordinary shares is treated as acquired or issued as part of a plan or series of related transactions that include the distribution or such related transactions. The process for determining whether an acquisition or issuance triggering these provisions has occurred is complex, inherently factual and subject to interpretation of the facts and circumstances of a particular case. For this purpose, any acquisitions or issuances of Covidien ordinary shares within two years after the distribution generally are presumed to be part of such a plan, although we or Covidien, as applicable, may be able to rebut that presumption. We are not aware of any acquisitions or issuances of Covidien ordinary shares triggers the aplan or series of related transactions apart of a plan or series of related transactions or issuance of our ordinary shares or Covidien ordinary shares triggers the application Section 355(e) of the Code, Covidien or we could incur significant U.S. federal income tax liabilities attributable to certain related transactions undertaken in anticipation of the distribution.

The Treasury Regulations require certain shareholders that receive stock in the distribution to attach a detailed statement setting forth certain information relating to the separation to their U.S. federal income tax returns for the year in which the distribution occurs. Within a reasonable period after the distribution, Covidien will provide shareholders who receive our ordinary shares in the distribution with the information necessary to comply with such requirement. In addition, all shareholders are required to retain permanent records relating to the amount, basis and fair market value of our ordinary shares received in the distribution and to make those records available to the IRS upon request.

-169-

## **Material Irish Tax Consequences**

The following is a summary of the material Irish tax consequences for certain beneficial owners of Covidien ordinary shares who receive Mallinckrodt ordinary shares pursuant to the separation and who are the beneficial owners of such Mallinckrodt ordinary shares. The summary does not purport to be a comprehensive description of all of the tax considerations that may be relevant to each of the shareholders. The summary is based upon Irish tax laws and the practice of the Irish Revenue Commissioners in effect on the date of this information statement and correspondence with the Irish Revenue Commissioners. Changes in law and/or administrative practice may result in alteration of the tax considerations described below.

The summary does not constitute tax advice and is intended only as a general guide. The summary is not exhaustive and shareholders should consult their own tax advisors about the Irish tax consequences (and tax consequences under the laws of other relevant jurisdictions) of the separation and of the acquisition, ownership and disposal of our ordinary shares. The summary applies only to shareholders who will own our ordinary shares as capital assets and does not apply to other categories of shareholders, such as dealers in securities, trustees, insurance companies, collective investment schemes and shareholders who have, or who are deemed to have, acquired our ordinary shares by virtue of an Irish office or employment (performed or carried on in Ireland).

#### Irish Tax on Chargeable Gains

*Non-resident Shareholders.* The rate of tax on chargeable gains (where applicable) in Ireland is 33%. Our shareholders that are not resident or ordinarily resident in Ireland for Irish tax purposes and do not hold their shares in connection with a trade or business carried on by such shareholders through an Irish branch or agency will not be liable for Irish tax on chargeable gains realized on a subsequent disposal of our ordinary shares.

Covidien shareholders that are not resident or ordinarily resident in Ireland for Irish tax purposes and do not hold their shares in connection with a trade or business carried on by such shareholders through an Irish branch or agency will not be subject to Irish tax on chargeable gains on the receipt of new Mallinckrodt ordinary shares pursuant to the separation.

*Irish Resident Shareholders.* Our shareholders that are resident or ordinarily resident in Ireland for Irish tax purposes, or that hold their shares in connection with a trade or business carried on by such persons through an Irish branch or agency will, subject to the availability of any exemptions and reliefs, be subject to Irish tax on chargeable gains arising on a subsequent disposal of our ordinary shares.

Covidien shareholders that are resident or ordinarily resident in Ireland for Irish tax purposes, or shareholders that hold their shares in connection with a trade or business carried on by such persons through an Irish branch or agency, will not be subject to Irish tax on chargeable gains on the receipt of new Mallinckrodt ordinary shares pursuant to the separation but will rather be treated for Irish tax purposes as having acquired their shares in Mallinckrodt at the same time and for the same cost as they acquired their original shares in Covidien. Such shareholders may, however, be subject to Irish tax on chargeable gains on the receipt of any cash in lieu of fractional shares pursuant to the separation as they will be deemed to have made a part disposal of their shares in Covidien.

#### Stamp Duty

The rate of stamp duty (where applicable) on transfers of shares of Irish incorporated companies is 1% of the price paid or the market value of the shares acquired, whichever is greater. Where Irish stamp duty arises, it is generally a liability of the transferee.

The distribution will be exempt from the charge to Irish stamp duty.

-170-

Irish stamp duty may, depending on the manner in which the shares in Mallinckrodt are held, be payable in respect of transfers of Mallinckrodt ordinary shares after the separation.

Shares Held Through DTC. A transfer of our ordinary shares effected by means of the transfer of book entry interests in DTC will not be subject to Irish stamp duty. On the basis that most of our ordinary shares are expected to be held through DTC, it is anticipated that most transfers of ordinary shares will be exempt from Irish stamp duty.

Shares Held Outside of DTC or Transferred Into or Out of DTC. A transfer of our ordinary shares where any party to the transfer holds such shares outside of DTC may be subject to Irish stamp duty. Shareholders wishing to transfer their shares into (or out of) DTC may do so without giving rise to Irish stamp duty provided:

- there is no change in the beneficial ownership of such shares; and
- at the time of the transfer into DTC there is no agreement in place for the sale of the shares by the beneficial owner to a third party.

Due to the potential Irish stamp charge on transfers of our ordinary shares, it is strongly recommended that any person who wishes to acquire our ordinary shares after the separation acquires such shares through DTC (or through a broker who in turn holds such shares through DTC).

Mallinckrodt currently intends to pay (or cause one of our affiliates to pay) stamp duty, if any, in connection with share transfers made in the ordinary course of trading by a seller who holds shares directly to a buyer who will hold the acquired shares beneficially. In other cases Mallinckrodt may, in its absolute discretion, pay (or cause one of its affiliates to pay) any stamp duty. Mallinckrodt's articles of association as they will be in effect after the distribution provide that, in the event of any such payment, Mallinckrodt (i) may seek reimbursement from the buyer, (ii) will have a lien against the Mallinckrodt ordinary shares acquired by such buyer and any dividends paid on such shares and (iii) may set-off the amount of the stamp duty against future dividends on such shares. Parties to a share transfer may assume that any stamp duty arising in respect of a transaction in Mallinckrodt ordinary shares has been paid unless one or both of such parties is otherwise notified by Mallinckrodt.

## Withholding Tax on Dividends

Distributions made by us will, in the absence of one of many exemptions, be subject to Irish dividend withholding tax ("DWT") at a rate of 20%.

For DWT purposes, a distribution includes any distribution that may be made by us to our shareholders, including cash dividends, non-cash dividends and additional stock taken in lieu of a cash dividend. Where an exemption does not apply in respect of a distribution made to a particular shareholder, we are responsible for withholding DWT prior to making such distribution.

*General Exemptions.* Irish domestic law provides that a non-Irish resident shareholder is not subject to DWT on dividends received from us if such shareholder is beneficially entitled to the dividend and is either:

- an individual resident for tax purposes in a "relevant territory" (including the U.S.) and is neither resident nor ordinarily resident in Ireland (for a list of "relevant territories" for DWT purposes, please see Annex A to this information statement);
- a company resident for tax purposes in a "relevant territory," provided such company is not under the control, whether directly or indirectly, of a person or persons who is or are resident in Ireland;
- a company, wherever resident, that is controlled, directly or indirectly, by persons resident in a "relevant territory" and who is or are (as the case may be) not controlled by, directly or indirectly, persons who are not resident in a "relevant territory";

-171-

- a company, wherever resident, whose principal class of shares (or those of its 75% direct or indirect parent) is substantially and regularly traded on a recognized stock exchange either in a "relevant territory" or on such other stock exchange approved by the Irish Minister for Finance; or
- a company, wherever resident, that is wholly owned, directly or indirectly, by two or more companies where the principal class of shares of each of such companies is substantially and regularly traded on a recognized stock exchange in a "relevant territory" or on such other stock exchange approved by the Irish Minister for Finance;

and provided, in all cases noted above, the shareholder has furnished the relevant Irish Revenue Commissioners' DWT forms (the "DWT Forms") to:

- its broker (and the relevant information is further transmitted to us or any qualifying intermediary appointed by us) before the record date for the dividend if its shares are held through DTC, or
- our transfer agent at least seven business days before such record date if its shares are held outside of DTC.

Links to the various DWT Forms are available at: http://www.revenue.ie/en/tax/dwt/forms/index.html.

For shareholders that cannot avail themselves of one of Ireland's domestic law exemptions from DWT, it may be possible for such shareholders to rely on the provisions of a double tax treaty to which Ireland is party to reduce the rate of DWT.

Shares Held by U.S. Resident Shareholders. Dividends paid in respect of our ordinary shares that are owned by U.S. residents and held through DTC will not be subject to DWT provided the addresses of the beneficial owners of such shares in the records of the broker holding such shares are in the U.S. It is strongly recommended that such shareholders ensure that their information is properly recorded by their brokers (so that such brokers can further transmit the relevant information to a qualifying intermediary appointed by us).

Dividends paid in respect of our ordinary shares that are owned by residents of the U.S. and held outside of DTC will not be subject to DWT if such shareholders provide a completed W-9 form to Computershare, our transfer agent, to confirm their U.S. residence at least seven business days before the record date for the first dividend payment to which they are entitled. It is strongly recommended that such shareholders complete a W-9 form and provide it to our transfer agent as soon as possible after acquiring their shares.

If any shareholder that is resident in the U.S. receives a dividend from which DWT has been withheld, the shareholder may be entitled to apply for a refund of such DWT from the Irish Revenue Commissioners.

Shares Held by Residents of "Relevant Territories" Other than the U.S. Shareholders that are residents of "relevant territories," other than the U.S. and regardless of when such shareholders acquired their shares, must satisfy the conditions of one of the exemptions referred to above under the heading "—General Exemptions," including the requirement to furnish completed DWT Forms, in order to receive dividends without them being subject to DWT. If such shareholders hold their shares through DTC, they must provide the appropriate DWT Forms to their brokers (so that such brokers can further transmit the relevant information to a qualifying intermediary appointed by us) before the record date for the first dividend to which they are entitled. If such shareholders hold their shares outside of DTC, they must provide the appropriate DWT Forms to our transfer agent at least seven business days before such record date. It is strongly recommended that such shareholders complete the appropriate DWT Forms and provide them to their brokers or our transfer agent, Computershare, as the case may be, as soon as possible.

If any shareholder who is resident in a "relevant territory" receives a dividend from which DWT has been withheld, the shareholder may be entitled to a refund of DWT from the Irish Revenue Commissioners.

-172-

Shares Held by Residents of Ireland. Most Irish tax resident or ordinarily resident shareholders will be subject to DWT in respect of dividends paid on our ordinary shares.

Shareholders that are residents of Ireland, but are entitled to receive dividends without DWT, must complete the appropriate DWT Forms and provide them to their brokers (so that such brokers can further transmit the relevant information to a qualifying intermediary appointed by us) before the record date for the first dividend to which they are entitled (in the case of shares held through DTC), or to our transfer agent Computershare at least seven business days before such record date (in the case of shares held outside of DTC).

Shares Held by Other Persons. Our shareholders that do not fall within any of the categories specifically referred to above may nonetheless fall within other exemptions from DWT. If any shareholders are exempt from DWT, but receive dividends subject to DWT, such shareholders may apply for refunds of such DWT from the Irish Revenue Commissioners.

Shares Held by Existing Covidien Shareholders. To the extent that existing Covidien shareholders resident in the U.S. or another relevant territory have previously provided our transfer agent, Computershare, or any qualifying intermediary appointed by us with appropriate forms or addresses to support their claim for an exemption from Irish DWT in respect of their shareholding in Covidien, Computershare and that qualifying intermediary (and any other qualifying intermediary in the payment chain) can rely upon these forms and addresses and will not be required to obtain new documentation from such shareholders until these forms have expired or these addresses have changed.

*Qualifying Intermediary.* Prior to paying any dividend, we will put in place an agreement with an entity that is recognized by the Irish Revenue Commissioners as a "qualifying intermediary," which will provide for certain arrangements relating to distributions in respect of our ordinary shares that are held through DTC (the "Deposited Securities"). The agreement will provide that the qualifying intermediary shall distribute or otherwise make available to Cede & Co., as nominee for DTC, any cash dividend or other cash distribution with respect to the Deposited Securities after we deliver or cause to be delivered to the qualifying intermediary the cash to be distributed.

We will rely on information received directly or indirectly from our qualifying intermediary, brokers and our transfer agent in determining where shareholders reside, whether they have provided the required U.S. tax information and whether they have provided the required DWT Forms. Shareholders that are required to file DWT Forms in order to receive dividends free of DWT should note that such forms are generally valid, subject to a change in circumstances, until December 31 of the fifth full year after the year of issue of the forms.

#### Income Tax on Dividends Paid on Mallinckrodt Shares

Irish income tax may arise for certain persons in respect of dividends received from Irish resident companies.

A shareholder that is not resident or ordinarily resident in Ireland and that is entitled to an exemption from DWT generally has no liability to Irish income tax or the universal social charge on a dividend from us. An exception to this position may apply where such shareholder holds our ordinary shares through a branch or agency in Ireland through which a trade is carried on.

A shareholder that is not resident or ordinarily resident in Ireland and that is not entitled to an exemption from DWT generally has no additional Irish income tax liability or a liability to the universal social charge. The DWT deducted by us discharges the liability to income tax. An exception to this position may apply where the shareholder holds our ordinary shares through a branch or agency in Ireland through which a trade is carried on.

Irish resident or ordinarily resident shareholders may be subject to Irish tax and/or the universal social charge on dividends received from us.

-173-

## **Capital Acquisitions Tax**

Irish Capital Acquisitions Tax ("CAT") could apply to a gift or inheritance of Irish situate shares irrespective of the place of residence, ordinary residence or domicile of the parties. Our ordinary shares held in book entry form may be regarded as property situated in Ireland as our share register must be held in Ireland. The person who receives the gift or inheritance has primary liability for CAT.

CAT is levied at a rate of 33% above certain tax-free thresholds. The appropriate tax-free threshold is dependent upon (1) the relationship between the donor and the donee and (2) the aggregation of the values of previous gifts and inheritances received by the donee from persons within the same group threshold. Gifts and inheritances passing between spouses are exempt from CAT. Children have a tax-free threshold of  $\notin$ 225,000 in respect of taxable gifts or inheritances received from their parents. Our shareholders should consult their own tax advisors as to whether CAT is creditable or deductible in computing any domestic tax liabilities.

# THE IRISH TAX CONSEQUENCES SUMMARIZED ABOVE ARE FOR GENERAL INFORMATION ONLY.

## Irish Restrictions on Import and Export of Capital

The Financial Transfers Act 1992 provides that the Irish Minister for Finance can make provision for the restriction of financial transfers between Ireland and other countries. For the purposes of this Act, "financial transfers" include all transfers which would be movements of capital or payments within the meaning of the treaties governing the European Communities if they had been made between Member States of the Communities. This Act has been used by the Minister for Finance to implement European Council Directives, which provide for the restriction of financial transfers to certain countries, organizations and people including the Al-Qaeda network and the Taliban, Belarus, Burma (Myanmar), Democratic People's Republic of Korea, Democratic Republic of Congo, Egypt, Eritrea, Iran, Iraq, Ivory Coast, Lebanon, Liberia, Republic of Guinea, Somalia, Sudan, Syria, Tunisia, Yugoslavia (Slobodan Milosevic and associated persons) and Zimbabwe.

-174-

## INDEBTEDNESS

## The Notes

In connection with the separation, MIFSA, a wholly owned subsidiary of Covidien that will become our wholly owned subsidiary upon completion of the distribution, has issued \$300 million aggregate principal amount of 3.50% senior unsecured notes due 2018 and \$600 million aggregate principal amount of 4.75% senior unsecured notes due 2023 (collectively, the "notes"). Mallinckrodt plc will guarantee the notes on an unsecured and unsubordinated basis upon completion of the distribution. MIFSA will pay interest on the notes semi-annually in arrears on April 15 and October 15 of each year, commencing on October 15, 2013.

The notes were issued and sold in a private placement to qualified institutional buyers pursuant to Rule 144A under the Securities Act and non-U.S. persons pursuant to Regulation S under the Securities Act.

MIFSA may redeem all of the notes at any time, and some of the notes, from time to time, at a redemption price equal to the principal amount of the notes redeemed plus a make-whole premium.

The Indenture contains covenants limiting the ability of Mallinckrodt and its restricted subsidiaries to incur certain liens and to enter into sale and leaseback transactions and the ability of MIFSA and Mallinckrodt to merge or consolidate with any other person or sell or convey all or substantially all of their respective assets to any person.

The preceding summary of the terms of the Indenture is qualified in its entirety by reference to the Indenture filed as Exhibit 4.1 to the Form 10, which is incorporated herein by reference.

The net proceeds to MIFSA from the issuance and sale of the notes were approximately \$889.3 million. It is anticipated that, upon completion of the distribution, MIFSA will retain for general corporate purposes an amount of the net proceeds of the notes offering that, together with cash held by its subsidiaries, equals approximately \$168 million, and the remainder will be retained by Covidien.

Covidien anticipates using these funds for general corporate purposes and does not intend to repay any of its own indebtedness with these funds.

### **Revolving Credit Facility**

MIFSA has entered into a 5-year revolving credit facility with a borrowing capacity up to \$250 million (the "credit facility") that we expect will be undrawn at the time the separation is completed. Mallinckrodt plc will guarantee the credit facility on an unsecured and unsubordinated basis upon completion of the distribution. Borrowings under this facility will initially bear interest at LIBOR plus 1.50 percent per annum (subject to adjustment based upon a ratings-based pricing grid). The credit facility provides for customary fees, including commitment fees and other fees.

The credit facility contains customary affirmative and negative covenants, that among other things, will limit or restrict the ability of non-guarantor subsidiaries to incur indebtedness, our ability to incur liens, our ability to consolidate, merge or sell all or substantially all of our assets or the assets of the Specialty Pharmaceuticals segment or the Global Medical Imaging segment, our ability to pay dividends or make other distributions on or repurchase or redeem our capital stock, our ability to enter into transactions with affiliates, our ability to engage in sale and leaseback transactions, and our ability to enter into agreements restricting the ability of our subsidiaries to pay dividends. The credit facility also contains financial maintenance covenants, including a leverage ratio covenant and interest coverage ratio covenant.

-175-

Our ability to borrow under the credit facility is subject to satisfaction of the following material conditions:

- the completion of the distribution in a manner substantially consistent with the description provided to the lender prior to the date hereof; and
- other customary closing deliverables.

The preceding summary of the terms of the credit facility is qualified in its entirety by reference to the credit facility filed as Exhibit 10.4 to the Form 10, which is incorporated herein by reference.

## DESCRIPTION OF MALLINCKRODT'S SHARE CAPITAL

Mallinckrodt's memorandum and articles of association will be amended and restated in connection with the separation. The following is a summary of the material terms of Mallinckrodt's share capital that will be contained in the amended and restated memorandum and articles of association. The summaries and descriptions below do not purport to be complete statements of the relevant provisions of the memorandum and articles of association to be in effect at the time of the distribution. The summary is qualified in its entirety by reference to these documents, which you must read (along with the applicable provisions of Irish law) for complete information on Mallinckrodt's share capital as of the time of the distribution. The memorandum and articles of association to be in effect at the time of the distribution will be included as an exhibit to Mallinckrodt's registration statement on Form 10, of which this information statement forms a part.

#### Legal Name; Formation; Fiscal Year; Registered Office

The legal name of the newly formed Irish company is Mallinckrodt public limited company. Mallinckrodt was incorporated in Ireland as a public limited company on January 9, 2013 with company registration number 522227. Mallinckrodt's fiscal year ends on the last Friday in September and Mallinckrodt's registered address is 1st Floor, 20 On Hatch, Lower Hatch Street, Dublin 2, Ireland.

#### **Share Capital**

The authorized share capital of Mallinckrodt will be  $\notin$ 40,000 and \$200,000,000, divided into 40,000 ordinary A shares with a par value of  $\notin$ 1.00 per share, 500,000,000 ordinary shares with a par value of \$0.20 per share and 500,000,000 preferred shares with a par value of \$0.20 per share. The authorized share capital includes 40,000 ordinary A shares with a par value of  $\notin$ 1.00 per share in order to satisfy minimum statutory requirements for the granting of a trading certificate to an Irish public limited company. These ordinary A shares carry no voting or dividend rights. All current outstanding ordinary A shares, together with the seven ordinary shares held by the current nominee shareholders of Mallinckrodt, will be acquired and canceled by Mallinckrodt for no consideration contemporaneously with the distribution being effected.

Mallinckrodt may issue shares subject to the maximum prescribed by its authorized share capital contained in its memorandum of association. Upon completion of the distribution, based on approximately 458 million Covidien ordinary shares outstanding as of May 24, 2013, we expect that Mallinckrodt will have issued approximately \$11 million of its authorized share capital of \$200,000,000, with such issued share capital comprised of approximately 57 million ordinary shares with a par value of \$0.20 each. This means that Mallinckrodt would be able to issue approximately 443 million additional ordinary shares with a total nominal value of approximately \$89 million, 500,000,000 preferred shares with a nominal value of \$0.20 each (as well as 40,000 ordinary A shares with a par value of €1.00 per share).

As a matter of Irish company law, the directors of a company may cause the company to issue new ordinary or preferred shares without shareholder approval once authorized to do so by the articles of association of the company or by an ordinary resolution adopted by the shareholders at a general meeting. An ordinary resolution requires over 50% of the votes of a company's shareholders cast at a general meeting (in person or by proxy). The authority conferred can be granted for a maximum period of five years, at which point it must be renewed by the shareholders of the company by an ordinary resolution. The articles of association of Mallinckrodt will authorize the board of directors of Mallinckrodt to issue new ordinary or preferred shares without shareholder approval for a period of five years from the date of adoption of the amended and restated articles of association.

The authorized share capital may be increased or reduced by way of an ordinary resolution of Mallinckrodt's shareholders, but not below the number of shares then outstanding. The shares comprising the authorized share capital of Mallinckrodt may be divided into shares of such par value as the resolution prescribes.

The rights and restrictions to which the ordinary shares will be subject will be prescribed in Mallinckrodt's articles of association. Mallinckrodt's articles of association will entitle the board of directors, without shareholder

-177-

approval, to determine the terms of the preferred shares issued by Mallinckrodt. Preferred shares may be preferred as to dividends, rights on a winding up or voting in such manner as the directors of Mallinckrodt may resolve. The preferred shares may also be redeemable at the option of the holder of the preferred shares or at the option of Mallinckrodt, and may be convertible into or exchangeable for shares of any other class or classes of Mallinckrodt, depending on the terms of such preferred shares. The issuance of preferred shares is subject to applicable law, including the Irish Takeover Rules.

Irish law does not recognize fractional shares held of record; accordingly, Mallinckrodt's articles of association do not provide for the issuance of fractional ordinary shares of Mallinckrodt, and the official Irish register of Mallinckrodt will not reflect any fractional ordinary shares.

#### Pre-emption Rights, Share Warrants and Share Options

Certain statutory pre-emption rights apply automatically in favor of Mallinckrodt's shareholders where shares in Mallinckrodt are to be issued for cash. However, Mallinckrodt has opted out of these pre-emption rights in its articles of association as permitted under Irish company law. Irish law provides that this opt-out expires after five years unless renewed by a special resolution of the shareholders. A special resolution requires not less than 75% of the votes of Mallinckrodt's shareholders cast at a general meeting (in person or by proxy). If the opt-out is not renewed, shares issued for cash must be offered to pre-existing shareholders of Mallinckrodt pro-rata to their existing shareholding before the shares can be issued to any new shareholders. The statutory pre-emption rights do not apply where shares are issued for non-cash consideration.

The articles of association of Mallinckrodt provide that, subject to any shareholder approval requirement under any laws, regulations or the rules of any stock exchange to which Mallinckrodt is subject, the board is authorized, from time to time, in its discretion, to grant such persons, for such periods and upon such terms as the board deems advisable, options to purchase such number of shares of any class or classes or of any series of any class as the board may deem advisable, and to cause warrants or other appropriate instruments evidencing such options to be issued. The Irish Companies Acts provide that directors may issue share warrants or options without shareholder approval once authorized to do so by the articles of association or an ordinary resolution of shareholders. Under Irish law, the board may issue shares upon exercise of validly issued warrants or options without shareholder approval of certain equity compensation plans.

#### Dividends

Under Irish law, dividends and distributions may only be made from "distributable reserves." Distributable reserves, broadly, means the accumulated realized profits of Mallinckrodt less accumulated realized losses of Mallinckrodt. In addition, no distribution or dividend may be made unless the net assets of Mallinckrodt are equal to, or in excess of, the aggregate of Mallinckrodt's share capital which has been paid up or which is payable in the future plus undistributable reserves and the distribution does not reduce Mallinckrodt's net assets below such aggregate. Undistributable reserves include the share premium account, the capital redemption reserve fund and Mallinckrodt's net unrealized profits.

The determination as to whether or not Mallinckrodt has sufficient distributable reserves to fund a dividend must be made by reference to the "relevant accounts" of Mallinckrodt. The "relevant accounts" will be either the last set of unconsolidated annual audited financial statements or unaudited financial statements prepared in accordance with the Irish Companies Acts, which give a "true and fair view" of Mallinckrodt's unconsolidated financial position and accord with accepted accounting practice. The relevant accounts must be filed in the Companies Registration Office (the official public registry for companies in Ireland).

Although Mallinckrodt will not have any distributable reserves immediately following the distribution, we are taking steps to create such distributable reserves. See "Risk Factors" and "Dividends—Creation of Distributable Reserves."

-178-

The mechanism as to who declares a dividend and when a dividend becomes payable is governed by the articles of association of Mallinckrodt. Mallinckrodt's articles of association authorize the directors to declare such dividends as appear justified from the profits of Mallinckrodt without the approval of the shareholders at a general meeting. The board of directors may also recommend a dividend to be approved and declared by the shareholders at a general meeting. Although the shareholders may direct that the payment be made by distribution of assets, shares or cash, no dividend issued may exceed the amount recommended by the directors. The dividends can be declared and paid in the form of assets, shares or cash.

The directors of Mallinckrodt may deduct from any dividend payable to any shareholder all sums of money (if any) payable by such shareholder to Mallinckrodt in relation to the ordinary shares of Mallinckrodt.

The directors of Mallinckrodt are also entitled to issue shares with preferred rights to participate in dividends declared by Mallinckrodt. The holders of such preferred shares may, depending on their terms, be entitled to claim arrears of a declared dividend out of subsequently declared dividends in priority to ordinary shareholders.

For information about the Irish tax issues relating to dividend payments, see "Material Tax Consequences-Material Irish Tax Consequences."

#### **Share Repurchases and Redemptions**

## Overview

Article 3(d) of Mallinckrodt's articles of association provides that any ordinary share which Mallinckrodt has acquired or agreed to acquire is deemed to be a redeemable share. Accordingly, for Irish company law purposes, the repurchase of ordinary shares by Mallinckrodt will technically be effected as a redemption of those shares as described below under "—Share Repurchases and Redemptions—Repurchases and Redemptions by Mallinckrodt." If the articles of association of Mallinckrodt did not contain Article 3(d), repurchases by Mallinckrodt would be subject to many of the same rules that apply to purchases of Mallinckrodt ordinary shares by subsidiaries described below under "—Share Repurchases and Redemptions—Purchases by Subsidiaries of Mallinckrodt," including the shareholder approval requirements described below and the requirement that any on-market purchases be effected on a "recognized stock exchange." Except where otherwise noted, when we refer elsewhere in this information statement to repurchasing or buying back ordinary shares of Mallinckrodt, we are referring to the redemption of ordinary shares by Mallinckrodt pursuant to Article 3(d) of the articles of association or the purchase of ordinary shares of Mallinckrodt by a subsidiary of Mallinckrodt, in each case in accordance with the Mallinckrodt articles of association and Irish company law as described below.

#### **Repurchases and Redemptions by Mallinckrodt**

Under Irish law, a company can issue redeemable shares and redeem them out of distributable reserves (which are described above under "—Dividends") or the proceeds of a new issue of shares for that purpose. Although Mallinckrodt will not have any distributable reserves immediately following the distribution, we are taking steps to create such distributable reserves. See "Risk Factors" and "Dividends—Creation of Distributable Reserves." The issue of redeemable shares may only be made by Mallinckrodt where the nominal value of the issued share capital that is not redeemable is not less than 10% of the nominal value of the total issued share capital of Mallinckrodt. All redeemable shares must also be fully paid and the terms of redemption of the shares must provide for payment on redemption. Redeemable shares may, upon redemption, be cancelled or held in treasury. Shareholder approval will not be required to redeem Mallinckrodt ordinary shares pursuant to Article 3(d) of Mallinckrodt's articles of association.

The board of directors of Mallinckrodt will also be entitled to issue preferred shares which may be redeemed at the option of either Mallinckrodt or the shareholder, depending on the terms of such preferred shares. For additional information on redeemable shares, see "—Share Capital."

Repurchased and redeemed shares may be cancelled or held as treasury shares. The nominal value of treasury shares held by Mallinckrodt at any time must not exceed 10% of the nominal value of the issued share capital of Mallinckrodt. While Mallinckrodt holds shares as treasury shares, it cannot exercise any voting rights in respect of those shares. Treasury shares may be cancelled by Mallinckrodt or re-issued subject to certain conditions.

### Purchases by Subsidiaries of Mallinckrodt

Under Irish law, it may be permissible for an Irish or non-Irish subsidiary to purchase ordinary shares of Mallinckrodt either on-market or off-market. A general authority of the shareholders of Mallinckrodt is required to allow a subsidiary of Mallinckrodt to make on-market purchases of Mallinckrodt ordinary shares; however, as long as this general authority has been granted, no specific shareholder authority for a particular on-market purchase by a subsidiary of Mallinckrodt ordinary shares is required. We expect that Mallinckrodt will seek such general authority, which must expire no later than 18 months after the date on which it was granted, at the first annual general meeting of Mallinckrodt in 2014 and at subsequent annual general meetings. In order for a subsidiary of Mallinckrodt to make an on-market purchase of Mallinckrodt's ordinary shares, such shares must be purchased on a "recognized stock exchange." The NYSE, on which the ordinary shares of Mallinckrodt will be listed following the distribution, is specified as a recognized stock exchange for this purpose by Irish company law. For an off-market purchase by a subsidiary of Mallinckrodt, the proposed purchase contract must be authorized by special resolution of the shareholders of Mallinckrodt before the contract is entered into. The person whose shares are to be bought back cannot vote in favor of the special resolution and, for at least 21 days prior to the special resolution, the purchase contract must be on display or must be available for inspection by shareholders at the registered office of Mallinckrodt.

The number of shares held by the subsidiaries of Mallinckrodt at any time will count as treasury shares and will be included in any calculation of the permitted treasury share threshold of 10% of the nominal value of the issued share capital of Mallinckrodt. While a subsidiary holds ordinary shares of Mallinckrodt, it cannot exercise any voting rights in respect of those shares. The acquisition of the ordinary shares of Mallinckrodt by a subsidiary must be funded out of distributable reserves of the subsidiary.

#### **Bonus Shares**

Under Mallinckrodt's articles of association, the board may resolve to capitalize any amount credited to any reserve or fund available for distribution or the share premium account or any other undistributable reserve of Mallinckrodt through the issuance of fully paid-up bonus shares to shareholders on the same basis of entitlement as would apply in respect of a dividend distribution.

#### Consolidation and Division; Subdivision

Under its articles of association, Mallinckrodt may, by ordinary resolution, consolidate and divide all or any of its share capital into shares of larger par value than its existing shares or subdivide its shares into smaller amounts than is fixed by its articles of association.

## **Reduction of Share Capital**

Mallinckrodt may, by special resolution, reduce its authorized share capital in any way. Mallinckrodt also may, by special resolution and subject to confirmation by the High Court of Ireland, reduce or cancel its issued share capital (which includes share premium) in any way. The creation of distributable reserves discussed in "Dividends—Creation of Distributable Reserves" involves a reduction of share capital, namely the share premium account of Mallinckrodt, for purposes of Irish law.

-180-

#### **General Meetings of Shareholders**

Mallinckrodt will be required to hold an annual general meeting within 18 months of incorporation and at intervals of no more than 15 months thereafter, provided that an annual general meeting is held in each calendar year following the first annual general meeting, no more than nine months after Mallinckrodt's fiscal year end. The first annual general meeting of Mallinckrodt may be held outside Ireland. Thereafter, any annual general meeting may be held outside Ireland if a resolution so authorizing has been passed at the preceding annual general meeting. Because of the 15-month requirement described in this paragraph, Mallinckrodt's articles of association include a provision reflecting this requirement of Irish law.

Extraordinary general meetings of Mallinckrodt may be convened by (i) the board of directors, (ii) on requisition of the shareholders holding not less than 10% of the paid-up share capital of Mallinckrodt carrying voting rights or (iii) on requisition of Mallinckrodt's auditors upon their resignation. Extraordinary general meetings are generally held for the purposes of approving shareholder resolutions of Mallinckrodt as may be required from time to time.

Notice of a general meeting must be given to all shareholders of Mallinckrodt and to the auditors of Mallinckrodt. The minimum notice periods are 21 days' notice in writing for an annual general meeting or an extraordinary general meeting to approve a special resolution and 14 days' notice in writing for any other extraordinary general meetings. General meetings may be called by shorter notice, but only with the consent of the auditors of Mallinckrodt and all of the shareholders entitled to attend and vote thereat. Because of the 21-day and 14-day requirements described in this paragraph, Mallinckrodt's articles of association include provisions reflecting these requirements of Irish law.

In the case of an extraordinary general meeting convened by shareholders of Mallinckrodt, the proposed purpose of the meeting must be set out in the requisition notice. The requisition notice can contain any resolution. Upon receipt of this requisition notice, the board of directors has 21 days to convene a meeting of Mallinckrodt's shareholders to vote on the matters set out in the requisition notice. This meeting must be held within two months of the receipt of the requisition notice. If the board of directors does not convene the meeting within such 21-day period, the requisitioning shareholders, or any of them representing more than one half of the total voting rights of all of them, may themselves convene a meeting, which meeting must be held within three months of the receipt of the requisition notice.

The only matters which must, as a matter of Irish company law, be transacted at an annual general meeting are the presentation of the annual accounts, balance sheet and reports of the directors and auditors, the appointment of auditors and the fixing of the auditor's remuneration (or delegation of same). If no resolution is made in respect of the reappointment of an auditor at an annual general meeting, the previous auditor will be deemed to have continued in office.

If the directors become aware that the net assets of Mallinckrodt are half or less of the amount of Mallinckrodt's called-up share capital, the directors of Mallinckrodt must convene an extraordinary general meeting of Mallinckrodt's shareholders not later than 28 days from the date that they learn of this fact. This meeting must be convened for the purposes of considering whether any, and if so what, measures should be taken to address the situation.

#### Voting

Where a vote is to be taken at a general meeting, every shareholder has one vote for each ordinary share that he or she holds as of the record date for the meeting. Voting rights may be exercised by shareholders registered in Mallinckrodt's share register as of the record date for the meeting or by a duly appointed proxy of such a registered shareholder, which proxy need not be a shareholder. Where interests in shares are held by a nominee trust company, this company may exercise the rights of the beneficial holders on their behalf as their proxy. All proxies must be appointed in the manner prescribed by Mallinckrodt's articles of association. The articles of association of Mallinckrodt permit the appointment of proxies by the shareholders to be notified to Mallinckrodt electronically.

Mallinckrodt's articles provide that all resolutions are decided by a show of hands unless a vote is demanded by the Chairman, by at least three shareholders as of the record date for the meeting or by any shareholder or shareholders holding not less than 10% of the total voting rights of Mallinckrodt as of the record date for the meeting. Each Mallinckrodt ordinary shareholder of record as of the record date for the meeting has one vote at a general meeting on a show of hands. Treasury shares and shares held by subsidiaries will not be entitled to vote at general meetings of shareholders.

Irish company law requires "special resolutions" of the shareholders at a general meeting to approve certain matters. A special resolution requires not less than 75% of the votes cast of Mallinckrodt's shareholders present in person or by proxy at a general meeting. This may be contrasted with "ordinary resolutions," which require a simple majority of the votes of Mallinckrodt's shareholders cast in person or by proxy at a general meeting. Examples of matters requiring special resolutions include:

- amending the objects (*i.e.*, main purposes) of Mallinckrodt;
- amending the articles of association of Mallinckrodt;
- approving a change of name of Mallinckrodt;
- authorizing the entering into of a guarantee or provision of security in connection with a loan, quasi-loan or credit transaction to a director or a person who is deemed to be "connected" to a director for the purposes of the Irish Companies Acts;
- opting-out of pre-emption rights on the issuance of new shares;
- re-registration of Mallinckrodt from a public limited company to a private company;
- variation of class rights attaching to classes of shares;
- purchasing Mallinckrodt's ordinary shares off-market;
- any reduction of Mallinckrodt's issued share capital;
- resolving that Mallinckrodt be wound up by the Irish courts;
- resolving in favor of a shareholders' voluntary winding-up;
- re-designation of shares into different share classes; and
- setting the re-issue price of treasury shares.

## Variation of Class Rights Attaching to Shares

Variation of all or any special rights attached to any class of shares of Mallinckrodt is addressed in the articles of association of Mallinckrodt as well as the Irish Companies Acts. Any variation of class rights attaching to the issued shares of Mallinckrodt must be approved by a special resolution of the shareholders of the class affected. Mallinckrodt's articles of association expressly provide that any issue of preferred shares (whatever the rights attaching to them) will be deemed not to be a variation of the rights of ordinary shareholders.

### **Quorum for General Meetings**

The presence, in person or by proxy, of the holders of shares in Mallinckrodt entitling them to exercise a majority of the voting power of Mallinckrodt constitutes a quorum for the conduct of business. No business may take place at a general meeting of Mallinckrodt if a quorum is not present in person or by proxy. The board of directors has no authority to waive quorum requirements stipulated in the articles of association of Mallinckrodt. Abstentions and broker non-votes will be counted as present for purposes of determining whether there is a quorum in respect of the proposals.

## Requirements for Advance Notification of Director Nominations and Proposals of Shareholders

Our articles of association provide that with respect to an annual or extraordinary general meeting of shareholders, nominations of persons for election to our board of directors and the proposal of business to be

-182-

considered by shareholders may be made only pursuant to Mallinckrodt's notice of meeting; by the board of directors; by any shareholders pursuant to the valid exercise of power granted to them under the Irish Companies Acts; or by a shareholder who is entitled to vote at the meeting and who has complied with the advance notice procedures provided for in our articles of association.

In order to comply with the advance notice procedures of our articles of association, a shareholder must give written notice to Mallinckrodt's Secretary on a timely basis. To be timely for an annual general meeting, notice must be delivered not earlier than the close of business on the 120th day and not later than the close of business on the 90th day prior to the first anniversary of the preceding year's annual general meeting, subject to certain exceptions. To be timely for an extraordinary general meeting, notice must be delivered not earlier than the close of business on the 120th day prior to the date of such extraordinary general meeting and not later than the close of business on the 90th day prior to the date of such extraordinary general meeting or, if the first public announcement of the date of such extraordinary general meeting, notice the date of the date of such extraordinary general meeting, the 10th day following the day on which public announcement is first made of the date of the extraordinary general meeting, notice must be so delivered not later than the 10th day following the day on which public announcement of the date of such extraordinary general meeting, notice must be so delivered not later than the 10th day following the day on which public announcement of the date of such extraordinary general meeting. With respect to the 2014 annual general meeting, notice must be so delivered not later than the 10th day following the day on which public announcement of the date of such meeting is first made by Mallinckrodt.

In addition, whether relating to an annual or extraordinary general meeting, to be timely, a shareholder's notice must be updated and supplemented, if necessary, so the information provided or required to be provided is true and correct as of the record date for the meeting and as of the date that is ten business days prior to the meeting or any adjournment or postponement thereof. For nominations to the board, the notice must include all information about the director nominee that is required to be disclosed by SEC rules regarding the solicitation of proxies for the election of directors pursuant to Regulation 14A under the Exchange Act, a description of all direct and indirect compensation and other material monetary agreements during the past three years, any other material relationships with the proposed nominee and his or her affiliates and associates and such other information as Mallinckrodt may reasonably require to determine the eligibility of the proposed nominee, as well as a completed questionnaire, representation and agreement signed by the proposed nominee regarding the background, qualification and certain existing relationships of the proposed nominee. For other business that a shareholder proposes to bring before the meeting, the notice must include a brief description of the business, the reasons for proposing the business at the meeting, a discussion of any material interest of the shareholder in the business and a description of all arrangements with any other person or persons in connection with the proposal. Whether the notice relates to a nomination to the board of directors or to other business to be proposed at the meeting, the notice also must include information about the shareholder, the shareholder's holdings of Mallinckrodt shares (as well as "derivative instruments," "short interests" with respect to Mallinckrodt shares, as defined in our articles of association), any arrangements giving the shareholder the right to vote shares of Mallinckrodt, any rights to dividends on the Mallinckrodt shares that are separated or separable from the underlying Mallinckrodt shares, any performance-related fees (other than an asset-based fee) that the shareholder is entitled to based on any increase or decrease in the value of the Mallinckrodt shares or "derivative instruments," any significant equity interests or any derivative instruments in any of Mallinckrodt's principal competitors held by the shareholder and any interest of the shareholder in any contract with Mallinckrodt or any of its affiliates or principal competitors.

In addition, the Irish Companies Acts provide that shareholders holding not less than 10% of the total voting rights may call an extraordinary general meeting for the purpose of considering director nominations or other proposals, as described above under "—General Meetings of Shareholders."

#### **Unanimous Shareholder Consent to Action Without Meeting**

The Irish Companies Acts provide that shareholders may approve an ordinary or special resolution of shareholders without a meeting only if (a) *all* shareholders sign the written resolution and (b) the company's

articles of association permit written resolutions of shareholders (Mallinckrodt's articles of association contain the appropriate authorizations for this purpose). Mallinckrodt's articles of association permit unanimous written resolutions of shareholders.

#### **Inspection of Books and Records**

Under Irish law, shareholders have the right to: (1) receive a copy of the memorandum and articles of association of Mallinckrodt and any act of the Irish Government which alters the memorandum of association of Mallinckrodt; (2) inspect and obtain copies of the minutes and resolutions of general meetings of Mallinckrodt; (3) inspect and receive a copy of the register of shareholders, register of directors and secretaries, register of directors' interests and other statutory registers maintained by Mallinckrodt; (4) receive copies of balance sheets and directors' and auditors' reports which have previously been sent to shareholders prior to an annual general meeting; and (5) receive balance sheets of a subsidiary company of Mallinckrodt which have previously been sent to shareholders prior to an annual general meeting for the preceding 10 years. The auditors of Mallinckrodt will also have the right to inspect all books, records and vouchers of Mallinckrodt. The auditors' report must be circulated to the shareholders 21 days before the annual general meeting with Mallinckrodt's financial statements prepared in accordance with the Irish Companies Acts, and must be read to the shareholders at Mallinckrodt's annual general meeting.

#### Acquisitions and Appraisal Rights

There are a number of mechanisms for acquiring an Irish public limited company, including:

- (a) a court-approved scheme of arrangement under the Irish Companies Acts. A scheme of arrangement with shareholders requires a court order from the High Court of Ireland and the approval of: (1) 75% of the voting shareholders by value; and (2) 50% in number of the voting shareholders, at a meeting called to approve the scheme;
- (b) through a tender offer by a third party for all of the shares of Mallinckrodt. Where the holders of 80% or more of Mallinckrodt's shares have accepted an offer by a bidder for their shares in Mallinckrodt, the remaining shareholders may be statutorily required to also transfer their shares to such bidder. If the bidder does not exercise its "squeeze out" right, then the non-accepting shareholders also have a statutory right to require the bidder to acquire their shares on the same terms. If shares of Mallinckrodt were listed on the official list of the Irish Stock Exchange or another regulated stock exchange in the E.U., this threshold would be increased to 90%; and
- (c) it is also possible for Mallinckrodt to be acquired by way of a merger with an E.U.-incorporated public company under the E.U. Cross Border Merger Directive 2005/56. Such a merger must be approved by a special resolution. If Mallinckrodt is being merged with another E.U. public company under the E.U. Cross Border Merger Directive 2005/56 and the consideration payable to Mallinckrodt's shareholders is not all in the form of cash, Mallinckrodt's shareholders may be entitled to require their shares to be acquired at fair value.

Under Irish law, there is no requirement for a company's shareholders to approve a sale, lease or exchange of all or substantially all of a company's property and assets. However, Mallinckrodt's articles of association provide that the affirmative vote of the holders of a majority of the outstanding voting shares on the relevant record date is required to approve a sale, lease or exchange of all or substantially all of its property or assets.

## **Disclosure of Interests in Shares**

Under the Irish Companies Acts, subject to certain limited exceptions, a shareholder of Mallinckrodt must notify Mallinckrodt (but not the public at large) if as a result of a transaction the shareholder will be interested in 5% or more of any class of shares of Mallinckrodt carrying voting rights; or if as a result of a transaction a shareholder who was interested in more than 5% of any class of shares of Mallinckrodt carrying voting rights ceases to be so interested. Where a shareholder is interested in more than 5% of shares of Mallinckrodt carrying voting rights ceases to be so interested. Where a

-184-

rights, any alteration of his or her interest that brings his or her total holding through the nearest whole percentage number, whether an increase or a reduction, must be notified to Mallinckrodt (but not the public at large). The relevant percentage figure is calculated by reference to the aggregate par value of the class of shares in which the shareholder is interested as a proportion of the entire par value of the issued shares of that class. Where the percentage level of the shareholder's interest does not amount to a whole percentage, this figure may be rounded down to the next whole number. All such disclosures must be notified to Mallinckrodt within five business days of the transaction or alteration of the shareholder's interests that gave rise to the requirement to notify. Where a person fails to comply with the notification requirements described above, no right or interest of any kind whatsoever in respect of any shares in Mallinckrodt concerned, held by such person, will be enforceable by such person, whether directly or indirectly, by action or legal proceeding. However, such person may apply to the court to have the rights attaching to the shares concerned reinstated.

In addition to the above disclosure requirement, Mallinckrodt, under the Irish Companies Acts, may by notice in writing require a person whom Mallinckrodt knows or has reasonable cause to believe to be or, at any time during the three years immediately preceding the date on which such notice is issued, to have been interested in shares comprised in Mallinckrodt's relevant share capital: (a) to indicate whether or not it is the case, and (b) where such person holds or has during that time held an interest in any class of shares of Mallinckrodt carrying voting rights to give such further information as may be required by Mallinckrodt, including particulars of such person's own past or present interests in such class of shares of Mallinckrodt. Any information given in response to the notice is required to be given in writing within such reasonable time as may be specified in the notice.

Where such a notice is served by Mallinckrodt on a person who is or was interested in any class of shares of Mallinckrodt carrying voting rights and that person fails to give Mallinckrodt any information required within the reasonable time specified, Mallinckrodt may apply to the court for an order directing that the affected shares be subject to certain restrictions.

Under the Irish Companies Acts, the restrictions that may be placed on the shares by the court are:

- (a) any transfer of those shares, or in the case of unissued shares any transfer of the right to be issued with shares and any issue of shares, is void;
- (b) no voting rights are exercisable in respect of those shares;
- (c) no further shares may be issued in right of those shares or in pursuance of any offer made to the holder of those shares; and
- (d) no payment may be made of any sums due from Mallinckrodt on those shares, whether in respect of capital or otherwise.

Where the shares in Mallinckrodt are subject to these restrictions, the court may order the shares to be sold and may also direct that the shares will cease to be subject to these restrictions.

## **Anti-Takeover Provisions**

#### **Business Combinations with Interested Shareholders**

Mallinckrodt's articles of association include a provision similar to Section 203 of the Delaware General Corporation Law, which generally prohibits Mallinckrodt from engaging in a business combination with an interested shareholder for a period of three years following the date the person became an interested shareholder, unless, in general:

- Mallinekrodt's board of directors approved the transaction which resulted in the shareholder becoming an interested shareholder;
- upon consummation of the transaction which resulted in the shareholder becoming an interested shareholder, the shareholder owned at least 85% of the voting shares outstanding at the time of

commencement of such transaction, excluding for purposes of determining the number of voting shares outstanding (but not the outstanding voting shares owned by the interested shareholder), voting shares owned by persons who are directors and also officers and by certain employee share plans; or

the business combination is approved by Mallinckrodt's board of directors and authorized at an annual or extraordinary general meeting of shareholders by the affirmative vote of the holders of at least 66 2/3% of the outstanding voting shares that are not owned by the interested shareholder.

A "business combination" is generally defined as a merger, scheme of arrangement, asset or share sale or other transaction resulting in a financial benefit to the interested shareholder. An "interested shareholder" is generally defined as a person who, together with affiliates and associates, owns or, within three years prior to the date in question, owned 15% or more of the outstanding voting shares of Mallinckrodt.

#### Shareholder Rights Plans and Share Issuances

Irish law does not expressly prohibit companies from issuing share purchase rights or adopting a shareholder rights plan (commonly known as a "poison pill") as an anti-takeover measure. However, there is no directly relevant case law on the validity of such plans under Irish law. In addition, such a plan would be subject to the Irish Takeover Rules described below.

Mallinckrodt's articles of association allow the board to adopt a shareholder rights plan upon such terms and conditions as the board deems expedient and in the best interests of Mallinckrodt, subject to applicable law.

Subject to the Irish Takeover Rules described below, the board also has power to cause Mallinckrodt to issue any of its authorized and unissued shares on such terms and conditions as the board may determine (as described under "—Share Capital") and any such action must be taken in the best interests of Mallinckrodt. It is possible, however, that the terms and conditions of any issue of preferred shares could discourage a takeover or other transaction that holders of some or a majority of the ordinary shares believe to be in their best interests or in which holders might receive a premium for their shares over the then market price of the shares.

#### Irish Takeover Rules

A transaction by virtue of which a third party is seeking to acquire 30% or more of the voting rights of Mallinckrodt will be governed by the Irish Takeover Panel Act 1997 and the Irish Takeover Rules made thereunder and will be regulated by the Irish Takeover Panel. The "General Principles" of the Irish Takeover Rules and certain important aspects of the Irish Takeover Rules are described below.

General Principles. The Irish Takeover Rules are built on the following General Principles which will apply to any transaction regulated by the Irish Takeover Panel:

- in the event of an offer, all classes of shareholders of the target company should be afforded equivalent treatment and, if a person acquires control of a company, the other holders of securities must be protected;
- the holders of securities in the target company must have sufficient time and information to allow them to make an informed decision regarding the offer;
- the board of a company must act in the interests of the company as a whole. If the board of the target company advises the holders of securities as regards the offer, it must advise on the effects of the implementation of the offer on employment, employment conditions and the locations of the target company's place of business;
- false markets (*i.e.*, a market based on erroneous, imperfect or unequally disclosed information) in the securities of the target company or any other company concerned by the offer must not be created;
- a bidder can only announce an offer after ensuring that he or she can pay in full the consideration offered;

-186-

- a target company may not be hindered longer than is reasonable by an offer for its securities. This is a recognition that an offer will disrupt the dayto-day running of a target company particularly if the offer is hostile and the board of the target company must divert its attention to resist the offer; and
- acquisitions of securities (whether such acquisition is to be effected by one transaction or a series of transactions) will only be allowed to take place
  at an acceptable speed and subject to adequate and timely disclosure. Specifically, the acquisition of 10% or more of the issued voting shares within a
  seven day period that would take a shareholders' holding to or above 15% of the issued voting shares (but less than 30%) is prohibited, subject to
  certain exemptions.

*Mandatory Bid.* If an acquisition of shares or other securities were to increase the aggregate holding/entitlement of an acquirer and its concert parties to 30% or more of the voting rights in Mallinckrodt, the acquirer and, depending on the circumstances, its concert parties would be required (except with the consent of the Irish Takeover Panel) to make a cash offer for the outstanding shares at a price not less than the highest price paid for the shares by the acquirer or its concert parties during the previous 12 months. This requirement would also be triggered by an acquisition of shares or other securities by a person holding (together with its concert parties) shares or other securities carrying between 30% and 50% of the voting rights in Mallinckrodt if the effect of such acquisition were to increase the percentage of the voting rights held by that person (together with its concert parties) by 0.05% within a twelve-month period. A single holder (that is, a holder excluding any parties acting in concert with the holder) holding or entitled to more than 50% of the voting rights of a company is not subject to this rule.

Voluntary Bid; Requirements to Make a Cash Offer and Minimum Price Requirements. A voluntary offer is an offer that is not a mandatory offer. If a bidder or any of its concert parties has acquired ordinary shares of Mallinckrodt within the period of three months prior to the commencement of the voluntary offer, the offer price must be not less than the highest price paid for Mallinckrodt ordinary shares by the bidder or its concert parties during that period. The Irish Takeover Panel has the power to extend the "look back" period to 12 months if the Irish Takeover Panel, having regard to the General Principles, believes it is appropriate to do so.

If the bidder or any of its concert parties has acquired more than 10% of the ordinary shares of Mallinckrodt (i) during the period 12 months prior to the commencement of the voluntary offer period or (ii) at any time after the commencement of the voluntary offer period, the offer must be in cash (or accompanied by a full cash alternative) and the price per Mallinckrodt ordinary share must be not less than the highest price paid by the bidder or its concert parties during, in the case of (i), the period of 12 months prior to the commencement of the voluntary offer and, in the case of (ii), the offer period. The Irish Takeover Panel may apply this rule to a bidder who, together with its concert parties, has acquired less than 10% of the total ordinary shares of Mallinckrodt in the 12-month period prior to the commencement of the Irish Takeover Panel, having regard to the General Principles, considers it just and proper to do so.

A voluntary offer period will generally commence on the date of the first announcement of the offer or proposed offer.

Substantial Acquisition Rules. The Irish Takeover Rules also contain rules governing substantial acquisitions of shares that restrict the speed at which a person may increase his or her holding of voting shares and rights over voting shares to an aggregate of between 15% and 30% of the voting rights of Mallinckrodt. Except in certain circumstances, an acquisition or series of acquisitions of shares or rights over shares representing 10% or more of the voting rights is prohibited if such acquisition(s), when aggregated with shares or rights already held, would result in the acquirer holding 15% or more but less than 30% of the voting rights of shares or rights of Mallinckrodt and such acquisitions are made within a period of seven days. These rules also require accelerated disclosure of acquisitions of shares or rights over shares relating to such acquisitions.

*Frustrating Action.* Under the Irish Takeover Rules, the board of directors of Mallinckrodt is not permitted to take any action which might frustrate an offer for the shares of Mallinckrodt once the board of directors has received an approach which may lead to an offer, or has reason to believe an offer is imminent, except as noted below. Potentially

-187-

frustrating actions such as (i) the issue of shares, options or convertible securities, (ii) material disposals, (iii) entering into contracts other than in the ordinary course of business or (iv) any action, other than seeking alternative offers, which may result in frustration of an offer, are prohibited during the course of an offer or at any time during which the board has reason to believe an offer is imminent. Exceptions to this prohibition are available:

- (a) where the action is approved by the offeree at a general meeting; or
- (b) with the consent of the Irish Takeover Panel where:
  - (i) the Irish Takeover Panel is satisfied the action would not constitute a frustrating action;
  - (ii) the holders of 50% of the voting rights state in writing that they approve the proposed action and would vote in favor of it at a general meeting;
  - (iii) such action is in accordance with a contract entered into prior to the announcement of the offer; or
  - (iv) the decision to take such action was made before the announcement of the offer and either has been at least partially implemented or is in the ordinary course of business.

For other provisions that could be considered to have an anti-takeover effect, see above at "—Pre-emption Rights, Share Warrants and Share Options," "— Disclosure of Interests in Shares," "—Requirements for Advance Notification of Director Nominations and Proposals of Shareholders" and "—Unanimous Shareholder Consent to Action Without Meeting," in addition to "—Election of Directors," "—Vacancies on Board of Directors" and "—Amendment of Governing Documents" below.

#### **Corporate Governance**

The articles of association of Mallinckrodt delegate the day-to-day management of Mallinckrodt to its board of directors. The board of directors may then delegate management of Mallinckrodt to committees, executives or to a management team, but regardless, the directors will remain responsible, as a matter of Irish law, for the proper management of the affairs of Mallinckrodt.

## **Election of Directors**

The Irish Companies Acts provide for a minimum of two directors. Mallinckrodt's articles of association provides for a minimum of two directors and a maximum of 15 directors. The shareholders of Mallinckrodt may from time to time increase or reduce the maximum number, or increase the minimum number, of directors by a special resolution amending the articles of association.

Directors are elected by the affirmative vote of a majority of the votes cast by shareholders at an annual general meeting (present in person or by proxy) and serve for one-year terms. Any nominee for director who does not receive a majority of the votes cast is not elected to the board. However, because Irish law requires a minimum of two directors at all times, in the event that an election results in no directors being elected, each of the two nominees receiving the greatest number of votes in favor of his or her election shall hold office until his or her successor is elected. In the event that an election results in only one director being elected, that director will be elected and serve for a one-year term, and the nominee receiving the greatest number of votes in favor of his or her election will hold office until his or her successor is elected.

## Vacancies on the Board of Directors

Mallinckrodt's articles of association provide that the directors have the authority to appoint one or more directors to Mallinckrodt's board, subject to the maximum number of directors allowed for in the articles of association. A vacancy caused by the removal of a director may be filled at the meeting at which the director is removed by resolution of Mallinckrodt's shareholders. If not, it may be filled by the board of directors.

Any director so appointed will hold office until the next annual general meeting of Mallinckrodt. During any vacancy on the board, the remaining directors will have full power to act as the board.

-188-

#### **Removal of Directors**

The Irish Companies Acts provide that notwithstanding anything contained in the articles of association of a company or in any agreement between that company and a director, the shareholders may by an ordinary resolution remove a director from office before the expiration of his or her term. Accordingly, the shareholders of Mallinckrodt may by an ordinary resolution remove a director from office before the expiration of his or her term. The power of removal is without prejudice to any claim for damages for breach of contract (*e.g.*, employment contract) which the director may have against Mallinckrodt in respect of his or her removal.

## **Amendment of Governing Documents**

Irish companies, including Mallinckrodt, may only alter their memorandum of association and articles of association with the approval of the holders of at least 75% of the company's shares present and voting in person or by proxy at a general meeting of the company.

#### Duration; Dissolution; Rights upon Liquidation

Mallinckrodt's corporate existence will have unlimited duration. Mallinckrodt may be dissolved at any time by way of either a shareholders' voluntary winding up or a creditors' voluntary winding up. In the case of a shareholders' voluntary winding up, a special resolution of the shareholders of Mallinckrodt is required (*i.e.*, 75% of the votes cast, in person or by proxy, at a general meeting of shareholders). Mallinckrodt may also be dissolved by way of court order on the application of a creditor, or by the Companies Registration Office as an enforcement measure where Mallinckrodt has failed to file certain returns.

The rights of the shareholders to a return of Mallinckrodt's assets on dissolution or winding up, following the settlement of all claims of creditors, may be prescribed in Mallinckrodt's articles of association or the terms of any preferred shares issued by the directors of Mallinckrodt from time to time. The holders of preferred shares in particular may have the right to priority in a dissolution or winding up of Mallinckrodt. If the articles of association contain no specific provisions in respect of a dissolution or winding up, then, subject to the priorities of any creditors, the assets will be distributed to shareholders in proportion to the paid-up par value of the shares held. Mallinckrodt's articles provide that the ordinary shareholders of Mallinckrodt are entitled to participate pro rata in a winding up, but their right to do so may be subject to the rights of any preferred shareholder to participate under the terms of any series or class of preferred shares.

#### **Uncertificated Shares**

Holders of ordinary shares of Mallinckrodt will not have the right to require Mallinckrodt to issue certificates for their shares. Mallinckrodt will only issue uncertificated ordinary shares.

## **Stock Exchange Listing**

Mallinckrodt intends to apply for authorization to list its ordinary shares on the New York Stock Exchange under the symbol "MNK." We do not plan to be listed on the Irish Stock Exchange at the present time.

### **No Sinking Fund**

The shares have no sinking fund provisions.

#### No Liability for Further Calls or Assessments

The shares to be issued in the distribution will be duly and validly issued and fully paid.

-189-

## **Transfer and Registration of Shares**

Mallinckrodt's official share register will be maintained by its transfer agent and the transfer agent's affiliates. Registration in this share register will be determinative of membership in Mallinckrodt. A shareholder of Mallinckrodt who holds shares beneficially will not be the holder of record of such shares. Instead, the depository (*e.g.*, Cede & Co., as nominee for DTC) or other nominee will be the holder of record of such shares. Accordingly, a transfer of shares from a person who holds such shares beneficially to a person who also holds such shares beneficially through the same depository or other nominee will not be registered in Mallinckrodt's official share register, as the depository or other nominee will remain the record holder of such shares.

A written instrument of transfer is required under Irish law in order to register on Mallinckrodt's official share register any transfer of shares (i) from a person who holds such shares directly to any other person, (ii) from a person who holds such shares beneficially to a person who holds such shares directly, or (iii) from a person who holds such shares beneficially to another person who holds such shares beneficially where the transfer involves a change in the depository or other nominee that is the record owner of the transferred shares. An instrument of transfer also is required for a shareholder who directly holds shares to transfer those shares into his or her own broker account (or vice versa). Such instruments of transfer may give rise to Irish stamp duty. A person wishing to acquire shares directly may need to purchase the shares through a broker account and then transfer such shares into his or her own name.

Mallinckrodt currently intends to pay (or cause one of our affiliates to pay) stamp duty, if any, in connection with share transfers made in the ordinary course of trading by a seller who holds shares directly to a buyer who will hold the acquired shares beneficially. In other cases Mallinckrodt may, in its absolute discretion, pay (or cause one of its affiliates to pay) any stamp duty. Mallinckrodt's articles of association as they will be in effect after the distribution provide that, in the event of any such payment, Mallinckrodt (i) may seek reimbursement from the buyer, (ii) will have a lien against the Mallinckrodt ordinary shares acquired by such buyer and any dividends paid on such shares and (iii) may set-off the amount of the stamp duty against future dividends on such shares. Parties to a share transfer may assume that any stamp duty arising in respect of a transaction in Mallinckrodt ordinary shares has been paid unless one or both of such parties is otherwise notified by Mallinckrodt.

Mallinckrodt's articles of association as they will be in effect after the separation delegate to Mallinckrodt's Secretary and certain other persons and delegates the authority to execute an instrument of transfer on behalf of a transferring party. In order to help ensure that the official share register is regularly updated to reflect trading of Mallinckrodt ordinary shares occurring through normal electronic systems, we intend to regularly produce any required instruments of transfer in connection with any transactions for which we pay stamp duty (subject to the reimbursement and set-off rights described above). In the event that we notify one or both of the parties to a share transfer that we believe stamp duty is required to be paid in connection with such transfer and that we will not pay such stamp duty, such parties may either themselves arrange for the execution of the required instrument of transfer (and may request a form of instrument of transfer from Mallinckrodt for this purpose) or request that Mallinckrodt execute an instrument of transfer on behalf of the transferring party in a form determined by Mallinckrodt. In either event, if the parties to the share transfer have the instrument of transfer duly stamped (to the extent required) and then provide it to Mallinckrodt's transfer agent, the transferee will be registered as the legal owner of the relevant shares on Mallinckrodt's official Irish share register (subject to the matters described below).

The directors of Mallinckrodt may decline to recognize any instrument of transfer unless (i) it is accompanied by such evidence as the directors may reasonably require to show the right of the transferor to make the transfer; (ii) it is in respect of one class of share only; (iii) it is in favor of not more than four transferees; and (iv) it is lodged at the registered office of Mallinckrodt or at such other place as the directors may appoint. In the case of a transfer of shares by means other than a sale through a stock exchange on which the shares are listed, the directors have absolute discretion to decline to register such transfer of a share that is not fully paid or that is transferred to or by a minor or person of unsound mind.

The registration of transfers may be suspended by the directors at such times and for such period, not exceeding in the whole 30 days in each year, as the directors may from time to time determine.

-190-

#### Limitations on Liability, Indemnification of Directors and Officers and Insurance

Under Irish law, a company may not exempt its directors from liability for negligence or a breach of duty. However, where a breach of duty has been established, directors may be statutorily exempted by an Irish court from personal liability for negligence or breach of duty if, among other things, the court determines that they have acted honestly and reasonably, and that they may fairly be excused as a result.

The Irish Companies Acts only permit a company to pay the costs or discharge the liability of a director or the Secretary where judgment is given in his/her favor in any civil or criminal action in respect of such costs or liability, or where an Irish court grants relief because the director or Secretary acted honestly and reasonably and ought fairly to be excused. This restriction does not apply to executives who are not directors or the Secretary of Mallinckrodt. Any obligation of an Irish company which purports to indemnify a director or secretary of an Irish company over and above this will be void under Irish law, whether contained in its articles of association or any contract between the director and the company.

In addition, the articles of association of Mallinckrodt also contain an indemnity for officers (other than the Secretary).

The directors of Mallinckrodt may on a case-by-case basis decide at their discretion that it is in the best interest of Mallinckrodt to indemnify an individual director from any liability arising from his or her position as a director of Mallinckrodt. However, this discretion must be exercised bona fide in the best interests of Mallinckrodt as a whole.

Irish companies may take out directors' and officers' liability insurance, as well as other types of insurance, for their directors and officers.

In connection with the separation, we expect that Mallinckrodt and one of its subsidiaries will enter into indemnification agreements with each of the directors of Mallinckrodt and its Secretary that will provide for indemnification and expense advancement (except in cases where Mallinckrodt or any of its subsidiaries is proceeding against the indemnitee) and include related provisions meant to facilitate the indemnitee's receipt of such benefits.

The limitation of liability and indemnification provisions described above may discourage shareholders from bringing a lawsuit against directors for breach of their fiduciary duty. These provisions may also have the effect of reducing the likelihood of derivative litigation against Mallinckrodt's directors and officers, even though such an action, if successful, might otherwise benefit Mallinckrodt and its shareholders. However, these provisions will not limit or eliminate Mallinckrodt's rights, or those of any shareholder, to seek non-monetary relief such as injunction or rescission in the event of a breach of a director's duty of care. The provisions will not alter the liability of directors under the federal securities laws. In addition, your investment may be materially adversely affected to the extent that, in a class action or direct suit, Mallinckrodt pays the costs of settlement and damage awards against directors and officers pursuant to these indemnification provisions. There is currently no pending material litigation or proceeding against any Mallinckrodt director, officer or employee for which indemnification is being sought.

#### Sale of Unregistered Securities

Upon its incorporation on January 9, 2013, Mallinckrodt issued one ordinary share of 0.20 to each of seven nominee companies (*i.e.*, seven ordinary shares in total) to hold on trust for an Irish corporate services provider. On January 11, 2013, Mallinckrodt issued 40,000 ordinary A shares of 0.20 each to the abovementioned Irish corporate services provider. Mallinckrodt did not register either of these issuances under the Securities Act because such issuances did not constitute public offerings and therefore were exempt from registration pursuant to Section 4(2) of the Securities Act. Each share has been issued for cash at its par value.

#### **Transfer Agent and Registrar**

After the distribution, the transfer agent and registrar for our ordinary shares will be Computershare.

-191-

## WHERE YOU CAN FIND MORE INFORMATION

We have filed a registration statement on Form 10 with the SEC with respect to the ordinary shares of Mallinckrodt being distributed as contemplated by this information statement. This information statement is a part of, and does not contain all of the information set forth in, the registration statement and the exhibits and schedules to the registration statement. For further information with respect to us and our ordinary shares, please refer to the registration statement, including its exhibits and schedules. Statements made in this information statement relating to any contract or other document are not necessarily complete, and you should refer to the exhibits attached to the registration statement for copies of the actual contract or document. You may review a copy of the registration statement, including its exhibits and schedules, at the SEC's public reference room, located at 100 F Street, N.E., Washington, D.C. 20549, by calling the SEC at 1-800-SEC-0330 or via the Internet website maintained by the SEC at www.sec.gov. Information contained on any website referenced in this information statement.

As a result of the distribution, we will become subject to the information and reporting requirements of the Exchange Act and, in accordance with the Exchange Act, we will file periodic reports, proxy statements and other information with the SEC.

We intend to furnish holders of our ordinary shares with annual reports containing consolidated financial statements prepared in accordance with accounting principles generally accepted in the U.S. and audited and reported on, with an opinion expressed, by an independent registered public accounting firm.

You should rely only on the information contained in this information statement or to which we have referred you. We have not authorized any person to provide you with different information or to make any representation not contained in this information statement.

-192-

# INDEX TO COMBINED FINANCIAL STATEMENTS THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC

	Page
Annual Audited Combined Financial Statements	
Report of Independent Registered Public Accounting Firm	F-2
Combined Statements of Income for fiscal years ended September 28, 2012, September 30, 2011 and September 24, 2010	F-3
Combined Statements of Comprehensive Income for fiscal years ended September 28, 2012, September 30, 2011 and September 24, 2010	F-4
Combined Balance Sheets at September 28, 2012 and September 30, 2011	F-5
Combined Statements of Parent Company Equity for fiscal years ended September 28, 2012, September 30, 2011 and September 24, 2010	F-6
Combined Statements of Cash Flows for fiscal years ended September 28, 2012, September 30, 2011 and September 24, 2010	F-7
Notes to Combined Financial Statements	F-8
Interim Unaudited Condensed Combined Financial Statements	
Condensed Combined Statements of Income for the six months ended March 29, 2013 and March 30, 2012 (unaudited)	F-46
	F 47

Condensed Combined Statements of Comprehensive Income for the six months ended March 29, 2013 and March 30, 2012 (unaudited)	F-47
Condensed Combined Balance Sheets at March 29, 2013 and September 28, 2012 (unaudited)	F-48
Condensed Combined Statements of Cash Flows for the six months ended March 29, 2013 and March 30, 2012 (unaudited)	F-49
Notes to Condensed Combined Financial Statements (unaudited)	F-50

## MALLINCKRODT PLC

Audited Balance Sheet at Capitalization	
Report of Independent Registered Public Accounting Firm	F-66
Balance Sheet at January 11, 2013	F-67
Notes to Financial Statements	F-68
Interim Unaudited Financial Statements	-
Statement of Comprehensive Loss for the period from January 11, 2013 (date of capitalization) to March 29, 2013 (unaudited)	F- 69
Balance Sheets at March 29, 2013 and January 11, 2013 (date of capitalization) (unaudited)	F- 70
Statement of Retained Deficit for the period from January 11, 2013 (date of capitalization) to March 29, 2013 (unaudited)	F- 71
Statement of Cash Flows for the period from January 11, 2013 (date of capitalization) to March 29, 2013 (unaudited)	F- 72
Notes to Financial Statements (unaudited)	F- 73

## REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Shareholders of Covidien plc:

We have audited the accompanying combined balance sheets of the Pharmaceuticals business of Covidien plc (such business referred to as the "Company") as of September 28, 2012 and September 30, 2011 and the related combined statements of income, comprehensive income, parent company equity and cash flows for each of the three fiscal years in the period ended September 28, 2012. The Company's management is responsible for these combined financial statements. Our responsibility is to express an opinion on these combined financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the combined financial statements present fairly, in all material respects, the financial position of the Company as of September 28, 2012 and September 30, 2011, and the results of its operations and its cash flows for each of the three fiscal years in the period ended September 28, 2012, in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note 1 to the combined financial statements, the Company is comprised of the assets and liabilities used in managing the Pharmaceuticals business of Covidien plc. The combined financial statements include expense allocations for certain corporate functions historically provided by Covidien plc. These allocations may not be reflective of the actual expenses which would have been incurred had the Company operated as a separate entity apart from Covidien plc.

/s/ DELOITTE & TOUCHE LLP St. Louis, Missouri February 1, 2013

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC COMBINED STATEMENTS OF INCOME Fiscal Years Ended September 28, 2012, September 30, 2011 and September 24, 2010 (in millions)

	2012	2011	2010
Net sales (including sales to a related party of \$54.2, \$52.4 and \$50.5)	\$2,056.2	\$2,021.8	\$2,047.6
Cost of sales (including purchases from a related party of \$34.7, \$41.1 and \$38.1)	1,091.4	1,106.9	1,115.2
Gross profit	964.8	914.9	932.4
Selling, general and administrative expenses	551.7	532.5	565.3
Research and development expenses	144.1	141.5	119.1
Separation costs	25.5	2.9	—
Restructuring charges, net	11.2	8.4	11.5
Gain on divestitures	(2.9)	(11.1)	(3.9)
Operating income	235.2	240.7	240.4
Other income, net (including royalties from a related party of \$0.9, \$2.9 and \$3.5)	1.0	2.9	3.4
Interest expense	(0.5)	(0.6)	(0.7)
Interest income	0.4	0.2	0.1
Income from continuing operations before income taxes	236.1	243.2	243.2
Provision for income taxes	94.8	86.2	97.3
Income from continuing operations	141.3	157.0	145.9
(Loss) income from discontinued operations, net of income taxes	(6.7)	(6.3)	54.7
Net income	\$ 134.6	\$ 150.7	\$ 200.6

See Notes to Combined Financial Statements.

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC COMBINED STATEMENTS OF COMPREHENSIVE INCOME Fiscal Years Ended September 28, 2012, September 30, 2011 and September 24, 2010 (in millions)

	2012	2011	2010
Net income	\$134.6	\$150.7	\$200.6
Other comprehensive income (loss), net of tax			
Currency translation:			
Currency translation	(2.9)	(0.5)	(12.1)
Currency translation reclassified to net income due to business divestitures		—	3.3
	(2.9)	(0.5)	(8.8)
Defined benefit plans:			
Unrecognized net loss arising during the period	(18.5)	(9.2)	(25.9)
Prior service credit resulting from plan amendments	—	17.0	—
Amortization of prior service credit and net actuarial loss	3.4	4.1	7.8
Plan settlements and curtailments included in net periodic pension costs	(0.2)	5.0	7.5
	(15.3)	16.9	(10.6)
Income tax benefit (provision) relating to defined benefit plans	4.6	(4.5)	3.6
Total other comprehensive (loss) income, net of tax	(13.6)	11.9	(15.8)
Comprehensive income	\$121.0	\$162.6	\$184.8

See Notes to Combined Financial Statements.

F-4

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC COMBINED BALANCE SHEETS At September 28, 2012 and September 30, 2011 (in millions)

(		
	2012	2011
Assets		
Current Assets:		
Accounts receivable trade, less allowance for doubtful accounts of \$9.4 and \$5.7	\$ 291.1	\$ 302.2
Inventories	435.3	373.5
Prepaid expenses and other current assets	31.0	37.7
Deferred income taxes	119.9	130.5
Total current assets	877.3	843.9
Property, plant and equipment, net	945.2	906.3
Goodwill	507.5	507.5
Intangible assets, net	365.6	379.5
Other assets	179.0	186.2
Total Assets	\$2,874.6	\$2,823.4
Liabilities and Parent Company Equity		
Current Liabilities:		
Current maturities of long-term debt	\$ 1.3	\$ 1.3
Accounts payable	112.5	121.2
Accrued payroll and payroll-related costs	60.3	56.1
Accrued and other current liabilities	221.7	230.2
Total current liabilities	395.8	408.8
Long-term debt	8.9	10.4
Pension and postretirement benefits	189.6	202.9
Environmental liabilities	136.5	154.8
Deferred income taxes	73.7	76.1
Other liabilities	178.2	181.7
Total Liabilities	982.7	1,034.7
Commitments and contingencies (note 20)		
Parent Company Equity:		
Parent company investment	1,807.0	1,690.2
Accumulated other comprehensive income	84.9	98.5
Total Parent Company Equity	1,891.9	1,788.7

Total Liabilities and Parent Company Equity

See Notes to Combined Financial Statements.

\$2,874.6

\$2,823.4

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC COMBINED STATEMENTS OF PARENT COMPANY EQUITY Fiscal Years Ended September 28, 2012, September 30, 2011 and September 24, 2010 (in millions)

	Parent Company Investment	Com	umulated Other prehensive ncome	Total Parent Company Equity
Balance at September 25, 2009	\$ 1,914.0	\$	102.4	\$2,016.4
Net income	200.6		—	200.6
Other comprehensive loss, net of tax	—		(15.8)	(15.8)
Net transfers to parent	(365.3)			(365.3)
Balance at September 24, 2010	1,749.3		86.6	1,835.9
Net income	150.7		—	150.7
Other comprehensive income, net of tax	—		11.9	11.9
Net transfers to parent	(209.8)			(209.8)
Balance at September 30, 2011	1,690.2		98.5	1,788.7
Net income	134.6		—	134.6
Other comprehensive loss, net of tax	—		(13.6)	(13.6)
Net transfers to parent	(17.8)	_	_	(17.8)
Balance at September 28, 2012	\$ 1,807.0	\$	84.9	\$1,891.9

See Notes to Combined Financial Statements.

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC COMBINED STATEMENTS OF CASH FLOWS Fiscal Years Ended September 28, 2012, September 30, 2011 and September 24, 2010 (in millions)

	2012	2011	2010
Cash Flows from Operating Activities:			
Net income	\$ 134.6	\$ 150.7	\$ 200.6
Loss (income) from discontinued operations, net of income taxes	6.7	6.3	(54.7)
Income from continuing operations	141.3	157.0	145.9
Adjustments to reconcile net cash provided by continuing operating activities:			
Depreciation and amortization	130.9	119.8	114.2
Share-based compensation	10.7	10.3	12.0
Deferred income taxes	9.0	36.4	(7.4)
Gain on divestitures	(2.9)	(11.1)	(3.9)
Other non-cash items	(7.8)	(0.3)	7.0
Changes in assets and liabilities, net of the effects of divestitures:			
Accounts receivable, net	8.7	5.2	(32.4)
Inventories	(62.8)	12.2	2.9
Accounts payable	(8.3)	4.6	22.6
Income taxes	79.4	36.0	99.5
Accrued and other liabilities	(54.2)	(8.0)	18.0
Other	11.8	8.1	1.0
Net cash provided by continuing operating activities	255.8	370.2	379.4
Cash Flows from Investing Activities:			
Capital expenditures	(144.2)	(120.4)	(103.5)
Divestitures, net of cash retained by businesses sold	(3.8)	7.9	286.3
Purchase of product rights	(13.2)	—	(55.0)
Restricted cash	5.6	0.1	
Cash paid under license agreement		—	(15.0)
Other	3.4	(0.2)	1.5
Net cash (used in) provided by continuing investing activities	(152.2)	(112.6)	114.3
Cash Flows from Financing Activities:			
Repayment of capital leases	(1.3)	(1.3)	(1.2)
Excess tax benefit from stock-based compensation	1.7	1.8	1.0
Net transfers to parent	(104.0)	(258.1)	(505.0)
Net cash used in continuing financing activities	(103.6)	(257.6)	(505.2)
Discontinued Operations:			
Net cash provided by discontinued operating activities			22.8
Net cash used in discontinued investing activities	_		(11.3)
Net cash provided by discontinued operations			11.5
Net change in cash	\$	<u>\$                                    </u>	\$ _
Supplementary Cash Flow Information:			
Interest paid	\$ 0.6	\$ 0.6	\$ 0.7
Income taxes paid, net of refunds	\$ 4.9	\$ 11.6	\$ 23.2
	ψ 1.7	φ 11.0	φ <b>2</b> 0.2

See Notes to Combined Financial Statements.

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC NOTES TO COMBINED FINANCIAL STATEMENTS

## 1. Basis of Presentation

## Separation

On December 15, 2011, Covidien plc ("Covidien" or "parent") announced a plan to spin off its Pharmaceuticals business into a separate, publicly traded company. Upon completion of the separation, Mallinckrodt plc will be the parent company which will own the Pharmaceuticals business.

## **Basis of Presentation**

The Pharmaceuticals business of Covidien plc (such business referred to as the "Company"), presented herein, represents a combined reporting entity comprising the assets and liabilities used in managing and operating the Company, including corporations, branches and operations that have been carved out which relate to Covidien's Pharmaceuticals business. The combined financial statements have been presented on a standalone basis and are derived from the consolidated financial statements of Covidien. The combined financial statements have been prepared in United States ("U.S.") dollars and in accordance with accounting principles generally accepted in the U.S. ("GAAP"). The Company's combined financial statements may not be indicative of the Company's future performance and do not necessarily reflect what the results of operations, financial position and cash flows would have been had it operated as an independent, publicly traded company during the periods presented.

Intercompany transactions between the Company and Covidien have been included in these combined financial statements and are considered to be effectively settled for cash in the combined financial statements at the time the transaction is recorded. The total net effect of the settlement of these intercompany transactions is reflected in the combined statements of cash flows as a financing activity and in the combined balance sheets as parent company investment.

The combined financial statements include expense allocations for certain functions provided by Covidien, including, but not limited to, general corporate expenses related to finance, legal, information technology, human resources, communications, employee benefits and incentives, insurance and stock-based compensation. These expenses have been allocated to the Company on the basis of direct usage when identifiable, with the remainder allocated on the basis of operating expenses, headcount or other measures. During fiscal 2012, 2011 and 2010, the Company was allocated \$49.2 million, \$56.3 million and \$60.8 million, respectively, of general corporate expenses incurred by Covidien which are included within selling, general and administrative expenses in the combined statements of income. Management considers the bases on which the expenses have been allocated to reasonably reflect the utilization of services provided to or the benefit received by the Company during the periods presented. The allocations may not, however, reflect the expense the Company would have incurred as an independent, publicly traded company for the periods presented. Actual costs that may have been incurred if the Company had been a standalone company would depend on a number of factors, including the organizational structure, what functions were outsourced or performed by employees and strategic decisions made in areas such as information technology and infrastructure. The Company is unable to determine what such costs would have been had the Company been independent. Following the separation, the Company will perform these functions using its own resources or purchased services. For an interim period, however, some of these functions will continue to be provided by Covidien under a transition services agreement, particularly in relation to areas outside the U.S.

The combined financial statements include certain assets and liabilities that have historically been recorded at the Covidien corporate level but are specifically identifiable or otherwise allocable to the Company. The cash and cash equivalents held by Covidien at the corporate level are not specifically identifiable to the Company. Accordingly, cash and cash equivalents have not been allocated to the Company for any of the periods presented.

Covidien's debt and the related interest expense have not been allocated to the Company for any of the periods presented since the Company is not the legal obligor of the debt and Covidien's borrowings were not directly attributable to the Company's business. Debt incurred by the Company directly has been included in the combined financial statements.

Covidien maintains self-insurance programs at the corporate level. The Company was allocated a portion of the expenses associated with these programs as part of the general corporate overhead expense allocation. In addition, certain product liability reserves have been allocated to the Company. No other self-insurance reserves have been allocated to the Company as the remaining self-insurance reserves represent obligations of Covidien, which are not transferrable.

The Company has disposed of some of the operations previously owned. Where appropriate, these operations have been reflected as discontinued operations in the combined financial statements. Divestitures of product lines not representing businesses have been reflected in operating income.

#### Fiscal Year

The Company reports its results based on a "52-53 week" year ending on the last Friday of September. Fiscal 2012 and 2010 consisted of 52 weeks and ended on September 28, 2012 and September 24, 2010, respectively. Fiscal 2011 ended on September 30, 2011 and consisted of 53 weeks. Unless otherwise indicated, references in the combined financial statements to 2012, 2011 and 2010 are to the Company's fiscal year ended September 28, 2012, September 30, 2011 and September 24, 2010, respectively.

## **Principles of Combination**

Entities in which Covidien owns or controls more than fifty percent of the voting shares or has the ability to control through similar rights are included in the combined financial statements to the extent they relate to Covidien's Pharmaceuticals business. All intracompany transactions and accounts between the Company's businesses have been eliminated. The results of entities disposed of are included in the combined financial statements up to the date of disposal.

## Use of Estimates

The preparation of the combined financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amount of assets and liabilities, disclosure of contingent assets and liabilities, and the reported amounts of revenues and expenses. Actual results may differ from those estimates.

## 2. Summary of Significant Accounting Policies

## **Revenue Recognition**

The Company recognizes revenue for product sales when title and risk of loss have transferred from the Company to the buyer, which may be upon shipment or upon delivery to the customer site, based on contract terms or legal requirements in non-U.S. jurisdictions. The Company sells products direct to retail pharmacies and end user customers and through distributors who resell the products to retail pharmacies, institutions and end user customers. Chargebacks and rebates represent credits that are provided to certain distributors and customers for either the difference between the Company's contracted price with a customer and the distributor's invoice price paid to the Company or for contractually agreed volume price discounts. When the Company recognizes net sales, it simultaneously records an adjustment to revenue for estimated chargebacks, rebates, product returns and other sales deductions. These provisions are estimated based upon: historical experience, estimated future trends, estimated customer inventory levels, current contracted sales terms with customers, level of utilization of the Company's products and other competitive factors. The Company adjusts these reserves to reflect differences between estimated activity and actual experience. Such adjustments impact the amount of net sales recognized by the Company in the period of adjustment.

Taxes collected from customers relating to product sales and remitted to governmental authorities are accounted for on a net basis. Accordingly, such taxes are excluded from both net sales and expenses.

## Shipping and Handling Costs

Shipping costs, which are costs incurred to physically move product from the Company's premises to the customer's premises, are classified as selling, general and administrative expenses. Handling costs, which are costs incurred to store, move and prepare product for shipment, are classified as cost of sales. Shipping costs included in selling, general and administrative expenses were \$59.1 million, \$57.3 million and \$68.2 million in fiscal 2012, 2011 and 2010, respectively.

### **Research and Development**

Internal research and development costs are expensed as incurred. Research and development expenses include salary and benefits, allocated overhead and occupancy costs, clinical trial and related clinical manufacturing costs, contract services and other costs.

Upfront and milestone payments made to third parties under license arrangements are expensed as incurred up to the point of regulatory approval of the product. Milestone payments made to third parties subsequent to regulatory approval are capitalized as an intangible asset and amortized to cost of sales over the estimated useful life of the related product.

## Advertising

Advertising costs are expensed when incurred. Advertising expense for continuing operations was \$8.8 million, \$9.7 million and \$9.6 million in fiscal 2012, 2011 and 2010, respectively, and is included in selling, general and administrative expenses.

## **Currency Translation**

For the Company's non-U.S. subsidiaries that transact in a functional currency other than U.S. dollars, assets and liabilities are translated into U.S. dollars using fiscal year-end exchange rates. Revenues and expenses are translated at the average exchange rates in effect during the related month. The net effect of these translation adjustments is shown in the combined financial statements as a component of accumulated other comprehensive income within parent company equity. For subsidiaries operating in highly inflationary environments or where the functional currency is different from the local currency, non-monetary assets and liabilities are translated at the rate of exchange in effect on the date the assets and liabilities were acquired or assumed, while monetary assets and liabilities are translated at fiscal year-end exchange rates. Translation adjustments of these subsidiaries are included in net income. Gains and losses resulting from foreign currency transactions are included in net income.

#### Accounts Receivable and Allowance for Doubtful Accounts

Trade accounts receivable are presented net of an allowance for doubtful accounts. The allowance for doubtful accounts reflects an estimate of losses inherent in the Company's accounts receivable portfolio determined on the basis of historical experience, specific allowances for known troubled accounts and other available evidence. Accounts receivable are written off when management determines they are uncollectible.

#### Inventories

Inventories are recorded at the lower of cost or market value, primarily using the first-in, first-out convention. The Company reduces the carrying value of inventories for those items that are potentially excess, obsolete or slow-moving based on changes in customer demand, technology developments or other economic factors.

## Property, Plant and Equipment

Property, plant and equipment are stated at cost. Major renewals and improvements are capitalized, while routine maintenance and repairs are expensed as incurred. Depreciation for property, plant and equipment assets, other than land and construction in progress, is based upon the following estimated useful lives, using the straight-line method:

Buildings	5 to 45 years
Leasehold improvements	2 to 14 years
Machinery and equipment	3 to 20 years

The Company capitalizes certain computer software and development costs incurred in connection with developing or obtaining software for internal use. These costs are included in machinery and equipment and are amortized over the estimated useful lives of the software.

Upon retirement or other disposal of property, plant and equipment, the cost and related amount of accumulated depreciation are eliminated from the asset and accumulated depreciation accounts, respectively. The difference, if any, between the net asset value and the proceeds is included in net income.

The Company assesses the recoverability of assets using undiscounted cash flows whenever events or circumstances indicate that the carrying value of an asset may not be recoverable. If an asset is found to be impaired, the amount recognized for impairment is equal to the difference between the carrying value of the asset and the present value of future cash flows or other reasonable estimate of fair value.

#### Acquisitions

Amounts paid for acquisitions are allocated to the tangible assets acquired and liabilities assumed based on their estimated fair values at the date of acquisition. The Company then allocates the purchase price in excess of net tangible assets acquired to identifiable intangible assets, including purchased research and development. The fair value of identifiable intangible assets is based on detailed valuations that use information and assumptions provided by management. The Company allocates any excess purchase price over the fair value of the net tangible and intangible assets acquired to goodwill.

The Company's purchased research and development represents the estimated fair value as of the acquisition date of in-process projects that have not reached technological feasibility. The primary basis for determining technological feasibility of these projects is obtaining regulatory approval.

The value of in-process research and development ("IPR&D") is determined using the discounted cash flow method. In determining the value of IPR&D, the Company considers, among other factors, appraisals, the stage of completion of the projects, the technological feasibility of the projects, whether the projects have an alternative future use and the estimated residual cash flows that could be generated from the various projects and technologies over their respective projected economic lives. The discount rate used is determined at the time of acquisition and includes a rate of return which accounts for the time value of money, as well as risk factors that reflect the economic risk that the cash flows projected may not be realized.

The value attributable to IPR&D projects at the time of acquisition is capitalized as an indefinite-lived intangible asset and tested for impairment until the project is completed or abandoned. Upon completion of the project, the indefinite-lived intangible asset is then accounted for as a finite-lived intangible asset and amortized on a straight-line basis over its estimated useful life. If the project is abandoned, the indefinite-lived intangible asset is charged to expense. As of September 28, 2012, the Company had no IPR&D.

#### Goodwill and Other Intangible Assets

Goodwill represents the excess of the purchase price of an acquired entity over the amounts assigned to assets and liabilities assumed in a business combination. The Company tests goodwill for impairment during the fourth quarter of each year, or more frequently if impairment indicators arise. The Company utilizes a two-step approach. The first step requires a comparison of the carrying value of the reporting units to the fair value of these units. The Company estimates the fair value of its reporting units through internal analyses and valuation, utilizing an income approach based on the present value of future cash flows. If the carrying value of a reporting unit exceeds its fair value, the Company will perform the second step of the goodwill impairment test to measure the amount of impairment loss, if any. The second step of the goodwill impairment test compares the implied fair value of a reporting unit's goodwill with its carrying value. The implied fair value of goodwill is determined in the same manner that the amount of goodwill recognized in a business combination is determined. The Company allocates the fair value of a reporting unit to all of the assets and liabilities of that unit, including intangible assets, as if the reporting unit had been acquired in a business combination. Any excess of the value of a reporting unit over the amounts assigned to its assets and liabilities is the implied fair value of goodwill.

Intangible assets acquired in a business combination are recorded at fair value, while intangible assets acquired in other transactions are recorded at cost. Intangible assets with finite useful lives are subsequently amortized using the straight-line method over the following estimated useful lives of the assets:

Completed technology	5 to 25 years
License agreements	8 to 30 years
Trademarks	30 years

Amortization expense related to completed technology and certain other intangible assets is included in cost of sales, while amortization expense related to intangible assets that contribute to the Company's ability to sell, market and distribute products is included in selling, general and administrative expenses. The Company reviews intangible assets for impairment by comparing the fair value of the assets, estimated using an income approach, with their carrying value. If the carrying value exceeds the fair value of the intangible asset, the amount recognized for impairment is equal to the difference between the carrying value of the asset and the present value of future cash flows. The Company assesses the remaining useful life and the recoverability of finite-lived intangible assets whenever events or circumstances indicate that the carrying value of an asset may not be recoverable. Indefinite-lived intangible assets are tested for impairment at least annually.

#### Contingencies

The Company is subject to various patent, product liability, government investigations, environmental liability and other legal proceedings in the ordinary course of business. The Company records accruals for contingencies when it is probable that a liability has been incurred and the amount can be reasonably estimated. The Company discounts environmental liabilities using a risk-free rate of return when the obligation is fixed or reasonably determinable. The impact of the discount in the combined balance sheets was not material in any period presented. Legal fees, other than those pertaining to environmental and asbestos matters, are expensed as incurred. Insurance recoveries related to potential claims are recognized up to the amount of the recorded liability when coverage is confirmed and the estimated recoveries are probable of payment. Assets and liabilities are not netted for financial statement presentation.

#### Asset Retirement Obligations

The Company's obligations to decommission two facilities upon a cessation of its radiological licensed operations are included on the combined balance sheets as asset retirement obligations. In addition, the Company establishes asset retirement obligations for certain assets at the time they are installed. The present value of an asset retirement obligation is recorded as a liability when incurred. The liability is subsequently adjusted in future periods as accretion expense is recorded or as revised estimates of the timing or amount of cash flows required to retire the asset are obtained. The corresponding asset retirement costs are capitalized as part of the carrying value of the related long-lived asset and depreciated over the asset's useful life.

#### Income Taxes

Income taxes as presented are calculated on a separate tax return basis (inclusive of certain loss benefits), although the Company's operations have historically been included in Covidien's U.S. federal and state tax returns or the tax returns of non-U.S. jurisdictions. Accordingly, the income taxes presented may not be reflective of the results that would have occurred on a standalone basis.

With the exception of certain non-U.S. entities, the Company does not maintain taxes payable to or from Covidien and the Company is deemed to settle the annual current tax balances immediately with the legal tax-paying entities in the respective jurisdictions. These settlements are reflected as changes in parent company investment.

Deferred tax assets and liabilities are recognized for the expected future tax consequences of events that have been reflected in the combined financial statements. Deferred tax assets and liabilities are determined based on the differences between the book and tax bases of assets and liabilities and operating loss carryforwards, using tax rates expected to be in effect for the years in which the differences are expected to reverse. A valuation allowance is provided to reduce net deferred tax assets if, based upon the available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized.

The Company determines whether it is more likely than not that a tax position will be sustained upon examination. The tax benefit of any tax position that meets the more-likely-than-not recognition threshold is calculated as the largest amount that is more than 50% likely of being realized upon resolution of the uncertainty. To the extent a full benefit is not expected to be realized on the uncertain tax position, an income tax liability is established. Interest and penalties on income tax obligations, including uncertain tax positions, are included in the provision for income taxes.

The calculation of the Company's tax liabilities involves dealing with uncertainties in the application of complex tax regulations in a multitude of jurisdictions across the Company's global operations. Due to the complexity of some of these uncertainties, the ultimate resolution may result in a payment that is materially different from current estimates of the tax liabilities. If the Company's estimate of tax liabilities proves to be less than the ultimate assessment, an additional charge to expense would result. If payment of these amounts ultimately proves to be less than the recorded amounts, the reversal of the liabilities may result in income tax benefits being recognized in the period when it is determined that the liabilities are no longer necessary. A significant portion of these potential tax liabilities are recorded in other liabilities on the combined balance sheets as payment is not expected within one year.

#### **Parent Company Investment**

Parent company investment in the combined balance sheets represents Covidien's historical investment in the Company, the Company's accumulated net earnings after income taxes, and the net effect of transactions with and allocations from Covidien.



## 3. Discontinued Operations and Divestitures

## **Discontinued Operations**

During fiscal 2010, the Specialty Chemicals business (formerly known as "Mallinckrodt Baker"), which was part of the Company's Specialty Pharmaceuticals segment, was sold for net cash proceeds of \$273.3 million. Mallinckrodt Baker was sold because its products and customer bases were not aligned with the Company's long-term strategic objectives. This business met the discontinued operations criteria and, accordingly, is included in discontinued operations for all periods presented.

In connection with this transaction, the Company recorded a \$20.4 million pre-tax gain on the sale of Mallinckrodt Baker during fiscal 2010. Included within this gain was a \$17.7 million pre-tax charge associated with indemnification obligations to the purchaser, which are discussed in note 13.

During fiscal 2011, the Company recorded a \$9.1 million pre-tax loss on the sale of Mallinckrodt Baker, primarily for pension settlements related to employees of this business. In addition, during fiscal 2012, the Company recorded an additional \$6.7 million loss, primarily related to the indemnification obligations to the purchaser, which are discussed in note 13.

Net sales, income from operations and (loss) income on disposition for discontinued operations are as follows:

(Dollars in Millions)	2012	2011	2010
Net sales	\$	<u></u> \$—	\$400.4
Income from operations, net of income tax provision of \$, \$ and \$28.3	\$—	\$—	\$ 32.6
(Loss) income on disposition, net of income tax benefit of \$, \$2.8 and \$1.7	(6.7)	(6.3)	22.1
(Loss) income from discontinued operations, net of income taxes	\$(6.7)	\$(6.3)	\$ 54.7

### Divestitures

During fiscal 2011, the Company sold the rights to market TussiCaps extended-release capsules, a cough suppressant, for an upfront cash payment of \$11.5 million. As a result of this transaction, the Company recorded an \$11.1 million gain. The purchaser also may be obligated to make contingent payments to the Company of up to \$11.5 million from December 31, 2011 through September 30, 2015, payable in equal quarterly installments until such time as a new competitive generic product is introduced into the market. In addition, the Company would receive a \$1.0 million contingent payment if certain sales targets are achieved over the same time period. The Company received \$2.9 million of contingent payments during fiscal 2012.

During fiscal 2010, the Company sold its nuclear radiopharmacies in the U.S. for net cash proceeds of \$13.0 million. As a result of this transaction, the Company recorded a \$3.9 million net gain. In connection with this sale, the Company also entered into a supply agreement, under which the purchaser committed to annual purchase volumes through December 31, 2014.

### 4. Product Acquisitions

*Roxicodone*—In August 2012, the Company's Specialty Pharmaceuticals segment paid \$13.2 million under an agreement to acquire all of the rights to Xanodyne Pharmaceuticals, Inc.'s Roxicodone, which was capitalized as an intangible asset. Roxicodone is an immediate-release oral formulation of oxycodone hydrochloride indicated for the management of moderate to severe pain where the use of an opioid analgesic is appropriate. Roxicodone is the Reference Listed Drug for one of the Company's generic products and is important to the Company's product pipeline. There are no ongoing royalty payments under this agreement.

*Exalgo*—In June 2009, the Company's Specialty Pharmaceuticals segment acquired the rights to market and distribute the pain management drug Exalgo in the U.S., for an upfront cash payment of \$10.0 million, which was included in research and development expenses during fiscal 2009. Under the license arrangement, the Company is obligated to make additional payments of up to \$73.0 million based upon the successful completion of specified development and regulatory milestones. During fiscal 2009, \$10.0 million of such milestone payments were made and included in research and development expenses. During fiscal 2010, the U.S. Food and Drug Administration ("FDA") approved the Exalgo New Drug Application ("NDA") for the 8 mg, 12 mg and 16 mg tablet dosage forms, resulting in additional payments of \$55.0 million, which were capitalized as an intangible asset. In addition, during fiscal 2012, the Company received FDA approval to market a 32 mg tablet dosage form. The Company is also required to pay royalties on sales of the product. During fiscal 2012, 2011 and 2010, the Company paid royalties of \$16.1 million, \$5.5 million and \$4.4 million, respectively.

#### 5. License Agreements

Depomed, Inc.—In October 2009, the Company's Specialty Pharmaceuticals segment licensed worldwide rights to utilize Depomed, Inc.'s ("Depomed") Acuform gastric retentive drug delivery technology for the exclusive development of four products. Under this license agreement, the Company paid Depomed upfront and development payments of \$5.3 million during fiscal 2009. In addition to these payments, the Company may be obligated to pay up to \$64 million in additional development milestone payments. The Company will also pay Depomed a royalty on sales of products developed under this license agreement. During fiscal 2012 and 2010, an insignificant amount of milestone payments were expensed as incurred since regulatory approval had not yet been received. No milestone payments were made in fiscal 2011. In addition, no royalties have been paid through fiscal 2012.

*Pennsaid*—In June 2009, the Company's Specialty Pharmaceuticals segment entered into a licensing agreement which granted it rights to market and distribute Pennsaid and MNK-395, product candidates for the treatment of osteoarthritis of the knee. This license arrangement included an upfront cash payment of \$10.0 million, which was included in research and development expenses during fiscal 2009. The Company is also responsible for all future development activities and expenses. In addition, the Company may be required to make additional payments of up to \$120 million based upon the successful completion of specified regulatory and sales milestones, and is required to pay royalties on sales of the products. During fiscal 2010, upon FDA approval of the Pennsaid NDA, the Company made a milestone payment of \$15.0 million, which was capitalized as an intangible asset. During fiscal 2012, the Company paid royalties of \$7.5 million associated with this product. The amount of royalties the Company paid during fiscal 2011 and 2010 were insignificant. MNK-395 is currently under FDA review.

#### 6. Restructuring and Related Charges, Net

During fiscal 2011 and fiscal 2009, Covidien launched restructuring programs designed to improve its cost structure. The 2009 program is substantially completed. The Company expects to incur charges under the 2011 program as the specific actions required to execute on these initiatives are identified and approved, most of which are expected to be incurred by the end of fiscal 2014.

Net restructuring and related charges by segment are as follows:

(Dollars in Millions)	2012	2011	2010
Specialty Pharmaceuticals	<u>2012</u> \$11.3	\$ 6.5	2010 \$ 3.3
Global Medical Imaging	7.9	3.8	8.4
Corporate	—	(0.3)	(0.2)
	19.2	10.0	11.5
Less: accelerated depreciation	(8.0)	(1.6)	
Restructuring charges, net	\$11.2	\$ 8.4	\$11.5

Net restructuring and related charges are comprised of the following:

(Dollars in Millions)	2012	2011	2010
2011 program	\$21.0	\$ 9.4	\$ —
2009 program	(1.8)	0.6	11.5
	19.2	10.0	11.5
Less: non-cash charges, including accelerated depreciation	(6.2)	(1.6)	(0.3)
Total charges expected to be settled in cash	\$13.0	\$ 8.4	\$11.2

The following table summarizes cash activity for restructuring reserves that are specifically identifiable to the Company, substantially all of which relates to employee severance and benefits:

(Dollars in Millions)	2011 Program	2009 Program	Total
Balance at September 25, 2009	\$	\$ 13.0	\$ 13.0
Charges	_	11.7	11.7
Changes in estimate	—	(0.5)	(0.5)
Cash payments	—	(14.8)	(14.8)
Reclassifications <sup>(1)</sup>		(4.6)	(4.6)
Currency translation	—	(0.3)	(0.3)
Balance at September 24, 2010		4.5	4.5
Charges	7.8	1.8	9.6
Changes in estimate	—	(1.2)	(1.2)
Cash payments	(0.2)	(3.3)	(3.5)
Reclassifications <sup>(1)</sup>	_	(1.6)	(1.6)
Currency translation	(0.2)	—	(0.2)
Balance at September 30, 2011	7.4	0.2	7.6
Charges	12.5	0.3	12.8
Changes in estimate	0.3	(0.1)	0.2
Cash payments	(11.3)	(0.2)	(11.5)
Reclassifications <sup>(1)</sup>	(0.1)	(0.1)	(0.2)
Balance at September 28, 2012	\$ 8.8	\$ 0.1	\$ 8.9

<sup>(1)</sup> Primarily represents the reclassification of pension and other postretirement benefits from restructuring reserves to pension and post retirement obligations.

Net restructuring and related charges, including associated asset impairments, incurred cumulative to date related to the 2011 program are as follows:

(Dollars in Millions)	2011	Program
Specialty Pharmaceuticals	\$	16.7
Global Medical Imaging		13.7
Total	\$	30.4

Restructuring reserves are reported on the Company's combined balance sheets in accrued and other current liabilities.

### 7. Income Taxes

The U.S. and non-U.S. components of income from continuing operations before income taxes were as follows:

(Dollars in Millions)	2012	2011	2010
U.S.	\$174.6	\$134.9	\$152.8
Non-U.S.	61.5	108.3	90.4
	\$236.1	\$243.2	\$243.2

Significant components of income taxes related to continuing operations are as follows:

(Dollars in Millions)	2012	2011	2010
Current:			
United States:			
Federal	\$61.1	\$19.2	\$ 58.3
State	7.2	2.4	11.8
Non-U.S.	17.5	28.2	34.6
Current income tax provision	85.8	49.8	104.7
Deferred:			
United States:			
Federal	5.3	37.8	(5.7)
State	2.4	4.3	(0.6)
Non-U.S.	1.3	(5.7)	(1.1)
Deferred income tax provision (benefit)	9.0	36.4	(7.4)
	\$94.8	\$86.2	\$ 97.3

The reconciliation between U.S. federal income taxes at the statutory rate and the Company's provision for income taxes on continuing operations is as follows:

(Dollars in Millions)	2012	2011	2010
Notional U.S. federal income taxes at the statutory rate	\$82.6	\$ 85.1	<b>2010</b> \$ 85.1
Adjustments to reconcile to the income tax provision:			
U.S. state income tax provision, net	7.1	5.9	7.0
Rate differences between non-U.S. and U.S. jurisdictions <sup>(1)</sup>	(3.5)	(16.8)	(10.7)
Adjustments to accrued income tax liabilities and uncertain tax positions	2.3	0.9	10.4
Withholding tax, net	0.4	3.8	1.1
Credits, principally research	(0.8)	(4.1)	(0.7)
Nondeductible expenses	8.1	8.4	7.4
Other	(1.4)	3.0	(2.3)
Provision for income taxes	\$94.8	\$ 86.2	\$ 97.3

<sup>(1)</sup> Excludes non-deductible charges and other items which are broken out separately in the statutory rate reconciliation presented.

As of September 28, 2012, September 30, 2011 and September 24, 2010, the amounts of unrecognized tax benefits for which the Company is legally and directly liable and would be required to remit cash if not sustained were \$13.4 million, \$14.2 million and \$15.9 million, respectively. Historically, the Company's operations have been included in tax returns filed by Covidien or certain of its subsidiaries not included in the combined financial statements. As a result, some federal uncertain tax positions related to the Company's operations result in

unrecognized tax benefits that are obligations of entities not included in the combined financial statements. Because the activities that give rise to these unrecognized tax benefits relate to the Company's operations, the impact of these items (presented in the table below) have been charged to the income tax provision through parent company investment, a component of parent company equity.

The following table summarizes the activity related to the Company's unrecognized tax benefits, excluding interest:

(Dollars in Millions)	2012	2011	2010
Balance at beginning of fiscal year	\$168.4	\$175.7	\$183.5
Additions related to current year tax positions	1.3	2.2	2.8
Additions related to prior period tax positions	1.6	1.1	1.1
Reductions related to prior period tax positions	(1.9)	(3.9)	(8.5)
Settlements	(1.7)	(6.7)	(0.2)
Lapse of statute of limitations	(2.2)		(3.0)
Balance at end of fiscal year	165.5	168.4	175.7
Cash advance paid in connection with proposed settlements	(23.5)	(23.5)	_
Balance at end of fiscal year, net of cash advance	\$142.0	\$144.9	\$175.7
-			

During fiscal 2011, Covidien made a \$35.1 million advance payment to the U.S. Internal Revenue Service in connection with the proposed settlement of certain tax matters. This payment was comprised of \$23.5 million of tax and \$11.6 million of interest.

Unrecognized tax benefits are reported in the following combined balance sheet captions in the amount shown:

(Dollars in Millions)	2012	2011
Other liabilities	\$ 13.4	\$ 14.2
Parent company investment	152.1	154.2
	\$165.5	\$168.4

The Company had unrecognized tax benefits of \$144.3 million, \$144.8 million and \$149.8 million as of September 28, 2012, September 30, 2011 and September 24, 2010, respectively, which if settled favorably would benefit the effective tax rate. The remaining \$21.2 million, \$23.6 million and \$25.9 million of unrecognized tax benefits as of September 28, 2012, September 30, 2011 and September 24, 2010, respectively, would be offset by the write off of related deferred and other tax assets, if recognized. During fiscal 2012, 2011 and 2010, the Company accrued additional interest of \$1.4 million, \$3.8 million and \$6.5 million, respectively. The total amount of accrued interest related to uncertain tax positions was \$33.9 million, \$32.5 million and \$40.3 million at September 28, 2012, September 24, 2010, respectively, of which \$26.0 million, \$24.8 million and \$32.3 million was included in parent company investment as of September 28, 2012, September 30, 2011 and September 24, 2010, respectively. Non-current income taxes payable also includes anticipated refunds and other items not related to uncertain tax positions.

Income taxes payable is reported in the following combined balance sheet captions in the amounts shown:

(Dollars in Millions)	2012	2011
Accrued and other current liabilities	\$ 2.6	2011 \$ 0.7
Other liabilities	19.4	19.9
	\$22.0	\$20.6

Covidien continues to be examined by various tax authorities. The resolution of these tax matters could result in a significant change in the Company's unrecognized tax benefits. However, the Company does not expect that the total amount of unrecognized tax benefits will significantly change over the next twelve months.

As of September 28, 2012, tax years that remain subject to examination in the Company's major tax jurisdictions are as follows:

Jurisdiction	Earliest Open Year
United States—federal and state	1996
Australia	2008
Canada	2004
France	2000
Germany	2003
Ireland	2008
Italy	2005
Japan	2006
Netherlands	2005
Switzerland	2004
United Kingdom	2009

Deferred income taxes result from temporary differences between the amount of assets and liabilities recognized for financial reporting and tax purposes. The components of the net deferred tax asset at the end of each fiscal year are as follows:

(Dollars in Millions)	2012	2011
Deferred tax assets:		
Accrued liabilities and reserves	\$ 47.4	\$ 46.0
Inventories	36.4	40.7
Environmental liabilities	66.4	75.6
Rebate reserves	38.1	38.7
Indemnification reserves	14.9	15.3
Postretirement benefits	67.7	74.5
Other	20.8	24.9
	291.7	315.7
Deferred tax liabilities:		
Property, plant and equipment	(139.9)	(150.0)
Intangible assets	(89.1)	(94.2)
	(229.0)	(244.2)
Net deferred tax asset before valuation allowances	62.7	71.5
Valuation allowances	(15.3)	(15.6)
Net deferred tax asset	\$ 47.4	\$ 55.9

Deferred taxes are reported in the following combined balance sheet captions in the amounts shown:

(Dollars in Millions)	2012	2011
Deferred income taxes (current assets)	<u>2012</u> \$119.9	<u>2011</u> \$130.5
Other assets	3.8	3.2
Accrued and other current liabilities	(2.6)	(1.7)
Deferred income taxes (non-current liabilities)	(73.7)	(76.1)
Net deferred tax asset	\$ 47.4	\$ 55.9

At September 28, 2012, the Company had approximately \$4.6 million of net operating loss carryforwards in certain non-U.S. jurisdictions, of which \$2.4 million have no expiration, and the remaining \$2.2 million will expire in future years through 2021.

The valuation allowances for deferred tax assets of \$15.3 million and \$15.6 million at September 28, 2012 and September 30, 2011, respectively, relate principally to the uncertainty of the utilization of certain deferred tax assets, primarily certain reserves in non-U.S. jurisdictions and unrealized capital losses in the U.S. The Company believes that it will generate sufficient future taxable income to realize the tax benefits related to the remaining net deferred tax assets.

During fiscal 2012 and 2011, the Company provided for U.S. and non-U.S. income and withholding taxes in the amount of \$0.4 million and \$3.8 million, respectively, on earnings that were or are intended to be repatriated. In general, the remaining earnings of the Company's subsidiaries are considered to be permanently reinvested. Income taxes are not provided on undistributed earnings of U.S. and non-U.S. subsidiaries that are either indefinitely reinvested or can be distributed on a tax-free basis. It is not practicable to determine the cumulative amount of undistributed earnings and tax liability that would arise if these earnings were remitted.

#### 8. Inventories

At the end of fiscal 2012 and 2011, inventories were comprised of:

(Dollars in Millions)	2012	2011
Raw materials and supplies	$\frac{2012}{\$74.1}$	<b>2011</b> \$ 73.7
Work in process	184.7	161.3
Finished goods	176.5	138.5
Inventories	\$435.3	\$373.5

## 9. Property, plant and equipment

At the end of fiscal 2012 and 2011, property, plant and equipment at cost and accumulated depreciation were:

(Dollars in Millions)	2012	2011
Land	\$ 60.0	\$ 60.4
Buildings and related improvements	297.3	276.6
Machinery and equipment	1,212.7	1,134.2
Construction in progress	181.4	154.5
	1,751.4	1,625.7
Less: accumulated depreciation	(806.2)	(719.4)
Property, plant and equipment, net	\$ 945.2	\$ 906.3

The amounts above include property under capital leases of \$17.0 million and \$17.9 million at September 28, 2012 and September 30, 2011, respectively, consisting primarily of buildings. Accumulated amortization of capitalized lease assets was \$14.3 million and \$14.1 million at the end of fiscal 2012 and 2011, respectively. In addition, machinery and equipment includes capitalized software costs of \$59.9 million and \$52.2 million at September 28, 2012 and September 30, 2011, respectively. Accumulated amortization of capitalized software was \$43.3 million and \$38.2 million at the end of fiscal 2012 and 2011, respectively.

Depreciation expense, including amounts related to capitalized leased assets, was \$103.6 million, \$92.8 million and \$90.8 million in fiscal 2012, 2011 and 2010, respectively.

## 10. Goodwill and Intangible Assets

Goodwill for the Company's operating segments consisted of the following:

(Dollars in Millions)	September 28, 2012	mber 30, 2011
Specialty Pharmaceuticals	\$ 287.8	\$ 287.8
Global Medical Imaging	219.7	219.7
	\$ 507.5	\$ 507.5

The gross carrying amount and accumulated amortization of intangible assets at the end of fiscal 2012 and 2011 were as follows:

	:	2012	:	2011
(Dollars in Millions)	Gross Carrying Amount	Accumulated Amortization	Gross Carrying Amount	Accumulated Amortization
Amortizable:				
Completed technology	\$ 376.1	\$ 173.7	\$ 362.8	\$ 159.0
Licenses	191.1	67.1	191.1	54.8
Trademarks	7.7	3.5	7.7	3.3
Total	\$ 574.9	\$ 244.3	\$ 561.6	\$ 217.1
Non-Amortizable:				
Trademarks	\$ 35.0		\$ 35.0	

Intangible asset amortization expense for fiscal 2012, 2011 and 2010 was \$27.3 million, \$27.0 million and \$23.4 million, respectively. The estimated aggregate amortization expense on intangible assets owned by the Company as of September 28, 2012 is expected to be \$29.7 million in fiscal 2013 through fiscal 2016 and \$28.2 million in fiscal 2017.

In fiscal 2008, the Company's Global Medical Imaging segment acquired completed technology, which facilitates the injection of contrast media. In fiscal 2010, the Company decided to market the technology for sale. However, the Company subsequently realized that a design flaw of the technology would prohibit the sale of the products without further investment. The Company decided not to make any further investment in the technology and, accordingly, recorded an impairment charge of \$4.6 million to write off the entire amount of the intangible asset, which is included in research and development expenses in fiscal 2010. The Company recorded total intangible asset impairments of \$6.4 million during fiscal 2010.

### 11. Related Party Transactions

The combined financial statements have been prepared on a standalone basis and are derived from the consolidated financial statements and accounting records of Covidien.

#### **Related Party Sales and Purchases**

During fiscal 2012, 2011 and 2010, the Company sold products to other Covidien businesses in the amount of \$54.2 million, \$52.4 million and \$50.5 million, respectively, which is included in net sales in the combined statements of income. The Company also purchases inventories from other Covidien businesses. The Company purchased and recognized in cost of sales inventory from Covidien of \$34.7 million, \$41.1 million and \$38.1 million in fiscal 2012, 2011 and 2010, respectively. As of September 28, 2012 and September 30, 2011, the aggregate amount of inventories purchased from other Covidien businesses that remained on the Company's combined balance sheets was insignificant.

#### **Royalty Income**

During fiscal 2012, 2011 and 2010, a subsidiary of Covidien paid royalties to the Company for use of certain trademarks and technology. Amounts outstanding under these agreements are considered settled for cash in the combined financial statements at the end of each reporting period and, as such, are included in parent company investment. During fiscal 2012, 2011 and 2010, the Company recognized royalty income of \$0.9 million, \$2.9 million and \$3.5 million, respectively, which is included in other income in the combined statements of income.

## Parent Company Equity

Covidien uses a centralized approach to cash management and financing of its operations, excluding debt directly incurred by any of its businesses. The Company's cash is transferred to Covidien daily and Covidien funds the Company's operating and investing activities as needed. Cash transfers to and from Covidien's cash management system are reflected as a component of parent company investment within parent company equity in the combined balance sheets.

Net transfers to parent are included within parent company investment on the combined statements of parent company equity. The components of the net transfers to parent for fiscal 2012, 2011 and 2010 are as follows:

(Dollars in Millions)	2012	2011	2010
Cash pooling and general financing activities	\$(84.0)	\$(258.2)	\$(209.8)
Corporate expense allocation	49.2	56.3	60.8
Cash transfer from (to) parent for divestitures	3.8	(7.9)	(286.3)
Cash transfer from parent for purchase of product rights and license	13.2		70.0
Total net transfers to parent	\$(17.8)	\$(209.8)	\$(365.3)

#### 12. Debt

At the end of fiscal 2012 and 2011, debt was comprised of:

(Dollars in Millions)	2012	2011
Current maturities of long-term debt:		
Capital lease obligation	\$ 1.3	\$ 1.3
Long-term debt:		
7% debentures due December 2013	5.8	5.8
Capital lease obligation	3.1	4.6
Total	8.9	10.4
Total debt	\$10.2	\$11.7

Since quoted market prices for the Company's debentures are not available in an active market, they are classified as level 2 for purposes of developing an estimate of their fair value. The fair value of the Company's debt did not differ significantly from its carrying value at September 28, 2012 and September 30, 2011.

The Company's capital lease obligation relates to a non-U.S. manufacturing facility. This lease expires in December 2015. The aggregate amounts of debt, including capital lease obligation, maturing during the next five fiscal years are as follows:

(Dollars in Millions)	
Fiscal 2013	\$ 1.3
Fiscal 2014	7.1
Fiscal 2015	1.4
Fiscal 2016	0.4
Fiscal 2017	—

#### 13. Guarantees

In disposing of assets or businesses, the Company often provides representations, warranties and indemnities to cover various risks and liabilities, including unknown damage to the assets, environmental risks involved in the sale of real estate, liability to investigate and remediate environmental contamination at waste disposal sites and manufacturing facilities and unidentified tax liabilities related to periods prior to disposition. Except as discussed below, the Company generally does not have the ability to estimate the potential liability from such indemnities because they relate to unknown conditions. However, the Company has no reason to believe that these uncertainties would have a material adverse effect on its financial condition, results of operations and cash flows.

In connection with the sale of Mallinckrodt Baker, the Company agreed to indemnify the purchaser with respect to various matters, including certain environmental, health, safety, tax and other matters. The indemnification obligations relating to environmental, health and safety matters have a term of 17 years, while some of the other indemnification obligations have an indefinite term. The amount of the liability relating to all of these indemnification obligations included in other liabilities on the Company's combined balance sheets at both September 28, 2012 and September 30, 2011 was \$22.4 million, of which \$18.3 million related to environmental, health and safety matters. The value of the environmental, health and safety indemnity was measured based on the probability-weighted present value of the costs expected to be incurred to address environmental, health and safety claims made under the indemnity. The aggregate fair value of these indemnification obligations did not differ significantly from their aggregate carrying value at September 28, 2012 and September 30, 2011. As of September 28, 2012, the maximum future payments the Company could be required to make under all of these indemnification obligations was \$76.5 million. The Company was required to pay \$30.0 million into an escrow account as collateral for all of these indemnification obligations to the purchaser, of which \$24.5 million and \$30.0 million remained in other assets on the combined balance sheet at September 28, 2012 and September 30, 2011, respectively.

The Company has recorded liabilities for known indemnification obligations included as part of environmental liabilities, which are discussed in note 20. In addition, the Company is liable for product performance; however in the opinion of management, such obligations will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

The Company is required to provide the Nuclear Regulatory Commission financial assurance demonstrating its ability to cover the cost of decommissioning its Maryland Heights, Missouri radiopharmaceuticals production facility upon closure, though the Company does not intend to close this facility. The Company has provided this financial assurance in the form of a \$58.0 million surety bond. In addition, as of September 28, 2012, the Company had a \$21.1 million letter of credit to guarantee decommissioning costs associated with its St. Louis, Missouri plant.

As of September 28, 2012, the Company had various other letters of credit and guarantee and surety bonds totaling \$25.8 million. In addition, at September 28, 2012, Covidien had outstanding letters of credit and guarantee and surety bonds totaling \$108.4 million, which supported multiple Covidien businesses, including the Company.

### 14. Derivative Instruments

The Company is exposed to certain risks relating to its business operations. Risks that relate to foreign exchange exposure and certain commodity price exposures are managed by participating in the centralized hedging functions of Covidien which are designed to minimize exposure to such risks. Foreign currency option and forward contracts are used to economically manage the foreign exchange exposures of operations outside the U.S. Swap contracts on commodities are periodically entered into to manage the price risk associated with forecasted purchases of commodities used in the Company's manufacturing processes. The associated derivative assets and liabilities have not been included on the Company's combined balance sheet since derivative activity is centrally managed by Covidien. Changes in the derivative financial instrument's fair value which related to the Company's business operations, however, have been recognized in the Company's earnings unless specific hedge criteria are met. Covidien has designated certain commodity swap contracts as cash flow hedges. Covidien has not designated the foreign currency forward and option contracts as hedging instruments.

## Foreign Exchange Exposure

Derivatives not designated as hedging instruments—The Company has foreign exchange exposure on the translation of the financial statements and on transactions denominated in foreign currencies. Covidien's policy is to use various forward and option contracts to economically manage foreign currency exposures on accounts and notes receivable, accounts payable, intercompany loans and forecasted transactions that are denominated in certain foreign currencies. Covidien generally manages its exposure for forecasted transactions for the upcoming 12 months. These contracts do not meet the necessary criteria to qualify for hedge accounting; accordingly, changes in fair value are recognized in earnings.

The amount of net (loss) gain on foreign exchange forward and option contracts not designated as hedging instruments and related hedged items were recorded as follows:

2012	2011	2010
$\overline{\$(0.3)}$	\$(3.7)	<u>\$</u> —
0.1	0.1	(3.1)
\$(0.2)	\$(3.6)	\$(3.1)
	\$(0.3) 0.1	$     \begin{array}{r} \$(0.3) & \$(3.7) \\ \hline       0.1 & 0.1 \\ \end{array} $

## **Commodities** Exposure

Covidien has entered into gas commodity swap contracts on behalf of the Company. The amounts of the net losses on these contracts were recorded as follows:

(Dollars in Millions) Cost of sales	<u>2012</u> \$0.9	<u>2011</u> \$0.8	<u>2010</u> \$1.1
Selling, general and administrative expenses	2.3	2.4	1.7
	\$3.2	\$3.2	\$2.8

## 15. Financial Instruments and Fair Value Measurements

Fair value is defined as the exit price that would be received from the sale of an asset or paid to transfer a liability, using assumptions that market participants would use in pricing an asset or liability. The fair value guidance establishes a three-level hierarchy, which maximizes the use of observable inputs and minimizes the use of unobservable inputs used in measuring fair value. The levels within the hierarchy are as follows:

Level 1— observable inputs such as quoted prices in active markets for identical assets or liabilities;

Level 2— significant other observable inputs that are observable either directly or indirectly; and

Level 3— significant unobservable inputs in which there is little or no market data, which requires the Company to develop its own assumptions.

The following tables provide a summary of the significant assets and liabilities that are measured at fair value on a recurring basis at September 28, 2012:

		Basis of Fair Value M Quoted Prices in Active Markets for Identical Assets and Liabilities	leasurement Significant Other Observable Inputs
(Dollars in Millions)	Total	(Level 1)	(Level 2)
Assets			
Debt and equity securities held in rabbi trust	\$25.2	\$ 13.7	\$ 11.5
Liabilities			
Deferred compensation liabilities	<u>\$ 9.3</u>	<u>\$ 9.3</u>	<u>\$                                    </u>

The following tables provide a summary of the significant assets and liabilities that are measured at fair value on a recurring basis at September 30, 2011:

		Quote in A Mar Identical	Basis of Fair Value M ed Prices Active kets for I Assets and bilities	Sig ( Obs	nificant Other servable nputs
(Dollars in Millions)	Total		evel 1)		evel 2)
Assets					
Debt and equity securities held in rabbi trust	\$22.5	\$	8.8	\$	13.7
Liabilities					
Deferred compensation liabilities	\$ 6.4	\$	6.4	\$	

Debt and equity securities held in rabbi trust—Debt securities held in the rabbi trust primarily consist of U.S. government and agency securities and corporate bonds. When quoted prices are available in an active market, the investments are classified as level 1. When quoted market prices for a security are not available in an active market, they are classified as level 2. Equity securities held in the rabbi trust primarily consist of U.S. common stocks, which are valued using quoted market prices reported on nationally recognized securities exchanges.

Deferred compensation liabilities—Covidien maintains a non-qualified deferred compensation plan in the U.S., which permits eligible employees of the Company to defer a portion of their compensation. A recordkeeping account is set up for each participant and the participant chooses from a variety of funds for the deemed investment of their accounts. The measurement funds generally correspond to the funds offered in Covidien's U.S. tax-qualified defined contribution retirement plan and the account balance fluctuates with the investment returns on those funds.

### Financial Instruments Not Measured at Fair Value

The fair value of restricted cash is equivalent to its carrying value of \$24.6 million and \$30.2 million as of September 28, 2012 and September 30, 2011, respectively (level 1), substantially all of which is included in other assets on the combined balance sheets. The Company's life insurance contracts are carried at cash surrender value (level 3). The fair value of these contracts approximates the carrying value of \$47.6 million and \$46.6 million at September 28, 2012 and September 30, 2011, respectively.

### Concentration of Credit and Other Risks

Financial instruments that potentially subject the Company to concentrations of credit risk primarily consist of accounts receivable. The Company does not require collateral from customers. A portion of the Company's trade accounts receivable outside the U.S., however, include sales to government-owned or supported healthcare systems in several countries, which are subject to payment delays. Payment is dependent upon the financial stability and creditworthiness of those countries' national economies. Deteriorating credit and economic conditions in parts of Western Europe, particularly in Spain and Italy, may continue to increase the average length of time it takes the Company to collect its accounts receivable in certain regions within these countries.

The Company routinely evaluates all government receivables for potential collection risks associated with the availability of government funding and reimbursement practices. While the Company's accounts receivable, net of allowance for doubtful accounts, in Greece is insignificant, during fiscal 2012, the Company recorded a \$4.4 million charge to write down its outstanding accounts receivable in Greece. This charge is included within selling, general and administrative expenses. The Company has not incurred any other significant losses on government receivables; however, if the financial condition of customers or the countries' healthcare systems continue to deteriorate such that their ability to make payments is uncertain, additional allowances may be required in future periods.

The Company's accounts receivable, net of the allowance for doubtful accounts, in Spain and Italy at the end of each period are as follows:

(Dollars in Millions)	2012	2011
Spain	\$15.0	\$26.6
Italy	12.5	14.7

Net sales to customers in Spain and Italy totaled \$55.0 million, \$60.2 million and \$58.7 million for fiscal 2012, 2011 and 2010, respectively.

The following table shows net sales attributable to distributors that accounted for 10% or more of the Company's total net sales:

	2012	2011	2010
Cardinal Health, Inc.	19%	19%	15%
McKesson Corporation	14%	13%	11%
AmerisourceBergen Corporation	9%	10%	8%

The following table shows accounts receivable attributable to distributors that accounted for 10% or more of the Company's gross accounts receivable at the end of each period:

	2012	2011
Cardinal Health, Inc.	19%	19%
McKesson Corporation	20%	16%
AmerisourceBergen Corporation	10%	12%

The following table shows net sales attributable to products that accounted for 10% or more of the Company's total net sales:

	2012	2011	2010
Optiray (CMDS)	17%	19%	17%
Acetaminophen products (API)	11%	11%	10%

Molybdenum-99 ("Mo-99") is a key raw material in the Company's Ultra-Technekow DTE technetium generators that are sold by its Global Medical Imaging segment. There are only eight suppliers of this raw material worldwide. The Company has agreements to obtain Mo-99 from three nuclear research reactors and relies predominantly on two of these reactors for its Mo-99 supply. Accordingly, a disruption in the commercial supply or a significant increase in the cost of this material from these sources could have a material adverse effect on the Company's financial condition, results of operations and cash flows.

#### 16. Retirement Plans

## **Defined Benefit Plans**

The Company sponsors a number of defined benefit retirement plans covering certain of its U.S. employees and non-U.S. employees. As of September 28, 2012, U.S. plans represented 73% of the Company's total pension plan assets and 78% of the Company's total projected benefit obligation. The Company generally does not provide postretirement benefits other than retirement plan benefits for its employees. However, certain of the Company's U.S. employees participate in postretirement benefit plans that provide medical benefits. These plans are unfunded.

During fiscal 2011, the Company amended one of its U.S. retiree medical plans to eliminate coverage for future retirees unless certain conditions were met. The plan amendment reduced the Company's overall obligation to participants by \$17.0 million and impacted both prior and future benefits under the plan. As a result of this amendment, the Company's net periodic benefit cost decreased by approximately \$8.6 million during fiscal 2011.

During fiscal 2011, the Company incurred settlement charges of \$11.1 million resulting from the level of lump-sum payments paid out of one of its U.S. pension plans, a significant portion of which were driven by the divestiture of Mallinckrodt Baker. During fiscal 2010, the Company incurred settlement charges of \$7.4 million resulting from the level of lump-sum payments paid out of one of its U.S. pension plans stemming primarily from restructuring actions.

The net periodic benefit cost (credit) for pension and postretirement benefit plans is as follows:

	Р	ension Benefit	s	Postretirement Benefits		
(Dollars in Millions)	2012	2011	2010	2012	2011	2010
Service cost	\$ 5.0	\$ 6.2	\$ 7.4	\$ 0.1	\$ 0.2	\$ 1.0
Interest cost	21.2	23.5	24.9	3.1	3.8	4.9
Expected return on plan assets	(24.5)	(25.3)	(23.8)		—	
Amortization of prior service cost (credit)	0.7	0.8	1.8	(9.2)	(9.0)	(5.8)
Amortization of net actuarial loss	11.7	11.8	11.5	0.2	0.5	0.3
Plan settlements (gain) loss	(0.2)	11.1	7.4	—	—	_
Curtailments	—	1.9	0.1		(4.6)	
Special termination benefits		0.1	1.8			_
Net periodic benefit cost (credit)	\$ 13.9	\$ 30.1	\$ 31.1	\$(5.8)	\$(9.1)	\$ 0.4

The following table represents the changes in benefit obligations, plan assets and the net amounts recognized on the combined balance sheet for pension and postretirement benefit plans at the end of fiscal 2012 and 2011:

(Dollars in Millions)	Pension 2012	Pension Benefits 2012 2011		ent Benefits 2011
Change in benefit obligations:			2012	
Projected benefit obligations at beginning of year	\$ 491.1	\$ 498.9	\$ 80.1	\$ 100.0
Service cost	5.0	6.2	0.1	0.2
Interest cost	21.2	23.5	3.1	3.8
Employee contributions	0.3	0.3	—	
Actuarial loss (gain)	53.3	12.8	2.8	(4.3)
Benefits and administrative expenses paid	(32.3)	(21.8)	(5.8)	(6.0)
Plan amendments	—		—	(17.0)
Plan settlements	(0.3)	(30.0)	—	
Curtailments	—		—	3.4
Currency translation	(5.1)	1.2		
Projected benefit obligations at end of year	\$ 533.2	\$ 491.1	\$ 80.3	\$ 80.1
Change in plan assets:				
Fair value of plan assets at beginning of year	\$ 383.6	\$ 379.3	\$ —	\$ —
Actual return on plan assets	63.0	25.0	—	
Employer contributions	23.4	30.2	5.8	6.0
Employee contributions	0.3	0.3	—	
Benefits and administrative expenses paid	(32.3)	(21.8)	(5.8)	(6.0)
Plan settlements	(0.3)	(30.0)	—	
Currency translation	(5.7)	0.6		
Fair value of plan assets at end of year	\$ 432.0	\$ 383.6	<u>\$                                    </u>	\$
Funded status at end of year	\$(101.2)	\$(107.5)	\$ (80.3)	\$ (80.1)
Amounts recognized on the combined balance sheet:				
Non-current assets	\$ 17.7	\$ 25.6	\$ —	\$ —
Current liabilities	(2.2)	(2.2)	(7.4)	(8.1)
Non-current liabilities	(116.7)	(130.9)	(72.9)	(72.0)
Net amount recognized on the combined balance sheet	\$(101.2)	\$(107.5)	\$ (80.3)	\$ (80.1)
Amounts recognized in accumulated other comprehensive income consist of:				
Net actuarial loss	\$(127.5)	\$(123.2)	\$ (12.1)	\$ (9.5)
Prior service (cost) credit	(1.8)	(2.5)	20.8	30.0
Net amount recognized in accumulated other comprehensive income	\$(129.3)	\$(125.7)	\$ 8.7	\$ 20.5

The estimated amounts that will be amortized from accumulated other comprehensive income into net periodic benefit cost (credit) in fiscal 2013 are as follows:

		Postretirement
(Dollars in Millions)	Pension Benefits	Benefits
Amortization of net actuarial loss	\$ 12.1	\$ 0.4
Amortization of prior service cost (credit)	0.6	(9.2)

The accumulated benefit obligation for all pension plans at the end of fiscal 2012 and 2011 was \$527.6 million and \$487.0 million, respectively.

Additional information related to pension plans is as follows:

(Dollars in Millions)	2012	2011
Pension plans with accumulated benefit obligations in excess of plan assets:		
Accumulated benefit obligation	\$414.3	\$386.3
Fair value of plan assets	295.4	253.3

The accumulated benefit obligation and fair value of plan assets for pension plans with projected benefit obligations in excess of plan assets do not significantly differ from the amounts in the table above since substantially all of the Company's pension plans are frozen.

#### Actuarial Assumptions

Weighted-average assumptions used to determine net periodic benefit cost for the Company's pension plans are as follows:

		U.S. Plans		Non-U.S. Plans		
	2012	2011	2010	2012	2011	2010
Discount rate	4.4%	4.9%	5.5%	5.2%	4.7%	6.2%
Expected return on plan assets	7.5%	7.6%	7.2%	4.0%	4.0%	4.1%
Rate of compensation increase	2.8%	2.8%	2.1%	3.7%	3.7%	3.1%

Weighted-average assumptions used to determine benefit obligations for the Company's pension plans are as follows:

		U.S. Plans			Non-U.S. Plans			
	2012	2011	2010	2012	2011	2010		
Discount rate	3.5%	4.4%	4.9%	4.0%	5.2%	4.7%		
Rate of compensation increase	— %	2.8%	2.8%	3.7%	3.7%	3.7%		

For the Company's U.S. plans, the discount rate is based on the market rate for a broad population of Moody's AA-rated corporate bonds over \$250 million. For the Company's non-U.S. plans, the discount rate is generally determined by reviewing country and region specific government and corporate bond interest rates.

In determining the expected return on pension plan assets, Covidien and the Company consider the relative weighting of plan assets by class and individual asset class performance expectations as provided by external advisors in reaching conclusions on appropriate assumptions. The investment strategy for the pension plans has been governed by Covidien. Covidien's overall investment objective is to obtain a long-term return on plan assets that is consistent with the level of investment risk that is considered appropriate. Investment risks and returns are reviewed regularly against benchmarks to ensure objectives are being met.

The weighted-average discount rate used to determine net periodic benefit cost and obligations for the Company's postretirement benefit plans are as follows:

	2012	2011	2010
Net periodic benefit cost	4.1%	4.6%	5.4%
Benefit obligations	3.2%	4.1%	4.6%

Healthcare cost trend assumptions for postretirement benefit plans are as follows:

	2012	2011
Healthcare cost trend rate assumed for next fiscal year	7.5%	7.8%
Rate to which the cost trend rate is assumed to decline	4.5%	4.5%
Fiscal year the ultimate trend rate is achieved	2029	2029

A one-percentage-point change in assumed healthcare cost trend rates would have the following effects:

(Dollars in Millions)	1-Percentage-Point Increase	1-Percentage-Point Decrease
Effect on total of service and interest cost	\$ 0.2	\$ (0.2)
Effect on postretirement benefit obligation	3.5	(3.5)

## Plan Assets

The Company's U.S. pension plans have a target allocation of either 59% equity securities and 41% debt securities or 33% equity securities and 67% debt securities, depending on the status and duration of liabilities of the plan. Various asset allocation strategies are in place for non-U.S. pension plans depending upon local law, status, funding level and duration of liabilities. The weighted-average target allocation for the Company's non-U.S. pension plans at the end of fiscal 2012 is as follows:

Equity securities	15%
Debt securities	80
Real estate	5
Total	100%

Pension plans have the following weighted-average asset allocations at the end of each fiscal year:

	<u>U.S.</u> P	U.S. Plans		Plans
	2012	2011	2012	2011
Equity securities	58%	56%	8%	10%
Debt securities	40	42	89	86
Cash and cash equivalents	1	1	_	—
Real estate and other	1	1	3	4
Total	100%	100%	100%	100%

The following tables provide a summary of plan assets held by the Company's pension plans that are measured at fair value on a recurring basis at the end of fiscal 2012 and 2011:

			surement	
(Dollars in Millions)	Total	Quoted Prices in Active Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)
Equity securities:				
U.S. small mid cap	\$ 24.0	\$ 24.0	\$	\$
U.S. large cap	101.2	101.2		—
International	66.8	57.2	9.6	
Debt securities:				
Diversified fixed income funds <sup>(1)</sup>	97.4	97.4	—	
High yield bonds	15.9	15.9	—	—
Emerging market funds	12.0	12.0	—	
Diversified/commingled funds	2.2	_	2.2	
Insurance contracts	105.1		—	105.1
Other	7.4	3.8	3.6	—
Total	\$432.0	\$ 311.5	\$ 15.4	\$ 105.1

		Basis of Fair Value Measurement			ent		
(Dollars in Millions)	Total	in Mar Identi	ed Prices Active kets for ical Assets evel 1)	Observ	cant Other able Inputs evel 2)	Unobsei	nificant vable Inputs evel 3)
Equity securities:							
U.S. small mid cap	\$ 19.6	\$	19.6	\$	—	\$	
U.S. large cap	87.4		87.4		—		—
International	55.9		44.6		11.3		
Debt securities:							
Diversified fixed income funds <sup>(1)</sup>	85.7		85.7				
High yield bonds	17.1		17.1				
Emerging market funds	10.0		10.0				_
Diversified/commingled funds	2.5				2.5		_
Insurance contracts	97.8						97.8
Other	7.6		3.0		4.6		
Total	\$383.6	\$	267.4	\$	18.4	\$	97.8

(1) Diversified fixed income funds consist of U.S. Treasury bonds, mortgage-backed securities, corporate bonds, asset-backed securities and U.S. agency bonds.

*Equity securities*—Equity securities primarily consist of mutual funds with underlying investments in foreign equity and domestic equity markets. The fair value of these investments is based on net asset value of the units held in the respective fund, which are determined by obtaining quoted prices on nationally recognized securities exchanges (level 1) or through net asset values provided by the fund administrators that can be corroborated by observable market data (level 2).

*Debt securities*—Debt securities are primarily invested in mutual funds with underlying fixed income investments in U.S. government and corporate debt, U.S. dollar denominated foreign government and corporate debt, asset-backed securities, mortgage-backed securities and U.S. agency bonds. The fair value of these investments is based on the net asset value of the units held in the respective fund which are determined by obtaining quoted prices on nationally recognized securities exchanges.

*Diversified/commingled funds*—Diversified/commingled funds held by the Company primarily consist of corporate debt securities and mutual funds invested in U.S. and non-U.S. equity securities. The fair value of these investments is determined using other inputs, such as net asset values provided by the fund administrators that can be corroborated by observable market data.

*Insurance contracts*—Insurance contracts held by the Company are issued by Delta Lloyd, a well-known, highly rated insurance company. The fair value of these insurance contracts is based upon the present value of future cash flows under the terms of the contracts and therefore the fair value of these assets has been classified as level 3 within the fair value hierarchy. Significant assumptions used in determining the fair value of these contracts are the amount and timing of future cash flows and counterparty credit risk. The objective of the insurance contracts is to provide the Company with future cash flows that will match the estimated timing and amount of future pension benefit payments. Delta Lloyd's insurance subsidiaries have a Standard & Poor's credit rating of A.

*Other*—Other includes cash and cash equivalents invested in a money market mutual fund, the fair value of which is determined by obtaining quoted prices on nationally recognized securities exchanges (level 1). In addition, other includes real estate funds, the fair value of which is determined using other inputs, such as net asset values provided by the fund administrators that can be corroborated by observable market data (level 2).

The following table provides a summary of the changes in the fair value measurements that used significant unobservable inputs (level 3) for fiscal 2011 and 2012:

(Dollars in Millions)	Insurance Contracts
Balance at September 24, 2010	\$ 76.6
Net unrealized gains	18.4
Net purchases, sales and issuances	2.6
Currency translation	0.2
Balance at September 30, 2011	97.8
Net unrealized gains	15.1
Net purchases, sales and issuances	(2.9
Currency translation	(4.9
Balance at September 28, 2012	\$ 105.1

Covidien shares are not a direct investment of the Company's pension funds; however, the pension funds may indirectly include Covidien shares. The aggregate amount of the Covidien shares are not material relative to the total pension fund assets.

#### **Contributions**

Covidien and the Company's funding policy is to make contributions in accordance with the laws and customs of the various countries in which the Company operates as well as to make discretionary voluntary contributions from time to time. The Company anticipates that Covidien will make contributions of \$42.8 million to the Company's defined benefit pension plans in fiscal 2013. In addition, the Company anticipates that Covidien will make contributions of \$7.5 million to the Company's postretirement benefit plans in fiscal 2013.

### **Expected Future Benefit Payments**

Benefit payments expected to be paid, reflecting future expected service as appropriate, are as follows:

(Dollars in Millions)	Pension Benefits	Postretirement Benefits
Fiscal 2013	\$ 45.2	\$ 7.5
Fiscal 2014	34.4	7.1
Fiscal 2015	33.9	6.7
Fiscal 2016	33.4	6.4
Fiscal 2017	32.7	6.0
Fiscal 2018-2022	153.1	25.1

### **Defined Contribution Retirement Plans**

The Company maintains, through Covidien, one active tax-qualified 401(k) retirement plan in the U.S., which provides for an automatic Company contribution of three percent of an eligible employee's pay. The Company also makes a matching contribution generally equal to 50% of each employee's elective contribution to the plan up to six percent of the employee's eligible pay. Total 401(k) expense related to continuing operations was \$20.9 million, \$19.3 million and \$18.4 million for fiscal 2012, 2011 and 2010, respectively.

#### **Deferred** Compensation Plans

As discussed in note 15, the Company maintains, through Covidien, one active non-qualified deferred compensation plan in the U.S., which permits eligible employees to defer a portion of their compensation. Deferred compensation expense for each period presented was insignificant.

### Rabbi Trusts and Other Investments

The Company maintains several rabbi trusts, the assets of which are used to pay retirement benefits. The rabbi trust assets are subject to the claims of the Company's creditors in the event of the Company's insolvency. Plan participants are general creditors of the Company with respect to these benefits. The trusts primarily hold life insurance policies and debt and equity securities, the value of which is included in other assets on the combined balance sheets. Note 15 provides additional information regarding the debt and equity securities. The carrying value of the 70 life insurance contracts held by these trusts was \$37.8 million and \$36.4 million at September 28, 2012 and September 30, 2011, respectively. These contracts have a total death benefit of \$93.9 million and \$94.2 million at September 28, 2012 and September 30, 2011, respectively. However, there are outstanding loans against the policies amounting to \$16.9 million and \$16.3 million at September 28, 2012 and September 30, 2011, respectively.

Covidien has insurance contracts which serve as collateral for certain of the Company's non-U.S. pension plan benefits, \$9.8 million and \$10.2 million of which have been allocated to the Company at September 28, 2012 and September 30, 2011, respectively. These amounts were also included in other assets on the combined balance sheets.

### 17. Share Plans

Compensation costs related to share-based transactions are recognized in the combined financial statements based on fair value. Total equity-based compensation cost related to continuing operations for fiscal 2012 and 2011 was \$11.1 million and \$10.6 million, respectively, and has been included in selling, general and administrative expenses. Total equity-based compensation for fiscal 2010 was \$14.1 million, of which \$12.6 million related to continuing operations and was included in selling, general and administrative expenses. The Company recognized a related tax benefit associated with this expense of \$3.8 million, \$3.4 million and \$4.8 million during fiscal 2012, 2011 and 2010, respectively.

## Stock Compensation Plans

As of September 28, 2012, all equity awards held by employees of the Company were granted under Covidien's amended and restated 2007 Stock and Incentive Plan or predecessor plans. The following disclosures represent the Company's portion of such plans.

Share options—Options are granted to purchase Covidien ordinary shares at prices that are equal to the fair market value of the shares on the date the option is granted. Options generally vest in equal annual installments over a period of four years and expire 10 years after the date of grant. The grant-date fair value of options, adjusted for estimated forfeitures, is recognized as expense on a straight-line basis over the requisite service period, which is generally the vesting period. Forfeitures are estimated based on historical experience.

Option activity and information is as follows:

	Shares	Weighted- Average Exercise Price	Weighted- Average Remaining Contractual Term (in years)	Aggregate Intrinsic Value (dollars in millions)
Outstanding at September 30, 2011	2,012,713	\$ 40.79		
Granted	722,720	47.00		
Exercised	(576,616)	38.87		
Expired/Forfeited	(135,241)	43.83		
Outstanding at September 28, 2012	2,023,576	43.35	7.20	\$ 32.5
Vested and unvested expected to vest as of September 28, 2012	1,876,242	43.15	7.08	30.5
Exercisable at September 28, 2012	680,731	39.91	4.80	13.3

As of September 28, 2012, there was \$7.7 million of total unrecognized compensation cost related to unvested Covidien options, which is expected to be recognized over a weighted-average period of 1.4 years.

The grant date fair value of options has been estimated using the Black-Scholes pricing model. Use of a valuation model requires management to make certain assumptions with respect to selected model inputs. The expected volatility assumption is based on the historical and implied volatility of Covidien's peer group with similar business models. The expected life assumption is based on the contractual and vesting term of the option, employee exercise patterns and employee post-vesting termination behavior. The expected annual dividend per share is based on Covidien's dividend rate on the date of grant. The risk-free interest rate is based on U.S. Treasury zero-coupon issues with a remaining term equal to the expected life assumed at the date of grant. The weighted-average assumptions used in the Black-Scholes pricing model for options granted during each year, along with the weighted-average grant-date fair values, were as follows:

	2012	2011	2010
Expected stock price volatility	27.9%	27.0%	27.0%
Risk-free interest rate	1.18%	1.79%	2.29%
Expected annual dividend per share	\$ 0.90	\$0.80	\$ 0.72
Expected life of options (years)	5.6	5.3	5.4
Fair value per option	\$10.27	\$9.46	\$11.46

The total intrinsic value of options exercised during fiscal 2012, 2011 and 2010 was \$8.5 million, \$11.1 million and \$5.3 million, respectively. The related tax benefit for fiscal 2012, 2011 and 2010 was \$3.0 million, \$3.5 million, respectively.

*Restricted stock units*—Recipients of restricted stock units ("RSUs") have no voting rights and receive dividend equivalent units which vest upon the vesting of the related shares. RSUs generally vest in equal annual installments over a four-year period. Restrictions on RSUs lapse upon normal retirement, death or disability of the employee. The grant-date fair value of RSUs, adjusted for estimated forfeitures, is recognized as expense on a straight-line basis over the service period. The fair market value of RSUs is determined based on the market value of Covidien's shares on the date of grant.

RSU activity is as follows:

	Shares	Ğr	ted-Average ant-Date ir Value
Non-vested at September 30, 2011	162,760	\$	42.85
Granted	167,086		47.35
Vested	(55,496)		42.03
Forfeited	(21,334)		44.87
Non-vested at September 28, 2012	253,016		45.83

The weighted-average grant-date fair value of Covidien RSUs granted to employees of the Company during fiscal 2012, 2011 and 2010 was \$47.35, \$43.85 and \$47.05, respectively. The total fair value of RSUs vested for employees of the Company during fiscal 2012, 2011 and 2010 was \$2.6 million, \$5.8 million and \$6.8 million, respectively. The related tax benefit for fiscal 2012, 2011 and 2010 was \$0.9 million, \$2.0 million and \$2.4 million, respectively. As of September 28, 2012, there was \$6.6 million of unrecognized compensation cost related to RSUs, which is expected to be recognized over a weighted-average period of 1.4 years.

*Performance share units*—Similar to recipients of RSUs, recipients of performance share units ("PSUs") have no voting rights and receive dividend equivalent units which vest upon the vesting of the related shares. The grant-date fair value of PSUs, adjusted for estimated forfeitures, is generally recognized as expense on a straight-line basis from the grant date through the end of the performance period. The vesting of PSUs is generally based on relative total shareholder return (total shareholder return for Covidien as compared to total shareholder return of a healthcare industry index), measured over a three-year performance period. The healthcare industry index is comprised of many healthcare companies which replicate Covidien's mix of businesses. Depending on Covidien's relative performance during the performance period, a recipient of the award is entitled to receive a number of ordinary shares equal to a percentage, ranging from 0% to 200%, of the award granted.

PSU activity is as follows<sup>(1)</sup>:

	Shares	Ğr	ted-Average ant-Date ir Value
Non-vested at September 30, 2011	193,521	\$	54.52
Granted	21,803		61.85
Performance metric adjustment <sup>(2)</sup>	(4,003)		42.20
Vested	(65,957)		42.65
Forfeited	(12,771)		59.59
Non-vested at September 28, 2012 <sup>(3)</sup>	132,593		61.52

<sup>(1)</sup> The number of shares disclosed in this table are at the target number of 100%.

(2) Represents the adjustment to awards granted in fiscal 2009 for the three-year performance cycle award period ended September 30, 2011, based on the actual total shareholder return achievement of 94%.

(3) Approximately 100,000 shares of Covidien were earned for awards that were granted in fiscal 2010 for the three-year performance cycle award period ended September 28, 2012, based on the actual total shareholder return achievement of 200%.

The Monte Carlo model was used to estimate the probability of satisfying the performance criteria and the resulting fair value of PSU awards. The assumptions used in the Monte Carlo model for PSUs granted during each year were as follows:

	2012	2011	2010
Covidien expected stock price volatility	28.7%	31.4%	30.2%
Covidien peer group stock price volatility	29.1%	33.3%	32.5%
Correlation of returns	47.5%	49.7%	47.3%

The weighted-average grant-date fair value per share of PSUs granted to employees of the Company during fiscal 2012, 2011 and 2010 was \$61.85, \$58.05 and \$62.53, respectively. The total fair value of PSUs vested during fiscal 2012 was \$2.9 million and the related tax benefit was \$1.0 million. The total fair value of PSUs vested and related tax benefit during fiscal 2011 and 2010 was not significant. As of September 28, 2012, there was \$1.6 million of unrecognized compensation cost related to PSUs, which is expected to be recognized over a weighted-average period of 0.8 years.

### **Employee Stock Purchase Plans**

Substantially all full-time employees of the Company's U.S. subsidiaries and employees of certain qualified non-U.S. subsidiaries are eligible to participate in Covidien's employee stock purchase plan. Eligible employees authorize payroll deductions to be made for the purchase of shares. Covidien matches the first \$25,000 of an employee's contribution by contributing an additional 15% of the employee's payroll deduction. All shares purchased under the plan are purchased on the open market by a designated broker.

Covidien also maintains a Savings Related Share Plan for the benefit of employees of certain qualified non-U.S. subsidiaries in the United Kingdom. The terms of this plan provide for Covidien to grant to certain employees the right to purchase shares at a stated price and receive certain tax benefits. Under this plan, eligible Company employees in the United Kingdom are granted options to purchase shares of Covidien at the end of a three-year period at 85% of the fair market value of a Company share on the day before the date such employees were invited to apply for the grant of options. Options under the plan are generally exercisable after a period of three years from the invitation date and expire six months after the date of vesting. Compensation cost related to options granted under this plan was insignificant during fiscal 2012, 2011 and 2010.

Impact of the separation—Prior to completion of the separation from Covidien, the board of directors of Mallinckrodt plc is expected to adopt, with the approval of the current shareholders of Mallinckrodt plc, stock incentive plans, which provide for future awards to Company employees. In connection with the separation from Covidien, PSUs are expected to be converted into RSUs based on performance achieved on or about the distribution date. In addition, upon separation from Covidien, all outstanding equity awards held by active employees of the Company are expected to be converted into like-kind equity awards of the Company. Such equity awards will be converted at equivalent value determined using the intrinsic value method. The original vesting provisions will remain in effect for all equity awards.

### 18. Accumulated Other Comprehensive Income

The components of accumulated other comprehensive income are as follows:

(Dollars in Millions)	Currency Translation	Unrecognized (Loss) Gain on Benefit Plans	Accumulated Other Comprehensive Income
Balance at September 25, 2009	\$ 169.3	\$ (66.9)	\$ 102.4
Pre-tax change	(8.8)	(10.6)	(19.4)
Income tax benefit		3.6	3.6
Balance at September 24, 2010	160.5	(73.9)	86.6
Pre-tax change	(0.5)	16.9	16.4
Income tax provision		(4.5)	(4.5)
Balance at September 30, 2011	160.0	(61.5)	98.5
Pre-tax change	(2.9)	(15.3)	(18.2)
Income tax benefit		4.6	4.6
Balance at September 28, 2012	<u>\$ 157.1</u>	\$ (72.2)	\$ 84.9

## 19. Leases

The Company has facility, vehicle and equipment leases that expire at various dates. Rental expense under facility, vehicle and equipment operating leases related to continuing operations, a portion of which has been allocated to the Company, was \$15.5 million, \$14.4 million and \$16.5 million for fiscal 2012, 2011 and 2010, respectively. The Company also has facility and equipment commitments under capital leases.

The following is a schedule of minimum lease payments for non-cancelable leases as of September 28, 2012:

(Dollars in Millions)	Operating Leases	Capital Leases
Fiscal 2013	\$ 11.3	\$ 1.4
Fiscal 2014	11.3	1.4
Fiscal 2015	6.9	1.4
Fiscal 2016	6.3	0.4
Fiscal 2017	5.8	
Thereafter	12.7	
Total minimum lease payments	\$ 54.3	4.6
Less interest portion of payments		(0.2)
Present value of minimum lease payments		\$ 4.4

The Company exchanged title to \$11.3 million of its plant assets in return for an equal amount of Industrial Revenue Bonds (IRB) issued by the St. Louis County. The Company also simultaneously leased such assets back from the County under a capital lease expiring December 2022, the terms of which provide the Company with the right of offset against the IRBs. The lease also provides an option for the Company to repurchase the assets at the end of the lease for nominal consideration. These transactions collectively result in a property tax abatement 10 years from the date the property is placed in service. Due to right of offset, the capital lease obligation and IRB asset are recorded net in the combined balance sheets.

#### 20. Commitments and Contingencies

The Company has purchase obligations related to commitments to purchase certain goods and services. At September 28, 2012, such obligations were as follows:

(Dollars in Millions)	
Fiscal 2013	\$70.1
Fiscal 2014	24.6
Fiscal 2015	21.2
Fiscal 2016	21.2
Fiscal 2017	—

The Company is subject to various legal proceedings and claims, including patent infringement claims, product liability matters, environmental matters, employment disputes, contractual disputes and other commercial disputes, including those described below. Management believes that these legal proceedings and claims likely will be resolved over an extended period of time. Although it is not feasible to predict the outcome of these matters, management is of the opinion that their ultimate resolution should not have a material adverse effect on the Company's financial position, results of operations and cash flows.

#### **Governmental Proceedings**

On January 7, 2009, the Company received a subpoent from the U.S. Attorney's Office for the Northern District of California requesting production of documents relating to the sales and marketing of its Tofranil-PM, Restoril and Magnacet products. The Company is complying as required by the terms of the subpoent. The Company believes that the amount accrued related to this matter is adequate, the amount of which is not significant.

#### Patent/Antitrust Litigation

*Tyco Healthcare Group LP, et al. v. Mutual Pharmaceutical Company, Inc.* The Company filed a patent infringement suit in the U.S. District Court for the District of New Jersey against Mutual Pharmaceutical Co., Inc., et al. (collectively, "Mutual") on March 20, 2007 pursuant to procedures set out in the Drug Price Competition and Patent Term Restoration Act of 1984, after Mutual submitted an Abbreviated New Drug Application ("ANDA") to the FDA seeking to sell a generic version of the Company's 7.5 mg Restoril sleep aid product. Mutual also filed antitrust and unfair competition counterclaims. The patents at issue have since expired or been found invalid. On January 18, 2013, the trial court issued an opinion and order granting the Company's motion for summary judgment regarding Mutual's antitrust and unfair competition counterclaims. Mutual has the right to appeal this decision. While it is not possible at this time to determine with certainty the ultimate outcome of the counterclaims, the Company believes that the final resolution of the claims will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

In addition, the Company was involved in patent infringement litigation involving two patents owned by the Company. During fiscal 2010, the counterparty agreed to pay the Company \$19.3 million in exchange for the Company's release of all claims associated with the two patents, of which \$15.0 million was received in fiscal 2010 and the remainder in fiscal 2011. The settlement amount was allocated to both past and future royalties through 2014. Accordingly, during fiscal 2012, 2011 and 2010, the Company recorded income of \$1.8 million, \$1.8 million and \$12.0 million, respectively, related to this settlement.

#### **Pricing Litigation**

Two cases are pending against the Company that allege generally that the Company and numerous other pharmaceuticals companies reported false pricing information in connection with certain drugs that are reimbursable under Medicaid, resulting in overpayment by state Medicaid programs for those drugs. These cases,

brought by state Attorneys General in Utah and Louisiana, generally seek monetary damages and attorneys' fees. The Company is named as a defendant in *State* of Utah v. Actavis US, Inc., et al. filed May 8, 2008, which is pending in the Third Judicial Circuit of Salt Lake County, Utah and in *State of Louisiana v. Abbott* Laboratories Inc., et al. filed November 3, 2010, which is pending in the 19th Judicial District, Parish of East Baton Rouge, Louisiana. The Company intends to contest these cases and to explore other options as appropriate. The Company believes that the amount accrued related to these cases, the amount of which is not significant, is adequate.

### **Commercial Litigation**

During fiscal 2012, the Company recorded a legal charge of \$4.3 million to settle a longstanding dispute with General Electric Company ("GE"), which is included in selling, general and administrative expenses. GE had alleged breach of a manufacturing and supply agreement claiming that the Company failed to manufacture and supply the imaging agent Optison<sup>TM</sup> at certain times.

## Environmental Remediation and Litigation Proceedings

The Company is involved in various stages of investigation and cleanup related to environmental remediation matters at a number of sites, including those described below. The ultimate cost of site cleanup and timing of future cash outlays is difficult to predict, given the uncertainties regarding the extent of the required cleanup, the interpretation of applicable laws and regulations, and alternative cleanup methods. The Company concluded that, as of September 28, 2012, it was probable that it would incur remedial costs in the range of \$151.5 million to \$264.9 million. The Company concluded that, as of September 28, 2012, the best estimate within this range was \$151.5 million, of which \$15.0 million was included in accrued and other current liabilities and the remainder was included in environmental liabilities on our combined balance sheet at September 28, 2012.

*Orrington, Maine*—The Company is a successor to a company which owned and operated a chemical manufacturing facility in Orrington, Maine from 1967 until 1982. The Company is currently responsible for the costs of completing an environmental site investigation required by the U.S. Environmental Protection Agency ("EPA") and the Maine Department of Environmental Protection ("MDEP"). Based on the site investigation, the Company submitted a Corrective Measures Study plan and identified a preferred alternative which was submitted to the EPA and MDEP for approval in 2004. MDEP disagreed with the proposed alternative and served a compliance order on the Company and United States Surgical Corporation, a subsidiary of Covidien, in December 2008. The compliance order included a directive to remove a significant volume of soils at the site. On December 19, 2008, the Company filed an appeal with the Maine Board of Environmental Protection ("Maine Board") to challenge the terms of the compliance order. A hearing before the Maine Board began on January 25, 2010 and concluded on February 4, 2010. On August 19, 2010, the Maine Board modified the MDEP order and issued a final order requiring removal of two landfills, capping of the remaining three landfills, installation of a groundwater extraction system and long-term monitoring of the site and the three remaining landfills.

On September 17, 2010, the Company appealed the final order issued by the Maine Board in Maine Superior Court. On appeal, the Company has requested that the Superior Court invalidate the Maine Board's final order in its entirety or, in the alternative, reverse or modify the final order to eliminate the requirements that it remove one of the two landfills and recap the remaining three landfills. The Company also appealed certain administrative requirements of the final order. On November 1, 2012, the Superior Court affirmed the Maine Board's final order. The Company has appealed the Superior Court's decision to the Maine Supreme Judicial Court. The Company has assessed the status of this matter and has concluded that it is more likely than not that the Maine Board's final order will be either invalidated, reversed or modified, and, further, intends to vigorously pursue all available means to achieve such result.

The Company estimates that, as of September 28, 2012, the cost to comply with the proposed remediation alternatives at this site ranges from \$95.8 million to \$170.3 million. However, there are still significant

uncertainties in the outcome of the pending litigation, and the Company continues to disagree with the level of remediation outlined in the Maine Board's final order. At September 28, 2012, estimated future investigation and remediation costs of \$95.8 million were accrued for this site.

Penobscot River and Bay—Since April 2000, the Company has also been involved in a lawsuit, Maine People's Alliance and Natural Resources Defense Council, Inc. v. HoltraChem Manufacturing Company, LLC and Mallinckrodt US LLC, filed in the U.S. District Court for the District of Maine by the Natural Resources Defense Council and the Maine People's Alliance. Plaintiffs sought an injunction requiring the Company to conduct extensive studies of mercury contamination of the Penobscot River and Bay and options for remediating such contamination, and to perform appropriate remedial activities, if necessary.

On July 29, 2002, following a March 2002 trial, the District Court entered an opinion and order which held that conditions in the Penobscot River and Bay may pose an imminent and substantial endangerment and that the Company was liable for the cost of performing a study of the river and bay. The District Court subsequently appointed a study panel to oversee the study and ordered the Company to pay costs associated with the study. The study panel conducted Phase I studies and proposed a Phase II study which was approved by the District Court. The Phase II study calls for several additional years of field work, followed by an additional year for data synthesis. The Company has accrued for the cost of the studies as estimated by the study panel; however, due to the uncertainties involved pending completion of the study panel's work, it is not possible to estimate the costs, if any, which might result from an order to conduct remediation in the Penobscot River and Bay. Accordingly, costs of any such remediation are not included in the range of estimated aggregate environmental remediation costs.

The entity with liability for the investigation and remediation described under "Orrington, Maine" and "Penobscot River and Bay" will not be transferred to Mallinckrodt plc as part of the separation. Accordingly, this will be a liability of a Covidien entity following the separation.

*Crab Orchard National Wildlife Refuge Superfund Site, near Marion, Illinois*—The Company is a successor in interest to International Minerals and Chemicals Corporation ("IMC"). Between 1967 and 1982, IMC leased portions of the AUS Operable Unit at the Crab Orchard Superfund Site (the "Site") from the government and manufactured various explosives for use in mining and other operations. In March 2002, the Department of Justice ("DOJ"), the U.S. Department of the Interior and the EPA (together, the "Government Agencies") issued a special notice letter to General Dynamics Ordnance and Tactical Systems, Inc. ("General Dynamics"), one of the other potentially responsible parties ("PRPs") at the Site, to compel General Dynamics to perform the remedial investigation and feasibility study ("RI/FS") for the AUS Operable Unit. General Dynamics negotiated an Administrative Order on Consent with the Government Agencies to conduct an extensive RI/FS at the Site under the direction of the U.S. Fish and Wildlife Service. General Dynamics asserted in August 2004 that the Company is jointly and severally liable, along with approximately eight other lessees and operators at the AUS Operable Unit, for alleged contamination of soils and groundwater resulting from historic operations and has threatened to file a contribution claim against the Company and other pRPs who received demand letters from General Dynamics before the initiation of formal PRP negotiations to address resolution of these alleged claims. While it is not possible at this time to determine with certainty the ultimate outcome of this case, the Company believes that the final resolution of all known claims, after taking into account amounts already accrued, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

Mallinckrodt Veterinary, Inc., Millsboro, Delaware—The Company previously operated a plant in Millsboro, Delaware (the "Millsboro Site") that manufactured various animal healthcare products. In 2005, the Delaware Department of Natural Resources and Environmental Control found trichloroethylene ("TCE") in the Millsboro public water supply at levels that exceeded the federal drinking water standards. Further investigation to identify the TCE plume in the ground water indicated that the plume has extended to property owned by a

third party near the Millsboro Site. The Company, and other former owners assumed responsibility for the Millsboro Site cleanup under the Alternative Superfund Program administered by the EPA. The Company and other PRPs entered into an Administrative Order on Consent with the EPA on May 10, 2010 which was subsequently amended in November 2010 and January 2011 to investigate the potential source of TCE contamination and to evaluate options to abate, mitigate and/or eliminate the release or threat of release of hazardous substances at the Millsboro Site. The Company, along with other parties, continues to conduct the studies and prepare remediation plans in accordance with the amended Administrative Order on Consent. While it is not possible at this time to determine with certainty the ultimate outcome of this case, the Company believes that the final resolution of all known claims, after taking into account amounts already accrued, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

*Coldwater Creek, St. Louis County, Missouri*—The Company is one of several companies named as defendants in three tort complaints (*McClurg, et al. v. MI Holdings, Inc., et al.*, filed April 10, 2012 and *Steinmann et al. v. MI Holdings, Inc., et al.*, filed October 23, 2012) with numerous plaintiffs pending in the U.S. District Court for the Eastern District of Missouri. These cases allege personal injury for alleged exposure to radiological substances present in Coldwater Creek in Missouri. Plaintiffs lived in various locations in St. Louis County, Missouri near Coldwater Creek. Radiological residues which may have been present in the creek have been remediated by the U.S. Army Corps of Engineers. The Company believes that it has meritorious defenses to these complaints and is vigorously defending against them. The Company is unable to estimate a range of reasonably possible losses for the following reasons: (i) the proceedings are in early stages; (ii) the Company has not received and reviewed complete information regarding the plaintiffs and their medical conditions; and (iii) there are significant factual issues to be resolved. While it is not possible at this time to determine with certainty the ultimate outcome of these cases, the Company believes that the final resolution of all known claims will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

The Company has also recorded asset retirement obligations for the estimated future costs primarily associated with legal obligations to decommission two facilities within the Global Medical Imaging segment. Substantially all of these obligations are included in other liabilities on the combined balance sheets. The following table provides a summary of the changes in the Company's asset retirement obligations:

(Dollars in Millions)	2012	2011
Balance at beginning of period	2012 \$45.9	<u>2011</u> \$ 73.9
Accretion expense	2.5	4.3
Revisions in estimated cash flows	_	(32.2)
Currency translation and other	(2.0)	(0.1)
Balance at end of period	\$46.4	\$ 45.9

The Company believes that any potential payment of such estimated amounts will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

#### Supply Agreement

During fiscal 2010, the Company amended an existing supply agreement. In accordance with the amendment, the Company will receive \$6.1 million over a four-year period in exchange for decreasing the purchase requirements under the supply agreement. As a result of this contract amendment, the Company recorded a \$5.5 million gain during fiscal 2010, which was included in selling, general and administrative expenses.

### **Products Liability Litigation**

The Company is one of four manufacturers of Gadolinium-Based Contrast Agents, such as our Optimark product, involved in litigation alleging that administration of these agents causes development of nephrogenic

systemic fibrosis in a small number of patients with advanced renal impairment. The complaints generally allege design and manufacturing defects, failure to warn, breach of warranty, fraud and violations of various state consumer protection laws. The litigation includes a federal multi-district litigation in the U.S. District Court for the Northern District of Ohio (*In re Gadolinium-Based Contrast Agents Product Liability Litigation*, which was established on February 27, 2008) and cases in various state courts. The Company believes that it has meritorious defenses to these complaints and is defending against them. When appropriate, the Company settles cases. As of January 31, 2013, there were four remaining cases in which the plaintiffs have either documented or specifically alleged use of the Company's Optimark product. The Company believes that it has adequate amounts recorded related to these matters. While it is not possible at this time to determine with certainty the ultimate outcome of these cases, the Company believes that the final resolution of all known claims, after taking into account amounts already accrued, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

Beginning with lawsuits brought in July 1976, the Company is also named as a defendant in personal injury lawsuits based on alleged exposure to asbestoscontaining materials. A majority of the cases involve product liability claims, based principally on allegations of past distribution of products containing asbestos. A limited number of the cases allege premises liability, based on claims that individuals were exposed to asbestos while on the Company's property. Each case typically names dozens of corporate defendants in addition to the Company. The complaints generally seek monetary damages for personal injury or bodily injury resulting from alleged exposure to products containing asbestos. The Company's involvement in asbestos cases has been limited because it did not mine or produce asbestos. Furthermore, in the Company's experience, a large percentage of these claims have never been substantiated and have been dismissed by the courts. The Company has not suffered an adverse verdict in a trial court proceeding related to asbestos claims, and intends to continue to defend these lawsuits. When appropriate, the Company settles claims; however, amounts paid to settle and defend all asbestos claims have been immaterial. As of January 31, 2013, there were approximately 11,600 asbestos-related cases pending against the Company.

The Company estimates pending asbestos claims and claims that were incurred but not reported, as well as insurance recoveries. The Company's estimate of its liability for pending and future claims is based on claims experience over the past five years and covers claims either currently filed or expected to be filed over the next seven years. The Company believes that it has adequate amounts recorded related to these matters. While it is not possible at this time to determine with certainty the ultimate outcome of these asbestos-related proceedings, the Company believes that the final outcome of all known and anticipated future claims, after taking into account amounts already accrued and insurance coverage, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

#### **Other Matters**

The Company is a defendant in a number of other pending legal proceedings incidental to present and former operations, acquisitions and dispositions. The Company does not expect the outcome of these proceedings, either individually or in the aggregate, to have a material adverse effect on the Company's financial condition, results of operations and cash flows.

#### 21. Segment and Geographic Data

The Company is engaged in the development, manufacture and distribution of pharmaceuticals and diagnostic imaging agents. The Company manages and operates its business through the following two segments:

- Specialty Pharmaceuticals produces and markets branded and generic pharmaceuticals and active pharmaceutical ingredients ("API"), comprised of medicinal opioids, synthetic controlled substances, acetaminophen and other active ingredients;
- Global Medical Imaging develops, manufactures and markets contrast media and delivery systems and radiopharmaceuticals (nuclear medicine).

Management measures and evaluates the Company's operating segments based on segment net sales and operating income. Management excludes corporate expenses from segment operating income. In addition, certain amounts that management considers to be non-recurring or non-operational are excluded from segment operating income because management evaluates the operating results of the segments excluding such items. These items include revenues and expenses associated with related party sales of products to other Covidien businesses, intangible asset amortization, net restructuring and related charges, and separation costs. Although these amounts are excluded from segment operating income, as applicable, they are included in reported combined operating income and in the reconciliations presented below. Selected information by business segment is as follows:

(Dollars in Millions)	2012	2011	2010
Net sales <sup>(1)</sup> :			
Specialty Pharmaceuticals	\$1,005.2	\$ 909.4	\$ 869.0
Global Medical Imaging	996.8	1,060.0	1,128.1
Net sales of operating segments	2,002.0	1,969.4	1,997.1
Net sales to related parties <sup>(2)</sup>	54.2	52.4	50.5
Net sales	\$2,056.2	\$2,021.8	\$2,047.6
Operating income:			
Specialty Pharmaceuticals	\$ 162.8	\$ 121.5	\$ 139.6
Global Medical Imaging	214.3	232.4	221.5
Segment operating income	377.1	353.9	361.1
Unallocated amounts:			
Corporate and allocated expenses <sup>(3)</sup>	(69.9)	(73.3)	(85.8)
Intangible asset amortization	(27.3)	(27.0)	(23.4)
Restructuring and related charges, net	(19.2)	(10.0)	(11.5)
Separation costs	(25.5)	(2.9)	
Operating income	\$ 235.2	\$ 240.7	\$ 240.4

<sup>(1)</sup> Amounts represent sales to external customers. There are no intersegment sales.

<sup>(2)</sup> Represents products that were sold to other Covidien businesses, which is discussed in note 11.

<sup>(3)</sup> Includes administration expenses and certain compensation, environmental and other costs not charged to the Company's operating segments.

(Dollars in Millions) Depreciation and amortization <sup>(4)</sup> :	2012	2011	2010
Specialty Pharmaceuticals	\$ 88.7	\$ 77.5	\$ 68.2
Global Medical Imaging	42.2	42.3	46.0
Depreciation and amortization	\$130.9	\$119.8	\$114.2

<sup>(4)</sup> Depreciation for certain shared facilities is allocated based on occupancy percentage.

Net sales by products within the Company's segments are as follows:

(Dollars in Millions)	2012	2011	2010
Generic Pharmaceuticals and API	\$ 848.8	\$ 824.7	\$ 781.8
Brands Pharmaceuticals	156.4	84.7	87.2
Specialty Pharmaceuticals	1,005.2	909.4	869.0
Contrast Media and Delivery Systems	542.0	595.5	609.1
Nuclear Imaging	454.8	464.5	519.0
Global Medical Imaging	996.8	1,060.0	1,128.1
Net sales of operating segments	2,002.0	1,969.4	1,997.1
Net sales to related parties <sup>(1)</sup>	54.2	52.4	50.5
Net sales	\$2,056.2	\$2,021.8	\$2,047.6

<sup>(1)</sup> Represents products that were sold to other Covidien businesses, which is discussed in note 11.

Selected information by geographic area is as follows:

(Dollars in Millions)	2012	2011	2010
Net sales <sup>(2)</sup> :			
United States	\$1,350.2	\$1,293.8	\$1,380.5
Europe, Middle East and Africa	411.0	419.7	393.8
Other	295.0	308.3	273.3
	\$2,056.2	\$2,021.8	\$2,047.6
Long-lived assets <sup>(3)</sup> :			
United States	\$ 847.7	\$ 802.0	\$ 802.9
Europe, Middle East and Africa (including \$45.5, \$48.9 and \$49.6 in Ireland)	72.2	81.3	85.6
Other	52.1	48.1	50.6
	\$ 972.0	\$ 931.4	\$ 939.1

<sup>(2)</sup> Net sales are attributed to regions based on the location of the entity that records the transaction, none of which relate to the country of Ireland.

<sup>(3)</sup> Long-lived assets are primarily composed of property, plant and equipment.

## 22. Subsequent Events

Subsequent events have been evaluated for adjustment through November 15, 2012, the date at which the parent's consolidated financial statements were completed and issued, and February 1, 2013, for purposes of evaluating disclosures in these combined financial statements.

On October 1, 2012, the Company's Specialty Pharmaceuticals segment acquired all the outstanding equity of CNS Therapeutics, Inc. ("CNS Therapeutics"), a specialty pharmaceuticals company focused on developing and commercializing products for site-specific administration to the central nervous system to treat neurological disorders and intractable chronic pain, for total consideration of \$95.0 million. The total consideration was comprised of an upfront cash payment of \$88.1 million (net of cash acquired of \$3.6 million) and the fair value of contingent consideration of \$6.9 million. This contingent consideration, which could potentially total a maximum of \$9.0 million, is primarily based on whether the FDA approves another dosage form of Gablofen on or before December 31, 2016. The acquisition of CNS Therapeutics expanded the Company's branded pharmaceuticals portfolio and supports the Company's strategy of leveraging its therapeutic expertise and core capabilities in manufacturing, regulatory and commercialization to serve patients.

The following amounts represent the preliminary estimate of the fair value of the identifiable assets acquired and liabilities assumed:

(Dollars in millions)	
Current assets <sup>(1)</sup>	\$ 13.6
Intangible assets	91.9
Goodwill (non-tax deductible)	24.3
Total assets acquired	129.8
Current liabilities	4.0
Deferred tax liabilities (non-current)	27.2
Contingent consideration (non-current)	6.9
Total liabilities assumed	38.1
Net assets acquired	<u>\$ 91.7</u>

(1) This amount includes \$3.3 million of accounts receivable, which is also the gross contractual value. As of the acquisition date, the fair value of accounts receivable approximated carrying value.

Intangible assets acquired consist of the following:

		Weighted-Average
(Dollars in Millions)	Amount	Amortization Period
Completed technology	\$ 73.1	13 years
Trademark	0.2	3 years
In-process research and development	18.6	Non-Amortizable
	\$ 91.9	

The in-process research and development projects primarily relate to three intrathecal pain products. As of the date of acquisition, these pain products were in various stages of development. Development, testing, clinical trials and regulatory submission are required in order to bring these products to market. The Company estimates that the total costs to complete these products will be approximately \$18.0 million. In addition, the Company expects that regulatory approvals will occur between 2015 and 2018. The Company determined the valuation of in-process research and development using, among other factors, appraisals. The value was primarily based on the discounted cash flow method and was discounted at a 35% rate, which was considered commensurate with the risks and stages of development of the pain products. Future residual cash flows that could be generated from the products were determined based upon management's estimate of future revenue and expected profitability of the products. These projected cash flows were then discounted to their present values taking into account management's estimate of future expenses that would be necessary to bring the products to completion.

The Company has not yet finalized its deferred tax assets and liabilities for the CNS Therapeutics acquisition, the impact of which is not expected to have a material effect on the Company's financial condition.

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC CONDENSED COMBINED STATEMENTS OF INCOME (UNAUDITED) Six Months Ended March 29, 2013 and March 30, 2012 (in millions)

	Six Montl	Six Months Ended	
	March 29, 2013	March 30, 2012	
Net sales (including sales to a related party of \$25.9 and \$27.9)	\$1,089.3	\$1,026.8	
Cost of sales (including purchases from a related party of \$22.0 and \$17.0)	582.3	538.5	
Gross profit	507.0	488.3	
Selling, general and administrative expenses	307.5	274.1	
Research and development expenses	77.6	72.3	
Separation costs	26.4	10.2	
Restructuring charges, net	6.6	5.5	
Gain on divestiture	(1.4)	(1.4)	
Operating income	90.3	127.6	
Other income, net (including royalties from a related party of \$0.5 and \$0.5)	0.2	0.7	
Interest expense	(0.2)	(0.3)	
Interest income	0.1	0.3	
Income from continuing operations before income taxes	90.4	128.3	
Provision for income taxes	36.1	49.4	
Income from continuing operations	54.3	78.9	
Loss from discontinued operations, net of income taxes	(1.1)	(3.7)	
Net income	\$ 53.2	\$ 75.2	

See Notes to Condensed Combined Financial Statements.

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC CONDENSED COMBINED STATEMENTS OF COMPREHENSIVE INCOME (UNAUDITED) Six Months Ended March 29, 2013 and March 30, 2012

(in millions)

	Six M	Six Months Ended	
	March 29, 2013	March 30, 2012	
Net income	\$ 53.2	<b>2012</b> \$ 75.2	
Other comprehensive loss, net of tax			
Currency translation	(8.2)	(0.2)	
Unrecognized loss on benefit plans	(1.7)	(4.4)	
Unrecognized loss on derivatives	(4.0)	—	
Other comprehensive loss, net of tax	(13.9)	(4.6)	
Comprehensive income	\$ 39.3	\$ 70.6	

See Notes to Condensed Combined Financial Statements.

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC CONDENSED COMBINED BALANCE SHEETS (UNAUDITED) At March 29, 2013 and September 28, 2012 (in millions)

	March 29, 2013	September 28, 2012
Assets		
Current Assets:		
Accounts receivable trade, less allowance for doubtful accounts of \$8.8 and \$9.4	\$ 364.3	\$ 291.1
Inventories	461.4	435.3
Prepaid expenses and other current assets	166.3	150.9
Total current assets	992.0	877.3
Property, plant and equipment, net	969.3	945.2
Goodwill	532.0	507.5
Intangible assets, net	439.8	365.6
Other assets	184.9	179.0
Total Assets	\$3,118.0	\$ 2,874.6
Liabilities and Parent Company Equity		
Current Liabilities:		
Current maturities of long-term debt	\$ 7.1	\$ 1.3
Accounts payable	100.9	112.5
Accrued and other current liabilities	286.8	282.0
Total current liabilities	394.8	395.8
Long-term debt	2.3	8.9
Pension and postretirement benefits	153.4	189.6
Environmental liabilities	134.4	136.5
Other liabilities	293.7	251.9
Total Liabilities	978.6	982.7
Commitments and contingencies (note 12)		
Parent Company Equity:		
Parent company investment	2,068.4	1,807.0
Accumulated other comprehensive income	71.0	84.9
Total Parent Company Equity	2,139.4	1,891.9
Total Liabilities and Parent Company Equity	\$3,118.0	\$ 2,874.6

See Notes to Condensed Combined Financial Statements.

# THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC CONDENSED COMBINED STATEMENTS OF CASH FLOWS (UNAUDITED) Six Months Ended March 29, 2013 and March 30, 2012

(in millions)

	Six Mon March 29,	Six Months Ended March 29, March 30,	
	2013	2012	
Cash Flows From Operating Activities:			
Net income	\$ 53.2	\$ 75.2	
Loss from discontinued operations, net of income taxes	<u> </u>	3.7	
Income from continuing operations	54.3	78.9	
Adjustments to reconcile net cash (used in) provided by continuing operating activities:			
Depreciation and amortization	66.9	65.2	
Share-based compensation	6.6	5.5	
Deferred income taxes	3.5	4.5	
Other non-cash items	(2.8)	(5.1)	
Changes in assets and liabilities, net of the effects of acquisitions:			
Accounts receivable, net	(74.4)	6.6	
Inventories	(23.1)	(32.8)	
Accounts payable	(12.0)	(9.0)	
Income taxes	27.3	39.1	
Accrued and other liabilities	(41.8)	(49.2)	
Other	(12.3)	3.6	
Net cash (used in) provided by operating activities	(7.8)	107.3	
Cash Flows From Investing Activities:			
Capital expenditures	(76.7)	(63.3)	
Acquisition, net of cash acquired	(88.1)		
Other	(0.2)	1.1	
Net cash used in investing activities	(165.0)	(62.2)	
Cash Flows From Financing Activities:			
Repayment of capital leases	(0.7)	(0.6)	
Excess tax benefit from stock-based compensation	3.0	1.6	
Debt issuance costs	(2.3)		
Net transfer from (to) parent	172.8	(46.1)	
Net cash provided by (used in) financing activities	172.8	\$ (45.1)	
Net change in cash	\$	\$ —	

See Notes to Condensed Combined Financial Statements.

## THE PHARMACEUTICALS BUSINESS OF COVIDIEN PLC NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)

#### 1. Basis of Presentation

#### Separation

On December 15, 2011, Covidien plc ("Covidien" or "parent") announced a plan to spin off its Pharmaceuticals business into a separate, publicly traded company. Upon completion of the separation, Mallinckrodt plc will be the parent company which will own the Pharmaceuticals business.

#### **Basis of Presentation**

The Pharmaceuticals business of Covidien plc (such business referred to as the "Company"), presented herein, represents a combined reporting entity comprising the assets and liabilities used in managing and operating the Company, including corporations, branches and operations that have been carved out which relate to Covidien's Pharmaceuticals business. The unaudited combined financial statements have been presented on a standalone basis and are derived from the consolidated financial statements of Covidien. The unaudited combined financial statements have been prepared in United States ("U.S.") dollars and in accordance with accounting principles generally accepted in the U.S. ("GAAP"). The preparation of the unaudited combined financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amount of assets and liabilities, disclosure of contingent assets and liabilities, and the reported amounts of revenues and expenses. Actual results may differ from those estimates. In management's opinion, the unaudited combined financial statements contain all normal recurring adjustments necessary for a fair presentation of the interim results reported. The year-end balance sheet data were derived from audited combined financial statements, but do not include all of the annual disclosures required by GAAP; accordingly, these financial statements should be read in conjunction with the Company's audited combined financial statements included elsewhere in this information statement.

Intercompany transactions between the Company and Covidien have been included in these condensed combined financial statements and are considered to be effectively settled for cash in the condensed combined financial statements at the time the transaction is recorded. The total net effect of the settlement of these intercompany transactions is reflected in the condensed combined statements of cash flows as a financing activity and in the condensed combined balance sheets as parent company investment.

The Company's unaudited combined financial statements may not be indicative of the Company's future performance and do not necessarily reflect what the results of operations, financial position and cash flows would have been had it operated as an independent, publicly traded company during the periods presented. The unaudited combined financial statements include expense allocations for certain functions provided by Covidien, including, but not limited to, general corporate expenses related to finance, legal, information technology, human resources, communications, employee benefits and incentives, insurance and stock-based compensation. These expenses have been allocated to the Company on the basis of direct usage when identifiable, with the remainder allocated on the basis of operating expenses, headcount or other measures. During the six months ended March 29, 2013 and March 30, 2012, the Company was allocated \$25.5 million and \$22.7 million, respectively, of general corporate expenses incurred by Covidien which are included within selling, general and administrative expenses in the unaudited combined statements of income. Management considers the bases on which the expenses have been allocated to reasonably reflect the utilization of services provided to or the benefit received by the Company during the periods presented. The allocations may not, however, reflect the expense the Company would have incurred as an independent, publicly traded company for the periods presented. Actual costs that may have been incurred if the Company had been a standalone company would depend on a number of factors, including the organizational structure, what functions were outsourced or performed by employees and strategic decisions made in areas such as information technology and infrastructure. In addition, as part of Covidien, the Company shared in other costs of Covidien, including costs of Covidien's international

infrastructure. The Company's portion of these costs were \$20.6 million and \$21.2 million during the first six months of 2013 and 2012, respectively. The Company is unable to determine what such costs would have been had the Company been independent. Following the separation, the Company will perform these functions using its own resources or purchased services. For an interim period, however, some of these functions will continue to be provided by Covidien under a transition services agreement, particularly in relation to areas outside the U.S.

The unaudited combined financial statements include certain assets and liabilities that have historically been recorded at the Covidien corporate level but are specifically identifiable or otherwise allocable to the Company. The cash and cash equivalents held by Covidien at the corporate level are not specifically identifiable to the Company. Accordingly, cash and cash equivalents have not been allocated to the Company for any of the periods presented. Covidien's debt and the related interest expense have not been allocated to the Company for any of the periods presented since the Company is not the legal obligor of the debt and Covidien's borrowings were not directly attributable to the Company's business. Debt incurred by the Company directly has been included in the unaudited combined financial statements.

#### 2. Related Party Transactions

#### **Related Party Sales and Purchases**

During the six months ended March 29, 2013 and March 30, 2012, the Company sold inventory to another Covidien business in the amount of \$25.9 million and \$27.9 million, respectively, which is included in net sales in the condensed combined statements of income. The Company also purchases inventories from other Covidien businesses. The Company recognized cost of sales from the inventory purchased from Covidien of \$22.0 million and \$17.0 million during the six months ended March 29, 2013 and March 30, 2012, respectively. As of March 29, 2013 and September 28, 2012, the aggregate amount of inventories purchased from other Covidien businesses that remained on the Company's condensed combined balance sheets was \$6.3 million and \$4.5 million, respectively.

#### **Royalty Income**

During the six months ended March 29, 2013 and March 30, 2012, a subsidiary of Covidien paid royalties to the Company for use of certain trademarks and technology. Amounts outstanding under these agreements are considered settled for cash in the condensed combined financial statements at the end of each reporting period and, as such, are included in parent company investment. During both the six months ended March 29, 2013 and March 30, 2012, the Company recognized royalty income of \$0.5 million, which is included in other income in the condensed combined statements of income.

#### 3. Restructuring and Related Charges, Net

Net restructuring and related charges by segment are as follows:

	Six Month	is Ended
(Dollars in Millions)	March 29, 2013	March 30, 2012
Specialty Pharmaceuticals	\$ 6.6	\$ 7.9
Global Medical Imaging	1.3	3.0
	7.9	10.9
Less: accelerated depreciation	(1.3)	(5.4)
Restructuring charges, net	\$ 6.6	\$ 5.5

Net restructuring and related charges are comprised of the following:

	Six Months Ended	
(Dollars in Millions)	March 29, 2013	March 30, 2012
2011 program	\$ 7.2	\$ 10.8
2009 program	(0.1)	0.1
Acquisitions	0.8	
Restructuring and related charges, net	7.9	10.9
Less: non-cash charges, including accelerated depreciation	(1.4)	(5.4)
Total charges expected to be settled in cash	\$ 6.5	\$ 5.5

The following table summarizes cash activity for restructuring reserves that are specifically identifiable to the Company, substantially all of which relates to employee severance and benefits under the 2011 program:

(Dollars in Millions)	Total
Balance at September 28, 2012	<u>Total</u> \$ 8.9
Charges	7.4
Changes in estimate	(0.9)
Currency translation and other	(0.2)
Cash payments	(6.2)
Balance at March 29, 2013	<u>\$ 9.0</u>

Net restructuring and related charges, including associated asset impairments, incurred cumulative to date related to the 2011 program are as follows:

Program
\$ 22.7
14.9
14.9 \$ 37.6

Substantially all of the restructuring reserves are included in accrued and other current liabilities on the Company's condensed combined balance sheets.

## 4. Acquisition

*CNS Therapeutics*—On October 1, 2012, the Company's Specialty Pharmaceuticals segment acquired all the outstanding equity of CNS Therapeutics, Inc. ("CNS Therapeutics"), a specialty pharmaceuticals company focused on developing and commercializing products for site-specific administration to the central nervous system to treat neurological disorders and intractable chronic pain, for total consideration of \$95.0 million. The total consideration was comprised of an upfront cash payment of \$88.1 million (net of cash acquired of \$3.6 million) and the fair value of contingent consideration of \$6.9 million. This contingent consideration, which could potentially total a maximum of \$9.0 million, is discussed further in note 11. The acquisition of CNS Therapeutics expanded the Company's branded pharmaceuticals portfolio and supports the Company's strategy of leveraging its therapeutic expertise and core capabilities in manufacturing, regulatory and commercialization to serve patients.

F-52

The following amounts represent the preliminary estimate of the fair value of the identifiable assets acquired and liabilities assumed:

(Dollars in Millions)	
Current assets <sup>(1)</sup>	\$ 13.3
Intangible assets	91.9
Goodwill (non-tax deductible) <sup>(2)</sup>	24.5
Total assets acquired	129.7
Current liabilities	4.0
Deferred tax liabilities (non-current)	27.1
Contingent consideration (non-current)	6.9
Total liabilities assumed	<u>38.0</u> \$ 91.7
Net assets acquired	\$ 91.7
-	

This amount includes \$3.3 million of accounts receivable, which is also the gross contractual value. As of the acquisition date, the fair value of accounts receivable approximated carrying value.
 (2) Conductil relates to the Comparison of bility to explore the comparison of the comparison of

Goodwill relates to the Company's ability to exploit CNS Therapeutics' technologies.

Intangible assets acquired consist of the following:

(Dollars in Millions)	Amount	Weighted-Average Amortization Period
Completed technology	\$ 73.1	13 years
Trademark	0.2	3 years
In-process research and development	18.6	Non-Amortizable
	\$ 91.9	

The in-process research and development projects primarily relate to three intrathecal pain products. As of the date of acquisition, these pain products were in various stages of development. Development, testing, clinical trials and regulatory submission are required in order to bring these products to market. The Company estimates that the total costs to complete these products will be approximately \$18.0 million. In addition, the Company expects that regulatory approvals will occur between 2015 and 2018. The Company determined the valuation of in-process research and development using, among other factors, appraisals. The value was primarily based on the discounted cash flow method and was discounted at a 35% rate, which was considered commensurate with the risks and stages of development of the pain products. Future residual cash flows that could be generated from the products were determined based upon management's estimate of future revenue and expected profitability of the products. These projected cash flows were then discounted to their present values taking into account management's estimate of future expenses that would be necessary to bring the products to completion.

The Company has not yet finalized its deferred tax assets and liabilities for the CNS Therapeutics acquisition, the impact of which is not expected to have a material effect on the Company's financial condition.

#### 5. Inventories

Inventories were comprised of the following at the end of each period:

(Dollars in Millions)	March 29, 		
Raw materials and supplies	\$ 84.3	\$	74.1
Work in process	181.6		184.7
Finished goods	195.5		176.5
Inventories	\$ 461.4	\$	435.3

## 6. Goodwill and Intangible Assets

The changes in the carrying amount of goodwill were as follows:

(Dollars in Millions)	Specialty Pharmaceuticals	Global Medical Imaging	Total
Goodwill at September 28, 2012	\$ 287.8	\$ 219.7	\$507.5
Acquisition	24.5		24.5
Goodwill at March 29, 2013	\$ 312.3	\$ 219.7	\$532.0

The gross carrying amount and accumulated amortization of intangible assets at the end of each period were as follows:

	h 29, 2013		ber 28, 2012
Gross Carrying <u>Amount</u>	Accumulated Amortization	Gross Carrying Amount	Accumulated Amortization
\$ 449.2	\$ 185.1	\$ 376.1	\$ 173.7
191.1	73.2	191.1	67.1
7.9	3.7	7.7	3.5
\$ 648.2	\$ 262.0	\$ 574.9	\$ 244.3
\$ 35.0		\$ 35.0	
18.6			
\$ 53.6		\$ 35.0	
	Gross Carrying Amount \$ 449.2 191.1 7.9 \$ 648.2 \$ 35.0 18.6	Gross Carrying Amount         Accumulated Amortization           \$ 449.2         \$ 185.1           191.1         73.2           7.9         3.7           \$ 648.2         \$ 262.0           \$ 35.0         18.6	Gross Carrying Amount         Accumulated Amortization         Gross Carrying Amount           \$ 449.2         \$ 185.1         \$ 376.1           191.1         73.2         191.1           7.9         3.7         7.7           \$ 648.2         \$ 262.0         \$ 574.9           \$ 35.0         \$ 35.0         \$ 35.0

Intangible asset amortization expense for the six months ended March 29, 2013 and March 30, 2012 was \$17.7 million and \$13.5 million, respectively. The estimated aggregate amortization expense on intangible assets owned by the Company is expected to be as follows:

(Dollars in Millions)	
Fiscal 2013	\$35.4
Fiscal 2014	35.4
Fiscal 2015	35.4
Fiscal 2016	35.3
Fiscal 2017	33.9

#### 7. Debt

In November 2012, Mallinckrodt International Finance S.A. ("MIFSA"), a wholly owned subsidiary of Covidien that will become a wholly owned subsidiary of Mallinckrodt plc upon completion of the distribution, was formed in connection with the separation. MIFSA is a holding company established to directly, or indirectly, own substantially all of the operating subsidiaries of the Company, to issue debt securities and perform treasury operations.

In March 2013, MIFSA entered into a \$250 million five-year senior unsecured revolving credit facility that matures in June 2018 (the "credit facility"). Borrowings under the credit facility will initially bear interest at LIBOR plus 1.50% per annum (subject to adjustment based upon a ratings-based pricing grid). The credit facility agreement contains customary covenants, including a financial maintenance covenant that limits Mallinckrodt plc's ratio of debt to earnings before interest, income taxes, depreciation and amortization, as adjusted for certain

items, and another financial maintenance covenant that requires Mallinckrodt plc's ratio of earnings before interest, income taxes, depreciation and amortization, as adjusted for certain items, to interest expense to exceed certain thresholds. MIFSA will not be permitted to draw upon the credit facility until certain conditions are met, including completion of the distribution.

#### 8. Retirement Plans

The net periodic benefit cost for the Company's defined benefit pension plans was as follows:

	Six Month	s Ended
(Dollars in Millions)	March 29, 2013	March 30, 2012
Service cost	\$ 2.4	\$ 2.6
Interest cost	9.1	10.7
Expected return on plan assets	(14.7)	(12.3)
Amortization of net actuarial loss	6.0	5.8
Amortization of prior service cost	0.3	0.3
Net periodic benefit cost	<u>\$ 3.1</u>	\$ 7.1

During the six months ended March 29, 2013, Covidien made a \$37.5 million voluntary contribution to the Company's pension plans.

The net periodic benefit credit for postretirement benefit plans for the six months ended March 29, 2013 and March 30, 2012 was \$3.1 million and \$2.9 million, respectively, the components of which were not material.

#### 9. Guarantees

In disposing of assets or businesses, the Company often provides representations, warranties and indemnities to cover various risks and liabilities, including unknown damage to the assets, environmental risks involved in the sale of real estate, liability to investigate and remediate environmental contamination at waste disposal sites and manufacturing facilities, and unidentified tax liabilities related to periods prior to disposition. Except as discussed below, the Company generally does not have the ability to estimate the potential liability from such indemnities because they relate to unknown conditions. However, the Company has no reason to believe that these uncertainties would have a material adverse effect on its financial condition, results of operations and cash flows.

In connection with the sale of the Specialty Chemicals business (formerly known as "Mallinckrodt Baker"), in fiscal 2010 the Company agreed to indemnify the purchaser with respect to various matters, including certain environmental, health, safety, tax and other matters. The indemnification obligations relating to environmental, health and safety matters have a term of 17 years, while some of the other indemnification obligations have an indefinite term. The amount of the liability relating to all of these indemnification obligations included in other liabilities on the Company's combined balance sheets at both March 29, 2013 and September 28, 2012 was \$22.4 million, of which \$18.3 million related to environmental health and safety matters. The value of the environmental, health and safety indemnity was measured based on the probability-weighted present value of the costs expected to be incurred to address environmental, health and safety claims made under the indemnity. The aggregate fair value of these indemnification obligations did not differ significantly from their aggregate carrying value at March 29, 2013 and September 28, 2012. As of March 29, 2013, the maximum future payments the Company could be required to make under all of these indemnification obligations to the purchaser, of which \$23.7 million and \$24.5 million remained in other assets on the condensed combined balance sheets at March 29, 2013 and September 28, 2012, respectively.



The Company has recorded liabilities for known indemnification obligations included as part of environmental liabilities, which are discussed in note 12. In addition, the Company is liable for product performance; however in the opinion of management, such obligations will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

The Company is required to provide the Nuclear Regulatory Commission financial assurance demonstrating its ability to cover the cost of decommissioning its Maryland Heights, Missouri radio pharmaceuticals production facility upon closure, though the Company does not intend to close this facility. The Company has provided this financial assurance in the form of a \$58.0 million surety bond. In addition, as of March 29, 2013, the Company had a \$21.1 million letter of credit to guarantee decommissioning costs associated with its St. Louis, Missouri plant.

As of March 29, 2013, the Company had various other letters of credit and guarantee and surety bonds totaling \$20.5 million. In addition, at March 29, 2013, Covidien had outstanding letters of credit and guarantee and surety bonds totaling \$132.1 million, which supported multiple Covidien businesses, including the Company.

#### **10. Derivative Instruments**

The Company is exposed to certain risks relating to its business operations. Risks that relate to foreign exchange exposure and certain commodity price exposures are managed by participating in the centralized hedging functions of Covidien which are designed to minimize exposure to such risks. Foreign currency option and forward contracts are used to economically manage the foreign exchange exposures of operations outside the U.S. Swap contracts on commodities are periodically entered into to manage the price risk associated with forecasted purchases of commodities used in the Company's manufacturing processes. The associated derivative assets and liabilities for these types of instruments have not been included on the Company's condensed combined balance sheet since derivative activity is centrally managed by Covidien. Changes in the derivative financial instrument's fair value which related to the Company's business operations, however, have been recognized in the Company's earnings unless specific hedge criteria are met. Covidien has designated certain commodity swap contracts as cash flow hedges. Covidien has not designated the foreign currency forward and option contracts as hedging instruments.

Risks that relate to interest rate exposure are managed by using derivative instruments. In March 2013, MIFSA entered into forward interest rate lock contracts to hedge the risk of variability in the market interest rates prior to the issuance of fixed rate debt. These transactions have been reflected in the financial statements, including the condensed combined balance sheet, since the transactions were solely entered into in connection with the separation and were not centrally managed by Covidien. At March 29, 2013, the Company had a \$1.3 million liability related to an open interest rate lock contract, which was included in accrued and other current liabilities on the condensed combined balance sheet.

#### Foreign Exchange Exposure

Derivatives not designated as hedging instruments—The Company has foreign exchange exposure on the translation of the financial statements and on transactions denominated in foreign currencies. Covidien's policy is to use various forward and option contracts to economically manage foreign currency exposures on accounts and notes receivable, accounts payable, intercompany loans and forecasted transactions that are denominated in certain foreign currencies. Covidien generally manages its exposure for forecasted transactions for the upcoming 12 months. These contracts do not meet the necessary criteria to qualify for hedge accounting; accordingly, all associated changes in fair value are recognized in earnings.

The location and amount of the net (loss) gain on foreign exchange forward and option contracts not designated as hedging instruments and related hedged items were recorded as follows:

	Six Month	Six Months Ended	
(Dollars in Millions)	March 29, 2013	March 30, 2012	
Cost of sales	\$ (2.4)	\$ 1.3	
Selling, general and administrative expenses	2.3	(0.5)	
	\$ (0.1)	\$ 0.8	

#### **Commodities** Exposure

Covidien has entered into gas commodity swap contracts on behalf of the Company. The amounts of the net losses on these contracts were recorded as follows:

	Six Month	Six Months Ended	
(Dollars in Millions)	March 29, 2013	March 30, 2012	
Cost of sales	\$ 0.2	\$ 0.5	
Selling, general and administrative expenses	0.6	1.4	
	\$ 0.8	\$ 1.9	

## Interest Rate Exposure

In March 2013, MIFSA entered into a forward interest rate lock contract with a \$300 million notional value. MIFSA designated the interest rate lock contract as a cash flow hedge against the risk of variability in market interest rates prior to its anticipated issuance of fixed rate senior notes due April 2023. The interest rate lock contract was considered to be highly effective; accordingly the \$2.7 million loss that resulted upon settlement of the contract on March 26, 2013, was recorded in accumulated other comprehensive income. The notes, which were originally anticipated to be issued in March 2013, were not issued until April 2013 (see note 14). This delay resulted in exposure to potential market interest rate variability for the period subsequent to March 26, 2013 until the issuance of the notes. To offset this risk, in March 2013, MIFSA entered into a new forward interest rate lock contract. This contract also had a \$300 million notional value, was designated as a cash flow hedge against the risk of variability in market interest rates prior to debt issuance and was deemed to be highly effective. As of March 29, 2013, the fair value of this contract was a loss of \$1.3 million, which was recorded in accumulated other comprehensive income. No additional loss was recorded upon settlement of this contract on April 2, 2013. As of March 29, 2013, a \$4.0 million loss resulting from both interest rate lock contracts remained in accumulated other comprehensive income and will be amortized to interest expense over the ten-year term of the notes.

In April 2013, MIFSA entered into an additional forward interest rate lock contract with a \$300 million notional value to hedge the exposure to potential market interest rate variability for the period subsequent to April 2, 2013 until the issuance of the notes. This hedging relationship was considered to be highly effective; accordingly the \$3.6 million loss that resulted upon settlement of this interest rate lock contract in April 2013 will also be recorded in accumulated other comprehensive income and amortized to interest expense over the ten-year term of the notes.

F-57

#### 11. Financial Instruments and Fair Value Measurements

Fair value is defined as the exit price that would be received from the sale of an asset or paid to transfer a liability, using assumptions that market participants would use in pricing an asset or liability. The fair value guidance establishes a three-level fair value hierarchy, which maximizes the use of observable inputs and minimizes the use of unobservable inputs used in measuring fair value. The levels within the hierarchy are as follows:

- Level 1- observable inputs such as quoted prices in active markets for identical assets or liabilities;
- Level 2- significant other observable inputs that are observable either directly or indirectly; and
- Level 3— significant unobservable inputs in which there is little or no market data, which requires the Company to develop its own assumptions.

The following tables provide a summary of the significant assets and liabilities that are measured at fair value on a recurring basis at March 29, 2013 and September 28, 2012:

(Dollars in Millions) Assets:	March 29, 	Basi Quoted Prices in Active Markets for Identical Assets (Level 1)	s of Fair Value Measureme Significant Other Observable Inputs (Level 2)	ent Significant Unobservable Inputs (Level 3)
Debt and equity securities held in rabbi trust	\$ 23.9	\$ 13.2	\$ 10.7	\$
Liabilities:				
Deferred compensation liabilities	\$ 9.9	\$ 9.9	\$ —	\$ —
Contingent consideration	6.9		—	6.9
Interest rate lock	1.3		1.3	
Total liabilities at fair value	\$ 18.1	\$ 9.9	\$ 1.3	\$ 6.9

		Bas	is of Fair Value Measureme	ent
(Dollars in Millions)	September 28, 2012	Quoted Prices in Active Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)
Assets:			<u> </u>	
Debt and equity securities held in rabbi trust	\$ 25.2	\$ 13.7	<u>\$ 11.5</u>	\$
Liabilities:				
Deferred compensation liabilities	\$ 9.3	\$ 9.3	<u>\$                                    </u>	\$

Debt and equity securities held in rabbi trust—Debt securities held in the rabbi trust primarily consist of U.S. government and agency securities and corporate bonds. When quoted prices are available in an active market, the investments are classified as level 1. When quoted market prices for a security are not available in an active market, they are classified as level 2. Equity securities held in the rabbi trust primarily consist of U.S. common stocks, which are valued using quoted market prices reported on nationally recognized securities exchanges.

Deferred compensation liabilities—Covidien maintains a non-qualified deferred compensation plan in the U.S., which permits eligible employees of the Company to defer a portion of their compensation. A recordkeeping account is set up for each participant and the participant chooses from a variety of funds for the deemed investment of their accounts. The measurement funds generally correspond to the funds offered in Covidien's U.S. tax-qualified defined contribution retirement plan and the account balance fluctuates with the investment returns on those funds.

*Contingent consideration*—During the first six months of fiscal 2013, the Company recorded contingent consideration of \$6.9 million upon the acquisition of CNS Therapeutics. This contingent consideration, which could potentially total a maximum of \$9.0 million, is primarily based on whether the FDA approves another dosage form of Gablofen on or before December 31, 2016. The fair value of the contingent payments was measured based on the probability-weighted present value of the consideration expected to be transferred using a discount rate of 1.0%.

(Dollars in Millions)	
Balance at September 28, 2012	\$—
Acquisition date fair value of contingent consideration	6.9
Balance at March 29, 2013	\$ 6.9

#### Financial Instruments Not Measured at Fair Value

The fair value of restricted cash is equivalent to its carrying value of \$23.7 million and \$24.6 million as of March 29, 2013 and September 28, 2012, respectively (level 1), substantially all of which is included in other assets on the condensed combined balance sheets. The Company's life insurance contracts are carried at cash surrender value (level 3). The fair value of these contracts approximates the carrying value of \$50.7 million and \$47.6 million at March 29, 2013 and September 28, 2012, respectively. Since quoted market prices for the Company's debentures are not available in an active market, they are classified as level 2 for purposes of developing an estimate of their fair value. The fair value of the Company's debt approximates the carrying value of \$9.4 million and \$10.2 million at March 29, 2013 and September 28, 2012, respectively.

## Concentration of Credit and Other Risks

Financial instruments that potentially subject the Company to concentrations of credit risk primarily consist of accounts receivable. The Company does not require collateral from customers. A portion of the Company's trade accounts receivable outside the U.S., however, include sales to government-owned or supported healthcare systems in several countries, which are subject to payment delays. Payment is dependent upon the financial stability and creditworthiness of those countries' national economies. Deteriorating credit and economic conditions in parts of Western Europe, particularly in Spain and Italy, may continue to increase the average length of time it takes the Company to collect its accounts receivable in certain regions within these countries.

The Company routinely evaluates all government receivables for potential collection risks associated with the availability of government funding and reimbursement practices. The Company has not incurred any significant losses on government receivables; however, if the financial condition of customers or the countries' healthcare systems continue to deteriorate such that their ability to make payments is uncertain, additional allowances may be required in future periods.

The Company's accounts receivable, net of the allowance for doubtful accounts, in Spain and Italy at the end of each period are as follows:

(Dollars in Millions)	March 29, 2013	ember 28, 2012
Spain	\$ 15.7	\$ 15.0
Italy	13.9	12.5

Net sales to customers in Spain and Italy totaled \$26.3 million and \$29.0 million for the six months ended March 29, 2013 and March 30, 2012, respectively.

The following table shows net sales attributable to distributors that accounted for 10% or more of the Company's total net sales:

	Six Month	Six Months Ended	
	March 29, 2013	March 30, 2012	
Cardinal Health, Inc.	20%	18%	
McKesson Corporation	16%	12%	
AmerisourceBergen Corporation	7%	8%	

The following table shows accounts receivable attributable to distributors that accounted for 10% or more of the Company's gross accounts receivable at the end of each period:

	March 29, 	September 28, 2012
Cardinal Health, Inc.	20%	19%
McKesson Corporation	23%	20%
AmerisourceBergen Corporation	9%	10%

The following table shows net sales attributable to products that accounted for 10% or more of the Company's total net sales:

	Six Month	Six Months Ended	
	March 29,	March 30,	
	2013	2012	
Optiray (CMDS)	14%	17%	
Acetaminophen products (API)	10%	11%	

Molybdenum-99 ("Mo-99") is a key raw material in the Company's Ultra-Technekow DTE technetium generators that are sold by its Global Medical Imaging segment. There are only eight suppliers of this raw material worldwide. The Company has agreements to obtain Mo-99 from three nuclear research reactors and relies predominantly on two of these reactors for its Mo-99 supply. Accordingly, a disruption in the commercial supply or a significant increase in the cost of this material from these sources could have a material adverse effect on the Company's financial condition, results of operations and cash flows.

## 12. Commitments and Contingencies

The Company is subject to various legal proceedings and claims, including patent infringement claims, product liability matters, environmental matters, employment disputes, contractual disputes and other commercial disputes, including those described below. Management believes that these legal proceedings and claims likely will be resolved over an extended period of time. Although it is not feasible to predict the outcome of these matters, management is of the opinion that their ultimate resolution should not have a material adverse effect on the Company's financial position, results of operations and cash flows.

#### **Governmental Proceedings**

On January 7, 2009, the Company received a subpoena from the U.S. Attorney's Office for the Northern District of California requesting production of documents relating to the sales and marketing of its Tofranil-PM, Restoril and Magnacet products. The Company is complying as required by the terms of the subpoena. The Company believes that the amount accrued related to this matter is adequate, the amount of which is not significant.

#### Patent/Antitrust Litigation

*Tyco Healthcare Group LP, et al. v. Mutual Pharmaceutical Company, Inc.* The Company filed a patent infringement suit in the U.S. District Court for the District of New Jersey against Mutual Pharmaceutical Co., Inc., et al. (collectively, "Mutual") on March 20, 2007 pursuant to procedures set out in the Drug Price Competition and Patent Term Restoration Act of 1984, after Mutual submitted an Abbreviated New Drug Application ("ANDA") to the FDA seeking to sell a generic version of the Company's 7.5 mg Restoril sleep aid product. Mutual also filed antitrust and unfair competition counterclaims. The patents at issue have since expired or been found invalid. On January 18, 2013, the trial court issued an opinion and order granting the Company's motion for summary judgment regarding Mutual's antitrust and unfair competition counterclaims. On May 1, 2013, Mutual appealed this decision to the U.S. Court of Appeals for the Federal Circuit. While it is not possible at this time to determine with certainty the ultimate outcome of the counterclaims, the Company believes that the final resolution of the claims will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

## **Pricing Litigation**

Two cases are pending against the Company that allege generally that the Company and numerous other pharmaceuticals companies reported false pricing information in connection with certain drugs that are reimbursable under Medicaid, resulting in overpayment by state Medicaid programs for those drugs. These cases, brought by state Attorneys General in Utah and Louisiana, generally seek monetary damages and attorneys' fees. The Company is named as a defendant in *State of Utah v. Actavis US, Inc., et al.* filed May 8, 2008, which is pending in the Third Judicial Circuit of Salt Lake County, Utah and in *State of Louisiana v. Abbott Laboratories Inc., et al.* filed November 3, 2010, which was pending in the 19th Judicial District, Parish of East Baton Rouge, Louisiana. In May 2013, we agreed to terms of settlement with the Attorney General for the state of Louisiana resolving all claims in *State of Louisiana v. Abbott Laboratories Inc., et al.*, involving alleged reporting of false pricing information in connection with certain drugs that are reimbursable under Medicaid, resulting in overpayment by the state Medicaid program for those drugs. The Company intends to contest the Utah case and to explore other options as appropriate. The Company believes that the amount accrued related to these cases, the amount of which is not significant, is adequate.

#### **Environmental Remediation and Litigation Proceedings**

The Company is involved in various stages of investigation and cleanup related to environmental remediation matters at a number of sites, including those described below. The ultimate cost of site cleanup and timing of future cash outlays is difficult to predict, given the uncertainties regarding the extent of the required cleanup, the interpretation of applicable laws and regulations, and alternative cleanup methods. The Company concluded that, as of March 29, 2013, it was probable that it would incur remedial costs in the range of \$144.6 million to \$251.7 million. The Company concluded that, as of March 29, 2013, the best estimate within this range was \$144.6 million, of which \$10.2 million was included in accrued and other current liabilities and the remainder was included in environmental liabilities on the condensed combined balance sheet at March 29, 2013.

*Orrington, Maine*—The Company is a successor to a company which owned and operated a chemical manufacturing facility in Orrington, Maine from 1967 until 1982. The Company is currently responsible for the costs of completing an environmental site investigation required by the U.S. Environmental Protection Agency ("EPA") and the Maine Department of Environmental Protection ("MDEP"). Based on the site investigation, the Company submitted a Corrective Measures Study plan and identified a preferred alternative which was submitted to the EPA and MDEP for approval in 2004. MDEP disagreed with the proposed alternative and served a compliance order on the Company and United States Surgical Corporation, a subsidiary of Covidien, in December 2008. The compliance order included a directive to remove a significant volume of soils at the site. On December 19, 2008, the Company filed an appeal with the Maine Board of Environmental Protection ("Maine Board") to challenge the terms of the compliance order. A hearing before the Maine Board began on January 25,

2010 and concluded on February 4, 2010. On August 19, 2010, the Maine Board modified the MDEP order and issued a final order requiring removal of two landfills, capping of the remaining three landfills, installation of a groundwater extraction system and long-term monitoring of the site and the three remaining landfills.

On September 17, 2010, the Company appealed the final order issued by the Maine Board in Maine Superior Court. On appeal, the Company has requested that the Superior Court invalidate the Maine Board's final order in its entirety or, in the alternative, reverse or modify the final order to eliminate the requirements that it remove one of the two landfills and recap the remaining three landfills. The Company also appealed certain administrative requirements of the final order. On November 1, 2012, the Superior Court affirmed the Maine Board's final order. The Company has appealed the Superior Court's decision to the Maine Supreme Judicial Court. The Company has assessed the status of this matter and has concluded that it is more likely than not that the Maine Board's final order will be either invalidated, reversed or modified, and, further, intends to vigorously pursue all available means to achieve such result.

The Company estimates that, as of March 29, 2013, the cost to comply with the proposed remediation alternatives at this site ranges from \$94.7 million to \$166.2 million. However, there are still significant uncertainties in the outcome of the pending litigation, and the Company continues to disagree with the level of remediation outlined in the Maine Board's final order. At March 29, 2013, estimated future investigation and remediation costs of \$94.7 million were accrued for this site.

Penobscot River and Bay—Since April 2000, the Company has also been involved in a lawsuit, Maine People's Alliance and Natural Resources Defense Council, Inc. v. HoltraChem Manufacturing Company, LLC and Mallinckrodt US LLC, filed in the U.S. District Court for the District of Maine by the Natural Resources Defense Council and the Maine People's Alliance. Plaintiffs sought an injunction requiring the Company to conduct extensive studies of mercury contamination of the Penobscot River and Bay and options for remediating such contamination, and to perform appropriate remedial activities, if necessary.

On July 29, 2002, following a March 2002 trial, the District Court entered an opinion and order which held that conditions in the Penobscot River and Bay may pose an imminent and substantial endangerment and that the Company was liable for the cost of performing a study of the river and bay. The District Court subsequently appointed a study panel to oversee the study and ordered the Company to pay costs associated with the study. The study panel conducted a Phase I study and completed a Phase II study, which included several years of field work and data collection. The study panel issued the Phase II study report on April 17, 2013. The Company and its outside consultants are currently reviewing the Phase II report to assess whether or not an order to conduct remediation is probable. Given the length and complexity of the report and the extensive analysis required to evaluate the study panel recommendations, it is not possible at this time to estimate the costs, if any, that might result from the issuance of the Phase II report or to determine the type and extent of an order to conduct remediation in the Penobscot River and Bay, if any. Accordingly, costs of any such remediation are not included in the range of estimated aggregate environmental remediation costs.

The entity with liability for the investigation and remediation described under "Orrington, Maine" and "Penobscot River and Bay" will not be transferred to Mallinckrodt plc as part of the separation. Accordingly, this will be a liability of a Covidien entity following the separation.

*Crab Orchard National Wildlife Refuge Superfund Site, near Marion, Illinois*—The Company is a successor in interest to International Minerals and Chemicals Corporation ("IMC"). Between 1967 and 1982, IMC leased portions of the AUS Operable Unit at the Crab Orchard Superfund Site (the "Site") from the government and manufactured various explosives for use in mining and other operations. In March 2002, the Department of Justice ("DOJ"), the U.S. Department of the Interior and the EPA (together, the "Government Agencies") issued a special notice letter to General Dynamics Ordnance and Tactical Systems, Inc. ("General Dynamics"), one of the other potentially responsible parties ("PRPs") at the Site, to compel General Dynamics to perform the remedial investigation and feasibility study ("RI/FS") for the AUS Operable Unit. General Dynamics negotiated

an Administrative Order on Consent with the Government Agencies to conduct an extensive RI/FS at the Site under the direction of the U.S. Fish and Wildlife Service. General Dynamics asserted in August 2004 that the Company is jointly and severally liable, along with approximately eight other lessees and operators at the AUS Operable Unit, for alleged contamination of soils and groundwater resulting from historic operations and has threatened to file a contribution claim against the Company and other parties for recovery of its costs incurred in connection with the RI/FS activities being conducted at the AUS Operable Unit. The Company and other PRPs who received demand letters from General Dynamics have explored settlement alternatives, but have not reached settlement to date. The Company and other PRPs are awaiting completion of the RI/FS by General Dynamics before the initiation of formal PRP negotiations to address resolution of these alleged claims. While it is not possible at this time to determine with certainty the ultimate outcome of this case, the Company believes that the final resolution of all known claims, after taking into account amounts already accrued, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

*Mallinckrodt Veterinary, Inc., Millsboro, Delaware*—The Company previously operated a plant in Millsboro, Delaware (the "Millsboro Site") that manufactured various animal healthcare products. In 2005, the Delaware Department of Natural Resources and Environmental Control found trichloroethylene ("TCE") in the Millsboro public water supply at levels that exceeded the federal drinking water standards. Further investigation to identify the TCE plume in the ground water indicated that the plume has extended to property owned by a third party near the Millsboro Site. The Company, and other former owners assumed responsibility for the Millsboro Site cleanup under the Alternative Superfund Program administered by the EPA. The Company and other PRPs entered into an Administrative Order on Consent with the EPA on May 10, 2010 which was subsequently amended in November 2010 and January 2011 to investigate the potential source of TCE contamination and to evaluate options to abate, mitigate and/or eliminate the release or threat of release of hazardous substances at the Millsboro Site. The Company, along with other parties, continues to conduct the studies and prepare remediation plans in accordance with the amended Administrative Order on Consent. While it is not possible at this time to determine with certainty the ultimate outcome of this case, the Company believes that the final resolution of all known claims, after taking into account amounts already accrued, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

*Coldwater Creek, St. Louis County, Missouri*—The Company is one of several companies named as defendants in four tort complaints (*McClurg, et al. v. MI Holdings, Inc., et al.*, filed February 28, 2012; *Adams, et al. v. MI Holdings, Inc., et al.*, filed April 10, 2012, *Steinmann et al. v. MI Holdings, Inc., et al.*, filed October 23, 2012 and *Schneider, et al. v. MI Holdings, Inc., et al.*, filed April 19, 2013) with numerous plaintiffs pending in the U.S. District Court for the Eastern District of Missouri. These cases allege personal injury for alleged exposure to radiological substances present in Coldwater Creek in Missouri. Plaintiffs lived in various locations in St. Louis County, Missouri near Coldwater Creek. Radiological residues which may have been present in the creek have been remediated by the U.S. Army Corps of Engineers. The Company believes that it has meritorious defenses to these complaints and is vigorously defending against them. The Company is unable to estimate a range of reasonably possible losses for the following reasons: (i) the proceedings are in early stages; (ii) the Company has not received and reviewed complete information regarding the plaintiffs and their medical conditions; and (iii) there are significant factual issues to be resolved. While it is not possible at this time to determine with certainty the ultimate outcome of these cases, the Company believes that the final resolution of all known claims will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

The Company has also recorded asset retirement obligations for the estimated future costs primarily associated with legal obligations to decommission two facilities within the Global Medical Imaging segment. Substantially all of these obligations are included in other liabilities on the condensed combined balance sheets. The following table provides a summary of the changes in the Company's asset retirement obligations:

(Dollars in Millions)	
Balance at September 28, 2012	\$46.4
Accretion expense	1.5
Payments	(0.2)
Currency translation	(0.1)
Balance at March 29, 2013	(0.1) \$47.6

The Company believes that any potential payment of such estimated amounts will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

## **Products Liability Litigation**

The Company is one of four manufacturers of Gadolinium-Based Contrast Agents, such as the Company's Optimark product, involved in litigation alleging that administration of these agents causes development of nephrogenic systemic fibrosis in a small number of patients with advanced renal impairment. In May 2013, the Company agreed to terms of settlement with the plaintiffs in four previously disclosed lawsuits involving its Optimark product. These settlements resolved cases that were included in federal multi-district litigation pending in the U.S. District Court for the Northern District of Ohio (In re Gadolinium-Based Contrast Agents Product Liability Litigation, which was established on February 27, 2008) and cases in various state courts.

Beginning with lawsuits brought in July 1976, the Company is also named as a defendant in personal injury lawsuits based on alleged exposure to asbestoscontaining materials. A majority of the cases involve product liability claims based principally on allegations of past distribution of products containing asbestos. A limited number of the cases allege premises liability based on claims that individuals were exposed to asbestos while on the Company's property. Each case typically names dozens of corporate defendants in addition to the Company. The complaints generally seek monetary damages for personal injury or bodily injury resulting from alleged exposure to products containing asbestos. The Company's involvement in asbestos cases has been limited because it did not mine or produce asbestos. Furthermore, in the Company's experience, a large percentage of these claims have never been substantiated and have been dismissed by the courts. The Company has not suffered an adverse verdict in a trial court proceeding related to asbestos claims, and intends to continue to defend these lawsuits. When appropriate, the Company settles claims; however, amounts paid to settle and defend all asbestos claims have been immaterial. As of April 30, 2013, there were approximately 11,600 asbestos-related cases pending against the Company.

The Company estimates pending asbestos claims and claims that were incurred but not reported, as well as insurance recoveries. The Company's estimate of its liability for pending and future claims is based on claims experience over the past five years and covers claims either currently filed or expected to be filed over the next seven years. The Company believes that it has adequate amounts recorded related to these matters. While it is not possible at this time to determine with certainty the ultimate outcome of these asbestos-related proceedings, the Company believes that the final outcome of all known and anticipated future claims, after taking into account amounts already accrued and insurance coverage, will not have a material adverse effect on the Company's financial condition, results of operations and cash flows.

#### **Other Matters**

The Company is a defendant in a number of other pending legal proceedings incidental to present and former operations, acquisitions and dispositions. The Company does not expect the outcome of these proceedings, either individually or in the aggregate, to have a material adverse effect on the Company's financial condition, results of operations and cash flows.

## 13. Segment Data

Selected information by business segment is as follows:

	Six Month	hs Ended
(Dollars in Millions)	March 29, 2013	March 30, 2012
Net sales <sup>(1)</sup> :		
Specialty Pharmaceuticals	\$ 604.6	\$ 491.6
Global Medical Imaging	458.8	507.3
Net sales of operating segments	1,063.4	998.9
Net sales to related parties <sup>(2)</sup>	25.9	27.9
Vet sales	\$1,089.3	\$1,026.8
Operating income:		
Specialty Pharmaceuticals	\$ 140.0	\$ 77.4
Global Medical Imaging	68.0	111.4
Segment operating income	208.0	188.8
Jnallocated amounts:		
Corporate and allocated expenses <sup>(3)</sup>	(65.7)	(26.6
Intangible asset amortization	(17.7)	(13.5)
Restructuring and related charges, net (note 3)	(7.9)	(10.9)
Separation costs	(26.4)	(10.2)
Dperating income	\$ 90.3	\$ 127.6

Amounts represent sales to external customers. Intersegment sales are not significant.
 Paragents products that ware sold to other Covidian businesses, which is discussed in

Represents products that were sold to other Covidien businesses, which is discussed in note 2.

(3) Includes administration expenses and certain compensation, environmental and other costs not charged to the Company's operating segments.

## 14. Subsequent Events

Subsequent events have been evaluated for adjustment through April 29, 2013, the date at which the parent's consolidated financial statements were completed and issued, and May 31, 2013, for purposes of evaluating disclosures in these condensed combined financial statements.

In April 2013, MIFSA, a wholly owned subsidiary of Covidien that will become a wholly owned subsidiary of Mallinckrodt plc upon completion of the distribution, issued \$300 million aggregate principal amount of 3.50% senior unsecured notes due 2018 and \$600 million aggregate principal amount of 4.75% senior unsecured notes due 2023 (collectively, the "notes"). Mallinckrodt plc will guarantee the notes on an unsecured and unsubordinated basis upon completion of the distribution. MIFSA will pay interest on the notes semiannually in arrears on April 15 and October 15 of each year, commencing on October 15, 2013. The notes were issued and sold in a private placement. The net proceeds to MIFSA from the issuance and sale of the notes were approximately \$889.3 million. It is anticipated that, upon completion of the distribution, MIFSA will retain for general corporate purposes an amount of the notes offering that, together with cash held by its subsidiaries, equals approximately \$168 million, and the remainder will be retained by Covidien.

## **REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM**

To the Board of Directors and Shareholders of Mallinckrodt plc:

We have audited the accompanying balance sheet of Mallinckrodt plc (the "Company") as of January 11, 2013 (date of capitalization). This financial statement is the responsibility of the Company's management. Our responsibility is to express an opinion on this financial statement based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statement is free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audit included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial statement, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, such balance sheet presents fairly, in all material respects, the financial position of the Company as of January 11, 2013 (date of capitalization), in conformity with accounting principles generally accepted in the United States of America.

/s/ DELOITTE & TOUCHE LLP Boston, Massachusetts February 1, 2013

# MALLINCKRODT PLC

## BALANCE SHEET At January 11, 2013 (date of capitalization) (in thousands of U.S. dollars)

Assets	
Current Assets:	
Cash	\$ 53
Total Assets	\$ 53 \$ 53
Commitments and contingencies (Note 4)	
Stockholders' Equity	
Ordinary A shares; €1.00 par value, 40,000 shares authorized, 40,000 shares issued and outstanding	\$ 53
Ordinary shares; \$0.20 par value, 300,000 shares authorized, 7 shares issued and outstanding	_
Total Stockholder's Equity	\$ 53

See Notes to Financial Statements.

## MALLINCKRODT PLC NOTES TO FINANCIAL STATEMENTS

#### 1. History and Description of the Company

Mallinckrodt plc (the "Company") was incorporated in Ireland, as a public limited company, on January 9, 2013 and capitalized on January 11, 2013 as a holding company for the purposes of being the parent company of Covidien plc's Pharmaceuticals business. The Company has not engaged in any business or other activities.

### 2. Basis of Presentation and Summary of Significant Accounting Policies

### **Basis of Presentation**

The balance sheet and accompanying notes have been prepared in accordance with accounting principles generally accepted in the U.S.

## Cash

Cash as of January 11, 2013 was received in exchange for ordinary shares issued.

#### **Functional Currency**

Items included in these financial statements are measured using the currency of the primary economic environment in which Mallinckrodt plc operates. The financial statements are presented in U.S. dollars, which is the Company's functional and presentation currency.

#### Subsequent Events

The Company has evaluated subsequent events for recognition or disclosure through February 1, 2013, the date the financial statements were available to be issued.

#### 3. Share Capital

Ordinary A shares have no voting or dividend rights. In addition, in the event of the liquidation of the Company, the holders of any ordinary A shares then outstanding would be entitled to payment only after the holders of ordinary shares have received amounts owed to them in accordance with the articles of association.

## 4. Commitments and Contingencies

The Company is not currently a party to any loss contingencies or litigation.

#### Tax Matters

Upon separation, Covidien will transfer its interest in the businesses comprising its Pharmaceuticals segment to the Company. In addition, assets and liabilities related to certain non-operating activities of Covidien, primarily intercompany transactions, will also be transferred to the Company; historically, these activities were not managed by the Pharmaceuticals segment. Most of these assets and liabilities are not significant. However, as measured as of September 28, 2012, the Company expects to assume a net income tax payable of approximately \$125 million, primarily consisting of non-current contingent tax liabilities associated with unresolved tax matters related to these non-operating activities. In addition, the Company expects to enter into a tax matters agreement, pursuant to which Covidien will indemnify it, net of certain tax benefits realized by us, for any payments related to these liabilities which in the aggregate (after taking into account certain tax benefits realized by us) exceed \$200 million, and which pertain to periods prior to the distribution date. In addition, the Company expects to assume a net deferred tax liability of approximately \$134 million as measured as of September 28, 2012, primarily resulting from different treatment for book and tax purposes of certain intercompany transactions and the deferred tax effect of the aforementioned contingent tax liabilities. As the Company is not currently obligated to pay any of these liabilities and will not be obligated to do so until the separation has been completed, they are not included in the Company's balance sheet.

# MALLINCKRODT PLC STATEMENT OF COMPREHENSIVE LOSS (UNAUDITED) For the period from January 11, 2013 (date of capitalization) to March 29, 2013 (in thousands of U.S. dollars)

Net sales	\$—
Selling general and administrative expenses	2
Net loss	(2)
Other comprehensive loss	
Comprehensive loss	<u>\$(2)</u>

See Notes to Unaudited Financial Statements.

# MALLINCKRODT PLC BALANCE SHEETS (UNAUDITED) At March 29, 2013 and January 11, 2013 (date of capitalization) (in thousands of U.S. dollars)

	ch 29, 013	2 (da	ary 11, 013 ate of lization)
Assets	 		
Current Assets:			
Cash	\$ 51	\$	53
Total Assets	\$ 51	\$	53
Commitments and contingencies (Note 3)			
Shareholders' Equity			
Ordinary A shares; €1.00 par value, 40,000 shares authorized, 40,000 shares issued and outstanding	\$ 53	\$	53
Ordinary shares; \$0.20 par value, 300,000 shares authorized, 7 shares issued and outstanding			—
Retained deficit	(2)		
Total Shareholders' Equity	\$ 51	\$	53

See Notes to Unaudited Financial Statements.

## MALLINCKRODT PLC STATEMENT OF RETAINED DEFICIT (UNAUDITED) For the period from January 11, 2013 (date of capitalization) to March 29, 2013 (in thousands of U.S. dollars)

	Retained Deficit
Balance at January 11, 2013	<u> </u>
Net loss	(2)
Balance at March 29, 2013	\$ (2)

See Notes to Unaudited Financial Statements.

## MALLINCKRODT PLC STATEMENT OF CASH FLOWS (UNAUDITED) For the period from January 11, 2013 (date of capitalization) to March 29, 2013 (in thousands of U.S. dollars)

Cash Flows From Operating Activities:	
Net loss	\$ (2)
Adjustments to reconcile net cash provided by continuing operations:	
Non-cash effect of currency rate changes	2
Net cash used in operating activities	—
Effect of currency rate changes on cash	(2)
Net decrease in cash	(2)
Cash at beginning of period	53
Cash at end of period	53 \$ 51

See Notes to Unaudited Financial Statements.

## MALLINCKRODT PLC NOTES TO UNAUDITED FINANCIAL STATEMENTS

#### 1. History and Description of the Company

Mallinckrodt plc (the "Company") was incorporated in Ireland, as a public limited company, on January 9, 2013 and capitalized on January 11, 2013 as a holding company for the purposes of being the parent company of Covidien plc's Pharmaceuticals business. The Company has not engaged in any business or other activities.

## 2. Basis of Presentation

#### **Basis of Presentation**

The accompanying financial statements reflect the operations of Mallinckrodt plc, a company incorporated in Ireland. The unaudited financial statements have been prepared in U.S. dollars, in accordance with accounting principles generally accepted in the U.S. ("GAAP"). In management's opinion, the unaudited financial statements contain all normal recurring adjustments necessary for a fair presentation of the interim results reported. The January 11, 2013 balance sheet data were derived from audited financial statements, but do not include all of the annual disclosures required by GAAP; accordingly, these financial statements should be read in conjunction with the Company's audited financial statements for the period ended January 11, 2013, which can be found in this information statement.

#### Subsequent Events

The Company has evaluated subsequent events for recognition or disclosure through May 31, 2013, the date the financial statements were available to be issued.

#### 3. Commitments and Contingencies

The Company is not currently a party to any loss contingencies or litigation.

#### Tax Matters

Upon separation, Covidien will transfer its interest in the businesses comprising its Pharmaceuticals segment to the Company. In addition, assets and liabilities related to certain non-operating activities of Covidien, primarily intercompany transactions will also be transferred to the Company; historically, these activities were not managed by the Pharmaceuticals segment. Most of these assets and liabilities are not significant. However, as measured as of March 29, 2013, the Company expects to assume a net income tax payable of approximately \$111 million, primarily consisting of non-current contingent tax liabilities associated with unresolved tax matters related to these non-operating activities. In addition, the Company expects to enter into a tax matters agreement which will allocate liability between the Company and Covidien for certain taxes associated with the distribution as well as certain current and historic taxes. The Company's liability pursuant to the tax matters agreement for certain historic taxes and certain taxes associated with the distribution (after taking into account certain tax benefits realized by the Company) will be subject to a \$200 million limitation. The decrease in the estimated net income tax payable, compared with the amount disclosed in the notes to the January 11, 2013 audited financial statements primarily resulted from a change in the determination of how certain historic tax liabilities will be allocated between the Company and Covidien pursuant to the tax matters agreement. In addition, the Company expects to assume a net deferred tax liabilities will be allocated between the Company and Covidien pursuant to the tax matters agreement. In addition, the Company expects to assume a net deferred tax liabilities will be allocated between the Company and Covidien pursuant to the tax matters agreement. In addition, the Company expects to assume a net deferred tax liabilities will be allocated between the Company and Covidien pursuant to the tax matters agreement. In addition, the

# ANNEX A

## Relevant Territories as of May 31, 2013

Albania Armenia Australia Austria Bahrain Belarus Belgium Bosnia & Herzegovina Bulgaria Canada Chile China Croatia Cyprus Czech Republic Denmark Egypt

Estonia Finland France Georgia Germany Greece Hong Kong Hungary Iceland India Israel Italy Japan Korea Kuwait Latvia Lithuania

Luxembourg Macedonia Malaysia Malta Mexico Moldova Montenegro Morocco Netherlands New Zealand Norway Pakistan Panama Poland Portugal Qatar Romania

Russia Saudi Arabia Serbia Singapore Slovak Republic Slovenia South Africa Spain Sweden Switzerland The Republic of Turkey Ukraine United Arab Emirates United Kingdom United States Uzbekistan Vietnam Zambia

# WACHTELL, LIPTON, ROSEN & KATZ

MARTIN LIPTON HERBERT M WACHTELL BERNARD W. NUSSBAUM LAWRENCE B. PEDOWITZ PAUL VIZCARRONDO, JR. PETER C. HEIN HAROLD S. NOVIKOFF MEYER G. KOPLOW THEODORE N. MIRVIS EDWARD D. HERLIHY DANIEL A. NEFF FRIC M ROTH ANDREW R. BROWNSTEIN MICHAEL H. BYOWITZ PAUL K. ROWE MARC WOLINSKY DAVID GRUENSTEIN STEPHEN G. GELLMAN STEVEN A. ROSENBLUM STEPHANIE J. SELIGMAN

JOHN F. SAVARESE SCOTT K CHARLES DAVID S. NEILL JODI J. SCHWARTZ ADAM O. EMMERICH GEORGE T. CONWAY III RALPH M LEVENE RICHARD G. MASON MICHAEL J. SEGAL DAVID M SILK ROBIN PANOVKA DAVID A KATZ ILENE KNABLE GOTTS DAVID M. MURPHY JEFFREY M. WINTNER TREVOR S. NORWITZ BEN M GERMANA ANDREW J. NUSSBAUM RACHELLE SILVERBERG STEVEN A. COHEN

#### 51 WEST 52ND STREET NEW YORK, N.Y. 10019-6150 TELEPHONE: (212) 403 - 1000 FACSIMILE: (212) 403 - 2000

GEORGE A. KATZ (1965-1989) JAMES H. FOGELSON (1967-1991)

OF COUNSEL

ROBERT M. MORGENTHAU ERIC S. ROBINSON PATRICIA A. ROBINSON\* LEONARD M. ROSEN MICHAEL W. SCHWARTZ ELLIOTT V. STEIN WARREN R. STERN PATRICIA A. VLAHAKIS J. BRYAN WHITWORTH AMY R. WOLF

PHILIP MINDLIN \* ADMITTED IN THE DISTRICT OF COLUMBIA

COUNSEL

DAVID M. ADLERSTEIN MICHELE J. ALEXANDER AMANDA K. ALLEXON\*\* LOUIS J. BARASH DIANNA CHEN ANDREW J.H. CHEUNG PAMELA EHRENKRANZ KATHRYN GETTLES-ATWA

WILLIAM T. ALLEN

PETER C. CANELLOS

DAVID M. EINHORN KENNETH B. FORREST

THEODORE GEWERTZ MAURA R. GROSSMAN

RICHARD D. KATCHER

THEODORE A. LEVINE

DOUGLAS K. MAYER

ROBERT B. MAZUR

PAULA N. GORDON NANCY B. GREENBAUM MARK A. KOENIG J. AUSTIN LYONS SABASTIAN V. NILES AMANDA N. PERSAUD JEFFREY A. WATIKER

\*\* ADMITTED IN THE STATE OF ILLINOIS

May 31, 2013

DEBORAH L. PAUL DAVID C. KARP RICHARD K. KIM JOSHUA R. CAMMAKER MARK GORDON JOSEPH D. LARSON LAWRENCE S. MAKOW JEANNEMARIE O'BRIEN WAYNE M. CARLIN STEPHEN R. DIPRIMA NICHOLAS G. DEMIMO IGOR KIRMAN JONATHAN M. MOSES T. EIKO STANGE DAVID A. SCHWARTZ JOHN F. LYNCH WILLIAM SAVITT ERIC M. ROSOF MARTIN J.E. ARMS GREGORY E. OSTLING DAVID B. ANDERS ADAM J. SHAPIRO NELSON O. FITTS JEREMY L. GOLDSTEIN JOSHUA M. HOLMES DAVID E. SHAPIRO DAMIAN G. DIDDEN ANTE VUCIC IAN BOCZKO MATTHEW M. GUEST DAVID E. KAHAN DAVID K. LAM BENJAMIN M. ROTH JOSHUA A. FELTMAN ELAINE P. GOLIN EMIL A. KLEINHAUS KARESSA L. CAIN RONALD C.CHEN GORDON S. MOODIE DONGJU SONG BRADLEY R. WILSON

#### VIA HAND DELIVERY AND EDGAR

Mr. Jeffrey P. Riedler Assistant Director U.S. Securities and Exchange Commission Division of Corporation Finance 100 F Street, N.E. Washington, D.C. 20549

> Re: Mallinckrodt plc Amendment No. 2 to Registration Statement on Form 10-12B Filed May 8, 2013 File No. 001-35803

Dear Mr. Riedler:

On behalf of our client, Mallinckrodt plc (the "<u>Company</u>"), we are providing the Company's responses to the comments of the Staff of the Division of Corporation Finance (the "<u>Staff</u>") of the U.S. Securities and Exchange Commission (the "<u>Commission</u>") set forth in your letter, dated May 20, 2013, with respect to the filing referenced above.

Jeffrey P. Riedler U.S. Securities and Exchange Commission May 31, 2013 Page 2 of 3

This letter and Amendment No. 3 ("<u>Amendment No. 3</u>") to the Registration Statement on Form 10 (File No. 001-35803) (the "<u>Registration Statement</u>") are being filed electronically via the EDGAR system today. In addition to the EDGAR filing, we are delivering a hard copy of this letter, along with six copies of Amendment No. 3 marked to indicate changes from Amendment No. 2 to the Registration Statement filed on May 8, 2013.

For the Staff's convenience, the text of the Staff's comments is set forth below in bold, followed in each case by the Company's response. Terms not otherwise defined in this letter shall have the meanings set forth in Amendment No. 3. All references to page numbers in these responses are to the pages of the information statement filed as Exhibit 99.1 (the "<u>Information Statement</u>") in the marked version of Amendment No. 3.

# <u>Exhibit 99.1</u>

# <u>General</u>

- 1. Please revise all financial statements and data to provide updated unaudited interim six months financial statements for both Mallinckrodt and the Pharmaceuticals business of Covidien plc as follows pursuant to Rule 3-12 of Regulation S-X:
  - Summary Historical and Unaudited Proforma Combined Financial Data
  - Unaudited Proforma Condensed Combined Financial Statements
  - Selected Historical Combined Financial Data
  - MD&A
  - Combined Financial Statements

Response: The Information Statement has been revised in response to the Staff's comment.

## Unaudited Pro Forma Condensed Combined Financial Statements, pages 45-50

2. As a continuing reminder for future amendments, please quantify your proforma adjustments, proforma earnings per share and proforma weighted-average shares outstanding. Please be sure to disclose the relevant assumptions in the proforma amounts quantified.

Response: The Information Statement has been revised in response to the Staff's comment.

\* \* \*

We hope that the foregoing, and the revisions set forth in Amendment No. 3, have been responsive to the Staff's comments. If you have any questions or comments regarding the foregoing, please do not hesitate to contact me at (212) 403-1005 or by email at VGoldfeld@wlrk.com.

Jeffrey P. Riedler U.S. Securities and Exchange Commission May 31, 2013 Page 3 of 3

Sincerely,

/s/ Victor Goldfeld

Victor Goldfeld

Enclosures

cc: Jack Kapples Vice President and Corporate Secretary Covidien plc

> Miriam Singer Vice President and Corporate Secretary, Pharmaceutical Products Covidien plc